

Mowry Village Project

Environmental Impact Report

August 2, 2023

Prepared for:

City of Newark, Community Development Department 37101 Newark Boulevard Newark, California 94560

Prepared by:

Stantec Consulting Services Inc, 1340 Treat Boulevard, Suite 300 Walnut Creek, California 94597 This page intentionally left blank.

Table of Contents

ACRC	ONYMS AND	DABBREVIATION	VII
EXEC	UTIVE SUN	IMARY	ES-1
ES.1	SUMMAR'	Y OF PROPOSED PROJECT	ES-1
	ES.1.1	Project Objectives	ES-1
	ES.1.2	Project Approvals	ES-1
	ES.1.3	Responsible and Trustee Agencies	ES-2
ES.2	AREAS O	F CONTROVERSY/ISSUES TO BE RESOLVED	ES-2
ES.3	ALTERNA	TIVES TO THE PROPOSED PROJECT	ES-3
ES.4	SUMMAR'	Y OF IMPACTS AND MITIGATION MEASURES	ES-4
ES.5	REVIEW (OF THE DRAFT EIR	ES-35
10		CTION	4.4
1.0			1-1 • •
1.1		RUNMENTAL REVIEW PROCESS	
	1.1.1	Purpose and Authority	1-1 1 0
	1.1.2	Load Agoney Determination	۲-۱ ۱ 2
10			1-2
1.2	300PE 0	F DRAFT EIR	I-Z 1 7
1 0			۲-۱۱۰۲
1.3			1-7
1.4		JF THE DRAFT EIR	
	1.4.1	Electively Commenting on an Elk	1-10 1 10
	1.4.2		1-10
2.0	PROJECT	DESCRIPTION	2-1
2.1	PROJECT	OVERVIEW	
	2.1.1	Project Site	2-1
	2.1.2	General Plan and Zoning	2-1
	2.1.3	Existing Site Conditions	2-7
	2.1.4	Existing Operations	2-13
	2.1.5	Surrounding Land Uses	2-13
	2.1.6	Residential Development	2-13
	2.1.7	Residential Open Space	2-13
	2.1.8		2-19
	2.1.9	Lanoscaping	
	2.1.10	Lightung	
	2.1.11	Alternative Transportation	
2.2			2-33 2 22
2.2		Construction Schedule and Equipment	
	2.2.1 222	Demolition and On-Site Soil Remediation	∠-33 2_33
	223	Grading Fill and Drainage	2 <u>.</u> 35
23		OBJECTIVES AND REOURED PROJECT APPROVALS	2.25 2_25
2.0	231	Objectives	2-35 2 <u>-</u> 35
	2.3.2	Approvals	



3.0	ENVIRON	IMENTAL IMPACT ANALYSIS	3-1
3.1	AESTHET	۲ICS	3-5
	3.1.1	Environmental Setting	3-5
	3.1.2	Regulatory Setting	3-6
	3.1.3	Environmental Impacts	3-8
3.2	AGRICUL	TURAL AND FORESTRY RESOURCES	
•	3.2.1	Environmental Setting	
	3.2.2	Regulatory Setting	3-13
	3.2.3	Environmental Impacts	3-14
3.3	AIR QUAL	_ITY	
	3.3.1	Environmental Setting	
	3.3.2	Regulatory Setting	
	3.3.3	Environmental Impacts	3-30
3.4	BIOLOGI		
	3.4.1	Environmental Setting	
	3.4.2	Regulatory Setting	3-61
	3.4.3	Environmental Impacts	3-68
3.5	CULTURA	AL RESOURCES	
	3.5.1	Environmental Setting	
	3.5.2	Regulatory Setting	3-94
	3.5.3	Environmental Impacts	3-96
3.6	ENERGY	·	3-103
	3.6.1	Environmental Setting	3-103
	3.6.2	Regulatory Setting	3-103
	3.6.3	Environmental Impacts	3-106
3.7	GEOLOG	Y AND SOILS	3-111
	3.7.1	Environmental Setting	3-111
	3.7.2	Regulatory Setting	3-114
	3.7.3	Environmental Impacts	3-116
3.8	GREENH	OUSE GAS EMISSIONS	3-127
	3.8.1	Environmental Setting	3-127
	3.8.2	Regulatory Setting	3-129
	3.8.3	Environmental Impacts	3-136
3.9	HAZARDS	S AND HAZARDOUS MATERIALS	3-147
	3.9.1	Environmental Setting	3-147
	3.9.2	Regulatory Setting	3-150
	3.9.3	Environmental Impacts	3-155
3.10	HYDROL	OGY AND WATER QUALITY	3-163
	3.10.1	Environmental Setting	3-163
	3.10.2	Regulatory Setting	3-165
	3.10.3	Environmental Impacts	3-169
3.11	LAND US	E AND PLANNING	3-181
	3.11.1	Environmental Setting	3-181
	3.11.2	Regulatory Setting.	3-183
	3.11.3	Environmental Impacts	3-186
3.12	MINERAL	RESOURCES	3-201



	3.12.1	Environmental Setting	3-201
	3.12.2	Regulatory Setting	3-201
	3.12.3	Environmental Impacts	3-202
3.13	NOISE		3-205
	3.13.1	Environmental Setting	3-205
	3.13.2	Regulatory Setting	3-211
	3.13.3	Environmental Impacts	3-221
3.14	POPULAT	ION AND HOUSING	3-235
	3.14.1	Environmental Setting	3-235
	3.14.2	Regulatory Setting	3-238
	3.14.3	Environmental Impacts	3-239
3.15	PUBLIC S		3-243
	3.15.1	Environmental Setting	3-243
	3.15.2	Regulatory Setting	3-244
	3.15.3	Environmental Impacts	3-246
3.16	RECREAT	۲ION	3-251
	3.16.1	Environmental Setting	3-251
	3.16.2	Regulatory Setting	3-251
	3.16.3	Environmental Impacts	3-252
3.17	TRANSPO		3-255
	3.17.1	Environmental Setting	3-255
	3.17.2	Regulatory Setting.	3-257
	3.17.3	Environmental Impacts	3-261
3.18	TRIBAL C	ULTURAL RESOURCES	3-275
	3.18.1	Environmental Setting	3-275
	3.18.2	Regulatory Setting.	3-276
	3.18.3	Environmental Impacts	3-278
3.19	UTILITIES	AND SERVICE SYSTEMS	3-281
	3.19.1	Environmental Setting	3-281
	3.19.2	Regulatory Setting	3-283
	3.19.3	Environmental Impacts	3-286
3.20	WILDFIRE		3-295
	3.20.1	Environmental Setting	3-295
	3.20.2	Regulatory Setting	3-295
	3.20.3	Environmental Impacts	3-297
		·	
4.0	CUMULA	TIVE EFFECTS	4-1
4.1	INTRODU	CTION	4-1
4.2	CUMULA	ΓΙVE IMPACT SETTING	4-1
4.3	GEOGRA	PHIC SCOPE	4-2
4.4	LIST OF F	RELATED PLANS AND PROJECTS	4-3
4.5	CUMULA	TIVE IMPACT ANALYSIS	4-9
	4.5.1	Aesthetics	
	4.5.2	Agricultural and Forestry Resources	4-11
			4 4 4
	4.5.3	Air Quality	4-11
	4.5.3 4.5.4	Air Quality Biological Resources	4-11



	4.5.6	Energy	4-13
	4.5.7	Geology and Soils	4-13
	4.5.8	Greenhouse Gas Emissions and Climate Change	4-13
	4.5.9	Hazards and Hazardous Materials	4-14
	4.5.10	Hydrology and Water Quality	4-14
	4.5.11	Land Use and Planning	4-14
	4.5.12	Mineral Resources	4-15
	4.5.13	Noise and Vibration	4-15
	4.5.14	Population and Housing	4-16
	4.5.15	Public Services	4-16
	4.5.16	Recreation	4-18
	4.5.17	Transportation	4-18
	4.5.18	Tribal Cultural Resources	4-19
	4.5.19	Utilities and Service Systems	4-19
	4.5.20	Wildfire	4-20
50			5-1
5.1		CTION	5-1
5.2	REOURE	MENTS FOR THE CONSIDERATION OF ALTERNATIVES	5_1
0.2	521	No Project Alternative	5_2
	522	Consistency with Project Objectives	5-2 5_2
	523	Feasibility	
	524	Potential to Avoid or Lessen Significant Environmental Effects	5-3
53		DI OGY AND SCREENING CRITERIA	5-4
5.0			
5.4			5 /
		Alternative Leastion Alternative	
	5.4.1 5.4.2		
	5.4.Z	Small Drojoct Altornativo	
5 5			5-5 5 6
5.5		No Project Alternative	
	5.5.1	Alternative 2 Multi family Residential Alternative	5 10
	5.5.2	Alternative 3 – Reduced Density Alternative	5_16
	554	Alternative $4 - 100$ Percent Affordable Housing Alternative	5_22
56		MENTALLY SUPERIOR ALTERNATIVE	5_28
0.0			
6.0	OTHER C	EQA CONSIDERATIONS	6-1
6.1	GROWTH	-INDUCING IMPACTS	6-1
	6.1.1	Population Growth	6-2
	6.1.2	Removal of Barrier to Growth	6-2
6.2	SIGNIFIC	ANT AND UNAVOIDABLE IMPACTS	6-3
	6.2.1	Transportation	6-3
6.3	SIGNIFIC	ANT IRREVERSIBLE ENVIRONMENTAL CHANGES	6-3
6.4	MANDAT	DRY FINDINGS OF SIGNIFICANCE	6-5
70			74
1.0			



8.0	REFERENCES	8-'	1
		-	-

LIST OF TABLES

Table ES-1: Executive Summary of Impacts and Mitigation MeasuresE	ES-7
Table 1.2-1: Comments Received on the NOP	. 1-3
Table 3.0-1: Environmental Resource Abbreviations	. 3-3
Table 3.3-1: Hayward-La Mesa Air Quality Monitoring Station Data	3-21
Table 3.3-2. California and National Ambient Air Quality Standards	3-23
Table 3.3-3. San Francisco Bay Area Air Basin Designations for State and National	
Ambient Air Quality	3-24
Table 3.3-4. Anticipated Construction Schedule	3-30
Table 3.3-5. Construction Equipment Assumptions	3-31
Table 3.3-6: BAAQMD Air Quality CEQA Thresholds of Significance	3-35
Table 3.3-7: Unmitigated Remediation and Construction Criteria Pollutant and Precursor	
Emissions	3-41
Table 3.3-8: Mitigated Remediation and Construction Criteria Pollutant and Precursor	
Emissions	3-42
Table 3.3-9: Existing Use Criteria Pollutant and Precursor Emissions	3-43
Table 3.3-10: Operational Criteria Pollutant and Precursor Emissions	3-43
Table 3.3-11: Summary of the Cumulative Health Impacts at the Project Site at Project	
Buildout	3-47
Table 3.3-12: Screening Levels for Potential Odor Sources	3-49
Table 3.5-1: Previously Documented Resources within the Study Area	3-93
Table 3.8-1: Construction Greenhouse Gas Emissions	-139
Table 3.8-2: Operational Greenhouse Gas Emissions	-140
Table 3.8-3: Newark Climate Action Plan Consistency Analysis	-141
Table 3.8-4: SB 32 Scoping Plan Consistency Analysis	-142
Table 3.8-5: Project Consistency with 2022 Scoping Plan Greenhouse Gas Reduction	
Strategies	-143
Table 3.11-1: General Plan Consistency Analysis	·188
Table 3.11-2: Newark Areas 3 and 4 Specific Plan Consistency Analysis	·196
Table 3.13-1: Typical A-Weighted Sound Levels	·205
Table 3.13-2: Definition of Sound Measurements	-206
Table 3.13-3: Guideline Vibration Annoyance Potential Criteria	-208
Table 3.13-4: Guideline Vibration Damage Potential Criteria	-208
Table 3.13-5: Vibration Source Levels for Construction Equipment	-209
Table 3.13-6: Short-Term Noise Measurement Results	-211
Table 3.13-7: EPA Impact Guidelines	-222
Table 3.13-8: Traffic Peak Hour Counts and Estimated Noise Increase - Existing	-223
Table 3.13-9: Traffic Peak Hour Counts and Estimated Noise Increase - 2040	-223
Table 3.13-10: Construction Stage Equipment	-227
Table 3.13-11: Federal Highway Administration Roadway Construction Noise Model	
Source Noise Levels	-228
Table 3.13-12: Calculated Noise Level from Each Construction Stage	-228
Table 3.13-13: Vibration Source Levels for Construction Equipment	232
Table 3.14-1: Newark Historic Population Growth	-235
Table 3.14-2: ABAG Projections 2040	-236

Table 3.14-3: Newark Historic Housing Units Growth	3-236
Table 3.14-4: Housing Need Allocation	3-237
Table 4.3-1: Geographic Scope of Cumulative Impact and Method of Evaluation	4-2
Table 4.4-1: List of Related Projects	4-7
Table 5.6-1: Project Alternative Impacts Comparison	5-29
Table 5.6-2: Project Alternatives Comparison to Project Objectives	5-31

LIST OF FIGURES

Figure 2-1: Regional Location	2-3
Figure 2-2: Project Site	
Figure 2-3: Existing Zoning	
Figure 2-4: Proposed Zoning	2-11
Figure 2-5: Project Site Plan	2-15
Figure 2-6: Proposed Elevations	2-17
Figure 2-7: Mowry Avenue Roadway Improvements	2-21
Figure 2-8: Landscape Plan	2-23
Figure 2-9: Water Line Improvements	2-27
Figure 2-10: Stormwater Improvements	
Figure 3-1: Biological Resources Study Area	3-53
Figure 3-2: Biological Resources Habitat Map	3-57
Figure 3-3: Aquatic Resources Delineation Map	3-59
Figure 3-4: Area of Potential Effects	3-91
Figure 3-5: City of Newark Noise Compatibility Guidelines	3-215
Figure 4-1: Related Projects	4-5

LIST OF APPENDICES

- Appendix B CalEEMod and Energy Calculations
- Appendix C Biological Resources Technical Report
- Appendix D Cultural Resources Assessment
- Appendix E Due Diligence and Design Level Geotechnical Reports
- Appendix F Phase I and II Environmental Site Assessment
- Appendix G Noise Modeling
- Appendix H Transportation Impact Analysis Report

Draft Environmental Impact Report

Acronyms and Abbreviation

AB	Assembly Bill
ABAG	Association of Bay Area Governments
AC Transit	Alameda-Contra Costa (AC) Transit
ACBM	Asbestos-Containing Building Materials
ACC	Advanced Clean Cars
ACE	Altamont Commuter Express
ACFC&WCD	Alameda County Flood Control and Water Conservation District
ACFD	Alameda County Fire Department
ACWD	Alameda County Water District
ADA	Americans with Disabilities Act
ADFW	average dry weather flow
ADT	average daily trip
AEP	Association of Environmental Professionals
AFY	acre-feet per year
AICUZ	Air Installation Compatible Land Use Zone
AMSL	above mean sea level
ANSI	America National Standards Institute
APE	Area of Potential Effect
APN	Assessor's Parcel Number
Applicant	Mowry Project Owner, LLC
AQP	Air Quality Plan
AWWTP	Alvarado Wastewater Treatment Plan
BAAQMD	Bay Area Air Quality Management District
BART	Bay Area Rapid Transit
Basin Plan	San Francisco Bay Basin Water Quality Control Plan
BCDC	San Francisco Bay Conservation and Development Commission
Bgs	below ground surface
BMP	best management practice
BTEX	benzene, toluene, ethylbenzene, and xylenes
CAA	Clean Air Act
CAAQS	California Ambient Air Quality Standards
CalEEMod	California Emissions Estimator Model
Cal-IPC	California Invasive Plant Council
CAL EMA	California Emergency Management Agency
Cal EPA	California Environmental Protection Agency
CAL FIRE	California Department of Forestry and Fire Protection
Cal OES	Governor's Office of Emergency Services
CALGreen	California Green Building Standards Code



CalOSHA	California Division of Occupational Safety and Health Administration
CalRecycle	California Department of Resources Recycling and Recovery
Caltrans	California Department of Transportation
CAP	Climate Action Plan
CAPCOA	California Air Pollution Control Officers Association
CARB	California Air Resources Board
CBC	California Building Code
CCAA	California Clean Air Act
CCR	California Code of Regulations
CDFW	California Department of Fish and Wildlife
CEC	California Energy Commission
CEQA	California Environmental Quality Act
CESA	California Endangered Species Act
CFC	California Fire Code
CFGC	California Fish and Game Code
CFR	Code of Federal Regulations
CHL	California Historical Landmark
CH ₄	methane
City	City of Newark
CNDDB	California Natural Diversity Database
CNEL	community noise equivalent level
CNPS	California Native Plant Society
со	carbon monoxide
CO ₂	carbon dioxide
CPUC	California Public Utilities Commission
CREC	Controlled Recognized Environmental Conditions
CRHR	California Register of Historical Resources
CRPR	California Rare Plant Rank
СТС	County Transportation Commission
CUPA	Certified Unified Program Agencies
CWA	Clean Water Act
CY	cubic yards
dB	decibel
dB(A)	A-weighted decibel
DBH	diameter at breast height
DOC	California Department of Conservation
DOF	Department of Finance
DPM	diesel particulate matter
DTSC	Department of Toxic Substances Control
DWR	California Department of Water Resources
EDD	Employment Development Department
EIR	Environmental Impact Report



EO	Executive Order
ESA	Environmental Site Assessment
ESL	Environmental Screening Level
FEMA	Federal Emergency Management Agency
FERC	Federal Energy Regulatory Commission
FESA	Federal Endangered Species Act
FHWA	Federal Highway Administration
FMMP	Farmland Mapping and Monitoring Program
FRA	Federal Railroad Administration
FTA	Federal Transit Administration
GC	Government Code
GHG	greenhouse gas
GIS	geographic information system
GSA	Groundwater Sustainability Agency
GSP	Groundwater Sustainability Plan
GWP	global warming potential
НАР	hazardous air pollutant
HCD	Department of Housing and Community Development
HCP	Habitat Conservation Plan
Helix	Helix Environmental Planning, Inc.
HFC	hydrofluorocarbons
НОА	Homeowner's Association
HOV	High Occupancy Vehicle
HREC	Historical Recognized Environmental Conditions
HSC	Health and Safety Code
Hz	Hertz
IPaC	Information for Planning and Consultation
I-880	Interstate 880
LAFCO	Alameda County Local Agency Formation Commission
lbs/day	pounds per day
LCFS	Low Carbon Fuel Standard
Ldn	day-night sound level
Leq	equivalent sound level
LID	Low Impact Development
Lmax	maximum sound level
Lmin	minimum sound level
LOS	level of service
LTA	local responsibility area
Lxx	percentile-exceeding sound level
MACT	Maximum Achievable Control Technologies
МВТА	Migratory Bird Treaty Act
MEIR	maximally exposed individual receptor



Mgd	million gallons per day
mg/m ³	milligrams per cubic meter
MTCO2e	Metric tons of carbon dioxide equivalent
MMTCO2e	Million metric tons of carbon dioxide equivalent
Mph	miles per hour
MPO	Metropolitan Planning Organization
ММТ	million metric tons
MRZ	Mineral Resource Zone
MTC	Metropolitan Transportation Commission
MWh	megawatt-hour
NAAQS	national ambient air quality standards
NCCP	Natural Community Conservation Planning
NDF	Newark Desalination Facility
NEHRP	National Earthquake Hazards Reduction Program
NEHRPA	National Earthquake Hazards Reduction Program Act
NESHAP	National Emissions Standards for Hazardous Air Pollutants
NF ₃	nitrogen trifluoride
NGVD	National Geodetic Vertical Datum
NHTSA	National Highway Traffic Safety Administration
NMFS	National Marine Fisheries Service
NO	nitric oxide
NOC	Notice of Completion
NOI	Notice of Intent
NOP	Notice of Preparation
NOx	nitrogen oxide
NO ₂	nitrogen oxide
NPD	Newark Police Department
NPDES	National Pollutant Discharge Elimination System
NPPA	California Native Plant Protection Act
NRCS	Natural Resources Conservation Service
NRHP	National Register of Historic Places
NUSD	Newark Unified School District
NWIC	Northwest Information Center
N ₂ O	nitrous oxide
OCP	organochloride pesticides
OEHAA	California Office of Environmental Health Hazard Assessment
OHP	Office of Historic Preservation
OITC	Outside-Inside Transmission Class
OPR	Governor's Office of Planning and Research
ORC	oxygen release compound
O ₃	Ozone
РАН	polycyclic aromatic hydrocarbons



Pb	lead
PBMP	Pedestrian and Bicycle Master Plan
PD-RS-6000	Residential Single-Family with Planned Unit Development Overlay
PFC	perfluorocarbons
PG&E	Pacific Gas and Electric
PI	Plasticity Index
PM _{2.5}	particulate matter 2.5 microns or less in diameter
PM10	particulate matter ten microns or less in diameter
POTW	publicly owned treatment works
Ppm	parts per million
PPC	peak particle velocity
PRC	Public Resources Code
RCNM	Roadway Construction Noise Model
RCRA	Resource Conservation and Recovery Act
REC	Recognized Environmental Condition
RHNA	Regional Housing Needs Allocation
ROC	Reactive organic compounds
ROG	Reactive Organic Gas
ROW	Right of Way
RPC-SAT	California Ocean Protection Council Science Advisory Team
RRFB	Rectangular Rapid Flashing Beacon
RSP	RSP Groups, Inc.
RTP/SCS	regional transportation plan/sustainable communities strategy
RWQCB	Regional Water Quality Control Board
RWS	Regional Water System
SAF	San Andreas Fault
SAFE	Safer Affordable Fuel Efficient
SB	Senate Bill
SFBAAB	San Francisco Bay Area Air Basin
SF ₆	sulfur hexafluoride
SGMA	Sustainable Groundwater Management Act
SIP	State Implementation Plan
SLCP	Short-Lived Climate Pollutant
SLF	Sacred Lands File
SMARA	California Surface Mining and Reclamation Act
SO ₂	sulfur dioxide



SO4	sulfates
SR	State Route
SSA	Streambed Alteration Agreement
SSC	Species of Special Concern
Stantec	Stantec Consulting Services Inc.
STC	Sound Transmission Class
SWMP	Stormwater Management Plan
SWP	State Water Project
SWPPP	Stormwater Pollution Prevention Plan
SWRCB	State Water Resources Control Board
TAC	toxic air contaminants
TAZ	Traffic Analysis Zone
ТСР	Traffic Control Plan
TDM	Transportation Demand Management
TIA	Transportation Impact Analysis
ТРН	petroleum hydrocarbons
TPHd	petroleum hydrocarbons as diesel
TPHg	petroleum hydrocarbons as gasoline
TPHmo	petroleum hydrocarbons as motor oil
TPY	tons per year
UPRR	Union Pacific Railroad
USACE	U.S. Army Corps of Engineers
USD	Union Sanitary District
USEPA	U.S. Environmental Protection Agency
USFS	U.S. Forest Service
USFWS	U.S. Fish and Wildlife Service
USGS	United States Geological Survey
UWMP	Urban Water Management Plan
VMT	Vehicle Miles Traveled
VOC	volatile organic compounds
WDR	Waste Discharge Requirements
WSA	Water Supply Assessment
WTP	Water Treatment Plant
ZEV	zero-emission vehicle
μg/m³	micrograms per liter



EXECUTIVE SUMMARY

This document is a Draft Environmental Impact Report (EIR) for the Mowry Village Project (proposed project). This section of the Draft EIR provides a summary of the proposed project, the anticipated environmental impacts of the proposed project, the alternatives, and areas of known controversy to be resolved.

ES.1 SUMMARY OF PROPOSED PROJECT

The proposed project is located in the City of Newark (City) in southwestern Alameda County, California, southwest of the intersection of Mowry Avenue and the Union Pacific Railroad (UPRR) tracks, west of Cherry Street. The project site is approximately 29 acres and consists of three parcels identified as Assessor's Parcel Numbers (APNs) 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00. The proposed project involves the demolition of existing structures located on the project site and remediation of the site to construct 203 single-family detached homes. The proposed project would include additional on- and off-site improvements including, but not limited to, construction of on-street parking, drive aisles, utility improvements, landscaping and widening of Mowry Avenue.

A detailed description of the proposed project is provided in Section 2.0, Project Description.

ES.1.1 Project Objectives

The primary objective of the proposed project is to provide low density residential housing that incorporates multi-modal transportation for the future residents of Newark. Specific project objectives include the following:

- Implement the City's General Plan by developing the site with low-density residential;
- Support the City in meeting its Regional Housing Needs Allocation (RHNA) target assigned by the Association of Bay Area Governments (ABAG);
- Provide high quality residential development including a mix of lot sizes;
- Minimize environmental impacts associated with residential development by siting the project on developed and disturbed lands;
- Remediate contaminated soil on-site to levels suitable for residential development;
- Create a residential development that integrates multi-modal transportation (pedestrian, bicycle, automobile) and connects the development to existing, nearby bus transit stops and active centers by improving Mowry Avenue and upgrading the at-grade vehicular and pedestrian crossing along Mowry Avenue at the UPRR railroad tracks to increase safety.

ES.1.2 Project Approvals

The proposed project requires, but may not be limited to, the following approvals from the City:



Draft Environmental Impact Report Executive Summary

- Rezone from Park to RS-6000: Residential Single-Family with Planned Unit Development Overlay
- Planned Unit Development
- Specific Plan Amendment
- Vesting Tentative Map
- Design Review
- Grading, Building, and Encroachment Permits
- Alameda County Local Agency Formation Commission (LAFCO) Annexation Approval

ES.1.3 Responsible and Trustee Agencies

Under the California Environmental Quality Act (CEQA), a responsible agency is a public agency, other than the lead agency, that has responsibility to carry out or approve a project (Public Resources Code [PRC] Section 21069). A trustee agency is a state agency that has jurisdiction by law over natural resources that are held in trust for the people of the State of California (PRC Section 21070).

The following agencies may serve as responsible and/or trustee agencies:

- California Department of Transportation, District 4
- State Water Resources Control Board
- Regional Water Quality Control Board #2
- Bay Area Air Quality Management
- Alameda County LAFCO
- Alameda County Water District (ACWD)
- Union Sanitary District (USD)

ES.2 AREAS OF CONTROVERSY/ISSUES TO BE RESOLVED

Section 15123 of the State CEQA Guidelines requires that a summary of an EIR identify areas of controversy known to the lead agency, including issues raised by agencies and the public. The City distributed a Notice of Preparation (NOP) of the Draft EIR for the proposed project beginning on November 30, 2021. The NOP was circulated for a 30-day public review and comment period, ending on January 3, 2022. 11 commentors submitted written responses to the NOP. Seven written responses were from interested individuals or organizations and four written comments were from agencies including the Alameda County Transportation Commission, Native American Heritage Commission, Alameda County Water District, and Union Sanitary. An additional three comments were received orally from interested individuals and organizations during the Public Scoping Meeting held on December 14, 2021. The NOP



Draft Environmental Impact Report Executive Summary

and written comments received are included in Appendix A. Comments in response to the NOP generally identified the following areas of potential concern.

- Sea level rise
- Circulation system safety
- Hazards such as contaminated soils and unstable soil conditions
- Increased demand on infrastructure and public services

This Draft EIR contains substantial evidence to support the conclusions presented herein. It is possible that there will be disagreements among various parties regarding these conclusions, although the City of Newark is not aware of any disputed conclusions at the time of this writing. Both the CEQA Guidelines and case law clearly provide the standards for treating disagreement among experts. Where evidence and opinions conflict on an issue concerning the environment, and the lead agency knows of these controversies in advance, the EIR must acknowledge the controversies, summarize the conflicting opinions of the experts, and include sufficient information to allow the public and decision-makers to make an informed judgement about the environmental consequences of the proposed project.

ES.3 ALTERNATIVES TO THE PROPOSED PROJECT

The project alternatives and their potential impacts are discussed in Section 5.0, Alternatives to the Proposed Project, of this Draft EIR. As authorized under CEQA, the alternatives are discussed in less detail than the proposed project.

No Project Alternative

CEQA Guidelines Section 15126.6(e)(1) requires that the no project alternative be described and analyzed, "to allow decision-makers to compare the impacts of approving the proposed project with the impacts of not approving the proposed project." The no project analysis is required to discuss, "the existing conditions at the time the Notice of Preparation is published . . . as well as what would be reasonably expected to occur in the foreseeable future if the project were not approved, based on current plans and consistent with available infrastructure and community services" (Section 15126.6(e)(2)).

The No Project Alternative assumes that no additional development would occur on the project site and would continue to use the existing structures on the project site for commercial services.

While the no project alternative would avoid the significant and unavoidable impact to transportation, specifically Vehicle Miles Traveled (VMT), it would not meet any of the project objectives, including providing housing in the City.

Multi-family Residential Alternative (Alternative 2)

The Multi-family Residential Alternative would construct 405 multi-family residential units on the 29 acre project site, resulting in a density of approximately 14 units per acre. This alternative would require modifications to the site plan and layout and would require a General Plan Amendment from low density residential to medium density residential to increase density on-site. This alternative would include the



Draft Environmental Impact Report Executive Summary

demolition of existing uses and remediation of the existing groundwater and soil contamination on-site and would include on- and off-site utility and circulation improvements. This alternative would provide multi-family residential units that range from garden apartments and condominiums to townhomes and row houses that would have a maximum building height of 60 feet, in accordance with the development standards of the Areas 3 and 4 Specific Plan. The Multi-family Residential Alternative would not reduce the significant and unavoidable impact to VMT.

Reduced Density Alternative (Alternative 3)

The Reduced Density Alternative would develop the 29 acre project site with 64 single-family detached residential homes. The Reduced Density Alternative would have a resulting density of approximately two units per acre which is within the allowable density of 8.7 units per acre for the low density residential land use designation. This alternative would also include the demolition of existing uses and structure on-site, remediation of the existing groundwater and soil contamination on-site and would include on- and off-site utility and circulation improvements. Though the Reduced Density Alternative's density of two units per acre would be within the allowable density for low density residential General Plan land use designation, it would underutilize the allowable density of the designation of the project site. The Reduced Density Alternative would not reduce the significant and unavoidable impact to VMT.

100 Percent Affordable Housing Alternative (Alternative 4)

The 100 Percent Affordable Housing Alternative would develop the 29 acre project site with 405 multifamily residential units, similar to the Multi-family Residential Alternative. However, under this alternative, 100 percent of the residential units developed would be provided as affordable housing. This alternative would also include the demolition of existing uses and structure on-site, remediation of the existing groundwater and soil contamination on-site and would include on- and off-site utility and circulation improvements. This alternative would result in a density of approximately 14 units per acre and therefore, would require a General Plan Amendment from low density residential to medium density residential to allow for the increased density. This alternative assumed that no density bonus law would be applied. This alternative would provide multi-family residential units that range from garden apartments and condominiums to townhomes and row houses that would have a maximum building height of 60 feet, in accordance with the development standards of the Areas 3 and 4 Specific Plan. The 100 Percent Affordable Housing Alternative would not reduce the significant and unavoidable impact to VMT.

ES.4 SUMMARY OF IMPACTS AND MITIGATION MEASURES

Table ES-1, Executive Summary of Impacts and Mitigation Measures, summarizes the potential environmental effects of the proposed project, the recommended mitigation measures, if applicable, and the level of significance after mitigation. Per CEQA Section 15093, should the project be approved as proposed, any impact noted in the summary as "significant" after mitigation would require the adoption of a statement of overriding considerations. As shown in Table ES-1, development of the proposed project would result in significant and unavoidable impacts. Therefore, a statement of overriding considerations would be required.

Additionally, CEQA requires public agencies to establish a mitigation monitoring and reporting program for the purpose of ensuring compliance with those mitigation measures identified in an EIR and/or



Draft Environmental Impact Report Executive Summary

adopted as conditions of approval in order to mitigate or avoid significant environmental impacts identified in an EIR. A mitigation monitoring and reporting program, incorporating the mitigation measures set forth in this document, will be adopted at the time of certification of the Final EIR.

Draft Environmental Impact Report Executive Summary

This page intentionally left blank.

Draft Environmental Impact Report Executive Summary

Table ES-1: Executive Summary of Impacts and Mitigation Measures

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
Section 3.3 Air Quality			
Section 3.3 Air Quality Impact AIR-1: The proposed project could conflict with or obstruct implementation of the applicable air quality plan.	Potentially Significant Impact	 MM AIR-1: Implement Construction Best Management Practices. The applicant shall require all construction contractors to implement the basic construction mitigation measures recommended by the BAAQMD to reduce fugitive dust emissions. Emission reduction measures will include, at a minimum, the following measures. Additional measures may be identified by the BAAQMD or contractor as appropriate: All exposed surfaces (e.g., parking areas, staging areas, soil piles, graded areas, and unpaved access roads) will be watered two times per day. All haul trucks transporting soil, sand, or other loose material off-site will be covered. All visible mud or dirt track-out onto adjacent public roads will be removed using wet power vacuum street sweepers at least once per day. The use of dry power sweeping is prohibited. All vehicle speeds on unpaved roads shall be limited to 15 miles per hour. All roadways, driveways, and sidewalks to be paved will be completed as soon as possible. Building pads will be laid as soon as possible after grading unless seeding or soil binders are used. Idling times shall be minimized either by shutting equipment off when not in use or by reducing the maximum idling time to 5 minutes (as required by the California Airborne Toxics Control Measure Title 13, Section 2485 of California Code of Regulations; clear signage shall be provided for construction workers at all entered by the provided for construction workers at all entered by the provided for construction workers at all 	Less Than Significant Impact with Mitigation
		 All construction equipment shall be maintained and properly tuned in accordance with manufacturer's specifications. 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 All equipment shall be checked by a certified visible emissions evaluator or checked by a certified mechanic and determined to be running in proper condition prior to operation. 	
		 Post a publicly visible sign with the telephone number and person to contact at the City regarding dust complaints. This person will respond and take corrective action within 48 hours. The Bay Area Air Quality Management District's phone number will also be visible to ensure compliance with applicable regulations. 	
		MM AIR-2: Tier 4 Certified Construction Equipment. The project applicant or designated contractor shall specify, and the City shall verify that all grading, building, and other construction permits for the project, include the following requirement: All diesel-powered off-road equipment used for project remediation and construction activities shall be U.S. EPA Tier 4 certified, or have CARB approved engine retrofit kits certified to have emissions equivalent to Tier 4 standards.	
Impact AIR-2: The proposed project could result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or State ambient air quality standard.	Potentially Significant Impact	Mitigation Measures AIR-1 and AIR-2 are required.	Less Than Significant Impact with Mitigation
Impact AIR-3: The proposed project could expose sensitive receptors to substantial pollutant concentrations.	Potentially Significant Impact	Mitigation Measure AIR-1 and HAZ-1 are required.	Less Than Significant Impact with Mitigation
Section 3.4 Biological Resources	•		
Impact BIO-1: The proposed project could have a substantial adverse effect, either directly or through habitat modifications on any species in local or regional plans, policies, or regulations, or by the California Department of Fish and Wildlife or U.S. Fish and Wildlife Service.	Potentially Significant Impact	 MM BIO-1: Standard Construction Best Management Practices. The applicant shall require implementation of standard construction BMPs outlined, but not limited to, throughout construction. The project limit shall be delineated and/or fenced prior to the start of construction. 	Less Than Significant Impact with Mitigation



Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 Appropriate erosion control measures shall be used (e.g., hay bales, filter fences, vegetative buffer strips or other accepted equivalents) to reduce siltation and contaminated runoff from leaving the proposed project site. The integrity and effectiveness of the BMPs shall be inspected daily by the resident engineer. Corrective actions and repairs shall be carried out immediately. Construction by-products and pollutants such as petroleum products, chemicals, or other deleterious materials shall not be allowed into off-site wetlands or marsh habitats. A plan for the emergency clean-up of any spills of fuel or other materials shall be available when construction equipment is in use. Construction vehicles and equipment shall be maintained to prevent contamination of soil or water from external grease and oil or from leaking hydraulic fluid, fuel, oil, and grease. Leaking vehicles and equipment shall be removed from the site. Building materials storage areas containing hazardous or potentially toxic materials such as herbicides and petroleum products shall have an impermeable membrane between the ground and the hazardous material and shall be bermed to prevent the discharge of pollutants to ground water and runoff water. Equipment shall be re-fueled and serviced at designated construction staging areas, a minimum of 100 feet from any aquatic resource. All construction material and fill shall be stored and contained in a designated area that is located away from aquatic habitats to prevent transport of materials into adjacent water bodies. 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 A silt fence shall be installed to collect any discharge, and adequate materials shall be available for spill clean-up and during storm events. 	
		 No litter, debris, or side cast shall be dumped or permitted to enter wetlands or marsh habitats. During project activities, all trash that may attract predators shall be properly contained, removed from the work site, and disposed of regularly. Following construction, all trash and construction debris shall be removed from work areas. 	
		MM BIO-2: Environmental Awareness. Prior to initial ground disturbance, all project personnel shall attend an environmental awareness and compliance training. The training program shall present applicable environmental regulations and permit conditions. The training program shall include applicable measures established for the proposed project to minimize impacts to avoid sensitive resources, habitats, and species. Subsequent training events shall be scheduled to support the training of new personnel. Dated sign-in sheets for attendees at these meetings shall be maintained and submitted to the City prior to the issuance of grading permit.	
		MM BIO-3: Pre-Construction Nesting Bird Survey. If project activities such as vegetation removal activities commence during the avian breeding season (February 1 through August 31), a qualified biologist shall conduct a pre-construction nesting bird survey no more than 14 days prior to initiation of project activities. The survey area should include suitable raptor nesting habitat within 500 feet of the project boundary (inaccessible areas outside of the project parcels can be surveyed from the parcel or from public roads using binoculars or spotting scopes). Pre-construction surveys are not required in areas where project activities have been continuous since prior to February 1, as determined by a qualified biologist. Areas that have been inactive for more than 14 days during the avian breeding season must be re-surveyed prior to	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		resumption of project activities. If no active nests are identified, no further mitigation is required. If active nests are identified, the following measure shall be implemented:	
		 A suitable buffer (e.g., 300 to 500 feet for northern harrier and white-tailed kite; 200 to 300 feet for common raptors; 50 to 100 feet for non-raptors) shall be established by a qualified biologist around active nests and no construction activities within the buffer shall be allowed until a qualified biologist has determined that the nest is no longer active (i.e., the nestlings have fledged and are no longer reliant on the nest, or the nest has failed). Encroachment into the buffer may occur at the discretion of a qualified biologist. Any encroachment into the buffer shall be monitored by a qualified biologist to determine whether nesting birds are being impacted. 	
		MM BIO-4: Burrowing Owl Pre-Construction Surveys. Pre- construction surveys for burrowing owls shall be completed no more than 15 days prior to the start of construction in areas planned for fill placement and construction areas in general conformance with the California Burrowing Owl Consortium's and the CDFW Staff Report (2012) protocols. Because owls are known to occur in the vicinity of the project site, these surveys shall be completed no more than 15 days prior (rather than 30 days prior, as per the Consortium's protocol) to the start of importing fill and construction to minimize the probability of immigration of owls between the time surveys are completed and the initiation of grading. If the initial disturbance is followed by periods of inactivity exceeding 15 days, or if the development is phased spatially and/or temporally such that an area in which construction activities are to commence has not been disturbed by construction activities within the prior 15-day period, a new burrowing owl pre-construction survey shall be completed prior to the start of disturbance. If burrowing owls are detected on one within 250 foot of the site	
		 burrowing owls are detected on or within 250 feet of the site, the mitigation measures below shall be implemented. If burrowing owl is located during the non-breeding 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 buffer zone shall be maintained around the occupied burrow(s) if practicable. If such a buffer is not practicable, then a buffer adequate to avoid injury or mortality of owls shall be maintained, or the birds shall be evicted as described below. During the breeding season (generally 1 February to 31 August), a 250-foot buffer, within which no new activity shall be permissible, shall be maintained between project activities and occupied burrows. Owls on site after 1 February shall be assumed to be nesting unless direct observations indicate otherwise. This protected buffer area shall remain in effect until 31 August, or based upon monitoring evidence, until the young owls are foraging independently, or the nest is no longer active. Owls that are not nesting can be evicted using the methods below during the period from 1 February to 31 August. If construction would directly impact occupied burrows, eviction of owls may occur outside the nesting season (or during the nesting season if the owls are determined to be not nesting) to prevent injury or mortality of individual owls. No burrowing owls shall be evicted from burrows during the nesting season (1 February through 31 August) unless evidence indicates that nesting is not actively occurring (e.g., because the owls have not yet begun nesting early in the season). Relocation of owls during the nobreeding season shall be completed by a qualified biologist using one-way doors, which shall be installed in all burrows within the impact area and left in place for at least two nights. These one-way doors shall then be removed, and the burrows backfilled immediately prior to the initiation of grading. If resident burrowing owl(s) are found in the proposed project site during pre-construction surveys and eviction is necessary to facilitate construction, Mitigation Measure BIO-6 shall be implemented. These measures do not apply to short-term use of the site by a burrowing owl for 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		foraging, as a stop-over during migration, or temporary use of the site by wintering birds or dispersing juveniles.	
		MM BIO-5: Burrowing Owl Mitigation . To reduce impacts of the project on the local (South Bay) burrowing owl population, habitat shall be preserved and managed for burrowing owls off-site if eviction of resident owls is required. California burrowing owl mitigation guidelines recommend that 6.5 acres of foraging habitat be preserved and managed per occupied burrowing owl burrow (whether by a pair or singly) in mitigation sites. Therefore, mitigation shall be required for each pair or single resident burrowing owl that is evicted, up to a maximum of 13 acres. Mitigation may take the form of off-site habitat preservation and management (in which case all the monitoring and habitat requirements in the following owl mitigation bank. Because the nearest burrowing owl mitigation banks are currently located outside of the South Bay, this mitigation may occur outside the region. If off-site habitat is to be preserved, a mitigation and	
		submitted to the City and the CDFW for review and approval. This plan shall detail the following:	
		Areas to be preserved for owls.	
		Methods for managing habitat for owls and their prey.	
		 Plans to enhance burrow availability within the mitigation site (potentially including the provision of artificial burrows, although long-term management for ground squirrels would be important as well). 	
		 A monitoring program and adaptive management program. 	
		At least 50 percent of the mitigation area must consist of upland habitat suitable for use by burrowing mammals, and no wetlands supporting tall vegetation shall be included within the mitigation site. The mitigation area must be contiguous with habitat that is permanently preserved as open space to avoid having the site surrounded by development in the future. The	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		mitigation area shall be protected in perpetuity through a conservation easement, deed restriction, conveyance to a qualified land trust or the Don Edwards National Wildlife Refuge, or through equivalent means.	
		Assuming burrowing owl habitat mitigation would occur off-site, some on-site enhancements shall also be made to reduce impacts of the project on the local (South Bay) burrowing owl population. Such enhancements shall include the provision of two artificial burrow complexes on the sides of the adjacent levees (if allowed by levee managers) and management of at least portions of levee side slopes around these burrow complexes to provide suitable conditions for burrowing owls and ground squirrels (e.g., periodic mowing to maintain short vegetation). Given the extent of natural habitat with short vegetation, and the continued presence of seasonal wetlands near the proposed project site, providing and maintaining burrows for use by owls is expected to maintain some burrowing owl presence near the proposed project site even if most or all the owl habitat mitigation occurs off-site.	
		Signage shall be placed in appropriate locations to prohibit individuals from entering areas where the artificial burrow complexes would be located. Signage shall be placed along the levee slopes to instruct recreational users of these levees against leaving the levee tops to protect sensitive species such as the burrowing owl.	
		MM BIO-6: Special Status Bat Species. A survey for roosting bats shall be completed prior to the removal of any building or tree with potential for day-roosting by bats, or prior to the initiation of any construction activities within 250 feet of such potential roost sites. The survey shall be completed by a qualified biologist. If suitable roost sites are found but a visual survey is not adequate to determine presence or absence of bats (which would be particularly likely in the case of potential roost trees), acoustical equipment could be used to determine occupancy. This survey shall be completed prior to the beginning of the breeding season (i.e., prior to 1 March) in the year in which construction or demolition in each area is scheduled to occur so that adequate measures can be	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		implemented, if necessary, to evict the bats during the non- breeding season. The survey results shall be provided to the Community Development Director for review and approval prior to the start any construction related activities.	
		 Because the initial surveys would be completed prior to the breeding season, several months may pass between that survey and the initiation of construction or demolition in each area. Therefore, a second pre-demolition/pre-construction survey for roosting bats, following the methods described above, shall be completed within 15 days prior to the commencement of these activities in each area to determine whether bats have occupied a roost in or near the development impact areas. This survey shall be facilitated considerably by information (e.g., on potential roost trees) gathered during the previous survey. If bats are found to be roosting, additional mitigation as follows must be implemented. If a maternity roost of any special-status bat species is found, the bat biologies the allowed by an area to an an area to be an area to be all determines the avtent of a species is found. 	
		construction-free buffer around the active roost that would be maintained. This buffer would be maintained from 1 March until the young are flying, typically after 31 August. If a roost of any kind is found in an area (e.g., a building or tree) that would not be disturbed by construction, or that can be avoided, the roost structure shall not be impacted.	
		 If a day roost is found in a building, or in a tree that is to be completely removed or replaced, individual bats shall be safely evicted under the direction of a qualified biologist. Eviction of bats shall occur at dusk, so that bats would have less potential for predation compared to daytime roost abandonment. Eviction shall occur between 1 September and 31 March, outside the maternity season, but shall not occur during long periods of inclement or cold weather (as determined by the bat biologist) when prey is not available, or bats are in torpor. If a day roost is found within a building, eviction shall occur by opening the roosting area to allow air flow through the cavity. Demolition may then follow no sooner than the following 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 day (i.e., there must be no less than one night between initial disturbance for air flow and the demolition). This action shall allow bats to leave during dark hours, thus increasing their chance of finding new roosts with a minimum of potential predation during daylight. If feasible, one-way doors shall also be used to evict bats from tree roosts. If use of a one-way door is not feasible, or the exact location of the roost entrance in a tree is not known, the trees with roosts that need to be removed shall first be disturbed by removal of some of the trees' limbs not containing the bats. Such disturbance shall occur at dusk to allow bats to escape during the dark hours. These trees would then be removed the following day. These activities shall be performed under the supervision of the qualified biologist. If a day roost for pallid bats or another rare bat would be impacted, an alternative bat roost structure shall be provided. The design and placement of this structure would be determined by a qualified biologist based on the location of the original roost and which species is located. This bat structure shall be erected at least one month (and preferably a year or more) prior to removal of the original roost structure. This structure shall be checked during the breeding season for up the three years following completion of the development, or until it is found to be occupied by bats, to provide information for future development projects regarding the effectiveness of such structures in minimizing impacts to bats. 	
		MM BIO-7: Reduce the Spread of Invasive Species. Prior to issuance of any building or grading permits, the project shall develop and implement an Invasive Species Management Plan to reduce the presence and spread of non-native, invasive plant species for the area to be developed. The Plan shall be developed prior to importing any fill material required to elevate building sites and prior to grading any areas on the Specific Plan site. The overarching goal of this mitigation is to	
		Specific Plan site. The overarching goal of this mitigation is to halt the further expansion of existing invasive species and introduction of new invasive species into sensitive habitats on	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		site. The Invasive Species Management Plan shall include, but not be limited to, the following, summarized below:	
		 Prior to construction, map populations of invasive species within all areas proposed to be graded; quantify the extent and location of invasive populations in sensitive habitats. 	
		 Areas identified to have weed infestations shall be treated prior to ground disturbance according to weed control methods detailed below and Best Management Practices within all upland areas to be graded, after review and approval of methodologies by the City of Newark. 	
		 Weed control treatments shall include all legally permitted herbicide, manual, and mechanical methods approved for application. The timing of the weed control treatment shall be determined for each plant species with the goal of controlling populations before they start producing seeds and/or encroach into adjacent areas from rhizomatous shoots. Consultation with a City of Newark approved wildlife biologist or plant ecologist shall be required prior to weed control treatments in sensitive habitats with the intent of avoiding any adverse impacts to special-status species in the area. 	
		 Surveying and monitoring for weed infestations shall occur annually while grading operations are occurring. Treatment of all identified weed populations shall occur at a minimum of once annually. 	
		 During project construction, all seeds and straw materials used on site shall be weed free rice straw, and all gravel and fill material shall be certified weed free. 	
		 During project construction, vehicles and all equipment shall be washed before and after entering the project area. 	
		MM BIO-8: Post-Construction Predator Management Plan and Program. This program shall focus on education of occupants of the new residential areas regarding measures to minimize the potential for subsidizing predator species and to minimize the potential effects of pets on sensitive species and	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		enforcement of the program's measures, and restrictions on certain activities that could increase predation of sensitive species. A plan shall be developed. The details of the program would be developed during construction but shall include, at a minimum, the following:	
		• Minimize disturbance from the development by educating the public about the importance of preserving the ecological integrity of the adjacent natural areas instructing recreational users to stay on the levee tops out of sensitive habitats and keep dogs on leashes.	
		• To prevent the spread of invasive non-native plants into the nearby sensitive habitats, plants contained on the California Exotic Plant Pest Council List of Invasive Plants shall be barred from use within the landscaping of the Mowry Village development area. A list of plants suitable for landscape use shall be provided to property buyers.	
		 Feeding pets outdoors shall be prohibited so that pet food does not attract or subsidize the diets of nuisance species. 	
		 Pets shall be prohibited from ranging freely (off-leash dogs shall be prohibited in off-site wetland areas and no free-roaming outdoor cats shall be permitted), to prevent their entry into sensitive species habitat. 	
		• All food waste shall be contained so that it does not attract or subsidize the diets of predators.	
Impact BIO-5: The proposed project could conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy.	Potentially Significant Impact	 MM BIO-9: Tree Preservation Plan. Prior to the issuance of any construction-phase permit, a construction-phase Tree Preservation Plan shall be prepared by a certified arborist to the satisfaction of the City's Community Development Director for all areas with trees. The construction-phase Tree Preservation Plan shall include the following tree protection measures which are based on guidelines established by the International Society of Arboriculture: Establish Tree Protection Zones. 	Less Than Significant Impact with Mitigation
		Protect tree root systems.	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		Install wood bark mulch.	
		• Trees removed by the project shall be replaced at a 3:1 ratio unless the City's Community Development Director determines that a higher ratio is required.	
		• Trees greater than 18 inches in diameter at breast height (DBH) shall not be removed unless a Tree Removal Permit, or equivalent, has first been approved for the removal of such trees.	
		• The species and exact number of trees to be planted on the site during the construction phase shall be determined in consultation with the City and to the satisfaction of the Community Development Director.	
		 In the event the developed portion of the development site does not have sufficient area to accommodate the required tree mitigation, one or more of the following measures shall be implemented at the development permit stage: 	
		• An alternative site(s) shall be identified for additional tree planting. Alternative sites may include local parks or schools, or installation of trees on adjacent properties for screening purposes, to the satisfaction of the City's Community Development Director. The size of a 15-gallon replacement tree can be increased to 24-inch box and counted as two replacement trees.	
Section 3.5 Cultural Resources			
Impact CUL-2: The proposed project could cause a substantial adverse change in the significance of an archaeological resource pursuant to Section 15064.5.	Potentially Significant Impact	 MM CUL-1: Archaeological Monitoring Program. Prior to the start of construction activities, the Applicant shall prepare a monitoring program that would be implemented during demolition and construction activities. The monitoring program shall include: Retention of a Qualified Archaeologist. Qualified archaeological shall be retained to implement a monitoring and recovery program during all ground-disturbing activity associated with the proposed project, including grubbing, grading, and excavation activities. The qualified 	Less Than Significant Impact with Mitigation



Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		archaeologist shall meet the Secretary of Interior's Professional Standards for prehistoric and historic archaeology.	
		• Agreement of Disposition of Recovered Artifacts. A written agreement shall be secured with a recognized museum repository regarding the final disposition and permanent storage and maintenance of any unique archaeological resources or historical resources recovered as a result of the archaeological monitoring, as well as corresponding geographic site data that might be recovered as a result of the specified monitoring program.	
		• Preconstruction Briefing. Construction personnel shall be briefed by the qualified archaeologist on procedures to be followed in the event that unique archaeological resources, historical resources, tribal cultural resources, or human remains are encountered during construction. The qualified archaeologist shall be required to provide a telephone number where they can be reached by the construction contractor, as necessary.	
		• Construction Monitoring. An archaeological monitor, working under the supervision of the qualified archaeologist shall observe all ground disturbing activities associated with the proposed project, including grubbing, grading, and excavation activities on- and off-site. The monitor shall be authorized to halt construction in the immediate area where buried cultural resources are encountered. In the event that any historic or prehistoric subsurface cultural resources are discovered during ground disturbing activities, all work within 50 feet of the resources shall be halted and the project Applicant shall notify the City and consult with a qualified archaeologist, as applicable, to assess the significance of the find. If any find is determined to be significant, appropriate avoidance measures recommended by the consultant and approved by the City must be followed unless avoidance is determined unnecessary or infeasible by the City. Feasibility of avoidance shall be determined with consideration of factors such as the nature of the find,	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		project design, costs, and other considerations. If avoidance is unnecessary or infeasible, other appropriate measures (e.g., data recovery, excavation) shall be instituted. Work may proceed on other parts of the project site while measures for the find are implemented.	
		 Monitoring Report. A complete set of daily monitoring logs shall be kept on-site throughout the earth moving activities and be available for inspection. The daily monitoring log shall be keyed to a location map to indicate the area monitored, date, assigned personnel, and results of monitoring, including the recovery of archaeological material, sketches of recovered materials, and associated geographic site data. Within 90 days of the completion of archaeological monitoring, a monitoring report shall be submitted to the City and filed with NWIC. Tribal Cultural Pasources. If buried resources 	
		• Tribal Cultural Resources. In buried resources discovered during construction are identified to be tribal cultural resources, a Native American tribal representative from a tribe that is traditionally and culturally affiliated to the geographic area where the project is located will be consulted to determine the significance of the discovered resource and determine the appropriate avoidance or preservation measures or action to take in accordance with PRC Section 21084.3.	
Impact CUL-3: The proposed project could disturb any human remains, including those interred outside of dedicated cemeteries.	Potentially Significant Impact	MM CUL-2: Human Burials Encountered During Construction. In the event of accidental discovery or recognition of human remains during project construction activities, PRC Section 5097.98 shall be followed and the following steps shall be taken:	Less Than Significant Impact with Mitigation
		1. There shall be no further excavation or disturbance of the specific location or any nearby area reasonably suspected to overlie adjacent human remains until the County Coroner is contacted to determine if the remains are Native American and if an investigation of the cause of death is required. If the coroner determines the remains are Native America, the coroner shall contact the NAHC within 24 hours, and the NAHC shall identify the person or	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation	
		 persons it believes to be the "most likely descendant" of the deceased Native America. The most likely descendant may make recommendations to the landowner or the person responsible for the excavation work, for means of treating or disposing of, with appropriate dignity, the human remains, and any associated grave goods as provided in PRC Section 5097.98, or 2. Where the following conditions occur, the landowner or his/her authorized representative shall rebury the Native American human remains and associated grave goods with appropriate dignity either in accordance with the recommendations of the most likely descendant or on the project area in a location not subject to further subsurface disturbance: The NAHC is unable to identify a most likely descendent failed to make a recommendation within 48 hours after being notified by the commission; The landowner or his authorized representative rejects the recommendation of the descendent, and 		
		the mediation of the NAHC fails to provide measures acceptable to the landowner.		
Section 3.7 Geology and Soils				
 Impact GEO-1: The proposed project could directly or indirectly cause potential substantial adverse effects, including risk of loss, injury, or death involving: i) Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault. 	Potentially Significant Impact	Mitigation Measure HAZ-1 is required.	Less Than Significant Impact with Mitigation	


	Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
ii) iii) iv)	Strong seismic ground-snaking. Seismic-related ground failure, including liquefaction. Landslides.		MM GEO-1: Implement Recommendations included in the Geotechnical Studies. Prior to issuance of grading permits, the Applicant shall be required to incorporate all mitigation measures and design recommendations contained within the preliminary and design level geotechnical reports prepared by Berlogar Stevens and Associates in 2019 and 2020 into relevant project plans and specifications. These specifications pertain to but are not limited to expansive soils, building foundations, foundation drainage, and backfill of excavations. The geotechnical reports prepared for the proposed project includes recommendations such as the use of post-tension concrete slab-on-grade foundation to support structures placed on expansive soils, excavation of uncontrolled fill, utility trench excavation and backfill requirements, placement of retaining walls, and implementation of California Building Code Seismic Design Parameters. The project site plans shall be submitted to the City and reviewed as part of the development review process.	
			MM GEO-2: Mitigation of Liquefaction-Induced Impacts on Buildings. Prior to issuance of grading permits, the Applicant shall be required to incorporate all mitigation measures and design recommendations relating to liquefaction contained within the preliminary and design-level geotechnical reports prepared by Berlogar Stevens and Associates in 2019 and 2020 into relevant project plans and specifications. Structural mitigation for liquefaction should, at a minimum, include the use of relatively stiff structural concrete slab-on-grade foundations, such as post-tensioned concrete slabs-on-grade, to ensure foundations resist the effects of liquefaction-induced differential settlement. Strengthening connections within each structure may also be necessary, subject to review by the City. The project site plans, with all geotechnical design measures incorporated, shall be submitted to the City and reviewed as part of the development review process.	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
Impact GEO-2: The proposed project could result in substantial soil erosion or the loss of topsoil.	Potentially Significant Impact	Mitigation Measure HYD-1 is required.	Less Than Significant Impact with Mitigation
		Mitigation Measure GEO-1, GEO-2, and HAZ-1 are required.	
Impact GEO-3: The proposed project could be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse.	Potentially Significant Impact	MM GEO-3: Prepare and Implement Dewatering and Shoring Plans. A dewatering plan shall be submitted to the City for approval prior to the issuance of a grading permit. At a minimum, the dewatering plan shall detail dewatering methods, location of dewatering activities, equipment, groundwater sampling, disposal, and discharge point in accordance with the applicable waste discharge requirements of the San Francisco Bay Regional Water Quality Control Board. In the event that shoring methods are implemented for any excavations, shoring plans shall be prepared and submitted to the City for approval prior to issuance of a grading permit. Shoring activities required for the jack and bore operations to connect utility lines under the Union Pacific Railroad tracks shall be prepared in accordance with the Union Pacific Engineering Project Specifications and Guidelines for Temporary Shoring.	Less Than Significant Impact with Mitigation
Impact GEO-4: The proposed project could be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property.	Potentially Significant Impact	Mitigation Measure GEO-1 and HAZ-1 are required.	Less Than Significant Impact with Mitigation
Impact GEO-6: The proposed project could directly or indirectly destroy a unique paleontological resource or site or unique geologic feature.	Potentially Significant Impact	MM GEO-4: Procedures for Paleontological Resources Discovered During Construction. If any paleontological resources are encountered during ground-disturbing or subsurface construction activities (e.g., trenching, grading), all construction activities within a 50-foot radius of the identified resource shall cease and the City shall immediately be notified. The applicant shall retain a qualified paleontologist (as approved by the City) to evaluate the find and recommend appropriate treatment of the inadvertently discovered	Less Than Significant Impact with Mitigation

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		paleontological resource. The appropriate treatment of an inadvertently discovered paleontological resource shall be implemented to ensure that impacts to the resource are avoided.	
Section 3.9 Hazards and Hazardous Materials			
Impact HAZ-1: The proposed project could create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials.	Potentially Significant Impact	 MM HAZ-1: Remediation of On-Site Contamination. Prior to the start of construction activities, the Applicant shall remediate existing on-site contamination at the project site to residential standards. Remediation activities shall be completed prior to site grading and construction of buildings. Remediation activities shall include: The Applicant shall implement the Corrective Action Plan and Remedial Excavation Work Plan as approved by the ACWD. The Corrective Action Plan and Remedial Excavation Work Plan provides detailed plans to remediate contaminated soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. The project site shall be remediated below relevant screening levels under oversight by an appropriate regulatory agency, in this case the ACWD. The oversight agency shall be responsible for overseeing and directing all site investigation and cleanup activities in a manner that ensures that the standards and requirements of the State of California are fully addressed. Confirmation sampling shall be completed after the conclusion of remediation activities to ensure existing onsite contamination has been remediated to residential standards. Prior to any demolition of the existing buildings (Pick-n-Pull), an asbestos survey is required by local authorities and/or National Emissione Standards for Hazardous Air 	Less Than Significant Impact with Mitigation

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		Pollutants (NESHAP) guidelines. NESHAP guidelines require the removal of potentially friable Asbestos- Containing Building Material (ACBMs) prior to building demolition or renovation that may disturb the ACBM. The results of the survey shall be submitted to the City for review and approval, prior to issuance of demolition permits.	
Impact HAZ-2: The proposed project could create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment.	Potentially Significant Impact	Mitigation Measure HYD-1 and HAZ-1 are required.	Less Than Significant Impact with Mitigation
Impact HAZ-4: The proposed project could be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would create a significant hazard to the public or the environment.	Potentially Significant Impact	Mitigation Measure HAZ-1 is required.	Less Than Significant Impact with Mitigation
Section 3.10 Hydrology and Water Quality			
Impact HYD-1: The proposed project could violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or groundwater quality.	Potentially Significant Impact	Mitigation Measure HAZ-1 is required.	Less Than Significant Impact with Mitigation



Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 MM HYD-1: Prepare and Implement a SWPPP. Prior to the issuance of any construction-related permits, the applicant shall prepare and submit a Notice of Intent (NOI) to the SWRCB and prepare a SWPPP in compliance with the NPDES General Construction Permit. The SWPPP shall include a detailed, site-specific listing of the potential sources of stormwater pollution; pollution prevention measures (erosion and sediment control measures and measures to control non-stormwater discharges and hazardous spills); description of the type and location of erosion and sediment control BMPs to be implemented at the project site; and a BMP monitoring and maintenance schedule to determine the amount of pollutants leaving the project site. A copy of the SWPPP must be current and remain on-site. Water quality BMPs identified in the SWPPP could include but are not limited to the following: a. Surface water runoff shall be controlled by directing flowing water away from critical areas and by reducing runoff velocity. Diversion structures, such as terraces, dikes, and ditches, shall collect and direct runoff water around vulnerable areas to prepared drainage outlets. b. Surface roughening, berms, check dams, hay bales, or similar devices shall be used to reduce runoff velocity and erosion. c. Sediment shall be contained when conditions are too extreme for treatment by surface protection. Temporary sediment traps, filter fabric fences, inlet protectors, vegetative filters and buffers, or settling basins shall be used to detain runoff water long enough for sediment particles to settle out. Construction materials, including topsoil and chemicals, shall be stored, covered, and isolated to prevent runoff losses and contamination of 	
		 d. Topsoil removed during construction shall be carefully stored and treated as an important resource. Berms shall be placed around topsoil stockpiles to prevent runoff during storm events. 	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		e. Fuel and vehicle maintenance areas shall be established away from all drainage courses, and these areas shall be designed to control runoff.	
		f. Temporary erosion control measures, such as silt fences, staked straw bales, and temporary revegetation, shall be employed for disturbed areas. No disturbed surfaces will be left without erosion control measures in place during the winter and spring months.	
		g. A spill prevention and countermeasure plan shall be developed to identify proper storage, collection, and disposal measures for potential pollutants (such as fuel, fertilizers, pesticides, etc.) used on-site. The plan will also require the proper storage, handling, use, and disposal of petroleum products.	
		Construction activities shall be scheduled to reduce land disturbance during peak runoff periods and to the immediate area required for construction. Soil conservation practices shall be completed during the fall or late winter to reduce erosion during spring runoff. Existing vegetation will be retained where possible. To the extent feasible, grading activities shall be limited to the immediate area required for construction.	
Impact HYD-2: The proposed project could substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the project may impede sustainable groundwater management of the basin.	Potentially Significant Impact	Mitigation Measure GEO-3 is required.	Less Than Significant Impact with Mitigation
 Impact HYD-3: The proposed project could substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would: i) Result in substantial erosion or siltation on- or off-site; ii) Substantially increase the rate or amount of surface runoff in a manner which would result in flooding on, or off site; 	Potentially Significant Impact	Mitigation Measure HYD-1 is required.	Less Than Significant Impact with Mitigation



Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
 iii) Create or contribute runoff water which would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff; or 			
IV) Impede or redirect flood flows.			
Impact HYD-5: The proposed project could conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan.	Potentially Significant Impact	Mitigation Measures HYD-1 and GEO-3 are required.	Less Than Significant Impact with Mitigation
Section 3.13 Noise			
Impact NOI-1: The proposed project could result in a generation of a substantial temporary or permanent increase in ambient noise level in the vicinity of the project in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies.	Potentially Significant Impact	 MM NOI-1: Project-Specific Acoustical Study. A project-specific acoustical analysis shall be completed at the time detailed development plans are prepared, so that the design of the residential units would be sufficient to adequately reduce interior noise level to 45 dBA Ldn or lower. Building sound insulation requirements shall include the provision of forced-air mechanical ventilation for all new units with direct line of sight to significant transportation noise sources or railroad lines in the project vicinity. Special building sound insulation treatment may be required. These treatments shall include, but are not limited to, sound rated windows and doors, sound rated wall construction, acoustical caulking, protected ventilation openings, etc. The specific determination of what treatments are necessary shall be determined on a unit-by-unit basis. The results of the analysis, including the description of the necessary noise control treatments to achieve the acceptable noise levels inside the living units, shall be submitted to the City along with building plans and shall be reviewed and approved by the Community Development Director prior to the issuance of a building permit. The project-specific acoustical study shall have the following minimum attributes: Be the responsibility of the development applicant. 	Less Than Significant Impact with Mitigation

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		 Be prepared by qualified persons experienced in the fields of environmental noise assessment ad architectural acoustics. 	
		 Include representative noise level measurements with sufficient sampling periods and locations to adequately describe existing local conditions. 	
		 Include estimates for existing and projected (20 years hence) noise levels in terms of (a) Ldn or CNEL and (b) any future noise regulations to be adopted by the City. Those existing and projected noise levels shall be compared to the adopted policies of the Newark General Plan Noise Element. 	
		 Include recommended mitigation measures to achieve compliance with the adopted policies and standards of the Newark General Plan Noise Element. Where the noise source in question consists of intermittent single events, the report should address the effects of maximum noise levels in sleeping rooms and potential sleep disturbance issues. 	
		 Include estimates for interior and exterior noise exposure after the prescribed mitigation measures have been implemented. 	
		 Describe a post-project assessment program that could be used to evaluate the effectiveness of the proposed mitigation measures. 	
		MM NOI-2: Project Fixed-Source Noise. The noise from all mechanical equipment associated with the project, including air conditioning units, shall comply with the requirements in Section 17.24.100, Paragraph A.2.a in the Newark Municipal Code. When the actual on-site equipment is selected, a noise analysis shall be prepared by a qualified acoustical engineer and the equipment shall incorporate measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels that may affect nearby properties, including adjacent homes. Noise levels from the project's fixed-source equipment	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
		dB(A) between the hours of 7:00 a.m. and 9:00 p.m. or 60 dB(A) between the hours of 9:00 p.m. and 7:00 a.m.	
		MM NOI-3: Construction Activity. All construction and remediation activity shall follow the City's time and noise reduction measure requirements for construction activity.	
		Development of the project site shall include the following construction-noise mitigation measures, to reduce noise impacts from project construction to a less than significant level.	
		• Restrict noise-generating activities at the construction site or in areas adjacent to the construction site to the hours of 7:00 a.m. to 6:00 p.m., Monday through Friday, and between 8:00 a.m. to 5:00 p.m. on Saturdays. Construction shall be prohibited on Sundays and holidays.	
		• Equip all internal combustion engine driven equipment with intake and exhaust mufflers that are in good condition and appropriate for the equipment.	
		 Unnecessary idling of internal combustion engines shall be strictly prohibited. 	
		 Locate stationary noise generating equipment such as air compressors or portable power generators as far as possible from sensitive receptors. 	
		 Construct temporary noise barriers to screen stationary noise generating equipment when located near adjoining sensitive land uses. Temporary noise barriers could reduce construction noise levels by 5 dBA. 	
		 Utilize "quiet" air compressors and other stationary noise sources where technology exists. 	
		• Route all construction traffic to and from the project site via designated truck routes where possible. Prohibit construction related heavy truck traffic in residential areas where feasible.	
		• Control noise from construction workers' radios to a point where they are not audible at existing residences bordering the project site.	

Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation				
		 The contractor shall prepare and submit to the City for approval a detailed construction plan identifying the schedule for major noise-generating construction activities. Designate a "disturbance coordinator" who would be responsible for responding to any local complaints about construction noise. The disturbance coordinator will determine the cause of the noise complaint (e.g., starting too early, bad muffler, etc.) and will require that reasonable measures warranted to correct the problem be implemented. Conspicuously post a telephone number for the disturbance coordinator at the construction site and include in it the notice sent to neighbors regarding the construction schedule. 					
Section 3.17 Transportation	Section 3.17 Transportation						
Impact TRANS-1: The proposed project would conflict or be inconsistent with CEQA Guidelines Section 15064.3, Subdivision(b).	Potentially Significant Impact	 MM TRANS-1: Implementation of a Transportation Demand Management (TDM) Plan. The applicant shall prepare and implement a TDM Plan prior to the start of construction activities. Impacts on VMT can be reduced through implementing a robust TDM program to reduce VMT through measures that discourage the use of single-occupant automobiles and encourage the use of other travel modes. The TDM Plan would reduce VMT, as well as automobile trip generation and parking demand. Due to the project location, type of development, availability of transit service, and other area characteristics, limited TDM Plan could include the following strategies: Explore the feasibility and, if feasible, coordinate with other nearby developments and/or AC Transit to provide shuttle or bus service between the project site and a BART station and/or other major destinations. Offer to provide free parking spaces for at least two car share vehicles (Zipcar, etc.) for residents to use to reduce the need for personal vehicle ownership. Offer to provide carpool matching to project residents. 	Significant and Unavoidable Impact				



Impact	Level of Significance Before Mitigation	Mitigation Measure	Level of Significance After Mitigation
Section 3.18 Tribal Cultural Resources			
 Impact TRIB-1: The proposed project could cause a substantial adverse change in the significance of a tribal cultural resource, defined in PRC Section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to California Native American tribe, and that is: i) Listed or eligible for listing in the CRHR, or in a local register of historical resources as defined in PRC Section 5020.1(k), or ii) A resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of PRC Section 5024.1. In applying the criteria set forth in subdivision (c) of PRC Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe. 	Potentially Significant Impact	Mitigation Measures CUL-1 and CUL-2 are required.	Less Than Significant Impact with Mitigation

MOWRY VILLAGE PROJECT Draft Environmental Impact Report Executive Summary



ES.5 REVIEW OF THE DRAFT EIR

The Draft EIR will be available for public review for the statutory 45-day review period and will circulate from August 2, 2023, to September 18, 2023. The document will be available for public review at the locations listed below. In addition, the Draft EIR is available electronically on the City of Newark's project webpage: <u>https://www.newark.org/departments/community-development/planning-division/projects-under-environmental-review</u>.

Agencies, organizations, and interested parties will have the opportunity to comment on this Draft EIR during the 45-day public review period. The City of Newark encourages the electronic submission of comments. Please indicate a contact person for your agency or organization and send your comments to: <u>ART.INTERIANO@newark.org</u>. Please include Mowry Village Project in the subject line.

Written comments on this Draft EIR should be addressed to:

City of Newark Attention: Art Interiano, Deputy Community Development Director 37101 Newark Boulevard, Newark, CA 94560 Phone: (510) 578-4330



Draft Environmental Impact Report Executive Summary



1.0 INTRODUCTION

This Draft Environmental Impact Report (EIR) is prepared in accordance with the California Environmental Quality Act (CEQA) to evaluate the potential environmental impacts associated with the Mowry Village Project (proposed project). This document is prepared in conformance with CEQA (California Public Resources Code [PRC] Section 21000, et seq.) and the CEQA Guidelines (California Code of Regulations [CCR], Title 14, Section 15000, et seq.). This Draft EIR is intended to serve as an informational document for the public agency decision makers and the public regarding the proposed project.

1.1 THE ENVIRONMENTAL REVIEW PROCESS

CEQA requires public agencies to identify, disclose, and consider the potential environmental impacts of proposed discretionary actions that lead agencies are considering for approval. A project that may have a significant impact on the environment cannot be approved unless the lead agency makes the approval contingent upon the implementation of mitigation measures that would reduce or avoid that impact to the extent feasible. When a project may have significant environmental impacts, the lead agency must prepare an EIR before it considers whether to approve the project.

The City of Newark (City), as the lead agency for the proposed project, has prepared this Draft EIR for public review and comment. As discussed below, the Draft EIR will be available for review and comment by public agencies and the general public for a period of 45 days. Prior to considering the proposed project, the City will prepare a Final EIR that includes the Draft EIR, the comments received on the Draft EIR, written responses to those comments, a list of commenters, and any revisions being made to the Draft EIR in response to the comments. The Final EIR will be considered by the City's discretionary bodies when taking action on the proposed project.

1.1.1 Purpose and Authority

This Draft EIR has been prepared pursuant to the State CEQA Guidelines (14 CCR 15000 et seq.). CEQA requires that State and local government agencies consider the environmental consequences of projects over which they have discretionary authority before acting on those projects (PRC 21000 et seq.).

According to CEQA Guidelines Section 15064(f)(1), preparation of an EIR is required whenever a project may result in a significant adverse environmental impact. The purpose of this Draft EIR is to analyze the potential environmental impacts of the proposed project, to indicate ways to reduce or avoid potential environmental impacts associated with the proposed project, and to identify alternatives to the project that reduce or avoid significant environmental impacts. CEQA requires that each public agency mitigate or avoid the significant environmental effects of projects it approves or implements whenever feasible.

An EIR is an informational document used in state, regional, and local planning and decision-making processes to meet the requirements of CEQA. The purpose of the EIR is not to recommend approval or

denial of a project. However, the City's decision whether to approve or to deny the project must take into consideration the information provided by the EIR. A public agency may approve a project even if it would result in significant and unavoidable environmental impacts, provided the agency adopts a statement of overriding considerations.

The Draft EIR must disclose the following: the proposed project's environmental effects, including those that cannot be avoided; the proposed project's growth inducing effects; the project-related effects found not to be significant; alternatives to the proposed project; and cumulative impacts.

1.1.2 Types of Environmental Impact Report

In accordance with CEQA Guidelines Section 15161, this document is a project EIR that examines the environmental impacts of a specific project. This type of EIR focuses on the changes in the environment that would result from a specific project. In accordance with CEQA Guidelines Section 15161, a project EIR must examine the environmental effects of all phases of the project, including construction and operation. Additional resource-specific studies, such as air quality, biological resources, cultural resources, noise, and traffic, have been prepared for this Draft EIR to provide detailed information about the proposed project's potential impacts on the environment. The mitigation measures identified in this Draft EIR are sufficiently detailed to ensure that they would be effectively carried out to reduce the proposed project's impacts.

1.1.3 Lead Agency Determination

The City is designated as the lead agency for the proposed project. CEQA Guidelines Section 15367 defines the lead agency as, "...the public agency, which has the principal responsibility for carrying out or approving a project." Other public agencies may use this document in their decision making or permit processes (e.g., Department of Water Resources, Bay Area Air Quality Management District [BAAQMD], California Department of Transportation [Caltrans], etc.).

This Draft EIR was prepared by the City with technical assistance provided by Stantec Consulting Services Inc. (Stantec), an environmental consultant. Prior to public review, this Draft EIR was extensively reviewed and evaluated by the City staff and, as such, the Draft EIR reflects the independent judgment and analysis of the City, as required by CEQA. Lists of organizations and persons consulted and the report preparation personnel are provided in Section 7.0, List of Preparers, of this Draft EIR.

1.2 SCOPE OF DRAFT EIR

The City determined that an EIR was required for the proposed project and this Draft EIR addresses the potential environmental effects of the proposed project. The City distributed a Notice of Preparation (NOP) of a Draft EIR for the proposed project beginning on November 30, 2021. The NOP was distributed for a 30-day comment period that ended on January 3, 2022. The comments on the NOP were considered in the preparation of this Draft EIR. The scope of this Draft EIR includes the potential impacts identified in the NOP and issues raised by agencies and the public in response to the NOP.



Draft Environmental Impact Report Introduction

The City has determined that the proposed project has the potential to result in significant environmental impacts on the following resources, which are addressed in detail in this Draft EIR.

- Aesthetics
- Agricultural and Forestry Resources
- Air Quality
- Biological Resources
- Cultural Resources
- Energy
- Geology and Soils
- Greenhouse Gas Emissions
- Hazards and Hazardous Materials
- Hydrology and Water Quality

- Land Use and Planning
- Mineral Resources
- Noise
- Population and Housing
- Public Services
- Recreation
- Transportation
- Tribal Cultural Resources
- Utilities and Service Systems
- Wildfire

Table 1.2-1 lists the comment letters received during the project scoping period.

Table 1.2-1: Comments Received on the NOP

Affiliation	Signatory	Date	Comment Description	EIR Section Where Comment is Addressed
Private Parties – Written				
Interested Individual	Kelly	December 15, and 16, 2021	Concerns about sea level rise	 EIR Section 3.10: Hydrology and Water Quality
Interested Individual	Pat Callaway	December 18, 2021	Concerns about additional housing increasing crime, traffic congestion, and noise as well as burden on public service resources; concerns about sea level rise and water shortages; concern with being located in an earthquake liquefaction zone and shallow groundwater	 EIR Section 3.7: Geology and Soils EIR Section 3.9: Hazards and Hazardous Materials EIR Section 3.10: Hydrology and Water Quality EIR Section 3.13: Noise EIR Section 3.15: Public Services

\bigcirc

Draft Environmental Impact Report Introduction

Affiliation	Signatory	Date	Comment Description	EIR Section Where Comment is Addressed
				 EIR Section 3.17: Transportation EIR Section 3.19: Utilities and Service Systems
Interested Individual	Victoria Richard	December 3, 2021	Requested more designated park space (park for elementary age kids and enclosed space for dogs)	• EIR Section 3.16: Recreation
Interested Individual	Robert Lucey	December 7, 2021	Concerns with significant increase in vehicular traffic and design of proposed travel lanes along Mowry Avenue. Requests provision of four traffic lanes (two in each direction) from Cherry Street to the new subdivision.	• EIR Section 3.17: Transportation
Cargill, Incorporated	Ric Notini	December 29, 2021	Concerns with impacts related to air quality, transportation, and hydrology and water quality effects that may occur to the adjacent salt ponds from development of the project	 EIR Section 3.3: Air Quality EIR Section 3.10: Hydrology and Water Quality EIR Section 3.17: Transportation
Several Organizations – Joint Letter of Concern (Greenbelt Alliance, CCCR, Sierra Club, Center for Biological Diversity, San Francisco Baykeeper, Alameda Creek Alliance, Santa Clara Valley Aubudon Society, Ohlone Aubudon Society, Mission Peak Conservancy)	Zoe Siegel, Carin High, Martha Kreeger, Lisa Belenky, Eric Buescher, Jeff Miller, Shani Kleinhaus, William Hoppes, William Yragui	December 13, 2021	Concerns with sea level rise and site being located within a significant wildlife habitat	 EIR Section 3.4: biological Resources EIR Section 3.10: Hydrology and Water Quality
Alameda County Transportation Commission	Cathleen Sullivan	January 3, 2022	Outlined requirements for conducting transportation impact analysis and modeling for the project; outlined requirements for impacts and mitigation measures that should be discussed in the DEIR	• EIR Section 3.17: Transportation

Draft Environmental Impact Report Introduction

Affiliation	Signatory	Date	Comment Description	EIR Section Where Comment is Addressed	
Native American Heritage Commission	Katy Sanchez	December 6, 2021	Discussed compliance with AB 52 and SB 18 and recommendations for Cultural Resources Assessments	 EIR Section 3.5: Cultural Resources EIR Section 3.18: Tribal Cultural Resources 	
Alameda County Water District	Laura J. Hidas	January 3, 2022	Outlined that the WSA prepared in 2008 for the Specific Plan is still applicable unless Areas 3 and 4 demands have increased in which case would require ACWD review; outlined groundwater resource impacts that is required to be addressed in the EIR; outlined utilities and service system requirements for the project	 EIR Section 3.10: Hydrology and Water Quality EIR Section 3.19: Utilities and Service Systems 	
Citizens Committee to Complete the Refuge	Carin High	January 3, 2022	Concerns with sea level rise; concerns with effect of raising ground elevation of the project site resulting in displacement of flood waters to adjacent properties and flood control capabilities of Line D; concerns with the potential presence of wetlands adjacent to the project site and associated impacts and proximity of the site to significant wildlife habitat	 EIR Section 3.4: Biological Resources EIR Section 3.10: Hydrology and Water Quality 	
Union Sanitary	Rollie Arbolante	January 5, 2022	Outlines requirement of project to enter into agreement with Union Sanitary that the developer of the project would be responsible for replacement of the Cherry Street pump Station as the pump station does not have capacity to serve the development	• EIR Section 3.19: Utilities and Service Systems	
Private Parties – Oral ¹					
Interested Individual	Kelly Abreo	December 14, 2021	Questioning merits of the project, made references to preference of vertical development. Concerns with pedestrian safety, community continuity and land use planning, sea level rise, and flooding.	 EIR Section 3.1: Aesthetics EIR Section 3.10: Hydrology and Water Quality EIR Section 3.11: Land Use and Planning 	

Draft Environmental Impact Report Introduction

Affiliation	Signatory	Date	Comment Description	EIR Section Where Comment is Addressed
				• EIR Section 3.17: Transportation
Several Organizations	Josh Senefeld	December 14, 2021	Concerns with sea level rise and sea level rise revealing contamination that will seep into waterways; concern with preservation of wildlife habitats	 EIR Section 3.4: Biological Resources EIR Section 3.9: Hazards
				 and Hazardous Materials EIR Section 3.10: Hydrology and Water Quality
Sierra Club	William Yragui	December 14, 2021	Concerns about sea level rise, liquefaction, and believes building housing on built up dirt mound is irresponsible. Would like the habitat to be preserved.	 EIR Section 3.4: Biological Resources EIR Section 3.7: Geology and Soils
				EIR Section 3.10: Hydrology and Water Quality
City of Newark	John Becker	December 14, 2021	Concerns about only one access point in proposed development and emergency access to the site, would like to study access to the site. Concerns with sea level rise and noise as development would be near railroad and railroad crossing	 EIR Section 3.9: Hazards and Hazardous Materials
				 EIR Section 3.10: Hydrology and Water Quality EIR Section 3.13: Noise
				• EIR Section 3.17: Transportation
City of Newark	William Fitts	December 14, 2021	Concerns about safety aspect of transportation and would like to add second route out of site for public safety	 EIR Section 3.9: Hazards and Hazardous Materials
				• EIR Section 3.17: Transportation
City of Newark	Debbie Otterstetter	December 14, 2021	Concerns with soil contamination and hazardous impacts	 EIR Section 3.9: Hazards and Hazardous Materials

Draft Environmental Impact Report Introduction

Affiliation	Signatory	Date	Comment Description	EIR Section Where Comment is Addressed
City of Newark	Jeff Aguilar	December 14, 2021	Would like safety crossing at railroad evaluated, would like a crosswalk included	 EIR Section 3.9: Hazards and Hazardous Materials
				 EIR Section 3.17: Transportation

¹ Oral comments taken at the public Scoping Meeting held on December 14, 2021.

1.2.1 Location and Overview

The project site is located within the City of Newark in southwestern Alameda County, California, southwest of the intersection of Mowry Avenue and the Union Pacific Railroad (UPRR) tracks, west of Cherry Street. The 29 acre project site consists of three parcels identified as Assessor's Parcel Numbers (APNs) 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00. The project site is within the Newark Areas 3 and 4 Specific Plan and the proposed project lies within Sub Area D of Area 4.

Mowry Project Owner, LLC (Applicant) is proposing to demolish the existing on-site structures, remediate the site, and construct 203 single-family detached homes. Additional improvements would include onstreet parking, drive aisles, underground utilities, storm drainage systems with Low Impact Development (LID) techniques incorporated and water quality treatment areas, lighting, sidewalks, and landscaping. The proposed project would also include off-site improvements and widening of Mowry Avenue. Section 2.0, Project Description, includes more detailed information about the proposed project.

1.3 ORGANIZATION OF THE DRAFT EIR

This Draft EIR is arranged into the following sections, which contain the contents of an EIR as required by CEQA Guidelines Section 15120 through 15132.

Section ES: Executive Summary. The Executive Summary provides a summary of the proposed project and the project alternatives, including a summary of project impacts, recommended mitigation measures, and the level of significance after mitigation for each environmental issue.

Section 1.0: Introduction. The Introduction provides an overview of the proposed project and the CEQA process and describes the purpose, scope, and components of this Draft EIR.

Section 2.0: Project Description. The Project Description provides a detailed description of the proposed project, including the location and project characteristics. The intended uses of this Draft EIR, project background, project objectives, and required project approvals are also addressed.



Section 3.0: Environmental Impact Analysis. The Environmental Impact Analysis analyzes the environmental effects of the proposed project. Impacts are organized into major environmental topic areas. Each topic area includes a description of the regulatory setting, environmental setting, significance criteria, project impacts, mitigation measures, and level of significance after mitigation. The specific environmental topic areas that are addressed in Section 3.0 include the following:

- Section 3.1: Aesthetics
- Section 3.2: Agricultural and Forestry Resources
- Section 3.3: Air Quality
- Section 3.4: Biological Resources
- Section 3.5: Cultural Resources
- Section 3.6: Energy
- Section 3.7: Geology and Soils
- Section 3.8: Greenhouse Gas Emissions
- Section 3.9: Hazards and Hazardous Materials
- Section 3.10: Hydrology and Water Quality
- Section 3.11: Land Use and Planning
- Section 3.12: Mineral Resources
- Section 3.13: Noise
- Section 3.14: Population and Housing
- Section 3.15: Public Services
- Section 3.16: Recreation
- Section 3.17: Transportation
- Section 3.18: Tribal Cultural Resources
- Section 3.19: Utilities and Service Systems
- Section 3.20: Wildfire

Section 4.0: Cumulative Effects. Section 15130 of the CEQA Guidelines requires an EIR to discuss cumulative impacts of a project when the project's incremental effect is cumulatively



considerable. A cumulative impact consists of an impact created because of the combination of the project evaluated in the EIR together with other reasonably foreseeable projects causing related impact.

Section 5.0: Alternatives to the Proposed Project. Describes and compares the proposed project alternatives to the proposed project.

Section 6.0: Other CEQA Considerations. The Other CEQA Considerations section provides a summary of significant environmental effects, including unavoidable, irreversible, and growth-inducing impacts.

Section 7.0: List of Preparers and Organizations Consulted. The List of Preparers and Organizations Consulted section provides a list of the organizations and persons consulted, and the various individuals who contributed to the preparation of this Draft EIR. This section also includes a list of the lead agency personnel and technical consultants used to prepare this Draft EIR.

Section 8.0: References. This section provides a list of the technical studies and other documents used to prepare this Draft EIR.

Appendices. The appendices contain the NOP (including comments) and technical studies prepared to support the analyses and conclusions in this Draft EIR.

1.4 REVIEW OF THE DRAFT EIR

CEQA does not require formal hearings at any stage of the environmental review process (CEQA Guidelines Section 15202[a]). However, it does encourage, "wide public involvement, formal and informal, in order to receive and evaluate public reactions to environmental issues..." (CEQA Guidelines Section 15201). The City distributed an NOP of a Draft EIR for the project beginning on November 30, 2021. The NOP was distributed for a 30-day comment period that ended on January 3, 2022. The comments on the NOP were considered in the preparation of this Draft EIR. Appendix A contains the written comments received on the NOP.

The City of Newark has filed a Notice of Completion (NOC) with Governor's Office of Planning and Research (OPR) to begin the public review period (PRC, Section 21161). Concurrent with the NOC, this Draft EIR has been distributed to responsible and trustee agencies, other affected agencies, surrounding cities, and interested parties, as well as to all parties requesting a copy of the Draft EIR in accordance with PRC Section 21092(b)(3). During the public review period, the Draft EIR, including the technical appendices, are available for review online at: <u>https://www.newark.org/departments/community-development/planning-division/projects-under-environmental-review</u>.

Agencies, organizations, and interested parties have the opportunity to comment on this Draft EIR during the 45-day public review period, which will begin on August 2, 2023, and end on September 18, 2023. The City of Newark encourages the electronic submission of comments.



Draft Environmental Impact Report Introduction

Send your comments by email to: <u>ART.INTERIANO@newark.org</u>. Please include Mowry Village Project Draft EIR in the subject line.

Written comments on this Draft EIR should be addressed to:

City of Newark Attention: Art Interiano, Deputy Community Development Director 37101 Newark Boulevard Newark, CA 94560 Phone: (510) 578-4330

Upon completion of the public review period, written responses to all environmental issues raised will be prepared and made available for review by the commenting agencies at least 10 days prior to any public hearing on the proposed project at which the certification of the Final EIR will be considered. Comments received and the responses to comments will be included as part of the record for consideration by decision-makers for the proposed project.

1.4.1 Effectively Commenting on an EIR

Readers are invited to review and comment on the adequacy and completeness of this Draft EIR in describing the potential impacts of the proposed project, the level of severity of each impact, the mitigation measures being proposed to reduce or avoid those impacts, and the project alternatives being considered. The most effective comments are those that focus on the adequacy and completeness of the environmental analysis and that are supported by factual evidence. Comments that focus on whether the proposed project should be approved or denied are not comments on the adequacy of this Draft EIR.

1.4.2 Final EIR

After the end of the review period, the City will review the comments received, prepare written responses to those comments, make any related revisions to the Draft EIR, and publish the Final EIR, which will include the Draft EIR, comments on the Draft EIR, responses to comments and any revisions to the Draft EIR.

The Final EIR will be considered by the City's Planning Commission and City Council when taking action on the proposed project. If the proposed project is approved, CEQA requires the City to adopt findings describing how each of the significant impacts identified in the EIR is being mitigated. The findings are required to describe the reasons why significant unavoidable impacts, if any, cannot be mitigated; in this case, all significant effects of the project would be mitigated to less than significant levels by the adoption of feasible mitigation measures except for impacts to transportation. The findings will also describe the project alternatives analyzed in the EIR and explain whether or not any alternative or portion of an alternative has been adopted. Because the proposed project has significant and unavoidable impacts, the City would be required to adopt a statement of overriding considerations describing the benefits of the proposed project that outweigh its environmental impacts. Finally, the City would adopt a mitigation monitoring and reporting plan that describes how it will ensure the mitigation measures being required of the proposed project would be carried out.



2.0 PROJECT DESCRIPTION

2.1 PROJECT OVERVIEW

The Applicant is proposing development of the Mowry Village Project in the City of Newark, California. The project site consists of a 29 acre site within the Areas 3 and 4 Specific Plan that is currently developed as an auto part and scrap metal salvage lot, known as "Pick-n-Pull." The Applicant is proposing to demolish the existing on-site structures, remediate the site, and construct 203 single-family detached homes. Additional improvements would include on-street parking, drive aisles, underground utilities, LID drainage and water quality treatment areas, lighting, sidewalks, and landscaping. The proposed project would also include off-site improvements and widening of Mowry Avenue.

2.1.1 Project Site

The project site is located within the City of Newark in southwestern Alameda County, California, southwest of the intersection of Mowry Avenue and the UPRR tracks, west of Cherry Street (Figure 2-1). The project site consists of three parcels identified as APNs 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00 (Figure 2-2). The proposed project would include off-site utility and circulation improvements along Mowry Avenue.

2.1.2 General Plan and Zoning

The project site is designated Low Density Residential by the City of Newark 2013 General Plan and zoned Park. In accordance with the General Plan, the allowable density for Low Density Residential is less than 8.7 dwelling units per acre. The proposed project would be consistent with this requirement and construct 203 single-family detached units, resulting in a density of 7 units per acre. The City's General Plan defines this designation as:

 Low Density Residential (Less than 8.7 units per net acre). This designation is intended for single-family residential developments on lots larger than 5,000 square feet. It corresponds to most of Newark's residential neighborhoods. Multiple zoning districts apply within Low Density Residential areas to distinguish areas with different minimum lot sizes. Other compatible uses, such as schools, childcare centers, parks, and religious facilities may also locate in areas with this designation, subject to appropriate permitting requirements (City of Newark 2013a).

The project is proposing to rezone the project site from Park to RS-6000: Residential Single-Family with Planned Unit Development Overlay (PD-RS-6000). The rezoning request is to better align the zoning with the proposed use of the project and the existing General Plan designation. RS-6000 refers to residential single-family zoning district with a minimum lot size of 6,000 square feet. The residential single-family district implements the low density residential General Plan land use designation.



Draft Environmental Impact Report Project Description



Disclaimer: This document has been prepared based on information provided by others as cited in the Notes section. Stantec has not verified the accuracy and/or completeness of this information and shall not be responsible for any errors or omissions which may be incorporated herein as a result. Stantec assumes no responsibility for data supplied in electronic format, and the recipient accepts full responsibility for verifying the accuracy and completeness of the data.

Active\185705508\mxd\fig_2-1_regional_location.mxd Revised: 2023-04-19 By: KAEJOHN

Draft Environmental Impact Report Project Description



Disclaimer: This document has been prepared based on information provided by others as cited in the Notes section. Stantec has not verified the accuracy and/or completeness of this information and shall not be responsible for any errors or omissions which may be incorporated herein as a result. Stantec assumes no responsibility for data supplied in electronic format, and the recipient accepts full responsibility for verifying the accuracy and completeness of the data.

Draft Environmental Impact Report Project Description

Draft Environmental Impact Report Project Description

Due to the minimum lot size and setbacks prescribed in the RS-6000 zoning district, a Planned Unit Development is being proposed to allow a deviation from the standards listed in the RS-6000 development standards in the Newark Municipal Code. Figure 2-3 shows the existing zoning designation for the project site and surrounding areas and Figure 2-4 shows the proposed zoning designation.

Additionally, the project site is within the Areas 3 and 4 Specific Plan which is comprised of Area 3, encompassing 296 acres, and Area 4, encompassing 560 acres. Areas 3 and 4 are further divided into Sub Areas A through F. The proposed project lies within Sub Area D of Area 4 and is designated for a golf course or other recreational uses. The proposed project is not consistent with the land use identified by the Areas 3 and 4 Specific Plan. As such, the proposed project also requires a Specific Plan Amendment to change the use to residential single-family. The standards for single-family developments are outlined in the Specific Plan. Additionally, the Areas 3 and 4 Specific Plan allocated 1,260 residential units to be constructed within the Specific Plan area and designated specific areas within the Specific Plan area for the development of these allocated residential units. Though the Specific Plan's allocated 1,260 units have not yet all been developed, the Development Agreement for the Sanctuary West Project, which is located within the Specific Plan area east of the project site, were assigned the remaining allocated units within the Specific Plan area. Therefore, the development of the proposed project's 203 residential units would not be within the Specific Plan allocated residential units of 1,260 units and would be above the allowed number of units for the Specific Plan area. Therefore, approval of the proposed Specific Plan Amendment and rezoning would allow for the development of more units within the Specific Plan area above the planned number of units.

2.1.3 Existing Site Conditions

The project site is located in an area with agricultural and industrial uses in the southwestern portion of the City. The majority of the project site is developed as an auto part and scrap metal salvage lot, known as "Pick-n-Pull," that includes a 13,000 square foot warehouse, 1,500 square foot sales office, 3,000 square foot workshop, and a large parking area for storing vehicles that consists of crushed rock and asphalt. The northern parcel of the project site is currently undeveloped land.

The undeveloped northern parcel of the project site is roughly triangular in shape and occupies an area of approximately 10 acres. Site topography is generally flat although fill has been placed in the central area of the undeveloped parcel. The surface elevation is about 10 feet above mean sea level (AMSL) around the perimeter of the parcel with a mound up to about 15 feet AMSL in the center. The surface elevation of the middle and southern parcels of the project site is about 10 feet AMSL along the northern property line where it abuts the undeveloped northern parcel as well as along the Mowry Avenue frontage and in the southwestern area of the salvage yard where the warehouse building is located. The topography of the parcels throughout the main yard area varies from about 10 feet AMSL at the west to 5 feet AMSL at the far east end of the yard. There is a pair of constructed water quality basins along the eastern boundary and at the southern tip of the project site.



Draft Environmental Impact Report Project Description



San Francisco Oakland Tracy San Leandro Livermore San Mateo Fremont San Jose 3185 ft

Notes 1. Coordinate System: NAD 1983 StatePlane California III FIPS 0403 Feet 2. Data Sources: 3. Background: Sources: Esri, HERE, Garmin, Intermap, increment P Corp., GEBCO, USGS, FAO, NPS, NRCAN, GeoBase, IGN, Kadaster NL, Ordnance Survey, Esri Japan, METI, Esri China (Hong Kong), (c) OpenStreetMap contributors, and

Legend

- Project Site (includes boundaries of proposed roadway and utility improvements)
- Railroad
- **City of Newark Zoning Districts**
 - Business and Technology Park Limited Industrial
 - Park
 - **Public Facilities**
 - Residential Single Family-6000
 - **Resource Production**







Newark, California

Client/Project

City of Newark Mowry Village Project

Figure No. 2-3 Trite Existing General Plan Land Use Designations and Zoning Districts

Disclaimer: This document has been prepared based on information provided by others as cited in the Notes section. Stantec has not verified the accuracy and/or completeness of this information and shall not be responsible for any errors or ormissions which may be incorporated herein as a result. Stantec assumes no responsibility for data supplied in electronic format, and the recipient accepts full responsibility for verifying the accuracy and completeness of the data.

Draft Environmental Impact Report Project Description



San

Oakland San Leandro Livermore San Mateo Fremont

San Jose

3185 ft

 Acotes
 Coordinate System: NAD 1983 StatePlane California III FIPS 0403 Feet
 Data Sources:
 Background: Sources: Esri, HERE, Garmin, Intermap, increment P Corp., GEBCO, USGS, FAO, NPS, NRCAN, GeoBase, IGN, Kadaster NL, Ordnance Survey, Esri Japan, METI, Esri China (Hong Kong), (c) OpenStreetMap contributors, and

Legend

- Project Site (includes boundaries of proposed roadway and utility improvements)
- + Railroad
- City of Newark Zoning Districts
 - Business and Technology Park
 - Limited Industrial Park
 - Public Facilities
 - Residential Single Family-6000
 - Resource Production







Newark, California

Client/Project City of Newark Mowry Village Project

Figure No. 2-4 Title Proposed Zoning

Disclaimer: This document has been prepared based on information provided by others as cited in the Notes section. Stantec has not verified the accuracy and/or completeness of this information and shall not be responsible for any errors or omissions which may be incorporated herein as a result. Stantec assumes no responsibility for data supplied in electronic format, and the recipient accepts full responsibility for verifying the accuracy and completeness of the data.

Draft Environmental Impact Report Project Description
Draft Environmental Impact Report Project Description

2.1.4 Existing Operations

A portion of the project site is currently developed and occupied by an auto parts and scrap metal salvage lot known as "Pick-n-Pull." The Pick-n-Pull hours of operation are from 9:00 AM to 5:00 PM on weekdays and 8:00 AM to 5:00 PM on weekends.

2.1.5 Surrounding Land Uses

The project site is generally bounded to the northeast and east by the City's open space parcel, formerly used for agriculture. Mowry Avenue and Alameda County Flood Control and Water Conservation District's (ACFC&WCD) Line B canal lies to the west. The property to the south and southwest, known as the Harwinder Singh site, was previously developed with one warehouse type structure near Mowry Avenue, and the site was used as an auto wrecking yard. The building has since been demolished, and there are presently no buildings on the Harwinder Singh site. The UPRR tracks are approximately 300 to 1,000 feet northwest of the project site. Additionally, Cargill owns and operates salt production ponds located west of the project site, and Mowry Slough begins approximately 1,500 feet southwest of the site.

2.1.6 Residential Development

The project site plan is shown in Figure 2-5. The proposed project includes construction of 203 singlefamily detached homes on the 29 acre project site, resulting in a density of 7 units per acre. The proposed single-family homes would be located on three typical lot sizes that are 3,375 square feet, 3,600 square feet, or 4,000 square feet. Each home would be two stories tall and feature various floor plans with four to five bedrooms, a two-car garage, and a rear yard. The various lot sizes would feature New Traditional Mediterranean, Contemporary Spanish, or Farmhouse architectural styles. The New Traditional Mediterranean would provide a combination of horizontal stone masonry and sanded stucco, while the Contemporary Spanish would provide a red tile roof covering with prominent arches above the doors or windows. The Farmhouse architectural style would provide gable roofs, exposed wood, and covered porches (Figure 2-6).

According to the U.S. Census Bureau, the City's average household size between the years 2017 and 2021 was 3.32 people per household (U.S. Census 2022). The proposed project would construct 203 single-family homes, which would generate up to 674 residents if fully occupied.

2.1.7 Residential Open Space

The proposed project would provide 212,922 square feet (4.89 acres) of on-site open space which would include both common and private open space areas. The proposed project would provide 40,802 square feet (0.94 acres) of common open space consisting of landscaping and bioretention areas. The on-site open space area that would provide recreational opportunities would be located in the center of the project site and would include amenities such as a lawn, pedestrian path, and picnic tables. The proposed project would also provide a rear yard for each home, resulting in a total of 172,120 square feet (3.95 acres) of private open space.



Draft Environmental Impact Report Project Description



Project Site Plan

MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description





MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description



Draft Environmental Impact Report Project Description

2.1.8 Circulation

Mowry Avenue Roadway Improvements

Mowry Avenue along the project frontage to the UPRR rail crossing is proposed to be widened to include a single travel lane, a center median with left turn pocket, bike lanes with buffers, landscaped parkways, stormwater treatment, and sidewalk along the project frontage (Figure 2-7). The proposed project would widen the right-of-way (ROW) of Mowry Avenue, south of the UPRR tracks, from 49.5 feet to 96 to 104 to accommodate one 12 foot vehicle lane in the southbound direction, one 12 foot vehicle lane in the northbound direction, a 12 foot wide median and left turn pocket to access the project site. A 6 foot bicycle lane with 3 foot buffer would also be provided in each direction of travel. A 5 foot parkway strip, 5 foot sidewalk, and 3 foot landscape strip on the northbound side would be provided with a 4 foot landscape strip and a minimum 10 foot setback from face of curb to the top of bank of the ACFC&WCD's Line B channel on the southbound side.

The proposed sidewalk along the frontage of Mowry Avenue would conform to the existing UPRR crossing to the north. The proposed project would provide pedestrian crossing improvements at the UPRR crossing, which would be equipped with crossing arms, upgraded roadway panels, signage, striping, and pedestrian path and bicycle crossing improvements to encourage safer access to the proposed project, surrounding development, and recreation facilities. The UPRR crossing would also include any required gate signals, visual, and/or audio equipment, as required by UPRR or the Newark Municipal Code.

Additionally, existing Mowry Avenue north of the UPRR tracks and extending to Cherry Street would be re-striped and a mid-block crossing to the Silliman Center would be constructed. Re-striping the road would eliminate one travel lane in the southbound direction to accommodate a single 14 foot vehicle travel lane, a 3 foot bike buffer, a 6 foot bike lane and a 10 foot parking lane matching the northbound side of Mowry Avenue. These striping improvements would accommodate the proposed mid-block crossing proposed at the Silliman Center.

The mid-block crossing would be located approximately mid-point between the UPRR tracks and Cherry Street, along the Silliman Center frontage. Improvements would include construction of high visibility crosswalk markings, Rectangular Rapid Flashing Beacons (RRFBs), advance pedestrian crossing yield markings, advance pedestrian crossing signage, median refuge, curb extensions, and Americans with Disability Act (ADA) compliant curb ramps.

Vehicular Access/Street Design

The residential development would be accessible directly from Mowry Avenue and would be oriented along several internal streets serving the neighborhood. The proposed private streets include three east/west oriented roadways referred as, "A" Street, "B" Street, and "C" Street, with "A" Street and "B" Street functioning as the main arterials through the neighborhood. The proposed private streets also include four north/south oriented roadways referred as, "D" Avenue, "E" Avenue, "F" Avenue, and "G" Drive, all of which intersect with the east/west oriented roadways within the residential development. The proposed private streets would total approximately 7.3 acres of the project site. Each street would include two travel lanes, parking and sidewalks on both sides, and trees along the frontages. Two courts are



Draft Environmental Impact Report Project Description

proposed within the project site, branching off the private streets. The streets are proposed to be privately owned and maintained by a Homeowner's Association (HOA). Bulb-outs are proposed at street intersections to promote traffic calming and provide shorter street crossings for pedestrians.

Parking

Parking supply requirements are based on the Newark Municipal Code, Section 17.23.040 - Required Number of On-Site Parking Spaces. Based on the City's requirements of two spaces per unit for single-family homes (detached), a total of 508 parking spaces would be required. The proposed project would provide 962 parking spaces, including 406 off-street covered spaces, 406 off-street driveway spaces, and 150 on-street guest spaces. With a planned supply of 962 spaces, the proposed parking supply would exceed the City's requirements with a surplus of 454 spaces.

Pedestrian Circulation

Sidewalks would be provided along both sides of each private street and would connect to sidewalks along Mowry Avenue to be developed as part of the proposed project. Construction of the proposed project would include a sidewalk along the project frontage on Mowry Avenue. The sidewalk would connect to the proposed crossing provided at the UPRR tracks.

Fire Access

The minimum width available for driving or turning movements throughout the project site would be 20 feet. The neighborhood streets would be at least 36 feet wide. The project roadway and neighborhood design would provide adequate turning radii and drive areas for fire trucks and other emergency vehicles.

2.1.9 Landscaping

Although there is currently some landscaping on the project site, including existing trees and vegetation, the existing landscaping would be removed during the demolition phase and redeveloped as part of the proposed project. There are currently 13 trees within the project site proposed for removal. The project is proposing 213 trees to be planted to replace the existing removed trees and provide landscaping throughout the site. Any tree that would be removed would be required to comply with Chapter 8.16 - Preservation of Trees on Private Property, of the Newark Municipal Code. Landscaping for the proposed project would be placed along the project frontage on Mowry Avenue and along the new private streets created throughout the development (Figure 2-8). Landscaping for the proposed project would consist of trees, shrubs, and groundcover plants. Mailbox kiosks would be located within the landscaped areas and seating would be installed within these areas. These landscaping features would provide pedestrian-friendly frontages throughout the project site. All proposed landscaping would consist of low water use plants and would meet the requirement of the Model Water Efficient Landscape Ordinance and city, state, and water agency water conservation regulations.





MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description





MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description



Draft Environmental Impact Report Project Description

2.1.10 Lighting

The proposed project would provide exterior lighting throughout the site to illuminate the main entrances of the single-family homes, private streets, sidewalks, common space areas, and driveways for security and safety purposes. Additional lighting would be installed along the project frontage on Mowry Avenue. Outdoor lighting installed for the proposed project would be designed in accordance with lighting regulations and standards outlined under Newark Municipal Code Section 17.17.060, Lighting and Illumination.

2.1.11 Utility Infrastructure

Water Supply

Currently, municipal water is the primary sources of water for the project site. The Alameda County Water District (ACWD) approved a Water Supply Assessment (WSA), prepared November 2008, for the Areas 3 and 4 Specific Plan area that indicated sufficient supplies exist to meet the ACWD's projected demands for its service area as well as the Areas 3 and 4 Specific Plan's demands, including the proposed project, under normal year conditions. Though the WSA utilized the Specific Plan's proposed land uses to determine projected demand within the Specific Plan area, the change to the project site's proposed land use from golf course or other recreational use to residential would not result in changes to the determination included in the WSA as the proposed project's residential units would reduce the water demand for the site from what was assumed for the development of the golf course and the water demand for the Specific Plan area has been reduced since the preparation of the WSA. During critically dry or multiple dry years, the ACWD service area may be facing water supply shortages. Because the Specific Plan's demands are already factored into the 2020-2025 Urban Water Management Plan (UWMP), the development of these 203 homes which fall within the 1,260 residential units foreseen by the Specific Plan would not result in increased shortages beyond those which are already factored into ACWD's planning under current and foreseeable conditions (City of Newark 2014).

An 8 inch water main, totaling approximately 6,300 linear feet, is proposed to serve the residents on-site and connect to a proposed 12 inch water main within Mowry Avenue. The 12 inch water main is proposed to be extended from the terminus of the existing 16 inch water main on the north side of the UPRR tracks within Mowry Avenue toward the project site. The water main extension would be constructed through a jack-and-bore operation under the UPRR tracks, totaling approximately 1,850 linear feet. Fire hydrants are proposed throughout the development and along Mowry Avenue consistent with ACWD, Alameda County Fire Department (ACFD), and California Fire Code (CFC) standards.

In addition, potable and non-potable water mains would be extended approximately 900 linear feet from the southwest corner of the Sanctuary West Development, within the old Addition Road alignment adjacent and parallel to ACFC&WCD Line D channel, to the UPRR ROW (Figure 2-9). The potable and non-potable water mains would be jack-and-bored, approximately 250 linear feet, under the Line D channel and the UPRR ROW. The non-potable water main would be stubbed on the western edge of the UPRR ROW for future connection. The potable water main would extend 2,500 linear feet northwest along the UPRR ROW, within an existing utility easement, to Mowry Avenue and then an additional 500 linear feet down Mowry Avenue to the project entrance at future 'A' Street.



Draft Environmental Impact Report Project Description



MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description



Draft Environmental Impact Report Project Description

Wastewater

The Union Sanitary District (USD) would provide sanitary sewer service to the project site. Area 3 is within the existing USD service area boundary, but USD has indicated Area 4 would need to be annexed into their jurisdictional boundaries through Alameda County Local Agency Formation Commission (LAFCO). The proposed project would install 8 inch sanitary sewer lines throughout the development, totaling approximately 5,950 linear feet, to connect to a proposed 8 inch sanitary sewer line within Mowry Avenue. The wastewater flow would be directed north towards Mowry Avenue and routed to a proposed sewer pump station on Mowry Avenue, between the UPRR tracks and the project frontage. The flows would then be conveyed to the existing 8 inch sewer main on the north side of the UPRR tracks. The pump station system would provide redundant dual pump facilities, including backup generators, as required by USD for public use installations and would be designed to function independently in case of overload or mechanical failure. The sewer main extension would be constructed through a jack-and-bore operation under the UPRR tracks. The proposed project's wastewater would then be conveyed south to the Boyce Avenue pump station and ultimately to the Alvarado Treatment Plant.

Stormwater and Drainage

The proposed on- and off-site improvements would result in approximately 881,450 square feet of impervious area, and approximately 465,680 square feet of pervious area. The proposed project would install a storm drain system consisting of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes (Figure 2-10). The storm drainage system would utilize LID techniques which may include directing roof runoff to vegetated areas, storm drain stenciling, and site design that promotes infiltration. Storm drain pipes installed throughout the project site would range from 15 to 24 inches in size and would convey stormwater to the two on-site bioretention areas. The two bioretention areas would total approximately 24,665 square feet in size with 6 inches of ponding over the treatment areas. The bioretention treatment areas would be planted with water conserving grass species, shrubs, and trees that are adapted to bio-swale conditions. The bioretention treatment areas would discharge flow through the on-site storm drain system, into the adjacent City-owned open space parcel, consistent with the historic drainage path.

The proposed project would also involve off-site stormwater improvements associated with the widening of Mowry Avenue. These improvements would consist of storm drains ranging from 15 to 24 inches in size within Mowry Avenue, which would collect and convey flow towards the ACFC&WCD Line B at the terminus of Mowry Avenue. Additionally, the proposed roadway improvements would include four bioretention treatment areas along Mowry Avenue, totaling approximately 3,212 square feet.

Electricity, Gas, and Telecommunications

The proposed project would be 100 percent electric and electric services at the site would be provided by Pacific Gas and Electric Company (PG&E). Telecommunication services would be provided by AT&T and Comcast.



Draft Environmental Impact Report Project Description



Stantec

City of Newark Mowry Village Project

Figure No. **2-10** Title

Stormwater Improvements

MOWRY VILLAGE PROJECT Draft Environmental Impact Report Project Description



2.1.12 Alternative Transportation

Alameda-Contra Costa (AC) Transit provides local bus service in the East Bay and Transbay bus service to the Transbay Terminal in San Francisco. Bay Area Rapid Transit (BART) provides regional rail service connecting San Francisco, northern San Mateo County, and the East Bay. Altamont Commuter Express (ACE) and Amtrak also provide regional rail service within the San Francisco Bay Area and beyond.

Three AC Transit bus routes operate near the project site (Lines 200, 216, and 251). The closest stop to the project site for Lines 200 and 216 is at the Mowry Avenue/Cherry Street intersection located approximately 0.5 mile northeast of the project site, and the closest stop for Line 251 is at the Cherry Street/Jasmin Avenue intersection, located approximately 0.5 mile north of the project site. On weekends, the closest stop for Lines 216 and 251 is at the Silliman Recreation Center, located approximately 0.25 mile northeast of the project site. The closest BART Station to the project site is the Fremont Station, located approximately three miles north of the project site. The nearest ACE station and Amtrak station is located in the City of Fremont, on Fremont Boulevard near Peralta Boulevard, approximately 3.2 miles north of the project site.

2.2 PROJECT CONSTRUCTION

2.2.1 Construction Schedule and Equipment

It is anticipated construction of the proposed project would take approximately two years to complete, starting in Spring 2023 and ending Fall 2025. The proposed project would be completed in the following six phases:

- 1. Demolition, on-site soil remediation, and Mowry Avenue improvements
- 2. Site preparation
- 3. Grading/underground utilities
- 4. Paving
- 5. Building construction
- 6. Architectural coating

Construction of the residential development and Mowry Avenue improvements would use typical heavy construction equipment, such as backhoes, bulldozers, front loaders, scrapers, graders, excavators, concrete saws, cranes, off-road trucks, and jackhammers. No pile driving is proposed. As required by the Newark Municipal Code, Section 17.24.100 – Noise, project construction activities would occur between 7:00 A.M. and 7:00 P.M., Monday through Saturday, and between 10:00 A.M. and 6:00 P.M. on Sundays and holidays. Construction workers would access the project site from Mowry Avenue. All construction equipment and materials would be stored on-site.

2.2.2 Demolition and On-Site Soil Remediation

Prior to project construction, the existing 13,000 square foot warehouse, 1,500 square foot sales office, 3,000 square foot workshop, and large asphalt parking area servicing the Pick-n-Pull operation would be demolished and removed. Due to the site's previous use as an auto part and scrap metal salvage lot, Phase I and Phase II Environmental Site Assessments (ESA) were conducted by Haley & Aldrich, Inc in



Draft Environmental Impact Report Project Description

January 2019 to assess whether known or suspect Recognized Environmental Conditions (RECs), Historical RECs (HRECs), or Controlled RECs (CRECs) are associated with the project site. The Phase II ESA determined contaminated soils are present in shallow portions of the project site ranging from 0.5 to 1.5 feet below ground surface (bgs); however, the site soils could be remediated to levels compatible with residential use through removal (Haley & Aldrich, Inc 2019). Additionally, contaminated groundwater was detected at the project site. Soil gas samples were planned to be collected for the preparation of the Phase II ESA; however, soil gas samples could not be collected at the site due to the presence of perched groundwater. Prior to construction, the proposed project is estimated to remove approximately 39,000 cubic yards (CY) of vegetation, contaminated soil, demolition debris, and other cleared materials.

A Corrective Action Plan and Remedial Excavation Work Plan was approved by the ACWD in March 2021 (Appendix A). The Corrective Action Plan and Remedial Excavation Work Plan provides detailed plans to remediate contaminated soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater containment through in-situ remediation, and surface water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. RPS Groups, Inc. (RPS) prepared the Corrective Action Plan and Remedial Excavation and Work Plan and is proposed to oversee the implementation of the plan and remediation activities. Below are brief summaries of the remediation plan for the project site.

Soil on the project site can be remediated to residential levels through removal of shallow soil in select portions of the auto wrecking yard. The top 0.5 feet of gravel and asphalt would be removed and stockpiled on-site. Once the gravel and asphalt layer has been removed, confirmation samples would be collected at surface elevation and, consistent with the prior soil investigation, at depths of one feet and two feet bgs, and analyzed for petroleum hydrocarbons (TPH), lead, arsenic, and polycyclic aromatic hydrocarbons (PAHs) in areas previously inaccessible due to the presence of cars. Soil scraped from the excavation areas would be stockpiled on-site and characterized for disposal or on-site reuse. Once soil remediation has been completed, a groundwater monitoring network will be developed with ACWD concurrence. Wells would be installed and developed after completion of soil remediation.

In accordance with the Corrective Action Plan issued for the proposed project, the impacted groundwater would be exposed via remedial excavation and treated with the rapid chemical oxidizing agent PersolfOx as well as a time-released oxygen release compound (ORC) to treat residual levels of benzene, toluene, ethylbenzene, and xylenes (BTEX) after the excavation has been backfilled. ORC will be added to any remedial excavation that extends to groundwater elevation. Benzene in soil gas would be released during the remedial soil excavation process. RSP would obtain a permit exemption by the Bay Area Air Quality Management District (BAAQMD) for any minor atmospheric VOC released from this process.

Surface water, upon receiving a USD discharge permit, would be removed from the retention basins and piped to a USD sewer inlet in conformance with USD permit requirements. Existing stormwater retention basins would be graded and replaced with an updated stormwater conveyance system. The surface water in all three stormwater retention basins would be removed and discharged to the sanitary sewer system under the permit. After all surface water has been removed from the northernmost stormwater retention basin, the top one foot of sediment would be removed, stockpiled on-site, and characterized for off-site disposal.



Draft Environmental Impact Report Project Description

2.2.3 Grading, Fill, and Drainage

The entire 29 acre project site would be disturbed during site preparation and grading. Construction of the proposed residential development and Mowry Avenue improvements would involve the import of approximately 252,000 CY of clean fill. Clean imported fill would be used to elevate the proposed pad grades for the homes above the Federal Emergency Management Agency (FEMA) 100-year flood plain elevation. The lowest proposed pad elevation for the proposed project is 13.0 feet National Geodetic Vertical Datum (NGVD) with the average pad elevation for the proposed project at 14.2 feet NGVD. The proposed pad grades would exceed the City's standards, which state residential structures shall be "elevated to or above the base flood elevation or to a minimum of six inches above the building pad which shall be at a minimum elevation of 11.25 feet on the NGVD, whichever affords the greater degree of flood damage protection." Additionally, the site is being elevated to accommodate the San Francisco Bay Conservation and Development Commission's (BCDC) currently adopted sea level rise guidance which recommends a minimum building pad elevation of 12.2 feet. Additionally, the proposed project's pad elevation would meet the minimum pad elevation requirements of the California Ocean Protection Council Science Advisory Team's (RPC-SAT) projected likely range for sea level rise by the year 2100 of 3.4 feet. Proposed grading conforms are also proposed along the northeast property line of the project site on the City's open space parcel adjacent to the project. Clean imported fill would contain no deleterious matter or rocks greater than 4 inches in largest dimension and have Plasticity Index (PI) less than 20. Clean imported fill materials would be subject to the evaluation by the Geotechnical Engineer prior to their use. The properties of the soil with respect to corrosivity would also be evaluated.

2.3 PROJECT OBJECTIVES AND REQUIRED PROJECT APPROVALS

2.3.1 Objectives

The primary objective of the Mowry Village Project is to provide low density residential housing that incorporates multi-modal transportation for the future residents of Newark. Specific project objectives include the following:

- Implement the City's General Plan by developing the site with low-density residential;
- Support the City in meeting its Regional Housing Needs Allocation (RHNA) target assigned by the Association of Bay Area Governments (ABAG);
- Provide high quality residential development including a mix of lot sizes;
- Minimize environmental impacts associated with residential development by siting the project on developed and disturbed lands;
- Remediate contaminated soil on-site to levels suitable for residential development;
- Create a residential development that integrates multi-modal transportation (pedestrian, bicycle, automobile) and connects the development to existing, nearby bus transit stops and active centers by improving Mowry Avenue and upgrading the at-grade vehicular and pedestrian crossing along Mowry Avenue at the UPRR railroad tracks to increase safety.



Draft Environmental Impact Report Project Description

2.3.2 Approvals

The proposed project requires, but may not be limited to, the following approvals from the City:

- Rezone from Park to PD-RS-6000
- Planned Unit Development
- Specific Plan Amendment
- Vesting Tentative Map
- Design Review
- Grading, Building, and Encroachment Permit
- Alameda County LAFCO Annexation Approval

The proposed project requires, but may not be limited to, the following approvals from the other agencies:

- USD Stormwater Discharge Permit
- State Water Resources Control Board (SWRCD) National Pollutant Discharge Elimination System (NPDES) Construction General Permit

The following agencies may serve as responsible and/or trustee agencies:

- California Department of Transportation (Caltrans), District 4
- San Francisco Bay SWRCB
- Regional Water Quality Control Board (RWQCB) #2
- BAAQMD
- Alameda County LAFCO
- ACWD



3.0 ENVIRONMENTAL IMPACT ANALYSIS

Approach to Environmental Analysis

In accordance with CEQA Guidelines Section 15126.2, this Draft EIR identifies and focuses on the significant direct and indirect environmental effects of the proposed project, giving due consideration to both its short- and long-term effects. Short-term effects are generally those associated with construction of the proposed project, while long-term effects are generally those associated with operation of project components. As described in Section 1.0, Introduction, this analysis focuses on all environmental topics required to be analyzed by CEQA. Section 3.1 through 3.20 of this Draft EIR contain discussions of the potential environmental impacts related to the construction and operation of the proposed project.

Environmental Resource Topics

The potential environmental effects associated with the implementation of the proposed project are evaluated in the following environmental resource areas:

- Aesthetics
- Agricultural and Forestry Resources
- Air Quality
- Biological Resources
- Cultural Resources
- Energy
- Geology and Soils
- Greenhouse Gas Emissions
- Hazards and Hazardous Material
- Hydrology and Water Quality
- Organization of Environmental Resource Section

Each environmental resource section contains the following:

Regulatory Setting presents the laws, regulations, plans, and policies that are relevant to each resource topic. Regulations originating from the federal, state, and/or local levels are each discussed as appropriate.

Environmental Setting presents the existing environmental conditions on the project site and within the surrounding area as appropriate, in accordance with CEQA Guidelines Section 15125. The extent of the environmental setting area evaluated (the project study area) differs among resources, depending on the



- Land Use and Planning
- Mineral Resources
- Noise
- Population and Housing
- Public Services
- Recreation
- Transportation
- Tribal Cultural Resources
- Utilities and Service Systems
- Wildfire

Draft Environmental Impact Report Environmental Impact Analysis

locations where impacts would be expected. For example, air quality impacts are assessed for the air basin (macro-scale), as well as the site vicinity (micro-scale), whereas aesthetic impacts are assessed for the project vicinity only.

In determining the level of significance of environmental impacts associated with the proposed project, the analysis in this Draft EIR assumes that the proposed project would comply with relevant federal and state laws and regulations, and City of Newark General Plan policies, ordinances, and other adopted City documents, unless otherwise noted. Therefore, such mandatory policies, ordinances, and standards are not identified as mitigation measures, but rather are discussed as part of the "regulatory setting" governing the proposed project.

Thresholds of Significance identifies the thresholds of significance used to determine the level of significance of the environmental impacts for each resource topic, in accordance with CEQA Guidelines Sections 15126, 15126.2, and 15143. The thresholds of significance used in this Draft EIR are based on the checklist presented in Appendix G of the CEQA Guidelines; best available data; and regulatory standards of federal, state, and local agencies.

Project Impacts identify the level of each environmental impact by comparing the effects of the proposed project to the environmental setting. Key methods and assumptions used to frame and conduct the impact analysis, as well as issues or potential impacts not discussed further (i.e., such issues for which the project would have no impact), are also described.

Project impacts are organized numerically in each subsection (e.g., Impact AES-1, Impact AES-2, Impact AES-3). A bold-font environmental impact statement precedes the discussion of each impact while its level of significance succeeds the discussion of each impact. The discussion that follows the impact summary includes the substantial evidence supporting the impact significance conclusion.

Mitigation Measures describe any feasible measures that could rectify, reduce, or compensate for significant adverse impacts, with measures having to be fully enforceable through incorporation into the project (PRC Section 21081.6[b]). Mitigation measures are not required for environmental impacts that are found to be less than significant. Where feasible mitigation for a significant environmental impact is available, it is described following the impact. Where sufficient feasible mitigation is not available to reduce environmental impacts to a less than significant level, or where the lead agency lacks the authority to ensure that the mitigation is implemented when needed, the impacts are identified as significant and unavoidable.

Level of Significance After Mitigation describes the level of impact significance remaining after mitigation measures are implemented.

Level of Significance

Determining the severity of project impacts is fundamental to achieving the objectives of CEQA. CEQA Guidelines Section 15091 requires that decision makers mitigate, to the maximum extent feasible, the significant impacts identified in the Final EIR. If the EIR identifies any significant unmitigated impacts, CEQA Guidelines Section 15093 requires decision-makers to balance the economic, legal, social, technological or other benefits of a proposed project against its unavoidable environmental risks when determining whether to approve the project. If a lead agency approves a project which will result in the



Draft Environmental Impact Report Environmental Impact Analysis

occurrence of significant effects which are identified in the final EIR but are not avoided or substantially lessened, the decision-maker must adopt a statement of overriding considerations that explains why the benefits of the project outweigh the adverse environmental consequences identified in the EIR.

The level of significance for each impact examined in this Draft EIR is determined by considering the predicted magnitude of the impact against the applicable threshold. Thresholds were developed using criteria from the CEQA Guidelines and Appendix G Checklist; federal, state, and local regulatory schemes; regional and local plans and ordinances; accepted practice; consultation with recognized experts; and other professional opinions.

Format Used for Impact Analysis and Mitigation Measures

The format adopted in this Draft EIR to present the evaluation of environmental impacts is described and illustrated below.

Summary Heading of Impact

Impact AIR-1: An impact summary heading appears immediately preceding the impact description (Summary Heading of Impact in this example). The impact abbreviation identifies the section of the report (AIR for Air Quality in this example) and the sequential order of the impact (1 in this example) within that section. To the right of the impact number is the impact statement, which identifies the potential impact.

Impact Analysis

A narrative analysis follows the impact statement.

Level of Significance Before Mitigation

This section identifies the level of significance of the impact before any mitigation is proposed.

Mitigation Measures

In some cases, following the impact discussion, reference is made to federal and state regulations and agency policies that would fully or partially mitigate the impact. In addition, policies and programs from applicable local land use plans that partially or fully mitigate the impact may be cited.

Project-specific mitigation measures, beyond those contained in other documents, are set off with a summary heading and described using the format presented below:

MM AIR-1: Project-specific mitigation is identified that would reduce the impact to the lowest degree feasible. The mitigation number links the particular mitigation to the impact with which it is associated (AIR-1 in this example);

Abbreviations used in the mitigation measure numbering are shown in Table 3.0-1.

Table 3.0-1: Environmental Resource Abbreviations

Code	Environmental Issue
AES	Aesthetics
AG	Agricultural and Forestry Resources



Draft Environmental Impact Report Environmental Impact Analysis

Code	Environmental Issue
AIR	Air Quality
BIO	Biological Resources
CUL	Cultural and Historical Resources
EN	Energy
GEO	Geology and Soils
GHG	Greenhouse Gas Emissions and Climate Change
HAZ	Hazards and Hazardous Materials
HYD	Hydrology and Water Quality
LU	Land Use and Planning
MIN	Mineral Resources
NOI	Noise and Vibration
POP	Population and Housing
PUB	Public Services
REC	Recreation
TRANS	Transportation
TRIB	Tribal Cultural Resources
UTIL	Utilities and Service Systems
WF	Wildfire

Level of Significance After Mitigation

This section identifies the resulting level of significance of the impact following mitigation.

3.1 **AESTHETICS**

This section describes the environmental and regulatory setting for aesthetics. It also describes existing conditions and potential impacts related to aesthetics that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.1.1 Environmental Setting

Regional Visual Character

The City of Newark contains a diverse variety of neighborhoods and districts, including a range of established and planned residential neighborhoods, commercial districts, business parks, industrial facilities, and the Don Edwards National Wildlife Refuge. The Southwest Newark Residential and Recreational Focus Area, where the project site is located, covers an area of approximately 637 acres. Its western portion largely consists of vacant fields with some existing industrial development. A panoramic view of the East Bay Hills to the southeast and south is available from vantage points in the vicinity of the Sportsfield Park. Most of the western portion of the Southwest Newark Residential and Recreational Area is flat, vacant land with little vegetation, except some emergent wetland areas, particularly near Mowry Slough which runs through a portion of the focus area and along its western boundaries. The City does not contain any officially designated scenic corridors or highways (City of Newark 2013b).

There are no officially designated scenic vistas or view corridors in the City. However, views of the undeveloped Coyote Hills to the northwest, of Mission Peak to the east, and the East Bay Hills to the east and southeast are available from open spaces within the City. Additionally, views of low-lying wetlands fronting San Francisco Bay are available from vantage points along the western perimeter of Newark (City of Newark 2013b).

Project Site Visual Character

The proposed project is located on an approximately 29 acres partially developed site. The southern portion of the site is currently being operated by Pick-n-Pull Auto Dismantlers as an auto wrecking and scrap metal salvage yard. The northern portion of the site is vacant and covered with wild grasses. The northern parcel of the site is roughly triangular in shape and is undeveloped. The surface elevation is approximately 10 feet around the perimeter of the site is slightly sloping with an approximate elevation of 10 feet on the west and 4.5 feet on the east. There are no existing developments surrounding the site and the project site does not contain any General Plan designated scenic resources.

Light and Glare Conditions

The urbanized portion of Newark between the freeways and along major arterial roads are developed with a mix of commercial, industrial, and residential uses. Existing sources of light and glare in the City are similar to those found in any urbanized area, and include streetlamps, parking lot lighting, storefront and signage lighting, and car headlamps.

Though the project site is developed with existing uses, no substantial light and glare sources exist onsite or along adjacent lands. Lighting at the Pick-n-Pull site consists of outdoor lighting located on existing



structures. There is currently no street lighting located along the portion of Mowry Avenue in front of the project site and vehicle headlight glare is limited in the area, as this section of Mowry Avenue does not have a lot of vehicle traffic. There are no electrical signs, billboards, or flashing or oscillating light sources at the project site.

3.1.2 Regulatory Setting

State

California State Scenic Highway Program

The California Scenic Highway Program, maintained by Caltrans, protects scenic state highway corridors from changes which would diminish the aesthetic value of lands adjacent to the highways. There are no officially designated or eligible State scenic highways within or adjacent to the project site.

California Energy Code Title 24 Parts 1 and 6 - Outdoor Lighting Zones

In 2001, the California Legislature passed a bill requiring the California Energy Commission (CEC) to adopt energy efficient standards for outdoor lighting for both the public and private sector. In November 2003, the CEC adopted changes to the Building Energy Efficient Standards within Title 24. These standards became effective on October 1, 2005, and specify outdoor lighting requirements for residential and nonresidential development. The intent of the new standards is to improve the quality of outdoor lighting and help reduce the impacts of light pollution, light trespass, and glare. The standards regulate lighting on and off. Different lighting standards are set by classifying areas by lighting zone. The classification is based on population figures in the 2003 Census and the areas can be designated as LZ1 (dark), LZ2 (low), LZ3 (medium), or LZ4 (high). Lighting requirements for dark and rural areas are stricter in order to protect the areas from new sources of light pollution and light trespass.

Local

City of Newark General Plan

The following lists goals and policies from the City of Newark General Plan pertaining to aesthetics that are applicable to the proposed project.

Goal LU-1: Quality of Life. Maintain a desirable quality of life in Newark by preserving a small town, neighborhood-oriented atmosphere and sustaining a balanced mix of land uses.

• **Policy LU-1.12: Large Scale Development.** Plan and design Newark's remaining large-scale development sites in a manner which sensitively integrates these areas with existing uses and adjacent neighborhoods, and which includes a mix of uses that makes these areas more conductive to walking, bicycling, and transit use.

Goal LU-2: Land Use Compatibility. Ensure compatibility between adjacent land uses.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy LU-2.2: Context-Sensitive Design.** Require that new structures, additions, and major renovations are aesthetically compatible with existing structures and the surrounding context and contribute positively to the visual quality of neighborhoods.
- **Policy LU-2.4: Buffering from Transportation Facilities.** Ensure that the design of new residential development near rail lines, truck routes, freeways, or major thoroughfares includes setbacks, landscape screening, and other provisions to minimize exposure to negative impacts such as noise and air pollution.
- **Policy LU-2.7: Design Guidelines.** Maintain design guidelines and a design review process that applies to building and site design throughout the city.

Goal LU-4: Community Design and Identity. Enhance Newark's identity as a city of high quality development that is distinctive from other cities in the Bay Area.

- **Policy LU-4.5: Gateways.** Maintain high standards for the design and appearance of development at major gateways into Newark, and along major arterials. Public art, landscaping, paving, lighting, and signage should be used to create a positive visual impression at these locations.
- **Policy LU-4.6: Streetscapes.** Ensure that medians, sidewalks, planting strips and other areas within the right-of-way of major thoroughfares are attractively landscaped and well maintained.
- **Policy LU-4.7: Lighting.** Manage exterior lighting to reduce potential light and glare impacts, improve public safety, and enhance the character of the streetscape.
- Policy LU-4.13: Bayfront Identity. Reinforce Newark's identity as a bayfront city by orienting new development on the western and southern edges of the city toward the bay and shoreline areas. Future projects in these areas should enhance views to the water and wetlands and be compatible with the area's scenic and recreational qualities. The bayfront identity should be emphasized in gateways and public art as well.
- **Policy LU-4.14: View Protection.** Protect and enhance panoramic views and vistas of horizon features such as Coyote Hills, Mission Peak, the East Bay and Peninsula Hills, and San Francisco Bay.

Goal LU-7: Southwest Newark Residential and Recreational Project. Develop the Southwest Newark Residential and Recreational Project as one of the Silicon Valley's premier new neighborhoods, with executive housing and high quality recreation.

• Policy LU-7.1: Southwest Newark Residential and Recreational Project (Area 3 and 4 Development). Facilitate the development of the 637 acres formerly known as "The Area 3 and 4 project" consistent with previously approved plans for this area. The residential holding capacity of this area shall be 1,260 units.



Draft Environmental Impact Report Environmental Impact Analysis

Areas 3 and 4 Specific Plan

The Areas 3 and 4 Specific Plan contains siting standards and architectural design guidelines which apply to the Southwest Newark Residential and Recreational Focus Area. The standards and guidelines provide guidance on heights, setbacks, lot coverage, architectural theme, street design, and landscaping.

For single-family lots, five lot sizes are contemplated in the Specific Plan along with attached or stacked housing. No criteria are provided for custom lots since they will be irregularly shaped. The Community Development Director will establish appropriate setbacks during site and architectural review of a particular home on a custom or irregular lot. All residential streets will be standard City of Newark "Type I – Minor Streets" with a 56-foot right-of-way. Cul-de-sacs will be designed per City standards as well with a curb radius of 45 feet and right-of-way radius of 50 feet.

Newark Zoning Code

The Newark Zoning Code identifies development standards for the various zoning districts throughout the City, and describes the purpose, intent, and uses allowed in each zoning district. The relevant development standards related to visual resources in the City include building coverage, height, setbacks, and lighting. The development standards for residential districts are provided under Chapter 17.07.030 of the City's Zoning Code.

Chapter 17.34, Design Review

Chapter 17.34 of the Newark Municipal Code establishes the design review procedure to ensure that new development supports the goals and objectives of the general plan and other adopted plans and guidelines. Design review is required for all projects that require a permit for new construction, reconstruction, rehabilitation, alteration, or other improvements to the exterior of a structure, site, or parking area. The Planning Commission has the design review authority for all projects requiring planning commission approval such as conditional use permits and variances.

3.1.3 Environmental Impacts

This section analyzes the project's potential to result in significant aesthetics impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

Analysis of the proposed project's visual impacts is based on an evaluation of the changes to the existing visual resources that would result from implementation of the proposed project. In determining the extent and implications of the visual changes, consideration was given to: the existing visual quality of the affected environment; specific changes in the visual character and quality of the affected environment; the extent to which the affected environment contains places or features that provide unique visual experiences or that have been designated in plans and policies for protection or special consideration; and the sensitivity of viewers and their activities and the extent to which these activities are related to the aesthetic qualities affected by the proposed project.



Draft Environmental Impact Report Environmental Impact Analysis

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's aesthetic impacts are significant. Would the proposed project:

- Have a substantial adverse effect on a scenic vista?
- Substantially damage scenic resources, including but not limited to trees, rock outcroppings, and historic buildings within a state scenic highway?
- In non-urbanized areas, substantially degrade the existing visual character or quality of the public views of the site and its surroundings. (Public views are those that are experienced from publicly accessible vantage point). If the project is in an urbanized area, would the project conflict with applicable zoning and other regulations governing scenic quality?
- Create a new source of substantial light or glare that would adversely affect day- or nighttime views in the area?

Project Impact Analysis and Mitigation Measures

Scenic Vista

Impact AES-1 The proposed project would not have a substantial adverse effect on a scenic vista.

Impact Analysis

The General Plan EIR indicates that there are no officially designated scenic vistas or view corridors in the City and the General Plan does not identify any specific vistas or views for special protection in the future (City of Newark 2013b). From the project site, there are views of Mission Peak to the east and northeast of the project site. Additionally, views of hills located across the bay are available from the project site. The proposed project would construct 203 new single-family residences which would be two stories tall. Given the amount of separation between the project site and these views (across the bay), and no official designated scenic vistas or view corridors being located in the City, the development of the proposed project would not substantially alter views and the proposed project would not have a substantial adverse effect on a scenic vista. Therefore, there would be no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Scenic Resources within a State Scenic Highway

Impact AES-2 The proposed project would not substantially damage scenic resources, including, but not limited to, trees, rock outcroppings, and historic buildings within a State scenic highway.

Impact Analysis

According to the General Plan EIR and the Caltrans California State Scenic Highway System Map, there are no officially designated state scenic highways within the City or near the project site (Caltrans 2021). The closest officially designated state scenic highway is a section of Interstate 680 located approximately 5.5 miles northeast of the project site. The project site does not contain vegetation, trees, rock outcroppings, or historic buildings that are identified as a scenic resource by the General Plan. Therefore, the proposed project would not substantially damage scenic resources within a state scenic highway and there would be no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Visual Character and Scenic Quality

Impact AES-3 In an urbanized area, the proposed project would not conflict with applicable zoning and other regulations governing scenic quality.

Impact Analysis

The Newark Zoning Code designates the project site as Park. However, the proposed project includes rezoning of the project site from Park to PD-RS-6000 to better align the zoning designation of the project site with the proposed use of the project and the existing Low Density Residential General Plan land use designation. Due to the minimum lot size and setbacks prescribed in the RS-6000 zoning district, a Planned Unit Development is being proposed to allow a deviation from the standards listed in the RS-6000 development standards in the Newark Zoning Code. The proposed project would be designed and constructed in accordance with all applicable regulations governing scenic quality to ensure that the proposed project is consistent with the proposed zoning district. The proposed project would be reviewed during the City's design review and project review process to ensure that the proposed project does not conflict with applicable zoning standards and requirements. The proposed project would comply with the City's Zoning Code provisions related to scenic quality including Citywide regulation related to landscaping described under Chapter 17.21 of the City's Zoning Code and the lighting and illumination regulations described under Chapter 17.17.060 of the City's Zoning Code, As such, with the approval of rezoning and compliance with the City's Zoning Code regulations related to scenic guality, the proposed project would not conflict with any applicable zoning or other regulations governing scenic quality, and impacts would be less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Light and Glare

Impact AES-4	The proposed project would not create a new source of substantial light or glare
	which would adversely affect day or nighttime views in the area.

Impact Analysis

The project site is developed with an auto wrecking and scrap metal salvage yard and sources of light and glare exists at the project site. Existing sources of lighting at the project site include outdoor lighting located on existing structures. There is no existing street lighting along Mowry Avenue and vehicle lights and glare from passing cars is limited due to the location of the project site and low volume of traffic.

The proposed project would include new sources of nighttime lighting at the project site. The proposed project would provide exterior lighting throughout the site to illuminate the main entrances of the singlefamily homes, private streets, sidewalks, common space areas, and driveways for security and safety purposes. Additional lighting would be installed along the project frontage on Mowry Avenue. All site entrances will be visible from a public street and well lit. The proposed project would be required to comply with General Plan Policy LU-4.7 which requires management of exterior lighting to reduce potential light and glare impacts. The proposed project would install lighting according to the City's lighting standards and requirements as outlined in the Newark Municipal Code Section 17.17.060 and would include shielding to ensure light spillage does not occur on adjacent properties. The proposed project would also introduce new sources of glare at the site through the construction of new structures and increased vehicle usage of the area; however, the proposed project would minimize the use of reflective exterior building materials to minimize new sources of glare at the site. Reflective surfaces within the new detached single-family homes would be generally limited to window glazing and the traditional architectural styles proposed within this project would not result in any imbalanced window to wall ratios that could create new glare impacts. The proposed project would be required to comply with Newark Municipal Code Section 17.24.070 which requires that no use shall be operated such that significant, direct glare is visible beyond the boundaries of the lot where the use is located. Compliance with the City's requirements would ensure that light and glare impacts associated with the proposed project would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis
Draft Environmental Impact Report Environmental Impact Analysis

3.2 AGRICULTURAL AND FORESTRY RESOURCES

This section describes the environmental and regulatory setting for agricultural and forestry resources. It also describes existing conditions and potential impacts related to agricultural and forestry resources that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.2.1 Environmental Setting

The City of Newark is located near the southern part of San Francisco Bay in a highly urbanized area of Alameda County. According to the General Plan EIR, there are no forestry resources or timberland resource zones in the City or the surrounding area, and there is no active timberland production in the general vicinity of the City (City of Newark 2013b). There are no lands classified as Prime Farmland, Unique Farmland, or Farmland of Statewide Importance as mapped by the California Resources Agency within the City. However, the General Plan EIR does identify some parcels under Williamson Act contracts within the City. Parcels under Williamson Act contracts within the City are comprised of approximately 3,000 acres of privately owned properties used for salt harvesting.

The California Department of Conservation (DOC) Farmland Mapping and Monitoring Program (FMMP) classifies the project site as Urban and Built-Up Land (DOC 2021). The project site is designated as Low Density Residential by the General Plan and is zoned Park by the Newark Zoning Code. The proposed project includes rezoning the project site from Park to PD-RS-6000. Additionally, the project site is within the Areas 3 and 4 Specific Plan area and is designated for a golf course or other recreational uses by the Specific Plan. Therefore, the proposed project also includes a Specific Plan Amendment to change the use to residential single-family. While the Specific Plan land use designation and zoning would change following approval of the proposed project, the Specific Plan amendment and rezone would better align the project site's allowed uses with the General Plan designation.

3.2.2 Regulatory Setting

Federal

Natural Resources Conservation Service (NRCS)

The NRCS maps soils and farmland uses to provide comprehensive information necessary for understanding, managing, conserving, and sustaining the country's limited soil resources. The NRCS manages the Farmland Protection Program, which provides funds to help purchase development rights to keep productive farmland in agricultural uses.

State

California Department of Conservation

In 1982, the State of California created the FMMP within the DOC to continue the mapping activity of the NRCS. The DOC administers the California Land Conversion Act of 1965. Also known as the Williamson Act.



Draft Environmental Impact Report Environmental Impact Analysis

California Land Conversion Act (Williamson Act)

The Williamson Act is the only established agricultural program that directly involves state government in an administrative or fiscal capacity. The Williamson Act creates an arrangement whereby private landowners voluntarily restrict their land to agricultural and compatible open space uses under a rolling two-year contract through the Farmland Security Zones provisions. In return, parcels are assessed for property tax purposes at a rate consistent with their actual use, rather than their potential market value.

Local

City of Newark General Plan

The City's General Plan does not include any goals or policies related to agricultural and forestry resources.

3.2.3 Environmental Impacts

This section analyzes the project's potential to result in significant agricultural and forestry impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of documents pertaining to the project site, including the General Plan, General Plan EIR, and the DOC Important Farmland Map.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's agricultural and forestry impacts are significant. Would the proposed project:

- Convert Prime Farmland, Unique Farmland, or Farmland of Statewide Importance (Farmland), as shown on the maps prepared pursuant to the Farmland Mapping and Monitoring Program of the California Resources Agency, to non-agricultural use?
- Conflict with existing zoning for agricultural use, or a Williamson Act contract?
- Conflict with existing zoning for, or cause rezoning of, forest land (as defined in PRC section 12220(g)), timberland (as defined by PRC section 4526), or timberland zoned Timberland Production (as defined by Government Code section 51104(g))?
- Result in the loss of forest land or conversion of forest land to non-forest use?
- Involve other changes in the existing environment which, due to their location or nature, could result in conversion of Farmland, to non-agricultural use or conversion of forest land to non-forest use?



Project Impact Analysis and Mitigation Measures

		الم منتقر ا
Convert	Farm	lana

Impact AG-1	The proposed project would not convert Prime Farmland, Unique Farmland, or
	Farmland of Statewide Importance (Farmland), as shown on the maps prepared
	pursuant to the Farmland Mapping and Monitoring Program of the California
	Resources Agency, to non-agricultural use.

Impact Analysis

As described above under Section 3.2.1, Environmental Settings, the project site is designated by the DOC's FMMP as Urban and Built-Up Land and is designated Low Density Residential by the City's General Plan. The project site is not designated or zoned as farmland and does not allow for agricultural uses. The proposed project includes a Specific Plan Amendment and rezone to develop the site for residential uses. While the Specific Plan designation and zoning would change following approval of the proposed project, the proposed project would not convert farmland to non-agricultural uses. Therefore, the proposed project would not result in the conversion of Prime Farmland, Unique Farmland, or Farmland of Statewide Importance to non-agricultural uses and no impact would occur.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Agricultural Zoning

Impact AG-2 The proposed project would not conflict with existing zoning for agricultural use or a Williamson Act contract.

Impact Analysis

The project site is currently zoned Park and is not under a Williamson Act contract. The proposed project includes rezoning the project site from Park to PD-RS-6000. Neither the Park nor RS-6000 zoning districts allow for agricultural uses. Therefore, the project site is not zoned for agricultural uses and the proposed project would not conflict with existing zoning for agricultural use or a Williamson Act contract. There would be no impacts from the proposed project.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Forest Land or Timberland Zoning

Impact AG-3	The proposed project would not conflict with existing zoning for, or cause
	rezoning of, forestland (as defined in Public Resources Code section 12220[g]),
	timberland (as defined by Public Resources Code section 4526), or timberland
	zoned Timberland Production (as defined by Government Code section
	51104[g]).

Impact Analysis

The project site does not contain forestland (as defined in PRC Section 12220[g]), or timberland (as defined by PRC Section 4526). Furthermore, the project site is not zoned Timberland Production (as defined by Government Code section 51104[g]). The project site would not require rezoning of forestland or timberland production. Therefore, the proposed project would not conflict with existing zoning or cause rezoning of forestland or timberland and there would be no impacts.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Loss or Conversion of Forest Land

Impact AG-4 The proposed project would not result in the loss of forestland or conversion of forestland to non-forest use.

Impact Analysis

The project site is designated Low Density Residential by the General Plan and there are no forestland resources that exist on the project site. Therefore, the proposed project would not result in the loss of forestland or conversion of forestland to non-forest use and no impact would occur.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Change to Existing Environment

Impact AG-5 The proposed project would not involve other changes in the existing environment which, due to their location or nature, could result in conversion of Farmland to non-agricultural use or conversion of forestland to non-forest use.

Impact Analysis

The project site is classified as Urban and Built-Up Land by the DOC and surrounding areas are classified as Other Land or Grazing Land. The project site is designated by the City's General Plan as Low Density Residential and lands surrounding the project site are designated as Low Density Residential; Park and Recreation Facilities; and Salt Harvesting, Refining and Production. Lands located to the west of the project site are used for salt production ponds. The project site is generally surrounded to the northeast and east by land that was formerly used for agriculture, however, the parcels are currently being used by the City as open space. The area surrounding the project site is not under agricultural use and therefore, the proposed project would not cause changes to the existing environment that could result in conversion of farmland outside the project site boundary to non-agricultural use and there would be no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.3 AIR QUALITY

This section describes the environmental and regulatory setting for air quality. It also describes existing conditions and potential impacts related to air quality that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.3.1 Environmental Setting

San Francisco Bay Area Air Basin

The proposed project is in Alameda County within the San Francisco Bay Area Air Basin (SFBAAB), which is under the jurisdiction of the BAAQMD. The SFBAAB comprises all of Alameda, Contra Costa, Marin, Napa, San Francisco, San Mateo, and Santa Clara counties, southern Sonoma County, and the southwestern portion of Solano County. Air quality in this area is determined by such natural factors as topography, meteorology, and climate, in addition to the presence of existing air pollution sources and ambient conditions.

Climate and Meteorology

The regional climate within the San Francisco Bay Area is driven by a summertime high-pressure cell centered over the northeastern Pacific Ocean that dominates the summer climate of the West Coast. The persistence of this high-pressure cell generally results in negligible precipitation during the summer, and meteorological conditions are typically stable with a steady northwesterly wind flow. This flow causes upwelling of cold ocean water from below the surface, which produces a band of cold water off the California coast. The cool and moisture-laden air approaching the coast from the Pacific Ocean is further cooled by the presence of the cold-water band, resulting in condensation and the presence of fog and stratus clouds along the Northern California coast. In the winter, the Pacific high-pressure cell weakens and shifts to the south, resulting in wind flows offshore, the absence of upwelling, and an increase in the occurrence of storms. Winter stagnation episodes are characterized by nocturnal drainage wind flows in coastal valleys. Drainage is a reversal of the usual daytime air-flow patterns; air moves from the Central Valley toward the coast and back down toward the Bay from the smaller valleys within the Air Basin.

Air Pollutants of Concern

Criteria air pollutants are defined as pollutants for which the federal and state governments have established ambient air quality standards for outdoor concentrations. The federal and state standards have been set at levels above which concentrations could be harmful to human health and welfare. These standards are designed to protect the most sensitive persons such as children, pregnant women, and the elderly, from illness or discomfort. Criteria air pollutants include ozone (O₃), nitrogen dioxide (NO₂), carbon monoxide (CO), sulfur dioxide (SO₂), particulate matter 2.5 microns or less in diameter (PM_{2.5}), particulate matter ten microns or less in diameter (PM₁₀), and lead (Pb). Note that Reactive Organic Gases (ROGs), which are also known as reactive organic compounds (ROCs) or volatile organic compounds (VOCs), and nitrogen oxide (NOx) are not classified as criteria pollutants. However, ROGs and NOx are widely emitted from land development projects and participate in photochemical reactions in the atmosphere to form O₃; therefore, NOx and ROGs are relevant to the proposed project and are of concern in the air basin and are listed below along with the criteria pollutants (USEPA 2022).



Draft Environmental Impact Report Environmental Impact Analysis

- **Ozone.** O₃ is a gas that is formed when NOx and ROGs, both byproducts of internal combustion engine exhaust and other sources, undergo slow photochemical reactions in the presence of sunlight. Ozone concentrations are generally highest during the summer months when the combination of direct sunlight, light wind, and warm temperature conditions create conditions favorable to the formation of this pollutant. Exposure to O₃ can cause throat irritation, damage to the airways, increase susceptibility to lung infection, aggravate lung diseases (i.e. asthma, emphysema, and chronic bronchitis), and increase the frequency of asthma attacks.
- **Reactive Organic Gases.** ROGs are compounds composed primarily of atoms of hydrogen and carbon. Internal combustion associated with motor vehicle usage is the major source of these hydrocarbons. Adverse effects on human health are not caused directly by ROGs, but rather by reactions of ROGs to form secondary air pollutants, including ozone. ROG exposure can result in eye, nose, and throat irritation, headaches, and damage to the liver, kidney, and nervous system. The health impacts from ROG varies greatly based on the toxicity of the chemical.
- Nitrogen Dioxide and Nitrogen Oxides. Fuel combustion produces nitrogen which combines with oxygen to produce nitric oxide (NO). Further oxidation of NO results in the formation of NO₂, which is a criteria pollutant. NO₂ is a reddish-brown, highly reactive gas which acts as an acute irritant and, in equal concentrations, is more injurious than NO. NO and NO₂ are referred to together as oxides of nitrogen. As noted above, NO_x is involved in photochemical reactions that produce ozone. NOx exposure can also result in severe respiratory problems. Short-term NOx exposure can irritate the airways and aggravate respiratory symptoms while long-term exposure may contribute to the development of asthma and increased susceptibility to respiratory infections.
- **Carbon Monoxide.** CO is a colorless, odorless gas produced by the incomplete combustion of fuels. CO concentrations tend to be the highest during winter mornings, with little to no wind, when surface-based inversions trap the pollutant at ground levels. Because CO is emitted directly from internal combustion engines and motor vehicles operating at slow speeds, the highest ambient CO concentrations are generally found near congested transportation corridors and intersections. CO exposure can reduce the amount of oxygen that is transported in the blood stream to critical organs. High levels of exposure, which are possible in enclosed spaces, can dizziness, confusion, unconsciousness, and death.
- **Sulfur dioxide.** SO₂ is a colorless, extremely irritating gas or liquid. It enters the atmosphere as a pollutant mainly as a result of burning high-sulfur-content fuel oils and coal and from chemical processes occurring at chemical plants and refineries. When sulfur dioxide oxidizes in the atmosphere, it forms sulfates (SO₄). People with asthma, particularly children, are sensitive to the health effects of SO₂. Exposure to SO₂ can harm the human respiratory system and make breathing difficult.
- Respirable Particulate Matter. PM₁₀ consists of extremely small, suspended particles or droplets 10 microns or smaller in diameter. Some sources of PM₁₀, like pollen and windstorms, are naturally occurring. However, in populated areas, most PM₁₀ is caused by road dust, diesel soot, and combustion products, abrasion of tires and brakes, and construction activities. PM₁₀ exposure can lead to premature death in people with heat or lung disease, nonfatal heart attacks,



Draft Environmental Impact Report Environmental Impact Analysis

irregular heartbeat, aggravated asthma, decreased lung function, and increased respiratory symptoms.

- Fine Particulate Matter. PM_{2.5} refers to particulate matter that is 2.5 microns or smaller in size. The sources of PM_{2.5} include fuel combustion from automobiles, power plants, wood burning, industrial processes, and diesel-powered vehicles such as buses and trucks. These fine particles are also formed in the atmosphere when gases such as sulfur dioxide, NO_x, and VOCs are transformed in the air by chemical reactions. PM_{2.5} has the same health risks can PM₁₀ exposure. However, PM_{2.5} has increased levels of risk as particulates 2.5 microns or smaller can penetrate deeper into the lungs.
- Lead. Pb occurs in the atmosphere as particulate matter. The combustion of leaded gasoline is the primary source of airborne lead in the Basin. The use of leaded gasoline is no longer permitted for on-road motor vehicles, so most such combustion emissions are associated with off-road vehicles such as racecars that use leaded gasoline. Other sources of Pb include the manufacturing and recycling of batteries, paint, ink, ceramics, ammunition, and secondary lead smelters. Lead exposure can cause health impacts to almost every organ and system in the human body. Once exposure, lead is circulated through the blood and stored in bones along with calcium. Depending on the level of exposure, health effects of lead in adults include cardiovascular effects, decreased kidney function, and reproductive problems in men and women. Exposure to children may result in behavior and learning problems, lower IQ, anemia, and slowed growth.

Ambient Air Quality

Each year, BAAQMD summarizes data collected from the Bay Area air quality monitoring stations. The nearest air quality monitoring station to the project is Hayward-La Mesa at 3466 La Mesa Drive, located approximately 10 miles northwest of the project site. Table 3.3-1 includes a summary of the air quality monitoring data from 2019 to 2021. The table shows the number of times each station recorded pollutant concentrations above federal and state air quality standards and the highest annual reading for each pollutant. The Hayward-La Mesa Station only monitors 1-hour and 8-hour ozone.

Pollutant	Air Pollutant, Averaging Time (Units)	2019	2020	2021
Ozone	Maximum 1-hour	0.106	0.116	0.097
(ppm)	California 1-hour number of days over standard	2	3	1
	Maximum 8-hour	0.085	0.092	0.082
	National 8-hour number of days over standard	2	4	3
	California 8-hour number of days over standard	2	5	3

Table 3.3-1: Hayward-La Mesa Air Quality Monitoring Station Data

Notes: ppm = parts per million Source: CARB 2022



Draft Environmental Impact Report Environmental Impact Analysis

Toxic Air Contaminants

In addition to the criteria pollutants discussed above, toxic air contaminants (TACs) are another group of pollutants of concern. TACs are considered either carcinogenic or noncarcinogenic based on the nature of the health effects associated with exposure to the pollutant. For regulatory purposes, carcinogenic TACs are assumed to have no safe threshold and health impacts are assumed to occur at any level. Cancer risks are expressed as excess cancer cases per one million. Noncarcinogenic TACs differ in that there is generally assumed to be a safe level of exposure below which no negative health impact is believed to occur. These levels are determined on a pollutant-by-pollutant basis.

The California Air Resources Board (CARB) has designated 244 compounds as TACs. CARB has implemented control measures for a number of compounds that pose high risks and show potential for effective control. The majority of the estimated health risks from TACs can be attributed to relatively few compounds. Diesel particulate matter (DPM) differs from other TACs in that it is not a single substance but rather a complex mixture of hundreds of substances, including 40 cancer-causing substances. Diesel exhaust is a complex mixture of particulates and gases produces when an engine burns diesel fuel. DPM is a concern because it causes lung cancer; many compounds found in diesel exhaust are carcinogenic.

Sensitive Receptors

Some land uses are considered more sensitive to air pollution than others due to the types of population groups or activities involved. Heightened sensitivity may be caused by health problems, proximity to the emissions source, or duration of exposure to air pollutants. Children, pregnant women, the elderly, and those with existing health problems are especially vulnerable to the effects of air pollution. Accordingly, land uses that are typically considered to be sensitive receptors include residences, schools, childcare centers, playgrounds, retirement homes, convalescent homes, hospitals, and medical clinics.

The nearest sensitive receptor is the Silliman Center Preschool which lies approximately 1,100 feet north of the site (and 100 feet east of the roadway improvements along Mowry Avenue). Other nearby receptors are in a residential development over 1,000 feet away to the northeast of the project site.

3.3.2 Regulatory Setting

Federal

Clean Air Act

The Clean Air Act (CAA) of 1970 and the CAA Amendments of 1971 required the U.S. Environmental Protection Agency (USEPA) to establish national ambient air quality standards (NAAQS), with requires retaining the option to adopt more stringent standards or to include other specific pollutants. On April 2, 2007, the Supreme Court found that carbon dioxide is an air pollutant covered by the CAA; however, no NAAQS have been established for carbon dioxide.

These standards are the levels of air quality considered safe, with an adequate margin of safety, to protect the public health and welfare. They are designed to protect those "sensitive receptors" most susceptible to further respiratory distress such as asthmatics, the elderly, very young children, people already weakened by other disease or illness, and persons engaged in strenuous work or exercise.



Draft Environmental Impact Report Environmental Impact Analysis

Healthy adults can tolerate occasional exposure to air pollutant concentrations considerably above these minimum standards before adverse effects are observed.

The federal and state ambient air quality standards are listed below in Table 3.3-2, and the attainment status for the criteria pollutants are listed in Table 3.3-3.

	Averaging	California Standards	s National Standards		
Pollutant	Time	Concentration	Primary	Secondary	
	1 hour	0.09 ppm (180 µg/m ³)	—	Somo oo primoru	
Ozone	8 hour	0.070 ppm (137 µg/m³)	0.070 ppm (137 μg/m³)	standard	
Despirable	24 hour	50 µg/m³	150 μg/m³	Somo oo primoru	
particulate matter	Annual arithmetic mean	20 µg/m³	_	same as primary standard	
Fine perticulate	24 hour	—	35 µg/m³	Somo oo primoru	
matter	Annual arithmetic mean	12 µg/m³	12 µg/m³	standard	
	1 hour	20 ppm (23 mg/m ³)	35 ppm (40 mg/m ³)	_	
Carbon monoxide	8 hour	9.0 ppm (10 mg/m ³)	9 ppm (10 mg/m ³)		
	8 hour (Lake Tahoe)	6 ppm (7 mg/m³)	—	—	
	1 hour	0.18 ppm (339 µg/m ³)	100 ppb (188 µg/m³)	—	
Nitrogen dioxide	Annual arithmetic mean	0.030 ppm (57 μg/m³)	0.053 ppm (100 μg/m³)	Same as primary standard	
	1 hour	0.25 ppm (655 µg/m³)	75 ppb (196 µg/m³)	_	
	3 hour	—	—	0.5 ppm (1,300 μg/m³)	
Sulfur dioxide	24 hour	0.04 ppm (105 µg/m³)	0.14 ppm (for certain areas)		
	Annual arithmetic mean	_	0.030 ppm (for certain areas)	_	
	30-day average	1.5 µg/m³	—	_	
Lead	Calendar quarter	—	1.5 µg/m³	Samo as Primany	
	Rolling 3-month average	—	0.15 µg/m³	Standard	
Visibility-reducing particles	8 hour	See Footnote ¹			
Sulfates	24 hour	25 µg/m³	No Nationa	l Standards	
Hydrogen sulfide	1 hour	0.03 ppm (42 µg/m ³)	/m ³) /m ³)		
Vinyl chloride	24 hour	0.01 ppm (26 µg/m ³)			

Table 3.3-2. California and National Ambient Air Quality Standards



Draft Environmental Impact Report Environmental Impact Analysis

Pollutant	Averaging	California Standards National Standards		Standards
	Time	Concentration	Primary	Secondary

Notes:

¹ In 1989, the CARB converted both the general statewide 10-mile visibility standard and the Lake Tahoe 30-mile visibility standard to instrumental equivalents, which are "extinction of 0.23 per kilometer" and "extinction of 0.07 per kilometer" for the statewide and Lake Tahoe Air Basin standards, respectively. μg/m³ =micrograms per liter

 mg/m^3 = milligrams per cubic meter

Source: CARB 2016

The USEPA has classified air basins (or portions thereof) as being in attainment, nonattainment, or unclassified for each criteria air pollutant, based on whether or not the NAAQS have been achieved. If an area is designated unclassified, it is because inadequate air quality data were available as a basis for a nonattainment or attainment designations. As summarized in Table 3.3-3, the SFBAAB is designated as nonattainment for state ozone, PM_{2.5}, and PM₁₀ standards, as well as national ozone and PM_{2.5} standards.

 Table 3.3-3. San Francisco Bay Area Air Basin Designations for State and National

 Ambient Air Quality

Criteria Pollutants	State Designation	National Designation
Ozone	Non-attainment	Non-attainment
Carbon monoxide	Attainment	Attainment
PM ₁₀	Non-attainment	Unclassified
PM _{2.5}	Non-attainment	Non-attainment
Carbon monoxide	Attainment	Unclassified/attainment
Nitrogen dioxide	Attainment	Unclassified/attainment
Sulfur dioxide	Attainment	Attainment
Sulfates	Attainment	_
Lead	Attainment	Unclassified/attainment
Hydrogen sulfide	Unclassified	_
Visibility reducing particles	Unclassified	_

Notes:

 $PM_{2.5}$ = particulate matter less than 2.5 microns in aerodynamic diameter

 PM_{10} = particulate matter between 2.5 and 10 microns in aerodynamic diameter

Source: BAAQMD 2017b

National Ambient Air Quality Standards

The CAA required USEPA to establish NAAQS. As shown in Table 3.3-2, USEPA has established primary and secondary NAAQS for the following criteria air pollutants: O₃, CO, NO₂, SO₂, PM₁₀, PM_{2.5}, and lead. The primary standards protect the public health, and the secondary standards protect public welfare. The CAA also required each state to prepare an air quality control plan, referred to as a State Implementation Plan (SIP). The federal CAA amendments of 1990 added requirements for states with nonattainment areas to revise their SIPs to incorporate additional control measures to reduce air pollution. The SIP is modified periodically to reflect the latest emissions inventories, planning documents,



Draft Environmental Impact Report Environmental Impact Analysis

and rules and regulations of the air basins as reported by their jurisdictional agencies. USEPA is responsible for reviewing all SIPs to determine whether they conform to the mandates of the CAA and its amendments, and whether implementation would achieve air quality goals. If USEPA determines a SIP to be inadequate, a federal implementation plan that imposes additional control measures may be prepared for the nonattainment area. If an approvable SIP is not submitted or implemented within the mandated timeframe, sanctions may be applied to transportation funding and stationary air pollution sources in the air basin.

Hazardous Air Pollutants

USEPA and CARB regulate hazardous air pollutants (HAPs) and TACs through statutes and regulations that generally require the use of the maximum available control technology or best available control technology for TACs to limit emissions, respectively. These, in conjunction with additional rules set forth by BAAQMD, described further below, establish the regulatory framework for TACs.

Under federal law, 187 substances are currently listed as HAPs. Major sources of specific HAPs are subject to the requirements of the National Emissions Standards for Hazardous Air Pollutants (NESHAP) program. The USEPA is establishing regulatory schemes for specific source categories and requires implementation of the Maximum Achievable Control Technologies (MACT) for major sources of HAPs in each source category. State law has established the framework for California's TAC identification and control program, which is generally more stringent than the federal program and is aimed at HAPs that are a problem is California. The state has formally identified 244 substances as TACs and is adopting appropriate control measures for each. Once adopted at the state level, each air district will be required to adopt a measure that is equally or more stringent.

State

The California Legislature enacted the California Clean Air Act (CCAA) in 1988 to address air quality issues. CARB is the agency responsible for coordination and oversight of state and local air pollution control programs in California and for implementing the CCAA. California law authorizes CARB to set ambient (outdoor) air pollution standards (California Health and Safety Code [HSC] Section 39606) in consideration of public health, safety, and welfare (California Ambient Air Quality Standards [CAAQS]) (Table 3.3-2).

California Clean Air Act of 1988

The CCAA allows the state to adopt ambient air quality standards and other regulations provided that they are at least as stringent as federal standards. CARB, a part of the California Environmental Protection Agency (Cal EPA), is responsible for the coordination and administration of both federal and state air pollution programs within California and for implementing the CCAA. California law authorizes CARB to set ambient (outdoor) air pollution standards (California HSC Section 39606) in consideration of public health, safety, and welfare (Table 3.3-2).

California Ambient Air Quality Standards

CARB has established CAAQS for sulfates, hydrogen sulfide, vinyl chloride, visibility-reducing particulate matter, and the above-mentioned criteria air pollutants. In most cases, the CAAQS are more stringent



Draft Environmental Impact Report Environmental Impact Analysis

than the NAAQS. Differences in the standards are generally explained by the health effects studies considered during the standard-setting process and the interpretation of the studies. In addition, the CAAQS incorporate a margin of safety to protect sensitive individuals.

CCAA requires that all local air districts in the state endeavor to achieve and maintain CAAQS by the earliest date practicable. CCAA specifies that local air districts should focus attention on reducing the emissions from transportation and area-wide emission sources and provides districts with the authority to regulate indirect sources.

Among CARB's other responsibilities are overseeing local air district compliance with federal and state laws, approving local air quality plans, submitting SIPs to Cal EPA, monitoring air quality, determining and updating area designations and maps, and setting emissions standards for new mobile sources, consumer products, small utility engines, off-road vehicles, and fuels.

California State Implementation Plan

The federal CAA (and its subsequent amendments) requires each state to prepare an air quality control plan referred to as a SIP. The SIP is a living document that is periodically modified to reflect the latest emissions inventories, plans, and rules and regulations of air basins as reported by the agencies with jurisdiction over them. The CAA Amendments released in 1992 dictate that states containing areas violating the NAAQS revise their SIPs to include extra control measures to reduce air pollution. The SIP includes strategies and control measures to attain the NAAQS by deadlines established by the CAA. The Cal EPA has the responsibility to review all SIPs to determine if they conform to the requirements of the CAA.

State law makes CARB the lead agency for all purposes related to the SIP. Local air districts and other agencies prepare SIP elements and submit them to CARB for review and approval. CARB then forwards SIP revisions to the Cal EPA for approval and publication in the Federal Register. The 2017 *Clean Air Plan, Spare the Air, Cool the Climate* is the SIP for SFBAAB. The 2017 Clean Air Plan is a regional blueprint for achieving air quality standards and healthful air in the SFBAAB. The 2017 Clean Air Plan focuses on two closely-related goals: protecting public health and protecting the climate. Consistent with the GHG reduction targets adopted by the state of California, the plan lays the groundwork for a long-term effort to reduce Bay Area GHG emissions 40 percent below 1990 levels by 2030 and 80 percent below 1990 levels by 2050. The 2017 plan also addressed a multi-pollutant strategy to simultaneously reduce emissions and ambient concentrations of ozone, fine particulate matter, toxic air contaminants, as well as GHG's. The control strategy focuses on the following priorities: reduce emissions of criteria air pollutants and TACs from all key sectors; reduce emissions of "super-GHGs" such as methane, black carbon, and fluorinated gases; decrease demand for fossil fuels (gasoline, diesel and natural gas); and decarbonize the energy system.

California Air Toxics "Hot Spots"

TACs in California are regulated primarily through the Tanner Air Toxics Act (Assembly Bill [AB] 1807, Chapter 1047, Statutes of 1983) and the Air Toxics "Hot Spots" Information and Assessment Act of 1987 (AB 2588, Chapter 1252, Statutes of 1987). AB 1807 sets forth a formal procedure for CARB to designate substances as TACs. Research, public participation, and scientific peer review are required before CARB



Draft Environmental Impact Report Environmental Impact Analysis

can designate a substance as a TAC. To date, CARB has identified more than 21 TACs, including diesel particulate matter, and has adopted USEPA's list of HAPs as TACs.

Once a TAC is identified, CARB adopts an airborne toxics control measure for sources that emit that particular TAC. If a safe threshold exists for a substance at which there is no toxic effect, the control measure must reduce exposure below that threshold. If no safe threshold exists, the source must incorporate best available control technology for toxics to minimize emissions.

CARB has adopted diesel exhaust control measures and more stringent emission standards for various on-road mobile sources of emissions, including transit buses, and off-road diesel equipment (e.g., tractors, generators). Recent milestones included the low-sulfur diesel fuel requirement and stricter emissions standards for heavy-duty diesel trucks (effective in 2007 and subsequent model years) and off-road diesel equipment (2011). Over time, replacing older vehicles would result in a vehicle fleet that produces substantially lower levels of TACs than under current conditions. Mobile-source emissions of TACs (e.g., benzene, 1,3-butadiene, DPM) in California have been reduced substantially over the last decade; such emissions will be reduced further through a progression of regulatory measures (e.g., low-emission vehicles, clean fuels, and Phase II reformulated-gasoline regulations) and control technologies.

The Hot Spots Act requires that existing facilities that emit toxic substances above a specified level prepare an inventory of toxic emissions and a risk assessment if emissions are significant, notify the public of significant risk levels, and prepare and implement risk reduction measures.

In March 2015, the California Office of Environmental Health Hazard Assessment (OEHHA) adopted "The Air Toxics Hot Spots Program Guidance Manual for Preparation of Health Risk Assessments" in accordance with the Health and Safety Code, Section 44300. The Final Guidance Manual incorporates the scientific basis from earlier developed Technical Support Documents to assess risk from exposure to facility emissions. The 2015 OEHHA Final Guidance has key changes including greater age sensitivity in particular for children, decreased exposure durations, and higher breathing rate profiles. Because cancer risk could be up to three times greater using this new guidance, it may result in greater mitigation requirements, more agency backlog, and increased difficulty in getting air permits.

Regional

Bay Area Air Quality Management District

BAAQMD is the public agency that regulates stationary sources of air pollution in the nine counties that comprise the San Francisco Bay Area: Alameda, Contra Costa, Marin, Napa, San Francisco, San Mateo, Santa Clara, southwestern Solano, and southern Sonoma. BAAQMD attains and maintains air quality conditions in Napa County through a comprehensive program of planning, regulation, enforcement, technical innovation, and promotion of the understanding of air quality issues. The clean air strategy of BAAQMD includes the preparation of plans and programs for the attainment of NAAQS and CAAQS, adoption and enforcement of rules and regulations, and issuance of permits for stationary sources. BAAQMD also inspects stationary sources, responds to citizen complaints, monitors ambient air quality and meteorological conditions, and implements other programs and regulations required by the CAA and CCAA.



Draft Environmental Impact Report Environmental Impact Analysis

As mentioned above, BAAQMD adopts rules and regulations. All projects are subject to BAAQMD's rules and regulations in effect at the time of construction. Specific rules applicable to project construction may include, but are not limited to:

- Regulation 2, Rule 1, General Permit Requirements: Includes criteria for issuance or denial of permits, exemptions, appeals against decisions of the Air Pollution Control Officer and BAAQMD actions on applications.
- Regulation 2, Rule 2, New Source Review: Applies to new or modified sources and contains requirements for Best Available Control Technology and emission offsets. Rule 2 implements federal New Source Review and Prevention of Significant Deterioration requirements.
- Regulation 6, Rule 1, General Requirements: Limits the quantity of particulate matter in the atmosphere by controlling emission rates, concentration, visible emissions, and opacity.
- Regulation 7, Odorous Substances: Regulation 7 places general limitations on odorous substances and specific emission limitations on certain odorous compounds. A person (or facility) must meet all limitations of this regulation but meeting such limitations shall not exempt such person from any other requirements of BAAQMD, state, or national law. The limitations of this regulation shall not be applicable until BAAQMD receives odor complaints from 10 or more complainants within a 90-day period alleging that a person has caused odors perceived at or beyond the property line of such person and deemed to be objectionable by the complainants in the normal course of their work, travel, or residence. When the limits of this regulation become effective as a result of citizen complaints described above the limits shall remain effective until such time as no citizen complaints have been received by BAAQMD for 1 year. The limits of this regulation shall become applicable again if BAAQMD receives odor complaints from five or more complainants within a 90-day period. BAAQMD staff shall investigate and track all odor complaints they receive and shall attempt to visit the site, identify the source of the objectionable odor, and assist the owner or facility in finding a way to reduce the odor.
- Regulation 8, Rule 3, Architectural Coatings: Limits the quantity of volatile organic compounds in architectural coatings supplied, sold, offered for sale, applied, solicited for application, or manufactured for use within BAAQMD.

Bay Area Air Quality Management District Toxic Air Contaminants

At the local level, air pollution control or management districts may adopt and enforce CARB control measures. Under BAAQMD Regulation 2, Rule 1, General Permit Requirements, and Regulation 2, Rule 2, New Source Review, all sources that possess the potential to emit TACs are required to obtain permits from the district. Permits may be granted to these operations if they are constructed and operated in accordance with applicable regulations including new-source-review standards and air-toxics control measures. BAAQMD limits emissions and public exposure to TACs through programs including the Community Air Risk Evaluation Program, which estimates and reports both local and regional impacts of TACs in the Bay Area. BAAQMD administers certain portions of the Air Toxics "Hot Spots" Information and Assessment Act (AB 2588, 1987, Connelly), which serves to collect data, identify specific facilities that produce localized impacts, assess health risks, notify nearby residents of risks, and reduce those



Draft Environmental Impact Report Environmental Impact Analysis

significant risks to acceptable levels through 'Hot Spots' Risk Reduction Audits and Plans for specific facilities.

Bay Area Air Quality Management CEQA Air Quality Guidelines

The BAAQMD released the latest version of the CEQA Air Quality Guidelines in 2017. The Guidelines provide BAAQMD recommended procedures for evaluating air quality impacts within the SFBAAB during the CEQA process. The Guidelines also establish construction and operational related criteria air pollutant thresholds for ROG, NOx, PM10, and PM2.5. These thresholds establish the level at which a project's individual emissions of air pollutants or precursors would result in a cumulatively considerable contribution to existing air quality.

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project (City of Newark 2013a):

Goal CS-6: Green Building. Reduce the impacts of buildings and development on greenhouse gas levels and the environment in general.

• **Policy CS-6.2: Encouraging Greener Construction.** Encourage greener construction methods and greater use of recycled-content materials in new residential, commercial, and industrial construction projects.

Goal CS-7: Energy Conservation. Maximize opportunities for energy efficiency, conservation, and independence.

- **Policy CS-7.1: Reducing Energy Use.** Support measures to reduce energy consumption and increase energy efficiency in residential, commercial, industrial, and public buildings.
- Policy CS-7.2: Renewable Energy Sources. Support the expanded use of renewable energy sources such as wind and solar by Newark residents and businesses, the City of Newark, and other government officials.
- **Policy CS-7.3: Designing for Energy Efficiency.** Support building design, site planning, and subdivision design methods that reduce heating and cooling costs and achieve greater energy efficiency.

Goal LU-2: Land Use Compatibility. Ensure compatibility between adjacent land uses.

• **Policy LU-2.4: Buffering from Transportation Facilities.** Ensure that the design of new residential development near rail lines, truck routes, freeways, or major thoroughfares includes setbacks, landscape screening, and other provisions to minimize exposure to negative impacts such as noise and air pollution.

Goal HW-1: Air Quality. Air quality that meets state and federal standards and provide improved respiratory health for Newark residents.



Draft Environmental Impact Report Environmental Impact Analysis

• **Policy HW-1.3: Reducing Exposure to Air Pollution in New Development.** Use site planning and architectural design to reduce potential exposure of sensitive uses to major air pollution sources, including freeways and industrial activities.

3.3.3 Environmental Impacts

This section analyzes the project's potential to result in significant air quality impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

Construction

Construction of the Project has the potential to create air quality impacts through the use of heavy-duty construction equipment, construction workers' vehicle trips, truck hauling trips, and vendor truck trips. Offroad construction equipment and vehicle trips generate NOx and particulate matter. Diesel powered construction equipment and hauling trucks would emit DPM, a type of TAC. In addition, fugitive dust emissions would result from site disturbance, including grading and asphalt recycling, and fugitive ROG emissions would result from application of architectural coatings and paving. Short-term constructionrelated emissions of criteria air pollutants and precursors were calculated using the California Emissions Estimator Model (CalEEMod) Version 2020.4.0 computer program (California Air Pollution Control Officers Association 2017). CalEEMod was used to calculate emissions from construction of the parking lot, buildings, and paved areas. Modeling was based on project-specific information (e.g., building type and size, amount of demolition, area to be paved) where available, and default values in CalEEMod that are based on the project's location, land use type, and type of construction.

The construction activities and assumed duration, based on CalEEMod defaults with adjustments for anticipated debris and soil hauling, are shown in Table 3.3-4, Anticipated Construction Schedule.

Construction Activity	Start Date	End Date	Number of Working Days
Pick-n-Pull Inventory Removal	9/26/2023	1/25/2024	88
Demolition	1/26/2024	3/28/2024	45
Mobilization	3/29/2024	4/4/2024	5
Remedial Soil Cleanup	4/10/2024	5/22/2024	31
Soil Import	5/23/2024	10/16/2024	105
Grading	5/23/2024	10/18/2024	107
Underground Utilities	10/19/2024	9/25/2025	244
Jack and Bore Preparation	3/1/2025	3/10/2025	6
Jack and Bore	3/11/2025	4/21/2025	30
Jack and Bore Cleanup	4/22/2025	4/29/2025	6
Off-Site Street Improvements	5/1/2025	5/14/2025	10
Paving	9/26/2025	10/6/2025	7

Table 3.3-4. Anticipated Construction Schedule



Draft Environmental Impact Report Environmental Impact Analysis

Construction Activity	Start Date	End Date	Number of Working Days
Building Construction	10/7/2025	5/6/2026	152
Architectural Coating	11/6/2025	5/6/2026	130

Source: CalEEMod.

Off-road equipment such as excavators, graders, backhoes, loaders, pile-driving rigs, crushing equipment, pavers, water trucks, and forklifts would be used for demolition, geotechnical work, excavation, and grading, but also for building construction and hardscape and landscape materials installation. Miscellaneous construction equipment would include generators and air compressors, and possibly crushing and processing equipment and cement/mortar mixers. A variety of other smaller mechanical equipment would also be used at the Project site during the construction period, such as saw cutters, cutting/chopping saws, tile saws, stud impact guns, welding machines, and concrete boom pumps. Construction equipment estimates are based on CalEEMod defaults, adjusted for the anticipated construction schedule and site conditions. The modeled construction equipment for each activity is shown in Table 3.3-5, Construction Equipment Assumptions.

Equipment	Horsepower	Number	Hours per Day			
Pick-n-Pull Inventory Removal	Pick-n-Pull Inventory Removal					
Cranes	231	1	8			
Concrete/Industrial Saws	81	1	8			
Excavators	158	3	8			
Rubber Tired Dozers	247	2	8			
Demolition						
Concrete/Industrial Saws	81	1	8			
Excavators	158	3	8			
Rubber Tired Dozers	247	2	8			
Mobilization						
Cranes	231	1	2			
Tractors/Loaders/Backhoes	97	1	8			
Remedial Soil Cleanup						
Excavators	158	1	8			
Water Trucks	402	1	8			
Rubber Tired Dozers	247	1	8			
Rubber Tired Loaders	247	1	8			
Grading						
Excavators	158	2	8			
Graders	187	1	8			
Water Trucks	402	1	8			

Table 3.3-5. Construction Equipment Assumptions



Draft Environmental Impact Report Environmental Impact Analysis

Equipment	Horsepower	Number	Hours per Day
Rubber Tired Dozers	247	1	8
Scrapers	367	2	8
Tractors/Loaders/Backhoes	97	2	8
Underground Utilities			
Cranes	231	1	2
Excavators	158	2	8
Water Trucks	402	1	8
Rubber Tired Loaders	247	1	8
Tractors/Loaders/Backhoes	97	2	8
Jack and Bore Preparation	·		
Cranes	231	1	2
Excavators	158	1	8
Ski Steer Loaders	65	1	8
Tractors/Loaders/Backhoes	97	1	8
Jack and Bore	·		
Bore/Drill Rigs	221	1	8
Cranes	231	1	2
Excavators	158	1	8
Pumps	84	1	8
Jack and Bore Cleanup	·		
Cranes	231	1	2
Ski Steer Loaders	65	1	8
Tractors/Loaders/Backhoes	97	1	8
Off-Site Street Improvements			
Pavers	130	2	8
Paving Equipment	132	2	8
Rollers	80	2	8
Surfacing Equipment (Pavement Scarifier)	263	1	8
Paving			
Pavers	130	2	8
Paving Equipment	132	2	8
Rollers	80	2	8
Building Construction			
Cranes	231	1	2
Forklifts	89	3	8
Generator Sets	84	1	2
Tractors/Loaders/Backhoes	97	3	7
Architectural Coating			



Draft Environmental Impact Report Environmental Impact Analysis

Equipment	Horsepower	Number	Hours per Day
Air Compressors	78	1	6

Source: CalEEMod.

Subsurface project construction activities would also include construction of building foundations, and installation of subsurface utilities.

The areas to be excavated and/or graded were based on the project Tentative Map (CBG 2021). Approximately 39,000 CY of vegetation, contaminated soil, and old asphalt would be exported from the site during site preparation. Approximately 252,000 CY of soil and aggregate would be imported during grading to raise the level of the building pads and construct roadbeds. All grading and construction activity associated with site cleanup was included in the analysis of Project construction emissions. Project construction would also generate emissions from off-site truck trips for deliveries of concrete and other building materials, transportation of construction equipment to and from the site, hauling soils and debris from the site, and street sweepers.

All CalEEMod output files are provided in Appendix B.

Operations

Long-term operational emissions of criteria air pollutants and precursors were also calculated using CalEEMod. Operational emissions are generated from consumer products, landscape maintenance activities, and mobile-source emissions.

The project land uses were modeled as:

- 203 single-family residences with a default floor space of 1,800 square feet each
- approximately 7.13 acres of paved streets and sidewalks,
- approximately 5.89 of off-site street/sidewalk improvements,
- approximately 35,350 square feet of off-site utility improvements,
- approximately 4.89 acres of landscaping, storm water control features, and community open space.

Because the project would replace the existing automotive recycling and parts business on the project site (the Pick-n-Pull), the project's operational emissions would be offset by emissions from existing operations, which would be replaced. Emissions from existing operations were estimated in a separate model using CalEEMod. Existing land uses were modeled as 1,500 square feet of general office building; 3,000 square feet of general light industrial building; and 13,000 square feet of unrefrigerated warehouse-no rail.

Operational sources of criteria pollutant and precursor emissions in CalEEMod include area, energy and mobile:



Draft Environmental Impact Report Environmental Impact Analysis

- Area area sources include emissions from landscaping equipment, the use of consumer products, the reapplication of architectural coatings for maintenance, and gas fireplaces. Emissions associated with area sources were estimated using the CalEEMod default values for the project and the existing land use. Area sources in CalEEMod also include emissions from wood burning stoves and fireplaces. However, in accordance with the BAAQMD Regulation 6, Rule 3 *Wood-Burning Devices*, permanently installed wood-burning devices are not permitted in new development and the project would not include wood-burning stoves or wood-burning fireplaces (BAAQMD 2015).
- Energy The project would use electricity for lighting, heating and cooling. Electricity would be supplied by PG&E. Criteria pollutant and precursor emissions related to the generation of electrical power are emitted at the site of the generation facilities and are not included in the CalEEMod operation emissions. Criteria pollutant and precursor emissions related to the burning of natural gas in furnaces, water heaters and appliances are included in the CalEEMod operation emissions. The CalEEMod default natural gas usage rates for Alameda County were used for project and existing use modeling.
- Mobile Operational emissions from mobile sources are associated with project-related vehicle trip generation and trip length. Based on the trip generation rate from the project Transportation Impact Analysis (TIA), the project's single-family homes would generate 2,000 average daily trips ([ADT], and the existing land use generates 920 ADT (Fehr & Peers 2021a).

A project VMT analysis was completed as part of the TIA which concluded that the project would result in 27.9 average daily miles per resident in 2020 and 25.6 average daily miles per resident in 2040 (Fehr & Peers 2021a). Using linear interpolations between the VMT data points in the TIA, in 2027 the daily VMT would be 27.0 miles per resident. The project is anticipated to have a residential population of 682, resulting in a total project annual VMT of 6.72 million miles for the first full year of operation. The residential trip distances and purposes were adjusted in the model to result in an annual VMT of approximately 6.7 million miles.

All CalEEMod output files are provided in Appendix B.

Odors

Odors from a project are evaluated qualitatively. The analysis considers the screening level distance from typical odor producing land facilities (e.g., landfill, composting, etc.) when siting new receptors and the compliant history of the proposed land use.

Thresholds of Significance

In developing thresholds of significance for air pollutants, BAAQMD considered the emission levels for which a project's individual emissions would be cumulatively considerable. If a project exceeds the identified significance thresholds, its emissions would be cumulatively considerable, resulting in significant adverse air quality impacts to the region's existing air quality conditions. Therefore, additional analysis to assess cumulative impacts is unnecessary. The analysis to assess project-level air quality impacts should be as comprehensive and rigorous as possible (BAAQMD 2017a). Table 3.3-6 summarizes BAAQMD thresholds used for this analysis.



Table 3.3-6: BAAQMD Air Quality CEQA Thresholds of Significance

Pollutant	Construction-Related	Operation	al-Related	
Criteria Air Pollutants and Precursors (regional)	Average Daily Emissions (Ibs/day)	Average Daily Emissions (Ibs/day)	Maximum Annual Emissions (TPY)	
ROG	54	54	10	
NOx	54	54 54		
PM10	82 (exhaust)	82	15	
PM2.5	54 (exhaust)	54	10	
PM ₁₀ /PM _{2.5} (fugitive dust)	Best management practices	None		

Notes:

lbs/day = pounds per day

NOx = nitrogen oxides

PM_{2.5} = particulate matter less than 2.5 microns in aerodynamic diameter

PM₁₀ = particulate matter between 2.5 and 10 microns in aerodynamic diameter

ROG = reactive organic gases

TPY = tons per year

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's air quality impacts are significant. Would the proposed project:

- Conflict with or obstruct implementation of the applicable air quality plan?
- Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or state ambient air quality standard?
- Expose sensitive receptors to substantial pollutant concentrations?
- Result in other emissions (such as those leading to odors) adversely affecting a substantial number of people?

Project Impact Analysis and Mitigation Measures

Air Quality Plan

Impact AIR-1 The proposed project could conflict with or obstruct implementation of the applicable air quality plan.

Impact Analysis

The BAAQMD's 2017 Clean Air Plan is the regional air quality plan (AQP) for the Air Basin (BAAQMD 2017b). It identifies strategies to bring regional emissions into compliance with federal and State air quality standards. It also provides a control strategy to reduce ozone, PM, air toxics, and GHGs. The BAAQMD's Guidance provides two criteria for determining if a plan-level project is consistent with the current AQP control measures. However, the BAAQMD does not provide a threshold of significance for



Draft Environmental Impact Report Environmental Impact Analysis

project-level consistency analysis. Therefore, the following criteria will be used for determining a project's consistency with the AQP.

- Criterion 1: Does the project support the primary goals of the AQP?
- Criterion 2: Does the project include applicable control measures from the AQP?
- Criterion 3: Does the project disrupt or hinder implementation of any AQP control measures?

Criterion 1

The primary goals of the 2017 Clean Air Plan, the current AQP, are to:

- Protect public health through the attainment air quality standards
- Protect the climate

As discussed in impact discussions AIR-2, AIR-3, and AIR-4 the proposed project would not significantly contribute to cumulative nonattainment pollutant violations, expose sensitive receptors to substantial pollutant concentrations, or create objectionable odors affecting a substantial number of people after implementation of Mitigation Measure AIR-1. Therefore, the project is consistent with criterion 1 with incorporation of Mitigation Measure AIR-1, which would require all construction contractors to implement the basic construction mitigation measures recommended by the BAAQMD to reduce fugitive dust emissions.

Regarding climate protection, the proposed project's GHG emissions were determined to be less than significant, and the proposed project was consistent with applicable greenhouse gas reduction plans adopted to protect the climate (See Section 3.8, Greenhouse Gas Emissions). Accordingly, the proposed project would be consistent with criterion 1 for climate protection.

Criterion 2

The 2017 Clean Air Plan contains 85 control measures aimed at reducing air and climate pollutants in the Bay Area. For purposes of consistency with climate planning efforts at the state level, the control strategy in the Clean Air Plan is based upon the same economic sector framework used by the CARB for its 2014 update to the AB 32 Scoping Plan. The sectors are as follows:

- Stationary Sources
- Transportation
- Energy
- Buildings
- Agriculture
- Natural and Working Lands
- Waste Management



- Water
- Super-Greenhouse Gas Pollutants

The proposed project's potential to conflict with each of these measures is discussed below. The Clean Air Plan recognizes that, to a great extent, community design¹ dictates individual travel modes and that a key long-term control strategy to reduce emissions of criteria pollutants, air toxics, and GHGs from motor vehicles is to channel future Bay Area growth into communities where goods and services are located nearby and people have a range of viable transportation options. Therefore, the 2017 Clean Air Plan includes 85 control measures aimed at reducing air pollutants and GHGs in the SFBAAB.

Stationary Source Control Measures. The Stationary Source Measures, which are designed to reduce emissions from stationary sources such as metal melting facilities, refineries, and glass furnaces, are incorporated into rules adopted by the BAAQMD and then enforced by the BAAQMD's Permit and Inspection programs. Since the proposed project is residential in nature would not include any stationary sources of emissions, the Stationary Source Measures of the Clean Air Plan are not applicable to the proposed project.

Transportation Control Measures. The BAAQMD identifies Transportation Measures as part of the Clean Air Plan to decrease emissions of criteria pollutants, TACs, and greenhouse gases (GHGs) by reducing demand for motor vehicle travel, promoting efficient vehicles and transit service, decarbonizing transportation fuels, and electrifying motor vehicles and equipment. The proposed project would develop new multifamily residences that would locate residents near existing and planned residential uses, commercial, office, and retail space uses, and public parks. The proposed project would be constructed in accordance with City standards and would be consistent with the BAAQMD's effort to encourage planning for bicycle and pedestrian facilities.

Energy Control Measures. The Clean Air Plan also includes Energy Control Measures, which are designed to reduce emissions of criteria air pollutants, TACs, and GHGs by decreasing the amount of electricity consumed in the Bay Area, as well as decreasing the carbon intensity of the electricity used by switching to less GHG-intensive fuel sources for electricity generation. Since these measures apply to electrical utility providers and local government agencies (and not individual projects), the Energy Control Measures of the Clean Air Plan are not applicable to the proposed project. However, the project applicant would be required to conform to the energy efficiency requirements of the California Building Standards Code, also known as Title 24. Specifically, the project must implement the requirements of the most recent Building Energy Efficiency Standards, which is the current version of Title 24.

Building Control Measures. The BAAQMD has authority to regulate emissions from certain sources in buildings such as boilers and water heaters but has limited authority to regulate buildings themselves. Therefore, the strategies in the control measures for this sector focus on working with local governments that do have authority over local building codes, to facilitate adoption of best GHG control practices and

¹ For people who live in low-density, car-oriented residential developments, the motor vehicle is often the only viable transportation option. In such situations, even the best strategies to promote alternative modes of travel can only have a very modest effect. Alternatively, compact communities with a mixture of land uses make it much easier to walk, cycle, or take transit for at least some daily trips.



Draft Environmental Impact Report Environmental Impact Analysis

policies. The proposed project would be required to comply with the latest California Green Building Standards Code (CALGreen) standards. Therefore, the Building Control Measures of the Clean Air Plan are not applicable to the proposed project.

Agriculture Control Measures. The Agriculture Control Measures are designed to primarily reduce emissions of methane. Since the proposed project does not include any agricultural activities, the Agriculture Control Measures of the Clean Air Plan are not applicable to the proposed project.

Natural and Working Lands Control Measures. The Natural and Working Lands Control Measures focus on increasing carbon sequestration on rangelands and wetlands, as well as encouraging local governments to ordinances that promote urban-tree plantings. Since the project does not include the disturbance of any rangelands or wetlands, the Natural and Working Lands Control Measures of the Clean Air Plan are not applicable to the proposed project.

Waste Management Control Measures. The Waste Management Measures focus on reducing or capturing methane emissions from landfills and composting facilities, diverting organic materials away from landfills, and increasing waste diversion rates through efforts to reduce, reuse, and recycle. The proposed project would comply with local requirements for waste management (e.g., recycling and composting services). Therefore, the proposed project would be consistent with the Waste Management Control Measures of the Clean Air Plan.

Water Control Measures. The Water Control Measures focus on reducing emissions of criteria pollutants, TACs, and GHGs by encouraging water conservation, limiting GHG emissions from publicly owned treatment works (POTWs), and promoting the use of biogas recovery systems. Since these measures apply to POTWs and local government agencies (and not individual projects), the Water Control Measures are not applicable to the proposed project.

Super-GHG Control Measures. The Super-GHG Control Measures are designed to facilitate the adoption of best GHG control practices and policies through the BAAQMD and local government agencies. Since these measures do not apply to individual projects, the Super-GHG Control Measures are not applicable to the proposed project.

As discussed above, most of the measures contained in the Clean Air Plan would not be applicable to the proposed project and will be implemented by BAAQMD using its permit authority or are aimed at cities or counties to adopt within general plans. The proposed project would not impede implementation of any measures contained in the Clean Air Plan and would be consistent with applicable measures outlined in the Clean Air Plan. For example, under Mitigation Measure AIR-2 the proposed project would comply with TR22 of the Clean Air Plan that encourages the early development of Tier 4 engines in construction. Moreover, the proposed project will be consistent with the goals and policies laid out within the City's General Plan to reduce air pollution. The proposed project would be constructed to be 100 percent electric and would be built to the latest California Green Buildings Standards for residential developments. As such, the proposed project would be consistent with General Plan Goal CS-7. As shown in Impact 2 through Impact 4 below, the proposed project's emissions, risk, and odor impacts were found to be less than significant. As such the proposed project would also comply with General Plan Goal HW-1. Therefore, the project would not disrupt or hinder implementation of a control measure from the Clean Air Plan.



Draft Environmental Impact Report Environmental Impact Analysis

Criterion 3

If the approval of a project would not cause a disruption, delay, or otherwise hinder the implementation of any clean air plan control measure it would be considered consistent with the 2017 Clean Air Plan. Examples of how a project may cause the disruption or delay of control measures include a project that precludes an extension of a transit line or bike path or proposes excessive parking beyond parking requirements. The project will not preclude extension of a transit line or bike path (control measure TR-9), propose excessive parking beyond parking requirements (control measure TR-13), or otherwise create an impediment or disruption to implementation of any AQP control measures.

Conclusion

The proposed project would be consistent with the criteria of the AQP with incorporation of Mitigation Measure AIR-1 and AIR-2. As such, with the incorporation of this mitigation measure this impact would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM AIR-1:** Implement Construction Best Management Practices. The applicant shall require all construction contractors to implement the basic construction mitigation measures recommended by the BAAQMD to reduce fugitive dust emissions. Emission reduction measures will include, at a minimum, the following measures. Additional measures may be identified by the BAAQMD or contractor as appropriate:
 - All exposed surfaces (e.g., parking areas, staging areas, soil piles, graded areas, and unpaved access roads) will be watered two times per day
 - All haul trucks transporting soil, sand, or other loose material off-site will be covered
 - All visible mud or dirt track-out onto adjacent public roads will be removed using wet power vacuum street sweepers at least once per day. The use of dry power sweeping is prohibited
 - All vehicle speeds on unpaved roads shall be limited to 15 miles per hour
 - All roadways, driveways, and sidewalks to be paved will be completed as soon as possible. Building pads will be laid as soon as possible after grading unless seeding or soil binders are used
 - Idling times shall be minimized either by shutting equipment off when not in use or by reducing the maximum idling time to 5 minutes (as required by the California Airborne Toxics Control Measure Title 13, Section 2485 of California Code of Regulations; clear signage shall be provided for construction workers at all access points
 - All construction equipment shall be maintained and properly tuned in accordance with manufacturer's specifications

Draft Environmental Impact Report Environmental Impact Analysis

- All equipment shall be checked by a certified visible emissions evaluator or checked by a certified mechanic and determined to be running in proper condition prior to operation
- Post a publicly visible sign with the telephone number and person to contact at the City regarding dust complaints. This person will respond and take corrective action within 48 hours. The Bay Area Air Quality Management District's phone number will also be visible to ensure compliance with applicable regulations.
- MM AIR-2: Tier 4 Certified Construction Equipment. The project applicant or designated contractor shall specify, and the City shall verify, that all grading, building, and other construction permits for the project, include the following requirement: All diesel-powered off-road equipment used for project remediation and construction activities shall be U.S. EPA Tier 4 certified, or have CARB approved engine retrofit kits certified to have emissions equivalent to Tier 4 standards.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Criteria Pollutants

Impact AIR-2 The proposed project could result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or State ambient air quality standard.

Impact Analysis

In developing thresholds of significance for air pollutants, the BAAQMD considered the emission levels for which a project's individual emissions would be cumulatively considerable. If a project exceeds the identified significance thresholds, its emissions would be cumulatively considerable, resulting in significant adverse air quality impacts to the region's existing air quality conditions. Project construction and operational impacts are assessed separately below.

Construction Emissions

Construction activities associated with development of the proposed project would include site preparation, grading, building construction, paving and architectural coatings. Emissions from construction-related activities are generally short-term in duration but may still cause adverse air quality impacts.

Table 3.3-7 provides the construction emissions estimate for the proposed project. Please refer to Appendix B for details regarding assumptions used to estimate construction emissions. The duration of construction activity and associated equipment represent a reasonable approximation of the expected construction fleet as require pursuant to CEQA guidelines. As shown in the table, the unmitigated construction emissions would exceed BAAQMD significance thresholds.



MOWRY VILLAGE PROJECT Draft Environmental Impact Report Environmental Impact Analysis

	Pollutant Emissions (pounds per day)							
Phase	ROG	NOx	со	SOx	Fugitive PM ₁₀	Exhaust PM ₁₀	Fugitive PM _{2.5}	Exhaust PM _{2.5}
Pick-n-Pull Inventory Removal	3.2	39.9	26.2	0.1	1.6	1.4	0.4	1.3
Demolition	2.8	24.5	23.4	0.1	0.6	1.1	0.1	1.0
Mobilization	0.2	2.3	2.8	0.0	0.0	0.1	0.0	0.1
Remedial Soil Cleanup	2.4	39.1	21.0	0.1	9.6	0.9	4.2	0.8
Soil Import	0.6	40.4	9.1	0.2	5.5	0.3	1.5	0.3
Grading	3.8	35.7	31.5	0.1	9.4	1.5	3.7	1.3
Underground Utilities	1.6	14.7	17.1	0.0	0.5	0.5	0.1	0.5
Jack and Bore Preparation	0.5	4.2	7.5	0.0	0.1	0.2	0.0	0.2
Jack and Bore	0.8	6.3	9.6	0.0	0.1	0.3	0.0	0.2
Jack and Bore Cleanup	0.3	2.9	4.2	0.0	0.1	0.1	0.0	0.1
Off-Site Street Improvements	4.5	10.5	16.6	0.0	0.1	0.5	0.0	0.5
Paving	6.2	37.4	21.4	0.1	3.9	0.7	1.1	0.6
Building Construction	0.9	7.9	11.5	0.0	0.4	0.3	0.1	0.3
Architectural Coating	37.5	1.3	3.5	0.0	0.7	0.1	0.2	0.1
Concurrent 2024 Soil Import and Grading	4.4	76.1	40.5	0.3	14.9	1.8	5.2	1.7
Concurrent 2025 Underground Utilities and Jack and Bore	2.2	19.5	26.6	0.1	0.6	0.7	0.1	0.7
Concurrent 2025 Underground Utilities and Street Improve.	6.0	23.6	33.6	0.1	0.6	0.9	0.2	0.9
Concurrent 2025 Building Const. and Arch. Coating	38.3	9.1	14.9	0.0	1.0	0.4	0.3	0.4
Maximum Daily	38.9	76.1	40.5	0.3	14.9	1.8	5.2	1.7
BAAQMD Daily Thresholds	54	54	none	none	BCMMs	84	BCMMs	54
Exceed Daily Threshold?	No	Yes	No	No	Yes	No	Yes	No

Table 3.3-7: Unmitigated Remediation and Construction Criteria Pollutant and Precursor Emissions

Source: CalEEMod.

As shown in Table 3.3-7, impacts for emission of NOx during remediation and construction would exceed the BAAQMD threshold, and the impact would be potentially significant. Implementation of Mitigation Measure AIR-2 would require project construction to utilize Tier 4 certified construction equipment. As shown in Table 3.3-8, implementation of Mitigation Measure AIR-2 would reduce the impact to a less than significant level and would not result in a significant impact. Additionally, implementation of BCMMs as required by Mitigation Measure AIR-1 would reduce impacts of emission of PM₁₀ and PM_{2.5} during construction and remediation to a less than significant level and would not result in a significant impact.



MOWRY VILLAGE PROJECT Draft Environmental Impact Report Environmental Impact Analysis

	Pollutant Emissions (pounds per day)							
Phase	ROG	NOx	со	SOx	Fugitive PM ₁₀	Exhaust PM ₁₀	Fugitive PM _{2.5}	Exhaust PM _{2.5}
Pick-n-Pull Inventory Removal	0.8	13.4	31.3	0.1	1.6	0.2	0.4	0.2
Demolition	0.7	3.0	29.6	0.1	0.4	0.1	0.1	0.1
Mobilization	0.1	0.3	3.1	0.0	0.0	0.0	0.0	0.0
Remedial Soil Cleanup	0.9	23.6	28.3	0.1	5.9	0.3	2.3	0.2
Soil Import	0.6	40.4	9.1	0.2	5.4	0.3	1.5	0.3
Grading	1.0	4.0	39.4	0.1	4.3	0.1	1.7	0.1
Underground Utilities	0.5	4.5	22.9	0.0	0.5	0.1	0.1	0.1
Jack and Bore Preparation	0.2	1.7	8.7	0.0	0.1	0.0	0.0	0.0
Jack and Bore	0.3	1.2	13.1	0.0	0.1	0.0	0.0	0.0
Jack and Bore Cleanup	0.1	1.4	4.7	0.0	0.1	0.0	0.0	0.0
Off-Site Street Improvements	3.8	1.6	20.7	0.0	0.1	0.0	0.0	0.0
Paving	5.6	30.0	24.1	0.1	3.9	0.3	1.1	0.3
Building Construction	0.3	1.3	12.2	0.0	0.4	0.0	0.1	0.0
Architectural Coating	37.3	0.3	3.5	0.0	0.7	0.0	0.2	0.0
Concurrent 2024 Soil Import and Grading	1.6	44.4	48.5	0.3	9.7	0.5	3.2	0.4
Concurrent 2025 Underground Utilities and Jack and Bore	0.8	5.7	36.0	0.1	0.6	0.1	0.1	0.1
Concurrent 2025 Underground Utilities and Street Improve.	4.4	6.1	43.6	0.1	0.6	0.1	0.2	0.1
Concurrent 2025 Building Const. and Arch. Coating	37.6	1.6	15.7	0.0	1.0	0.0	0.3	0.0
Maximum Daily ¹	37.6	44.4	48.5	0.3	9.7	0.5	3.2	0.4
BAAQMD Daily Thresholds	54	54	none	none	BCMMs	84	BCMMs	54
Exceed Daily Threshold?	No	No	No	No	No	No	No	No

Table 3.3-8: Mitigated Remediation and Construction Criteria Pollutant and Precursor Emissions

Source: CalEEMod.

¹ Includes MM AIR-1 to implement BCMMs and MM AIR-2 to require Tier 4 engines for all off-road equipment with 50 or more horsepower

Operational Emissions

As previously discussed, the pollutants of concern include ROG, NO_X, PM₁₀, and PM_{2.5}. Operational sources of criteria pollutant and precursor emissions in CalEEMod include area, energy, and mobile sources.

Because the project would replace the existing automotive recycling and parts business on the project site (the Pick-n-Pull), the project's operational emissions would be offset by emissions from existing



Draft Environmental Impact Report Environmental Impact Analysis

operations, which would be replaced. The existing project emissions are presented in Table 3.3-9 and the project emissions are presented in Table 3.3-10. The net operational emissions fall below BAAQMD project level thresholds and would not result in a significant impact.

	Pollutant Emissions							
Source	ROG	NOx	со	SOx	Fugitive PM ₁₀	Exhaust PM ₁₀	Fugitive PM _{2.5}	Exhaust PM _{2.5}
Daily Emissions (pounds per day)								
Area	0.4	<0.1	<0.1	<0.1	0	<0.1	0	<0.1
Energy	<.01	<0.1	<0.1	<0.1	0	<0.1	0	<0.1
Mobile	1.8	2.5	18.7	<0.1	4.6	<0.1	1.2	<0.1
Total Existing Use Emissions ¹	2.3	2.5	18.7	<0.1	4.6	<0.1	1.2	<0.1

Table 3.3-9: Existing Use Criteria Pollutant and Precursor Emissions

Source: CalEEMod. ¹ Totals may not sum due to rounding.

Table 3.3-10: Operational Criteria Pollutant and Precursor Emissions

	Pollutant Emissions							
Source	ROG	NOx	со	SOx	Fugitiv e PM ₁₀	Exhaust PM ₁₀	Fugitive PM _{2.5}	Exhaust PM _{2.5}
	۵	aily Emi	issions (pounds p	er day)			
Area	10.2	2.9	17.9	<0.1	0	0.3	0	0.3
Energy	0.2	2.0	0.8	<0.1	0	0.2	0	0.2
Mobile	4.7	6.9	51.9	0.1	14.1	<0.1	3.8	<.01
Total Project Emissions ^{1,2}	15.2	11.8	70.7	0.1	14.1	0.5	3.8	0.5
Existing Use Emissions	(2.3)	(2.5)	(18.7)	(<0.1)	(4.6)	(<0.1)	(1.2)	(<0.1)
Net Project Emissions ¹	12.9	9.3	52.0	0.1	9.5	0.5	2.6	0.5
BAAQMD Daily Thresholds	54	54	none	none	none	84	none	54
Exceed Daily Threshold?	No	No	No	No	No	No	No	No
	A	Annual E	missions	s (tons pe	r year)			
Area	1.77	0.03	1.52	<0.01	0	<0.01	0	<0.01
Energy	0.04	0.36	0.15	<0.01	0	0.03	0	0.03
Mobile	0.86	1.20	8.90	0.02	2.47	0.01	0.66	0.01
Total Project Emissions ¹	2.66	1.59	10.57	0.02	2.47	0.05	0.66	0.05
Existing Use Emissions	(0.41)	(0.43)	(3.19)	(<0.01)	(0.81)	(<0.01)	(0.22)	(<0.01)
Net Project Emissions ¹	2.25	1.16	7.38	0.02	1.66	0.05	0.44	0.05
BAAQMD Annual Thresholds	10	10	none	none	none	15	none	10
Exceed Annual Threshold?	No	No	No	No	No	No	No	No

Source: CalEEMod.

¹ Totals may not sum due to rounding.



Draft Environmental Impact Report Environmental Impact Analysis

² Maximum daily emissions of ROG and SOX occur during the summer, maximum daily emissions of NOX and CO occur during the winter, emissions of PM10 and PM2.5 are not seasonally dependent.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure AIR-1 and AIR-2 is required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Sensitive Receptors

Impact AIR-3 The proposed project could expose sensitive receptors to substantial pollutant concentrations.

Impact Analysis

This discussion addresses whether the proposed project would expose sensitive receptors to substantial pollutant concentrations. The localized pollutants that could impact sensitive receptors include: NOA, construction-generated fugitive dust (PM₁₀ and PM_{2.5}), construction generated DPM, CO hotspots and operational-related TACs. Project construction and operational impacts are assessed separately below.

Some land uses are considered more sensitive to air pollution than others due to the types of population groups or activities involved. Heightened sensitivity may be caused by health problems, proximity to the emissions source, and/or duration of exposure to air pollutants. Children, pregnant women, the elderly, and those with existing health problems are especially vulnerable to the effects of air pollution. Accordingly, the following are land uses where sensitive receptors are typically located:

- Long-term health care facilities
- Rehabilitation centers
- Convalescent centers
- Hospitals
- Retirement homes
- Residences
- Schools, playgrounds, and childcare centers

The nearest sensitive receptor is the Silliman Center Preschool. The preschool lies approximately 1,100 feet north of the project site and approximately 100 feet east of the off-site improvements along Mowry Avenue. As a residential development project, the proposed project itself would be considered a sensitive receptor once operational. Most emissions during construction are generated during the site preparation and grading phases when heavy equipment is used to prepare the land for construction. As site preparation and grading are anticipated to occur for the entire project site prior to the completion of



Draft Environmental Impact Report Environmental Impact Analysis

ground-up construction, emissions from grading and site preparation would not overlap with project operation. Earliest residential occupancy is expected to occur in 2024, following the completion of construction. Assuming the project is built in phases, construction activities following site preparation and grading would primarily include building construction, paving, painting, and landscaping. Relative to site preparation and grading activities, limited amounts of diesel equipment are used during these construction activities, which would not contribute substantially to the health risk during construction.

Construction Fugitive Dust

During construction (grading), fugitive dust (PM₁₀ and PM_{2.5}) would be generated from site grading and other earth-moving activities. Most of this fugitive dust will remain localized and will be deposited near the project site.

The BAAQMD does not have a quantitative threshold for fugitive dust. The BAAQMD's Air Quality Guidelines recommend that projects determine the significance for fugitive dust through application of best management practices (BMPs). Mitigation Measure AIR-1 requires the implementation of fugitive dust control measures that are consistent with BMPs established by the BAAQMD, which reduce the project's construction-generated fugitive dust impacts to a less than significant level.

Diesel Particulate Matter

The greatest potential for TAC emissions during construction would be from DPM emissions associated with heavy equipment operations. Heath effects from carcinogenic air toxics, such as DPM, are usually described in terms of an individual cancer risk. An individual cancer risk is the likelihood that a person continuously exposed to concentrations of TACs over a 70-year lifetime will contract cancer. Relative to the lifetime risk, construction would be short-term and temporary and would not result in a long-term source of DPM emissions. The BAAQMD recommends assessing the cancer risk and health hazards posed to sensitive receptors that lie within a 1,000 foot radius from the project site (BAAQMD 2017a). Review of the surrounding project site demonstrates that the nearest receptors are preschoolers located over 1,000 feet to the north of the project site. The preschool is located approximately 100 feet east of the off-site improvements along Mowry Avenue. The off-site improvements would update existing utility lines. Utility construction would move along the existing lines and would not be at any one location along Mowry Avenue for an extended period of time. Therefore, construction DPM emissions would result in a less than significant impact.

Naturally Occurring Asbestos

The California DOC and the United States Geological Survey (USGS) have published a guide for generally identifying areas that are likely to contain NOA. There are no NOA areas located in in the immediate vicinity of the project site. However, the project would require the demolition of an existing facility where there may be potential for asbestos. Implementation of Mitigation Measure HAZ-1, identified in Section 3.9, Hazards and Hazardous Materials, would require an asbestos survey prior to demolition and soil and ground water quality investigations prior to demolition activities and grading activities and reduce impacts to less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Operation

Carbon Monoxide Hotspot

Localized high levels of CO (CO hotspots) are associated with traffic congestion and idling or slowmoving vehicles. The BAAQMD recommends a screening analysis to determine if a project has the potential to contribute to a CO hotspot. The screening criteria identify when site-specific CO dispersion modeling is necessary. The project would result in a less than significant impact to air quality for local CO if any of the following screening criteria are met:

- The project is consistent with an applicable congestion management program established by the county congestion management agency for designated roads or highways, regional transportation plan, and local congestion management agency plans
- The project traffic would not increase traffic volumes at affected intersections to more than 44,000 vehicles per hour
- The project traffic would not increase traffic volumes at affected intersections to more than 24,000 vehicles per hour where vertical and/or horizontal mixing is substantially limited (e.g., tunnel, parking garage, bridge underpass, natural or urban street canyon, below-grade roadway)

Implementation of the proposed project would not conflict with the applicable congestion management program established by the Alameda Transportation Authority. According to the Transportation Impact Analysis Report prepared for the project by Fehr & Peers, the project would generate approximately 136 net new trips during the a.m. peak hour and 147 net new trips during the p.m. peak hour and would not substantially increase traffic volumes on nearby roadways above 44,000 vehicles per hour. Furthermore, the adjacent roadways are not located in an area where vertical and/or horizontal mixing, or the free movement of the air mass, is substantially limited by physical barriers such as bridge overpasses or urban or natural canyon walls. Therefore, the project would not significantly contribute to an existing or projected CO hotspot. Impacts are less than significant.

Toxic Air Contaminants

The proposed project would develop 203 single-family residences and would not generate substantial onsite TAC emissions during operation. Residential land uses are not land uses that are typically associated with TAC emissions and the proposed project does not include any features that would include more than usual TAC emission. As described in the Transportation Impact Analysis Report, the project is expected to generate a net increase of 1,080 average daily trips. The proposed project would primarily generate trips associated with residents and visitors traveling to and from the project site. The daily travel trips to and from the project site would primarily be generated by passenger vehicles. Because nearly all passenger vehicles are gasoline-combusted, the proposed project would not generate significant amount of DPM emissions during operation. Therefore, the proposed project would not result in significant health impacts to nearby sensitive receptors during operation.

Cumulative Health Risk Assessment

The BAAQMD recommends assessing the potential cumulative impacts from sources of TACs within 1,000 feet of a project. For a project-level analysis, the BAAQMD provides three tools for use in screening



Draft Environmental Impact Report Environmental Impact Analysis

potential sources of TACs. The BAAQMD-provided tools that were used to assess the potential cumulative impacts from TACs are described below (BAAQMD 2022a).

- Stationary Source Risk and Hazard Screening Tools. The BAAQMD prepared a geographic information system (GIS) tool with the location of permitted sources. For each emissions source, the BAAQMD provides conservative estimates of cancer risk and PM_{2.5} concentrations. Based on information from the GIS tool, there are three BAAQMD-permitted stationary sources within 1,000 feet of the project site.
- Health Risks for Local Roadways. The BAAQMD pre-calculated concentrations and the associated potential cancer risks and PM_{2.5} concentration increases for each county within their jurisdiction for roadways that carry at least 30,000 average daily trips. For certain areas, the BAAQMD also included local roadways that meet BAAQMD's "major roadway" criteria of 10,000 vehicles or 1,000 trucks per day. The latest available screening tool is in the form of a GIS raster file.
- **Freeway Screening Analysis Tool.** The BAAQMD prepared a GIS raster file that contains preestimated cancer risk and PM_{2.5} concentration increases for highways within the Bay Area.
- **Rail Screening Tool.** The BAAQMD prepared a GIS raster file that contains estimated cancer risks and PM_{2.5} concentrations from railroad operations at any point within the Air Basin.

The project would locate new sensitive receptors (residents) that could be subject to existing sources of TACs at the project site. However, the California Supreme Court concluded in California Building Industry Association v. BAAQMD that agencies subject to CEQA are not required to analyze the impact of existing environmental conditions on a project's future users or residents, except where the project would exacerbate an existing environmental condition. Although impacts from existing sources of TAC emissions on sensitive receptors on the project site are not subject to CEQA, the BAAQMD recommends assessing the potential cumulative impacts from sources of TACs within 1,000 feet of a project when siting new sensitive land uses. Therefore, for informational purposes and in the spirit of CEQA's full disclosure, the potential TAC risks to the project's future residents were analyzed. The BAAQMD's various screening tools, which quantify health risks from existing stationary and permitted sources, were used to estimate the health risks (associated with TAC sources within 1,000 feet of the project site) on future residents within the proposed project.

The cumulative health risk results for future receptors at the project site are summarized at project buildout in Table 3.3-11.

Table 3.3-11: Summary of the	Cumulative Health Impacts	at the Project Site at Project	t
Buildout			

Source	Cancer Risk in One Million	Chronic Inhalation Hazard Index	Annual PM _{2.5} Concentration (μg/m³)				
Existing Sources ¹							
Valassis, FACID 14486	0.01	0.00	0.00				
Apple Inc., FACID 17769_1	8.56	0.00	0.01				



Draft Environmental Impact Report Environmental Impact Analysis

Source	Cancer Risk in One Million	Chronic Inhalation Hazard Index	Annual PM₂.₅ Concentration (μg/m³)
Apple Inc., FACID 17769_2	8.46	0.00	0.01
Apple Inc., FACID 17769_3	8.86	0.00	0.01
Apple Inc., FACID 17769_4	9.01	0.00	0.01
Apple Inc., FACID 17769_5	8.66	0.00	0.01
Apple Inc., FACID 17769_6	5.57	0.00	0.01
Apple Inc., FACID 17769_9	1.08	0.00	0.00
Apple Inc., FACID 17769_10	0.62	0.00	0.00
Apple Inc., FACID 17769_11	1.60	0.00	0.00
Apple Inc., FACID 17769_12	1.32	0.00	0.00
Apple Inc., FACID 17769_13	1.26	0.00	0.00
Apple Inc., FACID 17769_14	1.20	0.00	0.00
Apple Inc., FACID 17769_15	1.45	0.00	0.00
Apple Inc., FACID 17769_16	2.26	0.00	0.00
Existing Roadways	0.50	ND	0.011
Existing Highway 880	3.52	ND	0.064
Cumulative	e Health Risks at the	Site	
Cumulative Total	63.93	0.017	0.15
BAAQMD's Cumulative Thresholds of Significance	100	10	0.8
Threshold Exceedance in Unmitigated Scenario?	No	No	No

Notes:

¹Table accounts for adjustments made for distance of the source from the project site. Multipliers for distance from BAAQMD Health Risk Distance Multiplier (BAAQMD 2022b). Roadways and highways do not include distance multiplier.

 $PM_{2.5}$ = particulate matter of 2.5 microns or less; $\mu g/m^3$ = micrograms per cubic meter; FACID = Facility Identification Number; MEIR = maximally exposed individual receptor; ND = no data available

Source: BAAQMD 2022b

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure AIR-1 and HAZ-1 are required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.


Draft Environmental Impact Report Environmental Impact Analysis

Odors

Impact AIR-4The proposed project would not result in other emissions (such as those leading
to odors) adversely affecting a substantial number of people.

Impact Analysis

As stated in the BAAQMD 2017 Air Quality Guidelines, odors are generally regarded as an annoyance rather than a health hazard and the ability to detect odors varies considerably among the populations and overall is subjective. The BAAQMD does not have a recommended odor threshold for construction activities. However, BAAQMD recommends screening criteria that are based on distance between types of sources known to generate odor and the receptor. The BAAQMD Guidelines identify wastewater treatment plants, oil refineries, asphalt plants, chemical manufacturing, painting/coating operations, coffee roasters, food processing facilities, recycling operations and metal smelters as odor sources of particular concern, and recommends buffer zones of 1 to 2 miles around them to avoid potential odor conflicts. For projects within the screening distances, the BAAQMD has the following threshold for project operations:

• An odor source with five (5) or more confirmed complaints per year averaged over three years is considered to have a significant impact on receptors within the screening distance shown in the BAAQMD's guidance (see Table 3.3-3).

The BAAQMD's 2017 Air Quality Guidelines provide a table with odor screening distances recommended by BAAQMD for a variety of land uses. Projects that would site an odor source or a receptor farther than the applicable screening distance, shown in Table 3.3-12 below, would not likely result in a significant odor impact.

Odor Generator	Distance
Wastewater Treatment Plant	2 miles
Wastewater Pumping Facilities	1 mile
Sanitary Landfill	2 miles
Transfer Station	1 mile
Composting Facility	1 mile
Petroleum Refinery	2 miles
Asphalt Batch Plant	2 miles
Chemical Manufacturing	2 miles
Fiberglass Manufacturing	1 mile
Painting/Coating Operations	1 mile
Rendering Plant	2 miles
Coffee Roaster	1 mile
Food Processing Facility	1 mile
Confined Animal Facility/Feed Lot/Dairy	1 mile
Green Waste and Recycling Operations	1 mile

Table 3.3-12: Screening Levels for Potential Odor Sources

Source: BAAQMD 2017a



Draft Environmental Impact Report Environmental Impact Analysis

Project Construction and Project Operation

The occurrence and severity of odor impacts depend on numerous factors, including the nature, frequency, and intensity of the source; wind speed and direction; and the presence of sensitive receptors. Although offensive odors rarely cause any physical harm, they still can be very unpleasant, leading to considerable distress and often generating citizen complaints to local governments and regulatory agencies. Project operations would not be anticipated to produce odorous emissions. The proposed project consists of a single family home development and would not result in any uses that generate odor sources of concern. Construction activities associated with the proposed project could result in short-term odorous emissions from diesel exhaust associated with construction equipment. However, these emissions would be intermittent and would dissipate rapidly from the source. In addition, this diesel-powered equipment would only be present on-site temporarily during construction activities. Therefore, construction would not create objectionable odors affecting a substantial number of people, and the impact would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation necessary.

Level of Significance After Mitigation Less Than Significant Impact.



3.4 **BIOLOGICAL RESOURCES**

This section describes the environmental and regulatory setting for biological resources which includes aquatic resources. It also describes impacts on biological resources that would result from implementation of the proposed project and mitigation measures for significant impacts.

The analysis in this section is based on the Biological Resources Technical Report and Aquatic Resources Delineation prepared by Helix Environmental Planning, Inc. (Helix) (2022) for the proposed project. These documents are provided in Appendix C of the Draft EIR. The Aquatic Resources Delineation is Appendix G of the Biological Technical Report. Results incorporated into these documents are based on biological and aquatic resources surveys conducted within the study area (the "study area" in this section refers to the 29-acre project site and off-site improvement areas that total 35 acres [Figure 3-1]) from 2019 to 2022 for the proposed project. The 35-acre study area is located in southwestern Alameda County in the City of Newark, California, near the southern end of Mowry Avenue. The area is comprised of three parcels identified as APNs 537-0850-001-11, -13, and -002-00, along with off-site improvements on adjacent lands (Mowry Avenue north of the parcels and north of the UPRR tracks, and a linear area paralleling the southern side of the UPRR tracks). The study area is located on the Newark 7.5-minute USGS quadrangle map within Township 5 South, Range 1 W, Sections 7 and 8, in addition to un-sectioned land not included in the Public Lands Survey. The center of the study area is approximately located at latitude 37.511991 north, longitude -122.011772 west, NAD 83, Mount Diablo Meridian (Figures 3-1).

3.4.1 Environmental Setting

Regional Setting

The study area is within the City of Newark in southwestern Alameda County, California, southwest of the intersection of Mowry Avenue and the UPRR tracks, west of Cherry Street (Figure 2-1).

The site is in un-sectioned lands not included in the Public Lands Survey, adjacent to Township 5 South, Range 1 W, Mount Diablo Meridian. The project site is depicted on the USGS 7.5-minute "Newark, California" quadrangle map.

Project Setting

The study area is located in an area with agricultural and industrial uses in the southwestern portion of the City. The 29-acre project site located at the existing Pick-n-Pull consists of three parcels identified as APNs 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00 (Figure 2-2). The three parcels total 29 acres and the 19-acre portion located in the southern portion of the site is disturbed and developed and primarily covered by existing structures and pavement. The triangular northern parcel which is approximately 10 acres is undeveloped.



This page intentionally left blank.



3-1 Title

Biological Resources Study Area

This page intentionally left blank.

Draft Environmental Impact Report Environmental Impact Analysis

Three land cover types (developed, ruderal/disturbed, and stormwater detention basins) are present within the study area and are described in the following subsections. No native vegetation communities occur within the study area (Figure 3-2). Thirty-eight native wildlife species, 32 bird species, five mammal species, and one amphibian species were observed during the 2019-2022 surveys. Most plant species observed (29 out of 35) during 2019-2022 surveys were non-native, and 11 of them had a California Invasive Plant Council (Cal-IPC) invasiveness rating including limited, moderate, and high (Cal-IPC 2023) (for additional detail refer to Appendix C).

Developed

Developed land refers to areas where permanent structures, pavement, hardscape, or other land uses prevent the growth of vegetation, or where vegetation is associated with maintained landscaping. Developed land on the study area includes auto parts and scrap metal salvage yard, portions of Mowry Avenue and road frontage along Mowry Avenue, and a segment of the UPRR tracks. Developed land generally lacks significant habitat value for plants and wildlife. Wildlife within developed areas is comprised of species that can tolerate regular human disturbance.

Ruderal/Disturbed

Ruderal/disturbed habitat refers to land that retains a soil substrate but is subject to recent or on-going disturbance that prevents the formation of natural vegetation communities. Vegetation in ruderal/disturbed areas is dominated by naturalized or invasive non-native species and ruderal native annuals. The species composition is determined by local colonization potential or past introductions. Ruderal/disturbed areas include dirt roads, trails, parking areas, weedy open areas, abandoned fields, and other places where the natural vegetation has been removed.

Ruderal/disturbed habitat on the study area is dominated by introduced species such as wild oats (*Avena fatua*), Italian ryegrass (*Festuca perennis*), yellow star thistle (*Centaurea solstitialis*), and black mustard (*Brassica nigra*). Ruderal/disturbed areas include a disked field next to the auto wrecking and scrap metal salvage yard, and small strips of habitat adjacent to Mowry Avenue. The ruderal/disturbed habitat of the study area provides marginal nesting and foraging habitat for bird species in the region as well as habitat for disturbance-tolerant wildlife. Striped skunk (*Mephitis mephitis*) and black-tailed jackrabbit (*Lepus californicus*) were observed in the ruderal/disturbed habitat along with numerous bird species including red-tailed hawk (*Buteo jamaicensis*) and red-shouldered hawk (*Buteo lineatus*).

Stormwater Detention Basins

There are two constructed stormwater detention basins on the eastern boundary of the study area, located within the existing Pick-n-Pull site (Figure 3-3). These basins are routinely maintained to remove vegetation for the purpose of maintaining capacity. Between maintenance events, the basins support patches of narrow-leaved cattail (*Typha latifolia*) and other rapidly colonizing wetland plants surrounding reaches of open water. The total area of these basins is approximately 0.90 acre.



This page intentionally left blank.



Biological Resources Habitat Map

This page intentionally left blank.



This page intentionally left blank.

3.4.2 Regulatory Setting

Regulatory authority over biological resources is shared by federal, state, and local authorities under a variety of legislative acts. The following section summarizes the federal, state, and local regulations for special-status species, jurisdiction over waters of the United States and State of California, and sensitive biological resources. This section provides a listing and overview of these federal and state laws; only select regulations are applicable to this project.

Federal

Clean Water Act Sections 404 and 401

Under Section 404 (33 USC. 1344) of the Clean Water Act (CWA), as amended, the U.S. Army Corps of Engineers (USACE) retains primary responsibility for permits to discharge dredged or fill material into waters of the United States. All discharges of dredged or fill material into jurisdictional waters of the United States that result in permanent or temporary losses of waters of the United States are regulated by USACE. A permit from USACE must be obtained before placing fill or grading in wetlands or other waters of the United States, unless the activity is exempt from CWA Section 404 regulation (for example, certain farming and forestry activities).

USACE defines wetlands as "those areas that are inundated or saturated by surface or ground water at a frequency and duration sufficient to support, and that under normal circumstances do support, a prevalence of vegetation typically adapted for life in saturated soil conditions" (USACE 1987).

In other words, the USACE defines wetlands by the presence of all three wetland indicators: hydrophytic vegetation, hydric soils, and wetlands hydrology.

Waters of the United States are defined at 33 CFR Part 328. They include traditional navigable waters; impoundments of Waters of the United States; tributaries of traditional navigable waters; certain wetlands; and certain intrastate lakes, ponds, and streams. The applicability of Section 404 permitting over discharges to wetlands is, therefore, a two-step process: (1) determining the areas that are wetlands, and (2) where a wetland is present, assessing the wetland's connection to traditional navigable waters and non-navigable tributaries to determine whether the wetland is jurisdictional under the CWA. A wetland is considered jurisdictional if it meets certain specified criteria.

USACE is required to consult with the U.S. Fish and Wildlife Service (USFWS) and/or National Marine Fisheries Service (NMFS) under Section 7 of the federal Endangered Species Act (FESA) if the action subject to CWA permitting could result in "Take" of federally listed species or an adverse effect to designated critical habitat. A "Take" is defined as harassing, harming (including significantly modifying or degrading habitat), pursuing, hunting, shooting, wounding, killing, trapping, capturing, or collecting wildlife species, or any attempt to engage in such conduct (16 USC 1531; 50 CFR 17.3). An activity can be defined as a Take, even if it is unintentional or accidental. Taking can result in civil or criminal penalties. The proposed project is within the jurisdiction of the Sacramento District of USACE.

Section 401 of the CWA (33 USC 1341) requires any applicant for a federal license or permit to conduct any activity that may result in a discharge of a pollutant into waters of the United States to obtain a certification from the state in which the discharge originates or would originate, or, if appropriate, from the



Draft Environmental Impact Report Environmental Impact Analysis

interstate water pollution control agency having jurisdiction over the affected waters at the point where the discharge originates or would originate. The discharge must comply with the applicable effluent limitations and water quality standards. A certification obtained for the construction of any facility must also pertain to the subsequent operation of the facility. The responsibility for the protection of water quality in California rests with the SWRCB and its nine RWQCBs. The proposed project is within the jurisdiction of the San Francisco Bay RWQCB.

Fish and Wildlife Coordination Act

The Fish and Wildlife Coordination Act (16 USC Sections 661-667e, March 10, 1994, as amended 1946, 1958, 1978, and 1995) requires that whenever waters or channel of a stream or other body of water are proposed or authorized to be modified by a public or private agency under a federal license or permit, the federal agency must first consult with USFWS and/or the NMFS, and with the head of the agency exercising administration over the wildlife resources of the state where construction would occur (in this case, the California Department of Fish and Wildlife [CDFW]). The Fish and Wildlife Coordination Act is intended to conserve birds, fish, mammals and all other classes of wild animals and all types of aquatic and land vegetation upon which wildlife is dependent.

If direct, permanent impacts occur to waters of the United States from a proposed project, then a permit from USACE under CWA Section 404 is required for the construction of the proposed project. USACE is required to consult with USFWS and/or NMFS as appropriate regarding potential impacts to federally listed species under the FESA. Such action may prompt consultation with CDFW, which would review the proposed project pursuant to California Endangered Species Act (CESA) and issue a consistency letter with USFWS and/or NMFS, if required.

Federal Endangered Species Act

The United States Congress passed the FESA in 1973 to protect species that are endangered or threatened with extinction. The FESA is intended to operate in conjunction with the National Environmental Policy Act to help protect the ecosystems upon which endangered and threatened species depend and within which they live. The USFWS and the NMFS are the designated federal agencies responsible for administering the FESA.

The FESA prohibits the Take of endangered or threatened wildlife species. Activities that could result in Take of a federally listed species require an incidental Take authorization resulting from a FESA Section 7 consultation or an FESA Section 10 consultation. Plants are legally protected under the FESA only if Take occurs on federal land or from federal actions, such as issuing a wetland fill permit.

A federally endangered species is one that is considered in danger of becoming extinct throughout all, or a significant portion, of its range. A federally threatened species is one that is likely to become endangered in the foreseeable future. The USFWS also maintains a list of species proposed for listing as threatened or endangered. Proposed species are those for which a proposed rule to list as endangered or threatened has been published in the Federal Register. In addition to endangered, threatened, and proposed species, the USFWS maintains a list of candidate species. Candidate species are those for which the USFWS has on file sufficient information to support issuance of a proposed listing rule.



Draft Environmental Impact Report Environmental Impact Analysis

Pursuant to the requirements of the FESA, an agency reviewing a proposed project within its jurisdiction must determine whether any federally listed endangered or threatened species may be present in the project area and determine whether the proposed project would have a potentially significant impact on such a species. In addition, the agency is required to determine whether the proposed project is likely to jeopardize the continued existence of any species proposed to be listed under the FESA or result in the destruction or adverse modification of critical habitat designated or proposed to be designated for such species (16 USC 1536[3], [4]). Project-related impacts to species on the FESA endangered or threatened list would be considered significant and would require mitigation.

Migratory Bird Treaty Act

The Migratory Bird Treaty Act (MBTA) of 1918 established federal responsibilities for the protection of nearly all species of birds, their eggs, and nests. The Migratory Bird Treaty Reform Act of 2004 further defined species protected under the act and excluded all non-native species. Section 16 USC 703–712 of the Act states "unless and except as permitted by regulations, it shall be unlawful at any time, by any means or in any manner, to pursue, hunt, take, capture, kill, attempt to take, capture, or kill" a migratory bird. A migratory bird is any species or family of birds that live, reproduce, or migrate within or across international borders at some point during their annual life cycle. Currently, there are 836 migratory birds protected nationwide by the Migratory Bird Treaty Act, of which 58 are legal to hunt. The MBTA also prohibits disturbance and harassment of nesting migratory birds at any time during their breeding season. The USFWS is responsible for enforcing the MBTA (16 USC 703). The migratory bird nesting season is generally considered to be between February 1 and September 15 and earlier for raptors.

State

Porter-Cologne Water Quality Act

The state and RWQCB also maintain independent regulatory authority over the placement of waste, including fill, into waters of the State under the Porter-Cologne Act. Waters of the State are defined by the Porter-Cologne Act as "any surface water or groundwater, including saline waters, within the boundaries of the state." The SWRCB protects all waters in its regulatory scope but has special responsibility for isolated wetlands and headwaters. These water bodies might not be regulated by other programs, such as Section 404 of the CWA. Waters of the State are regulated by the RWQCBs under the State Water Quality Certification Program, which regulates discharges of dredged and fill material under Section 401 of the CWA and the Porter-Cologne Water Quality Control Act. Projects that require a USACE permit, or fall under other federal jurisdiction, and have the potential to impact waters of the State, are required to comply with the terms of the Water Quality Certification Program. If a proposed project does not require a federal license or permit but does involve activities that may result in a discharge of harmful substances to waters of the State, the RWQCBs have the option to regulate such activities under their state authority in the form of Waste Discharge Requirements (WDRs) or certification of WDRs.

California Endangered Species Act

The state enacted the CESA in 1984. The CESA is similar to the FESA but pertains to state-listed endangered and threatened species. The CDFW is responsible for maintaining a list of threatened and endangered species designated under state law (California Fish and Game Code [CFGC] 2070). Section 2080 of the CFGC prohibits Take of any species that the commission determines to be an endangered or



Draft Environmental Impact Report Environmental Impact Analysis

threatened species. Take is defined in Section 86 of the CFGC as "to hunt, purse, catch, capture, or kill, or attempt to hunt, purse, catch, capture, or kill."

The state and federal lists of threatened and endangered species are generally similar; however, a species present on one list may be absent from the other. The CESA regulations are also somewhat different from the ESA in that the state regulations include threatened, endangered, and candidate plants on non-federal lands within the definition of Take. The CESA allows for incidental Take resulting from otherwise lawful development projects.

Pursuant to the requirements of the CESA, an agency reviewing a proposed project within its jurisdiction must determine whether any state-listed endangered or threatened species may be present in the proposed project area and determine whether the proposed project would have a potentially significant impact on such species. Project-related impacts to species on the CESA endangered or threatened list (or, in addition, designated by the CDFW as a "Species of Special Concern (SSC)," which is a level below threatened or endangered status) would be considered significant and would require mitigation.

California Environmental Quality Act

CEQA Guidelines Sections 15125(c) and 15380(d) provide that a species not listed on the federal or state list of protected species may be considered rare or endangered if the species can be shown to meet certain specified criteria. Thus, CEQA provides the ability to protect a species from potential project impacts until the respective government agencies have an opportunity to designate the species as protected, if warranted.

The California Native Plant Society (CNPS) maintains a list of plant species native to California whose populations that are significantly reduced from historical levels, occur in limited distribution, or are otherwise rare or threatened with extinction. This information is in the online Rare Plant Inventory (CNPS 2021). Taxa with a California Rare Plant Rank (CRPR) of 1A, 1B, 2A, 2B, and 3 in the CNPS inventory consist of plants that meet the definitions of the CESA of the CFGC, are eligible for state listing, and meet the definition of Rare or Endangered under CEQA Guidelines Sections 15125 (c) and 15380(d). Some taxa with a CRPR 4 may meet the definitions of the CESA of the CFGC. Populations with a CRPR of 4 may qualify for consideration under CEQA if they are peripheral or disjunct populations, represent the type locality of the species, or exhibit unusual morphology and/or occur on unusual substrates.

Additionally, CDFW maintains lists of special animals and plants. These lists include a species conservation ranking status from multiple sources, including FESA, CESA, federal departments with unique jurisdictions, CNPS, and other non-governmental organizations. Based on these sources, CDFW assigns a heritage rank to each species according to their degree of imperilment (as measured by rarity, trends, and threats). These ranks follow NatureServe's Heritage Methodology, in which all species are listed with a G (global) and S (state) rank. Species with state ranks of SI-S3 are also considered highly imperiled.

CEQA Guidelines Appendix G checklist section IV (b) calls for the consideration of riparian habitats and sensitive natural communities. Sensitive vegetation communities are natural communities and habitats that are either unique, of relatively limited distribution in the region, or of particularly high wildlife value. However, these communities may or may not necessarily contain special-status species. Sensitive natural communities are usually identified in local or regional plans, policies, or regulations, or by the CDFW (i.e.,

Draft Environmental Impact Report Environmental Impact Analysis

the California Natural Diversity Database [CNDDB] and VegCAMP programs) or the USFWS. Impacts to sensitive natural communities and habitats must be considered and evaluated under CEQA (CCR Title 14, Div. 6, Chap. 3, Appendix G).

Although sensitive natural communities do not (at present) have legal protection, CEQA calls for an assessment of whether any such resources would be affected and requires a finding of significance if there would be substantial losses. High quality occurrences of natural communities with heritage ranks of 3 or lower are considered by CDFW to be significant resources and fall under the CEQA Guidelines for addressing impacts. Local planning documents (such as general plans) often identify these resources as well. Avoidance, minimizations, or mitigation measures should be implemented if project-affected stands of rare vegetation types or natural communities are considered high-quality occurrences of the given community.

As a trustee agency under CEQA, CDFW reviews potential project impacts to biological resources, including wetlands. In accordance with the CEQA thresholds of significance for biological resources, areas that meet the state criteria of wetlands and could be impacted by a project must be analyzed. Pursuant to CFGC Section 2785, CDFW defines wet areas as "lands which may be covered periodically or permanently with shallow water and which include saltwater marshes, freshwater marshes, open or closed brackish water marshes, swamps, mudflats, fens, and vernal pools."

California Fish and Wildlife Code Section 1600

Streams, lakes, and riparian vegetation as habitat for fish and other wildlife species are subject to jurisdiction by CDFW under Sections 1600-1616 of the CFGC with regard to any activity that would do one or more of the following: (1) substantially obstruct or divert the natural flow of a river, stream, or lake; (2) substantially change or use any material from the bed, channel, or bank of a river, stream, or lake; or (3) deposit or dispose of debris, waste, or other material containing crumbled, flaked, or ground pavement where it can pass into a river, stream, or lake generally require a Streambed Alteration Agreement (SAA).

The term "stream," which includes creeks and rivers, is defined in the CCR as follows: "a body of water that flows at least periodically or intermittently through a bed or channel having banks and supports fish or other aquatic life." This includes watercourses having a surface or subsurface flow that supports or has supported riparian vegetation (14 CCR 1.72).

In addition, the term stream can include ephemeral streams, dry washes, watercourses with subsurface flows, canals, aqueducts, irrigation ditches, and other means of water conveyance if they support aquatic life, riparian vegetation, or stream-dependent terrestrial wildlife. Riparian is defined as "on, or pertaining to, the banks of a stream;" therefore, riparian vegetation is defined as, "vegetation which occurs in and/or adjacent to a stream and is dependent on, and occurs because of, the stream itself." Removal of riparian vegetation also requires an SAA from CDFW.

California Fish and Wildlife Code Sections 3503 and 3513

According to Section 3503 of the CFGC, it is unlawful to take, possess, or needlessly destroy the nest or eggs of any bird (except house sparrows [*Passer domesticus*] and European starlings [*Sturnus vulgaris*]). Section 3503.5 specifically protects birds in the orders Falconiformes and Strigiformes (birds-of-prey). Section 3513 essentially overlaps with the MBTA, prohibiting the Take or possession of any migratory



Draft Environmental Impact Report Environmental Impact Analysis

non-game bird. Disturbance that causes nest abandonment and/or loss of reproductive effort is considered Take by the CDFW.

Fully Protected Species and Species of Special Concern

The classification of "fully protected" was CDFW's initial effort to identify and provide additional protection to those animals that were rare or faced possible extinction. Lists were created for fish, amphibian and reptiles, birds, and mammals. Most of the species on these lists have subsequently been listed under CESA and/or FESA. The CFGC sections (fish at Sec. 5515, amphibian and reptiles at Sec. 5050, birds at Sec. 3511, and mammals at Sec. 4700) dealing with "fully protected" species states that these species "... may not be taken or possessed at any time and no provision of this code or any other law shall be construed to authorize the issuance of permits or licenses to take any fully protected species," although Take may be authorized for necessary scientific research. This language makes the "fully protected" designation the strongest and most restrictive regarding the Take of these species. In 2003, the code sections dealing with fully protected species were amended to allow CDFW to authorize Take resulting from recovery activities for state-listed species.

SSC are broadly defined as animals not listed under the CESA, but that are nonetheless of concern to CDFW because they are declining at a rate that could result in listing, or historically occurred in low numbers and known threats to their persistence currently exist. This designation is intended to result in special consideration for these animals by CDFW, land managers, consulting biologists, and others, and is intended to focus attention on the species to help avert the need for costly listing under CESA and cumbersome recovery efforts that might ultimately be required. This designation also is intended to stimulate collection of additional information on the biology, distribution, and status of poorly known at-risk species, and focus research and management attention on them. Although the SSC designation provides no special legal status, they are given special consideration under CEQA during project review.

California Native Plant Protection Act of 1977

The California Native Plant Protection Act (NPPA) of 1977 (CFGC Sections 1900-1913) empowers the Fish and Game Commission to list native plant species, subspecies, or varieties as endangered or rare following a public hearing. To the extent that the location of such plants is known, CDFW must notify property owners that a listed plant is known to occur on their property. Where a property owner has been so notified by CDFW, the owner must notify CDFW at least 10 days in advance of any change in land use (other than changing from one agricultural use to another), so that CDFW may salvage listed plants that would otherwise be destroyed. Currently, 64 species of native plants have been listed as rare under the NPPA.

Natural Community Conservation Planning Act

The primary objective of the Natural Community Conservation Planning (NCCP) Act of 1991 is to conserve natural communities at the ecosystem scale while accommodating compatible land use. The NCCP Act is an effort by the state and numerous private and public partners that is broader in its orientation and objectives than the CESA and FESA (refer to discussions above). The NCCP Act seeks to anticipate and prevent the controversies and gridlock caused by species listings by focusing on the long-term stability of wildlife and plant communities and including key interests in the process.



Local

City of Newark General Plan

The City's General Plan includes the following goals and policies relevant to the proposed project:

Goal LU-7: Southwest Newark Residential and Recreational Project. Develop the Southwest Newark Residential and Recreational Project as one of the Silicon Valley's premier new neighborhoods, with executive housing and high-quality recreation.

- **Policy LU-7.2: Wetland Enhancement.** Create or enhance wetland habitat areas within nondeveloped portions of the Southwest Newark project area to offset loss of wetlands and aquatic habitat and provide additional habitat opportunities for rare plant and wildlife species.
- **Policy LU-7.3: Biological Resources Protection.** Maintain, protect, and enhance the natural biological resources of the Southwest Newark Residential and Recreational Project Areas, particularly sensitive habitats and associated rare plants and animals, while integrating development and human activity. Disturbance of wetland and aquatic habitat should be avoided to the maximum extent feasible.
- **Policy LU-7.4: Controlling Invasive Plants.** Avoid the introduction and spread of non-native invasive weeds as a result of development activities in this area. Require management plans to control the population of invasive species prior to grading, fill, and development activities.
- **Policy LU-7.8: Mitigating Construction Impacts.** Avoid and mitigate construction impacts on wetlands, aquatic habitat, wildlife, and water quality as development takes place in the Southwest Newark Residential and Recreational Project. Measures to minimize such impacts should be included in project approvals, consistent with state and federal agency oversight and regulations.

Goal CS-1: Environmental Protection. Protect Newark's natural environment, landscape, and physical features.

- **Policy CS-1.1: Environmental Impacts of Development.** Ensure that development minimizes its impacts on Newark's environment and natural resources through sound planning, design, and management.
- **Policy CS-1.2: Conservation of Sensitive Areas.** Support the conservation of environmentally sensitive areas and unique natural resources in the city.

Goal CS-2: Wetland Conservation. Conserve Newark's wetlands and bay lands.

- **Policy CS-2.1:** Wildlife and Habitat Protection. Preserve and protect Newark's plant and animal species and habitats, including wetlands, salt marshes, creeks, and lakes. Ensure that land use decisions avoid and mitigate potential impacts on wildlife habitat to the extent feasible.
- **Policy CS-2.2:** Special-Status Species. Ensure that adverse impacts on special-status species, including those deemed rare, threatened, endangered, or candidate species for protection, are avoided and mitigated to the greatest extent feasible as development takes place.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy CS-2.4: Wetlands Delineation.** Encourage the owners of large potentially developable properties to enter into early discussions with appropriate federal agencies to conduct wetlands delineation studies. Such studies should be used to identify areas to be conserved as permanent open space, as well as appropriate mitigation measures to offset any wetland impacts.
- **Policy CS-2.5**: Development Near Wetlands. Manage land use and development of upland sites in a manner that minimizes off-site impacts to nearby wetlands.

Goal CS-4: Urban Forest. Conserve and manage the City's tree resources and urban forest.

- **Policy CS-4.1: Tree Preservation.** Maintain and improve City programs for protecting and preserving trees.
- **Policy CS-4.2: Trees and Public Improvements.** Manage the City's trees in a way that preserves the life of public improvements such as curbs, gutters, and sidewalks. Ensure that trees that are removed due to their age, health, or potential to damage property, are replaced in kind with new trees that are appropriate for their locations.
- **Policy CS-4.4: Street Trees as Community Amenity.** Encourage the use of street trees and landscaping to distinguish major thoroughfares and neighborhoods, beautify the city, encourage walking, and create a stronger sense of identity.

3.4.3 Environmental Impacts

This section analyzes the project's potential to result in significant biological impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology

Biological surveys conducted in 2019-2022 included a general biological survey, special-status plant surveys (botanical surveys), a burrowing owl (*Athene cunicularia*) habitat assessment and protocol surveys, an assessment of black rail (*Laterallus jamaicensis*) and other special-status bird species with the potential to occur in adjacent marsh habitats, an arborist inventory, and an aquatic resources delineation (Table 3.4-1). Surveys within the approximately 35-acre study area began on January 4, 2019. Most of the surveys were conducted in 2019. Two surveys occurred in 2021 and the most recent survey occurred on March 8, 2022. Table 3.4-1 presents the surveys conducted and the date conducted. The results of this assessment are documented in the Biological Resources Technical Report (Appendix C) and include a table of species observed within the study area.

Table 3.4-1 Biological	Surveys Condu	ucted for The Pro	posed Project
------------------------	---------------	-------------------	---------------

Date	Survey
January 4, 2019	general biological survey,
	jurisdictional delineation,
	burrowing owl habitat assessment
April 16, 2019	Arborist inventory,
	botanical survey,



Draft Environmental Impact Report Environmental Impact Analysis

	burrowing owl survey	
May 2, 2019	botanical survey	
May 8, 2019	botanical survey	
May 22, 2019	botanical survey	
May 23, 2019	burrowing owl survey	
June 18, 2019	burrowing owl survey	
June 18, 2019	burrowing owl survey	
July 3, 2019	jurisdictional delineation	
November 16, 2021	general biological survey, burrowing owl survey, jurisdictional delineation of off-site improvement areas	
December 8, 2021	Arborist inventory of off-site improvements	
March 8, 2022	general biological, botanical surveys	

Prior to conducting field surveys, the following resources were reviewed to determine the potential for the project to impact sensitive biological resources:

- California Department of Fish and Wildlife (CDFW). California Natural Diversity Database (CNDDB); For: San Leandro, Hayward, Dublin, Redwood Point, Newark, Niles, Palo Alto, Mountain View, and Milpitas USGS 7.5-minute series quadrangles, Sacramento, CA (CDFW 2021)
- California Native Plant Society (CNPS). Inventory of Rare and Endangered Plants (online edition, v8-03 0.39) For: San Leandro, Hayward, Dublin, Redwood Point, Newark, Niles, Palo Alto, Mountain View, and Milpitas USGS 7.5-minute series quadrangles, Sacramento, CA (CNPS 2021)
- U.S. Fish and Wildlife Service (USFWS). 2. Information for Planning and Consultation (IPaC) for the Proposed Project (USFWS 2021a)
- USFWS National Wetlands Inventory (USFWS 2021b)

Information for this section also references the following two sources that document previous biological and wetland studies conducted within the study area:

- Recovery Plan for Tidal Marsh Ecosystems of Northern and Central California (USFWS 2013)
- The Baylands and Climate Change: What We Can Do. Baylands Ecosystem Habitat Goals Science Update 2015 prepared by the San Francisco Bay Area Wetlands Ecosystem Goals Project (San Francisco Bay Area Wetlands Ecosystem Goals Project 2015)

A list of special-status species with the potential to occur in the study area was compiled by performing a CNDDB query for the USGS quadrangle containing the study area (Newark) and the eight surrounding quadrangles (*San Leandro, Hayward, Dublin, Redwood Point, Niles, Palo Alto, Mountain View, and Milpitas*) and reviewing species data provided by the USFWS.



Draft Environmental Impact Report Environmental Impact Analysis

The study area lacks any form of a natural habitat corridor (e.g., riparian areas along streams, rivers, or other natural features) that would allow for wide ranging plants and animals from other habitats ingress and egress to the study area.

Special-Status Species

The following sections describe the potential for special-status species to occur within the study area.

Special-Status Plants

Special-status plant species are defined in accordance with the CEQA Guidelines, Section 15380 and the *Protocols for Surveying and Evaluating Impacts to Special-status Native Plant Populations and Natural Communities* (CDFW 2018) and include species that are:

- Federally or state-listed, or proposed for listing, as rare, threatened, or endangered (CDFW 2021),
- Special Plant as defined by the CNDDB (CDFW 2021), or
- Listed by the CNPS in the online version of its Inventory of Rare and Endangered Plants of California with a California Rare Plant Ranking (CDFW 2021).

The CNDDB query returned a list of 45 special-status plant species (CDFW 2021). The USFWS data did not include any additional special-status plant species. A table of the 45 species is included as an Appendix in the Biological Resources Report and includes a potential to occur column (Appendix C). The Biological Resources Technical Report considered the distances of mapped sensitive plant occurrences from the study area and the conditions on-site to determine that the study area does not contain suitable habitat for any of the special-status plant species that resulted from the CNDDB query because the study area does not contain any native vegetation communities.

Special-Status Wildlife

Special-status wildlife species are defined in accordance with the CEQA Guidelines, Section 15380 and include species that are:

- Listed, proposed for listing, or candidates for listing as threatened or endangered under the federal Endangered Species Act (USFWS 2021a)
- Listed or candidates for listing as threatened or endangered under the California Endangered Species Act (CDFW 2021)
- Designated as SSC by the CDFW (CDFW 2021)
- Designated as Fully Protected by the CDFW (CDFW 2021), or
- Otherwise meet the definition of rare, threatened, or endangered, as described in the CEQA Guidelines, Section 15380.



Draft Environmental Impact Report Environmental Impact Analysis

The CNDDB search performed as part of the Biological Resources Technical Report Section 3-71 returned a list of 63 special-status wildlife species (Appendix C). The USFWS data called out an additional four federally listed species. Based on the field surveys, the following five special-status wildlife species were determined to have the moderate or high potential to occur within the study area (Appendix C): northern harrier (*Circus hudsonius*), white-tailed kite (*Elanus leucurus*), pallid bat (*Antrozous pallidus*), Townsend's big-eared bat (*Corynorhinus townsendii*), and burrowing owl (*Athene cunicularia*) (Table 3.4-2). Suitable salt marsh habitat for salt marsh harvest mouse (*Reithrodontomys raviventris*) does not occur within the study area but is located within 300 feet of the study area. The acreage believed to be necessary to sustain a healthy salt marsh harvest mouse population is 150 acres or more (USFWS 2013, 2010). A 14-inch-tall exclusion fence is located along the southern border of the study area.

Scientific Name Common Name	State Status ¹	Habitat Suitability
<i>Antrozous pallidus</i> pallid bat	SSC	Structures and trees in the study area provide roosting habitat for bats.
Athene cunicularia burrowing owl	SSC	There is suitable habitat with mammal burrows in the ruderal/disturbed portion of the study area.
<i>Circus hudsonius</i> northern harrier	SSC	There is suitable nesting and foraging habitat in the study area and surrounding area.
<i>Corynorhinus townsendii</i> Townsend's big-eared bat	SSC	Structures and trees in the study area provide roosting habitat for bats.
<i>Elanus leucurus</i> white-tailed kite	FP	There is suitable nesting and foraging habitat in the study area and surrounding area.
Reithrodontomys raviventris salt marsh harvest mouse ²	FE/SE	There is no suitable habitat in the study area. However, nearby tidal salt marsh habitat may provide habitat.

Table 3.4-2 Regionally Occurring Special-Status Species with Potential to Occur

Notes: ¹ FP=State fully protected; SSC – state species of special concern; FE – FESA endangered; SE – CESA endangered ² Salt Marsh Harvest Mouse has no potential to occur on the site but is evaluated due to the presence of salt marsh habitat in the vicinity.

Pallid Bat (Antrozous pallidus)

The pallid bat is a CDFW SSC that occurs throughout California except for the high Sierra Nevada and the northern Coast Ranges. Habitats include grasslands, shrublands, woodlands, and forests from sea level to 6,000 feet. This species is most common in open, dry habitats with rocky areas for roosting; roosts also include cliffs, abandoned buildings, bird boxes, hollow trees and under bridges (Bolster ed. 1998). This species is primarily a crevice dweller, but recent studies have shown that they are also dependent upon tree roosts (Bolster ed. 1998). Particularly, in northern California pallid bat is more dependent upon oak woodland and oak savannah in lower elevations and may be found in coniferous forest, including redwoods at mid to higher elevations (Bolster ed. 1998). This is species is also intolerant of roost disturbance and it has a high loyalty to roosting sites. If this species experiences frequent disturbance at a roost site, they will abandon the roost (Bolster ed. 1998).



Draft Environmental Impact Report Environmental Impact Analysis

No pallid bat or other bat species were observed on site during surveys conducted within the study area from 1999 to 2021. The nearest CNDDB reported occurrence of pallid bat is located approximately seven miles east of the study area in in the coastal mountains (CDFW 2021). The record is dated 2001 and the site is confidential (CDFW 2021). The site is located at approximately 400 feet elevation in riparian, coastal oak woodland, and non-native annual grassland (CDFW 2021).

Ruderal habitat dominated by non-native annual grasses and forbs in the study area provides potential foraging habitat for pallid bat and tall gum/eucalyptus (*Eucalyptus* sp.) trees in the study area with woodpecker cavities provide possible roosting habitat. Pallid bat could also utilize crevices in structures at the Pick-n-Pull auto salvage wrecking yard; although, any roost site would be subject to constant disturbance as a result of daily activities associated with wrecking and salvaging vehicles. Due to the presence of marginally suitable habitat within the study area and adjacent to the site, pallid bat has the potential to occupy the site prior to construction.

Townsend's Big-Eared Bat (Corynorhinus townsendii)

The Townsend's big-eared bat, a CDFW SSC, is widely distributed throughout California except alpine and subalpine habitats. This species eats moths, beetle, and other insects which it catches on the wing or by gleaning from vegetation and is typically found near water. This species uses caves, mines, tunnels, buildings, and human made structures for roosting. Maternity roosts are typically in warm sites. Hibernation sites are typically cold, but not freezing. This species is very sensitive to disturbance and may abandon its roost after one visit (Zeiner et al. 1990).

No Townsend's big-eared bat or other bat species were observed on site during the 1999 to 2022 biological surveys. The nearest reported occurrence of the species is 14.2 miles west of the study area in Portola Valley (CDFW 2021).

Ruderal habitat dominated by annual grasses and forbs in the study area provides potential foraging habitat for Townsend's big-eared bat and tall eucalyptus trees in the study area with woodpecker cavities provide possible roosting habitat. Townsend's big-eared bat could also utilize crevices in structures at the Pick-n-Pull auto salvage wrecking yard, although any roost site would be subject to constant disturbance as a result of daily activities associated with wrecking and salvaging vehicles. Due to the presence of marginally suitable habitat within the study area and adjacent to the site, Townsend's big-eared bat has the potential to occupy the site prior to construction.

Burrowing Owl (Athene cunicularia)

Burrowing owl, a CDFW SSC, is often found in open, dry grasslands, agricultural and range lands, and desert habitats. It can also inhabit grass, forb, and shrub stages of pinyon and ponderosa pine habitats. Burrowing owl occurs at elevations ranging from 200 feet below mean sea level to over 9,000 feet AMSL. In California, the highest elevation where burrowing owls are known to occur is 5,300 feet AMSL in Lassen County. In addition to natural habitats, burrowing owl can be found in urban habitats such as at the margins of airports and golf courses and in vacant urban lots. Burrowing owl nests in underground burrows and commonly perch on nearby fence posts or mounds. The owl also uses ground squirrel (*Otospermophilus beecheyi*) burrows, badger (*Taxidea taxus*) dens, or artificial burrows such as a abandoned pipes or culverts (CDFG 2012).



Draft Environmental Impact Report Environmental Impact Analysis

Although, the more northern burrowing owl populations migrate seasonally, burrowing owl is a year-round resident in much of California. The owl often forms loose colonies with nest burrows 46 to 2,952 feet apart (ICF 2012). The nesting season for burrowing owl can begin as early as February 1 and continues through August 31. Burrowing owl forages in adjacent grasslands and other suitable habitats primarily for insects and small mammals and less often for reptiles, amphibians, and other small birds. Burrowing owl has been documented foraging up to 1.7 miles from its nest in Saskatchewan, Canada; however, this owl also showed an aversion to foraging in agricultural or other mixed-use areas (Haug and Oliphant 1990). In the southern Central Valley of California and Imperial Valley over 80 percent of foraging activity occurred within 2,000 feet of the burrow (Gervais et al. 2003). A study in Texas documented that foraging distances in an urban environment ranged from approximately 30 to 130 feet from the nest burrow (Chipman et al. 2008). The Texas study also noted that burrowing owl in urban settings tended to be more vigilant at the burrow and spend less time foraging (Chipman et al. 2008).

All biological surveys included searching for burrowing owl. In addition, protocol burrowing owl surveys were conducted in 2019 on April 16, May 2, May 23, and June 18. A non-breeding burrowing owl survey was conducted on November 16, 2021, and an additional biological reconnaissance survey was conducted on March 8, 2022. Burrowing owl, burrows, or sign were not observed within the study area during any of the surveys.

The nearest CNDDB recorded occurrence of burrowing owl is 875 feet north of the study area on the Campus of Ohlone College near Cherry Street, dated 2005 (Occurrence No. 270) (CDFW 2021). Two active burrows were observed at this location in 1998. Four adult pairs and nine juveniles were passively relocated in February 2005 (CDFW 2021). There are numerous CNDDB occurrences of burrowing owl in the western portions of the cities of Newark and Fremont (CDFW 2021); HELIX biologists have also observed burrowing owls on several occasions within two to three miles of the site. There is a small resident population of burrowing owl in the city and transient owls are frequently observed on undeveloped parcels in the region (Helix 2022).

The ruderal/disturbed areas of the study area, which occur within a largely urban area adjacent to an auto salvage yard, provide marginal nesting and foraging habitat for burrowing owl. However, this species has not been observed within the study area during numerous surveys. Based on the negative results of the surveys, burrowing owl is not expected to occur in the study area except as transient, nonbreeding individuals.

The only vegetation community on site that could provide habitat for burrowing owl is the ruderal/ disturbed community in the northern 1/3 of the Pick-n-Pull property, which is dominated by wild oats, Italian ryegrass, yellow star thistle, and black mustard. Burrowing owl habitat typically consists of short, sparse vegetation with scattered and isolated shrubs in locally flat terrain in well-drained soil with mammal burrows or other refuge sites (CDFG 2012). Records of burrowing owl or observations of burrowing owl sign within the previous three years is considered evidence of occupied habitat (CDFG 2012). Because there are no records of burrowing owl in the study area and no burrowing owl sign was observed during numerous biological surveys including protocol surveys, the potentially suitable habitat in the study area is presumed to be unoccupied. Additionally, the site is regularly disked, and existing California ground squirrel burrows are disked and turned over. Burrowing owls were not observed over several site visits timed to coincide with the activity period of burrowing owls during the breeding season. The site is not



Draft Environmental Impact Report Environmental Impact Analysis

favorable for burrowing owls compared to nearby sites and is unlikely to provide habitat for burrowing owl except as transient individuals moving through the site.

Northern Harrier (Circus hudsonius)

Northern harrier, a CDFW SSC, is widespread throughout North America from southern Canada to northern Mexico and is a year-round resident in California. Some harriers will migrate into California while others will migrate to Central America and South America (Zeiner et al. 1990). Northern harrier breeds in a variety of open habitats including marshes, wet meadows, weedy shorelines, grasslands, weed fields, pastures, sagebrush flats, desert sinks, and croplands (Zeiner et al. 1990). Northern harrier typically nests on the ground in patches of dense, tall vegetation in undisturbed areas. Harrier breeding occurs from March to August. Northern harrier feeds on a wide variety of vertebrate prey, including rodents, songbirds, waterfowl, and lizards (Zeiner et al. 1990).

Northern harrier was not observed on site during the 2019-2022 biological surveys. The nearest CNDDB occurrence of northern harrier documents two nests located approximately three miles southwest of the study area in salt marsh habitat (CDFW 2021). The record is dated 1971 and documents two pairs of breeding northern harriers, each nest with six eggs (CDFW 2021). HELIX biologists have observed foraging northern harrier on numerous occasions in the project region, but no nests have been documented (Helix 2022).

Ruderal habitat dominated by annual grasses and forbs in the study area provides potential nesting and foraging habitat for northern harrier. Freshwater marsh and salt marsh habitat on surrounding parcels provide suitable nesting habitat. In addition, small mammal prey is abundant and could support this species. Due to the presence of suitable habitat on and adjacent to the site, northern harrier has the potential to occupy the site prior to construction.

White-tailed Kite (Elanus leucurus)

White-tailed kite, a CDFW fully protected species, is a year-round resident in coastal and valley lowlands, where it inhabits herbaceous and open stages of most habitat types. Individuals forage in grasslands, farmlands, and wetlands, preying mostly on small diurnal mammals. Nests are built near the top of dense tree stands, usually near open foraging areas (Zeiner et al. 1990).

White-tailed kite was observed foraging in the vicinity of the study area during several of the biological surveys. No white-tailed kite nests were observed in or adjacent to the study area; although, suitable nest trees are present in the study area. The 1971 CNNDD record documented a nest in a willow (*Salix lasiolepis*) or sycamore (*Platanus racemosa*) tree; however, the area has since been developed (CDFW 2021).

Suitable nesting and foraging habitat for white-tailed kite is present in the study area. Several large eucalyptus trees that provide suitable nesting habitat are present along Mowry Avenue in the northern portion of the site adjacent to the ruderal/disturbed habitat. Open areas in the ruderal/disturbed habitat in and adjacent to the study area provide suitable foraging habitat for white-tailed kite. Due to the presence of suitable habitat on and adjacent to the site, white-tailed kite has the potential to occupy the site prior to construction.



Draft Environmental Impact Report Environmental Impact Analysis

Salt Marsh Harvest Mouse (Reithrodontomys raviventris)

The salt marsh harvest mouse is a federally listed and state listed endangered species and is endemic to tidal and brackish marsh habitat of the San Francisco Bay region. Salt marsh harvest mice are primarily found in the salt marshes along northern San Pablo Bay, surrounding Suisun Bay, and along southern San Francisco Bay (USFWS 1984). The salt marsh harvest mouse is critically dependent on dense cover and its preferred habitat is considered to be pickleweed dominated salt marsh wetlands.

No suitable habitat for salt marsh harvest mouse is present on the study area and no surveys have been conducted on the project site for this species. The nearest CNDDB reported occurrence of salt marsh harvest mouse documents tow individuals caught during a trapping survey approximately 1,500 feet south of the study area across the ACFC&WCD Line D. The record dates to 1985 and individuals were trapped on the edge of a salt marsh that abut a disked field (CFDW 2021). This record occurs within an expanse of salt marsh habitat.

The study area does not contain suitable habitat for salt marsh harvest mouse. The ruderal/developed habitat in the main portion of the study area is located over 2,000 feet north of suitable marsh habitat that could support this species and is separated from this habitat by the ACFC&WCD Line D which is a barrier to dispersal for salt marsh harvest mouse. The southern tip of the site that contains the auto wrecking yard and stormwater detention basins is located adjacent to salt marsh habitat that is suitable for use by salt marsh harvest mouse. However, the auto wrecking yard and stormwater detention basins themselves are developed and in active use and as such do not provide suitable habitat for salt marsh harvest mouse.

Protected Trees

The proposed project would remove or otherwise affect up to 45 existing trees protected by the City's tree preservation ordinance (Municipal Code Chapter 8.16, Preservation of Trees on Private Property). Most of the trees, 33 of 45, were non-native blue gum (*Eucalyptus globulus*). The 45 protected trees identified in the study area consisted of one native species, Fremont cottonwood (*Populus fremontii*). There were 44 non-native trees, with 11 rated in good condition. There were 26 of the 45 non-native trees with a diameter at breast height (DBH) greater than 18 inches. Additional information is provided in the Arborist Report which is as an Appendix in the Biological Resources Report (Appendix C).

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's impacts to biological resources are significant. Would the proposed project:

- Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special-status species in local or regional plans, policies, or regulations, or by the CDFW or USFWS?
- Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, and regulations, or by the California Department of Fish or USFWS?



Draft Environmental Impact Report Environmental Impact Analysis

- Have a substantial adverse effect on state or federally protected wetlands (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means?
- Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites?
- Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance?
- Conflict with the provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan?

Project Impact Analysis and Mitigation Measures

Candidate, Sensitive, or Special-Status Species

Impact BIO-1	The proposed project could have a substantial adverse effect, either directly or
	through habitat modifications on any species in local or regional plans, policies,
	or regulations, or by the California Department of Fish and Wildlife or U.S. Fish
	and Wildlife Service.

Impact Analysis

Following the completion of botanical surveys, the potential for these species to occur within the study area was assessed based on the habitats present within and adjacent to the study area, the proximity of known species occurrences, and knowledge of the species' range. It was determined that no special-status plant species are expected to occur within the study area due to lack of suitable habitat.

The following special-status wildlife species have moderate or high potential to occur within the study area: burrowing owl, northern harrier, white-tailed kite, pallid bat, and Townsend's big-eared bat. The other special-status species that were included in the database searches are not likely to occur or presumed absent within the study area due to the lack of suitable habitat. Migratory and resident nesting birds protected by MBTA and CDFW also have potential to occur in the study area. Direct impacts to sensitive bird species including nesting native resident and migratory birds, should they occur, include grading and ground-disturbing activities, increased noise levels from heavy equipment, and increased human presence. Construction during the breeding season could result in the displacement of breeding birds and the abandonment of active nests. If project construction occurs during the avian nesting season (generally considered to be between February 15 and September 15, with some raptors species nesting as early as January), impacts to protected migratory and resident nesting birds could occur. Nesting birds are expected to primarily occur adjacent to the study area but may forage or nest within the study area. Indirect impacts include human disturbance, fugitive dust, the spread of noxious non-native weed species, including the disruption of future breeding and foraging.

The proposed project could result in significant impacts to burrowing owl, northern harrier, white-tailed kite, nesting birds, and nesting colonies of pallid bats and Townsend's big eared bats. The implementation of the proposed project also has the potential to introduce or result in the spread of invasive species from the study area into adjacent sensitive habitats and impact special-status species.



Draft Environmental Impact Report Environmental Impact Analysis

Because special-status wildlife species may occur within the study area, there is potential for a substantial adverse effect on species as regulated by CDFW and/or the USFWS.

If construction and operation of the proposed project resulted in an impact to special-status species, these impacts would be considered significant. In accordance with General Plan Policy LU-7.8, all construction impacts to wildlife, wetlands, and aquatic habitats must be avoided and mitigated. In order to mitigate these potentially significant impacts, Mitigation Measures BIO-1 through BIO-5 would be implemented. These include site-wide BMPs (e.g., restriction on open trenches and guidelines for refueling near drainage features) (MM BIO-1), environmental awareness training to educate proposed project personnel regarding on-site plants and wildlife (MM BIO-2), pre-construction nesting bird surveys and avoidance measures for active nests (MM BIO-3), and pre-construction burrowing owl surveys and relocation of wildlife found within proposed project impact areas during pre-construction surveys (MM BIO-4 and BIO-5). These measures would ensure that potential impacts to special-status plant and wildlife species are reduced to a less than significant level during the construction phase.

Additionally, as the proposed project could also result in significant impacts to nesting colonies of pallid bats and Townsend's big eared bats, both identified as CDFW SSC, Mitigation Measure BIO-6 shall be implemented to minimize impacts to sensitive bat species. In addition to Mitigation Measures BIO-1 through BIO-6 identified to reduce impacts to wildlife species to a less than significant level, the proposed project would be required to implement Mitigation Measures BIO-7 and BIO-8. Mitigation Measure BIO-7 requires the preparation and implementation of an Invasive Species Management Plan to reduce the presence and spread of non-native, invasive species for the area to be developed. Mitigation Measure BIO-7 would comply with General Plan Policy LU-7.4 which requires management plans to control the population of invasive species. Mitigation Measure BIO-8 requires the preparation and implementation of a post-construction predator management plan and program to educate the project residents regarding measures to minimize the potential for subsidizing predator species and to minimize the potential effects of pets on sensitive species.

As the study area does not provide suitable habitat for salt marsh harvest mouse, no potential for direct impacts to this species is identified. However, the salt marsh harvest mouse does have potential to occur off-site in the area and therefore, may result in indirect impacts. As the project site is located within 300 feet of potentially suitable salt marsh harvest mouse habitat, a mouse-proof fence has been installed to prevent these species from entering the work area as a voluntary, precautionary, pre-construction protective measure to avoid potentially affecting salt marsh harvest mouse prior to any development activity. The current Pick-n-Pull owner has installed the mouse-proof fencing along the southern borders of the study area that are near suitable salt marsh habitat so that these species cannot enter and be harmed on a site that is zoned for development. As a result of site management post-implementation of the proposed project, potential indirect impacts to salt marsh habitat off-site are expected to be lower with the proposed project than with existing conditions because salt marsh harvest mouse have been excluded from the site through installation of a mouse-proof fence to prevent any potential for this endangered species to enter the site and be harmed, water quality from stormwater runoff into the adjacent habitats would be improved, and general avoidance measures to reduce predator and invasive species from entering nearby sensitive habitats will be incorporated into the proposed development. In addition, the development of the proposed project would not result in a significant change in the stormwater input to the adjacent salt marsh habitat and would not result in a significant alteration of hydrology in adjacent salt marsh habitats.



Because special-status wildlife species may occur within the study area, there is potential for a substantial adverse effect on species as regulated by CDFW and/or the USFWS. Implementation of Mitigation Measures BIO-1 through BIO-8 would ensure that impacts to special-status wildlife species would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM BIO-1:** Standard Construction Best Management Practices. The applicant shall require implementation of standard construction BMPs outlined, but not limited to, throughout construction.
 - The project limit shall be delineated and/or fenced prior to the start of construction.
 - Appropriate erosion control measures shall be used (e.g., hay bales, filter fences, vegetative buffer strips or other accepted equivalents) to reduce siltation and contaminated runoff from leaving the proposed project site. The integrity and effectiveness of the BMPs shall be inspected daily by the resident engineer. Corrective actions and repairs shall be carried out immediately.
 - Construction by-products and pollutants such as petroleum products, chemicals, or other deleterious materials shall not be allowed into off-site wetlands or marsh habitats.
 - A plan for the emergency clean-up of any spills of fuel or other materials shall be available when construction equipment is in use.
 - Construction vehicles and equipment shall be maintained to prevent contamination of soil or water from external grease and oil or from leaking hydraulic fluid, fuel, oil, and grease.
 - Leaking vehicles and equipment shall be removed from the site.
 - Building materials storage areas containing hazardous or potentially toxic materials such as herbicides and petroleum products shall have an impermeable membrane between the ground and the hazardous material and shall be bermed to prevent the discharge of pollutants to ground water and runoff water.
 - Equipment shall be re-fueled and serviced at designated construction staging areas, a minimum of 100 feet from any aquatic resource.
 - All construction material and fill shall be stored and contained in a designated area that is located away from aquatic habitats to prevent transport of materials into adjacent water bodies.
 - The preferred distance is 100 feet from any wetlands or marsh habitats.

- A silt fence shall be installed to collect any discharge, and adequate materials shall be available for spill clean-up and during storm events.
- No litter, debris, or side cast shall be dumped or permitted to enter wetlands or marsh habitats.
- During project activities, all trash that may attract predators shall be properly contained, removed from the work site, and disposed of regularly. Following construction, all trash and construction debris shall be removed from work areas.
- **MM BIO-2: Environmental Awareness.** Prior to initial ground disturbance, all project personnel shall attend an environmental awareness and compliance training. The training program shall present applicable environmental regulations and permit conditions. The training program shall include applicable measures established for the proposed project to minimize impacts to avoid sensitive resources, habitats, and species. Subsequent training events shall be scheduled to support the training of new personnel. Dated sign-in sheets for attendees at these meetings shall be maintained and submitted to the City prior to the issuance of grading permit.
- **MM BIO-3: Pre-Construction Nesting Bird Survey.** If project activities such as vegetation removal activities commence during the avian breeding season (February 1 through August 31), a qualified biologist shall conduct a pre-construction nesting bird survey no more than 14 days prior to initiation of project activities and again within 48 hours prior to initiation of project activities. The survey area should include suitable raptor nesting habitat within 500 feet of the project boundary (inaccessible areas outside of the project parcels can be surveyed from the parcel or from public roads using binoculars or spotting scopes). Preconstruction surveys are not required in areas where project activities have been continuous since prior to February 1, as determined by a qualified biologist. Areas that have been inactive for more than 14 days during the avian breeding season must be resurveyed prior to resumption of project activities. If no active nests are identified, no further mitigation is required. If active nests are identified, the following measure shall be implemented:
 - A suitable buffer (e.g., 300 to 500 feet for northern harrier and white-tailed kite; 200 to 300 feet for common raptors; 50 to 100 feet for non-raptors) shall be established by a qualified biologist around active nests and no construction activities within the buffer shall be allowed until a qualified biologist has determined that the nest is no longer active (i.e., the nestlings have fledged and are no longer reliant on the nest, or the nest has failed). Encroachment into the buffer shall be monitored by a qualified biologist. Any encroachment into the buffer shall be monitored by a qualified biologist to determine whether nesting birds are being impacted.
- **MM BIO-4: Burrowing Owl Pre-Construction Surveys.** Pre-construction surveys for burrowing owls shall be completed no more than 15 days prior to the start of construction in areas planned for fill placement and construction areas in general conformance with the California Burrowing Owl Consortium's and the CDFW Staff Report (2012) protocols.



Because owls are known to occur in the vicinity of the project site, these surveys shall be completed no more than 15 days prior (rather than 30 days prior, as per the Consortium's protocol) to the start of importing fill and construction to minimize the probability of immigration of owls between the time surveys are completed and the initiation of grading. If the initial disturbance is followed by periods of inactivity exceeding 15 days, or if the development is phased spatially and/or temporally such that an area in which construction activities are to commence has not been disturbed by construction activities within the prior 15-day period, a new burrowing owl pre-construction survey shall be completed prior to the start of disturbance. If burrowing owls are detected on or within 250 feet of the site, the mitigation measures below shall be implemented.

- If burrowing owl is located during the non-breeding season (generally 1 September to 31 January), a 150-foot buffer zone shall be maintained around the occupied burrow(s) if practicable. If such a buffer is not practicable, then a buffer adequate to avoid injury or mortality of owls shall be maintained, or the birds shall be evicted as described below. During the breeding season (generally 1 February to 31 August), a 250-foot buffer, within which no new activity shall be permissible, shall be maintained between project activities and occupied burrows. Owls on site after 1 February shall be assumed to be nesting unless direct observations indicate otherwise. This protected buffer area shall remain in effect until 31 August, or based upon monitoring evidence, until the young owls are foraging independently, or the nest is no longer active. Owls that are not nesting can be evicted using the methods below during the period from 1 February to 31 August.
- If construction would directly impact occupied burrows, eviction of owls may occur outside the nesting season (or during the nesting season if the owls are determined to be not nesting) to prevent injury or mortality of individual owls. No burrowing owls shall be evicted from burrows during the nesting season (1 February through 31 August) unless evidence indicates that nesting is not actively occurring (e.g., because the owls have not yet begun nesting early in the season, or because young have already fledged late in the season). Relocation of owls during the nonbreeding season shall be completed by a qualified biologist using one-way doors, which shall be installed in all burrows within the impact area and left in place for at least two nights. These one-way doors shall then be removed, and the burrows backfilled immediately prior to the initiation of grading.
- If resident burrowing owl(s) are found in the proposed project site during preconstruction surveys and eviction is necessary to facilitate construction, Mitigation Measure BIO-6 shall be implemented. These measures do not apply to short-term use of the site by a burrowing owl for foraging, as a stop-over during migration, or temporary use of the site by wintering birds or dispersing juveniles.
- MM BIO-5: Burrowing Owl Mitigation. To reduce impacts of the project on the local (South Bay) burrowing owl population, habitat shall be preserved and managed for burrowing owls off-site if eviction of resident owls is required. California burrowing owl mitigation guidelines recommend that 6.5 acres of foraging habitat be preserved and managed per occupied



Environmental Impact Analysis

burrowing owl burrow (whether by a pair or singly) in mitigation sites. Therefore, mitigation shall be required for each pair or single resident burrowing owl that is evicted, up to a maximum of 13 acres. Mitigation may take the form of off-site habitat preservation and management (in which case all the monitoring and habitat requirements in the following paragraphs would apply) or the purchase of credits in an off-site mitigation bank. Because the nearest burrowing owl mitigation banks are currently located outside of the South Bay, this mitigation may occur outside the region.

If off-site habitat is to be preserved, a mitigation and monitoring plan shall be prepared by a qualified biologist and submitted to the City and the CDFW for review and approval. This plan shall detail the following:

- Areas to be preserved for owls
- Methods for managing habitat for owls and their prey
- Plans to enhance burrow availability within the mitigation site (potentially including the provision of artificial burrows, although long-term management for ground squirrels would be important as well)
- A monitoring program and adaptive management program.

At least 50 percent of the mitigation area must consist of upland habitat suitable for use by burrowing mammals, and no wetlands supporting tall vegetation shall be included within the mitigation site. The mitigation area must be contiguous with habitat that is permanently preserved as open space to avoid having the site surrounded by development in the future. The mitigation area shall be protected in perpetuity through a conservation easement, deed restriction, conveyance to a qualified land trust or the Don Edwards National Wildlife Refuge, or through equivalent means.

Assuming burrowing owl habitat mitigation would occur off-site, some on-site enhancements shall also be made to reduce impacts of the project on the local (South Bay) burrowing owl population. Such enhancements shall include the provision of two artificial burrow complexes on the sides of the adjacent levees (if allowed by levee managers) and management of at least portions of levee side slopes around these burrow complexes to provide suitable conditions for burrowing owls and ground squirrels (e.g., periodic mowing to maintain short vegetation). Given the extent of natural habitat with short vegetation, and the continued presence of seasonal wetlands near the proposed project site, providing and maintaining burrows for use by owls is expected to maintain some burrowing owl presence near the proposed project site even if most or all the owl habitat mitigation occurs off-site.

Signage shall be placed in appropriate locations to prohibit individuals from entering areas where the artificial burrow complexes would be located. Signage shall be placed along the levee slopes to instruct recreational users of these levees against leaving the levee tops to protect sensitive species such as the burrowing owl.

Draft Environmental Impact Report Environmental Impact Analysis

- **MM BIO-6:** Special-Status Bat Species. A survey for roosting bats shall be completed prior to the removal of any building or tree with potential for day-roosting by bats, or prior to the initiation of any construction activities within 250 feet of such potential roost sites. The survey shall be completed by a qualified biologist. If suitable roost sites are found but a visual survey is not adequate to determine presence or absence of bats (which would be particularly likely in the case of potential roost trees), acoustical equipment could be used to determine occupancy. This survey shall be completed prior to the beginning of the breeding season (i.e., prior to 1 March) in the year in which construction or demolition in each area is scheduled to occur so that adequate measures can be implemented, if necessary, to evict the bats during the non-breeding season. The survey results shall be provided to the Community Development Director for review and approval prior to the start any construction related activities.
 - Because the initial surveys would be completed prior to the breeding season, several months may pass between that survey and the initiation of construction or demolition in each area. Therefore, a second pre-demolition/pre-construction survey for roosting bats, following the methods described above, shall be completed within 15 days prior to the commencement of these activities in each area to determine whether bats have occupied a roost in or near the development impact areas. This survey shall be facilitated considerably by information (e.g., on potential roost trees) gathered during the previous survey. If bats are found to be roosting, additional mitigation as follows must be implemented.
 - If a maternity roost of any special-status bat species is found, the bat biologist shall determine the extent of a construction-free buffer around the active roost that would be maintained. This buffer would be maintained from 1 March until the young are flying, typically after 31 August. If a roost of any kind is found in an area (e.g., a building or tree) that would not be disturbed by construction, or that can be avoided, the roost structure shall not be impacted.
 - If a day roost is found in a building, or in a tree that is to be completely removed or • replaced, individual bats shall be safely evicted under the direction of a qualified biologist. Eviction of bats shall occur at dusk, so that bats would have less potential for predation compared to daytime roost abandonment. Eviction shall occur between 1 September and 31 March, outside the maternity season, but shall not occur during long periods of inclement or cold weather (as determined by the bat biologist) when prey is not available, or bats are in torpor. If a day roost is found within a building, eviction shall occur by opening the roosting area to allow air flow through the cavity. Demolition may then follow no sooner than the following day (i.e., there must be no less than one night between initial disturbance for air flow and the demolition). This action shall allow bats to leave during dark hours, thus increasing their chance of finding new roosts with a minimum of potential predation during daylight. If feasible, one-way doors shall also be used to evict bats from tree roosts. If use of a one-way door is not feasible, or the exact location of the roost entrance in a tree is not known, the trees with roosts that need to be removed shall first be disturbed by removal of some of the trees' limbs not containing the bats. Such disturbance shall occur at dusk

to allow bats to escape during the dark hours. These trees would then be removed the following day. These activities shall be performed under the supervision of the qualified biologist.

- If a day roost for pallid bats or another rare bat would be impacted, an alternative bat roost structure shall be provided. The design and placement of this structure would be determined by a qualified biologist based on the location of the original roost and which species is located. This bat structure shall be erected at least one month (and preferably a year or more) prior to removal of the original roost structure. This structure shall be checked during the breeding season for up the three years following completion of the development, or until it is found to be occupied by bats, to provide information for future development projects regarding the effectiveness of such structures in minimizing impacts to bats.
- **MM BIO-7: Reduce the Spread of Invasive Species.** Prior to issuance of any building or grading permits, the project shall develop and implement an Invasive Species Management Plan to reduce the presence and spread of non-native, invasive plant species for the area to be developed. The Plan shall be developed prior to importing any fill material required to elevate building sites and prior to grading any areas on the Specific Plan site. The overarching goal of this mitigation is to halt the further expansion of existing invasive species and introduction of new invasive species into sensitive habitats on site. The Invasive Species Management Plan shall include, but not be limited to, the following, summarized below:
 - Prior to construction, map populations of invasive species within all areas proposed to be graded; quantify the extent and location of invasive populations in sensitive habitats.
 - Areas identified to have weed infestations shall be treated prior to ground disturbance according to weed control methods detailed below and Best Management Practices within all upland areas to be graded, after review and approval of methodologies by the City of Newark.
 - Weed control treatments shall include all legally permitted herbicide, manual, and mechanical methods approved for application. The timing of the weed control treatment shall be determined for each plant species with the goal of controlling populations before they start producing seeds and/or encroach into adjacent areas from rhizomatous shoots. Consultation with a City of Newark approved wildlife biologist or plant ecologist shall be required prior to weed control treatments in sensitive habitats with the intent of avoiding any adverse impacts to special-status species in the area.
 - Surveying and monitoring for weed infestations shall occur annually while grading operations are occurring. Treatment of all identified weed populations shall occur at a minimum of once annually.

- During project construction, all seeds and straw materials used on site shall be weed free rice straw, and all gravel and fill material shall be certified weed free.
- During project construction, vehicles and all equipment shall be washed before and after entering the project area.
- **MM BIO-8: Post-Construction Predator Management Plan and Program.** This program shall focus on education of occupants of the new residential areas regarding measures to minimize the potential for subsidizing predator species and to minimize the potential effects of pets on sensitive species and enforcement of the program's measures, and restrictions on certain activities that could increase predation of sensitive species. A plan shall be developed. The details of the program would be developed during construction but shall include, at a minimum, the following:
 - Minimize disturbance from the development by educating the public about the importance of preserving the ecological integrity of the adjacent natural areas instructing recreational users to stay on the levee tops out of sensitive habitats and keep dogs on leashes.
 - To prevent the spread of invasive non-native plants into the nearby sensitive habitats, plants contained on the California Exotic Plant Pest Council List of Invasive Plants shall be barred from use within the landscaping of the Mowry Village development area. A list of plants suitable for landscape use shall be provided to property buyers.
 - Feeding pets outdoors shall be prohibited so that pet food does not attract or subsidize the diets of nuisance species.
 - Pets shall be prohibited from ranging freely (off-leash dogs shall be prohibited in offsite wetland areas and no free-roaming outdoor cats shall be permitted), to prevent their entry into sensitive species habitat.
 - All food waste shall be contained so that it does not attract or subsidize the diets of predators.

Any neighborhood association established for new residential areas will be responsible for disseminating this information, and the neighborhood association and City will be responsible for enforcing the program.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.


Draft Environmental Impact Report Environmental Impact Analysis

Riparian Habitat or Natural Communities

Impact BIO-2	The proposed project would not have a substantial adverse effect on any riparian
	habitat or other sensitive natural community identified in local or regional plans,
	policies, and regulations, or by the California Department of Fish and Wildlife or
	U.S. Fish and Wildlife Service.

Impact Analysis

As identified in the Biological Resources Technical Report prepared for the proposed project, the study area does not contain any riparian habitat or other sensitive natural communities (Helix 2022). Results of the aquatic resources delineation and analysis showed that the study area contains two artificial stormwater basins, but these features do not contain any riparian habitat (Helix 2022). Therefore, the proposed project would have no impact on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, and regulations or by the CDFW or USFWS.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Protected Wetlands

Impact BIO-3 The proposed project would not have a substantial adverse effect on state or federally protected wetlands (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means.

Impact Analysis

The aquatic resources jurisdictional delineation surveys conducted in November 2021, March 2022, and September 2022 identified four potentially jurisdictional aquatic features; however, did not identify the presence of waters/wetlands subject to USACE, RWQCB or CDFW jurisdiction within the study area. The project site includes two constructed stormwater basins with wetland features identified as Storm Water Detention Basin 1 and Storm Water Detention Basin 2, in the southeastern portion of the project site (Figure 3-3). Additionally, the Aquatic Resources Delineation Report identifies two off-site potentially jurisdictional resources identified as Constructed Storm Drain and ACFC&WCD Line D, located approximately 1,500 feet east of the project site (Figure 3-3). Since the on-site basins were artificially constructed, they are not regulated by USACE, RWQCB, and CDFW.

The USACE provided a jurisdiction determination of the extent of navigable waters of the United States and waters of the United States in a letter dated January 25, 2023. The jurisdictional determination provided by the USACE identified that the Constructed Storm Drain, Storm Water Detention Basin 1, and Storm Water Detention Basin 2 are not jurisdictional resources. However, the jurisdictional determination identified the ACFC&WCD Line D channel as a jurisdictional aquatic resource, specifically Other Waters with In-stream Wetlands. On-site construction activities of the proposed project would have no impact on



Draft Environmental Impact Report Environmental Impact Analysis

this jurisdictional resource as it is located off-site. However, extension of the potable and non-potable water mains from the southwest corner of the Sanctuary West Development located to the east of the project site would occur within the vicinity of ACFC&WCD Line D. Impacts to ACFC&WCD Line D from off-site construction would be avoided through the use of jack-and-bore method to construct the extension. The extension of the potable and non-potable water mains would be jack-and-bored, approximately 250 linear feet, under the ACFC&WCD Line D channel and the UPRR ROW which would ensure that impacts to ACFC&WCD Line D are avoided. On-site and off-site construction activities would avoid all nearby jurisdictional aquatic features and would not result in substantial adverse effects on state or federally protected wetlands. Therefore, the potential impacts to wetlands are considered less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Migratory Wildlife Corridors

Impact BIO-4 The proposed project would not interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites.

Impact Analysis

Fish and wildlife corridors are segments of undisturbed land cover that connect larger, contiguous habitats. These corridors provide movement linkages for fish and wildlife. Corridors function as avenues along which wide-ranging animals can travel, genetic interchange can occur, and wildlife populations can move in response to environmental changes and natural disasters. Some fish and wildlife corridors include the Pacific flyway, linear riparian areas, streams, rivers, or other drainage features, and urban canyons dominated by native vegetation. Fish and wildlife corridors are recognized by federal agencies such as the USFWS and the State as important areas worthy of conservation. The study area does not act as a corridor for species dispersal or provide migration habitat connectivity to adjacent habitat and is not part of any defined essential connectivity areas as identified in the California Essential Habitat Connectivity Project (Spencer et al. 2010); therefore, the project would have no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Local Policies or Ordinances

Impact BIO-5 The proposed project could conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance.

Impact Analysis

The proposed project would remove or otherwise affect up to 45 existing trees protected by the City's tree preservation ordinance (Municipal Code Chapter 8.16, Preservation of Trees on Private Property). Most of the trees, 33 of 45, were non-native blue gum. The 45 protected trees identified in the study area consisted of one native, Fremont cottonwood, 44 non-native, 11 rated in good condition, and 26 with a DBH over 18 inches (refer to the Arborist Report in Appendix C). The project proposes to remove 13 existing trees within the project site; however, the project is proposing to replace the existing removed trees with 213 trees to be planted and provide landscaping throughout the site.

Under Chapter 8.16 of the Newark Municipal Code, it is unlawful to cut down, destroy, remove, or move any tree within the city limits on any parcel of land except a developed residential parcel 10,000 square feet or less in size unless a permit to do so has been obtained from the public works director. A tree is any live woody plant with at least one well defined perennial stem at least six inches in diameter measured four feet above ground level.

Because City protected trees may be impacted during project implementation, there is potential for a substantial adverse effect on the trees as regulated by the City. Impacts to City protected trees would be less than significant after the implementation of Mitigation Measure BIO-9.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM BIO-9: Tree Preservation Plan.** Prior to the issuance of any construction-phase permit, a construction-phase Tree Preservation Plan shall be prepared by a certified arborist to the satisfaction of the City's Community Development Director for all areas with trees. The construction-phase Tree Preservation Plan shall include the following tree protection measures which are based on guidelines established by the International Society of Arboriculture:
 - Establish Tree Protection Zones
 - Protect tree root systems
 - Install wood bark mulch
 - Trees removed by the project shall be replaced at a 3:1 ratio unless the City's Community Development Director determines that a higher ratio is required.
 - Trees greater than 18 inches in diameter at breast height (DBH) shall not be removed unless a Tree Removal Permit, or equivalent, has first been approved for the removal of such trees.



- The species and exact number of trees to be planted on the site during the construction phase shall be determined in consultation with the City and to the satisfaction of the Community Development Director.
- In the event the developed portion of the development site does not have sufficient area to accommodate the required tree mitigation, one or more of the following measures shall be implemented at the development permit stage:
- An alternative site(s) shall be identified for additional tree planting. Alternative sites
 may include local parks or schools, or installation of trees on adjacent properties for
 screening purposes, to the satisfaction of the City's Community Development
 Director. The size of a 15-gallon replacement tree can be increased to 24-inch box
 and counted as two replacement trees.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Habitat Conservation Plan

Impact BIO-6The proposed project would not conflict with the provisions of an adopted
habitat conservation plan, natural community conservation plan, or other
approved local, regional, or State habitat conservation plan.

Impact Analysis

The City is not within the jurisdiction of any adopted Habitat Conservation Plan (HCP) (City of Newark 2013b). The project would not conflict with any adopted HCPs or other approved local, regional, or state habitat conservation plan. Therefore, there would be no impacts.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.



3.5 CULTURAL RESOURCES

This section describes impacts on cultural resources that would result from implementation of the proposed project. Included is a summary of applicable policies and regulations related to cultural resources and review of existing conditions. It also describes impacts on cultural resources that would result from implementation of the proposed project, based on the Cultural Resources Assessment prepared by Helix Environmental Planning in December 2021 (Appendix D).

3.5.1 Environmental Setting

Cultural Setting

The following cultural resources discussion is modified from the Cultural Resources Assessment (Appendix D) unless otherwise referenced.

Prehistoric and Ethnographic Background

The San Francisco Bay Area supported a dense population of hunter-gatherers over thousands of years, leaving a rich and varied archaeological record. The Bay Area was a place of incredible language diversity, with seven languages spoken at the time of Spanish settlement in 1776. The diverse ecosystem of the Bay and surrounding lands supported an average of three to five persons per square mile but reached eleven persons per square mile in the North Bay. At the time of Spanish contact, the Native Americans living in the Bay Area were organized into local tribelets that defended fixed territories under independent leaders. Typically, individual Bay Area tribelets included 200 to 400 people distributed among three to five semi-permanent villages, within territories measuring approximately 10 to 12 miles in diameter.

At the time of European contact, the general Newark area was occupied by various tribelets that were part of the Ohlone (previously Costanoan) tribe of California Native Americans. These groups lived in approximately 50 separate and politically autonomous tribelet areas, each with one or more permanent villages, between the North San Francisco Bay and the lower Salinas River (Helix 2021). Refer to Section 3.18, Tribal Cultural Resources, for additional information regarding the ethnographic setting.

Historic Background City of Newark

European settlement of the area began with the founding of Mission San Jose in 1797. The City remained in control of Mission San Jose until approximately 1836 when the Mission was secularized and came under control of the Mexican government. In 1844, a large land area known as the Rancho Poterero de Los Cerritos grant, made by the Mexican governor gave what is now the entire City of Newark, as well as Coyote Hills and portions of Union City and Fremont to Augustine Alviso and Thomas Pacheco. When California became part of the U.S. in 1848, American settlers began moving to the Rancho potrero de los Cerritos area in great numbers. Among the first to settle in the Newark area was Origin Mowry, who in 1850 established Mowry's Landing which provided the main source of commerce to the area.



In 1878, the present day Union Pacific Railroad came into service in the City and the historic Old Town Newark train station was established. Railroad associated businesses were established in the vicinity and other industries within the area at the time included commercial salt ponds in the southwestern portion of the City. By 1880, the City had a population of 200 people. In 1909, construction of the first Bay bridge was completed which connected freight trains from Newark to Redwood City and ultimately all the way to San Francisco. By 1900, significant portions of marshland to the western edge of Newark were converted to salt evaporation ponds. From the 1950s until the 1990s, the City had been host to a variety of industrial uses, ranging from brick making, to chemical blending, to semi-truck assembly (City of Newark 2013b).

Archaeological Resources in the Project Area Records Search and Literature Review

A records search addressing the project site and 0.5 mile radius beyond the Area of Potential Effects (APE) boundary (together referred to as the study area) was conducted on April 17, 2018 at the Northwest Information Center (NWIC) by Helix Environmental Planning. The APE for the proposed project is defined as the geographic area where project activities may directly or indirectly cause changes in the character or use of historic properties of prehistoric or historic age, if any such properties exist. The APE for the proposed project measures 35.3 acres and corresponds to the project's maximum area of ground disturbance. The APE for the proposed project includes the project site located at the existing Pick-n-Pull site, the off-site expansion and modifications to Mowry Avenue, and the off-site infrastructure improvements (Figure 3-4).

Sources of information included previous surveys and cultural resources files; the National Register of Historic Places (NRHP); and California Register of Historic Resources (CRHR); the Office of Historic Preservation (OHP) Archaeological Determinations of Eligibility; the OHP Directory of Properties in the Historic Property Data File; and historical topographic maps and aerial photographs.

The records search identified 18 previous cultural resources studies that have been conducted within the study area. Of the 18 previous cultural resources studies within the study area, only two directly examined the current APE. Results from the records search indicate that no resources have been previously recorded within the APE; however, seven resources have been recorded within the 0.5 mile search radius. The seven resources include a prehistoric site with burials and habitation debris recorded in 1959; two prehistoric shell mounds recorded in the 1930s; a prehistoric site with burials, habitation debris and hearths/pit recorded in 1999; a historical era reburial of prehistoric remains recorded in 2000; an area of prehistoric habitation debris recorded in 2011; and a historic era trash scatter recorded in 2014. These resources are summarized in Table 3.5-1 below.





3-4 Title

Cultural Resources Area of Potential Effects

This page intentionally left blank.

Primary	Trinomial	Description	Year	Recorder	Affiliation
P-01-000079	CA-ALA-59	Prehistoric shell mounds with burials and habitation debris	1959	J.T. Davis	None
P-01-000112	CA-ALA-336	Possible prehistoric shell mound	ca. 1909	Nels Nelson	University of California, Berkeley
P-01-000113	CA-ALA-337	Possible prehistoric shell mound	ca. 1935	Nels Nelson	University if California, Berkeley
P-01-002267	CA-ALA-620	Prehistoric burial site with habitation debris	1999	Alan Levanthal, Rosemary Cambra	Ohlone Families Consulting Services
P-01-010491	None	Prehistoric reburial site	2002	Jason Claiborne	Archeo-Tec
P-01-011353	CA-ALA-641	Prehistoric habitation debris	2011	Randy Wiberg	Holman and Associated
P-01-011611	None	Historic era refuse	2014	Eric Strother, Kruger Frank	Garcia and Associated

Table 3.5-1: Previously Documented Resources within the Study Area

Notes:

The previously documented resources identified in this table are all located outside the APE but within the 0.5 mile buffer area

A search of the Historic Properties Database File for Alameda County was negative for historic properties within the study area and within 0.5 mile of the study area boundary.

Field Survey

On June 13, 2019, Helix Environmental Planning conducted an intensive pedestrian survey to characterize any prehistoric or historic era archaeological resources located within the APE. On November 16, 2021, Helix Environmental Planning conducted a second intensive pedestrian survey of the APE to examine areas recently added to the APE and to re-examine portions of the APE which were covered in dense grasses during the initial survey. Both surveys consisted of a pedestrian walk-over of the undeveloped portions of the APE in parallel transects spaces at 10 meter intervals. During the surveys, the ground surface was examined for the presence of historic era artifacts (e.g., metal, glass, ceramics), prehistoric artifacts (e.g., flaked stone tools, tool-making debris), and other features that might represent human activity that took place more than 50 years ago.

The 2019 and 2021 surveys did not cover the Pick-n-Pull property itself as this portion of the APE has been fully graded and capped with a layer of imported fill soil. Additionally, the two detention basins located on the eastern boundary of the Pick-n-Pull which mark the southeastern boundary of the project area were surveyed. Finally, the remainder of the APE which includes areas marked for underground excavations to the northwest, northeast, southeast, and southwest of the Pick-n-Pull property and located proposed to be modified on Mowry Avenue (adjacent west of the Pick-n-Pull) were also surveyed. No prehistoric or historic resources were found during the 2019 or 2021 filed surveys.



Historical Resources in the Project Area

The project site is not located within an identified historic district and there are no historical resources located on-site. A search of the Historic Properties Database File for Alameda County conduced for the Cultural Resources Assessment was negative for historic properties within the study area and within 0.5 mile of the study area boundary.

3.5.2 Regulatory Setting

Federal

National Historic Preservation Act

The National Historic Preservation Act, as amended, established the NRHP, which contains an inventory of the nation's significant prehistoric and historic properties. Under 36 Code of Federal Regulations (CFR) 60, a property is recommended for possible inclusion on the NRHP if it is at least 50 years old, has integrity, and meets one of the following criteria:

- It is associated with significant events in history or broad patterns of events.
- It is associated with significant people in the past.
- It embodies the distinctive characteristics of an architectural type, period, or method of construction; it is the work of a master or possesses high artistic value; or it represents a significant and distinguishable entity whose components may lack individual distinction.
- It has yielded or may yield information important in history or prehistory.

Certain types of properties usually are excluded from consideration for listing on the NRHP, but they can be considered if they meet special requirements in addition to meeting the criteria listed above. Such properties include religious sites, relocated properties, graves and cemeteries, reconstructed properties, commemorative properties, and properties that have achieved significance within the past 50 years.

State

California Environmental Quality Act

Pursuant to CEQA, a historical resource is a resource listed in, or eligible for listing in, the CRHR. In addition, resources included in local register of historic resources, or identified as significant in a local survey conducted in accordance with state guidelines, are also considered historic resources under CEQA, unless a preponderance of the facts demonstrates otherwise. According to CEQA, the fact that a resource is not listed in, or determined eligible for listing in, the CRHR or is not included in a local register or survey shall not preclude a Lead Agency, as defined by CEQA, from determining that the resource may be a historic resource as defined in PRC Section 5024.1(c).

CEQA applies to archaeological resources when (1) the archaeological resource satisfies the definition of a historic resource, or (2) the archaeological resource satisfies the definition of a "unique archaeological



Draft Environmental Impact Report Environmental Impact Analysis

resource." A unique archaeological resource is an archaeological artifact, object, or site that has a high probability of meeting any of the following criteria (PRC Section 21083.2(g)):

- 1. The archaeological resource contains information needed to answer important scientific research questions, and there is a demonstrable public interest in that information.
- 2. The archaeological resource has a special and particular quality such as being the oldest of its type or the best available example of its type.
- 3. The archaeological resource is directly associated with a scientifically-recognized important prehistoric or historic event or person.

California Register of Historic Resources

The CRHR is "an authoritative guide in California to be used by state and local agencies, private groups, and citizens to identify the state's historical resources and to indicate what properties are to be protected, to the extent prudent and feasible, from substantial adverse change" (PRC Section 5024.1(a)). Certain properties, including those listed in or formally determined eligible for listed in the NRHP and California Historical Landmarks (CHLs) numbered 770 and higher, are automatically included in the CRHR. Other properties recognized under the California Points of Historical Interest program, identified as significant in historic resources surveys, or designated by local landmarks programs may be nominated for inclusion in the CRHR.

Under PRC Section 5024.1(c), a resource, either an individual property or a contributor to a historic district, may be listed in the CRHR if the State Historical Resources Commission determines that it meets one or more of the following NRHP criteria (PRC Section 5024.1(c)):

- 1. It is associated with events that have made a significant contribution to the broad patterns of California's history and cultural heritage.
- 2. It is associated with the lives of persons important in our past.
- 3. It embodies the distinctive characteristics of a type, period, region, or method of construction; represents the work of an important creative individual; or possessed high artistic values.
- 4. It has yielded, or may be likely to yield, information important in history or prehistory.

Resource nominated to the CRHR must retain enough of their historic character of appearance to be recognized as historic resources and to convey the reasons for their significance. It is possible that a resource whose integrity does not satisfy NRHP criteria may still be eligible for listing in the CRHR. A resource that has lost its historic character or appearance may still have sufficient integrity for the CRHR if, under Criterion 4, it maintains the potential to yield significant scientific or historical information or specific data. Resources that have achieved significance within the past 50 years also may be eligible for inclusion in the CRHR, provided that enough time has lapsed to obtain a scholarly perspective on the events or individuals associated with the resource.



Draft Environmental Impact Report Environmental Impact Analysis

California Health and Safety Code and Public Resources Code

Broad provisions for the protection of Native American cultural resources are contained in the HSC, Division 7, Part 2, Chapter 5 (Sections 8010 through 8030). Several provisions of the PRC also govern archaeological finds of human remains and associated objects. Procedures are detailed under PRC Section 5097.98 through 5097.996 for actions to be taken whenever Native American remains are discovered.

Section 7050.5 of the HSC states that any person who knowingly mutilates or disinters, wantonly disturbs, or willfully removes human remains in or from any location other than a dedicated cemetery without authority of law is guilty of a misdemeanor, except as provided in Section 5097.99 of the PRC. Any person removing human remains without authority of law or written permission of the person or persons having the right to control the remains under PRC Section 7100 has committed a public offense that is punishable by imprisonment. PRC Chapter 1.7, Section 5097.5/5097.9 (Stats. 1965, c. 1136, p. 2792), entitled Archaeological and Historical Sites, defines any unauthorized disturbance or removal of remains on public land as a misdemeanor.

Local

City of Newark General Plan

The following lists goals and policies from the City of Newark General Plan pertaining to cultural resources that are applicable to the proposed project.

Goal LU-5: Historic Preservation. Identify, preserve, and maintain historic structures and sites to enhance Newark's sense of place and create living reminders of the city's heritage.

 Policy LU-5.5: Native American Resources. Coordinate with local tribal representatives and the Native American Heritage Commission to ensure the protection of Newark's Native American resources and to follow appropriate mitigation, preservation, and recovery procedures in the event that important resources are identified during development.

3.5.3 Environmental Impacts

This section analyzes the project's potential to result in significant cultural resources impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following impact analysis is based on the Cultural Resources Assessment prepared for the proposed project by Helix Environmental Planning in December 2021.

The Cultural Resources Assessment included a records search at the NWIC, literature review, and archaeological field survey. The records search and cultural resources survey were completed in accordance with the CEQA guidelines by the following actions: 1) identifying all cultural resources within the project site; 2) offering a significance evaluation of the identified cultural resources; 3) assessing

resource vulnerability to effects that could arise from project activities; and 4) offering suggestions designed to protect resource integrity, as warranted.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's cultural resources impacts are significant. Would the proposed project:

- Cause a substantial adverse change in the significance of a historical resource pursuant to Section 15064.5?
- Cause a substantial adverse change in the significance of an archaeological resource pursuant to Section 15064.5?
- Disturb any human remains, including those interred outside of formal cemeteries?

Project Impact Analysis and Mitigation Measures

Historical Resources

Impact CUL-1The proposed project would not cause a substantial adverse change in the
significance of a historical resource as identified in Section 15064.5.

Impact Analysis

No historic resources (eligible or likely eligible under state, federal, or local historic preservation criteria) were identified within or adjacent to the project site that would be impacted by the proposed project. We do not have an exact year of construction for the existing structures on-site; however, aerial photographs from 1963 shows that all structures that were present on-site prior to 1963 appears to have been demolished, and the land unused. Able Auto Wreckers operated an automobile wrecking yard on the project site from the late 1960s until they were acquired by the current owner, Pick-n-Pull, in 1996 and has continued to operate the automobile wrecking yard since that time (Helix 2021). According to CEQA, all buildings constructed over 50 years ago and possess architectural or historical significance may be considered potential historic resources. Historical aerial imagery available online does not show any structures present on the project site in the 1968 aerial imagery; however, the next available aerial imagery from 1979 shows the project site developed with structures that currently exist on-site (Netronline 2023). Therefore, some existing structures on the project site may be over 50 years old. However, existing structures on-site are typical metal structures found at industrial sites and do not possess any architectural or historical signification. The existing structures on-site would not eligible for listing as a historic resource under CEQA. Thus, the proposed project is not anticipated to have an impact on any known or potential historical resources. Therefore, there would be no impacts to historical resources.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures No mitigation is necessary.



Level of Significance After Mitigation

No Impact.

Archaeological Resources

Impact CUL-2 The proposed project could cause a substantial adverse change in the significance of an archaeological resource pursuant to Section 15064.5.

Impact Analysis

As described above under Section 3.5.1, the Cultural Resources Assessment did not identify any known archaeological resources on-site and therefore, the proposed project is not anticipated to have an impact on any known archaeological resources. However, the Cultural Resources Assessment identified six prehistoric resources and one historic era resource that have been documented in the study area or within 0.5 mile of the APE. The prehistoric resources include mounds, habitation debris, and human remains, all of which indicate repeated and/or long-term prehistoric occupation of the area. Mounds. which were ubiquitous in the region during prehistoric times, were historically bulldozed to create agricultural land, although they often contained human burials below grade that may remain undisturbed by shallow agricultural activity. Therefore, although the 2019 and 2021 intensive pedestrian survey were negative, the potential for the project area to contain prehistoric resources is considered to be high. The proposed project would include ground disturbing activities that could unearth previously undiscovered resources and therefore, the proposed project would require implementation of mitigation to reduce potential impacts to a less than significant level. The proposed project would be required to implement Mitigation Measure CUL-1, which requires the preparation and implementation of an archaeological monitoring program, to reduce potential impacts to a less than significant level. With implementation of Mitigation Measure CUL-1, potential impacts to undiscovered archaeological resources would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM CUL-1:** Archaeological Monitoring Program. Prior to the start of construction activities, the Applicant shall prepare a monitoring program that would be implemented during demolition and construction activities. The monitoring program shall include the following:
 - Retention of a Qualified Archaeologist. A Qualified Archaeologist shall be retained to implement a monitoring and recovery program during all ground-disturbing activity associated with the proposed project, including grubbing, grading, and excavation activities. The qualified archaeologist shall meet the Secretary of Interior's Professional Standards for prehistoric and historic archaeology.
 - Agreement of Disposition of Recovered Artifacts. A written agreement shall be secured with a recognized museum repository regarding the final disposition and permanent storage and maintenance of any unique archaeological resources or historical resources recovered as a result of the archaeological monitoring, as well as corresponding geographic site data that might be recovered as a result of the specified monitoring program.

- **Preconstruction Briefing.** Construction personnel shall be briefed by the qualified archaeologist on procedures to be followed in the event that unique archaeological resources, historical resources, tribal cultural resources, or human remains are encountered during construction. The qualified archaeologist shall be required to provide a telephone number where they can be reached by the construction contractor, as necessary.
- Construction Monitoring. An archaeological monitor, working under the supervision of the qualified archaeologist shall observe all ground disturbing activities associated with the proposed project, including grubbing, grading, and excavation activities onand off-site. The monitor shall be authorized to halt construction in the immediate area where buried cultural resources are encountered. In the event that any historic or prehistoric subsurface cultural resources are discovered during ground disturbing activities, all work within 50 feet of the resources shall be halted and the project Applicant shall notify the City and consult with a gualified archaeologist, as applicable, to assess the significance of the find. If any find is determined to be significant, appropriate avoidance measures recommended by the consultant and approved by the City must be followed unless avoidance is determined unnecessary or infeasible by the City. Feasibility of avoidance shall be determined with consideration of factors such as the nature of the find, project design, costs, and other considerations. If avoidance is unnecessary or infeasible, other appropriate measures (e.g., data recovery, excavation) shall be instituted. Work may proceed on other parts of the project site while measures for the find are implemented.
- Monitoring Report. A complete set of daily monitoring logs shall be kept on-site throughout the earth moving activities and be available for inspection. The daily monitoring log shall be keyed to a location map to indicate the area monitored, date, assigned personnel, and results of monitoring, including the recovery of archaeological material, sketches of recovered materials, and associated geographic site data. Within 90 days of the completion of archaeological monitoring, a monitoring report shall be submitted to the City and filed with NWIC.
- Tribal Cultural Resources. If buried resources discovered during construction are identified to be tribal cultural resources, a Native American tribal representative from a tribe that is traditionally and culturally affiliated to the geographic area where the project is located will be consulted to determine the significance of the discovered resource and determine the appropriate avoidance or preservation measures or action to take in accordance with PRC Section 21084.3.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.



Draft Environmental Impact Report Environmental Impact Analysis

Human Remains

Impact CUL-3 The proposed project could disturb any human remains, including those interred outside of dedicated cemeteries.

Impact Analysis

There are no known human remains within the project site. However, as identified above under Impact CUL-2, the Cultural Resources Assessment identified six prehistoric resources that have been documented in the study area or within 0.5 mile of the APE which include mounds, habitation debris, and human remains, all of which indicate repeated and/or long-term prehistoric occupation of the area. Mounds were historically bulldozed to create agricultural land, although they often contained human burials below grade that may remain undisturbed by shallow agricultural activity. Therefore, ground disturbing activities required during construction of the proposed project could have the potential to uncover previously unknown human remains. In the event of an accidental discovery or recognition of human remains during project related construction activities, the proposed project would be required to comply with PRC Section 5097.98. Additionally, the proposed project would be required to implement Mitigation Measure CUL-2 which requires proper handling and treatment of discovered remains. Implementation of Mitigation Measure CUL-2 would reduce potential impacts to human remains to a less than significant level.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM CUL-2: Human Burials Encountered During Construction.** In the event of an accidental discovery or recognition of human remains during project construction activities, PRC Section 5097.98 shall be followed and the following steps shall be taken:
 - 1. There shall be no further excavation or disturbance of the specific location or any nearby area reasonably suspected to overlie adjacent human remains until the County Coroner is contacted to determine if the remains are Native American and if an investigation of the cause of death is required. If the coroner determines the remains are Native America, the coroner shall contact the NAHC within 24 hours, and the NAHC shall identify the person or persons it believes to be the "most likely descendant" of the deceased Native America. The most likely descendant may make recommendations to the landowner or the person responsible for the excavation work, for means of treating or disposing of, with appropriate dignity, the human remains, and any associated grave goods as provided in PRC Section 5097.98, or
 - 2. Where the following conditions occur, the landowner or his/her authorized representative shall rebury the Native American human remains and associated grave goods with appropriate dignity either in accordance with the recommendations of the most likely descendant or on the project area in a location not subject to further subsurface disturbance:

- The NAHC is unable to identify a most likely descendent or the most likely descendent failed to make a recommendation within 48 hours after being notified by the commission;
- The descendent identified fails to make a recommendation; or
- The landowner or his authorized representative rejects the recommendation of the descendent, and the mediation of the NAHC fails to provide measures acceptable to the landowner.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

This page intentionally left blank.

3.6 ENERGY

This section describes the environmental and regulatory setting energy resources. It also describes existing conditions and potential impacts relative to energy resources that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.6.1 Environmental Setting

PG&E Company provides electricity service to the City of Newark (PG&E 2014). Upon buildout of the project site, electricity to the project site would be provided by PG&E. All electricity infrastructure would be located underground and would tie-in to existing infrastructure, see Appendix B.

In February 2018, PG&E announced that it had reached California's 2020 renewable energy goal three years ahead of schedule (Solar Industry 2018) and now delivers nearly 93 percent of its electricity from GHG-free resources. Approximately 50 percent of PG&E's electricity came from renewable resources including solar, wind, geothermal, biomass and small hydroelectric sources in 2021. Additionally, approximately 39 percent of PG&E's total electric power mix is from nuclear sources, 4 percent from large hydroelectric, and 7 percent natural gas (PG&E 2022).

3.6.2 Regulatory Setting

Federal

Federal Energy Regulatory Commission

The Federal Energy Regulatory Commission (FERC) is an independent agency that regulates the interstate transmission of electricity, natural gas, and oil. FERC also reviews proposals to build liquefied natural gas terminals and interstate natural gas pipelines, and licenses hydropower projects. Licensing of hydroelectric facilities under FERC's authority includes input from state and federal energy and power generation, environmental protection, fish and wildlife, and water quality agencies.

Federal Energy Conservation Policy Act

The Energy and Policy Conservation Act was enacted by Congress in 1975. This Act established the first fuel economy standards for on-road motor vehicles in the United States. Pursuant to the act, the National Highway Traffic Safety Administration (NHTSA) is responsible for establishing additional vehicle standards.

State

California Public Utilities Commission Requirements

The California Public Utilities Commission (CPUC) is a state agency created by a constitutional amendment to regulate privately-owned utilities providing telecommunications, electric, natural gas, water, railroad, rail transit, and passenger transportation services and in-state moving companies. CPUC is responsible for assuring that California utility customers have safe, reliable utility services at reasonable rates while protecting utility customers from fraud. CPUC regulates the planning and approval for the



Draft Environmental Impact Report Environmental Impact Analysis

physical construction of electric generation, transmission, or distribution facilities and local distribution pipelines of natural gas.

California Integrated Energy Policy

SB 1389 requires the CEC to "conduct assessments and forecasts of all aspects of energy industry supply, production, transportation, delivery and distribution, demand, and prices. The Energy Commission shall use these assessments and forecasts to develop energy policies that conserve resources, protect the environment, ensure energy reliability, enhance the state's economy, and protect public health and safety" (PRC Section 25301[a]). The CEC adopts an Integrated Energy Policy Report every two years and an update every other year. At the time of the NOP publication, the CEC had published its 2018 report and the 2020 report was circulated for public comments in January 2020. The report noted California's policy initiatives to reduce GHG and transform California's electricity system. The report also noted the additional efforts required to decarbonize California's overall energy system and invest in managing our aging energy infrastructure while planning for the future.

Title 20 and Title 24, California Code of Regulations

New buildings constructed in California must comply with the standards in Title 20, Energy Building Regulations, and Title 24, Energy Conservation Standards, of the CCR. Title 20 contains a range of standards, such as power plant procedures and siting, energy efficiency standards for appliances, and ensuring reliable energy sources are provided and diversified through energy-efficiency and renewable energy resources. Title 24 (AB 970) contains energy-efficiency standards for residential and nonresidential buildings based on a state mandate to reduce California's energy demand. Specifically, Title 24 addresses a number of energy-efficiency measures that impact energy used for lighting, water heating, heating, and air conditioning, including the energy impact of the building envelope such as windows, doors, skylights, wall/floor/ceiling assemblies, attics, and roofs. In addition, the new 2019 standards require rooftop solar on all new residential development under three stories. The 2022 Energy Code adopted on August 11, 2021, by the California Energy Commission made changes and updates to Title 24 standards. Buildings whose permit applications are applied for on or after January 1, 2023, must comply with the 2022 Energy Code. The new 2022 standards require all new homes and buildings constructed in 2023 or later to have all electric supply panels and circuitry to support all-electric appliances and heating to reduce natural gas appliance use in new developments.

Part 11 of Title 24 is the CALGreen code, which sets minimum and mandatory sustainability requirements to reduce environmental impact through better planning, design, and construction practices. CALGreen works along with the mandatory construction codes of Title 24 and is enforced at the local level. Any project-related construction would be required to comply with the Title 24 codes currently in place, including the CALGreen code. The existing 2019 standards became effective in January 2020.

Assembly Bill 1493 – Clean Car Standards (Pavley)

This bill was passed in 2002 and requires CARB to develop and implement regulations to reduce automobile and light truck GHG emissions, through mandating gradual reductions in global warming pollutants from cars and light trucks sold in California from 2009 through 2016. The average gram-permile reduction of GHG emissions from new California cars and light trucks is required to be about 30 percent in 2016 compared to model year 2004 vehicles.



Draft Environmental Impact Report Environmental Impact Analysis

CARB adopted the Advanced Clean Cars (ACC) program in 2012 in coordination with EPA and NHTSA. The ACC program combined the control of criteria pollutants and GHG emissions into a single coordinated set of requirements for model years 2015 through 2025. CARB adopted a new approach to passenger vehicles—cars and light trucks—by combining the control of smog-causing pollutants and GHG emissions into a single coordinated package of standards. The new approach also includes efforts to support and accelerate the numbers of plug-in hybrids and zero-emission vehicles in California. The new standard drops GHG emissions to 166 grams per mile, a reduction of 34 percent compared to 2016 levels, through 2025.

Senate Bill 350 and Senate Bill 100

Under SB 350, the state of California committed to reaching 50 percent renewable energy by December 31, 2030. SB 100 revises these goals to achieve 50 percent renewable resources by December 31, 2026 and achieve a 60 percent target by December 31, 2030 in order to plan for 100 percent of total retail sales of electricity in California to come from eligible renewable energy and zero carbon resources by December 31, 2045.

Executive Order B-48-18

The executive order (EO) was designed to boost the use of zero-emission vehicles (ZEVs), electric vehicle charging infrastructure, and hydrogen refueling infrastructure in California. The order will implement the Governor's target of 5 million ZEVs on the road by 2030 and 250,000 vehicle charging stations and 200 hydrogen refueling stations by 2025.

Warren-Alquist Energy Resources Conservation and Development Act

Initially passed in 1974 and amended since, the Warren-Alquist Energy Resources Conservation and Development Act (Warren-Alquist Act) created the CEC, California's primary energy and planning agency. The seven responsibilities of CEC are: forecasting future energy needs, promoting energy efficiency and conservation through setting standards, supporting energy related research, developing renewable energy resources, advancing alternative and renewable transportation fuels and technologies, certifying thermal power plants 50 megawatts or larger, and planning for and directing state responses to energy emergencies. CEC regulates energy resources by encouraging and coordinating research into energy supply and demand problems to reduce the rate of growth of energy resources and encourages CEC to explore renewable energy options that would be in line with environmental and public safety goals. (Warren-Alquist Act PRC section 25000 et seq., Public Resources Code Section 21100(b)(3), and CEQA Guidelines Section 15126.2(b))

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project (City of Newark 2013a):

Goal CS-5: Greenhouse Gas Reduction. Reduce GHG emissions in Newark and make reduction of the City's carbon output a high priority.



Draft Environmental Impact Report Environmental Impact Analysis

• **Policy CS-5.3: Alternative Fuel Vehicles.** Encourage the use of alternative fuel and electric vehicles and development of the necessary infrastructure for such vehicles to be viable in Newark.

Goal CS-7: Energy Conservation. Maximize opportunities for energy efficiency, conservation, and independence.

- **Policy CS-7.1: Reducing Energy Use.** Support measures to reduce energy consumption and increase energy efficiency in residential, commercial, industrial, and public buildings.
- Policy CS-7.2: Renewable Energy Sources. Support the expanded use of renewable energy sources such as wind and solar by Newark residents and businesses, the City of Newark, and other government agencies.
- **Policy CS-7.3: Designing for Energy Efficiency.** Support building design, site planning, and subdivision design methods that reduce heating and cooling costs and achieve greater energy efficiency.

City of Newark Climate Action Plan

In 2010, the City of Newark adopted their CAP Initial Framework to present measures to reduce local GHG emissions; meet state, regional, and local reduction targets; and streamline future environmental review. The CAP includes residential community action items including Item 4.3 that require developers to employ energy conservation strategies in developments. Consistency with the City's CAP is discussion in Section 3.8, Greenhouse Gas.

City of Newark Municipal Code

- Chapter 15.22: California Energy Code. Chapter 15.22 of the City's Municipal Code adopts the 2022 California Energy Code pursuant to the provisions of the California Government Code Section 50020 et seq.
- Chapter 15.23: California Green Buildings Standards Code. Chapter 15.23 adopts the 2022 California Green Building Standards Code pursuant to the provisions of California Code Section 50020 et seq.

3.6.3 Environmental Impacts

This section analyzes the project's potential to result in significant energy impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The energy requirements for the proposed project were determined using the construction and operational estimates generated from the Air Quality Analysis (refer to Appendix B). Short-term construction and long-term energy consumption are discussed below. Energy consumption calculations were prepared by Helix Environmental Planning Inc and are provided in Appendix B. The analysis then uses this information to determine whether the proposed project's energy use would be considered



Draft Environmental Impact Report Environmental Impact Analysis

wasteful, inefficient, or unnecessary, taking into account available energy supplies and existing use patterns, the proposed project's energy efficiency features, and compliance with applicable standards and policies aimed to reduce energy consumption.

Thresholds of Significance

In accordance with the Appendix G of the CEQA Guidelines, the following questions were analyzed and evaluated to determine whether the proposed project's impacts to energy are significant. Would the proposed project:

- Result in potentially significant environmental impacts due to wasteful, inefficient, or unnecessary consumption of energy resources, during project construction or operation?
- Conflict with or obstruct a state or local plan for renewable energy or energy efficiency?

Project Impact Analysis and Mitigation Measures

Wasteful, Inefficient, or Unnecessary Use of Energy

Impact EN-1 The proposed project would not result in potentially significant environmental impact due to wasteful, inefficient, or unnecessary consumption of energy resources, during project construction or operation.

Impact Analysis

Energy consumption from both short-term construction and long-term operations are discussed separately below.

Construction Energy Demand

Construction of the proposed project would require the use of fuels (primarily gasoline and diesel) for the operation of construction equipment and vehicles to perform a variety of activities, including excavation, hauling, paving, and vehicle travel. Energy in the form of electricity may also be consumed by some pieces of construction equipment, such as welding machines, power tools, lighting, etc. The proposed project would require 46,274 gallons of diesel fuel for construction off-road equipment. On-road vehicles would require 6,250 gallons of diesel and 9,739 gallons of gasoline during construction. In total, construction would require 52,524 gallons of diesel and 9,739 gallons of gasoline (Appendix B). Construction equipment used on-site would generally include, but would not be limited to, tractors, cranes, and excavators. The list of anticipated construction equipment can be found in Table 3.3-5 in Section 3.3, Air Quality. Construction equipment used on-site would be standard for new land development in the area. While the proposed project would include demolition and removal of existing structures and infrastructure, there are no other unique site features of project characteristics that would necessitate the use of construction equipment that would be less energy efficient than at comparable construction sites in other parts of the state. Construction equipment utilized for development of the proposed project would be U.S. EPA Tier 4 certified or have CARB approved engine retrofit kits certified to have emissions equivalent to Tier 4 standards and would meet the latest standards set forth by BAAQMD for construction equipment. Therefore, it is expected that construction fuel consumption associated with the proposed project would not be any more inefficient, wasteful, or unnecessary than at other construction sites in the region, and as such, impacts would be less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Long-Term Energy Demand Building Energy Demand

Buildings and infrastructure constructed pursuant to the proposed project would comply with the versions of CCR Titles 20 and 24, including CALGreen, that are applicable at the time that building permits are issued. The proposed project would be designed to be 100 percent electric and would not utilize natural gas. The proposed project is estimated to demand 974,468 kWh of electricity per year (Appendix B). This would represent an increase in demand for electricity

The proposed project is a new residential subdivision with home sizes and lots comparable to others in the area and the individual units would be built in accordance with the latest energy code to maximize efficiency. Therefore, it would be expected that building energy consumption associated with the proposed project would not be any more inefficient, wasteful, or unnecessary than for any other similar buildings in the region. Current state regulatory requirements for new building construction contained in the 2022 CALGreen and Title 24 standards would increase energy efficiency and reduce energy demand in comparison to existing commercial structures on the site, and therefore would reduce actual environmental effects associated with energy use from the proposed project. Additionally, the CALGreen and Title 24 standards have increased efficiency standards through each update.

Therefore, while the proposed project would result in increased electricity demand, the electricity would be consumed more efficiently and would be typical of townhome development. Compliance with future building code standards would result in increased energy efficiency.

Based on the above information, the proposed project would not result in the inefficient or wasteful consumption of electricity, and impacts would be less than significant.

Transportation Energy Demands

Annual vehicle fuel consumption for the proposed project is estimated to be 29,414.5 gallons of diesel and 215,961.9 gallons of gasoline. The proposed project would constitute new development within an established community and would not be opening a new geographical area for development such that it would draw mostly new trips or substantially lengthen existing trips. The proposed project would be well positioned to accommodate existing population. For these reasons, it would be expected that vehicular fuel consumption associated with the proposed project would not be any more inefficient, wasteful, or unnecessary than for any other similar land use activities in the region, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Conflict with Renewable Energy/Energy Efficiency Plan

Impact EN-2 The proposed project would not conflict with or obstruct a state or local plan for renewable energy or energy efficiency.

Impact Analysis

The City's General Plan includes a goal to maximize opportunities for energy efficiency, conservation, and independence. The City's Climate Action Plan (CAP) also includes strategies to reduce energy use and increase building efficiency. The proposed project would not conflict with the energy objectives of the General Plan nor the strategies in its CAP. The proposed project's consistency with the City's CAP is analyzed under Section 3.8, Greenhouse Gas Emissions. The proposed project would constitute development near an existing community. The site is located within a mile of other residential uses and located near commercial sites. As the site is located within the City, it would not be opening a new geographical area for development such that it would draw mostly new trips, or substantially lengthen existing trips. The proposed project would comply with the versions of CCR Titles 20 and 24, including CALGreen, that are applicable at the time that building permits are issued and with all applicable City measures.

For the above reasons, the proposed project would not conflict with or obstruct a state or local plan for renewable energy or energy efficiency, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.

This page intentionally left blank.

3.7 GEOLOGY AND SOILS

This section describes the environmental and regulatory setting for geology and soils. It also describes existing conditions and potential impacts related to geology and soils that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.7.1 Environmental Setting

Regional Setting

The City is located within the USGS Newark and Niles Quadrangle 7.5-minute topographic area. The City is typified by low topographic relief, with gentle slopes to the southwest in the direction of South San Francisco Bay. In general, the soils beneath Newark are dominated by very deep, poorly-drained, fine-grained soils such as clays and silty clay loams, with lesser areas of deep well-drained silty loam in the northeast corner of the City and very deep, very poorly drained clays in the tidelands that flank the southwest edge of the City's plan area (City of Newark 2013b). Many soil types found beneath the City are characterized by low permeability, are highly erodible, and have high shrink-swell potential. Surficial geology in the City is almost exclusively comprised of Quaternary alluvium and estuarine sediment.

Though no active faults pass through the City, the Bay Area is a seismically active region and the threat of earthquakes is pervasive (City of Newark 2013a). The City is located 2 miles west of the Hayward fault, 9 miles east of the San Andreas fault, and 13 miles west of the Calaveras fault. Even though no active faults pass directly through the City, an earthquake of moderate to high magnitude within the Bay Area could cause significant damage in the City. Strong ground shaking that occurs during earthquakes can induce other geologic hazards such as liquefaction, landslides, subsidence, lateral spreading, or collapse. The potential for these geologic hazards ranges from low to very high and depends on soil conditions, groundwater levels, and slope stability.

The 1972 Alquist-Priolo Earthquake Fault Zoning Act requires the California Geological Survey to establish regulatory Earthquake Fault Zones around the surface ruptures of active faults to reduce the hazard of surface fault rupture to structures built for human occupancy. There are no Alquist-Priolo Earthquake Fault Zones in the City (City of Newark 2013b).

Project Site Setting and Soils

The project site's topography and surrounding areas are relatively flat. The project site is currently developed as a Pick-n-Pull Auto Dismantler site and is composed of three parcels totaling approximately 29 acres. According to the Design Level Geotechnical Investigation Report (design level geotechnical report) prepared by Berlogar Stevens and Associates on June 21, 2021 for the project site, the northern parcel of the site has a surface elevation of about 10 feet around the perimeter of the site with surface elevation rising to 15 feet in the center of the northern parcel (Appendix E). The southern portion of the site is slightly sloping with an approximate elevation of 10 feet on the west and 4.5 feet on the east. Soil sampling conducted for the project site for the design level geotechnical report identified that the northern portion of the site was found to have between 2.5 and 4 feet of uncontrolled fill. The uncontrolled fill was composed of sandy clay, clayey sand and some concrete rubble. The uncontrolled fill was underlain by medium stiff silty clays and fat clays. The southern portion of the site was blanketed by approximately 1



Draft Environmental Impact Report Environmental Impact Analysis

foot of sandy gravel which was underlain by medium stiff silty clays and fat clays. The soil sampling procedures encountered clays and silty clays to depths between 30 and 35 feet. The clays were underlain by sand and silt sands to a depth of about 50 feet. According to Figure 4.5-3 in the General Plan EIR, the project site is in an area with liquefaction risk but is not in an area at risk of earthquake induced landslides (City of Newark 2013b).

The Applicant reached out to BCDC staff on June 29, 2023, to request confirmation on whether the three parcels within the project site was located within BCDC jurisdiction. A response was received from BCDC staff on July 17, 2023, where BCDC staff confirmed that project site parcels identified as APN 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00 are not located within BCDC jurisdiction (Yuri Jewett, personal communication, July 17, 2023). Therefore, the project site parcels are not located within BCDC jurisdiction. As a courtesy, City staff would follow up with BCDC during the public review period.

Seismic Hazards

The project site is located in the seismically active San Francisco Bay region. The San Andreas fault forms a portion of the boundary between two independent tectonic plates on the surface of the earth. Seismic hazards in the City include earthquake ground shaking and liquefaction, and geologic hazards that are not specifically related to earthquakes include unstable soils.

Faults

In the San Francisco Bay Area, movement across this plate boundary is concentrated on San Andreas fault (SAF) system. However, it is also disturbed to a lesser extent across a number of parallel and subparallel faults which include the Seal Cove-San Gregorio, Hayward, Rodgers Creek, Calaveras, Concord-Green Valley and Greenville faults. Together these faults are referred to as the SAF system.

Additionally, there are several other active or potentially active faults capable of producing ground shaking at the project site. Local faults that have a potential to cause ground shaking at the site include the Quimby, Evergreen, Silver Creek, Monte Vista-Shannon, Sargent, and Zayante-Vergeles faults. These faults could be triggered by activity within the Hayward Fault Zone or along the San Andreas Fault Zone.

The design level geotechnical report prepared for the proposed project identified that there are no known fault traces that cross the project site and the potential for fault rupture at the site is low (Berlogar Stevens and Associates 2021).

Ground Shaking and Ground Failure

Primary seismic hazard concerns include potential ground shaking and ground rupture along the surface trace of faults. Secondary seismic hazards are caused by the interaction of ground shaking with soft or unstable soils, resulting in liquefaction, settlement, and landslides. Ground shaking can vary over an area as a result of factors such as topography, bedrock type and the location and orientation of a fault rupture due to seismic activity. Ground settlement (i.e., subsidence) is the lowering of the ground surface during seismic activity and is caused by consolidation or the failure of the ground foundation, densification of soil material, or liquefaction (discussed below). Ground failure can cause serious direct damage or collapse of infrastructure caused by seismic activity and is considered the second "primary" earthquake hazard. The severity of ground failure depends on the strength and depth of the earthquake, but there are several



Draft Environmental Impact Report Environmental Impact Analysis

other contributing factors such as the regional geology, local topography and the site-specific ground characteristics within the project area.

The primary seismic hazard within the project area is associated with strong ground shaking from the nearby faults. Of the faults listed above, there are three major faults with the greatest potential for causing severe shaking at the site. The three faults are the Hayward, San Andreas, and Calaveras faults. The San Andreas fault, located approximately 14 miles southwest of the project site, dominates the structure and seismicity of the San Francisco Bay Area. The Hayward fault is located approximately 3 miles northeast of the project site and the Calaveras fault is located approximately 8 miles northeast of the project site.

According to the design level geotechnical report prepared for the proposed project, the probability of a magnitude of 6.7 or higher earthquake occurring between the years 2014 and 2044 along the Hayward fault is 14.3 percent, the Calaveras fault is 7.4 percent, and the San Andreas fault is 6.4 percent.

Lateral Spreading

Lateral spreading is a potential hazard associated with liquefaction. Liquefaction occurs when a subsurface soil layer liquefies and the upper non-liquefiable crust slides down gradient as large blocks over the liquefied soil toward a free-face (such as a descending slope, an incised river channel or open body of water), creating extensional ground cracking or fissures. Based on the results of the liquefaction analysis and the local topography, the design level geotechnical report identified that the potential for lateral spreading to occur at the project site is low (Berlogar Stevens and Associates 2021).

Liquefaction

Soil liquefaction occurs when ground shaking from an earthquake causes a sediment layer saturated with groundwater to lose strength and take on the characteristics of a fluid, thus becoming similar to quicksand. Factors determining liquefaction potential are soil type, the level and duration of seismic ground motions, the type and consistency of soils, and the depth to groundwater. Loose sands and peat deposits, along with recent Holocene age deposits, are more susceptible to liquefaction, while older deposits of clayey silts, silty clays, and clays deposited in freshwater environments are generally stable under the influence of seismic ground shaking.

Based on the analysis conducted for the design level geotechnical report, the liquefaction induced settlement potential at the project site was found to range between 1.6 and 2.8 inches. The analyses showed that the predominant contributor to the settlement potential was a sand layer at depths between 30 and 40 feet. Based on the analyses and geologic setting of the site, the design level geotechnical report estimates a liquefaction induced differential settlement of up to 0.5 inch across 100 feet (Berlogar Stevens and Associates 2021).

Geotechnical Reports Results

As mentioned above under Project Site Setting and Soils, a Design Level Geotechnical Investigation was conducted and prepared for the project site by Berlogar Stevens and Associates on June 15, 2021. Prior to the preparation of the design level geotechnical report, a Due Diligence Geotechnical Assessment was



Draft Environmental Impact Report Environmental Impact Analysis

prepared for the project site by Berlogar Stevens and Associates on April 1, 2019 (preliminary geotechnical report). The preliminary and design level geotechnical reports are provided as Appendix E.

The design level geotechnical report identified that the potential for fault rupture at the project site is very low due to the closest known active fault located with State-Designated Zone of Required Investigation is the Hayward fault, which is located approximately 3.2 miles to the northeast of the project site. Though fault rupture is not considered a concern at the project site, the project site should expect to experience at least one moderate to large earthquake during the lifespan of the development due to being located within a seismically active region. Some degree of structural damage due to strong seismic ground shaking should be expected at the site; however, the design level geotechnical report identified that the risk can be reduced through adherence with seismic design codes.

The design level geotechnical report identified that liquefaction induced settlement potential at the project site was found to range between 1.6 and 2.8 inches. The analysis determined that the predominant contributor to the settlement potential at the site is a sand layer at depths between 30 and 40 feet. Based on the results of the analyses and the geologic setting of the project site, the design level geotechnical report estimates that liquefaction induced differential settlement could be up to 0.5 inch across 100 feet. The design level geotechnical report also identified that liquefaction induced ground surface disruption potential and lateral spreading potential to occur at the project site is low and that there is no potential for seismic-induced compaction of unsaturated sands at the site. The design level geotechnical report identified that with the use of post-tensioned concrete slab-on-grade foundations for the planned residential structures, temporary softening of site soils would not have a significant impact on the proposed structures.

The preliminary geotechnical report and the design level geotechnical report both concluded that site conditions that could impact the proposed development included uncontrolled fill, seismic-induced (liquefaction) site settlement potential of 1 to 2.5 inches, moderately compressible soils, expansive soils, and corrosive soils. Both the preliminary and design level geotechnical reports included recommendations on methods to reduce potential impacts from construction of the proposed project. Recommendations included would be implemented into the proposed project design and consideration of remediation of contaminated soils would be taken into account for construction.

3.7.2 Regulatory Setting

Federal

Earthquake Hazards Reduction Act of 1977

The Earthquake Hazards Reduction Act of 1977 (FEMA 1977) established the National Earthquake Hazards Reduction Program (NEHRP) "to reduce the risks of life and property from future earthquakes in the United States through the establishment and maintenance of an effective earthquake hazards reduction program." The National Earthquake Hazards Reduction Program Act (NEHRPA) significantly amended this program in 1990 by refining the description of the agency responsibilities, program goals, and objectives. The four principal goals of the NEHRP are:

Develop effective practices and policies for earthquake loss reduction and accelerate their implementation;



Draft Environmental Impact Report Environmental Impact Analysis

- Improve techniques for reducing earthquake vulnerabilities of facilities and systems;
- Improve earthquake hazards identification and risk assessment methods, and their use; and
- Improve the understanding of earthquakes and their effects.

The NEHRPA designates the Federal Emergency Management Agency as the lead agency of the program and assigns it several planning, coordinating, and reporting responsibilities.

State

Alquist-Priolo Fault Zoning Act

In 1972, the Alquist-Priolo Earthquake Fault Zoning Act was passed to mitigate the effects of surface faulting on structures designed for human occupancy. This act required the State Geologist to delineate Earthquake Fault Zones along known active faults that have a relatively high potential for ground rupture. Faults that are zoned under the Alquist-Priolo Earthquake Fault Zoning Act must meet the strict definition of being "sufficiently active" and "well-defined" for inclusion as an Earthquake Fault Zone. The Earthquake Fault Zones are revised periodically, and they extend 200 to 500 feet on either side of identified fault traces. No structures for human occupancy may be built across an identified active fault trace. An area of 50 feet on either side of an active fault trace is assumed to be underlain by the fault, unless proven otherwise. Proposed construction in an Earthquake Fault Zone is permitted only following the completion of a fault location report prepared by a California Registered Geologist.

California Building Standards Code

The California Building Standards Code establishes building requirements for construction and renovation. The most recent version of the California Building Standards Code was published July 1, 2022, with an effective date of January 1, 2023. The California Building Standards Code is based on the International Code Council's Building and Fire Codes. Included in the California Building Standards Code are the Electrical Code, Mechanical Code, Plumbing Code, Energy Code, and Fire Code. Title 24, Part 2: California Building Code (CBC) of the California Building Standards Code of the CCR contains specific requirements for construction with respect to earthquakes and seismic hazards intended to be protective of public health. Chapter 16 Section 1613, Earthquake Loads, deals with structural design and requires that every structure, and portion thereof, including nonstructural components that are permanently attached to structures and their supports and attachments, shall be designed and constructed to resist the effects of earthquake motions.

California Seismic Hazards Mapping Act

The California Seismic Hazards Mapping Act of 1990 (PRC Section 1690-2699.6) addresses seismic hazards other than surface rupture, such as liquefaction and induced landslides. The Seismic Hazards Mapping Act specifies that the lead agency for a project may withhold development permits until geologic or soils investigations are conducted for specific sites and mitigation measures are incorporated into plans to reduce hazards associated with seismicity and unstable soil.



Draft Environmental Impact Report Environmental Impact Analysis

National Pollutant Discharge Elimination System Permit

In California, the SWRCB administers the USEPA's promulgated regulations (55 CFR 47990) requiring the permitting of stormwater-generated pollution under the NPDES. In turn, the SWRCB's jurisdiction is administered through RWQCBs. Pursuant to these federal regulations, an operator must obtain a General Permit under the NPDES Stormwater Program for all construction activities with ground disturbance of 1 acre or greater. The General Permit requires the implementation of BMPs to reduce pollutant loads into the waters of the State and measures to reduce sediment and erosion control. In addition, a SWPPP must be prepared. The SWPPP addresses water pollution control during construction. SWPPPs require that all stormwater discharges associated with construction activity, where clearing, grading, and excavating results in soil disturbances, must by law be free of site pollutants.

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:

Goal EH-1: Reducing Hazard Exposure. Reduce the potential for injury, harm, property damage, and loss of life resulting from environmental hazards.

• **Policy EH-1.1: Development Regulations and Code Requirements.** Establish and enforce development regulations and building code requirements to protect residents and workers from flooding, liquefaction, earthquakes, fires, and other hazards.

Goal EH-2: Geologic Hazards. Reduce risks to life and property associated with geologic hazards.

• **Policy EH-2.1: Earthquake Safety in New Construction.** Require new development to meet structural integrity standards which minimize the potential for damage during earthquakes.

Goal CS-1: Environmental Protection. Protect Newark's natural environment landscape, and physical features.

• **Policy CS-1.4: Soil Erosion.** Identify and eliminate soil erosion problems on public and private lands. The potential for erosion should be considered as a design and engineering factor in new development.

3.7.3 Environmental Impacts

This section analyzes the project's potential to result in significant geology and soils impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of documents pertaining to the project site, including the General Plan, General Plan EIR, USGS earthquake seismic hazard maps and the USGS land subsidence



Draft Environmental Impact Report Environmental Impact Analysis

in California Map. The following impact discussions consider the effects of the proposed project related to geology and soils in the City.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's geology and soils impacts are significant. Would the proposed project:

- Directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving:
- Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault?
- Strong seismic ground shaking?
- Seismic-related ground failure, including liquefaction?
- Landslides?
- Result in substantial soil erosion or the loss of topsoil?
- Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction or collapse?
- Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property?
- Have soils incapable of adequately supporting the use of septic tanks or alternative waste water disposal systems where sewers are not available for the disposal of waste water?
- Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?



Project Impact Analysis and Mitigation Measures

Seismic Hazard		
Impact GEO-1	The proposed project could directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving:	
	i) Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault. Refer to Division of Mines and Geology Special Publication 42.	
	ii) Strong seismic ground shaking.	
	iii) Seismic-related ground failure, including liquefaction.	
	iv) Landslides	

Impact Analysis

i. Fault Rupture

There are no Alquist-Priolo Earthquake Fault Zones located within or near the project site. The nearest known active fault, with a State-Designated Zone of Required Investigation, is the Hayward fault located approximately 3.2 miles northeast of the project site. Due to the lack of Alquist-Priolo fault zones near the project site, the design level geotechnical report prepared for the proposed project by Berlogar Stevens and Associated in 2021 determined that the risk of surface rupture near the project site is low and the potential for damage to structures at the project site due to rupture of a known earthquake fault is low. Thus, the proposed project would not exacerbate existing conditions by bringing people or structures into areas potentially susceptible to substantial effects, including fault rupture, that could result in substantial damage to proposed structures or infrastructure, or expose people to substantial risk of injury. Impacts associated with surface rupture from a known earthquake fault would be less than significant.

ii. Ground Shaking

Though fault rupture is not considered a concern for the project site, the site is located in a region with high seismicity and earthquake related ground shaking is expected to occur during the lifespan of the proposed project. The proposed project would comply with General Plan Policies EH-1.1 and EH-2.1 and would be constructed in conformance with the latest edition of the CBC and structural integrity standards, which includes engineering standards appropriate to withstand anticipated ground accelerations at the project site. Conformance with the earthquake design parameters of the CBC would be subject to City review as part of the development review process. Furthermore, the proposed project would be required to comply with the General Plan Action EH-2.A, which requires geotechnical studies that includes detailed investigations of ground shaking, liquefaction, soil stability, and other geologic hazards to be prepared for proposed development requirements. The required geotechnical studies under General Plan Action EH-2.A have been prepared for the proposed project and the proposed project would implement all recommendations included in the preliminary and design level geotechnical reports, as required by Mitigation Measure GEO-1. The recommendations include, but are not limited to, placement of fills,



Draft Environmental Impact Report Environmental Impact Analysis

stabilization of on-site soils, implementation of CBC seismic design standards, and placement of adequate foundations and retaining walls which would adequately support the new on-site structures during seismic events. Therefore, impacts related to ground shaking at the project site would be less than significant with implementation of Mitigation Measure GEO-1.

iii. Ground Failure, including Liquefaction

According to the City's General Plan EIR, the entire City is within a seismic hazard liquefaction zone (City of Newark 2013b). Buildout of the proposed project and adjacent off-site areas would place buildings and structures on areas susceptible to liquefaction. Therefore, the proposed project could potentially expose people and structures to substantial adverse effects associated with ground shaking, ground failure, and liquefaction. Ground failure due to liquefaction or expansive soils could compromise the structural stability of the buildings if they are not designed to accommodate liquefaction or lateral spreading.

As described above, the project design would be required to conform to the latest edition of the CBC and General Plan Action EH-2.A, which requires site-specific geotechnical reports to be prepared for all new developments in the City. The proposed project would comply with General Plan Policies EH-1.1 And EH-2.1 and would be designed and constructed to meet development regulations and building code requirements as well as structural integrity standards to minimize the potential for damage during earthquakes. The proposed project would also be required to comply with the Newark Municipal Code Section 15.50 which requires a liquefaction study be conducted and recommended mitigation measures included in the report be implemented into the project design. The design level geotechnical report was completed for the project site in 2021 which determined that there is a seismic-induced liquefaction site settlement potential of 1 to 2.5 inches at the project site and that the soils in the area have a moderate to high expansion potential. The preliminary and design level geotechnical reports recommend the use of post-tension concrete slab-on-grade foundations to mitigate impacts that could result from liquefaction or expansive soils. The recommendations included in the geotechnical reports relating to liquefaction hazard would be incorporated into the project design as part of Mitigation Measure GEO-2. Implementation of mitigation measures would reduce potential impacts. Additionally, the proposed project includes remediation activities of on-site soils prior to the start of construction activities, required by Mitigation Measure HAZ-1 as described in Section 3.9, Hazards and Hazardous Materials. Contaminated soils are present in shallow portions of the project site and therefore, the proposed project is estimated to remove approximately 39,000 CY of vegetation, contaminated soil, demolition debris, and other cleared materials prior to the start of construction activities. Removal of contaminated soils on-site would be completed through excavation of on-site soils from one foot to six feet bgs. Remediation activities may result in removal of some liquefiable soils present at the project site. Most soils excavated on-site would be disposed of off-site; however, some soils excavated may provide to be suitable for on-site reuse. Soils suitable for reuse would be stockpiled and may be reused on-site with appropriate regulatory concurrence. Clean imported fill would be brought in to replace removed soils and to elevate the proposed pad grades. The clean imported fill would be evaluated by a soil engineer prior to use in accordance with recommendations included in the design level geotechnical report. Therefore, impacts related to liquefaction would be less than significant with Mitigation Measure GEO-1, GEO-2, and HAZ-1 incorporated.



Draft Environmental Impact Report Environmental Impact Analysis

iv. Landslides

The project site is relatively flat and according to Figure 4.5-3 in the General Plan EIR, the project site is not located within or near an area at risk of earthquake induced landslides. Therefore, the potential for landslides to occur at or near the project site is low. No impact would occur.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HAZ-1 is required.

- MM GEO-1: Implement Recommendations included in the Geotechnical Studies. Prior to issuance of grading permits, the Applicant shall be required to incorporate all mitigation measures and design recommendations contained within the preliminary and design level geotechnical reports prepared by Berlogar Stevens and Associates in 2019 and 2021 into relevant project plans and specifications. These specifications pertain to but are not limited to expansive soils, building foundations, foundation drainage, and backfill of excavations. The geotechnical reports prepared for the proposed project includes recommendations such as the use of post-tension concrete slab-on-grade foundation to support structures placed on expansive soils, excavation of uncontrolled fill, utility trench excavation and backfill requirements, placement of retaining walls, and implementation of California Building Code Seismic Design Parameters. The project site plans shall be submitted to the City and reviewed as part of the development review process.
- **MM GEO-2: Mitigation of Liquefaction-Induced Impacts on Buildings.** Prior to issuance of grading permits, the Applicant shall be required to incorporate all mitigation measures and design recommendations relating to liquefaction contained within the preliminary and design level geotechnical reports prepared by Berlogar Stevens and Associates in 2019 and 2021 into relevant project plans and specifications. Structural mitigation for liquefaction should, at a minimum, include the use of relatively stiff structural concrete slab-on-grade foundations, such as post-tensioned concrete slabs-on-grade, to ensure. foundations resist the effects of liquefaction-induced differential settlement. Strengthening connections within each structure may also be necessary, subject to review by the City. The project site plans, with all geotechnical design measures incorporated, shall be submitted to the City and reviewed as part of the development review process.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.


Draft Environmental Impact Report Environmental Impact Analysis

Erosion

Impact GEO-2	The proposed project could result in substantial soil erosion or the loss of
	topsoil.

Impact Analysis

Construction activities associated with the proposed project would require demolition, grading, utility connections, building construction, construction of the new streets, development of 203 single-family residences, and landscaping on the 29 acre project site. The proposed project would comply with General Plan Policy CS-1.4 which requires identification and elimination of soil erosion problems and the potential for erosion to be considered as a design and engineering factor in new development. Construction activities could expose unprotected soils to stormwater runoff, causing erosion and loss of topsoil. Projects that disturb 1 acre or more of soils during construction are required to comply with the NPDES permitting program and implement a Stormwater Pollution Prevention Plan (SWPPP) that identifies BMPs to control the discharge of sediment and other pollutants during construction. As described in Section 3.10, Hydrology and Water Quality, the proposed project would implement a SWPPP and associated BMPs as part of Mitigation Measure HYD-1 to reduce erosion impacts. Therefore, soil erosion impacts associated with construction impacts would be less than significant with implementation of Mitigation Measure HYD-1.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HYD-1 is required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Unstable Geologic Unit or Soil

Impact GEO-3 The proposed project could be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse.

Impact Analysis

The proposed project is identified to be located in an area where there is potential for liquefaction and expansive soil hazards. The preliminary and design level geotechnical reports prepared by Berlogar Stevens and Associates identified that the project site would have a seismic induced liquefaction site settlement potential of 1 to 2.5 inches and has a moderate to high expansion potential. However, it also identified that potential for lateral spreading is low. The project site is not at risk of on- or off-site landslides. Furthermore, the project site is not adjacent to a stream bank, levee, or other open face that would be susceptible to lateral spreading.

The proposed project would be required to comply with the latest edition of the CBC and General Plan Action EH-2.A, which requires geotechnical studies such as a liquefaction study or soils engineering report be prepared for all new developments in the City. A preliminary and design level geotechnical

Draft Environmental Impact Report Environmental Impact Analysis

report has been prepared for the proposed project and the recommendations identified in the geotechnical reports would be incorporated into the project design as part of Mitigation Measures GEO-1 and GEO-2. The City would review the project design plans during the development review process to confirm these recommendations are incorporated into the proposed project.

The preliminary geotechnical report prepared for the proposed project identified that depth to groundwater at the project site vary from approximately 4 to 8.5 feet bgs and the groundwater map from the California Geologic Survey indicated historically-high groundwater at a depth of approximately 5 feet and therefore, groundwater could be encountered during construction activities. Excavation activities near or below the groundwater table has the potential to expose surrounding roadways, utilities, and UPRR tracks to unstable soils due to vertical movement and settlement. Therefore, excavations would require dewatering and shoring to protect the surrounding environment. Mitigation Measure GEO-3 would require the preparation and implementation of shoring and dewatering plans. Due to the proposed project requiring jack-and-bore operations under the UPRR tracks for utility line connections for existing water and sanitary sewer mains located on the north side of the UPRR tracks, all shoring would be constructed in accordance with UPRR requirements. Additional jack-and-bore would be required to extend the potable and non-potable water mains from the southwest corner of the Sanctuary Development located adjacent to the east of the project site, within the Old Addition Road alignment adjacent and parallel to ACFC&WCD Line D channel, to the UPRR ROW. The potable and non-potable water mains would be jack-and-bored under the Line D channel and the UPRR ROW and all shoring would be constructed in accordance with UPRR requirements. Additionally, the design level geotechnical report included recommendations for utility trench excavation and backfill which would be implemented into the project as required by Mitigation Measure GEO-1.

As required by Mitigation Measure HAZ-1, the proposed project includes remediation activities of on-site soils prior to the start of construction activities to remove shallow contaminated soils present on-site. Excavation activities required for remediation would require excavation from one foot to six feet bgs and remediation activities may result in the removal of unstable soils on-site. The proposed project is estimated to remove approximately 39,000 CY of vegetation, contaminated soil, demolition debris, and other cleared materials prior to the start of construction activities and clean imported fill would be brought in to replace removed soils. The clean imported fill would be evaluated by a soil engineer prior to use in accordance with recommendations included in the design level geotechnical report to ensure imported fill are suitable for use at the project site to mitigate any potential impacts from unstable soils. Most soils excavated on-site reuse. Soils suitable for reuse would be stockpiled and may be reused on-site with appropriate regulatory concurrence.

As such, with implementation of Mitigation Measures GEO-1, GEO-2, GEO-3, and HAZ-1, the proposed project would not be located on a geologic unit or soil that is unstable and impacts would be less than significant.

Level of Significance Before Mitigation Potentially Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

Mitigation Measure GEO-1, Mitigation Measure GEO-2, and Mitigation Measure HAZ-1 are required.

MM GEO-3: Prepare and Implement Dewatering and Shoring Plans. A dewatering plan shall be submitted to the City for approval prior to the issuance of a grading permit. At a minimum, the dewatering plan shall detail dewatering methods, location of dewatering activities, equipment, groundwater sampling, disposal, and discharge point in accordance with the applicable waste discharge requirements of the San Francisco Bay Regional Water Quality Control Board. In the event that shoring methods are implemented for any excavations, shoring plans shall be prepared and submitted to the City for approval prior to issuance of a grading permit. Shoring activities required for the jack-and-bore operations to connect utility lines under the Union Pacific Railroad tracks shall be prepared in accordance with the Union Pacific Engineering Project Specifications and Guidelines for Temporary Shoring.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Expansive Soil

Impact GEO-4 The proposed project could be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property.

Impact Analysis

The preliminary and design level geotechnical reports prepared by Berlogar Stevens and Associates identified that the surficial native clay soils in the area have a moderate to high expansion potential and therefore, could pose risks to the proposed project. The proposed project would be required to comply with the latest edition of the CBC and General Plan Action EH-2A, which requires geotechnical studies be prepared for new developments within the City. The design level geotechnical report prepared by Berlogar Stevens and Associates for the project site in 2021 included recommendations related to being located on expansive soils. The design level geotechnical report also provided recommendations related to site preparation and grading which would help reduce the potential impacts of expansive soils. Additionally, the design level geotechnical report identified the use of post-tensioned concrete slab-ongrade foundations to support residential structures on expansive soils (Berlogar Stevens and Associates 2021). The recommendations indicated in the geotechnical studies would be incorporated into the project as part of Mitigation Measure GEO-1. Compliance with Mitigation Measure GEO-1 would ensure that the proposed project is designed to withstand expansive soils and proper site preparation and grading techniques are utilized to ensure project site soils are prepared properly for placement of structures. Additionally, remediation activities required for the proposed project under Mitigation Measure HAZ-1 would require excavation of on-site soils from one foot to six feet bgs and may result in the removal of expansive soils on-site. Excavated areas would be backfilled with clean imported fill or excavated soils determined to be suitable for on-site reuse. The proposed project would be located on expansive soils; however, with implementation of Mitigation Measure GEO-1 and HAZ-1, potential impacts of expansive soils would be reduced and impacts would be less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure GEO-1 and Mitigation Measure HAZ-1 are required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Septic Tanks

Impact GEO-5 The proposed project would not have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal systems where sewers are not available for the disposal of wastewater.

Impact Analysis

The proposed project would connect directly to the City's municipal sewer system and would not require the construction of septic tanks or any other alternative wastewater disposal system. Therefore, no impact would occur.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Paleontological Resource or Geologic Feature

Impact GEO-6 The proposed project could directly or indirectly destroy a unique paleontological resource or site or unique geologic feature.

Impact Analysis

According to the City's General Plan EIR, the potential for fossil remains of significance are unlikely in the City due to the relatively recent age of Holocene Bay mud underlying the City. Additionally, there are no previous fossils found in the vicinity and there are no known paleontological resources in the City according to the University of California Museum of Paleontology Specimen Search database (City of Newark 2013b). Therefore, paleontological potential of the City is considered low. However, the proposed project would include ground disturbance during construction which could have the potential to directly or indirectly destroy an unknown unique paleontological or unique geologic feature. If unknown unique paleontological resources are discovered on-site during construction, all activities would be stopped within a 50 foot radius of the identified resource until a qualified paleontologist evaluates the finding as required by Mitigation Measure GEO-4. Implementation of Mitigation Measure GEO-4 would ensure that proper procedures for inadvertent discovery of paleontological resources are followed if unknown paleontological resources are discovered during ground disturbing activities. Therefore, impacts to paleontological or unique geologic features would be less than significant with implementation of Mitigation Measure GEO-4.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

MM GEO-4: Procedures for Paleontological Resources Discovered During Construction. If any paleontological resources are encountered during ground-disturbing or subsurface construction activities (e.g., trenching, grading), all construction activities within a 50-foot radius of the identified resource shall cease and the City shall immediately be notified. The applicant shall retain a qualified paleontologist (as approved by the City) to evaluate the find and recommend appropriate treatment of the inadvertently discovered paleontological resource. The appropriate treatment of an inadvertently discovered paleontological resource shall be implemented to ensure that impacts to the resource are avoided.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

This page intentionally left blank.

3.8 GREENHOUSE GAS EMISSIONS

This section describes the impacts on GHG emissions that would result from implementation of the proposed project. Included is a review of existing conditions, a summary of applicable policies and regulations related to GHG emissions, and analysis of environmental impacts of the proposed project. Where applicable, mitigation measures are included for significant impacts.

3.8.1 Environmental Setting

Greenhouse Gases

GHGs and climate change are cumulative global issues. The CARB and EPA regulate GHG emissions within the State of California and the U.S., respectively. While the CARB has the primary regulatory responsibility within California for GHG emissions, local agencies can also adopt policies for GHG emission reduction.

Many chemical compounds in the earth's atmosphere act as GHGs, as they absorb and emit radiation within the thermal infrared range. When radiation from the sun reaches the Earth's surface, some of it is reflected back into the atmosphere as infrared radiation (heat). GHGs absorb this infrared radiation and trap the heat in the atmosphere. Over time, the amount of energy from the sun to the Earth's surface should be approximately equal to the amount of energy radiated back into space, leaving the temperature of the earth's surface roughly constant. Many gases exhibit these "greenhouse" properties. Some of them occur in nature (water vapor, carbon dioxide [CO₂], methane [CH₄], and nitrous oxide [N₂O]), while others are exclusively human-made (like gases used for aerosols).

The principal climate change gases resulting from human activity that enter and accumulate in the atmosphere are listed below:

- **Carbon Dioxide**. CO₂ enters the atmosphere through the burning of fossil fuels (oil, natural gas, and coal), solid waste, trees and wood products, and chemical reactions (e.g., the manufacture of cement). CO₂ is also removed from the atmosphere (or "sequestered") when it is absorbed by plants as part of the biological carbon cycle.
- **Methane**. CH₄ is emitted during the production and transport of coal, natural gas, and oil. CH₄ emissions also result from livestock and agricultural practices and the decay of organic waste in municipal solid waste landfills.
- **Nitrous Oxide**. N₂O is emitted during agricultural and industrial activities as well as during combustion of fossil fuels and solid waste.
- **Hydrofluorocarbons (HFC)**. HFCs are one of several high global warming potential (GWP) gases that are not naturally occurring and are generated from industrial processes. HFC (refrigerant) emissions from vehicle air conditioning systems occur due to leakage, losses during recharging, or release from scrapping vehicles at end of their useful life.
- **Perfluorocarbons (PFC)**. PFCs are another high GWP gas that are not naturally occurring and are generated in a variety of industrial processes.



Draft Environmental Impact Report Environmental Impact Analysis

- **Sulfur Hexafluoride (SF₆)**. SF₆ is another high GWP gas that is not naturally occurring and is generated in a variety of industrial processes.
- Nitrogen Trifluoride (NF₃). NF₃ is an inorganic, colorless, odorless, toxic, nonflammable gas used in microelectronics. NF₃ is predominantly employed in the cleaning of the plasma-enhanced chemical vapor deposition chambers in the production of liquid crystal displays and silicon-based thin film solar cells.

Sources of Greenhouse Gas Emissions

On a global scale, GHG emissions are predominantly associated with activities related to energy production; changes in land use, such as deforestation and land clearing; industrial sources; agricultural activities; transportation; waste and wastewater generation; and commercial and residential land uses. World-wide, energy production including the burning of coal, natural gas, and oil for electricity and heat is the largest single source of global GHG emissions.

In 2020, GHG emissions within California totaled 369.2 million metric tons (MMT) of CO2e. Within California, the transportation sector is the largest contributor, accounting for approximately 38 percent of the total statewide GHG emissions. Emissions associated with industrial uses are the second largest contributor, totally roughly 24 percent. Electricity generation totaled roughly 16 percent. Residential, commercial, and agricultural/forestry made up the approximately 8 percent, 6 percent, and 9 percent, respectively, of the remaining GHG emissions (CARB 2022b).

The site is currently developed with a Pick-n-Pull, a self-service used auto parts store. The existing site generates GHGs from customer and employee vehicle trips to the site.

Potential Environmental Impacts

There are uncertainties as to exactly what the climate changes will be in various local areas of the earth. There are also uncertainties associated with the magnitude and timing of other consequences of a warmer planet: sea level rise, spread of certain diseases out of their usual geographic range, the effect on agricultural production, water supply, sustainability of ecosystems, increased strength and frequency of storms, extreme heat events, increased air pollution episodes, and the consequence of these effects on the economy.

Within California, climate changes would likely alter the ecological characteristics of many ecosystems throughout the state. Such alterations would likely include increases in surface temperatures and changes in the form, timing, and intensity of precipitation. For instance, historical records are depicting an increasing trend toward earlier snowmelt in the Sierra Nevada. This snowpack is a principal supply of water for the state, providing roughly 50 percent of state's annual runoff. If this trend continues, some areas of the state may experience an increased danger of floods during the winter months and possible exhaustion of the snowpack during spring and summer months. An earlier snowmelt would also impact the state's energy resources. An early exhaustion of the Sierra snowpack may force electricity producers to switch to more costly or non-renewable forms of electricity generation during spring and summer months. A changing climate may also impact agricultural crop yields, coastal structures, and biodiversity. As a result, resultant changes in climate will likely have detrimental effects on some of California's largest industries, including agriculture, wine, tourism, skiing, recreational and commercial fishing, and forestry.



3.8.2 Regulatory Setting

Federal

Executive Order 14008

EO 14008, Tackling the Climate Crisis at Home and Abroad was signed in January 2021 by President Biden. EO 14008 places the climate crisis at the forefront of foreign policy and national security planning.

Executive Order 14057

EO 14057, Catalyzing Clean Energy Industries and Jobs Through Federal Sustainability was signed in December 2021 by President Biden. EO 14057 states that it is the federal government's responsibility to lead the nation by example in order to achieve a carbon pollution free-electricity sector by 2035 and net-zero emissions economy-wide by no later than 2050. The EO lays out a series of energy efficiency and vehicle efficiency targets for the federal government to achieve, including a net-zero buildings portfolio by 2045.

EPA Reporting Rule

The U.S. EPA adopted a mandatory GHG reporting rule in September 2009. The rule would require suppliers of fossil fuels or entities that emit industrial GHGs, manufacturers of vehicles and engines, and facilities that emit 25,000 metric tons or more per year of GHG emissions to submit annual reports to the U.S. EPA.

Safer Affordable Fuel Efficient Vehicles Rule

The Safer Affordable Fuel Efficient (SAFE) Vehicle Rule was developed by the NHSTA and the U.S. EPA to update the corporate average fuel economy and GHG emissions standards for passenger cars and light trucks for model year 2021 through 2026.

State

Control of GHGs is generally regulated at the state level and is typically approached by setting emission reduction targets for existing sources of GHGs, setting policies to promote renewable energy and increase energy efficiency, and developing statewide action plans.

California has adopted statewide legislation addressing various aspects of climate change and GHG emissions mitigation. Much of this legislation establishes a broad framework for the state's long-term GHG reduction and climate change adaptation program. The governor has also issued several Eos related to the state's evolving climate change policy. Of particular importance are the following:

CEQA Guidelines

CEQA Guidelines (Title 14, Division 6, Chapter 3 of the California Code of Regulations) are administrative regulations governing the implementation of CEQA.

CEQA Guidelines Section 15064.4. Section 15064.4 of the CEQA Guidelines sets the criteria for a GHG analysis under CEQA. According to Section 15064.4, a lead agency has discretion to either quantify GHG



Draft Environmental Impact Report Environmental Impact Analysis

emissions resulting from a project and/or rely on a qualitative analysis or performance based standards. In determining significance, the lead agency should focus on the reasonably foreseeable incremental contribution of the project's emissions on the effects of climate change. The analysis should consider the following: (1) the project's contribution to GHG emissions as compared to the existing environment; (2) whether the project emissions exceed a thresholds of significance that the lead agency determines applies to the project; and (3) the extent to which the project complies with regulations or requirements adopted to implement a statewide, regional, or local plan for GHG reductions.

CEQA Guidelines Section 15126.4(c). Section 15126.4(c) sets guidelines for mitigation measures related to GHG emissions. Measures to mitigate the significant effects of GHG emissions may include: (1) measures in an existing plan or mitigation program that are required as part of the lead agency's decision; (2) reductions in emissions resulting from a project through project features or design; (3) off-site measures, including offsets; (4) measures that sequester GHGs; and (5) for plans, mitigation may include the identification of specific measures that may be implementation on a project by project basis.

Assembly Bill 32

The California State Legislature enacted AB 32, the California Global Warming Solutions Act of 2006. AB 32 required that GHGs emitted in California be reduced to 1990 levels by the year 2020. "Greenhouse gases" as defined under AB 32 include CO₂, CH₄, NO_X, hydrofluorocarbons, perfluorocarbons, and sulfur hexafluoride. Since AB 32 was enacted, a seventh chemical, nitrogen trifluoride, has also been added to the list of GHGs. CARB is the state agency charged with monitoring and regulating sources of GHGs. AB 32 states the following:

Global warming poses a serious threat to the economic well-being, public health, natural resources, and the environment of California. The potential adverse impacts of global warming include the exacerbation of air quality problems, a reduction in the quality and supply of water to the state from the Sierra snowpack, a rise in sea levels resulting in the displacement of thousands of coastal businesses and residences, damage to marine ecosystems and the natural environment, and an increase in the incidences of infectious diseases, asthma, and other human health-related problems.

CARB approved the 1990 GHG emissions level of 427 million metric tons of carbon dioxide equivalent (MMTCO₂e) on December 6, 2007 (CARB 2007). Therefore, to meet the state's target, emissions generated in California in 2020, were required to be equal to or less than 427 MMTCO₂e. In order to set a framework for the state to meet this target, CARB was tasked with creating a Scoping Plan (as described below). California announced in July 2018, that the state emitted 427 MMTCO₂e in 2016, and achieved AB 32 goals ahead of schedule.

Senate Bill 32

Senate Bill (SB) 32 was signed into law on September 8, 2016. SB 32 gives CARB the statutory responsibility to include the 2030 target previously contained in EO B-30-15 in the 2017 Scoping Plan Update. SB 32 states that "In adopting rules and regulations to achieve the maximum technologically feasible and cost-effective greenhouse gas emissions reductions authorized by this division, the state [air resources] board shall ensure that statewide greenhouse gas emissions are reduced to at least 40 percent below the statewide greenhouse gas emissions limit no later than December 31, 2030."



Draft Environmental Impact Report Environmental Impact Analysis

Assembly Bill 1279

AB 1279 was signed into law in 2022 and establishes the policy of the State to achieve carbon neutrality as soon as possible, but no later than 2045 and maintain net negative GHG emissions thereafter. AB 1279 would also ensure that by 2045 the statewide anthropogenic GHG emissions are reduced by at least 85 percent below 1990 levels. The bill would require CARB to ensure that an updated Scoping Plan identifies and recommends measures to achieve carbon neutrality, and to identify and implement policies and strategies that enable carbon dioxide removal and carbon capture, utilization, and storage technologies to complement AB 1279's emissions reduction requirements.

Climate Change Scoping Plan

In December 2008, CARB approved the AB 32 Scoping Plan outlining the state's strategy to achieve the 2020 GHG emissions limit. The Scoping Plan estimates a reduction of 174 MMTCO₂e (about 191 million U.S. tons) from the transportation, energy, agriculture, forestry, and high climate-change-potential sectors, and proposes a comprehensive set of actions designed to reduce overall GHG emissions in California, improve the environment, reduce dependence on oil, diversify California's energy sources, save energy, create new jobs, and enhance public health. The Scoping Plan was required to be updated every 5 years to evaluate the implementation of AB 32 policies to ensure that California remains on track to achieve its original 2020 GHG reduction goal.

The First Update to the Climate Change Scoping Plan was approved by the CARB on May 22, 2014. In 2016, the State Legislature passed SB 32, which codified a 2030 GHG emissions reduction target of 40 percent below 1990 levels. With SB 32, the State Legislature passed companion legislation AB 197, which provided additional direction for developing the Scoping Plan. On December 14, 2017, the CARB approved the Second Update to the Climate Change Scoping Plan, the 2017 Climate Change Scoping Plan: The Strategy for Achieving California's 2030 Greenhouse Gas Target (CARB 2017). The 2017 Scoping Plan identifies key sectors of the implementation strategy, which includes improvements in low carbon energy, industry, transportation sustainability, natural and working lands, waste management, and water. Through a combination of data synthesis and modeling, CARB determined that the target statewide 2030 emissions limit is 260 MMTCO₂e, and that further commitments will need to be made to achieve an additional reduction of 50 MMTCO₂e beyond current policies and programs. Key elements of the 2017 Update include a proposed 20 percent reduction in GHG emissions from refineries and an expansion of the Cap-and-Trade program to meet the aggressive 2030 GHG emissions goal.

CARB prepared the 2022 Scoping Plan for Achieving Carbon Neutrality, which builds upon the previous Scoping Plans in order to lay out a sector by sector blueprint for the state to achieve carbon neutrality by 2045, or earlier, while identifying a pathway to keep the state on track to meet SB 32. The 2022 Scoping Plan incorporates, coordinates, and leverages existing effort to reduce GHGs while identifying new clean technologies and energy. The actions and outcomes in the plan will achieve significant reductions in fossil fuel combustion, further reductions in short-lived climate pollutants, support for sustainable development, increased action on natural and working lands to reduce emissions and sequester carbon, and the capture and storage of carbon (CARB 2022a).



Assembly Bill 398

The Governor signed AB 398 on July 25, 2017, to extend the Cap-and-Trade Program to 2030. The legislation includes provisions to ensure that offsets used by sources are limited to 4 percent of their compliance obligation from 2021 to 2025 and 6 percent of their compliance obligation from 2026 through 2030. AB 398 also prevents air districts from adopting or implementing emission reduction rules from stationary sources that are also subject to the Cap-and-Trade Program (CARB 2017).

Senate Bill 375: The Sustainable Communities and Climate Protection Act of 2008

SB 375 was signed into law on September 30, 2008. According to SB 375, the transportation sector is the largest contributor of GHG emissions, which emits more than 40 percent of the total GHG emissions in California. SB 375 states, "Without improved land use and transportation policy, California will not be able to achieve the goals of AB 32." SB 375 does the following: (1) requires metropolitan planning organizations to include sustainable community strategies in their regional transportation plans for reducing GHG emissions, (2) aligns planning for transportation and housing, and (3) creates specified incentives for the implementation of the strategies.

CARB has prepared the Proposed Update to the SB 375 Greenhouse Gas Emission Reduction Targets that includes regional targets for each Metropolitan Planning Organization (MPO) for 2020 and 2035 to achieve significant additional GHG reductions from changed land use patterns and improved transportation. The Metropolitan Transportation Commissions and the Association of Bay Area Governments (MTC/ABAG), the MPO for the Bay Area, received a target of reduction transportation GHG emissions by 10 percent for 2020 and by 19 percent for 2035 (CARB 2022c).

Senate Bill 1368: Emission Performance Standards

In 2006, the State Legislature adopted SB 1368, which was subsequently signed into law by the governor. SB 1368 directs the California Public Utilities Commission to adopt a performance standard for GHG emissions for the future power purchases of California utilities. SB 1368 seeks to limit carbon emissions associated with electrical energy consumed in California by forbidding procurement arrangements for energy longer than 5 years from resources that exceed the emissions of a relatively clean, combined cycle natural gas power plant.

Because of the carbon content of its fuel source, a coal-fired plant cannot meet this standard because such plants emit roughly twice as much carbon as natural gas, combined cycle plants. Accordingly, the new law effectively prevents California's utilities from investing in, otherwise financially supporting, or purchasing power from new coal plants located in or out of the state. The California Public Utilities Commission adopted the regulations required by SB 1368 on August 29, 2007. The regulations implementing SB 1368 establish a standard for baseload generation owned by, or under long-term contract to publicly owned utilities, of 1,100 pounds of CO₂ per megawatt-hour (MWh).

Senate Bill 1078: Renewable Electricity Standards

On September 12, 2002, Governor Gray Davis signed SB 1078, requiring California to generate 20 percent of its electricity from renewable energy by 2017. SB 107 changed the due date to 2010 instead of 2017. On November 17, 2008, Governor Arnold Schwarzenegger signed EO S-14-08, which established



Draft Environmental Impact Report Environmental Impact Analysis

an RPS target for California requiring that all retail sellers of electricity serve 33 percent of their load with renewable energy by 2020. Governor Schwarzenegger signed EO S-21-09, which directed CARB to adopt a regulation by July 31, 2010, requiring the state's load serving entities to meet a 33 percent renewable energy target by 2020. CARB approved the Renewable Electricity Standard on September 23, 2010, by Resolution 10-23. In 2011, the State Legislature adopted this higher standard in SB X1-2. Renewable sources of electricity include wind, small hydropower, solar, geothermal, biomass, and biogas.

Senate Bill 350: Clean Energy and Pollution Reduction Act of 2015

The State Legislature approved and the governor then signed SB 350 on October 7, 2015, which reaffirms California's commitment to reducing its GHG emissions and addressing climate change. Key provisions include an increase in the RPS, higher energy efficiency requirements for buildings, initial strategies towards a regional electricity grid, and improved infrastructure for electric vehicle charging stations.

Senate Bill 100: California Renewables Portfolio Standard Program

The Governor approved SB 100 on September 10, 2018. The legislation revised the RPS goals to achieve the 50 percent renewable resources target by December 31, 2026, and to achieve a 60 percent target by December 31, 2030. The bill would require that retail sellers and local publicly owned electric utilities procure a minimum quantity of electricity products from eligible renewable energy resources so that the total kilowatt hours of those products sold to their retail end-use customers achieve 44 percent of retail sales by December 31, 2024; 52 percent by December 31, 2027; and 60 percent by December 31, 2030.

Executive Order S-3-05

On June 1, 2005, former California Governor Arnold Schwarzenegger announced EO S-3-05, which announced the following reduction targets for GHG emissions:

- By 2010, reduce GHG emissions to 2000 levels.
- By 2020, reduce GHG emissions to 1990 levels.
- By 2050, reduce GHG emissions to 80 percent below 1990 levels.

The 2050 reduction goal represents what some scientists believe is necessary to reach levels that would stabilize the climate. The 2020 goal was established to be a mid-term target. Because this is an EO, the goals are not legally enforceable for local governments or the private sector.

Executive Order B-30-15

On April 29, 2015, Governor Edmund G. Brown Jr. issued EO B-30-15 to establish a California GHG reduction target of 40 percent below 1990 levels by 2030. The Governor's EO aligns California's GHG reduction targets with those of leading international governments ahead of the United Nations Climate Change Conference in Paris in late 2015. The EO sets a new interim statewide GHG emission reduction target to reduce GHG emissions to 40 percent below 1990 levels by 2030 in order to ensure that



Draft Environmental Impact Report Environmental Impact Analysis

California meets its target of reducing GHG emissions to 80 percent below 1990 levels by 2050 and directs CARB to update the Climate Change Scoping Plan to express the 2030 target in terms of MMTCO₂e. The EO also requires the state's climate adaptation plan to be updated every 3 years and for the state to continue its climate change research program, among other provisions. As with EO S-3-05, this EO is not legally enforceable against local governments and the private sector. Legislation that would update AB 32 to provide post-2020 targets was signed by the Governor in 2016. SB 32 includes a 2030 mandate matching the requirements of the EO.

Executive Order S-01-07: Low Carbon Fuel Standard

The governor signed EO S 01-07 on January 18, 2007. The order mandates that a statewide goal shall be established to reduce the carbon intensity of California's transportation fuels by at least 10 percent by 2020. In particular, the EO established a Low Carbon Fuel Standard (LCFS) and directed the Secretary for Environmental Protection to coordinate the actions of the California Energy Commission, CARB, the University of California, and other agencies to develop and propose protocols for measuring the "life-cycle carbon intensity" of transportation fuels. This analysis supporting development of the protocols was included in the State Implementation Plan for alternative fuels (State Alternative Fuels Plan adopted by California Energy Commission on December 24, 2007) and was submitted to CARB for consideration as an "early action" item under AB 32. CARB adopted the Low Carbon Fuel Standard on April 23, 2009.

The LCFS was subject to legal challenge in 2011. Ultimately, CARB was required to bring a new LCFS regulation for consideration in February 2015. The proposed LCFS regulation was required to contain revisions to the 2010 LCFS as well as new provisions designed to foster investments in the production of the low-carbon fuels, offer additional flexibility to regulated parties, update critical technical information, simplify and streamline program operations, and enhance enforcement. The Office of Administrative Law approved the regulation on November 16, 2015. The regulation was last amended in 2018.

Executive Order S-13-08

EO S-13-08 states that "climate change in California during the next century is expected to shift precipitation patterns, accelerate sea level rise and increase temperatures, thereby posing a serious threat to California's economy, to the health and welfare of its population and to its natural resources." Pursuant to the requirements in the EO, the 2009 California Climate Adaptation Strategy was adopted, which is the "… first statewide, multi-sector, region-specific, and information-based climate change adaptation strategy in the United States." Objectives include analyzing risks of climate change in California, identifying and exploring strategies to adapt to climate change, and specifying a direction for future research.

Executive Order B-55-18

EO B-55-18 issued by Governor Brown on September 10, 2018, establishes a new statewide goal to achieve carbon neutrality as soon as possible, but no later than 2045, and to achieve and maintain net negative emissions thereafter. The EO directs CARB to work with relevant state agencies to develop a framework for implementation and accounting that tracks progress toward this goal.



Draft Environmental Impact Report Environmental Impact Analysis

California Energy Code

Compliance with the California Energy Code (Title 24, Part 6, of the CCR, California's Energy Efficiency Standards) and Title 20, Public Utilities and Energy, standards must occur for all new buildings constructed in California. These efficiency standards apply to new construction of both residential and nonresidential (i.e., maintenance buildings and pump station buildings associated with the Program) buildings, and they regulate energy consumed for heating, cooling, ventilation, water heating, and lighting. The building efficiency standards are enforced through the local building permit processes, and local government agencies may adopt and enforce energy standards for new buildings provided that these standards meet or exceed those provided in the Title 24 guidelines.

California Green Building Standards

CALGreen (Title 24, Part 11, of the CCR) is a statewide green building code that applies to the planning, design, operation, construction, use and occupancy of every newly constructed building in California. The code includes both mandatory and voluntary measures for residential and non-residential developments to increase energy efficiency, water efficiency and conservation, material conservation, and environmental quality.

Regional

ABAG Plan Bay Area 2050

ABAG and MTC adopted the Plan Bay Area 2050 in October 2021. Plan Bay Area 2050 is the region's regional transportation plan/sustainable communities strategy (RTP/SCS). Chapter 5 of the RTP/SCS addresses the potential transformation of the region in the coming decades from climate change and includes a series of strategies to create a more resilient environment. These strategies include expanding access to parks and open space, reducing climate emissions from vehicles, and reducing the risks from hazards.

Local

City of Newark General Plan

The City of Newark General Plan includes the following goals and policies related to GHGs that are applicable to the proposed project:

Goal CS-5: Greenhouse Gas Reduction. Reduce greenhouse gas emissions in Newark and make reduction of the City's carbon output a high priority.

- **Policy CS-5.2: Pedestrian and Bicycle Friendly Design.** Ensure that new development is planned and designed to facilitate walking and bicycling as well as driving. This can potentially reduce the number of vehicle trips and related greenhouse gas emissions.
- Policy CS-5.5: Consideration of Climate in Transportation Planning. Consider potential greenhouse gas emissions impacts when making changes to the transportation system. Give preference to solutions that reduce auto dependency and minimize emissions.



Draft Environmental Impact Report Environmental Impact Analysis

• **Policy CS-5.8: Planning for Sea Level Rise.** Require proposed development in low-lying areas to comply with applicable City of Newark standards for construction in flood hazard zones.

Goal CS-6: Green Building. Reduce the impacts of buildings and development on greenhouse gas levels and the environment in general.

- **Policy CS-6.2: Encouraging Greener Construction.** Encourage greener construction methods and greater use of recycled-content materials in new residential, commercial, and industrial construction projects.
- Policy CS-6.5: Minimizing Impervious Surface Coverage. Minimize impervious surface coverage and related stormwater runoff in new development areas by allowing narrower roads and shared driveways, and by encouraging the use of pervious materials on driveways and parking areas. Other means of reducing urban runoff, such as rain barrels and bioswales, also should be encouraged.

City of Newark Climate Action Plan

The City's Climate Action Plan was adopted by City Council in January 2010 to identify and evaluate feasible and effective policies to reduce GHG emissions in the public and private sectors. The CAP intends to achieve the following goals: (1) create an inventory of emissions from city government operations and community-wide activities; (2) present the inventory as a baseline against which to measure progress towards reducing GHG emissions; (3) develop a set of emission reduction goals for municipal operations; (4) present actions that the citizens and businesses of Newark can implement; and (5) present long-term planning efforts to layout future development (City of Newark 2010).

City of Newark Municipal Code, Chapter 15.44

The City's Municipal Code, Chapter 15.44, Green Building and Construction and Demolition Debris Recycling ordinance aims to minimize or avoid a variety of adverse impacts by regulating the design, construction, and operation of buildings and landscape within the City.

City of Newark Pedestrian and Bicycle Master Plan

The City adopted the Pedestrian and Bicycle Master Plan in February 2017. The Plan provides a blueprint for future pedestrian and bicycle improvement projects and programs in the City to fill the gap in existing networks as well as upgrade and repair existing facilities.

3.8.3 Environmental Impacts

This section analyzes the project's potential to result in significant greenhouse gas impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

GHG emissions that would result from implementation of the project and from the existing use of the project site were calculated using the CalEEMod, Version 2020.4.0. The proposed project would result in



Draft Environmental Impact Report Environmental Impact Analysis

both short- and long-term emissions of GHGs. Construction emissions would be generated from the exhaust of construction equipment, the exhaust of construction hauling trips, vendor truck trips, and worker commuter trips. In addition, construction GHG emissions are generated by the use of stationary equipment (e.g., generators and air compressors) and indirectly by the electricity used to power off-road equipment and to supply, treat, and distribute water that is used in the construction process.

The CalEEMod input and assumptions for construction modeling emissions are described in Section 3.3, Air Quality.

Operational sources of GHG emissions in CalEEMod include area, energy, mobile, water use, and solid waste. Operational project input and design features incorporated into CalEEMod for the project and existing use include:

- Area area sources include GHG emissions from landscaping equipment, the use of consumer products, and gas fireplaces. Emissions associated with area sources were estimated using the CalEEMod default values for the project. Area sources in CalEEMod also include emissions from wood burning stoves and fireplaces. However, in accordance with the BAAQMD Regulation 6, Rule 3 *Wood-Burning Devices*, permanently installed wood-burning devices are not permitted in new development and the project would not include wood-burning stoves or wood-burning fireplaces (BAAQMD 2019). The CalEEMod defaults for area sources were used in the project and existing use modeling.
- Energy The proposed project would use electricity for lighting, heating, and cooling. Some electricity generation entails the combustion of fossil fuels, including natural gas and coal, which results in GHG emissions at the power plant locations. Power plant GHG emissions may occur outside of the region or state. Electricity would be supplied by PG&E. Energy source emissions for the existing use were estimated using CalEEMod defaults. Energy source emissions for the project were estimated assuming implementation of energy-reducing project design features to comply with the 2019 Title 24 standards which include a requirement for on-site generation of electricity through photovoltaic (solar) panels. In accordance with 2019 Title 24, the project's estimated 284,000 square-feet of residential space (assuming 81,200 square feet of unconditioned garage space) in Alameda County climate zone 3 would require solar panels producing a minimum of 405.8 kW. Using a Capacity Factor of 20 percent to account for climate, amount of sunlight available per day, the pitch and orientation of the roof, and the efficiency of the electrical transmission, the project solar panel would produce approximately 711,028 kW-hours per year.
- **Mobile** Operational GHG emissions from mobile sources are associated with project-related vehicle trip generation and trip length, as described in the AQ section, above.
- Solid Waste Solid waste generated by the project would also contribute to GHG emissions. Treatment and disposal of solid waste produces emissions of methane. Modeling was conducted using CalEEMod default solid waste generation rates and GHG factors for Alameda County. For project modeling, a 75 percent reduction applied to account for residential AB 341 and local waste diversion mandates not accounted for in the model defaults.



Draft Environmental Impact Report Environmental Impact Analysis

> • Water Sources – Water-related GHG emissions are from the energy used and process emissions for the conveyance and treatment of water. The CEC's 2006 Refining Estimates of Water-Related Energy Use in California defines average energy values for water use. These values are used in CalEEMod to establish default water related emission factors. Modeling was conducted using these defaults. For the project modeling, a 20 percent reduction in potable water use and wastewater generation was applied in accordance with 2019 CALGreen standards.

Thresholds of Significance

The BAAQMD does not have an adopted threshold of significance for construction-related GHG emissions. However, the BAAQMD recommends quantification and disclosure of GHG construction emissions. Determining the significance of these construction-generated GHG emission impacts is recommended to be made in relation to meeting AB 32 GHG reduction goals, which requires the state to meet 1990 levels of GHG emissions by 2020. As discussed in the regulatory setting, California announced in July 2018 that the state emitted 427 MMTCO₂e in 2016 and achieved AB 32 goals.

Since GHG emissions are cumulative and construction emission are temporary and short term, it is common practice to amortize the total construction GHG emissions over 30 years to create an annual emissions rate that is combined with the operational GHG emissions for determining significance.

The BAAQMD's 2017 *CEQA Guidelines* provide numeric thresholds for GHG emissions during project operation for projects to demonstrate compliance with AB 32. A proposed land use development project would not have a significant GHG impact, if operation of the project would meet one of the following thresholds (BAAQMD 2017):

- Compliance with a qualified GHG Reduction Strategy;
- Annual emissions less than 1,100 metric tons per year (MT/yr) of CO2e; or
- 4.6 metric tons of CO2e (MTCO2e) per service population per year (MTCO2e/SP/yr).

In April 2022, the BAAQMD adopted updated GHG thresholds that would require a project to either comply with a qualified GHG Reduction Strategy or incorporate project design features and VMT reduction targets to demonstrate project significance. Since the Project's Notice of Preparation was released prior to BAAQMD adoption of the 2022 thresholds, the GHG thresholds disclosed in the BAAQMD's 2017 CEQA Guidelines are still applicable to the project. However, the Project would still be required to demonstrate compliance with SB 32 and the 2030 statewide GHG reductions. SB 32 requires the state to reduce emissions 40 percent below 1990 levels by 2030. According to the Association of Environmental Professionals (AEP) in *Final White Paper Beyond 2020 and Newhall: A Field Guide to New CEQA Greenhouse Gas Thresholds and Climate Action Plan Targets for California*, an efficiency threshold interpolated between a 2020 and a 2030 metric is appropriate to determine project significance (AEP 2016). As such, project with an operation date of 2020 would have a 0 percent reduction from BAAQMD's GHG thresholds are grequired by SB 32. Projects with an operation date between 2020 and 2030 would reduce the BAAQMD threshold based on a straight-line reduction calculation. Therefore, to determine significance and demonstrate consistent with SB 32, BAAQMD's



Draft Environmental Impact Report Environmental Impact Analysis

quantitative, per service population thresholds were reduced to 3.06 MT CO2e/year to account for the project's first year of operation in 2028.

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's GHG emissions impacts are significant. Would the proposed project:

- Generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?
- Conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases?

Project Impact Analysis and Mitigation Measures

Generation of Greenhouse Gases

Impact GHG-1 The proposed project would not generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?

Impact Analysis

Constructions Emission Inventory

Construction emissions would be generated from the exhaust of equipment and the exhaust of construction equipment, truck hauling trips, and material delivery trips and worker commuter trips. Detailed construction assumptions are provided in Section 3.3, Air Quality. The BAAQMD does not presently provide a construction-related GHG generation threshold but recommends that construction-generated GHGs be quantified and disclosed. MTCO₂e emissions during construction of the project are presented in Table 3.8-1.

Table 3.8-1: Construction Greenhouse Gas Emissions

Year	Emissions (MT CO ₂ e)
2023	328.1
2024	1,837.0
2025	605.2
2026	113.3
TOTAL ¹	2,883.6
Amortized Construction Emissions (30 years)	96.2

Source: CalEEMod.

¹ Totals may not sum due to rounding.

Operational Emission Inventory

Operational or long-term emissions occur over the life of the project. The operational emissions for the proposed project are shown in Table 3.8-2. Sources for operational emissions include motor vehicles, indirect electricity, water transport, and waste.



Because the proposed project would replace the existing automotive recycling and parts business on the project site (the Pick-n-Pull), the proposed project's operational emissions would be offset by emissions from existing operations, which would be replaced. The existing automotive recycling and parts business was modeled as 1,500 square feet of general office building; 3,000 square feet of general light industrial building; and 13,000 square feet of unrefrigerated warehouse-no rail. The model of existing land uses relies on traffic trip estimates from Fehr & Peers' Transportation Impact Analysis as well as water and waste reductions assumptions that are consistent with state reduction requirements. Other modeling inputs for the existing land uses were based on CalEEMod default assumptions. Detailed modeling details are provided in Section 3.3, Air Quality. The net GHG emissions are presented in Table 3.3-8.

During operation, the proposed project would result in net GHG emissions of approximately 2,041.4 MTCO₂e/year. The proposed project is estimated to serve approximately 674 residents. As shown in Table 3-8.2 the project would result in a generation of 3.03 MTCO₂e per service person per year. Estimated operational emissions would not exceed the BAAQMD adjusted significance thresholds; therefore, impacts would be less than significant.

Source	Annual Emissions (MT CO ₂ e)
Area	19.9
Energy	512.4
Mobile	1,997.9
Waste	108.2
Water	23.5
Operational Subtotal ¹	2,661.8
Amortized Construction Emissions (30 years)	96.2
Total Project Emissions	2,758.0
Existing Use Emissions	(716.6)
Net Project Emissions ¹	2,041.4
Efficiency (Net Emissions/Service Population; MT CO ₂ e/service population/year) ²	3.03
BAAQMD 2028 Adjusted Threshold (MT CO ₂ e/service population/year)	3.06
Exceed Threshold?	No

Table 3.8-2: Operational Greenhouse Gas Emissions

Source: CalEEMod.

¹ Totals may not sum due to rounding.

² Based on an estimated project population of 674.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Conflict with an Applicable Plan, Policy, or Regulation

Impact GHG-2 The proposed project would not conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases?

Impact Analysis

The following analysis assesses the proposed project's consistency with local and regional adopted plans to reduce GHG emissions. The City adopted their CAP Initial Framework in 2010, which was developed to present measures to reduce local GHG emissions; meet state, regional, and local reduction targets; and streamline future environmental review. Lastly, the State of California has developed Climate Change Scoping Plans, which are required to be updated every five years. The 2017 Scoping Plan outlines the strategy for achieving California's 2030 GHG target of 40 percent emissions reductions below 1990 levels. In December 2022, CARB adopted the 2022 Scoping Plan which builds upon the 2017 Scoping Plan to outline a blueprint for the state to meet SB 32 and AB 1279 reduction goals. The following provides a project-specific consistency analysis with each of these local, regional, and statewide plans:

City of Newark Climate Action Plan

The City's CAP includes a series of residential community action items, action items that are applicable to the proposed project are presented in Table 3.8-4.

Newark Climate Action Plan		Project Consistency
Residential Community Action Item 4.2	Encourage use of Alternative Fuel Vehicles.	Consistent. The proposed project would be consistent with CALGreen which requires at least 10 percent of parking for new projects to be equipped with EV chargers.
Residential Community Action Item 4.3	Energy Conservation	Consistent. The proposed project would be consistent with 2022 CALGreen building standards that include a series of energy conservation measures – such as installing ENERGY STAR appliances.
Residential Community Action Item 4.5	Increase Residential Recycling and Composting	Consistent. The proposed project would comply with all waste diversion measures for construction and would be serviced by a waste provider that would be required to meet city and state diversion goals.
Residential Community Action Item 4.6	Water Conservation	Consistent. The proposed project would be consistent with 2022 CALGreen which would require water conservation measures including low flow water fixtures. Moreover, all proposed landscaping would consist of low water use plants and would meet the requirement of the Model Water Efficient Landscape Ordinance.

Table 3.8-3: Newark Climate Action Plan Consistency Analysis

Draft Environmental Impact Report Environmental Impact Analysis

Newark Cl	imate Action Plan	Project Consistency
Residential Community Action Item 4.7	StopWaste.org Green Packages	Not Applicable. This action item is aimed at the city to create a financing option for the city to encourage developers to install energy efficient measures on homes. While the measure is not applicable, the Project would be consistent with 2022 CALGreen measures that would require energy efficient measures.
Residential Community Action Item 4.8	Multi-family Building Owners Assistance Newark	Not Applicable. The action item is aimed at the city to create a financing option to encourage developers to participate in the Green Points Rated or LEED Programs.

Notes:

CALGreen = California Green Building Standards Source: City of Newark, 2016.

California Climate Change Scoping Plan

CARB issued the Final 2017 Scoping Plan Update in November 2017 and establishes emissions reduction strategies necessary to meet SB 32's 2030 reduction goals. In 2022, CARB approved their 2022 Scoping Plan which expands on the Final 2017 Scoping Plan to assess progress in meeting SB 32 and reach AB 1279. Consistency with the 2017 and 2022 Scoping Plans are included in Tables 3.8-4 and 3.8-5, respectively, below.

Table 3.8-4: SB 32 Scoping Plan Consistency Analysis

Measure Name	Measure Description	Consistency Determination
SB 350 50% Renewable Mandate.	Utilities subject to the legislation will be required to increase their renewable energy mix from 33% in 2020 to 50% in 2030.	Consistent. The proposed project will purchase electricity from a utility subject to the SB 350 Renewable Mandate. In addition, the proposed project would be required to adhere to the latest Title 24 and CALGreen building standards.
Low Carbon Fuel Standard	This measure requires fuel providers to meet an 18% reduction in carbon content by 2030.	Consistent. Vehicles accessing the proposed project site would use fuel containing lower carbon content as the fuel standard is implemented.
Mobile Source Strategy (Cleaner Technology and Fuels Scenario)	Vehicle manufacturers will be required to meet existing regulations mandated by the LEV III and Heavy-Duty Vehicle programs. The strategy includes a goal of having 4.2 million ZEVs on the road by 2030 and increasing numbers of ZEV trucks and buses.	Consistent. Future residents can be expected to purchase increasing numbers of more fuel efficient and zero emission cars and trucks each year.



Draft Environmental Impact Report Environmental Impact Analysis

Measure Name	Measure Description	Consistency Determination
Short-Lived Climate Pollutant (SLCP) Reduction Strategy	The strategy requires the reduction of SLCPs by 40% from 2013 levels by 2030 and the reduction of black carbon by 50% from 2013 levels by 2030.	Consistent. SLCPs include HFCs, black carbon, and methane. Black carbon is created from the burning of fuels such as coal, diesel, and biomass. The proposed office and residential buildings would be 100 percent electric and generate very few diesel truck trips and would not contribute to black carbon pollution. HFCs are a group of industrial chemicals primarily used for air conditioning and refrigeration. CARB has already banned a series of HFCs including those used for residential refrigeration. The proposed project would comply with all applicable regulations.
SB 375 Sustainable Communities Strategies	Requires Regional Transportation Plans to include a sustainable communities' strategy for reduction of per capita vehicle miles traveled.	Not Applicable. This measure is aimed at Metropolitan Planning Organizations who prepare Regional Transportation Plans to include a sustainable communities' strategy to reduce vehicle miles traveled. The project would not interfere with implementation of this goal.
Post-2020 Cap-and- Trade Program	The Post 2020 Cap-and-Trade Program continues the existing program for another 10 years. The Cap-and-Trade Program applies to large industrial sources such as power plants, refineries, and cement manufacturers.	Consistent. The Post-2020 Cap-and- Trade Program indirectly affects people who use the products and services produced by the regulated industrial sources when increased cost of products or services (such as electricity and fuel) are transferred to the consumers. The Cap- and-Trade Program covers the GHG emissions associated with electricity consumed in California, whether generated in-state or imported. Accordingly, GHG emissions associated with CEQA projects' electricity usage are covered by the Cap- and-Trade Program. The Cap-and-Trade Program also covers fuel suppliers (natural gas and propane fuel providers and transportation fuel providers) to address emissions from such fuels and from combustion of other fossil fuels not directly covered at large sources in the program's first compliance period.

Source: CARB 2017

Table 3.8-5: Project Consistency with 2022 Scoping Plan Greenhouse Gas Reduction Strategies

Measure	Consistency Determination
Deploy ZEVs and reduce driving demand	Consistent. The proposed project would be consistent with CALGreen which requires at least 10 percent of parking for new projects to be equipped with EV chargers.

 \bigcirc

Draft Environmental Impact Report Environmental Impact Analysis

Measure	Consistency Determination
Coordinate supply of liquid fossil fuels with declining CA fuel demand	Not Applicable. This measure is aimed at petroleum refineries and fossil fuel extraction operations. The Project would not interfere with implementation of this measure.
Generate clean electricity	Consistent. This measure is aimed at the electric sector to decarbonize. The Project would purchase electricity from utility providers that are expanding GHG free electricity consistent with SB 350 Renewable Mandate.
Decarbonize Buildings	Consistent. The Project would comply with the latest California Green Building Standards that require energy and water efficient project design features that would reduce GHG emissions. In addition, the proposed Project would be 100 percent electric.
Decarbonize Industrial Energy Supply	Not Applicable. The Project would not include any industrial land uses.
Reduce non-combustion emissions (Methane)	Not Applicable. The Project would not include any land uses that generate significant levels of methane such as landfills or dairy farms.
Reduce non-combustion emissions (HFCs)	Consistent. HFCs are a group of industrial chemicals primarily used for air conditioning and refrigeration. CARB has already banned a series of HFCs including those used for residential refrigeration. The proposed project would comply with all applicable regulations.
Compensate for remaining emissions	Not Applicable. This measures is aimed at the state government to reduce statewide emissions to meet AB 1279 goals.

Source: CARB 2022a

In addition to the Plan level consistency analysis presented in Tables 3.8-3 through 3.8-5, the proposed project would be subject to Title 24 energy efficiency standards. Energy-efficient buildings require less electricity; therefore, increased energy efficiency reduces fossil fuel consumption and decreases GHG emissions. The proposed project would comply with the CALGreen, which includes requirements to increase recycling, reduce waste, reduce water use, increase bicycle use, and other measures that would reduce GHG emissions. Motor vehicle emissions associated with the proposed project would be reduced through compliance with State regulations on fuel efficiency and fuel carbon content. The proposed project would not conflict with the City's CAP or the statewide scoping plan adopted by the State of California to reduce GHG emissions therefore, impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance After Mitigation

Less Than Significant Impact.

This page intentionally left blank.

3.9 HAZARDS AND HAZARDOUS MATERIALS

This section describes the environmental and regulatory setting for hazards and hazardous materials. It also describes existing conditions and potential impacts related to hazards and hazardous materials that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.9.1 Environmental Setting

Hazardous Materials

Hazardous materials, as defined by CCR, are substances with certain physical properties that could pose a substantial present or future hazard to human health or the environment when improperly handled, disposed of, or otherwise managed. Hazardous materials are grouped into the following four categories, based on their properties:

- Toxic: Causes human health effects
- Ignitable: Has the ability to burn
- Corrosive: Causes severe burns or damage to materials
- Reactive: Causes explosions or generates toxic gases

Hazardous waste is any hazardous material that is discarded, abandoned, or slated to be recycled. The criteria that define a material as hazardous also define a waste as hazardous. If improperly handled, hazardous materials and hazardous waste can result in public health hazards if released into the soil or groundwater or through airborne releases in vapors, fumes, or dust.

California Government Code Section 65962.5 requires the Cal EPA to compile, maintain, and update specified lists of hazardous material release sites. The required lists of hazardous material release sites are commonly referred to as the "Cortese List," which are contained on internet websites, including the online EnviroStor database from the Department of Toxic Substances Control (DTSC) and the online GeoTracker database from the SWRCB. These two databases include hazardous material release sites along with other categories of sites or facilities specific to each agency's jurisdiction. A search of EnviroStor and GeoTracker databases in November 2021 revealed an open case at the project site on the GeoTracker database (DTSC 2021 SWRCB 2021). GeoTracker has the Pick-n-Pull Auto Dismantler listed as a "Cleanup Program Site." "Cleanup Program Sites" includes all "non-federally owned" sites that are regulated under the SWRCB's Site Cleanup Program. The site has been used as an automobile wrecking yard since the 1960s and is currently owned and operated by Pick-n-Pull Auto Dismantlers. The cleanup site's status is "OPEN - SITE ASSESSMENT" as of January 15, 2020 on the GeoTracker database. Cases with this status are those where site characterization, investigation, risk evaluation, and/or site conceptual model development are occurring at the site. A Corrective Action Plan and Remedial Excavation Work Plan was approved by the Alameda County Water District in March 2021. The Corrective Action Plan and Remedial Excavation Work Plan proposed to remediate the impacted soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater containment through in-situ remediation, and surface



Draft Environmental Impact Report Environmental Impact Analysis

water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. Additionally, the site directly adjacent to the southwest of the project site (Tolbertson Property) is listed in the GeoTracker database as a Cleanup Program Site. The Tolbertson Property case has a status of "Open – Verification Monitoring". Sites with status of "Open – Verification Monitoring" are sites where remediation phases are essentially complete, a monitoring/sampling program is occurring to confirm successful completion of cleanup at the site, and no active remediation or additional remediation is considered necessary. Due to the case status of the Tolbertson Property, this case is not expected to pose any risks to the proposed project.

Phase I and Phase II Environmental Site Assessment

A Phase I ESA and Phase II ESA were conducted by Haley and Aldrich, Inc for the proposed project site in January 2019 (Appendix F). The report identified that the project site was used as an automobile wrecking yard since the 1960s and is currently owned by Pick-n-Pull Auto Dismantlers who have operated an automobile scrap yard since 1996. The Phase I ESA identified two RECs at the project site. RECs are defined as "the presence or likely presence of any hazardous substances or petroleum products in, on, or at a property: (1) due to any release to the environment; (2) under conditions indicative of a release to the environment; or (3) under conditions that pose a material threat of a future release to the environment" (Haley and Aldrich 2019). REC #1 pertains to the automobile wrecking operations at the site. During approximately 50 years of automobile wrecking operations, significant guantities of hazardous materials have been handled and stored on the site, with documented spills and visibly stained soil. REC #2 pertains to historical agricultural operation that took place at the site prior to development as an automobile wrecking yard. Sites associated with historical agricultural uses commonly contain residual agricultural chemicals. Additionally, the Phase I ESA identified that de minimis staining was observed on paved and gravel surfaces at the project site. De minimis conditions are those conditions which "do not present a threat to human health or the environment and that generally would not be the subject of an enforcement action if brought to the attention of appropriate government agencies" (Haley and Aldrich 2019). Conditions determined to be *de minimis* are not RECs.

Due to potential environmental issues identified during the Phase I ESA process, a sampling program was conducted to assess soil and groundwater conditions at the project site and REC #1 and REC #2 identified above were investigated. Soil at the subject site was assessed at 23 locations within the auto wrecking yard and at seven locations in the undeveloped northern parcel. Soil samples were analyzed for California Title 22 metals (metals), VOCs, PAHs, organochlorine pesticides (OCPs), and total petroleum hydrocarbons as gasoline (TPHg), diesel (TPHd), and motor oil (TPHmo). Grab groundwater samples were collected from 12 locations and analyzed for VOCs and TPHg/d/mo (Haley and Aldrich 2019). The Phase II ESA results identified generally low levels of metals, VOCs, and PAHs, OCPs, TPHg, and TPHmo in shallow soil at the site. Metals and PAHs were detected in the soil in concentrations constant with background levels, with the exception of lead in two locations and PAHs in one location. No OCPs, VOCs, TPHg, or TPHmo were detected in soil above their respective residential direct exposure levels.

Environmental Screening Levels (ESLs) are used to determine the concentration of contaminants present in different environmental domains, mainly encompassing soil, groundwater resources, water vapors, and ambient air at the specific site, which may impact the exposed human population. ESLs provide conservative screening levels for chemicals found at sites with contaminated soil and groundwater. Tier 1 ESLs are used to determine the threshold of significance for concentrations of contaminants found at the



Draft Environmental Impact Report Environmental Impact Analysis

site. If a contaminant is found at concentrations exceeding its respective Tier 1 ESL, it may result in a significant impact to future residents if remediation for the contaminant is not completed. Lead was detected at concentrations exceeding its Tier 1 ESL in three samples collected between 0.5 and 2.5 feet bgs inside of the automobile wrecking yard.TPHd was detected at concentrations exceeding its respective Tier 1 ESLs in shallow soil in 16 locations across the automobile wrecking yard, generally at 0.5 and 1.5 feet bgs. Soil on the subject site can be remediated to residential levels through removal of shallow soil in select portions of the auto wrecking yard.

TPHg was detected above its Tier 1 ESL in one groundwater sample and TPHd was detected above its Tier 1 ESL in seven groundwater samples. Most VOCs were detected in groundwater samples at levels below their respective Tier 1 ESLs, with a few VOCs, notably benzene and ethylbenzene, exceeding their respective Tier 1 ESLs. Benzene and ethylbenzene were detected in groundwater at one location within the auto wrecking yard at concentrations exceeding their Tier 1 ESLs, which are based on potential drinking water concerns. In addition, the detected benzene and ethylbenzene concentrations at that location exceed the ESL to assess potential vapor intrusion concerns from groundwater.

On-site remediation would be required at the project site prior to start of construction activities to ensure all hazardous materials detected in soil and groundwater samples at the project site are below their respective Tier 1 ESLs. Trace levels of hazardous materials detected in soil and groundwater at the site may be present after on-site remediation activities take place. However, trace levels would not pose a substantial risk to future residents.

Schools

There are no existing schools located within 0.25 miles of the project site. The nearest schools are the Ohlone College Newark Center for Health Sciences and Technology, located approximately 0.5 mile northeast of the project site, and Newark Memorial High School, located approximately 0.66 mile northeast of the project site.

Airports

There are no public or private airports located within the City or within two miles of the City limits (City of Newark 2013b). The two nearest airports are located across the bay from Newark. Palo Alto Airport is located in the City of Palo Alto, approximately five miles southwest of Newark, and Moffett Federal Airfield is located in the City of Mountain View, approximately five miles south of Newark. The City is located outside of both airports influence areas (City of Newark 2013b).

Wildfire

According to the California Department of Forestry and Fire Protection (CAL FIRE), the City is not located in a local or state fire hazard severity zone (CALFIRE 2007, 2008).



3.9.2 Regulatory Setting

Federal

United States Environmental Protection Agency

The USEPA was established in 1970 to consolidate in one agency a variety of federal research, monitoring, standard-setting, and enforcement activities to ensure environmental protection. The USEPA's mission is to protect human health and to safeguard the natural environment—air, water, and land—upon which life depends. The USEPA works to develop and enforce regulations and implement environmental laws enacted by Congress, is responsible for researching and setting national standards for a variety of environmental programs, and delegates to states and tribes the responsibility for using permits and for monitoring and enforcing compliance. Where national standards are not met, the USEPA can issue sanctions and take other steps to assist the states and tribes to reach the desired levels of environmental quality. Laws and regulations established by the USEPA are enforced in Alameda County by the Cal EPA.

Resource Conservation and Recovery Act

The Resource Conservation and Recovery Act (RCRA) set up the federal regulatory program for hazardous substances and gives the USEPA the authority to regulate the generation, transport, treatment, and disposal of hazardous substances in a "cradle to grave" system. Under RCRA, the USEPA regulates the generation, transportation, treatment, storage, and disposal of hazardous substances. This regulatory system includes tracking all generators of hazardous waste.

1984 Hazardous and Solid Waste Amendment Act

RCRA was amended by the 1984 Hazardous and Solid Waste Amendment Act, which prohibited the use of certain techniques for the disposal of certain hazardous wastes. The Emergency Planning and Community Right-to-Know Act of 1986 imposes safety requirements to protect local communities in the event of accidental release of hazardous substances. The requirements provide measures so that the risks from interaction with hazardous materials, such as handling, storage, and disposal, are mitigated or prevented. This law protects human health and the environment if the unintended release of hazardous materials was to occur. The USEPA has delegated fulfillment of many of RCRA's requirements to the DTSC.

State

Hazardous Waste Control Act

The Hazardous Waste Control Act created the state hazardous waste management program. It is similar to, but more stringent than, the federal RCRA program. The act is implemented by regulations contained in CCR Title 26, which describes the following required aspects for the proper management of hazardous waste: identification and classification; generation and transportation; design and permitting of recycling treatment, storage and disposal facilities; operation of facilities and staff training; and closure of facilities and liability requirements.



Draft Environmental Impact Report Environmental Impact Analysis

These regulations list more than 800 materials that may be hazardous, and establish criteria for identifying, packaging, and disposing of such waste. Under the Hazardous Waste Control Act and Title 26, the generator of hazardous waste must complete a manifest that accompanies the waste from generator to transporter to the ultimate disposal location. Copies of the manifest must be filed with DTSC.

California Environmental Protection Agency and Department of Toxic Substances Control

The Cal EPA is responsible for creating and enforcing environmental regulations within California. Within Cal EPA is DTSC, which was formed under the Hazardous Waste Control Act. DTSC is responsible for regulating hazardous waste, remediating existing contamination, and identifying ways to reduce production of hazardous wastes. DTSC can delegate enforcement responsibilities to local jurisdictions.

State Water Resources Control Board

The San Francisco RWQCB is authorized by the SWRCB to enforce the provisions of the Porter-Cologne Water Quality Control Act of 1969. This act gives the San Francisco RWQCB authority to require groundwater investigations when the quality of groundwater or surface waters of the State is threatened and to require remediation actions, if necessary.

Unified Program

The unified hazardous waste and hazardous materials management regulatory program (Unified Program) is a unified hazardous materials management program that was established by California's Secretary for Environmental Protection following Senate Bill 1082 (1993). The Unified Program consolidates, coordinates, and makes consistent the administrative requirements, permits, inspections, and enforcement activities of the following programs:

- Hazardous Materials Release Response Plans and Inventories
- California Accidental Release Prevention Program
- Underground Storage Tank Program
- Above Ground Petroleum Storage Act Program
- Hazardous Waste Generator and On-site Hazardous Waste Treatment Programs
- California Uniform Fire Code: Hazardous Material Management Plans and Hazardous Material Inventory Statements

These six environmental programs are implemented at the local government level by Certified Unified Program Agencies (CUPAs). CUPAs provide a central permitting and regulatory agency for permits, reporting, and compliance enforcement. PRC Section 21151.4 sets special requirements for EIRs and negative declarations for projects that involve the construction or alteration of a facility within 0.25 mile of a school that creates the following conditions:

• The project might reasonably be anticipated to emit hazardous air emissions;



Draft Environmental Impact Report Environmental Impact Analysis

- The project would handle an extremely hazardous substance or a mixture containing extremely hazardous substances in a quantity equal to or greater than the state threshold quantity specified in Section 25532(j) of the HSC; or
- The project may pose a health or safety hazard to persons who would attend or would be employed at the school.

As part of the CEQA process, the lead agency preparing the EIR must consult with the appropriate school district regarding the potential impact of the project on the school, and the school district must be notified about the project in writing at least 30 days before the proposed certification of the EIR (PRC Section 21151.4; 14 CCR Section 15186[b]).

Cortese List Government Code Section 65962

Government Code Section 65962 was enacted in 1985 and was amended in 1992. It is used as a planning tool to comply with CEQA and requires information about locations of hazardous materials release sites. It states that through the combined efforts of DTSC, the Department of Health Services, the SWRCB, and local enforcement agencies, a list of potentially hazardous areas and sites will be compiled and remain up to date (at a minimum, updated annually). The list is consolidated by the Secretary for Environmental Protection and is distributed to each city and county in which sites on the list are located. The list can be found on DTSC's EnviroStor database, which includes information from the SWRCB's GeoTracker database.

California Department of Transportation

Caltrans manages interregional transportation, including the management and construction of the California highway system. In addition, Caltrans is responsible for the permitting and regulation of state roadways and requires that permits be obtained for transportation of oversized loads and transportation of certain materials, such as hazardous materials, and for construction-related traffic disturbance.

California Public Resources Code

PRC Section 21151.4 is another key state law pertaining to hazardous materials, and is presented verbatim below:

- a) An environmental impact report shall not be certified or a negative declaration shall not be approved for any project involving the construction or alteration of a facility within one-fourth of a mile of a school that might reasonably be anticipated to emit hazardous air emissions, or that would handle an extremely hazardous substance or a mixture containing extremely hazardous substances in a quantity equal to or greater than the state threshold quantity specified pursuant to subdivision (j) of Section 25532 of the Health and Safety Code, that may pose a health or safety hazard to persons who would attend or would be employed at the school, unless both of the following occur:
 - 1) The lead agency preparing the environmental impact report or negative declaration has consulted with the school district having jurisdiction regarding the potential impact of the Project on the school.



- 2) The school district has been given written notification of the Project not less than 30 days prior to the proposed certification of the environmental impact report or approval of the negative declaration.
- b) As used in this section, the following definitions apply:
 - 1) "Extremely hazardous substance" means an extremely hazardous substance as defined pursuant to paragraph (2) of subdivision (g) of Section 25532 of the Health and Safety Code.
 - 2) "Hazardous air emissions" means emissions into the ambient air of air contaminants that have been identified as a toxic air contaminant by the State Air Resources Board or by the air pollution control officer for the jurisdiction in which the Project is located. As determined by the air pollution control officer, hazardous air emissions also mean emissions into the ambient air of a substance identified in subdivisions (a) to (f), inclusive, of Section 44321 of the Health and Safety Code. [Amended by Stats. 2008, Ch. 148, Sec. 1. Effective January 1, 2009]

Division of Occupational Safety and Health

The California Division of Occupational Safety and Health Administration (CalOSHA) is responsible for enforcing workplace safety regulations and requirements in California, including hazardous materials requirements recorded under CCR Title 8. These regulations include requirements for safety training, availability of safety equipment, accident and illness prevention programs, warnings about hazardous substance exposure (such as asbestos), and preparation of emergency action and fire prevention plans.

CalOSHA also enforces hazard-communication program regulations that contain training and information requirements. Such requirements include procedures for identifying and labeling hazardous substances, communicating information about hazardous substances and their handling, and preparing health and safety plans to protect workers and employees at hazardous waste sites. Under the hazard-communication program, employers must make Safety Data Sheets available to employees and document employee information and training programs.

California Emergency Services Act

The California Emergency Services Act provides the basic authority for conducting emergency operations following a proclamation of emergency by the governor and/or appropriate local authorities. Local government and district emergency plans are considered to be extensions of the California Emergency Plan, established in accordance with the Emergency Services Act.

The California Emergency Management Agency (CAL EMA) is the state agency responsible for establishing emergency response and spill notification plans related to hazardous materials accidents. CAL EMA regulates businesses by requiring specific businesses to prepare an inventory of hazardous materials (CCR Title 19). CAL EMA is also the lead state agency for emergency management and is responsible for coordinating the state-level response to emergencies and disasters.



California Department of Forestry and Fire Protection

CAL FIRE has mapped fire threat potential throughout California. CAL FIRE ranks fire threat based on the availability of fuel and the likelihood of an area burning (based on topography, fire history, and climate). The rankings include no fire threat, moderate, high, and very high fire threat. Additionally, CAL FIRE produced a *2010 Strategic Fire Plan for California*, which contains goals, objectives, and policies to prepare for and mitigate for the effects of fire on California's natural and built environment.

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project (City of Newark 2013a):

Goal EH-4: Hazardous Materials. Protect Newark residents and workers from the potential adverse effects on hazardous materials.

- **Policy EH-4.1: Hazardous Materials Risk Reduction.** Seek to reduce the risk of hazardous materials accidents, spills and vapor releases, and minimize the effects of such incidents if they occur.
- **Policy EH-4.6: Hazardous Materials Transport.** Seek to reduce the risk of accidents in the transportation of hazardous materials. The City will require compliance with all hazardous waste transport standards established by state and federal agencies.
- **Policy EH-4.7: Railroad Cargo Safety.** Work with the Union Pacific Railroad and the California Public Utilities Commission (CPUC) to ensure safe conditions for the loading, unloading, and transport of hazardous materials along rail lines through Newark. UP should be encouraged to maintain its tracks and facilities in excellent condition, and minimize occasions where trains block railroad grade crossings.

Goal EH-5: Emergency Preparedness. Fast, efficient, and coordinated response to natural and manmade emergencies and disaster.

• **Policy EH-5.3: Adequacy of Emergency Response Access.** Avoid placing new development in areas where emergency response and evacuation cannot be provided within acceptable levels.

Goal HW-5: Reducing Hazard Exposure. A land use pattern that minimizes exposure of residents and workers to hazards associated with commercial and industrial uses.

• **Policy HW-5.3: Remediation.** Require remediation of soil and groundwater contamination to a level that is consistent with proposed land uses. All site cleanup shall be coordinated with state and federal regulatory agencies.

Local Hazard Mitigation Plan

The ABAG prepared a Local Hazard Mitigation Plan, Taming Natural Disasters, to help prepare for and mitigate the effects of potential hazards in the Bay Area. To supplement ABAG's Taming Natural



Draft Environmental Impact Report Environmental Impact Analysis

Disasters, Alameda County prepared an Annex particular to its region. The Annex describes the regional and local hazard mitigation planning process, recounts past occurrences of disasters, assesses various risks (e.g. urban land exposure, infrastructure exposure, critical health care facility exposure, etc.), details mitigation goals, and provides mitigation activities and priorities (City of Newark 2013b).

City of Newark Emergency Operations Plan and Chemical Emergency Preparedness Supporting Plan

The City has adopted two emergency response plans. The "Emergency Operations Plan" is the City's primary plan which provides operational procedures for responding to a variety of emergency conditions, including earthquakes, flooding, tsunamis, hazardous material incidents, and civil defense conditions. The guidelines included in this plan address the needs of the entire community and identify key responsible agencies and personnel. The City's second response plan is the "Chemical Emergency Preparedness Supporting Plan." This plan establishes standard operating procedures for responding to a chemical spill or hazardous materials incidents within the City (City of Newark 2013b).

3.9.3 Environmental Impacts

This section analyzes the project's potential to result in significant hazards and hazardous materials impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of documents pertaining to the project site, including the General Plan, General Plan EIR, and online regulatory compliance databases.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's hazards and hazardous materials impacts are significant. Would the proposed project:

- Create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials?
- Create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment?
- Emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within one-quarter mile of an existing or proposed school?
- Be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would it create a significant hazard to the public or the environment?



Draft Environmental Impact Report Environmental Impact Analysis

- For a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard or excessive noise for people rising or working in the project area?
- Impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan?
- Expose people or structures, either directly or indirectly, to a significant risk of loss, injury or death involving wildland fires?

Project Impact Analysis and Mitigation Measures

Routine Transport, Use, or Disposal of Hazardous Materials

Impact HAZ-1 The proposed project could create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials.

Impact Analysis

The proposed project consists of the development of the 29 acre project site with 203 single-family residences and associated infrastructure improvements. The proposed project would require remediation of the project site prior to the start of construction activities as required by General Plan Policy HW-5.3. On-site remediation would be required at the project site prior to start of construction activities to ensure all hazardous materials detected in soil and groundwater samples at the project site are below their respective Tier 1 ESLs. Construction activities associated with the proposed project could expose construction workers and the environment to existing on-site contamination. The Phase II ESA prepared for the proposed project identified that soil at the project site could be remediated to residential levels through the removal of shallow soils in select portions of the project site. The proposed project would be required to implement Mitigation Measure HAZ-1 which requires remediation of contaminated soil and groundwater. The Corrective Action Plan and Remedial Excavation Work Plan prepared for the proposed project proposed to remediate the impacted soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater contamination through in-situ remediation, and surface water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. ACWD would be the responsible agency for oversight and enforcement of the Corrective Action Plan and Remedial Excavation Work Plan. Confirmation sampling is required to be performed regularly to assess the effectiveness of Corrective Action Plan activities. Confirmation sampling conducted after the completion of Corrective Action Plan activities would ensure that hazardous site conditions are no longer present and ensure that development of the proposed project would not create a significant hazard to the public or the environment. Though trace levels of hazardous materials detected in soil and groundwater may be present after the completion of remediation activities, trace levels would not result in potential impacts to future residents and would not result in hazardous conditions being present at the site. Additionally, as the project site is developed with existing structures that would be demolished prior to the start of construction activities, the proposed project would be required to conduct an asbestos survey of the existing structures to determine if asbestos-containing building materials (ACBMs) are present. If ACBMs are identified within the existing structures, the proposed project would be required to properly dispose of hazardous materials in accordance with state and local guidelines. For the disposal of potential existing hazardous materials


Draft Environmental Impact Report Environmental Impact Analysis

located on-site and ACBMs identified within the existing structures, the proposed project would comply with General Plan Policy EH-4.6 which requires compliance with all hazardous waste transport standards established by state and federal agencies. All remediation activities would be completed prior to the start of construction of buildings and would take place prior to site grading. Compliance with all applicable regulations and implementation of Mitigation Measure HAZ-1 would ensure that impacts associated with the routine transport, use, and disposal of hazardous materials during remediation activities for the proposed project would result in a less than significant impact.

The proposed project consists of the development of the 29 acre project site with 203 single-family residences and associated infrastructure improvements. Construction of the proposed project would involve the minor routine transport, use, and disposal of hazardous substances such as diesel fuels, lubricants, solvents, asphalt, paints, building materials, finishing materials, pesticides, and fertilizers. The project contractor would be required to comply with all applicable federal, state, and local laws related to the transport, use, or disposal of hazardous materials, as overseen by the Cal EPA and DTSC. The proposed project would comply with General Plan Policy EH-4.1 which seeks to reduce the risk of hazardous materials accidents, spills and vapor releases, and minimize the effects of such incidents if they occur. With compliance with existing federal, state and local laws, the use of hazardous substances during construction would not result in a significant hazard to the public or environment through the routine transport, use, or disposal of hazardous materials.

Operational impacts associated with the proposed project would be limited to hazardous materials typically generated by residential uses. Hazardous materials used post construction would be those commonly found in other residential uses such as cleaning products, paints, oils, and pesticides for landscaping maintenance activities. These common household hazardous materials would be used in limited quantities and would not create a substantial hazard to the public or the environment. Therefore, impacts related to the routine transport, use, and disposal of hazardous materials during project construction and operation would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM HAZ-1: Remediation of On-Site Contamination.** Prior to the start of construction activities, the Applicant shall remediate existing on-site contamination at the project site to residential standards. Remediation activities shall be completed prior to site grading and construction of buildings. Remediation activities shall include:
 - The Applicant shall implement the Corrective Action Plan and Remedial Excavation Work Plan as approved by the ACWD. The Corrective Action Plan and Remedial Excavation Work Plan provides detailed plans to remediate contaminated soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater containment through in-situ remediation, and surface water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards.

- The project site shall be remediated below relevant screening levels under oversight by an appropriate regulatory agency, in this case the ACWD. The oversight agency shall be responsible for overseeing and directing all site investigation and cleanup activities in a manner that ensures that the standards and requirements of the State of California are fully addressed.
- As part of the Corrective Action Plan and Remedial Excavation Work Plan, confirmation sampling shall be completed after the conclusion of remediation activities to ensure existing on-site contamination has been remediated to residential standards.
- Prior to any demolition of the existing buildings (Pick-n-Pull) an asbestos survey is required by local authorities and/or National Emissions Standards for Hazardous Air Pollutants (NESHAP) guidelines. NESHAP guidelines require the removal of potentially friable Asbestos-Containing Building Material (ACBMs) prior to building demolition or renovation that may disturb the ACBM. The results of the survey shall be submitted to the City for review and approval, prior to issuance of demolition permits.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Accidental Release of Hazardous Materials

Impact HAZ-2 The proposed project could create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment.

Impact Analysis

Construction activities associated with the proposed project could result in release of existing on-site hazardous materials contamination into the environment and therefore, the proposed project would be required to comply with General Plan Policy HW-5.3 and implement Mitigation Measure HAZ-1 which requires remediation of existing on-site contamination prior to the start of construction activities. Remediation of on-site contamination prior to the start of construction activities would ensure that construction activities associated with the proposed project would not release existing on-site hazardous materials into the environment. In addition to remediation of on-site contamination, the proposed project would be required to conduct asbestos surveys to ensure that the existing structures proposed for demolition does not contain ACBMs and ensure demolition activities do not create a hazard to the public or environment. The asbestos survey is required by Mitigation Measure HAZ-1 and the results of the survey shall be submitted to the City for review and approval, prior to issuance of demolition permits. During all construction activities, including remediation activities, the proposed project would be required to implement a SWPPP, as required by Mitigation Measure HYD-1, to prevent contaminated runoff from leaving the project site. Implementation of Mitigation Measure HYD-1, Mitigation Measure HAZ-1, and compliance with applicable federal, state, and local regulation pertaining to hazardous materials would ensure remediation impacts related to the accidental release of hazardous materials are less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

As discussed in Impact HAZ-1, project construction would involve the use of hazardous materials such as fuels, solvents, and paints which has the potential to be released into the environment if not handled properly. The proposed project would be required to comply with applicable federal, state, and local regulations pertaining to the safe handling and storage of hazardous materials. Additionally, as required by Mitigation Measure HYD-1, the proposed project would be required to implement a SWPPP to prevent contaminated runoff from leaving the project site during construction activities. Compliance with existing regulations pertaining to safe handling and storage of hazardous materials used during construction and implementation of Mitigation Measure HYD-1 would ensure construction impacts related to accidental release of hazardous materials are less than significant.

Operational activities would involve limited use of common hazardous materials, including paints, solvents, fuels, oils, cleaners, and pesticides. The use of these substances is not expected to create a significant hazard to the public or the environment through reasonably foreseeable upset or accident as they would not be used in quantities that would cause significant impacts. The proposed project would be required to comply with applicable federal, state, and local laws pertaining to the safe handling, storage, and transport of hazardous materials. Therefore, impacts related to the release of hazardous materials into the environment would be less than significant with the implementation of Mitigation Measure HYD-1 and Mitigation Measure HAZ-1.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HYD-1 and Mitigation Measure HAZ-1 are required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Emission of Hazardous Materials near an Existing School

Impact HAZ-3 The proposed project would not emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within one-quarter mile of an existing or proposed school.

Impact Analysis

The project site is not located within 0.25 mile of an existing or proposed school. The nearest school is the Newark Memorial High School, approximately 0.87 mile northeast of the project site. The proposed project does not involve the development of a use that would emit hazardous materials, substances, or waste during operation. The use of heavy equipment and activities involving hazardous materials would be limited to the construction phase and confined to construction areas and within existing roadways. The use of hazardous materials would also be regulated by health and safety requirements under federal, state, and local laws, including handling, storage, and disposal of the materials, as well as emergency spill response. As such, the proposed project would have a less than significant impact related to the emission or handling of hazardous materials near a school.

Level of Significance Before Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Hazardous Materials Sites

Impact HAZ-4 The proposed project could be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would not create a significant hazard to the public or the environment.

Impact Analysis

The project site is currently operated as an auto wrecking yard since the 1960s and was previously used for agricultural purposes. The project site is included on a list of hazardous materials sites pursuant to Government Code Section 65962.5. The GeoTracker database identifies the Pick-n-Pull Auto Dismantler site as a "Cleanup Program Site." "Cleanup Program Sites" includes all "non-federally owned" sites that are regulated under the SWRCB's Site Cleanup Program. The cleanup site's status is "OPEN – SITE ASSESSMENT" as of January 15, 2020 on the GeoTracker database.

The Phase II ESA conducted for the proposed project identified concentrations of lead exceeding its Tier 1 ESLs in three samples and TPHd was detected at concentrations exceeding its Tier 1 ESL in shallow soil in 26 locations across the project site. The Phase II ESA also identified that the sampling conducted detected TPHq. TPHd. Benzene and ethylbenzene above their respective Tier 1 ESL in groundwater samples. The project site would require remediation activities prior to the start of construction activities to ensure there are no hazardous materials present in soil or groundwater at the site exceeding hazardous levels. The Phase II ESA identified that soils at the project site could be remediated to residential levels through the removal of shallow soils in select portions of the auto wrecking yard. The proposed project would comply with General Plan Policy HW-5.3 and would remediate soil and groundwater contamination to a level consistent with the proposed land use. The proposed project would be required to implement Mitigation Measure HAZ-1 which requires remediation of existing on-site contamination. As required by Mitigation Measure HAZ-1, the Applicant shall implement the Corrective Action Plan and Remedial Excavation Work Plan prepared for the site. The Corrective Action Plan and Remedial Excavation Work Plan proposes to remediate the impacted soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater contamination through in-situ remediation, and surface water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. Confirmation sampling is required to be performed regularly to assess the effectiveness of Corrective Action Plan activities. Confirmation sampling conducted after the completion of Corrective Action Plan activities as part of the Remedial Excavation Work Plan would ensure that hazardous site conditions are no longer present and ensure that development of the proposed project would not create a significant hazard to the public or the environment. Implementation of the mitigation measure would mitigate impacts caused by existing contamination and impacts would be a less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

Mitigation Measure HAZ-1 is required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Nearby Airport Hazard

Impact HAZ-5 The proposed project would not, for a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, result in a safety hazard for people residing or working in the project area.

Impact Analysis

The project site is not located within two miles of an airport and is not located within an airport land use plan. Therefore, the proposed project would not result in a safety hazard for people residing or working in the project site. No impact would occur.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Emergency Response or Evacuation Plan

Impact HAZ-6 The proposed project would not impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan.

Impact Analysis

The proposed project includes modifications and improvements to the existing roadways and may result in partial or complete road closures during construction activities. If road closures are necessary, the proposed project would prepare and implement a Traffic Control Plan (TCP) to ensure that construction activities do not impair or interfere with an adopted emergency response plan or emergency evacuation plan. The proposed project would comply with General Plan Policy EH-5.3 which requires avoiding placing new developments in areas where emergency response and evacuation cannot be provided within acceptable levels. The project site is located in an area already served by emergency services and development of the proposed project would not result in permanent modification to roadways that would interfere with any adopted emergency plans. Modification to roadways that is proposed as part of the proposed project would be constructed in accordance with the City's standards to provide adequate emergency access. Therefore, project construction and operation activities would not interfere with an emergency evacuation or response plan, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Wildland Fires

Impact HAZ-7	The proposed project would not expose people or structures, either directly or
	indirectly, to a significant risk of loss, injury or death involving wildland fires.

Impact Analysis

The project site is not located in or near a state responsibility area or a very high fire hazard zone of a locally responsible area (CAL FIRE 2007, 2008). Primary access to the project site would be through Mowry Avenue and adequate site ingress and egress to the project site for emergency vehicles would be provided.

All utilities required for the new development would be located underground and the proposed project includes installation of fire hydrants throughout the project site to mitigate fire hazards. The proposed project would be required to implement General Plan policies, such as General Plan Policy EH-5.3 which requires new developments be placed in areas where adequate emergency response can be provided, as well as the CFC and the Uniform Building Code which would reduce effects of development on wildland fire hazard impacts to a less than significant level. As such, the proposed project is not expected to be exposed to risks associated with wildland fires, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



3.10 HYDROLOGY AND WATER QUALITY

This section describes the environmental and regulatory setting for hydrology and water quality. It also describes existing conditions and potential impacts related to hydrology and water quality that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.10.1 Environmental Setting

Watershed and Regional Drainage

A watershed is the geographic area draining into a river system, ocean, or other body of water through a single outlet and includes the receiving waters. The City is located in the Lower Alameda Creek Watershed, which is further divided into three sub watersheds that are present within the City. The project site is located within the Mowry Slough watershed which consists of a network of storm drains and channels that have replaced small creeks that formerly drained into Mowry Slough (City of Newark 2013b).

Groundwater

The City is located within the Niles Cone Groundwater Basin, which is part of the larger Santa Clara Valley Groundwater Basin. The Niles Cone Groundwater Basin is bounded on the north by the boundary of Alameda County Water District and southern portion of Hayward, on the east by the Diablo Range, on the south by the Alameda-Santa Clara County border, and on the west by San Francisco Bay. The Niles Cone Groundwater Basin is designated as a medium-priority basin by the California Department of Water Resources.

The Niles Cone Basin is currently listed as having existing beneficial uses for groundwater and is the principal source of local supply for the ACWD. Groundwater recharge occurs through percolation of both local and imported water in Alameda Creek and the adjacent recharge ponds in the Quarry Lakes Regional Recreational Area. The water is subsequently recovered through ACWD's groundwater production wells and provides a potable supply to people in cities of Fremont, Newark, and Union City (City of Newark 2013b).

Water Quality

Most of the streams and creeks that originally flowed through the City have been replaced by a network of storm drains and channels that discharge urban runoff in to Newark Slough, Plummer Creek Slough, and Mowry Slough. The surface water bodies that exist in the City include engineered channels maintained by the ACFC&WCD, Plummer Creek, Newark Slough, Mowry Slough, tidal marshes, tidal flats, salt ponds, and small tidal estuaries (City of Newark 2013b). Pollutants could be present in stormwater runoff, including sediment, nutrients, oxygen-demanding substances, heavy metals, petroleum hydrocarbons, pathogenic bacteria, and viruses. Stormwater runoff is the principal source of pollution entering surface and ground water in the San Francisco Bay region. Typical pollutants include oil, grease, or antifreeze releases from cars or trucks; paint or paint products; leaves or yard waste; pesticides, herbicides, or



Draft Environmental Impact Report Environmental Impact Analysis

fertilizers from yards and gardens; solvents and household chemicals; animal wastes, litter, or sewer leakage; and construction debris such as fresh concrete, mortar, or cement.

Flooding

Flood hazard zones are identified on official Flood Insurance Rate Maps issued by the Federal Emergency Management Agency. The project site is designated as Zone X and Zone X (Shaded). The northern part of the project site is within an area of minimal flood hazard (Zone X); however, the majority of the project site is located in an area with a 0.2 percent annual chance of shallow flooding of less than one foot (Zone X Shaded) (FEMA 2021).

Seiches, Dam Inundation, and Tsunamis

According to the City's General Plan EIR and ABAG, the City is located within the inundation areas of three dams; Del Valle, Turner, and Calaveras, all of which are classified as high hazard dams because their failure could result in a significant loss of life and property damage (City of Newark 2013b). The Del Valle Dam is located approximately 15.5 miles northeast of City limits, James H. Turner Dam is located approximately 8.2 miles northeast of City limits, and Calaveras Dam is located approximately 9 miles east of City limits.

Seiches are standing waves oscillating in a landlocked body of water, typically caused by strong winds or seismic ground shaking. There are no large bodies of water or reservoir within the City and therefore, there is no likelihood of seiches occurring within the City. Tsunamis are tidal waves created by undersea fault movement. These waves are fast moving, create large swells of water, and upon reaching the coast can sweep inland with a large amount of force. The risk of flooding due to a tsunami event is considered to very low within the City and ABAG's tsunami evacuation maps for the Bay Area does not identify City as being located within tsunami inundation zones.

Sea Level Rise

According to the Areas 3 and 4 Specific Plan RDEIR, global temperatures have increased by approximately one degree Fahrenheit and sea level has risen by approximately 0.5 foot over the last century. The Areas 3 and 4 Specific Plan RDEIR included a range of potential future sea levels based on the Intergovernmental Panel on Climate Change's climate change scenarios. The mid-range projection of sea level change by 2058 was approximately 160 millimeters, or about six inches. Within the Areas 3 and 4 Specific Plan, the residential structures of Area 4 would be most directly impacted by global climate and sea level changes. According to the Areas 3 and 4 Specific Plan RDEIR, the most currently available estimates for sea level rise by 2050 range from 0.3 foot to 1.5 feet, and by 2100 from 0.6 foot to 4.8 feet. This would be consistent with the Adapting to Rising Tides (ART) Bay Area Sea Level Rise and Shorelines Analysis Maps' sea level rise projections. According to the ART map, 48 inches (four feet) of sea level rise would result in flooding to the project site (ART 2023). The ART Bay Area Sea Level Rise and Shorelines Analysis Maps' provides a regional-scale illustration of coastal flooding due to specific sea level rise and storm surge scenarios, and are intended to improve sea level rise awareness and preparedness.

New developments within the Areas 3 and 4 Specific Plan area are required to comply with the Newark Municipal Code Section 15.40.051, Standards of Construction, which requires new construction and



Draft Environmental Impact Report Environmental Impact Analysis

substantial improvements of any structure to have the lowest floor, including basement, elevated to or above the base flood elevation. Newark's Municipal Code calls for residential structures to be elevated to or above the base flood elevation or to a minimum of six inches above the building pad which shall be at a minimum elevation of 11.25 feet on the NGVD, whichever affords the greater degree of flood damage protection. This means the building pads for residential structures must be at 11.25 feet AMSL with the finished floor a minimum of six inches above the building pad (i.e. at 11.75 feet AMSL) (City of Newark 2015b). General Plan Policy EH-3.3 requires new residential developments to be constructed above the FEMA 100-year flood elevation. Additionally, the proposed project would be required to be elevated to accommodate BCDC's adopted sea level rise guidance which recommends a minimum pad elevation of 12.2 feet and the minimum pad elevation requirements of the RPC-SAT's projected likely range for sea level rise by the year 2100 of 3.4 feet.

San Francisco Bay Conservation and Development Commission

The Applicant reached out to BCDC staff on June 29, 2023, to request confirmation on whether the three parcels within the project site was located within BCDC jurisdiction. A response was received from BCDC staff on July 17, 2023, where BCDC staff confirmed that project site parcels identified as APN 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00 are not located within BCDC jurisdiction (Yuri Jewett, personal communication, July 17, 2023). Therefore, the project site parcels are not located within BCDC jurisdiction. As a courtesy, City staff would follow up with BCDC during the public review period.

3.10.2 Regulatory Setting

Federal

Federal Clean Water Act

The federal CWA (33 United States Code Section 1251 et seq.), formerly the Federal Water Pollution Control Act of 1972, was enacted with the intent of restoring and maintaining the chemical, physical, and biological integrity of the waters of the United States. The CWA requires states to set standards to protect, maintain, and restore water quality through the regulation of point source and certain non-point source discharges to surface water. Those discharges are regulated by the NPDES permit process (CWA Section 402). Section 401 of the CWA regulates surface water quality and a Water Quality Certification is required for federal actions (including construction activities) that may result in impacts to surface water. In California, NPDES permitting authority is delegated to, and administered by, the nine RWQCBs. The proposed project is located within Region 2, regulated by the San Francisco Bay RWQCB.

National Pollutant Discharge Elimination System

The NPDES permit program was established by the CWA to regulate municipal and industrial discharges to surface waters of the United States, including discharges from municipal separate storm sewer systems. Federal NPDES permit regulations have been established for broad categories of discharges, including point-source municipal waste discharges and non-point source stormwater runoff. NPDES permits generally identify effluent and receiving water limits on allowable concentrations and/or mass emissions of pollutants contained in the discharge; prohibitions of discharges not specifically allowed under the permit; and provisions that describe required actions by the discharger, including industrial pretreatment, pollution prevention, self-monitoring and other activities.



Draft Environmental Impact Report Environmental Impact Analysis

Developers whose projects disturb one or more acres of soil or which projects disturb less than one acre but are part of a larger common plan of development that in total disturbs one or more acres are required to file a notice on intent to obtain coverage under the NPDES Construction General Permit. The Construction General Permit required the preparation and implementation of a stormwater pollution prevention plan which must be completed before construction begins. The SWPPP should contain a site map that shows the construction site perimeter; existing and proposed buildings, lots, roadways, and stormwater collection and discharge points; general topography both before and after construction; and drainage patterns across the project site. The SWPPP must list best management practices the discharger will use to manage stormwater runoff and the placement of those BMPs.

State

Porter Cologne Water Quality Control Act

The State of California established the SWRCB, which oversees the nine RWQCBs, through the Porter-Cologne Water Quality Control Act (Porter-Cologne). Through the enforcement of Porter-Cologne, the SWRCB determines the beneficial uses of the waters (surface and groundwater) of the State, establishes narrative and/or numerical water quality standards, and initiates policies relating to water quality. The SWRCB and, more specifically, the RWQCB, are authorized to prescribe Waste Discharge Requirements for the discharge of waste, which may impact waters of the State. Furthermore, the development of water quality control plans, or Basin Plans, are required by Porter-Cologne to protect water quality. The SWRCB issues both General Construction Permits and Individual Permits under the auspices of the federal NPDES program.

State Water Resources Control Board and Regional Water Quality Control Boards

In California, the SWRCB has broad authority over water quality control issues for the state. The SWRCB is responsible for developing statewide water quality policy and exercises the powers delegated to the state by the federal government under the CWA.

Regional authority for planning, permitting, and enforcement is delegated to the nine RWQCBs. The regional boards are required to formulate and adopt water quality control plans for all areas in the region and establish water quality objectives in the plans. The City of Newark is within the jurisdiction of the San Francisco Bay RWQCB (Region 2). The San Francisco Bay RWQCB adopted a Water Quality Control Plan for the San Francisco Bay Basin (the Basin Plan) that designates beneficial uses, establishes water quality objectives, and contains implementation programs and policies to achieve those objectives for all waters addressed through the Basin Plan. Areas of the City of Newark that have shallow groundwater may require dewatering during excavation and trenching activities for new development. This activity is subject to the RWQCB construction dewatering permit requirements (R2-2012-0060). Discharge of any sediment-laden water from a dewatering site into waters of the State is prohibited. Discharge of uncontaminated groundwater from dewatering occur within an area of the City where previous groundwater contamination has been reported and still exists; the extracted groundwater would require treatment prior to discharge. The disposal of dewatered discharges would require a permit or a waiver (exemption) from the RWQCB for discharge to surface creeks and groundwater. The ACWD is the appropriate agency if



Draft Environmental Impact Report Environmental Impact Analysis

permits are required for dewatering wells and local agencies should be contacted if the discharge will be released to storm or sanitary sewers.

Sustainable Groundwater Management Act

The Sustainable Groundwater Management Act (SGMA) is a three-bill package that passed the California state legislature and was signed into California state law by Governor Jerry Brown in September 2014. SGMA establishes a framework for long-term sustainable groundwater management across California and requires local agencies to bring overdrafted basins into balanced levels of pumping and recharge The California Department of Water Resources (DWR) uses the California Statewide Groundwater Elevation Model Priority List to rank groundwater basins across the state according to priority levels of high, medium, low, or very low, and SGMA specifies deadlines for completion of Groundwater Sustainability Plans (GSPs) in order of basin priority. Under SGMA, high- and medium-priority basins, as designated by DWR, must establish GSPs in order of basin priority. Under SGMA, high- and medium-priority basins, as designated by DWR, must establish Groundwater Sustainability Agencies (GSAs) that oversee the preparation and implementation of a local GSP.

Local

San Francisco Bay Conservation and Development Commission

In 1969, the McAteer-Petris Act designated BCDC as the agency responsible for the protection of San Francisco Bay and its natural resources. BCDC fulfills this mission through the implementation of the San Francisco Bay Plan, an enforceable plan that guides the future protection and use of San Francisco Bay and its shoreline.

As a permitting authority along the San Francisco Bay shoreline, BCDC is responsible for granting or denying permits for any proposed fill, extraction of materials, or change in use of any water, land, or structure within the Commission's jurisdiction. BCDC has jurisdiction of Mowry Slough ending at the culvert at the Mowry Avenue bridge crossing (approximately 0.2 miles west of project site), at the bend of the channel near Plummer Creek (approx, and jurisdiction over managed wetlands in the Southwest Newark Residential and Recreational Focus Area. Projects located within BCDC jurisdiction that involve Bay fill must be consistent with the Bay Plan policies on the safety of fills and shoreline protection (City of Newark 2013b).

Alameda County Flood Control and Water Conservation District

The ACFC&WCD provides flood protection for Alameda County residents and businesses. The ACFC&WCD plans, designs, constructs, and maintains flood control projects such as natural creeks, channels, levees, pump stations, dams, and reservoirs. In 2016, the ACFC&WCD updated the Hydrology & Hydraulics Manual which serves as a guide for minimum design requirements and provides a hydrologic model for all of Alameda County. The ACFC&WCD is also charged with administering the Clean Water Program for unincorporated areas of Alameda County, the 14 cities of Alameda County, the ACFC&WCD, and the Zone 7 Water Agency. The ACFC&WCD provides administrative and contracting services for the Alameda Countywide Clean Water Program to help comply with federal and state requirements to improve water quality and better manage urban stormwater runoff.



Draft Environmental Impact Report Environmental Impact Analysis

Alameda County Clean Water Program

The City is subject to the Provision C.3 requirements that are post construction stormwater management requirements for new development and redevelopment projects. Provision C.3 requirements are separate from, and in addition to, requirements for erosion and sediment control and for pollution prevention measures during construction. These requirements apply to all new development or redevelopment projects that create or replace 10,000 square feet of impervious surfaces. Project applicants are required to implement site design measure, source control measure, and stormwater treatment measures to reduce stormwater pollution after construction of the project. The permit specifies methods to calculate the required size of treatment devices (City of Newark 2013b).

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:

Goal LU-7: Southwest Newark Residential and Recreational Project. Develop the Southwest Newark Residential and Recreational Project as one of the Silicon Valley's premier new neighborhoods, with executive housing and high quality recreation.

- **Policy LU-7.7: Maintaining Hydrologic Features.** Maintain the natural hydrologic features of the Southwest Newark Residential and Recreational Project area to the extent feasible, and maintain or improve the current quality of water leaving the site.
- **Policy LU-7.8: Mitigating Construction Impact.** Avoid and mitigate construction impacts on wetlands, aquatic habitat, wildlife, and water quality as development takes place in the Southwest Newark Residential and Recreational Project. Measures to minimize such impacts should be included in project approvals, consistent with state and federal agency oversight and regulations.

Goal CS-3: Water Resources. Conserve and enhance Newark's water resources,

- **Policy CS-3.1: Protection of Water Resources.** Ensure that land use decisions consider the availability of water for domestic and non-domestic uses, potential impacts on groundwater quality and groundwater recharge capacity, and potential off-site impacts on water quality.
- **Policy CS-3.2: Water Conservation Standards.** Promote water conservation through development standards, building requirements, irrigation requirements, landscape design guidelines, and other applicable City policies and programs.
- **Policy CS-3.4: Reducing Water Pollution.** Protect the quality of Newark's surface waters by supporting controls on point source and non-point sources of pollution.

Goal EH-3: Flooding Hazards. Reduce risk to life and property associated with flooding.

• **Policy EH-3.1: Planning to Avoid Flood Hazards.** Identify flood prone areas in Newark and utilize this data for land use and transportation planning purposes. Flood resistant construction techniques and minimum building elevations shall be required to reduce flood hazards.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy EH-3.2: Maintaining Drainage Patterns.** Prohibit development, grading, and land modification activities that would adversely affect Newark's drainage system or create unacceptable erosion impacts.
- **Policy EH-3.3: Residential Development in the Flood Plain.** Require that new residential development, including streets and other surface improvements, be constructed above the 100-year flood elevation.

Goal CSF-5: Infrastructure. Provide safe, reliable, and efficiently operated infrastructure which meets Newark's long-term water, sewer, and stormwater management needs.

- **Policy CSF-5.4: Flood Control.** Coordinate with Alameda County Flood Control and Water Conservation District and Alameda County Public Works to ensure that stormwater runoff is managed in a way that reduces flood hazards.
- **Policy CSF-5.5: Drainage with New Development.** Ensure that new development provides drainage and flood protection improvements which reduce on-site and downstream hazards such as ponding, flooding, and erosion. New development areas should be designed to minimize impervious surfaces in order to reduce associated site runoff and maximize groundwater recharge.

3.10.3 Environmental Impacts

This section analyzes the project's potential to result in significant hydrology and water quality impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The evaluation of potential hydrologic and water quality impacts was based on a review of City documents, including the General Plan and General Plan EIR. Mapping tools provided by FEMA were also reviewed. The information obtained from these sources are summarized to establish existing conditions and to identify potential environmental effects. In determining the level of significance, the analysis assumes that the proposed project would comply with relevant federal, state, and local ordinances and regulations.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's impacts to hydrology and water quality are significant. Would the proposed project:

- Violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or groundwater quality?
- Substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the project may impede sustainable groundwater management of the basin?



Draft Environmental Impact Report Environmental Impact Analysis

- Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would:
- Result in substantial erosion or siltation on- or off-site;
- Substantially increase the rate or amount of surface runoff in a manner which would result in flood on- or off-site;
- Create or contribute runoff water which would exceed the capacity of existing or planned stormwater drainage systems or provide substantially additional sources of polluted runoff; or
- Impede or redirect flood flows?
- In flood hazard, tsunami, or seiche zones, risk release of pollutants due to project inundation?
- Conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan?

Project Impact Analysis and Mitigation Measures

Surface and Groundwater Quality

Impact HYD-1 The proposed project could violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or groundwater quality.

Impact Analysis

Construction activities required for the proposed project would include demolition, site clearing, grading, utility connections, building construction, frontage improvements, and landscaping on-site. Construction activities would involve grading of the entire project site and permanent disturbance of the site. These activities have the potential to generate stormwater runoff and to discharge pollutants, such as fuel, solvents, oil, paints, and trash, into the City's storm drain system. The proposed project would be required to comply with General Plan Policy LU-7.8 and CS-3.4 which requires protection of water quality from pollution and mitigation of construction impacts on water quality. Protection of water quality during construction would be achieved with compliance and implementation of the NPDES General Construction Permit. The proposed project would comply with the NPDES General Construction Permit which requires the preparation of a SWPPP and incorporation of BMPs to control sedimentation, erosion, and hazardous materials from contacting stormwater, with the intent of keeping all products of erosion from moving offsite into receiving waters. The preparation and implementation of the SWPPP and applicable BMPs have been incorporated into Mitigation Measure HYD-1 to reduce potential water quality impacts to a less than significant level.

Additionally, as required by Mitigation Measure HAZ-1, the proposed project would include on-site remediation activities including the remediation of contaminated surface waters in the existing retention basins. Surface water would be removed from the existing retention basins and piped to a USD sewer inlet in conformance with USD permit requirements. The surface water in all three stormwater retention basins would be removed and discharged to the sanitary sewer system under permit. With conformance



Draft Environmental Impact Report Environmental Impact Analysis

to the USD permit requirements, the discharge of surface waters would not result in a violation of water quality requirements or waste discharge requirements.

Post construction impacts from development could affect drainage patterns and increase the overall amount of impervious surfaces, thus creating changes to stormwater flows and water quality. Water quality in stormwater runoff is regulated locally by the Alameda County Clean Water Program, which includes the C.3 provision set by the San Francisco Bay RWQCB. C.3 requirements apply to all new development or redevelopment projects that create or replace 10,000 square feet of impervious surfaces. C.3 requirements are separate from, and in addition to, requirements for erosion and sediment control and for pollution prevention measures during construction. Project applicants are required to implement site design measures, source control measures, and stormwater treatment measures to reduce stormwater pollution after construction of the proposed project. Implementation of site design measures, source control measures would ensure the proposed project is in compliance with General Plan Policy CS-3.4 which requires the protection of water quality from point source and non-point sources of pollution.

The proposed on- and off-site improvements from the proposed project would result in approximately 881,450 square feet of impervious surfaces and 465,680 square feet of pervious surfaces at the site. The proposed project would install a storm drain system consisting of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes. The storm drainage system would utilize LID techniques which may include directing roof runoff to vegetated areas, storm drain stenciling, and site design that promotes infiltration. Storm drain pipes installed throughout the project site would convey stormwater to the two on-site bioretention treatment areas which would be planted with water conserving grass species, shrubs, and trees that are adapted to bioswale conditions. The bioretention treatment areas would discharge flows through the on-site storm drain system, into the adjacent City owned open space parcel, consistent with the historic drainage path. The use of bioretention areas and discharge of flows consistent with the historic drainage path would ensure the proposed project would comply with General Plan Policy LU-7.7 which requires the maintenance of natural hydrologic features in the area to the extent feasible. The proposed project would also construct off-site stormwater improvements along Mowry Avenue. The proposed project would construct improvements to storm drains within Mowry Avenue which would collect and convey flow towards the ACFC&WCD Channel Line B at the terminus of Mowry Avenue and the proposed roadway improvement would also include four bioretention treatment areas along Mowry Avenue. The proposed drainage system improvements would be designed and constructed in accordance with the City's requirements and C.3 requirements. In addition, the proposed project would be required to prepare a Stormwater Management Plan (SWMP) that includes post construction BMPs that control pollutant levels. A Post Construction Stormwater Control Plan has been prepared for the proposed project by Carlson, Barbee and Gibson, Inc which includes LID design strategies, source control measures, and documents the proposed drainage design necessary to comply with applicable C.3 guidelines. The Post Construction Stormwater Control Plan would be submitted to the City upon finalization of the project design and would be implemented into the proposed project. Compliance with City requirements and C.3 standards and implementation of the post construction SWMP would ensure that operation of the proposed project does not result in violation of water quality standards or waste discharge requirements.



Draft Environmental Impact Report Environmental Impact Analysis

Therefore, operation of the proposed project would result in a less than significant impact regarding water quality degradation and construction of the proposed project would result in a less than significant impact with implementation of Mitigation Measure HYD-1 and Mitigation Measure HAZ-1.

Level of Significance Before Mitigation Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HAZ-1 is required.

- **MM HYD-1: Prepare and Implement a SWPPP.** Prior to the issuance of any construction-related permits, the applicant shall prepare and submit a Notice of Intent (NOI) to the SWRCB and prepare a SWPPP in compliance with the NPDES General Construction Permit. The SWPPP shall include a detailed, site-specific listing of the potential sources of stormwater pollution; pollution prevention measures (erosion and sediment control measures and measures to control non-stormwater discharges and hazardous spills); description of the type and location of erosion and sediment control BMPs to be implemented at the project site; and a BMP monitoring and maintenance schedule to determine the amount of pollutants leaving the project site. A copy of the SWPPP must be current and remain on-site. Water quality BMPs identified in the SWPPP could include but are not limited to the following:
 - Surface water runoff shall be controlled by directing flowing water away from critical areas and by reducing runoff velocity. Diversion structures, such as terraces, dikes, and ditches, shall collect and direct runoff water around vulnerable areas to prepared drainage outlets.
 - Surface roughening, berms, check dams, hay bales, or similar devices shall be used to reduce runoff velocity and erosion.
 - Sediment shall be contained when conditions are too extreme for treatment by surface protection. Temporary sediment traps, filter fabric fences, inlet protectors, vegetative filters and buffers, or settling basins shall be used to detain runoff water long enough for sediment particles to settle out. Construction materials, including topsoil and chemicals, shall be stored, covered, and isolated to prevent runoff losses and contamination of groundwater.
 - Topsoil removed during construction shall be carefully stored and treated as an important resource. Berms shall be placed around topsoil stockpiles to prevent runoff during storm events.
 - Fuel and vehicle maintenance areas shall be established away from all drainage courses, and these areas shall be designed to control runoff.
 - Temporary erosion control measures, such as silt fences, staked straw bales, and temporary revegetation, shall be employed for disturbed areas. No disturbed surfaces



will be left without erosion control measures in place during the winter and spring months.

- A spill prevention and countermeasure plan shall be developed to identify proper storage, collection, and disposal measures for potential pollutants (such as fuel, fertilizers, pesticides, etc.) used on-site. The plan will also require the proper storage, handling, use, and disposal of petroleum products.
- Construction activities shall be scheduled to reduce land disturbance during peak runoff periods and to the immediate area required for construction. Soil conservation practices shall be completed during the fall or late winter to reduce erosion during spring runoff. Existing vegetation will be retained where possible. To the extent feasible, grading activities shall be limited to the immediate area required for construction.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Groundwater Management

Impact HYD-2 The proposed project could substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the project may impede sustainable groundwater management of the basin.

Impact Analysis

Historically high groundwater at the project site has been reported at 5 feet bgs and based on the preliminary investigations and borings conducted at the project site, groundwater is estimated to be at a depth of 4 to 8.5 feet bgs (Berlogar, Stevens, and Associates 2019). Groundwater could be encountered during excavation activities and require dewatering. The Applicant would be required to comply with the waste discharge requirements of the San Francisco Bay RWQCB. Discharge of non-stormwater from an excavation that contains sediments or other pollutants to sanitary sewer, stormwater systems, creek beds, or receiving waters without treatment is prohibited. Discharge of uncontaminated groundwater from dewatering is a conditionally exempted discharge by the San Francisco Bay RWQCB. The proposed project would require implementation of Mitigation Measure GEO-3 which requires the preparation and implementation of a dewatering plan in accordance with the waste discharge requirements of the RWQCB. The dewatering plan would detail the location of dewatering activities, equipment, and discharge point in accordance with the requirements of the RWQCB. The dewatering plan would be required to be submitted to the City for review and approval prior to the start of construction activities. Implementation of Mitigation Measure GEO-3 would ensure the proposed project would comply with General Plan Policy CD-3.1 which requires the consideration of potential impacts to groundwater quality and water quality. Therefore, with implementation of Mitigation Measure GEO-3, the proposed project's construction activities would result in a less than significant impact to groundwater.

Water required for operation of the proposed project would be provided by the ACWD which utilizes treated groundwater as one of its sources for water supply. The proposed project would create 881,450 square feet of impervious surface at the project site, which could potentially impact groundwater because areas currently available for the infiltration of rainfall would be reduced. However, the proposed project



Draft Environmental Impact Report Environmental Impact Analysis

would incorporate 465,680 square feet of pervious surface at the project site consisting of landscaped areas and bioretention treatment areas. The proposed project would be required to comply with General Plan Policy CSF-5.5 which requires new development to provide drainage improvements to reduce runoff and maximize groundwater recharge The General Plan EIR identified that implementation of LID guidelines that include the use of permeable paving materials and on-site infiltration would increase the potential for groundwater recharge, and use of site design features required by the C.3 requirements and implementation of water use efficiency measures would ensure that groundwater supplies are not depleted (City of Newark 2013b). The proposed project would construct storm drain systems with LID techniques incorporated and pervious areas would include bioretention areas and landscaped areas which would allow for infiltration for groundwater recharge. The proposed drainage system improvements would be designed and constructed in accordance with the City's water efficiency guidelines and C.3 requirements which would ensure the proposed project's compliance with General Plan Policy CD-3.2 and ensure the promotion of water conservation through development standards, building requirements, irrigation requirement, and landscape design guidelines. Therefore, with implementation of standards and requirements applicable to groundwater recharge and groundwater supplies such as implementation of LID guidelines and construction of drainage systems in accordance with the City's water efficiency guidelines and C.3 requirements, operation of the proposed project would not substantially decrease groundwater supplies or impede groundwater recharge, and impacts would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures Mitigation Measure GEO-3 is required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.



Draft Environmental Impact Report Environmental Impact Analysis

Drainage Pattern

Impact HYD-3	The proposed project could substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river
	or through the addition of impervious surfaces, in a manner which would:
	i) Result in substantial erosion or siltation on- or off-site;
	 Substantially increase the rate or amount of surface runoff in a manner which would result in flooding on- or off-site;
	 iii) Create or contribute runoff water which would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff; or
	iv) Impede or redirect flood flows.

Impact Analysis

i. Result in substantial erosion or siltation on- or off-site

Construction of the proposed project would include ground disturbing work that would involve grading of portions of the project site, off-site ground disturbing work along Mowry Avenue, and the permanent disturbance of the 29 acre site. As a result, construction activities could result in erosion related impacts. As required by General Plan Policy LU-7.8, the proposed project is required to mitigate construction impacts on water quality, including those resulting from erosion. As outlined under Impact HYD-1, the proposed project would implement Mitigation Measure HYD-1 which requires the preparation of a SWPPP in accordance with the NPDES General Construction Permit. The SWPPP would include BMPs and pollution prevention measures (erosion and sediment control measures and measures to control non-stormwater discharges and hazardous spills) that would be implemented during construction activities to reduce the potential of erosion impacts. With implementation of Mitigation Measure HYD-1, the proposed project's construction would not result in substantial erosion or siltation on- or off-site.

Operation of the proposed project is not anticipated to result in substantial erosion or siltation. Storm drain pipes installed throughout the project site would convey impervious surface runoff to the on-site bioretention areas before discharging into the adjacent City owned open space parcel. These features would provide treatment, retention, and/or detention at the project site to reduce the volume of stormwater runoff and ensure that polluted runoff does not discharge into the storm drain system. Additionally, the proposed storm drainage system would be designed and constructed in accordance with C.3 requirements which requires implementation of site design measures, source control measures, and stormwater treatment measures to reduce stormwater pollution after construction of the proposed project. The new storm drain improvements constructed for the proposed project would ensure that the proposed project would comply with General Plan Policy CS-3.4, which requires the protection of water quality by supporting controls of point source and non-point sources of pollution, as the proposed project would treat runoff prior to it being discharged. Therefore, with implementation of applicable storm drainage requirements, operation of the proposed project would not result in substantial erosion or siltation on- or off-site. With implementation of Mitigation Measure HYD-1, the proposed project would result in less than significant impacts related to erosion and siltation.



Draft Environmental Impact Report Environmental Impact Analysis

ii. Substantially increase the rate or amount of surface runoff in a manner which would result in flooding on- or off-site

The proposed project involves the development of a 29 acre project site with 203 single-family residences. The proposed project would result in 881,450 square feet of impervious surfaces at the project site which would result in an increase in impervious surface at the site and increase the amount of stormwater runoff from the site. The proposed project would be required to comply with General Plan Policy CSF-5.5 which requires new development to provide drainage improvements to reduce site runoff. To control runoff, the proposed project would construct a storm drain system consisting of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes. Stormwater runoff would be directed to the storm drain pipes on-site which would convey stormwater to the two on-site bioretention areas. The bioretention treatment areas would then discharge flows through the on-site storm drain system into the adjacent City owned open space parcel, consistent with the historic drainage path. The proposed project would also construct improvements to the off-site storm drain system. The proposed project would construct improvements to storm drain pipes within Mowry Avenue which would collect and convey flow towards the ACFC&WCD Channel Line B at the terminus of Mowry Avenue and would construct four bioretention treatment areas along Mowry Avenue. The proposed project would design and construct storm drain systems in accordance with applicable requirements to control the volume of surface runoff at the project site. Therefore, storm drain systems would be designed to handle the increased surface runoff from the site and the proposed project would not result substantial increase in the rate or amount of surface runoff in a manner that would result in on- or off-site flooding and impacts would be less than significant.

iii. Create or contribute runoff water, which would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff

As described above, construction activities would have the potential to generate stormwater runoff and to discharge pollutants, such as fuel, solvents, oil, paints, and trash, into the City's storm drain system. In addition, the increase in impervious surface resulting from project implementation would alter the type and level of pollutants in stormwater runoff from the project site. To comply with General Plan Policy LU-7.8, which requires avoidance and mitigation of construction impacts on water quality, the proposed project would implement Mitigation Measure HYD-1 During construction activities, the proposed project would conform to the requirements of the NPDES General Construction Permit, which requires the preparation and implementation of a SWPPP. The SWPPP would specify BMPs to incorporate during construction to prevent, control, and reduce polluted runoff from entering the City's storm drain system and waterways. Implementation of these BMPs during construction is required as part of Mitigation Measure HYD-1.

As identified in the City's General Plan EIR, the City requires as a standard condition of approval that major development projects complete drainage and hydrology analysis to ensure that on- and off-site drainage facilities can accommodate increased stormwater flows. A Post Construction Stormwater Control Plan has been prepared for the proposed project by Carlson, Barbee and Gibson, Inc which includes LID design strategies, source control measures, and documents the proposed drainage design necessary to comply with applicable C.3 guidelines. The Post Construction Stormwater Control Plan would be submitted to the City upon finalization of the project design and would be implemented into the



Draft Environmental Impact Report Environmental Impact Analysis

proposed project. The proposed project would implement C.3 requirements for new development and construct a storm drain system with LID techniques incorporated and bioretention treatment areas onand off-site. Stormwater generated at the project site would be directed towards the bioretention treatment areas before being released into the existing drainage pattern to ensure polluted runoff does not enter the City's systems. The Areas 3 and 4 Specific Plan RDEIR assessed the capacity of existing or planned storm drain system to carry runoff from the planned development areas and found that impacts would be less than significant (City of Newark 2013b). With implementation of City requirements and C.3 requirements, the proposed project's operation would not create or contribute runoff that would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff. Construction of the proposed project would require implementation of Mitigation Measure HYD-1 to prevent and reduce polluted runoff from the construction site. Therefore, impacts would be less than significant with implementation of Mitigation Measure HYD-1.

iv. Impede or redirect flood flows

The project site is designated as Zone X and Zone X (Shaded). The northern part of the project site is within an area of minimal flood hazard (Zone X); however, the majority of the project site is located in an area with a 0.2 percent annual chance of shallow flooding of less than one foot (Zone X Shaded) (FEMA 2021). Project construction would require the import of clean fill to elevate the proposed pad grades for the homes above the FEMA 100-year flood plain elevation as required by Newark Municipal Code Section 15.40.051 and to accommodate BCDC's adopted sea level rise guidance. The import of clean fill to elevate the proposed pad grades would ensure compliance with General Plan Policy EH-3.3 which requires new residential developments to be constructed above the 100-year flood elevation. The proposed project would have a minimum pad elevation of 13.0 feet NGVD with an average pad elevation of 14.2 feet NGVD. The proposed project would elevate the proposed pad grades for the homes above the FEMA 100-year flood plain elevation of 9.3 feet and would exceed the City's requirement of having a minimum pad elevation of 11.25 feet. Additionally, the project site would be elevated to accommodate BCDC's adopted sea level rise guidance which recommends a minimum pad elevation of 12.2 feet and would meet the minimum pad elevation requirements of the RPC-SAT's projected likely range for sea level rise by the year 2100 of 3.4 feet. The proposed project could have the potential to exacerbate coastal squeeze which is defined as the loss of natural habitats or deterioration of their quality arising from placement of structures along the shoreline, preventing the landward transgression of those habitats that would naturally occur in response to sea level rise. However, given the baseline condition of the project site as already developed with existing structures, development of the proposed project would not exacerbate potential coastal squeeze impacts beyond what is already present. Therefore, this would not be a driving impact for the proposed project.

The proposed project would also include the construction of storm drain systems which would include bioretention treatment areas to accommodate surface runoff. The new constructed storm drainage system on-site would be designed and constructed to handle potential flood volumes and runoff from the site would be discharged through the on-site storm drain system into the adjacent City owned open space parcel, consistent with the historic drainage path. The proposed project would not substantially alter existing drainage pattern of the site in a matter which would impede or redirect flood flows and impacts would be less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HYD-1 is required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

Project Inundation

Impact HYD-4 The proposed project would not, in flood hazard, tsunami, or seiche zones, risk release of pollutants due to project inundation.

Impact Analysis

According to the ABAG, the City is located within the inundation area of three dams; however, the California Division of Safety of Dams inspects each dam on an annual basis to ensure the dam is safe, performing as intended, and is not developing problems. Additionally, the project site is not located in a tsunami or seiche zone. The risk of dam failure is extremely low and the potential for project inundation resulting from being located in a tsunami or seiche zone is low. Therefore, impacts would be less than significant.

As identified above under Section 3.10.1, the project site is designated as Zone X and Zone X (Shaded). The northern part of the project site is within an area of minimal flood hazard (Zone X); however, the majority of the project site is located in an area with a 0.2 percent annual chance of shallow flooding of less than one foot (Zone X Shaded) (FEMA 2021). In accordance with General Plan Policy EH-3.3, the proposed project would be constructed above the 100-year flood elevation. The proposed project would use clean imported fill to elevate the proposed pad grades for the homes and would have a minimum pad elevation of 13.0 feet NGVD and an average pad elevation of 14.2 feet NGVD. Therefore, the proposed project would be above the FEMA 100 year flood plain elevation of 9.3 feet which would provide the project site with flood damage protection. The proposed pad grades would exceed the City's standards for residential structures located in a flood hazard zone which requires a minimum pad elevation of 11.25 feet and would be above the BCDC minimum pad elevation recommendation for sea level rise guidance of 12.2 feet. Additionally, the proposed project's pad elevation would meet the minimum pad elevation requirements of the RPC-SAT's projected likely range for sea level rise by the year 2100 of 3.4 feet. Developing the proposed project to exceed the BCDC sea level rise guidance for minimum pad elevation would also ensure that the proposed project would not be inundated by potential sea level rise anticipated to affect the area in the future. Therefore, project inundation resulting from flood hazards would be a less than significant impact.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Water Quality Control Plan or Sustainable Groundwater Management Plan

Impact HYD-5 The proposed project could conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan.

Impact Analysis

The State Department of Water Resources identified the Niles Cone Groundwater Basin as a mediumpriority basin. The ACWD, who manages the groundwater of the Nile Cone Groundwater Basin, prepared and submitted an Alternative to a Groundwater Sustainability Plan in 2016 and was approved in 2019 by the DWR. The proposed project would comply with the Alternative to a Groundwater Sustainability Plan for the Niles Cone Groundwater Basin and would not conflict with or obstruct the implementation of a sustainable groundwater management plan.

The San Francisco Bay RWQCB adopted a Basin Plan that designates beneficial uses, establishes water quality objectives, and contains implementation programs and policies to achieve those objectors for all waters addressed through the Basin Plan. The proposed project would be constructed and operated in accordance with the Basin Plan. As required by Mitigation Measure HYD-1, the proposed project would obtain coverage under the NPDES General Construction Permit and prepare a SWPPP that includes BMPs to meet water discharge requirements. Additionally, if construction activities encounter groundwater, the proposed project would implement Mitigation Measure GEO-3 and prepare a dewatering plan in accordance with the discharge requirements of the RWQCB. Implementation of Mitigation Measures HYD-1 and GEO-3 would ensure that the proposed project would not conflict with any applicable water quality control plan. The proposed project would be constructed and operated in accordance with applicable water quality control plans and would not conflict with or obstruct implementation of any plans. Therefore, impacts would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measure HYD-1 and Mitigation Measure GEO-3 are required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.



Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.11 LAND USE AND PLANNING

This section describes the environmental and regulatory setting for land use and planning. It also describes existing conditions and potential impacts related to land use and planning that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.11.1 Environmental Setting

Regional Setting

The City of Newark covers an area of approximately 9,000 acres or approximately 14 square miles. The City is located between Interstate 880 and San Francisco Bay, south of State Route 84 and north of Stevenson Boulevard. The Don Edwards San Francisco Bay National Wildlife Refuge is located along the western perimeter of the City on the shore of San Francisco Bay and the freeways represent the northern and eastern limits of the City, separating Newark from surrounding Fremont. Approximately 1,800 acres of Newark's total area is in residential use, approximately 375 acres is in commercial use, 930 acres is in industrial or office-flex use, 250 acres is in public or institutional use, and 1,130 acres consists of roads and other ROWs. These areas represent approximately 50 percent of the land area of the City and the remaining 50 percent of Newark's land area consist of undeveloped land. Of the remaining 50 percent, approximately 960 acres is vacant and zoned for development, with 280 acres of "conservation" open space, and approximately 3,025 acres of salt evaporation ponds and ancillary facilities used for salt production (City of Newark 2013a).

Project Site Setting

The proposed project is located in the City of Newark, southwest of the intersection of Mowry Avenue and the UPRR tracks, west of Cherry Street. The project site is approximately 29 acres and consists of three parcels identified as APNs 537-0850-001-11, 537-0850-001-13, and 537-0850-002-00. The majority of the project site is currently developed as an auto part and scrap metal salvage yard. The auto part and scrap metal salvage lot, known as "Pick-n-Pull," that includes a 13,000 square foot warehouse, 1,500 square foot sales office, 3,000 square foot workshop, and a large parking area for storing vehicles that consists of crushed rock and asphalt. The northern parcel of the project site is currently undeveloped, agricultural land.

The undeveloped northern parcel of the project site is roughly triangular in shape and occupies an area of about 10 acres. Site topography is generally flat although fill has been placed in the central area of the undeveloped parcel. The surface elevation is about 10 feet AMSL around the perimeter of the parcel with a mound up to about 15 feet AMSL in the center. The surface elevation of the middle and southern parcels of the project site is about 10 feet AMSL along the northern property line where it abuts the undeveloped northern parcel as well as along the Mowry Avenue frontage and in the southwestern area of the salvage yard where the warehouse building is located. The topography of the parcels throughout the main yard area varies from about 10 feet AMSL at the west to 5 feet AMSL at the far east end of the yard.



Draft Environmental Impact Report Environmental Impact Analysis

Surrounding Land Uses

The project site is generally bounded to the northeast and east by the City's open space parcel, formerly used for agriculture. Mowry Avenue and ACFC&WCD Line B canal lies to the west. The property to the south and southwest, known as the Harwinder Singh site, was previously developed with one warehouse type structure near Mowry Avenue, and the site was used as an auto-wrecking yard. The building has since been demolished, and there are presently no buildings on the Harwinder Singh site. UPRR tracks are approximately 300 to 1,000 feet to the northwest of the project site. There is a pair of constructed water quality basins along the eastern boundary and at the southern tip of the project site. Additionally, Cargill owns and operates salt production ponds located west of the project site, and Mowry Slough begins approximately 1,500 feet southwest of the site.

General Plan Land Use Designation

The City's General Plan designated the project site as Low Density Residential. According to the General Plan, the allowable density of the Low Density Residential designation is less than 8.7 dwelling units per acre. The City's General Plan defines the Low Density Residential designation as intended for single-family residential development on lots larger than 5,000 square feet. It corresponds to most of Newark's residential neighborhoods. Multiple zoning districts apply within Low Density Residential areas to distinguish areas with different minimum lot sizes. Other compatible uses, such as schools, childcare centers, parks, and religious facilities may also locate in areas with this designation, subject to appropriate permitting requirements (City of Newark 2013a).

Zoning

The project site is zoned as Park. The project is proposing to rezone the project site from Park to PD-RS-6000.

The purpose of the RS residential single-family district is intended for residential densities up to 8.7 units per net acre. It provides for single-family residential developments on lots typically larger than 5,000 square feet. In addition to single-family homes, this district provides for other compatible uses, such as schools, childcare centers, parks, and community facilities that may be appropriate in a single-family residential neighborhood. This district implements the low density residential general plan land use designation.

The purpose of the PD Planned Development Overlay district is to provide one or more properties to be developed under a plan that provides for better coordinated development and incorporates development standards crafted to respond to site conditions in order to:

- A. Provide for greater flexibility in the design of the developments that is otherwise possible through the strict application of zoning district regulations;
- B. Ensure compliance with the general plan and provide various types of land use which can be combined in compatible relationship with each other as a part of a totally planned development; and
- C. Promote creativity in building design and innovation in development concepts.



Draft Environmental Impact Report Environmental Impact Analysis

Areas 3 and 4 Specific Plan

The project site is within the Areas 3 and 4 Specific Plan which is comprised of Area 3, encompassing 296 acres, and Area 4, encompassing 560 acres. Areas 3 and 4 are further divided into Sub Areas A through F. The proposed project lies within Sub Area D of Area 4 and is designated for a golf course or other recreational uses. The proposed project is not consistent with the land use identified by the Areas 3 and 4 Specific Plan. As such, the proposed project also requires a Specific Plan Amendment to change the use to residential single-family.

The Areas 3 and 4 Specific Plan allocated 1,260 residential units to be constructed within the Specific Plan area and designated specific parcels within the Specific Plan area for the development of these allocated residential units. The project site was designed for golf course or other recreational use development by the Specific Plan and though the Specific Plan's allocated 1,260 units have not yet all been developed, the Development Agreement for the Sanctuary West Project, which is located within the Specific Plan area east of the project site, were assigned the remaining allocated units within the Specific Plan area. Therefore, the development of the proposed project's 203 residential units would not be within the Specific Plan allocated residential units of 1,260 units, rather the proposed project's residential units would be above the allowed number of units for the Specific Plan area. As such, the approval of the proposed Specific Plan Amendment and rezoning of the project site would allow for the development of more units within the Specific Plan area above the planned number of units. The proposed project's 203 residential units and estimated number of residents resulting from development of the proposed project's RHNA.

3.11.2 Regulatory Setting

State

General Plans

The land use planning and zoning authority of local jurisdictions in California is set forth in the state's planning laws. California Government Code (GC) Section 65300, et seq. obliges cities and counties to adopt and implement general plans. The general plan is a comprehensive, long-term, and general document that describes plans for the physical development of a city or county and of any land outside its boundaries that, in the city's or county's judgment, bears relation to its planning. The general plan addresses a broad range of topics including, at a minimum, land use, circulation, housing, conservation, open space, noise, and safety. In addressing these topics, the general plan identifies the goals, objectives, policies, principles, standards, and plan proposals that support the city's or county's vision for the area. The general plan is a long-range document that typically addresses the physical character of an area over a 20-year period. Although the general plan serves as a blueprint for future development and identifies the overall vision for the planning area, it remains general enough to allow flexibility in the approach taken to achieve the plan's goals.

State Zoning Law

The State Zoning Law (California GC Section 65800, et seq.) establishes that zoning ordinances, which are laws that define allowable land uses within a specific district, are required to be consistent with the



Draft Environmental Impact Report Environmental Impact Analysis

general plan and any applicable specific plans. When amendments to the general plan are made, corresponding changes in the zoning ordinance may be required within a reasonable time to ensure the land uses designated in the general plan would also be allowable by the zoning ordinance (GC Section 65860, sub.[c]).

Local

San Francisco Bay Conservation and Development Commission

In 1969, the McAteer-Petris Act designated the San Francisco Bay BCDC as the agency responsible for the protection of San Francisco Bay and its natural resources. BCDC fulfills this mission through the implementation of the San Francisco Bay Plan, an enforceable plan that guides the future protection and use of San Francisco Bay and its shoreline.

As a permitting authority along the San Francisco Bay shoreline, BCDC is responsible for granting or denying permits for any proposed fill, extraction of materials, or change in use of any water, land, or structure within the Commission's jurisdiction. BCDC has jurisdiction for Mowry Slough ending at the culvert at the Mowry Avenue bridge crossing, at the bend of the channel near Plummer Creek, and jurisdiction over managed wetlands in the Southwest Newark Residential and Recreational Focus Area. Projects located within BCDC jurisdiction that involve Bay fill must be consistent with the Bay Plan policies on the safety of fills and shoreline protection (City of Newark 2013b).

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:

Goal LU-1: Quality of Life. Maintain a desirable quality of life in Newark by preserving a small town, neighborhood-oriented atmosphere and sustaining a balanced mix of land uses.

- Policy LU-1.2: Growth Focus Areas. Achieve a future growth pattern which includes new neighborhoods on vacant land along the southern and western edges of the city, and infill development in transit-served areas such as Old Town and the Greater NewPark Mall Area. Zoning and development review decisions should recognize these areas as the priority locations for growth and change over the next 20 years.
- **Policy LU-1.10: Vacant and Underutilized Sites.** Encourage the development of Newark's remaining vacant and underutilized sites for their highest and best use, consistent with the designations shown on the General Plan Diagram. Future growth in the city should generally be directed to areas identified in this General Plan.
- Policy LU-1.17: Sustainable Development Emphasis. Ensure that new development incorporates green building and sustainable design principles and encourage renovation of existing development to use water and energy more efficiently. Newark will reduce dependence on fossil fuels by citing homes, jobs, shopping, and services within walking distance of each other, and developing a circulation network that encourages walking, bicycling, and transit use.



Goal LU-4: Community Design and Identity. Enhance Newark's identity as a city of high-quality development that is distinctive from other cities in the Bay Area.

- **Policy LU-4.6**: **Streetscapes**. Ensure that medians, sidewalks, planting strips and other areas within the right-of-way of major thoroughfares are attractively landscaped and well maintained.
- **Policy LU-4.7**: Lighting. Manage exterior lighting to reduce potential light and glare impacts, improve public safety, and enhance the character of the streetscape. Exterior lighting includes streetlights for roads and parking areas, pedestrian lighting for sidewalks and walkways, building illumination, and accent lighting on special architectural and landscaping features. Lighting helps shape the character of the city and its neighborhoods through illumination level, light fixture type, finish, color, height, design, and location.
- Policy LU-4.13: Bayfront Identity. Reinforce Newark's identity as a bayfront city by orienting new development on the western and southern edges of the city toward the bay and shoreline areas. Future projects in these areas should enhance views to the water and wetlands and be compatible with the area's scenic and recreational qualities. The bayfront identity should be emphasized in gateways and public art as well.
- **Policy LU-4.14: View Protection**. Protect and enhance panoramic views and vistas of horizon features such as Coyote Hills, Mission Peak, the East Bay and Peninsula Hills, and San Francisco Bay.

Goal LU-7: Southwest Newark Residential and Recreational Project. Develop the Southwest Newark Residential and Recreation Project as one of the Silicon Valley's premier new neighborhoods, with executive housing and high-quality recreation.

- Policy LU-7.1: Southwest Newark Residential and Recreational Project (Areas 3 and 4 Development). Facilitate the development of 637 acres formerly known as the "Areas 3 and 4 project" consistent with previously approved plans for this area. The residential holding capacity of this area shall be 1,260 units.
- **Policy LU-7.2: Wetland Enhancement.** Create or enhance wetland habitat areas within nondeveloped portions of the Southwest Newark project area to offset loss of wetlands and aquatic habitat and provide additional habitat opportunities for rare plant and wildlife species.
- **Policy LU-7.3: Biological Resource Protection.** Maintain, protect, and enhance the natural biological resources of the Southwest Newark Residential and Recreational Project Area, particularly sensitive habitats and associated rare plants and animals, while integrating development and human activity. Disturbance of wetland and aquatic habitat should be avoided.
- **Policy LU-7.4: Controlling Invasive Plants**. Avoid the introduction and spread of non-native and invasive weeds as a result of development activities in this area. Require management plans to control the population of invasive species prior to grading, fill, and development activities.
- **Policy LU-7.5: Landscaping Palette**. Ensure that the choice of plants and landscaping in the Southwest Newark Residential and Recreational Project responds to soil conditions, wind conditions, and the cooler bayside climate. Landscaping should reinforce vista points, create



Draft Environmental Impact Report Environmental Impact Analysis

variations in textures and color, define circulation routes and pathways, and include features which provide a strong sense of identity.

- Policy LU-7.6: Open Space Amenities. Include a major open space and recreational amenity within the Southwest Newark Residential and Recreational Project boundary. The preferred amenity is an 18-hole golf course with clubhouse. The former solid waste disposal site at the west end of Mowry Avenue should be considered for inclusion in the Golf Course site. In the event a golf course in deemed infeasible, then another recreational use that is acceptable to the city shall be provided through developer fees. In addition, development in this area shall provide for neighborhood parks consistent with the ratios established by the General Plan.
- **Policy LU-7.7: Maintaining Hydrologic Features.** Maintain the natural hydraulic features of the Southwest Newark Residential and Recreational Project to the extent feasible and maintain or improve the current quality of water leaving the site.
- **Policy LU-7.8: Mitigating Construction Impacts**. Avoid and mitigate construction impacts on wetlands, aquatic habitat, wildlife, and water quality as development takes place in the Southwest Newark Residential Recreational Project. Measures to minimize such impacts should be included in project approvals, consistent with state and federal agency oversight and regulations.
- **Policy LU-7.9: Inclusionary Housing.** Address inclusionary housing requirements consistent with the Area 3 and 4 Development Agreement.

Newark Zoning Code

The Newark Zoning Code contains the City's Official Zoning Map that delineates the boundaries of zoning designations within the City and regulations that govern the use of land and placement of buildings and improvements within the various classes of districts. The purpose of the Newark Zoning Code is to implement the City's General Plan and to protect and promote the public health, safety, peace, comfort, convenience, prosperity and general welfare. The Zoning Code is included as Title 17 of the City's Code of Ordinances. Chapter 17.07 described the purpose, regulations, and standards of the residential zoning districts and Chapter 17.12 described the regulations and standards of the Planned Development overlay district.

3.11.3 Environmental Impacts

This section analyzes the project's potential to result in significant land use and planning impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The analysis of potential land use impacts considers the project's consistency with adopted plans and policies that regulate land use on the project site, and the project's compatibility with surrounding land uses. The determination of consistency with applicable land use policies and ordinances is based upon a review of the previously identified planning documents that regulate land use or guide land use decisions pertaining to the project site. CEQA Guidelines section 15125(d) requires that an EIR discuss



Draft Environmental Impact Report Environmental Impact Analysis

inconsistencies with applicable plans that the decision-makers should address. Evaluations are made to determine whether a project is consistent with such plans. Projects are considered consistent with regulatory plans if they are compatible with the general intent of the plans and would not preclude the attainment of their primary goals. The intent of the consistency evaluation is to determine if noncompliance with regulatory plans would result in a significant impact. The impact analysis was based on a review of the General Plan and the Areas 3 and 4 Specific Plan to identify planned land uses and policies applicable to the project. Existing land uses were determined from site reconnaissance and General Plan designations. The City's zoning regulations were also reviewed to determine the project's consistency with existing zoning.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's impacts to land use and planning are significant. Would the proposed project:

- Physically divide an established community?
- Cause a significant environmental impact due to a conflict with any land use plan, policy, or regulation adopted for the purpose of avoiding or mitigating an environmental effect?

Project Impact Analysis and Mitigation Measures

Established Community

Impact LU-1 The proposed project would not physically divide an established community.

Impact Analysis

The project site is currently developed with an auto parts and scrap metal salvage yard. The project site is bounded to the northeast and east by the City's open space parcel, and Mowry Avenue and ACFC&WCD Line B canal lies to the west. The property to the south and southwest, known as the Harwinder Singh site, was previously developed with a warehouse-type structure and was previously used as an auto-wrecking yard; however, the building has since been demolished and there are no buildings presently on the Harwinder Singh site. UPRR tracks are approximately 300 to 1,000 feet to the northwest of the project site. The project site and surrounding areas are not developed as an established community and the proposed project would not include the construction of new roadways that could divide an established community and there would be no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Conflict with Plans, Policies, or Regulations

Impact LU-2The proposed project would not cause a significant environmental impact due
to a conflict with any land use plan, policy, or regulation adopted for the purpose
of avoiding or mitigating an environmental effect.

Impact Analysis

General Plan Consistency Analysis

The City's General Plan designates the project site as Low Density Residential. The Low Density Residential designation of the General Plan has an allowable density of less than 8.7 dwelling units per acre. The proposed project would be consistent with this requirement and construct 203 single-family detached units, resulting in a density of 7 units per acre. The analysis in Table 3.11-1 demonstrates that the proposed project would not create inconsistencies with the applicable policies of the General Plan.

Table 3.11-1: General Plan Consistency Analysis

Policy	Consistency Analysis		
Land Use			
Goal LU-1: Maintain a desirable quality of life in Newark by preserving a small town, neighborhood-oriented atmosphere and sustaining a balanced mix of land uses.	Consistent. The proposed project would construct a new community of single-family homes with a neighborhood-oriented atmosphere which would include private streets, on-site open space areas, and landscaping throughout the project site.		
Policy LU-1.14: Cost Recovery. Ensure that new development generates sufficient revenue and pays its reasonable share to offset its cost impacts on public services and facilities.	Consistent. The proposed project would pay all associated fees to offset impacts on public services and facilities such as the parks impact fee and development impact fees.		
Policy LU-1.17: Sustainable Development Emphasis. Ensure that new development incorporates green building and sustainable design principles and encourage renovation of existing development to use water and energy more efficiently. Newark will reduce dependence on fossil fuels by citing homes, jobs, shopping, and services within walking distance of each other, and developing a circulation network that encourages walking, bicycling, and transit use.	Consistent. The proposed project would incorporate sustainability features into the project design such as using low water use plants for landscaping, constructing storm drain systems which would utilize LID techniques, and incorporating bioretention areas into the project design to provide on-site treatment and retention of stormwater. Additionally, the proposed project would be constructed to be 100 percent electric.		
Goal LU-2: Ensure compatibility between adjacent uses.	Consistent. The project site's surrounding lands are not developed with significant existing uses. Adjacent land uses include an open space parcel, salt production ponds, and a parcel previously developed as an auto- wrecking yard that has since been demolished. The proposed project would introduce new land uses to the project area. However, the proposed project would be designed and constructed to ensure compatibility with adjacent uses.		
Policy LU-2.4: Buffering from Transportation Facilities. Ensure that the design of new residential development near rail lines, truck routes, freeways, or major thoroughfares includes setbacks, landscape screening, and other provisions to minimize exposure to negative impacts such as noise and air pollution.	Consistent. UPRR tracks are located approximately 300 to 1,000 feet to the northwest of the project site and the project site does not directly border any transportation facilities. The proposed project would be adequately set back from the UPRR tracks and would include construction of a new crossing across the UPRR tracks and a mid-block crossing to the Silliman		

Policy	Consistency Analysis
	Center. Landscaping would be provided throughout the project site to provide screening of the UPRR tracks which would minimize exposure to existing transportation facilities.
Goal LU-4: Enhance Newark's identity as a city of high quality development that is distinctive from other cities in the Bay Area.	Consistent. The proposed project would construct a new single-family residential community and would be designed in accordance with the City's design guidelines to provide high quality development for the City.
Policy LU-4.6: Streetscapes. Ensure that medians, sidewalks, planting strips and other areas within the right-of-way of major thoroughfares are attractively landscaped and well maintained.	Consistent. The proposed project would construct new streets and construct improvements to existing roadways. Roadways would be designed and constructed in accordance with City standards and landscaping would be provided along roadways in accordance with City Zoning Code Chapter 17.21 which includes Citywide regulations related to landscaping.
Policy LU-4.7: Lighting. Manage exterior lighting to reduce potential light and glare impacts, improve public safety, and enhance the character of the streetscape.	Consistent. The proposed project would design and construct lighting in accordance with the City's lighting standards such as providing shielding and ensuring light and glare from the proposed project does not affect adjacent properties. Lighting installed for the proposed project would be constructed in accordance with City Zoning Code Chapter 17.17.060 which includes lighting and illumination regulations.
Goal LU-7: Develop the Southwest Newark Residential and Recreational Project as one of the Silicon Valley's premier new neighborhoods, with executive housing and high quality recreation.	Consistent. The proposed project would construct a new high quality residential community with 203 single-family homes within Area 4 of the Southwest Newark Residential and Recreational Project. The proposed project would also provide approximately 4.89 acres of on-site open space located within the project site that would provide recreational opportunities.
Policy LU-7.1: Southwest Newark Residential and Recreational Project (Area 3 and 4 Development). Facilitate the development of the 637 acres formerly known as "The Area 3 and 4 project" consistent with previously approved plans for this area. The residential holding capacity of this area shall be 1,260 units.	Consistent. The proposed project would construct 203 new single-family residential units in Area 4 of the Southwest Newark Residential and Recreational Project. The proposed project includes a Specific Plan Amendment and rezoning to allow the project site to be developed with residential uses and provide additional residential units above the holding capacity specified in the Specific Plan. With the approval of the Specific Plan Amendment and rezoning, the proposed project would be consistent with this policy.
Policy LU-7.6: Open Space Amenities. Include a major open space and recreational amenity within the Southwest Newark Residential and Recreational Project boundary. The preferred amenity is an 18-hole golf course with clubhouse. The former solid waste disposal site at the west end of Mowry Avenue should be considered for inclusion in the Golf Course site. In the event that a golf course is deemed infeasible, then another recreation use that is acceptable to the city shall be provided through developer fees. In addition, development in this area shall provide for neighborhood parks consistent with the ratios established by the General Plan.	Consistent. Though the proposed project would not include the development of a golf course, the proposed project would construct a new residential community that would include 4.89 acres of open space that would provide limited recreational opportunities to the residents of the proposed project. In addition to the on- site open space, the proposed project would pay a parks impact fee to ensure consistency with the parkland standard ratio established by the City.



Policy	Consistency Analysis
Policy LU-7.8: Mitigating Construction Impacts. Avoid and mitigate construction impacts on wetlands, aquatic habitat, wildlife, and water quality as development takes place in the Southwest Newark Residential and Recreational Project. Measures to minimize such impacts should be included in project approvals, consistent with state and federal agency oversight and regulations.	Consistent. The proposed project would include Mitigation Measure HYD-1 which requires the preparation and implementation of a SWPPP with BMPs identified to protect water quality. Additionally, the proposed project would implement Mitigation Measures BIO-1 through BIO-7 which would ensure there would be no construction related impact to wildlife, wetlands, or aquatic habitats.
Transp	ortation
Goal T-2: Create a citywide pedestrian and bicycle network that provides safe access to destinations within the city, connects to an integrated regional network, and is accessible to users of all ages, abilities, and means.	Consistent. The proposed project would construct bicycle lanes and crosswalks along Mowry Avenue to provide safe access from the project site to the surrounding areas.
Policy T-2.6: Pedestrian and Bicycle Provisions within New Development. Ensure safe and convenient pedestrian and bicycle access to and through new public and private developments. The City will use the development review process to ensure – and where appropriate to require – provisions for pedestrians and bicycles in new development areas.	Consistent. See discussion for Goal T-2.
Conservation ar	nd Sustainability
Goal CS-1: Protect Newark's natural environment, landscape, and physical features.	Consistent. The proposed project would be designed, constructed, and operated in accordance with all applicable City standards and regulations to protect Newark's natural environment, landscape, and physical features.
Policy CS-1.1: Environmental Impacts of Development. Ensure that development minimizes its impacts on Newark's environment and natural resources through sound planning, design, and management.	Consistent. The proposed project would be designed to minimize impacts on Newark's environment and natural resources. The proposed project would include Mitigation Measure HYD-1 which requires the implementation of a SWPPP to ensure the proposed project does not result in significant impacts to water quality. Additionally, the proposed project would implement Mitigation Measures BIO-1 through BIO-9 to protect natural resources within and around the project site.
Policy CS-1.4: Soil Erosion. Identify and eliminate erosion problems on public and private lands. The potential for erosion should be considered as a design and engineering factor in new development.	Consistent. The proposed project includes Mitigation Measure GEO-1 which requires the preparation of geotechnical studies to consider geological impacts such as erosion. Additionally, the proposed project includes Mitigation Measure HYD-1 which requires the implementation of a SWPPP which would include BMPs to control erosion and the discharge of sediment during construction.
Goal CS-2: Conserve Newark's wetlands and baylands.	Consistent. The proposed project would not include construction in wetland areas and would take steps to ensure wetland areas located adjacent to the project site are not affected.
Policy CS-2.1: Wildlife and Habitat Protection. Preserve and protect Newark's plant and animal species and habitats, including wetlands, salt marshes, creeks, and lakes. Ensure that land use decisions avoid and	Consistent. The proposed project is located adjacent to identified wetland areas. The proposed project would not include construction in the wetland areas and would include measures within the project to protect wildlife habitat.



Policy	Consistency Analysis	
mitigate potential impacts on wildlife habitat to the extent possible.		
Policy CS-2.2: Special Status Species. Ensure that adverse impacts on special status species, including those deemed rare, threatened, endangered, or candidate species for protection, are avoided and mitigated to the greatest extent feasible as development takes place.	Consistent. The proposed project would be required to implement Mitigation Measures BIO-1 through BIO-6 identified in Section 3.4, Biological Resources which include pre-construction mitigation for special-status species. Additionally, the proposed project would implement Mitigation Measure BIO-8, which include post-construction mitigation for the protection of special-status species. Implementation of mitigation measures identified in Section 3.4 would ensure that there are no significant impacts to special status species	
Policy CS-2.5: Development Near Wetlands. Manage land use and development of upland sites in a manner that minimizes off-site impacts to nearby wetlands.	Consistent. See discussion for Goal CS-2 and Policy CS-2.1.	
Goal CS-3: Conserve and enhance Newark's water resources.	Consistent. The proposed project would be constructed and operated in accordance with the City's water resource protection standards to conserve Newark's water resources.	
Policy CS-3.1: Protection of Water Resources. Ensure that land use decisions consider the availability of water for domestic and non-domestic uses, potential impacts on groundwater quality and groundwater recharge capacity, and potential off-site impacts on water quality.	Consistent. The proposed project's impacts to water resources is analyzed in this EIR. See Section 3.10 Hydrology and Water Quality and Section 3.19 Utilities and Service Systems for a detailed analysis of potential impacts to water resources. The proposed project would include Mitigation Measure HYD-1 which would ensure that there would be no potential impacts to water quality from construction of the proposed project. In addition to Mitigation Measure HYD-1, the proposed project would include remediation of onsite contaminated surface waters and groundwater. Remediation of contaminated surface waters and groundwater would ensure that there would be no potential impacts to water quality from existing contamination. Additionally, Mitigation Measure GEO-3 would ensure that discharge of contaminated groundwater does not occur during dewater activities. The proposed project would utilize storm drain systems with LID techniques incorporated and permeable paving materials which would increase the potential for groundwater recharge at the site. The proposed project would provide treatment of site runoff prior to it being discharged which would ensure that the proposed project would not result in water quality impacts.	
Policy CS-3.2: Water Conservation Standards. Promote water conservation through development standards, building requirements, irrigation requirements, landscape design guidelines, and other applicable City policies and programs.	Consistent. The proposed project would be designed and constructed in accordance with the City's water conservation standards. The proposed project would include the use of low water use plants for landscaping and storm drain systems with LID techniques incorporated to meet water conservation standards.	
Policy CS-3.4: Reducing Water Pollution. Protect the quality of Newark's surface waters by supporting controls of point source and non-point sources of pollution.	Consistent. The proposed project would include the preparation and implementation of an SWPPP, as required by Mitigation Measure HYD-1, which would include pollution prevention measures.	



Policy	Consistency Analysis
Goal CS-5: Reduce greenhouse gas emissions in Newark and make reduction of the City's carbon output a high priority.	Consistent . The proposed project would be designed and constructed in accordance with City standards and regulations for greenhouse gas and would include construction of features such as pedestrian and bicycle systems to encourage use of alternative transportation methods and reduce greenhouse gas emissions.
Policy CS-5.2: Pedestrian and Bicycle Friendly Design. Ensure that new development is planned and designed to facilitate walking and bicycling as well as driving. This can potentially reduce the number of vehicle trips and related greenhouse gas emissions.	Consistent. The proposed project would include improvements to Mowry Avenue to provide safe pedestrian and bicycle systems from the project site to the surrounding areas. Providing safe pedestrian and bicycle systems from the project site to surrounding areas would promote and facilitate the use of alternative transportation systems and reduce the use of personal vehicles resulting in a reduction of vehicle trips and related GHG emissions.
Policy CS-5.8: Planning for Sea Level Rise. Require proposed development in low-lying areas to comply with applicable City of Newark standards for construction in flood hazard zones.	Consistent. The proposed project involves the import of approximately 252,000 CY of clean fill to elevate the proposed pad grades for the homes above the 100 year flood plain elevation. The proposed pad grades would exceed the City's standard for construction in flood hazard zones and would accommodate the San Francisco Bay BCDC's currently adopted sea level rise guidance.
Goal CS-6: Reduce the impacts of buildings and development on greenhouse gas levels and the environment in general.	Consistent. As identified in Section 3.8, Greenhouse Gases, the proposed project would be consistent with the City's CAP and the California Climate Change Scoping Plan. Additionally, the proposed project would be designed and constructed in accordance with Title 24 CALGreen standards and requirements for energy efficiency. Additionally, the proposed project's structures would be 100 percent electric.
Policy CS-6.2: Encouraging Greener Construction. Encourage greener construction methods and greater use of recycled-content materials in new residential, commercial, and industrial construction projects.	Consistent. The proposed project would be constructed in compliance with CALGreen requirements.
Policy CS-6.5: Minimizing Impervious Surface Coverage. Minimize impervious surface coverage and related stormwater runoff in new development areas by allowing narrower roads and shared driveways and parking areas. Other means of reducing urban runoff, such as rain barrels and bioswales, also should be encouraged.	Consistent. The proposed project would include bioretention areas located within the project site to reduce urban runoff and minimize impervious surface coverage. The proposed project would include features within the project design to minimize impervious surface coverage. The proposed project would incorporate 465,680 square feet of pervious surfaces at the project site consisting of landscaped areas and biotreatment areas. Additionally, the proposed project would utilize permeable pavement materials.
Parks, Recreation	, and Open Space
Goal PR-2: Expand and improve Newark's parks and recreational facilities to meet existing and future needs.	Consistent. The proposed project includes the enhancement of open space areas around the stormwater pond areas for passive recreation.
Policy PR-2.2: Parks in New Development. Require new parks to be provided within large-scale new development. Where the provision of an on-site park is infeasible, require the payment of an in-lieu fee for parkland acquisition to serve that development.	Consistent. The proposed project would develop on- site open space areas located within the project site. However, the proposed project would pay the full parks impact fee to offset increased demand to park facilities from the new development.

 \bigcirc
Policy	Consistency Analysis
Policy PR-2.4: Pocket Parks. Allow a portion of the parkland dedication requirement to be met through the provision of on-site pocket parks and play lots in new development.	Consistent. The proposed project would develop on- site open space areas within the project site which would provide amenities such as a lawn, pedestrian path, and picnic tables. However, as the on-site open space areas would not rise to the level of a park due to limited development, the proposed project would pay the full parks impact fee to offset increased demand to park facilities.
Environmer	ntal Hazards
Goal EH-1: Reduce the potential for injury, harm, property damage, and loss of life resulting from environmental hazards.	Consistent. The proposed project would be designed and constructed in accordance with the CBC and the City's development standards to protect against environmental hazards. Additionally, the proposed project would implement Mitigation Measure GEO-1 which requires the implementation of recommendations includes in the geotechnical studies which include detailed investigation of ground shaking, liquefaction, soil stability and other geologic hazards. Additionally, the proposed project would include implementation of Mitigation Measure HAZ-1 which require remediation of existing on-site contamination at the project site. The Applicant would be required to implement the Corrective Action Plan and Remedial Excavation Work Plan which provides detailed plans to remediate contaminated soil, soil gas, groundwater, surface water, and sediments through a combination of removing the soil contamination through excavation, groundwater contaminated soil, not provide a tenuation, and surface water contamination through dewatering, and if conditions warrant, natural attenuation, to residential standards. Compliance with identified mitigation measures would ensure there would be no significant impacts from environmental hazards.
Policy EH-1.1: Development Regulations and Code Requirements. Establish and enforce development regulations and building code requirements to protect residents and workers from flooding, liquefaction, earthquakes, fires, and other hazards.	Consistent. See discussion for Goal EH-1. The proposed project would be designed and constructed in accordance with the CBC and the City's development regulations to protect against environmental hazards. Additionally, the proposed project would implement Mitigation Measure GEO-1 which requires the implementation of recommendations includes in the geotechnical studies which include detailed investigation of ground shaking, liquefaction, soil stability and other geologic hazards. Implementation of recommendations would be completed in accordance with established development regulations and CBC requirements.
Policy EH-1.5: Adequacy of Access. Require adequate access and clearance for fire equipment, fire suppression personnel, and evacuation for new development.	Consistent. The proposed project would be designed to provide adequate access throughout the site for emergency vehicles and personnel. The proposed project would be developed in accordance with City standards for emergency access.
Goal EH-2: Reduce risks to life and property associated with geologic hazards.	Consistent. The proposed project would be designed and constructed with geologic hazard mitigation measures to reduce risks associated with geologic hazards. Mitigation Measure GEO-1 and GEO-2 would require implementation of recommendations included in

Policy	Consistency Analysis
	the previously prepared geotechnical studies to ensure the proposed project is constructed to withstand geologic hazards thereby reducing the risks to life and property.
Policy EH-2.1: Earthquake Safety in New Construction. Require new development to meet structural integrity standards which minimize the potential for damage during earthquakes.	Consistent. The proposed project would be constructed in accordance with the earthquake design parameters of the CBC and would be constructed to minimize potential earthquake related impacts. Additionally, the proposed project would implement Mitigation Measure GEO-1 which would require implementation of recommendations included in the geotechnical studies prepared for the proposed project which include design recommendations to minimize potential damage during earthquakes.
Goal EH-3: Reduce the risks to life and property associated with flooding.	Consistent. The proposed project would be constructed to have elevated pad grades above the 100-year flood plain elevation as required by the City for new development located in a flood plain.
Goal EH-3.3: Residential Development in the Flood Plain. Require that new residential development, including streets and other surface improvements, be constructed above the 100-year flood elevation.	Consistent. See discussion for Goal EH-3.
Policy EH-3.7: Mitigating Downstream Flood Impacts. Design new development to reduce the potential for downstream flooding. Measures such as porous pavement and on-site drainage retention facilities should be considered to reduce downstream flooding.	Consistent. The proposed project would include storm drain systems with LID techniques incorporated and would construct bioretention areas within the project site to manage site runoff.
Goal EH-7: Ensure that new structures/uses are designed and constructed to preclude excessive, inappropriate, and undesirable noise effects.	Consistent. The proposed project would implement Mitigation Measure NOI-1 which requires preparation of a project-specific acoustical study to determine necessary noise control treatments to reduce interior noise levels. Additionally, the proposed project would implement Mitigation Measure NOI-2 which requires noise from all mechanical equipment associated with the project to comply with requirements identified in Newark Municipal Code Section 17.24.100, Paragraph A.2.a.
Policy EH-7.3: Reducing Exposure to Operational Noise. In new residential and mixed-use developments, require that stationary equipment (such as air conditioning units and condensers) be placed in separate spaces, rooftops, or other areas such that noise impacts to interior living areas will be reduced. Similarly, potentially noisy common areas, such as trash collection areas and loading zones, should be located away from residential units or other noise- sensitive spaces.	Consistent. The proposed project would implement Mitigation Measure NOI-2 which requires noise from all mechanical equipment associated with the project to comply with requirements identified in Newark Municipal Code Section 17.24.100, Paragraph A.2.a. Additionally, the proposed project would prepare a noise analysis once on-site equipment is selected and equipment would be designed to incorporate measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels.
Policy EH-7.6: New Noise Sources. Require new developments that have the potential to create long-term noise increases to mitigate potential impact to off-site receptor properties.	Consistent. See discussion for Policy EH-7.3.
Policy EH-7.7: Acoustical Study Requirements. Require acoustical studies for new developments in areas where noise levels exceed the 'normally acceptable' levels for the proposed land use. For residential uses	Consistent. Mitigation Measure NOI-1 identified in Section 3.13, Noise, requires the preparation of a project-specific acoustical study in accordance with General Plan Policy EH-7.7 to determine necessary

 \bigcirc

Policy	Consistency Analysis	
the analysis should include mitigation measures to limit the noise exposure in interior living spaces to 45 dB Ldn, consistent with California Title 24.	noise control treatments to reduce the impact of traffic noise on the interior of the residential units. Additionally, as identified in the discussion under Policy EH-7.3, the proposed project would implement Mitigation Measure NOI-2 which requires the preparation of a noise analysis once on-site equipment is selected and equipment would be designed to incorporate measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels	
Health and	d Wellness	
Goal HW-1: Air quality that meets state and federal standards and provides improved respiratory health for Newark residents.	Consistent. Section 3.3, Air Quality, identified that the proposed project would not result in air quality impacts and the proposed project would meet state and federal standards.	
Policy HW-1.3: Reducing Exposure to Air Pollution in New Development. Use site planning and architectural design to reduce potential exposure of sensitive uses to major air pollution sources, including freeways and industrial activities.	Consistent. Section 3.3, Air Quality, identified Mitigation Measures AIR-1 and AIR-2 to be implemented during construction to ensure that development of the proposed project would not result in substantial air pollution. Additionally, the proposed project would not be a major air pollution source nor is the project site located near major air pollution sources.	
Goal HW-7: Safe and secure neighborhoods and public spaces.	Consistent. The proposed project would construct a new residential community with 203 single-family homes. The proposed project would construct lighting throughout the site to illuminate the main entrances of the single-family homes, private streets, sidewalks, common space areas and driveways for security and safety purposes.	
Policy HW-7.1: Eyes on the Street. Design new development to encourage "eyes on the street" and discourage the potential for criminal activity.	Consistent. The proposed project would be designed with lighting provided throughout the site to ensure the site is well illuminated during nighttime and discourage the potential for criminal activity. Lighting installed for the proposed project would be constructed in accordance with City Zoning Code Chapter 17.17.060 which includes lighting and illumination regulations.	
Policy HW-7.2: Development Lighting. Require lighting plans for new development that ensures that common spaces and parking areas illuminated in a way that improves public safety.	Consistent. The proposed project would be designed and constructed in accordance with the City's lighting standards. Lighting installed for the proposed project would be constructed in accordance with City Zoning Code Chapter 17.17.060 which includes lighting and illumination regulations.	
Community Services and Facilities		
Goal CSF-2: Provide excellent schools that deliver high-quality educational services to Newark students while serving as neighborhood centers and fostering civic pride.	Consistent. The proposed project would pay a school impact fee to provide funding for schools within the City.	
Policy CSF-2.2: Mitigation of School Impacts. When new residential development is approved, require mitigation of school impacts to the full extent permitted by law. Work collaboratively with the Newark Unified School District to ensure the appropriate fees are collected and other appropriate mitigation measures are taken.	Consistent. The proposed project would pay a school impact fee to NUSD to mitigate impacts to schools from the proposed development.	



Draft Environmental Impact Report Environmental Impact Analysis

Policy	Consistency Analysis
Goal CSF-5: Provide safe, reliable, and efficiently operated infrastructure which meets Newark's long-term water, sewer and stormwater management needs.	Consistent. The proposed project would construct improvements to the existing water, sewer, and stormwater infrastructure along Mowry Avenue.
Policy CSF-5.5: Drainage within New Development. Ensure that new development provides drainage and flood protection improvements which reduce on-site and downstream hazards such as ponding, flooding, and erosion. New development areas should be designed to minimize impervious surfaces in order to reduce associated site runoff and maximize groundwater recharge.	Consistent. The proposed project would install a storm drain system on-site consisting of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes to manage site runoff. The storm drainage system would be designed to include LID techniques which may include directing roof runoff to vegetated areas, storm drain stenciling, and site design that promotes infiltration. Additionally, the proposed project would also involve off-site stormwater improvements along Mowry Avenue and would construct bioretention treatment areas along Mowry Avenue.

Specific Plan Consistency Analysis

The project site is located within the Areas 3 and 4 Specific Plan and the project site is designated for a golf course or other recreational uses. The proposed project is not consistent with the land use identified by the Areas 3 and 4 Specific Plan. As such, the proposed project also requires a Specific Plan Amendment to change the existing designated golf course or recreational use to a residential single-family use. While the Specific Plan designation would change following the approval of the proposed project, the change would result in a better alignment of the project site with the General Plan designation and the project site would be developed for residential uses, as planned by the General Plan. Additionally, the Specific Plan Amendment would allow for the development of additional residential units above the allocated maximum capacity of residential units for the Specific Plan area identified in the Specific Plan.

A detailed analysis of the proposed project's consistency with applicable goals and policies of the Areas 3 and 4 Specific Plan is provided in Table 3.11-2 to identify whether the proposed project would create an inconsistency with the General Plan. The analysis in Table 3.11-2 demonstrates that the proposed project would not create inconsistencies with the applicable policies of the Specific Plan.

Policy	Consistency Analysis
Goal: Maintain, protect and enhance the planning areas' natural biological resources particularly sensitive habitats and associated rare plants and animals while integrating development and human uses.	Consistent. The proposed project would be required to implement Mitigation Measures BIO-1 through BIO-9 identified in Section 3.4, Biological Resources of this EIR. Implementation of mitigation measures identified in Section 3.4 would ensure that construction and operation of the proposed project would not result in impacts to biological resources.
Policy 6-3: Development of the golf course should contain as much natural habitat as is feasible, such as unmaintained native grassland areas rather than turf	Not Applicable. Though the project site was identified for development as a golf course in the Areas 3 and 4 Specific Plan, the project proposes a Specific Plan Amendment to develop the site for residential uses.



Policy	Consistency Analysis
and native trees and other vegetation where appropriate.	The Specific Plan Amendment would better align the project site with the site's General Plan land use designation and the project site would be developed with residential uses, as planned by the General Plan. Additionally, the Specific Plan Amendment would allow for the development of additional residential units above the allocated maximum capacity of residential units for the Specific Plan area identified in the Specific Plan.
Policy 6-5: To maintain hydrology and water quality as currently exists in natural habitat areas, development of the golf course should use state of the art management methods such as a computerized irrigation system connected to an on-site weather station to limit watering to the exact needs of the course, sprinkler head designs to ensure a very even distribution of water to reduce water use and runoff, unmaintained native grasses in the outer roughs, designated irrigated and non-irrigated areas, retention of runoff (particularly off-season) within the golf course, accurate application of fertilizer to that required to eliminate contaminated runoff and retention of nuisance or off-season flows within the development area.	Not Applicable. See discussion under Policy 6-3.
Policy 6-6: Development of residential and golf course areas should be configured to optimize habitat areas (e.g., contiguous and large) for wildlife in remaining or preserved wetlands to provide needed habitat elements; limit disturbance from residences, the golf course, and recreational activities (e.g., hiking or dog walking along levees); avoid, to the extent feasible, or replace and enhance habitat for endangered species habitat lost; and allow for adequate movement for wildlife species within Area 4 with particular attention paid to waterbirds and special-status species found in the area: burrowing owls, peregrine falcons, tricolored blackbirds (colonies), salt marsh harvest mice, salt marsh wandering shrews, pallid bats, and Yuma bats and breeding northern harriers, Alameda song sparrows, Bryant's savannah sparrows, and San Francisco common yellowthroats.	Not Applicable. See discussion under Policy 6-3. However, development of the project site would implement Mitigation Measures BIO-1 through BIO-9 to ensure that development does not result in significant impacts to biological resources. Mitigation Measure BIO-1 through BIO-9 include both pre-construction and post-construction measures to protect special-status species and other natural resources in the area.
Policy 6-7: Temporary disturbance to all wetland and aquatic habitat should be avoided to the maximum extent feasible during construction activities using measures such as demarcation of construction areas with Environmentally Sensitive Area fencing	Consistent. As identified in the Biological Resources Technical Report (Appendix C), the project site does not include wetlands. Additionally, as identified in Section 3.4, Biological Resources, the potentially jurisdiction aquatic features located within or near the project site were artificially constructed and are not regulated by USACE, RWQCB, or CDFW. The proposed project would require extension of unity infrastructure through the ACFC&WCD Line D channel which is a jurisdiction aquatic feature. However, the extension would be completed with by jack-and-bore that would go underneath and avoid the Line D channel and therefore, construction of extension of the utility infrastructure would not disturb any wetland or aquatic habitat. The proposed project would not disturb any wetlands or aquatic habitats.
Policy 6-8: Minimize construction related impact on rare, threatened, endangered or other special-status species	Consistent. The proposed project would be required to implement Mitigation Measures BIO-1 through BIO-6



Draft Environmental Impact Report Environmental Impact Analysis

Policy	Consistency Analysis
particularly in natural, created or enhanced habitat areas remaining or preserved on-site such as burrowing owls, salt marsh harvest mice, salt marsh wandering shrews, pallid and Yuma bats, and nesting northern harriers, peregrine falcons, Alameda song sparrows, Bryant's savannah sparrows, San Francisco common yellowthroats, and tricolored blackbird colonies. Measures may include conducting pre-construction/pre- disturbance surveys, establishing buffer zones, avoiding habitat, creating alternate habitat, salvaging individuals, and during the breeding season: avoiding construction activities, excluding individuals from construction areas, removal of vegetation.	identified in Section 3.4, Biological Resources of this EIR which include pre-construction mitigation for special-status and sensitive wildlife species. Implementation of mitigation measures identified in Section 3.4 would ensure that construction of the proposed project would not result in impacts to biological resources.
Policy 6-9: Minimize construction related impact water quality degradation in natural, created or enhanced habitat areas remaining or preserved on-site using measures such as incorporating best management practices, minimizing soil disturbance adjacent to wetland and marsh habitat, suppressing dust during construction, and avoiding contamination of adjacent natural habitats during environmental cleanup of the auto wrecking yards.	Consistent. As required by Mitigation Measure HYD-1, the proposed project would include the preparation and implementation of an SWPPP which would include BMPs and measures to minimize construction related impacts to water quality.
Policy 6-10: The City of Newark shall require design and implementation of and must review and approval an Invasive Plant Species Management Plan prior to grading or importation of fill material as part of any proposed development in Specific Plan Areas 3 and 4 to reduce the potential establishment or spread of non- native, invasive weed populations as a result of development activities. This management plan will outline methods to control the existing populations of non-native, invasive species that are not a severe ecological threat and to remove those weed species present that pose a severe ecological threat from the accessible portion of the site to prevent the spread of their seed during and after construction and to prevent the invasion of graded area by invasive species.	Consistent. The proposed project would be required to implement Mitigation Measure BIO-7, identified in Section 3.4, Biological Resources of this EIR, which requires the preparation and implementation of an Invasive Species Management Plan. The plan would be submitted to the City for approval prior to the issuance of any building or grading permits.
Policy 6-11: The design of the golf course should minimize, to the extent practicable, disturbance by golfers of adjacent sensitive natural resources such as sensitive habitats, vegetation wildlife, and rare plant or animals with such measures as having high-use areas such as tees and greens set back from the edge of the golf course, broad rough/out-of-bounds areas along the interface between the golf course and sensitive habitats, "out of bounds" areas clearly marked, and focused lighting that does not extend into natural or habitat areas.	Not Applicable. See discussion under Policy 6-3. However, the proposed project would be designed to minimize disturbance of adjacent sensitive natural resources.

Newark Zoning Code Consistency

The Newark Zoning Code designates the project site as Park (City of Newark 2013a). The project is proposing to rezone the project site from Park to PD-RS-6000. The rezoning request is to better align the zoning with the proposed use of the project and the existing General Plan designation. Additionally, the



Draft Environmental Impact Report Environmental Impact Analysis

rezoning would allow the project site to develop residential units above the allocated housing unit capacity of the Areas 3 and 4 Specific Plan. Due to the minimum lot size and setbacks prescribed in the RS-6000 zoning district, a Planned Unit Development is being proposed to allow a deviation from the standards listed in the RS-6000 development standards in the Newark Municipal Code. Proposed deviations include a deviation from the minimum setbacks and minimum lot sizes. The proposed project would not meet the minimum lot size of 6,000 square feet for the RS-6000 zoning district and would not meet the minimum setbacks required at corners. While the zoning designation would change following the approval of the proposed project, the change would result in a better alignment of the project site with the General Plan designation and the project site would be developed for residential uses, as planned by the General Plan.

With the approvals required, the proposed project would be constructed in accordance with any land use plan, policy, and regulation adopted for the purpose of avoiding and mitigating environmental effects and this impact would be less than significant.

Level of Significance Before Mitigation Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.



This page intentionally left blank.

Environmental Impact Analysis

3.12 MINERAL RESOURCES

This section describes the environmental and regulatory setting for mineral resources. It also describes existing conditions and potential impacts related to mineral resources that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.12.1 Environmental Setting

According to the City's General Plan, there are no mining operations within the City itself and there is one mining operation located just west of the City limit. However, this is a gravel quarry located within the City of Fremont which was closed in 2007 with no expectation for it to reopen (City of Newark 2013a). The City's General Plan EIR determined that given the extent of urban development in the City and the proximity of sensitive environmental resources, future mineral extraction within the City's plan area is unlikely.

According to the Mineral Land Classification maps prepared by the DOC Division of Mines and Geology, the project site is located in a Mineral Resource Zone (MRZ) -3 zones. Land designated as MRZ-3 zones are areas containing mineral deposits the significance of which cannot be evaluated from available data (DOC 1996). Due to increased development of the City since the evaluation in 1996, ground disturbance activities resulting from increased development would have presumably unearthed unknown mineral resources in the City if they were present. If unknown mineral resources had been discovered, the evaluation would have been updated and the General Plan EIR would have analyzed potential impacts to mineral resources. As the General Plan EIR identified a less than significant impact relating to mineral resources from implementation of the General Plan (City of Newark 2013b), the need for updated evaluation to assess impacts are not required.

3.12.2 Regulatory Setting

Federal

There are no relevant federal regulations for mineral resources.

State

Surface Mining and Reclamation Act of 1975

Mining and mineral extraction operations throughout the state are subject to the California Surface Mining and Reclamation Act (SMARA). SMARA provides a comprehensive surface mining and reclamation policy with the regulation of surface mining operations to assure that adverse environmental impacts are minimized and mined lands are reclaimed to a usable condition. The purpose of SMARA is to identify and protect areas containing significant mineral resources. SMARA encourages the production, conservation, and protection of the state's mineral resources.



Local

City of Newark General Plan

The City's General Plan does not include any goals or policies related to mineral resources.

3.12.3 Environmental Impacts

This section analyzes the project's potential to result in significant mineral resources impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of the General Plan EIR and the DOC's Division of Mine Reclamation mineral lands classification maps.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's impacts to mineral resources are significant. Would the proposed project:

- Result in the loss of availability of a known mineral resource that would be of value to the region and the residents of the state?
- Result in the loss of availability of a locally-important mineral resource recovery site delineated on a local general plan, specific plan or other land use plan?

Project Impact Analysis and Mitigation Measures

Loss of Mineral Resource

Impact MIN-1 The proposed project would not result in the loss of availability of a known mineral resource that would be a value to the region and the residents of the state.

Impact Analysis

The DOC's Mineral Lands Classification Map of Aggregate Resources classifies the project site as being within an MRZ-3 zone. MRZ-3 zones are areas containing mineral deposits the significance of which cannot be evaluated from available data (DOC 1996). However, the City's General Plan did not identify any mineral resources of value on or near the site and no mineral extraction activities exist within City limits (City of Newark 2013b). Mineral extraction is not included as part of the proposed project and the project's proposed zoning would not allow mineral extraction. The proposed project would not result in the loss of availability of a known mineral resource that would be of value to the region and the residents of the state, and no impact would occur.

Level of Significance Before Mitigation

No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Mineral Resource Recovery Site

Impact MIN-2	The proposed project would not result in the loss of availability of a locally
	important mineral resource recovery site delineated on a local general plan,
	specific plan, or other land use plan.

Impact Analysis

The DOC Division of Mine Reclamation identifies the project site as an MRZ-3 zone. MRZ-3 zone classifications are given to areas that contain mineral deposits but the significance of it cannot be evaluated from the available data. There are no locally important mineral resource recovery sites delineated in the City's General Plan, specific plan or any other land use plan (City of Newark 2013b). Therefore, the proposed project would not result in the loss of availability of a locally important mineral resource recovery site, and no impact would occur.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

This page intentionally left blank.

Environmental Impact Analysis

3.13 NOISE

This section describes the environmental and regulatory setting for noise and vibration. It also describes existing conditions and potential impacts related to noise that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible. Descriptions and analysis in this section are based on noise modeling performed by HELIX Environmental Planning, Inc. The noise modeling output is included in this EIR as Appendix G.

3.13.1 Environmental Setting

Noise Fundamentals and Terminology

Noise is generally defined as unwanted sound that annoys or disturbs people and potentially causes an adverse psychological or physiological effect on human health. Because noise is an environmental pollutant that can interfere with human activities, evaluation of noise is necessary when considering the environmental impacts of a proposed project.

Sound is mechanical energy (vibration) transmitted by pressure waves over a medium such as air or water. Sound is characterized by various parameters that include the rate of oscillation of sound waves (frequency), the speed of propagation, and the pressure level or energy content (amplitude). In particular, the sound pressure level is the most common descriptor used to characterize the loudness of an existing sound level.

Although the decibel (dB) scale, a logarithmic scale, is used to quantify sound intensity, it does not accurately describe how sound intensity is perceived by human hearing. The perceived loudness of sound is dependent upon many factors, including sound pressure level and frequency content. The human ear is not equally sensitive to all frequencies in the entire spectrum, so noise measurements are weighted more heavily for frequencies to which humans are sensitive in a process called A-weighting, written as dB(A) and referred to as A-weighted decibels. There is a strong correlation between A-weighted sound levels and community response to noise. For this reason, the A-weighted sound level has become the standard tool of environmental noise assessment. Table 3.13-1 summarizes typical A-weighted sound levels for different common noise sources.

Table 3.13-1: Typical A-Weighted Sound Levels

Common Outdoor Activities	Noise Level (dBA)	Common Indoor Activities
Jet flyover at 1,000 feet	110	Rock band
Gas lawnmower at 3 feet	100	Food blender at 3 Feet
Diesel truck at 50 Feet at 50 miles per hour	90	Garbage Disposal at 3 Feet
Noisy urban area, daytime	80	Vacuum Cleaner at 10 Feet
Gas lawnmower, 100 feet	70	Normal Speech at 3 Feet
Commercial area	60	Large business office



Draft Environmental Impact Report Environmental Impact Analysis

Common Outdoor Activities	Noise Level (dBA)	Common Indoor Activities
Heavy traffic at 300 feet	50	Dishwasher in next room
Quiet urban daytime	40	Theater, large conference room (Background)
Quiet urban nighttime	20	Library
Quiet suburban nighttime	10	Bedroom at night, concert hall (Background)
Quiet rural nighttime	0	Broadcast/recording studio

Source: Caltrans 2013

Different types of measurements are used to characterize the time-varying nature of sound. These measurements include the equivalent sound level (Leq), the minimum and maximum sound levels (Lmin and Lmax, respectively), percentile-exceeded sound levels (such as L10, L20), the day-night sound level (Ldn), and the community noise equivalent level (CNEL). Ldn and CNEL values often differ by less than 1 dB. As a matter of practice, Ldn and CNEL values are considered to be equivalent and are treated as such in this assessment. Table 3.13-2 defines sound measurements and other terminology used in this report.

Table 3.13-2: Definition of Sound Measurements

Sound Measurements	Definition
Decibel (dB)	A unitless measure of sound on a logarithmic scale, which indicates the squared ratio of sound pressure amplitude to a reference sound pressure amplitude. The reference pressure is 20 micro-pascals.
A-Weighted Decibel (dB(A))	An overall frequency-weighted sound level in decibels that approximates the frequency response of the human ear.
Maximum Sound Level (Lmax)	The maximum sound level measured during the measurement period.
Minimum Sound Level (Lmin)	The minimum sound level measured during the measurement period.
Equivalent Sound Level (Leq)	The equivalent steady state sound level that in a stated period of time would contain the same acoustical energy.
Percentile-Exceeded Sound Level (Lxx)	The sound level exceeded xx percent of a specific time period. L10 is the sound level exceeded 10 percent of the time. L90 is the sound level exceeded 90 percent of the time. L90 is often considered to be representative of the background noise level in a given area.
Day-Night Level (Ldn)	The energy average of the A-weighted sound levels occurring during a 24- hour period, with 10 dB added to the A-weighted sound levels occurring during the period from 10:00 P.M. to 7:00 A.M.
Community Noise Equivalent Level (CNEL)	The energy average of the A-weighted sound levels occurring during a 24- hour period with 5 dB added to the A-weighted sound levels occurring during the period from 7:00 P.M. to 10:00 P.M. and 10 dB added to the A-weighted sound levels occurring during the period from 10:00 P.M. to 7:00 A.M.



Draft Environmental Impact Report Environmental Impact Analysis

Sound Measurements	Definition
Peak Particle Velocity (PPV)	A measurement of ground vibration defined as the maximum speed (measured in inches per second) at which a particle in the ground is moving relative to its inactive state. PPV is usually expressed in inches/second.
Frequency: Hertz (Hz)	The number of complete pressure fluctuations per second above and below atmospheric pressure.

Source: FHWA 2006

With respect to how humans perceive and react to changes in noise levels, a 1 dB(A) increase is imperceptible, a 3 dB(A) increase is barely perceptible, a 5 dB(A) increase is clearly noticeable, and a 10 dB(A) increase is subjectively perceived as approximately twice as loud. These subjective reactions to changes in noise levels were developed on the basis of test subjects' reactions to changes in the levels of steady-state pure tones or broadband noise and to changes in levels of a given noise source. These statistical indicators are thought to be most applicable to noise levels in the range of 50 to 70 dB(A), as this is the usual range of voice and interior noise levels. Numbers of agencies and municipalities have developed or adopted noise level standards, consistent with these and other similar studies to help prevent annoyance and to protect against the degradation of the existing noise environment.

For a point source such as a stationary compressor or construction equipment, sound attenuates based on geometry at a rate of 6 dB per doubling of distance. For a line source such as free-flowing traffic on a freeway, sound attenuates at a rate of 3 dB per doubling of distance. Atmospheric conditions including wind, temperature gradients, and humidity can change how sound propagates over distance and can affect the level of sound received at a given location. The degree to which the ground surface absorbs acoustical energy also affects sound propagation. Sound that travels over an acoustically absorptive surface, such as grass, attenuates at a slightly greater rate than sound that travels over a hard surface, such as pavement. The increased attenuation is typically in the range of 1–2 dB per doubling of distance. Barriers, such as buildings and topography that block the line of sight between a source and receiver, also increase the attenuation of sound over distance.

Decibel Addition

Because dBs are logarithmic units, sound pressure levels cannot be added or subtracted through ordinary arithmetic. On the dB scale, a doubling of sound energy corresponds to a 3 dB increase. In other words, when two identical sources are each producing sound of the same loudness, their combined sound level at a given distance would be 3 dB higher than one source under the same conditions. For example, if one source produces a sound pressure level of 70 dB(A), two identical sources would combine to produce 73 dB(A). The cumulative sound level of any number of sources can be determined using dB addition.

Vibration Standards

Vibration is like noise such that noise involves a source, a transmission path, and a receiver. While related to noise, vibration differs in that noise is generally considered to be pressure waves transmitted through air, whereas vibration usually consists of the excitation of a structure or surface. As with noise,



Draft Environmental Impact Report Environmental Impact Analysis

vibration consists of an amplitude and frequency. A person's perception to vibration depends on their individual sensitivity to vibration, as well as the amplitude and frequency of the source and the response of the system that is vibrating.

Vibration can be measured in terms of acceleration, velocity, or displacement. A common practice is to monitor vibration in terms of peak particle velocity (PPV) in inches per second. Standards pertaining to perception as well as damage to structures have been developed for vibration levels defined in terms of PPV.

Human and structural response to different vibration levels is influenced by a number of factors, including ground type, distance between source and receptor, duration, and the number of perceived vibration events. Table 3.13-3 notes the general threshold at which human annoyance could occur is 0.1 PPV for continuous/frequent sources. Table 3.13-4 indicates the threshold for damage to typical residential and commercial structures ranges from 0.3 to 0.5 PPV for continuous/frequent sources.

Table 3.13-3: Guideline Vibration Annoyance Potential Criteria

	Maximum Peak Particle Velocity (inches/second)			
Human Response	Transient Sources	Continuous/Frequent Sources		
Barely perceptible	0.035	0.012		
Distinctly perceptible	0.24	0.035		
Strongly perceptible	0.90	0.10		
Severe	2.0	0.40		

Notes: Transient sources create a single isolated vibration event, such as blasting or drop balls. Continuous/frequent intermittent sources include impact pile drivers, pogo-stick compactors, crack-and-seal equipment, vibratory pile drivers, and vibratory compaction equipment.

Source: Caltrans 2020

Table 3.13-4: Guideline Vibration Damage Potential Criteria

Structure and Canditian	Maximum Peak Particle Velocity (inches/second)			
Structure and Condition	Transient Sources	Continuous/Frequent Sources		
Extremely fragile historic buildings, ruins, ancient monuments	0.12	0.08		
Fragile buildings	0.30	0.12		
Historic and some old buildings	0.50	0.20		
Older residential structure	0.70	0.30		
New residential structures	1.2	0.50		
Modern industrial/commercial buildings	2.0	0.50		

Notes: Transient sources again create a single isolated vibration event, such as blasting or drop balls. Continuous/frequent intermittent sources include impact pile drivers, pogo-stick compactors, crack-and-seal equipment, vibratory pile drivers, and vibratory compaction equipment. Source: Caltrans 2020



Draft Environmental Impact Report Environmental Impact Analysis

The operation of heavy construction equipment, particularly pile driving and other impact devices, such as pavement breakers, create seismic waves that radiate along the surface of the ground and downward into the earth. These surface waves can be felt as ground vibration. Vibration from the operation of this equipment can result in effects ranging from annoyance of people to damage of structures. Varying geology and distance will result in different vibration levels containing different frequencies and displacements. In all cases, vibration amplitudes will decrease with increasing distance. Perceptible groundborne vibration is generally limited to areas within a few hundred feet of construction activities.

Table 7-4 "Vibration Source Levels for Construction Equipment" in the 2018 Federal Transit Administration (FTA) Transit Noise and Vibration Impact Assessment Manual lists vibration source levels for the construction equipment most likely to generate high levels of ground vibration (FTA 2018). The equipment listed in the FTA table includes impact and sonic pile drivers, clam shovel drops, hydromills, vibratory rollers, hoe rams, large and small bulldozers, caisson drilling, loaded trucks, and jackhammers. Table 3.13-5 below summarizes typical reference vibration levels generated by select construction equipment planned for this project.

Equipment	Peak Particle Velocity at 25 Feet
Vibratory roller	0.210
Large bulldozer	0.089
Caisson Drilling	0.089
Loaded trucks	0.076
Small bulldozer	0.003

Table 3.13-5: Vibration Source Levels for Construction Equipment

Source: FTA 2018

Vibration amplitude attenuates over distance and is a complex function of how energy is imparted into the ground and the soil conditions through which the vibration is traveling. The following equation can be used to estimate the vibration level at a given distance for typical soil conditions (FTA 2018). "PPVref" is the reference PPV from Table 3.13-5 and "Distance" is the distance between the source and the receptor:

PPV = PPVref x (25/Distance)^1.5

Existing Project Setting

Sensitive Receptors

Some land uses are more tolerant of noise than others. For example, schools, hospitals, churches, and residences are considered more sensitive to noise intrusion than are commercial or industrial activities. Ambient noise levels can also affect the perceived desirability or livability of a development.

The project site is located within an agricultural and industrial area in the southwestern portion of the City, and most of the project site is used as an auto parts and scrap metal salvage lot (the "Pick-n-Pull"). The northern parcel of the project site is currently undeveloped, open land. The parcel adjacent to the project site to the southwest is used for commercial vehicle storage. Mowry Avenue is adjacent to the project site's western boundary and ends approximately 700 feet south of the project site. Salt production ponds



Draft Environmental Impact Report Environmental Impact Analysis

are located on the west side of Mowry Avenue. The property to the north is undeveloped land formerly used for agriculture. The UPRR tracks are located approximately 300 feet north of the project site's northernmost point. The area to the east of the project site is permanent open space.

Noise-sensitive land uses are land uses that may be subject to stress and/or interference from excessive noise, such as residential dwellings, schools, hospitals, educational facilities, and libraries. Industrial and commercial land uses are generally not considered sensitive to noise. The closest noise-sensitive land use in the project vicinity is the Silliman Community Activity Center, located approximately 1,000 feet (0.19 mile) north of the project site. The outdoor recreation fields associated with the Silliman Activity Center are not typically considered noise-sensitive land uses. The closest residential land uses to the project site are multi-family homes approximately 2,100 (0.40 mile) to the northeast. The closest schools to the project site are the Ohlone College Newark Center for Health Sciences and Technology, approximately 2,650 feet (0.5 mile) northeast of the project site, and the Newark Memorial High School, approximately 3,483 feet (0.66 mile) northeast.

Existing Ambient Noise Levels

The existing noise environment in a project area is characterized by the area's general level of development due to the high correlation between the level of development and ambient noise levels. Areas that are not urbanized are relatively quiet, while areas that are more urbanized are noisier as a result of roadway traffic, industrial activities, and other human activities.

The predominant existing noise sources in the vicinity of the project site are vehicular traffic on Mowry Avenue associated with the existing Pick-n-Pull business, operational noise from the existing Pick-n-Pull business, passenger and freight trains on the UPRR tracks, overflights of commercial aircraft, and equipment such as tractors, graders and loaders associated with seasonal salt harvesting.

Two short-term ambient noise measurements (M1 and M2) were conducted during a site visit on April 30, 2019 by HELIX Environmental Planning, Inc. Site M1 was located in the vacant north portion of the project site, approximately 250 feet north of the Pick-n-Pull business and approximately 450 east of Mowry Avenue. Site M2 was located on the shoulder of Mowry Avenue, approximately 700 feet south of the UPRR tracks, and approximately 35 feet from the centerline of Mowry Avenue.

Measurements were taken using a Larson Davis SoundTrack LxT Sound Level Meter mounted on a tripod approximately 5 feet above the ground. The sound-level meter was field-calibrated immediately prior to the noise measurement to ensure accuracy using a Larson Davis Model CAL250 Calibrator. All measurements were made with meters that conform to the American National Standards Institute (ANSI) specifications for sound level meters (ANSI SI.4-1983 R2006). All instruments were maintained with National Institute of Standards and Technology traceable calibration per the manufacturers' standards.

Sources of noise noted during measurement M1 included operational noise from the Pick-n-Pull business, commercial aircraft overflying the site, and one passenger train (including train warning horns). Sources of noise noted during measurement M2 included traffic on Mowry Avenue, operational noise form the Pick-n-Pull business, commercial aircraft overflying the site, and three passenger trains (including train warning horns). The measured noise levels and related weather conditions for the short-term measurements are shown in Table 3.15-6, Short-term Noise Measurement Results. See Appendix G for survey notes from the short-term measurements.



Measurement	Location	Conditions	Time	dB(A) Leq	Notes
M1	North center of the site, approximately 250 feet from the Pick-n-Pull and 450 feet from Mowry Avenue.	59°F, 7 mph wind, 61 percent humidity, sunny	9:23 AM to 9:38 AM	48.5	Noise from Pick-n-Pull; aircraft at approximately 2- minute intervals; 1 train
M2	Shoulder of Mowry Avenue, approximately 700 feet south of the UPRR tracks and 35 feet from roadway centerline.	61°F, 8 mph wind, 59 percent humidity, sunny	9:49 AM to 10:01 AM	58.4	60 cars, 2 medium trucks, 3 heavy trucks; noise from Pick-n-Pull; aircraft at approximately 2 to 4- minute intervals; three trains

Source: HELIX Environmental Planning, Inc.

 $^{\circ}$ F = degrees Fahrenheit; mph = miles per hour

3.13.2 Regulatory Setting

Federal, state, and local agencies regulate different aspects of environmental noise. Generally, the federal government sets noise standards for transportation-related noise sources closely linked to interstate commerce. These include aircraft, locomotives, and trucks. No federal noise standards are directly applicable to this project. The state government sets noise standards for transportation noise sources such as automobiles, light trucks, and motorcycles. Noise sources associated with industrial, commercial, and construction activities are generally subject to local control through noise ordinances and general plan policies. Local general plans identify general principles intended to guide and influence development plans.

State

California Building Code

Part 2, Title 24 of the CCR California Noise Insulation Standards establishes minimum noise insulation standards to protect persons within new hotels, motels, dormitories, long-term care facilities, apartment houses, and dwellings other than single-family residences. Under Section 1207.11 "Exterior Sound Transmission Control", interior noise levels attributable to exterior noise sources cannot exceed 45 L_{dn} in any habitable room. Where such residences are located in an environment where exterior noise is 60 L_{dn} or greater, an acoustical analysis is required to ensure interior levels do not exceed the 45 L_{dn} interior standard. If the interior allowable noise levels are met by requiring that windows be kept closed, the design for the building must also specify a ventilation or air conditioning system to provide a habitable interior environment.

California Green Building Standards

The 2016 CALGreen establishes interior noise insulation standards for nonresidential occupied buildings, such as offices. CALGreen Section 5.507 "Environmental Comfort," states the following:



5.507.4.1 Exterior noise transmission. Wall and roof-ceiling assemblies exposed to the noise source making up the building or addition envelope or altered envelope shall meet a composite Sound Transmission Class (STC) rating of at least 50 or a composite Outside-Inside Transmission Class (OITC) rating of no less than 40, with exterior windows of a minimum STC of 40 or OITC of 30 in the following locations:

1. Within the 65 CNEL noise contour of an airport

Exceptions:

- 1. L_{dn} or CNEL for military airports shall be determined by the facility Air Installation Compatible Land Use Zone (AICUZ) plan.
- 2. L_{dn} or CNEL for other airports and heliports for which a land use plan that has not been developed shall be determined by the local general plan noise element.
- 3. Within the 65 CNEL or L_{dn} noise contour of a freeway or expressway, railroad, industrial source or fixed-guideway notice source as determined by the Noise Element of the General *Plan*.

5.507.4.1.1 Noise exposure where noise contours are not readily available. Buildings exposed to a noise level of 65 dB L_{eq} -1-hr during any hour of operation shall have building, addition or alteration exterior wall and roof-ceiling assemblies exposed to the noise source meeting a composite STC rating of at least 45 (or OITC 35), with exterior windows of a minimum STC of 40 (or OITC 30).

5.507.4.2 Performance method. For buildings located as defined in Section 5.507.4.1 or 5.507.4.1.1, wall and roof-ceiling assemblies exposed to the noise source making up the building or addition envelope or altered envelope shall be constructed to provide an interior noise environment attributable to exterior sources that does not exceed an hourly equivalent noise level $(L_{eq} - 1Hr)$ of 50 dB(A) in occupied areas during any hours of operations.

5.507.4.2.1 Site features. Exterior features such as sound walls or earth berms may be utilized as appropriate to the building, addition or alteration project to mitigate sound migration to the interior.

5.507.4.2.2 Documentation of compliance. An acoustical analysis documenting complying interior sound levels shall be prepared by personnel approved by the architect or engineer of record.

5.507.4.3 Interior sound transmission. Wall and floor-ceiling assemblies separating tenant spaces and tenant spaces shall have an STC of at least 40.

California Environmental Quality Act

The CEQA Guidelines Appendix G indicates a significant noise impact may occur if a project exposes persons to noise or vibration levels in excess of local general plans or noise ordinance standards or causes a substantial permanent or temporary increase in ambient noise levels. CEQA standards are discussed under the Thresholds of Significance criteria section.



Local

City of Newark General Plan

Chapter 9 "Environmental Hazards", Table EH-2 in the City of Newark General Plan (Figure 3-5) identifies land use compatibility noise standards for noise-sensitive land uses affected by transportation and non-transportation noise sources. As shown in the table below, the ranges for noise-sensitive single-family residential land uses impacted by noise are as follows:

- "Normally Acceptable" <60 dB(A) Ldn
- "Conditionally Acceptable" 55-70 dB(A) Ldn
- "Normally Unacceptable" 70-75 dB(A) Ldn
- "Clearly Unacceptable" Higher than 75 dB(A) Ldn

Sites with ambient noise at "conditionally acceptable" levels should be undertaken only after a detailed analysis of the noise reduction requirements is made and after needed noise insulation features are included in the design. Conventional construction, but with closed windows and fresh-air supply systems or air conditioning will normally suffice. New construction with exterior noise levels in the "Normally Unacceptable" range are discouraged. If new construction or development does proceed, a detailed analysis of the noise reduction requirements must be made and needed noise insulation features included in the design.



This page intentionally left blank.

TABLE EH-2 NOISE COMPATIBILITY GUIDELINES FOR NEWARK

	Interior CNEL or		Exteri CN	or Noi IEL or	se Exp L _{dn} (dE	osure, 3A)	
Land Uses	L _{dn} (dBA)	55	60	65	70	75	80
Residential-Low Density Single-Family, Duplex, Mobile Homes	45*		ìш			<i></i>	
Residential-Multiple Family	45*		111	йп	Ч7.	<i>.</i>	
Transient Lodging, Motels, Hotels	45*		'W	ĥп	1	////	~
Schools, Libraries, Churches, Hospitals, Nursing Homes	45*))))	ΥÌ //		<i></i>
Auditoriums, Concert Halls, Amphitheaters					H		
Sports Arena, Outdoor Spectator Sports							
Playgrounds, Neighborhood Parks				~~~~	<u>)) </u>		
Golf Courses, Riding Stables, Water Recreation, Cemeteries				////		<u>,,,,</u>	
Office Buildings, Businesses, Commercial and Professional	50			(///	ŤΠ	п₩	
Industrial, Manufacturing, Utilities, Agricultural					````	ŊЛ	
Normally Acceptable: Specified land use is satisfactory based upon the assumption that any buildings involved are of normal conventional construction, without any special noise insulation requirements.		Normally Un New constru discouraged. analysis of th and needed	nacceptab ction or de If new con le noise re noise insu	le: evelopme nstructior duction r lation fea	ent should n does pro equireme tures inclu	l generally oceed, a de ents must b uded in the	be etailed e made e design.
Conditionally Acceptable: New construction or development should be undertaken only after a detailed analysis of the noise reduction requirements is made and the needed noise insulation features are included in the design. Conventional construction, but with closed windows and fresh air supply systems or air conditioning will normally suffice. * Noise level requirement with closed windows, mechanical ventila	ation, or other	Clearly Unar New constru be undertak	cceptable: ction or de ken. tion shall l	evelopme	ent genera	ully should napter 12 S	not ection 1205 of

Source: Table EH-2, City of Newark General Plan, December 2013

Source: State of California General Plan Guidelines, 2003.

the Building Code.



Project Location Newark, California

Client/Project City of Newark Mowry Village Project

Figure No.

3-5 Title

City of Newark Noise Compatibility Guidelines

This page intentionally left blank.

Draft Environmental Impact Report Environmental Impact Analysis

The City of Newark General Plan also lists several noise actions and policies relevant to the proposed project:

Goal EH-6: Maintaining Peace and Quiet. Maintain the peace and quiet of Newark neighborhoods and promote an environment where noise does not adversely affect sensitive land uses.

- Policy EH-6.6: Construction Noise Regulating Construction Hours. Reduce noise associated with construction activities by prohibiting construction in residential neighborhoods between the hours of 7 PM and 7 AM Monday through Friday and at all times on Saturdays, Sundays, and State/federal holidays.
- Policy EH-6.7: Construction Noise Addressing Sources of Construction Noise. Reduce noise associated with construction activities by requiring properly maintained mufflers on construction vehicles, requiring the placement of stationary construction equipment as far as possible from developed areas, and requiring temporary acoustical barriers/shielding to minimize construction noise impacts at adjacent receptors. Special attention should be paid to noisesensitive receptors (including residential, hospital, school, and religious land uses).

Goal EH-7: Design of New Structures. Ensure that new structures/uses are designed and constructed to preclude excessive, inappropriate, and undesirable noise effects.

- Policy EH-7.1: Land Use Planning and Noise Compatibility. Use the noise compatibility guidelines in Table EH-2 and the future-conditions noise contour map in Figure EH-4 to plan for appropriate land uses near existing uses that generate noise. Noise mitigation should be included to ensure that new residential areas and other noise-sensitive uses are appropriately buffered from significant noise sources.
- Policy EH-7.2: Noise Compatibility Strategies. Where land use noise compatibility conflicts currently exist, explore the need for mitigation measures on noise sources that may adjacent to sensitive receptors. In planning for future developments, promote the use of buffer zones, barrier/shielding measures, and/or sound isolation building techniques to preclude noise impacts to noise-sensitive land uses.
- Policy EH-7.3: Reducing Exposure to Operation Noise. In new residential and mixed-use developments, require that stationary equipment (such as air conditioning units and condensers) be placed in separate spaces, rooftops, or other areas such that noise impacts to interior living areas will be reduced. Similarly, potentially noisy common spaces, such as trash collection areas and loading zones, should be located away from residential units or other noise-sensitive spaces.
- Policy EH-7.4: Residential Noise Standard Exterior. Plan for and implement strategies to maintain exterior noise levels that are consistent with the noise compatibility guidelines in Table EF-2. For residential areas, this limit is 60 dB(A) Ldn for outdoor living areas. Where this level is exceeded due to freeways, arterials, and/or railroads, the construction of berms, walls, buffer zones, and other noise-reduction measures to reduce noise to the greatest extent feasible will be required.



Draft Environmental Impact Report Environmental Impact Analysis

- Policy EH-7.5: Residential Noise Standard Interior. Use site planning and architectural design to protect occupants of new buildings from excessive noise, per California State Noise Insulation Standards (CCR Title 24) and Chapter 35 of the Uniform Building Code. For example, site planning should place bedrooms and other noise-sensitive rooms away from exterior noise sources and architectural design should use double-paned windows and other insulating measures to reduce interior noise.
- **Policy EH-7.6: New Noise Sources.** Require new developments that have the potential to create long-term noise increases to mitigate potential impact to off-site receptor properties.
- Policy EH-7.7: Acoustical Study Requirement. Require acoustical studies for new developments in areas where the noise levels exceed the "normally acceptable" levels for the proposed land use; based on Table EH-2. For residential uses, the analysis should include mitigation measures to limit the noise exposure in interior living spaces to 45 dB(A) Ldn, consistent with California Title 24.

Acoustical studies should have the following minimum attributes:

- Be the responsibility of the development applicant.
- Be prepared by qualified persons experienced in the field of environmental noise assessment and architectural acoustics.
- Include representative noise level measurements with sufficient sampling periods and locations to adequately describe existing local conditions.
- Include estimates for existing and projected (20 years hence) noise levels in terms of (a) Ldn or CNEL and (b) any future noise regulations to be adopted by the City. Those existing and projected noise levels shall be compared to the adopted policies of the Noise Element.
- Include recommended mitigation measures to achieve compliance with the adopted policies and standards of the Noise Element. Where the noise source in question consists of intermittent single events, the report should address the effects of maximum noise levels in sleeping rooms and potential sleep disturbance issues.
- Include estimates for interior and exterior noise exposure after the prescribed mitigation measures have been implemented.
- Describe a post-project assessment program that could be used to evaluate the effectiveness of the proposed mitigation measures.
- Action EH-7.A: Noise Mitigation. Use the development review process to ensure that noise
 impacts are mitigation through setbacks/buffer zones, earthen berms, sound walls, building
 siting/orientation, and other means.
- Action EH-7.B: Conditional Use Permits. Use the development review process, including conditional use permits, to limit activities which would generate high levels of noise during nighttime hours (i.e., from 7 PM to 7 AM).



City of Newark Municipal Code

Section 17.24.100 "Noise" in the City of Newark Municipal Code addresses noise as follows:

- A. Noise Limits. It shall be unlawful for any person to disturb the peace, quiet, and comfort of the community, or any portion thereof, or neighborhood therein, by creating or causing to be created any unreasonable noises.
 - 1. Applicability. The provisions of this subsection apply to noises from all sources within the city except the following:
 - a. Alarms and Warning Devises: Aural alarms or warning devices, including but not limited to fire alarms, burglar alarms, and emergency vehicle sirens and air horns. However, if a standard or minimum noise level is prescribed for particular type of aural alarm or warning device by the laws or regulations of the State of California, the noise emitted from such alarm or warning device shall not exceed such standard or minimum level by more than three dBA.
 - b. Emergency Response Activities: Noise from emergency response activities.
 - c. Events at Which No Mechanical or Amplifying Equipment is Employed: Noise from events conducted lawfully and without the use of sound of any kind that is mechanically produced or amplified or focused by any means.
 - d. Audio Equipment Used by Public Safety Officers: Noise from audio equipment used or operated by public safety officers in the performance of their duties.
 - e. Generators Required for Medical Purposes or During Power Outages: Noise from generators required for medical purposes or during power outages.
 - f. Permitted for Temporary Uses or Activities: Specific uses or activities for which a temporary exemption was granted through a conditional use permit, minor use permit, or other permit or authorization granted by the city.
 - 2. Noise Restriction by Decibel.
 - a. Residential Property Noise Limits.
 - *i.* No person shall produce or allow to be produced by human voice, machine, device, or any combination of same, on residential property, a noise level at any point outside of the property plane that exceeds 70 dBA between the hours of 7:00 a.m. and 9:00 p.m. or 60 dBA between the hours of 9:00 p.m. and 7:00 a.m.
 - ii. No person shall produce or allow to be produced by human voice, machine, device, or any combinations of same, on multifamily residential property, a noise level more than 60 dBA three feet from any wall, floor, or ceiling inside any dwelling unit on the same property, when the windows and doors of the dwelling unit are closed, except within the dwelling unit in which the noise source or sources may be located.



- b. Commercial and Industrial Property Noise Limits. Except for commercial and industrial property abutting residential property, no person shall produce or allow to be produced by human voice, machine, device, or any other combination of same, on commercial or industrial property, a noise level at any point outside of the property plane that exceeds 70 dBA.
 - Abutting Residential Property. Commercial and industrial property that abuts residential property shall be subject to the residential property noise limits set forth in subsections (a)(i) and (ii) above.
- c. Public Property Noise Limits. Except as otherwise provided in these regulations, no person shall produce or allow to be produced on public property, by human voice, machine, device, or any combination of same, a noise level that exceeds 60 dBA at a distance of 25 feet or more from the source. Noise from activities of the City of Newark is exempted from these regulations.
- 3. Construction and Landscaping Activities. Unless otherwise provided pursuant to a dulyissued permit or a condition of approval of a land use entitlement, the construction, alteration, or repair of structures and any landscaping activities, occurring between the hours of 10:00 a.m. and 6:00 p.m. on Sundays and holidays, and 7:00 a.m. and 7:00 p.m. on other days, shall be subject to the following:
 - a. No individual device or piece of equipment shall produce a noise level exceeding 83 dBA at a distance of twenty-five feet from the source. If the device or equipment is housed within a structure on the property, the measurement shall be made outside the structure at a distance as close as possible to twenty-five feet from the equipment.
 - b. The noise level at any point outside of the property plane shall not exceed 86 dBA. During all other times, the decibel levels set forth in Subsection 17.24.100.A.2, Noise Restriction by Decibel, control.
- B. Noise Creation and Noise Exposure.
 - 1. Acoustic Study. An acoustic study shall be required for any proposed project which could create or be subject to a noise exposure greater than that deemed "normally acceptable" by the general plan.
 - Noise Attenuation Measures. Any project subject to the acoustic study requirements of Subsection 17.24.100.A.1, Acoustic Study, may be required as a condition of approval to incorporate noise attenuation measures deemed necessary to ensure that noise standards are not exceeded.
 - a. New noise-sensitive uses (e.g., schools, hospitals, churches, and residences) shall incorporate noise attenuation measures to achieve and maintain an interior noise level of 45 dBA.
 - b. Noise attenuation measures identified in an acoustic study shall be incorporated into the project to reduce noise impacts to satisfactory levels.



c. Emphasis shall be placed upon site planning and project design measures. The use of noise barriers shall be considered and may be required only after all feasible design-related noise measures have been incorporated into the project.

3.13.3 Environmental Impacts

This section analyzes the project's potential to result in significant noise impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

Results from the HELIX Environmental Planning, Inc. site measurements were used to provide baseline noise conditions at nearby sensitive receptors and within the project site vicinity. For the purpose of this analysis, potential sensitive receptors were determined by reviewing current aerial photography and from notes from HELIX Environmental Planning, Inc.

Impacts from future project-related traffic were estimated using predicted traffic counts from the traffic volumes provided by HELIX Environmental Planning, Inc. and Fehr & Peers, 2021.

Noise from the Project's mechanical systems would operate regularly and are therefore required to comply with the requirements listed in Section 17.24.100 "Noise" in the City of Newark Municipal Code.

The Federal Highway Administration (FHWA) Roadway Construction Noise Model (RCNM) was used to estimate the impact from short-term construction activities. The RCNM is used as the FHWA's national standard for predicting noise generated from construction activities. The RCNM analysis includes the calculation of noise levels at a defined distance for a variety of construction equipment. The spreadsheet inputs include acoustical use factors and distance to receptors and calculates the expected Lmax values and Leq values at a selected receptor.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's noise impacts are significant. Would the proposed project result in:

- Generation of a substantial temporary or permanent increase in the ambient noise levels in the vicinity of the project noise levels in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies?
- Generation of excessive groundborne vibration or groundborne noise levels?
- For a project located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project expose people residing or working in the project area to excessive noise levels?



EPA Guidelines

The EPA has established guidelines (EPA 1978) for assessing the impact of an increase in noise levels. These guidelines have been used for several years as industry standards to determine the potential impact of noise increases on communities. Most people will tolerate a small increase in background noise (up to about 5 dB(A)) without complaint, especially if the increase is gradual over a period of years (such as from gradually increasing traffic volumes). Increases greater than 5 dB(A) may cause complaints and interference with sleep. Increases above 10 dB(A) (heard as a doubling of judged loudness) are likely to cause complaints and should be considered a serious increase. Table 3.7-7 defines each of the traditional impact descriptions, their quantitative range, and the qualitative human response to changes in noise levels.

Table 3.13-7: EPA Impact Guidelines

Increase over Existing or Baseline Sound Levels	Impact Per EPA Region Guidelines	Qualitative Human Perception of Difference in Sound Levels
0 dB to 5 dB	Minimum impact	Imperceivable or Slight difference
6 dB to 10 dB	Significant impact	Significant Noticeable difference—complaints possible
Over 10 dB	Serious impact	Loudness changes by a factor of two or greater. Clearly audible difference—complaints likely

Notes:

dB = decibels EPA = U.S. Environmental Protection Agency

Project Impact Analysis and Mitigation Measures

Noise Levels in Excess of Standards

Impact NOI-1 The proposed project could result in the generation of a substantial temporary or permanent increase in ambient noise levels in the vicinity of the project in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies.

Impact Analysis

Exterior Traffic Noise Level Impacts

Traffic noise depends primarily on vehicle speed (tire noise increases with speed), proportion of medium and large truck traffic (trucks generate engine, exhaust, and wind noise in addition to tire noise), and number of speed control devices, such as traffic lights and stop signs (accelerating and decelerating vehicles and trucks can generate more noise).

Changes in traffic volumes can also have an impact on overall traffic noise levels. For example, it takes 25 percent more traffic volume to produce an increase of only 1 dB(A) in the ambient noise level. For roads already heavy with traffic volume, an increase in traffic numbers could even reduce noise because



the heavier volumes could slow down the average speed of the vehicles. A doubling of traffic volume results in a 3 dB(A) increase in noise levels. Typically, a 6 dB increase or more over baseline or existing sound levels would be considered a significant impact.

To describe future noise levels due to traffic added from the proposed project, existing and 2040 peak hour existing traffic volumes (with and without the Project) listed in the traffic study, provided in Appendix H, were used to determine the percentage increase of traffic on the roads adjacent to the Project site and nearby sensitive receptors.

Tables 3-13.8 and 3-13.9 shows the peak hour counts associated with traffic on the local roadway network under the existing and existing plus project traffic conditions. The last columns in the table show the overall percentage change and the estimated difference in peak hour noise level in dB(A).

Roadway Section	Existing Peak Hour Traffic Count	Existing Peak Hour Traffic Count with Project	Percentage Change	Estimated dB(A) Change
Mowry Avenue				
Project to Cherry St.	414	561	35.5%	1.4
Cherry St to Cedar Blvd	1556	1630	4.8%	0.2
Cedar Blvd to Alpenrose Ct	2171	2245	3.4%	0.1
Alpenrose Ct to I-880	2763	2825	2.2%	0.1
Cherry Street				
Central Ave to Mowry Ave	2694	2731	1.4%	0.1
Mowry Ave to Stephenson Blvd	2210	2246	1.6%	0.1

Table 3.13-8: Traffic Peak Hour Counts and Estimated Noise Increase - Existing

Table 3.13-9: Traffic Peak Hour Counts and Estimated Noise Increase - 2040

Roadway Section	2040 Peak Hour Traffic Count	2040 Peak Hour Traffic Count with Project	Percentage Change	Estimated dB(A) Change
Mowry Avenue				
Project to Cherry St.	640	787	23.0%	0.9
Cherry St to Cedar Blvd	2560	2634	2.9%	0.1
Cedar Blvd to Alpenrose Ct	3490	3564	2.1%	0.1
Alpenrose Ct to I-880	4280	4342	1.5%	0.1
Cherry Street				
Central Ave to Mowry Ave	4420	4457	0.8%	0.03
Mowry Ave to Stephenson Blvd	3960	3996	0.9%	0.04

The project is expected to minimally increase traffic counts along all roads around the project site. There will be essentially no change in traffic noise (1.4 dB(A) or less) expected along these streets. Therefore,



Draft Environmental Impact Report Environmental Impact Analysis

the proposed project should not cause increased traffic noise levels over the baseline conditions at the neighboring sensitive receptors, and this would be a less than significant impact.

Interior Traffic Noise Level Impacts

Note – In accordance with the CBIA v. BAAQMD CEQA case law, the effect of the environment on the project is not generally a CEQA consideration, unless the project would exacerbate an existing condition. Although this issue is not a CEQA impact, it is a consideration for the City in determining project approval.

The California Building Code states that the interior noise levels attributable to exterior sources shall not exceed 45 dB(A) in any habitable room. The needed sound isolation requirements of a building's exterior façade system would be dependent on the following conditions:

- The dimension of the rooms with exterior windows
- The finishes within the rooms
- The ratio of clear glass to solid wall in the exterior wall assembly
- The exterior solid wall construction

Modern construction with punch windows typically provides a 25 dB(A) exterior-to-interior noise level reduction with windows closed. Therefore, sensitive receptors exposed to exterior noise levels of 70 dB(A) Ldn or less would typically comply with the required interior noise level standard as stated in the California Building Code. Modern construction using window walls, curtainwalls, or a high ratio of exterior clear glass would provide less reduction with the windows closed. Buildings using a large amount of glass are required to comply with the required interior noise level standard as stated in the California Building Code if exposure to exterior noise levels of 67 dB(A) Ldn or less is anticipated.

According to Policy EH-7.7 in the City of Newark General Plan, a project-specific acoustical analysis shall be completed at the time detailed development plans are prepared to determine appropriate mitigation to reduce interior noise levels to 45 dB(A) Ldn and include ways to incorporate such mitigation into the project design and implementation. Appropriate mitigation incorporated into the project design and implementation. Appropriate mitigation incorporated windows and doors, sound rated wall construction, acoustical caulking, and protected ventilation openings, Therefore, the proposed project would be required to implement Mitigation Measure NOI-1 which requires the preparation of a project-specific acoustical analysis to determine necessary noise control treatments. With the implementation of Mitigation Measure NOI-1, the impact of traffic noise on the interior of the residential units would be less than significant.

Train Noise

Note –Again, in accordance with the CBIA v. BAAQMD CEQA case law, the effect of the environment on the project is not generally a CEQA consideration, unless the project would exacerbate an existing condition. It would be difficult to argue that the project would result in more train traffic (and more train noise) on the UPRR tracks near the project site. Although this issue is not a CEQA impact, it is a consideration for the City in determining project approval. A preliminary analysis was completed to



Draft Environmental Impact Report Environmental Impact Analysis

determine if noise from the UPRR track would result in exterior and interior noise levels above the City standards for future project residences.

As noted from the studies by HELIX Environmental Planning, Inc., the UPRR tracks, approximately 300 feet north of the project site, carry freight trains and passenger trains for the Amtrak Capitol Corridor and Coast Starlight routes as well as the ACE route. The intersection of the UPRR tracks and Mowry Avenue is an "at-grade crossing." Trains approaching an at-grade crossing are required to sound their warning horns 15 to 20 seconds before reaching the crossing, but not more than 1,320 feet (0.25 mile) before the crossing. In addition, the at-grade crossing is equipped with an audible warning device when the roadway barriers are lowered.

The schedule for passenger rail on UPRR tracks between the cities of Fremont and San Jose (passing the project site) in 2019 (prior to reduction in service due to the pandemic in 2020 and 2021) was 24 trains per weekday, with 22 trains scheduled in the hours between 7:00 A.M. and 10:00 P.M., and 2 trains scheduled in the hours between 5:00 A.M. and 7:00 A.M. For national security reasons, neither the Federal Railroad Administration (FRA) nor the UPRR disclose freight train schedules or counts. A reasonable assumption for the average daily mix of traffic on the UPRR tracks is 24 passenger trains and 6 freight trains.

The FTA provides a spreadsheet that can be used to model the noise impacts from railroad traffic, including at-grade crossing warning horns and barrier alarms (FTA 2018). Based on an assumption of passenger trains averaging one diesel electric locomotive and six cars traveling at 50 mph, and freight trains averaging two diesel electric locomotives and 25 cars traveling at 40 mph, the predicted noise at the measurement location would be 74.4 dBA LDN, or 4.4 dBA above the measured level. Calibrating the model by subtracting 4.4 dBA from the results, the predicted noise level at the project residential lot closest to the UPRR tracks (lot 204, 300 feet from the tracks) would be 63.0 dB(A) Ldn (the FTA Transit Noise Assessment Model output is included in Appendix G). This would exceed the City residential standard of 60 dB(A) Ldn for exterior spaces.

Implementation of Mitigation Measure NOI-1 to require an acoustical study once the building shapes, orientations, and locations are known and to implement any design feature recommended by the study would reduce noise levels for project residences to levels at or below the City standards.

Salt Harvest Noise

The seasonal harvesting of salt from the crystallization beds across Mowry Avenue from the project site is typically completed using tractors, graders, and loaders. The noise from salt harvesting equipment would typically only last a few weeks per year and equipment use would be intermittent and would not be concentrated on the east side of the salt beds near Mowry Avenue and the project site. Therefore, future residents of the project would not be exposed to noise in excess of the City standards as a result of exiting salt harvesting operations.

Proposed Project Fixed-Source Noise

Typical single-family residential building construction would commonly involve new mechanical equipment, such as air conditioning units and exhaust fans. This equipment would generate noise that would radiate to neighboring properties. The noise from this equipment would be obliged to comply with

Draft Environmental Impact Report Environmental Impact Analysis

the maximum noise level limits listed in Section 17.24.100, Paragraph A.2.a in the City of Newark Municipal Code.

When the actual on-site equipment is selected, in accordance with Mitigation Measure NOI-2, a noise analysis will be prepared by a qualified acoustical engineer and the equipment would be designed to incorporate measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels that may affect nearby properties. Noise levels from the project's fixed-source equipment at any point outside of the property plane will not exceed 70 dB(A) between the hours of 7:00 A.M. and 9:00 P.M. or 60 dB(A) between the hours of 9:00 P.M. and 7:00 A.M.

Therefore, with the requirements listed in Section 17.24.100, Paragraph A.2.a (as noted in Mitigation Measure NOI-2), the impact of fixed-source noise to the neighboring properties would be less than significant.

Short-term Construction and Remediation Noise Impacts

Two types of short-term noise impacts could occur during construction and remediation. The first type of short-term noise impact is traffic noise from construction and remediation crew vehicular commutes on the access roads leading to and from the project site. Trucks hauling material or debris to and from the project site would primarily travel along Mowry Avenue from the project site to I-880, approximately 1.2 miles north. Mowry Avenue has residential land uses on both side of the road. There are no alternate haul routes that would not pass by residential neighborhoods. Haul trucks associated with the project's remediation and construction activities could result in a temporary increase in traffic noise on Mowry Avenue in excess of the City standard of 60 dBA LDN and would be a potentially significant impact. As required by General Plan Policies EH-6.6 and EH-6.7, noise associated with construction activities are prohibited to certain hours and sources of construction noise are required to be addressed. The proposed project would comply with General Plan Policies EH-6.6 and EH-6.7 through implementation of Mitigation Measure NOI-3. Mitigation Measure NOI-3 would restrict construction and remediation to the hours between 7:00 A.M. and 6:00 P.M., Monday through Friday; and between 8:00 A.M. and 5:00 P.M. on Saturdays; and no construction would be permitted on Sundays and holidays. Therefore, with implementation of Mitigation Measure NOI-3, construction and remediation traffic from the proposed project would not result in a substantial temporary increase in ambient noise levels in the vicinity of the project in excess of standards established in the City General Plan, and impacts would be less than significant.

The second type of short-term noise impact is related to noise generated during construction and remediation. Each construction stage, including remediation activities, has its own mix of equipment, and consequently, its own noise characteristics. The various construction operations would change the character of the noise generated at the project site and therefore, the noise level as construction progresses. The loudest stages of construction typically involve earthmoving and grading equipment.

The construction of the proposed project would be conducted in thirteen (13) stages and each stage will use different construction equipment. Remediation activities are included within the thirteen construction stages identified below. The main types of noise-producing equipment for each construction stage are shown in Table 3.13-10.



Table 3.13-10: Construction Stage Equipment

Construction Stage	Constructio	n Equipment
Pick-n-Pull Inventory Removal	Crane Concrete/Industrial Saw	Excavators (3) Rubber Tired Dozers (2)
Demolition	Concrete/Industrial Saw Excavators (3)	Rubber Tired Dozers (2)
Mobilization	Crane	Tractor
Remedial Soil Cleanup	Excavator Water Truck	Rubber Tired Dozer Front End Loader
Grading	Excavators (2) Grader Water Truck Rubber Tired Dozer	Scrapers (2) Tractor Front End Loader
Underground Utilities	Crane Excavators (2) Water Truck	Rubber Tired Dozer Tractor Front End Loader
Jack and Bore Preparation	Crane Excavator	Skid Steer Loader Tractor
Jack and Bore	Bore/Drill Rig Crane	Excavator Pump
Jack and Bore Cleanup	Crane Skid Steer Loader	Tractor
Off-Site Street Improvements	Pavers (2) Paving Equipment (2)	Rollers (2) Pavement Scarafier
Paving	Pavers (2) Paving Equipment (2)	Rollers (2)
Building Construction	Crane Forklifts (3) Generator	Tractor Front End Loader Backhoe
Architectural Coating	Air Compressor	

Table 3.13-11 lists the types of construction equipment and the maximum and average operational noise level as measured at 1,000 feet from the operating equipment. The 1,000-foot distance represents the approximate closest distance between the project site and the closest receptor at the Silliman Community Center.



Draft Environmental Impact Report Environmental Impact Analysis

Construction Equipment Source at the	Distance to	Sound Level at Receptor			
Project Site	Receptor	Lmax, dB(A)	Acoustical Use Factor (%)	Leq, dB(A)	
Backhoe	1,000 feet	51.5	40	47.6	
Compressor (air)	1,000 feet	51.6	40	47.7	
Concrete/Industrial Saw	1,000 feet	63.6	20	56.6	
Crane	1,000 feet	54.5	16	46.6	
Dozer	1,000 feet	55.6	40	51.7	
Drill Rig Truck	1,000 feet	53.1	20	46.1	
Excavator	1,000 feet	54.7	40	50.7	
Forklift	1,000 feet	57.4	40	53.4	
Front End Loader	1,000 feet	53.1	40	49.1	
Generator	1,000 feet	54.6	50	51.6	
Grader	1,000 feet	59.0	40	55.0	
Pavement Scarafier	1,000 feet	63.5	20	56.5	
Paver and Paving Equipment	1,000 feet	51.2	50	48.2	
Pumps	1,000 feet	54.9	50	51.9	
Roller	1,000 feet	54.0	20	47.0	
Scraper	1,000 feet	57.6	40	53.6	
Skid Steer Loader	1,000 feet	53.1	40	49.1	
Tractor	1,000 feet	58.0	40	54.0	
Water Truck	1,000 feet	50.5	40	46.5	

Table 3.13-11: Federal Highway Administration Roadway Construction Noise Model Source Noise Levels

Notes:

Source: Federal Highway Administration Road Construction Noise Model v1.1 2018

A worst-case condition for construction activity would assume all noise-generating equipment were operating at the same time and at the same distance from the closest noise-sensitive receptor. Using this assumption, the RCNM program calculated the following combined Leq and Lmax noise levels from each stage of construction as shown in Table 3.13-12.

Table 3.13-12: Calculated Noise Level from Each Construction Stage

Construction Phase	Distance to Closest Noise Sensitive Receptor	Calculated <u>Maximum</u> Sound Level, dB(A)	Calculated <u>Equivalent</u> Sound Level, dB(A)
Pick-n-Pull Inventory Removal	1,000 feet	66.2	60.6
Demolition	1,000 feet	65.9	60.4
Mobilization	1,000 feet	59.6	54.7
Draft Environmental Impact Report Environmental Impact Analysis

Construction Phase	Distance to Closest Noise Sensitive Receptor	Calculated <u>Maximum</u> Sound Level, dB(A)	Calculated <u>Equivalent</u> Sound Level, dB(A)	
Remedial Soil Cleanup	1,000 feet	59.9	55.9	
Grading	1,000 feet	65.8	61.9	
Underground Utilities	1,000 feet	63.4	59.1	
Jack and Bore Preparation	1,000 feet	61.5	57.0	
Jack and Bore	1,000 feet	60.4	55.6	
Jack and Bore Cleanup	1,000 feet	60.5	55.8	
Off-Site Street Improvements	1,000 feet	65.1	59.1	
Paving	1,000 feet	60.1	55.6	
Building Construction	1,000 feet	65.0	60.9	
Architectural Coating	1,000 feet	51.6	47.7	

Noise levels from construction and remediation should be within the "Normally Acceptable" land use compatibility range for residential properties as defined by the City of Newark General Plan and increases in noise levels from construction activity would be temporary. All construction activities at the site would follow the requirements listed in Mitigation Measure NOI-3.

In conclusion, construction and remediation noise would be short-term and intermittent. Furthermore, implementation of the Mitigation Measure NOI-3 and compliance with hours restrictions as dictated by the City would reduce construction and remediation noise to the closest noise-sensitive receptors to the extent feasible. Therefore, impacts from construction and remediation noise would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

MM NOI-1: Project-Specific Acoustical Study. A project-specific acoustical analysis shall be completed at the time detailed development plans are prepared, so that the design of the residential units would be sufficient to adequately reduce interior noise level to 45 dBA Ldn or lower. Building sound insulation requirements shall include the provision of forced-air mechanical ventilation for all new unit with direct line of sight to significant transportation noise sources or railroad lines in the project vicinity. Special building sound insulation treatment may be required. These treatments shall include, but are not limited to, sound rated windows and doors, sound rated wall construction, acoustical caulking, protected ventilation openings, etc. The specific determination of what treatments are necessary shall be determined on a unit-by-unit basis. The results of the analysis, including the description of the necessary noise control treatments to achieve the acceptable noise levels inside the living units, shall be submitted to the City along with building plans and shall be reviewed and approved by the Community Development Director prior to the issuance of a building permit.



Draft Environmental Impact Report Environmental Impact Analysis

The project-specific acoustical study shall have the following minimum attributes:

- Be the responsibility of the development applicant.
- Be prepared by qualified persons experienced in the fields of environmental noise assessment ad architectural acoustics.
- Include representative noise level measurements with sufficient sampling periods and locations to adequately describe existing local conditions.
- Include estimates for existing and projected (20 years hence) noise levels in terms of

 (a) Ldn or CNEL and (b) any future noise regulations to be adopted by the City.
 Those existing and projected noise levels shall be compared to the adopted policies
 of the Newark General Plan Noise Element.
- Include recommended mitigation measures to achieve compliance with the adopted policies and standards of the Newark General Plan Noise Element. Where the noise source in question consists of intermittent single events, the report should address the effects of maximum noise levels in sleeping rooms and potential sleep disturbance issues.
- Include estimates for interior and exterior noise exposure after the prescribed mitigation measures have been implemented.
- Describe a post-project assessment program that could be used to evaluate the effectiveness of the proposed mitigation measures.
- **MM NOI-2: Project Fixed-Source Noise.** The noise from all mechanical equipment associated with the project, including air conditioning units, shall comply with the requirements in Section 17.24.100, Paragraph A.2.a in the Newark Municipal Code. When the actual on-site equipment is selected, a noise analysis shall be prepared by a qualified acoustical engineer and the equipment shall incorporate measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels that may affect nearby properties, including adjacent homes. Noise levels from the project's fixed-source equipment at any point outside of the property plane will not exceeds 70 dB(A) between the hours of 7:00 A.M. and 9:00 P.M. or 60 dB(A) between the hours of 9:00 P.M. and 7:00 A.M.
- **MM NOI-3: Construction Activity.** All construction and remediation activity shall follow the City's time and noise reduction measure requirements for construction activity. Development of the project site shall include the following construction-noise mitigation measures, to reduce noise impacts from project construction to a less than significant level.
 - Restrict noise-generating activities at the construction site or in areas adjacent to the construction site to the hours of 7:00 A.M. to 6:00 P.M., Monday through Friday, and between 8:00 A.M. to 5:00 P.M. on Saturdays. Construction shall be prohibited on Sundays and holidays.



Draft Environmental Impact Report Environmental Impact Analysis

- Equip all internal combustion engine driven equipment with intake and exhaust mufflers that are in good condition and appropriate for the equipment.
- Unnecessary idling of internal combustion engines shall be strictly prohibited.
- Locate stationary noise generating equipment such as air compressors or portable power generators as far as possible from sensitive receptors.
- Construct temporary noise barriers to screen stationary noise generating equipment when located near adjoining sensitive land uses. Temporary noise barriers could reduce construction noise levels by 5 dBA.
- Utilize "quiet" air compressors and other stationary noise sources where technology exists.
- Route all construction traffic to and from the project site via designated truck routes where possible. Prohibit construction related heavy truck traffic in residential areas where feasible.
- Control noise from construction workers' radios to a point where they are not audible at existing residences bordering the project site.
- The contractor shall prepare and submit to the City for approval a detailed construction plan identifying the schedule for major noise-generating construction activities.
- Designate a "disturbance coordinator" who would be responsible for responding to any local complaints about construction noise. The disturbance coordinator will determine the cause of the noise complaint (e.g., starting too early, bad muffler, etc.) and will require that reasonable measures warranted to correct the problem be implemented. Conspicuously post a telephone number for the disturbance coordinator at the construction site and include in it the notice sent to neighbors regarding the construction schedule.

Level of Significance After Mitigation

Less Than Significant with Mitigation.

Excessive Groundborne Vibration

Impact NOI-2 The proposed project would not result in generation of excessive groundborne vibration or groundborne noise levels.

Impact Analysis

During construction of the proposed project, equipment such as trucks, bulldozers, and rollers may be used as close as 1,000 feet from the nearest sensitive receptor at the Silliman Community Center. Equipment used during project construction could generate vibration levels between 0.00001 PPV and 0.0008 PPV at 1,000 feet, as shown below in Table 3.13-13. All the groundborne vibration levels are below the FTA vibration threshold at which human annoyance could occur of 0.10 PPV. Additionally, construction activities would be temporary in nature and would likely occur during normal daytime working

Draft Environmental Impact Report Environmental Impact Analysis

hours. Therefore, construction vibrations are not predicted to cause damage to existing buildings or cause annoyance to sensitive receptors. As such, implementation of the project would have a less than significant impact related to vibration.

Type of Equipment	Peak Particle Velocity at 1,000 Feet	Threshold at which Human Annoyance Could Occur	Potential for Proposed Project to Exceed Threshold	
Large Bulldozer	0.0004	0.10	None	
Loaded Trucks	0.0003	0.10	None	
Caisson Drilling	0.0004	0.10	None	
Small Bulldozer	0.00001	0.10	None	
Vibratory Roller	0.0008	0.10	None	

Table 3.13-13: Vibration Source Levels for Construction Equipment

Source: Federal Transit Administration Transit Noise and Vibration Impact Assessment Manual (FTA 2018)

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Airport Land Use Plan

Impact NOI-3 For a project located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, the proposed project would not expose people residing or working in the project area to excessive noise levels.

Impact Analysis

The closest airport or private airstrip to the project site is the Moffett Federal Airport, approximately 6.5 miles southwest of the project. The project site is not within the airport influence area, or the 65 dBA CNEL noise contour for Moffett Federal Airport (Santa Clara County 2018). Commercial aircraft overfly the project site at altitudes between 4,000 and 10,000 feet above sea level while approaching or departing the three major commercial airports in the Bay Area: the San Francisco International Airport (approximately 20 miles northwest); the Oakland International Airport (approximately 17 miles northwest); and the Norman Mineta San Jose International Airport (approximately 10 miles southeast). According to the noise exposure maps for the three airports, the project site in not within the 65 dBA CNEL contour for any of the airports (San Francisco 2015; Port of Oakland 2006; San Jose 2020). Therefore, the project would not expose people residing or working in the project area to excessive noise levels from airports, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.14 POPULATION AND HOUSING

This section describes the environmental and regulatory setting for population and housing. It also describes existing conditions and potential impacts related to population and housing that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.14.1 Environmental Setting

Population Trends

Historic Growth

Newark went from a small town to a city during the 1960s and continued to grow through the 1970s, 1980s, and the 1990s at slightly declining rates. There were no significant gains in population between 2000 and 2009. The population of Newark grew by approximately 18 percent between 1970 and 1980 and grew by an additional 18 percent between 1980 and 1990. Population between 1990 and 2000 grew by approximately 12 percent, however, between 2000 and 2010, the population increased by only 0.8 percent. After 2010, the rate of population growth increased again, and the population of Newark grew by 11.3 percent between 2010 and 2020. The City's historic population growth between 1990 and 2022 is summarized in Table 3.14-1.

Year	Population	Change from Previous (Percent)
1990	37,861	
1995	39,213	3.6
2000	42,250	7.7
2005	42,524	0.6
2010	42,592	0.2
2015	44,426	4.3
2020	47,414	6.7
2022	47,229	-0.4

Table 3.14-1: Newark Historic Population Growth

Source: DOF 2007, 2012, 2022a, 2022b

Population provided in this table are estimates and may be different from the census count.

Current and Projected Population

According to the City's General Plan Housing Element, the City had a population of 42,327 in 2010 (City of Newark 2015a). Since 2010, the City's population is estimated to have increased by 10.9 percent to 47,229 people in 2022 (DOF 2022b). By the year 2030, the City's population is expected to increase to approximately 52,100 people (City of Newark 2015a).

According to the ABAG's Projection 2040 and as outlined in Table 3.14-2, the population of Newark is projected to increase to 47,720 by the year 2040 (ABAG 2018). This would represent a one percent increase from the current 2022 estimated population of Newark. The City's General Plan Housing



Draft Environmental Impact Report Environmental Impact Analysis

Element based its population estimates using the ABAG Projections 2013 which is superseded by ABAG Projections 2040 which was adopted in 2018.

Year	Population	Change from Previous (Percent)
2025	45,355	
2030	45,990	1.4
2035	46,355	0.79
2040	47,720	3.0

Table 3.14-2: ABAG Projections 2040

Source: ABAG 2018

Housing Trends

Housing Units and Average Household Size

The reported number of housing units in the City in 1990 was 12,284 units. The City's housing growth increased at a rate of approximately seven percent between 1990 and 2000. The housing growth rate significantly slowed down between the years 2005 and 2015 with the number of housing units increasing by less than one percent during the 10 year period. The rate of growth increased again to approximately 14.5 percent between 2015 and 2020. The City's housing growth between 1990 and 2022 is summarized in Table 3.14-3.

Table 3.14-3: Newark Historic Housing Units Growth

Year	Housing Units	Change from Previous (Percent)
1990	12,284	
1995	12,635	2.9
2000	13,123	3.9
2005	13,409	2.2
2010	13,414	0.04
2015	13,421	0.05
2020	15,371	14.5
2022	15,811	2.9

Source: DOF 2007, 2012, 2022a, 2022b

According to the City's General Plan Housing Element, Newark's average household size was 3.28 persons per household in 2010 (City of Newark 2015a). The Department of Finance (DOF) E-5 Population and Housing Estimates for Cities, Counties and the State estimated that the total housing in Newark, as of January 1, 2022, to be 15,811 units with an average household size of 3.07 persons per household. Additionally of the 15,811 existing units, approximately 15,329 housing units were occupied, resulting in a 3.0 percent housing vacancy rate (DOF 2022b).



Draft Environmental Impact Report Environmental Impact Analysis

Regional Housing Need Allocation

ABAG prepared the RHNA to allocate regional housing growth among different jurisdictions. The RHNA is the state-mandated process to identify the total number of housing units (by affordability level) that each jurisdiction must accommodate in its Housing Element for an eight year period. The RHNA indicated that the City is expected accommodate 1,078 new housing units within the four income levels between 2015 and 2023. Table 3.14-4 summarizes the regional housing needs allocation by income category. It indicates that approximately 54 percent of the housing need will be moderate to upper-income households, and 46 percent will be very low to low income households (ABAG 2013).

Table 3.14-4: Housing Need Allocation

Jurisdiction	Very Low Income (<50% of Area Median Income)	Low Income (50-80% of Area Median Income)	Moderate Income (80-120% of Area Median Income)	Above Moderate Income (> 120% of Area Median Income)	Total
Newark	330	167	158	423	1,078

Source: ABAG 2013

Employment Trends

According to the City's Housing Element, Newark was originally a railroad and manufacturing center. During the 1960s, Newark experienced a boom in housing construction and a 174 percent increase in population and became a community for people commuting to jobs outside of the City. Growth of industry and business from the 1970s onward created more jobs within the City and between 1990 and 2000, the number of jobs in the manufacturing and service sectors more than doubled (City of Redwood City 2015a).

According to the General Plan Housing Element, ABAG projected a 22 percent increase in jobs in Newark between 2010 and 2035. Agricultural and mining jobs in the City have disappeared altogether, and the projections show that will continue to be the case. Retail, manufacturing, and wholesale jobs are anticipated to increase slightly, but most new jobs are anticipated to be in the service sector. According to the General Plan Housing Element, the City's active labor force was approximately 23,706 total persons in 2012 and is projected to increase to 23,350 by 2030. According to the Employment Development Department's (EDD) Monthly Labor Force Data for Cities and Census Designated Places, in December 2021, Newark had a total labor force of 24,700 people with 23,900 employed people and 900 people unemployed which results in a 3.6 percent unemployment rate (EDD 2022).

The City's General Plan Housing Element includes ABAG's employment projections from 2010 through 2030. The Housing Element projected that the City's jobs would increase by 21 percent from 17,930 in 2010 to 21,720 by 2030 (City of Newark 2015a).



3.14.2 Regulatory Setting

State

California Housing Element Law

The state law requires each city and county to adopt a general plan for future growth. This plan must include a housing element that identifies housing needs for all economic segments and provides opportunities for housing development to meet that need. At the state level, Department of Housing and Community Development (HCD) estimates the relative share of California's projected population growth that would occur in each county in the state, based on DOF population projections and historic growth trends. Where there is a regional council of governments, HCD provides the regional housing need to the council. The council then assigns a share of the regional housing need to each of its cities and counties. The process of assigning shares provides cities and counties the opportunity to comment on the proposed allocations. HCD oversees the process to ensure that the council of governments distributes its share of the state's projected housing need.

Each city and county must update its general plan housing element on a regular basis (approximately every five years). Among other things, the housing element must incorporate policies and identify potential sites that would accommodate a county's share of the regional housing need. Before adopting an update to its housing element, a city or county must submit the draft to HCD for review. HCD will advise the local jurisdiction whether its housing element complies with the provisions of California Housing Element Law.

The councils of government are required to assign regional housing shares to the cities and counties within their region on a similar five-year schedule. At the beginning of each cycle, HCD provides population projections to the councils of government, which then allocate shares to their cities and counties. The shares of the regional need are allocated before the end of the cycle so that the cities and counties can amend their housing elements by mandated deadlines.

Local

Association of Bay Area Governments Projections

ABAG is the official comprehensive planning agency for the San Francisco Bay region, which is composed of the nine counties of Alameda, Contra Costa, Marin, Napa, San Francisco, San Mateo, Santa Clara, Solano, and Sonoma, and contains 101 cities. ABAG produces growth forecasts on fouryear cycles so that other regional agencies, including the MTC and BAAQMD, can use the forecast to make project funding and regulatory decisions. ABAG projections are the basis for the Regional Transportation Plan and the regional Ozone Attainment Plan. In this way, ABAG projections have practical consequences that shape growth and environmental quality. The General Plans, zoning regulations and growth management programs of local jurisdictions inform the ABAG projections. The ABAG projections are also developed to reflect the impact of "smart growth" policies and incentives that could be used to shift development patterns from historical trends toward a better jobs-housing balance, increased preservation of open space, and greater development and redevelopment in urban core and transit-accessible areas throughout the ABAG region.



Draft Environmental Impact Report Environmental Impact Analysis

Regional Housing Needs Allocation

The California Housing Element Law requires local jurisdictions to allow the construction of a share of the region's projected housing needs. This share is called the RHNA. The specific RHNA number for a jurisdiction is important because state law mandates that each jurisdiction provide sufficient land to accommodate a variety of housing opportunities for all economic segments of the community to meet or exceed this number of housing units. ABAG, as the regional planning agency, calculates the RHNA for individual jurisdictions within Alameda County, including Newark.

City of Newark General Plan

The City of Newark's 2015 – 2023 Housing Element was adopted in 2015 and is part of the City's General Plan. The Housing Element contains a description of Newark's population trends, housing characteristics, and employment trends, an analysis of the city's housing needs in relation to RHNA, an overview of sites available for housing, an analysis of potential constraints to housing development, evaluation of the previous housing element, and housing goals and policies. The City's General Plan includes the following goals and policies related to population and housing.

Goal LU-1: Quality of Life. Maintain a desirable quality of life in Newark by preserving a small town neighborhood oriented atmosphere and sustaining a balanced mix of land uses.

- Policy LU-1.2: Growth Focus Areas. Achieve a future growth pattern which includes new neighborhoods on vacant land along the southern and western edges of the city, and infill development in transit served areas such as Old Town and the Greater NewPark Mall Area. Zoning and development review decisions should recognize these areas as the priority locations for growth and change over the next 20 years.
- **Policy LU-1.3: Job Housing Balance.** Seek to balance housing and job growth. The City should strive to have a roughly equal number of jobs and employed residents, with a mix of housing types that meets the needs of the local workforce.
- **Policy LU-1.10: Vacant and Underutilized Sites.** Encourage the development of Newark's remaining vacant and underutilized sites for their highest and best use, consistent with the designations shown on the General Plan Diagram. Future growth in the city should generally be directed to the areas identified in this General Plan.

3.14.3 Environmental Impacts

This section analyzes the project's potential to result in significant population and housing impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following evaluation of potential population, housing, and employment impacts associated with the proposed project was based on data obtained from the U.S. Census, DOF, and applicable planning documents from the City. The following impact discussions consider the impacts of the proposed project related to employment, population, and housing in the City.



Draft Environmental Impact Report Environmental Impact Analysis

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's population and housing impacts are significant. Would the proposed project:

- Induce substantial unplanned population growth in an area, either directly (for example, by proposing new homes and businesses) or indirectly (for example, through extension of roads or other infrastructure)?
- Displace substantial numbers of existing people or housing, necessitating the construction of replacement housing elsewhere?

Project Impact Analysis and Mitigation Measures

Unplanned Population Growth

Impact POP-1 The proposed project would not induce substantial unplanned population growth in an area, either directly (for example, by proposing new homes and businesses) or indirectly (for example, through extension of roads or other infrastructure).

Impact Analysis

This analysis assessed the proposed project's potential to induce substantial population growth. There are two types of population growth: direct and indirect. Direct population growth can occur from the development of new residential units. Indirect population growth can occur from the creation of new employment opportunities or the removal of barrier to growth (e.g., the extension of urban infrastructure to an undeveloped area). The proposed project would not directly or indirectly induce substantial unplanned population growth, as explained below.

For the purposes of this analysis, the 2020 census population of 47,529 will be used to determine projected growth. Additionally, the U.S. Census Bureau identified an average persons per household of 3.32 between the years 2017 and 2021 which will be used to determine the proposed project's resident generation (U.S. Census 2022).

Direct Population Growth

The proposed project would develop 203 single-family detached homes, thereby directly inducing population growth at the project site. The City's General Plan identified that the average household size in the City of Newark was 3.28 persons per household in 2010. According to the U.S. Census Bureau, the City's average household size was 3.32 persons per household between the years of 2017 and 2021 and based on the census bureau's estimated household size, the proposed project would result in approximately 674 new residents.

As of the 2020 census, the City's population is identified at 47,529 people (U.S. Census 2022). As discussed, the City's General Plan estimated that the City's population would grow to 52,100 by 2030. The addition of 674 residents from the proposed project would represent approximately 14.8 percent of the City's growth anticipated by 2030 from the 2020 census population. The addition of 203 single-family



Draft Environmental Impact Report Environmental Impact Analysis

homes would also contribute to the City's RHNA. As discussed, City is expected to accommodate 1,078 new housing units between 2015 and 2023. As such, the proposed project would be within the 2030 population projections anticipated in the City's General Plan and would contribute to the City's RHNA. As the estimated number of residents generated from the proposed project would be within the population projected in the General Plan, it would not result in direct unplanned population growth as the increase in population from the proposed project was anticipated by the General Plan. The proposed project would not result in an increase in population over what was anticipated in the General Plan.

The Areas 3 and 4 Specific Plan planned for a maximum residential unit allocation of 1,260 units within the Specific Plan area. Though not all 1,260 units have been constructed, the remaining number of units available from the allocation has been designated for development by the Sanctuary West Project by its Development Agreement. Therefore, the proposed project's 203 units would not be within the maximum residential unit allocation within the Specific Plan. The proposed project is requesting a General Plan Amendment and rezoning of the site to allow for the development of residential units above the maximum allocation. With the approval of the rezoning and Specific Plan Amendment, the density of the Specific Plan area would be increased by a total of 254 units as the 29.21 acre project site would be zoned RS-6000 which has a maximum density requirement of 8.7 dwelling unit per acre. Though the rezoning and Specific Plan Amendment would result in unplanned population growth within the Specific Plan area, it would not result in substantial unplanned growth as it would still be within the General Plan growth projection for total housing units and resident population for the City and the proposed project would contribute to the City's RHNA. Therefore, the proposed project would not directly induce substantial unplanned population growth and impacts would be less than significant.

Indirect Population Growth

New extensions to area roads or other infrastructure and new employment opportunities are key factors in accounting for possible indirect population growth. The proposed project would not indirectly induce substantial unplanned population growth in the project site because it would not involve any new extensions to area roads or other infrastructure that could enable additional development in currently vacant areas not planned for growth and development in the General Plan. The proposed project includes off-site roadway and infrastructure improvements. The proposed project would include the extension of existing water and sanitary sewer main down Mowry Avenue from its current terminus located on the north side of the UPRR tracks. Additionally, the proposed project would include the extension of potable and non-potable water mains from the southwest corner of the Sanctuary West Project, located to the east of the project site. The potable and non-potable water mains would be jack-and-bored under the ACFC&WCD Line D channel and the UPRR ROW. The non-potable water main would be stubbed on the western edge of the UPRR ROW for future connection and the potable water main would extend northwest along the UPRR ROW, within an existing utility easement, to Mowry Avenue and then down Mowry Avenue to the project entrance at future 'A' Street. However, the extension of the water and sanitary sewer main would not result in indirect population growth as the project site is located near the terminus of Mowry Avenue and vacant developable lands located near the project site are limited. Due to the limitations on developable lands near the project site, the proposed project's extension of utility infrastructure would not result in indirect substantial growth. Therefore, the off-site roadway and utility infrastructure improvements would not result in a removal of barrier of growth and would not result in indirect population growth. Additionally, the proposed project does not include any elements outside of the residential and off-site improvement components and would not include new employment



Draft Environmental Impact Report Environmental Impact Analysis

opportunities. Therefore, the proposed project would not indirectly induce substantial unplanned population growth and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Displacement of People

Impact POP-2 The proposed project would not displace substantial numbers of existing people or housing, necessitating the construction of replacement housing elsewhere.

Impact Analysis

The majority of the project site is currently developed as an auto part and scrap metal salvage lot that includes a warehouse, sales office, workshop, and a large parking area. The northern parcel of the project site is currently undeveloped. The project site does not currently contain residential development and therefore, the construction of the proposed project would not result in the displacement of existing people or housing and would not necessitate the construction of replacement housing elsewhere. The proposed project would have no impact.

Level of Significance Before Mitigation

No Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.



3.15 PUBLIC SERVICES

This section describes the environmental and regulatory setting for public services. It also describes existing conditions and potential impacts related to public services that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.15.1 Environmental Setting

Fire Protection

The ACFD provides fire protection services to approximately 508 square miles in Alameda County, including Dublin, Newark, San Leandro, Union City, and the unincorporated areas. ACFD responds to structure fires, wildland fires, auto fires and extrications, medical emergencies, special rescues, and natural disasters (City of Newark 2013a). There are three fire stations serving Newark. The nearest fire station to the project site is ACFD Station 27, located at 39039 Cheery Street approximately 0.60 miles northeast of the project site.

In the Fiscal Year of 2019 to 2020, the ACFD responded to 42,363 total calls (ACFD 2021). ACFD has established an average response time goal of five minutes or less for 90 percent of the time for the first responding unit for a first alarm assignment, with the remaining units arriving within 10 minutes or less 90 percent of the time (City of Newark 2013b).

Police Protection

The Newark Police Department (NPD) provides law enforcement services to the City of Newark from its headquarters located at 37101 Newark Boulevard approximately 1.75 miles northwest of the project site (City of Newark 2013a). For the fiscal year of 2021 to 2022, the NPD has authorized a staff of one Chief, two Captains, three Lieutenants, eight Sergeants, 45 Police Officers, and 26 non-sworn (civilian) full-time positions (City of Newark 2021).

Schools

The City is served by the Newark Unified School District (NUSD) which operates eight elementary schools, one junior high, one high school, one continuation high school, and one independent study school and serves approximately 6,000 students (NUSD 2020). Birch Grove Primary Elementary School (Kindergarten through grade 2), Birch Grove Intermediate Elementary School (Grade 3 through 6), Newark Junior High and Newark Memorial High School serves the area surrounding the project site (NUSD 2021). In the 2021 to 2022 school year, Birch Grove Primary had an enrollment of approximately 360 students, Birch Grove Intermediate had an enrollment of 422 students, Newark Junior High had an enrollment of 679 students, and Newark Memorial High had an enrollment of 1,634 students (California Department of Education 2022).



Draft Environmental Impact Report Environmental Impact Analysis

Parks

The Newark Recreation and Community Services Department operates and maintains 131 acres of City parks and several recreational facilities. Of this total, 121 acres are owned by the City and 10 acres are leased from the Newark Unified School District. There are 13 parks in the City, including eight neighborhood parks, three community parks, and two special use parks (City of Newark 2013a). The closest park to the project site is the Silliman Center Sports Fields, which is a 29.6 acre community park that provides sports fields, an aquatic center, and a community activity center which includes amenities such as, but not limited to, gymnasiums, locker rooms, dance studio, and fitness center. The Silliman Center Sports Fields is located on Mowry Avenue at Cherry Street, approximately 0.25 mile from the project site.

3.15.2 Regulatory Setting

State

California Building Code

Title 24 of CCR, also known as the California Building Code, is a compilation of three types of building standards from three different origins:

- Building standards that have been adopted by state agencies without change from building standards contained in national model codes,
- Building standards that have been adopted and adapted from the national model code standards to meet California conditions, and
- Building standards authorized by the California legislature that constitute extensive additions not covered by the model codes that have been adopted to address particular California concerns.

The CFC is a component of the California Building Standards Code and contains fire safety-related building standards.

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:

Goal PR-2: Parkland Acquisition and Expansion. Expand and improve Newark's parks and recreational facilities to meet the existing and future needs.

• **Policy PR-2.2: Parks in New Development.** Require new parks to be provided within largescale new development. Where the provision of an on-site park is infeasible, require the payment of an in-lieu fee for parkland acquisition to serve that development.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy PR-2.3: Park Service Standards.** Establish the following park standards to determine where and how much parkland should be provided in Newark, and to calculate the amount of inlieu fees where appropriate:
 - Within the City, provide at least 3.0 acres of parkland per 1,000 population. This total shall exclude wetlands and other areas that are not accessible for active or passive recreation;
 - Provide one neighborhood park per 5,000 population, with a park located within ¹/₂-mile of each residence; and/or
 - Provide one community park per 15,000 population, with a park located within 2 miles of each residence.

These standards may be adjusted to facilitate high value and unique facilities such as linear trails, dog runs, formal gardens, and indoor facilities.

• **Policy PR-2.4: Pocket Parks.** Allow a portion of the parkland dedication requirement to be met through the provision of on-site pocket parks and play lots in new development.

Goal CSF-1: Community Services. Maintain community services and civic facilities that are readily accessible and respond to the needs of all Newark residents.

• **Policy CSF-1.1: Planning for Public Facilities.** Plan for adequate public facilities to meet Newark's current and future needs, based on demographic forecasts, fiscal and budgetary conditions, and adopted standards for municipal facilities and services.

Goal CSF-2: Educational Facilities. Provide excellent schools that deliver high-quality educational services to Newark students while serving the neighborhood centers and fostering civic pride.

• **Policy CSF-2.2: Mitigation of School Impacts.** When new residential development is approved, require mitigation of school impacts to the full extent permitted by law. Work collaboratively with the Newark Unified School Districts to ensure that appropriate fees are collected and other appropriate mitigation measures are taken.

Goal CSF-4: Public Safety. Provide responsive police, fire, and emergency medical services that ensure the safety of residents, employers, and visitors.

- **Policy CSF-4.1: Police Services.** Maintain professional, efficient, effective Police Department activities which promote a high level of public safety.
- **Policy CSF-4.2: Emergency Medical Services.** Ensure the provision of high-quality emergency medical response services, including paramedics and emergency medical technicians.
- **Policy CSF-4.4: Fire Prevention and Response Services.** Ensure the provision of fire prevention and response services which minimize fire risks and protect life and property.



3.15.3 Environmental Impacts

This section analyzes the project's potential to result in significant public services impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of the General Plan, General Plan EIR and the Newark Municipal Code.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's public services impacts are significant. Would the proposed project:

- Result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for any of the public services:
 - Fire protection?
 - Police protection?
 - Schools?
 - o Parks?
 - Other public facilities?

Draft Environmental Impact Report Environmental Impact Analysis

Project Impact Analysis and Mitigation Measures

Government F	acilities
Impact PUB-1	The proposed project would not result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times, or other performance objectives for any of the public services:
	Fire protection.
	Police protection.
	Schools.
	Parks.
	Other public facilities.

Impact Analysis

Fire Protection

The project site is located in an area already served by ACFD. However, development of the proposed project would incrementally increase the demand for fire protection services in the area. ACFD would provide fire protection services to the proposed residential development. The project site is located approximately 0.6 miles away from the nearest fire station and based on the short distance, emergency response to the project site would be able to meet the response time goal of five minutes. The proposed project would be required to pay a Development Impact Fee as required by Chapter 3.24 of the Newark Municipal Code. The funds collected from the Development Impact Fee would be used to provide public services and facilities, including fire protection services, within the City. Additionally, the proposed project would be designed and constructed in accordance with the City's fire protection requirements and the California Building Standards Code which includes the CFC. Implementation of the City's fire protection requirements and the California Building Standards Code would reduce the potential for fires at the site by providing fire hydrants throughout the site and ensuring the new development is equipped with fire protection measures such as alarms and sprinklers. Payment of required fees would offset cost of fire protection demands associated with the proposed project and incorporation of fire protection measures into the design of the development would minimize need for fire protection services. Therefore, though the proposed project would increase the need for fire protection services in the area it would not require the construction of new facilities or expansion of existing facilities as the proposed project would include measures to reduce and offset the increased demand. Impacts would be less than significant.

Police Protection

Law enforcement services for the project site would be provided by NPD. Implementation of the proposed project would result in an incremental increase in demand for police protection services at the project site and surrounding areas. The City's General Plan EIR identified that the NPD currently operates at a staffing level of less than 1.0 officers per 1,000 residents, which is below the national average of 1.5



Draft Environmental Impact Report Environmental Impact Analysis

officers per thousand residents (City of Newark 2013b). The City's General Plan EIR determined that buildout of the General Plan would result in a significant impact to police protection facilities as it would require a new building or expansion of the existing building. However, this impact was determined to be mitigated to a less than significant level with implementation of General Plan policies and collection of fees from new developments under the General Plan (City of Newark 2013b). The proposed project would be required to pay a Development Impact Fee as required by Newark Municipal Code Chapter 3.24. Payment of fees is determined to reduce impacts to a less than significant level as payment of impact fees would provide funding for the construction or expansion of policies facilities and funds are set aside to allow for construction of facilities as needed to accommodate future growth. Therefore, with the payment of required fees, the impacts to police protection services would be mitigated to a less than significant level and the proposed project would not require the construction of any unanticipated new or expanded police facilities.

Schools

The proposed project would include development of the project site with 203 single-family residential units and would increase demand for school facilities and services. The City's General Plan EIR identified that for planning purposes, NUSD estimated that on average new residential developments will generate 0.416 students per single-family unit (City of Newark 2013b). Therefore, using the General Plan generation estimate of 0.416 students per single-family unit, development of 203 single-family units would result in 84 new students. NUSD collects school impact fees on new residential construction within district boundaries on a per square foot basis. The fees serve to offset school facility costs associated with serving new students that result from new developments. Under SB 50, school districts may collect fees to offset the costs associated with increasing school capacity as a result of development and under the terms of this statute, payment of fees by property owners and developers is considered to mitigate in full for the purposes of CEQA any impacts to school facilities associated with a project. Therefore, with the payment of required fees, impacts associated with school facilities are considered mitigated and the proposed project would not result in the need for construction of unanticipated new or expanded school facilities and impacts would be less than significant.

Parks

Based on the City's General Plan EIR, the City currently operates at a parkland ratio of 3.01 acres of parkland per thousand residents, which meets the adopted standard. The proposed project would result in an increase of residents in the area of the project site and would result in increased uses of nearby parks. The City requires, as identified under General Plan Policy PR-2.2, that new development either dedicate land, pay a fee in lieu thereof, or both, at the option of the City, for park and/or recreational purposes. The proposed project would provide 4.89 acres of on-site open space. A majority of the on-site open space would be provided through private open space, provided as a rear yard for each home. However, the proposed project includes the development of 0.94 acres of common open space through the construction of a pocket park, landscaping, and bioretention areas. The common open space area located in the center of the project site would include amenities such as a lawn, pedestrian path, and picnic tables. Though the proposed project would provide on-site common open space areas, the common open space areas would not meet the requirements for park facilities and would not contribute acreage to the 3.0 acres per thousand resident parkland standard. Therefore, development of the proposed project may result in the City not being able to meet its adopted parkland standard. The



Draft Environmental Impact Report Environmental Impact Analysis

construction of the common open space would not meet the requirements for park dedication and therefore, the proposed project would be required to pay the full parks impact fee to offset increased demand to park facilities from development of the proposed project. Payment of required fees would ensure funding for the construction or expansion of policies facilities and funds are set aside to allow for construction of facilities as needed to accommodate future growth. Inclusion of open space within the proposed project design and payment of fees would mitigate the impacts on existing park facilities resulting from development of the proposed project and therefore, the impact would be less than significant.

Other Public Facilities

The addition of 203 new single-family residences would create an incremental increase in the demand for library facilities and other public facilities. In accordance with Newark Municipal Code Chapter 3.24, the proposed project would be required to pay Development Impact Fees to offset costs to public facilities. With the payment of required fees, the proposed project would not result in the need for construction of any unanticipated new or expanded public facilities and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.16 RECREATION

This section describes the environmental and regulatory setting for recreation. It also describes existing conditions and potential impacts related to recreation that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.16.1 Environmental Setting

The Newark Recreation and Community Services Department operates and maintains 131 acres of City parks and several recreational facilities. Of this total, 121 acres are owned by the City and 10 acres are leased from the Newark Unified School District. There are 13 parks in the City, including eight neighborhood parks, three community parks, and two special use parks (City of Newark 2013a). The closest park to the project site is the Silliman Center Sports Fields, which is a 29.6 acre community park that provides sports fields, an aquatic center, and a community activity center which includes amenities such as, but not limited to, gymnasiums, locker rooms, dance studio, and fitness center. The Silliman Center Sports Fields is located on Mowry Avenue at Cherry Street, approximately 0.25 mile from the project site.

3.16.2 Regulatory Setting

Federal

There are no federal regulations related to recreation.

State

Quimby Act

Section 66477 of the California Government Code, also known as the Quimby Act, was enacted in 1965 in an effort to promote the availability of park and open space areas in California. The Quimby Act authorizes cities and counties to enact ordinances requiring the dedication of land, or the payment of fees for park and/or recreational facilities in lieu thereof, or both, by developers of residential subdivisions as conditions to the approval of a tentative map or parcel map. The Quimby Act requires the provision of three acres of park area per 1,000 persons residing within a subdivision, unless the amount of existing neighborhood and community park exceeds that limit, in which case the city or county may adopt a higher standard not to exceed five acres per 1,000 residents. The Quimby Act also specifies acceptable uses and expenditures of funds from fees.

Local

City of Newark General Plan

The City of Newark General Plan includes the following goals and policies items relevant to the proposed project and recreation resources discussed in this section:

Goal PR-2: Parkland Acquisition and Expansion. Expand and improve Newark's parks and recreational facilities to meet existing and future needs.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy PR-2.2: Parks in New Development.** Require new parks to be provided within largescale new development. Where the provision of an on-site park is infeasible, require the payment of an in-lieu fee for parkland acquisition to serve that development.
- **Policy PR-2.3: Park Service Standards.** Establish the following park standards to determine where and how much parkland should be provided in Newark, and to calculate the amount of inlieu fees where appropriate:
 - within the city, provide at least 3.0 acres of parkland per 1,000 population. This total shall exclude wetlands and other areas that are not accessible for active or passive recreation,
 - $\circ~$ provide one neighborhood park per 5,000 population, with a park located within $\frac{1}{2}$ mile of each residence, and
 - provide one community park per 15,000 population, with a park located within 2 miles of each residence.

These standards may be adjusted to facilitate high value and unique facilities such as linear trails, dog runs, formal gardens, and indoor facilities.

- **Policy PR-2.4: Pocket Parks.** Allow a portion of the parkland dedication requirement to be met through the provision of on-site pocket parks and play lots in new development.
- **Policy PR-2.8: Natural Features in Parks.** Design new parks to respect and conserve important natural features. Wetlands and other environmentally sensitive areas located within park boundaries should be designated for protection and restored to the greatest extent possible.

3.16.3 Environmental Impacts

This section analyzes the project's potential to result in significant recreation impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of the General Plan, General Plan EIR and the Newark Municipal Code.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's recreation impacts are significant. Would the proposed project:

- Increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated?
- Does the project include recreational facilities or require the construction or expansion of recreational facilities which might have an adverse effect on the environment?



Project Impact Analysis and Mitigation Measures

Existing Parks	
Impact REC-1	The proposed project would not increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated.

Impact Analysis

_ . ..

The proposed project would permanently increase the City's residential population and would increase the use of existing parks and recreation facilities. Nearby park facilities that may be utilized by residents of the proposed project include the Silliman Center Sports Fields, which is the closest park to the project site located approximately 0.25 mile from the project site. As identified under General Plan Policy PR-2.3, the City of Newark has adopted a 3.0 acre of parkland per 1,000 residents standard for planning purposes and its Quimby Act fee is based on this ratio (City of Newark 2013b). As required by General Plan Policy PR-2.2, the City requires that new development either dedicate land, pay a fee in lieu thereof. or both, at the option of the City, for park or recreational purposes. The proposed project would provide 4.89 acres of on-site open space, with the majority of on-site open space being provided through private open space, provided as a rear yard for each home. However, the proposed project also includes the development of 0.94 acres of common open space through the construction of on-site recreational area. landscaping and bioretention areas. The on-site recreational area would be located in the center of the project site and would include amenities such as a lawn, pedestrian path, and picnic tables. This area would be designed to serve residents within the subdivision, helping to avoid over-usage (accelerated deterioration) of other existing parks in the City. However, the construction of common open space areas within the project site would not meet the City's requirements for park dedication and therefore, the proposed project would be required to pay the full parks impact fee to offset the increased demand to existing park and recreational facilities resulting from development of the proposed project. Payment of required fees would ensure funding for the construction or expansion of policies facilities and funds are set aside to allow for construction of facilities as needed to accommodate future growth. Inclusion of open space within the proposed project design and payment of fees would mitigate potential impacts on existing parks and recreational facilities caused by the proposed project. The proposed project would comply with all City standards and requirements related to recreation, and impacts would be less than significant.

Level of Significance Before Mitigation Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Recreational Facilities

Impact REC-2	The proposed project would not include recreational facilities or require the
	construction or expansion of recreational facilities which might have an adverse
	physical effect on the environment.

Impact Analysis

The proposed project would include 0.94 acres of common open space consisting of landscaped areas, a bioretention area, and on-site recreational area. The on-site recreation area would include amenities such as a lawn, picnic tables and a pedestrian path, none of which are expected to cause any adverse physical effects on the environment due to the small size of the park and the limited nature of the proposed improvements. The proposed project would not involve the construction or expansion of any off-site recreational facilities; however, as required by the City's General Plan Policy PR-2.2, the proposed project would be required to pay a parks impact fee to contribute funding for park and recreational facilities citywide. General Plan Policy PR-2.2 requires new developments to provide on-site parks or pay a fee in-lieu for parkland acquisition to serve the development (City of Newark 2013b). Funds collected through the parks impact fee would be used to help fund and pay for the construction or expansion of off-site improvements, the construction of which would be required to demonstrate it would not result in adverse environmental impacts. Therefore, since no adverse environmental impacts associated with construction of on site or off-site recreational facilities would occur, the impact is less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Environmental Impact Analysis

3.17 TRANSPORTATION

This section describes the existing transportation setting and potential effects on the project site and its surrounding area from proposed project implementation. Descriptions and analysis in this section are based on information contained in the Transportation Impact Analysis Report prepared in September 2021 by Fehr & Peers. The document is included in this Draft EIR as Appendix H.

3.17.1 Environmental Setting

Roadway System

Freeways

The proposed project is served by Interstate 880 (I-880), also known as the Nimitz Freeway, and State Route (SR) 84.

I-880 extends in a north-south direction on the east side of the San Francisco Bay, between Oakland in the north, and San Jose in the south. I-880 has four travel lanes in each direction in the project vicinity (three mixed-flow lanes and one High Occupancy Vehicle (HOV) lane), for a total of eight lanes. I-880 has an interchange at Mowry Avenue that provides access to the project site. Near the study area the average daily traffic (ADT) volume is approximately 217,000 vehicles.

SR 84 extends in an east-west direction and is located approximately three miles northwest of the project site. SR 84 has six travel lanes, with three lanes in each direction and a westbound HOV lane. Two interchanges are provided which serve the City of Newark at Thornton Avenue and Newark Boulevard. SR 84 west of the Thornton Avenue interchange is a toll road and is a freeway east of interchange. Near the study area the ADT volume is approximately 68,000 vehicles.

Arterials

The local roadway system within the City is composed of a hierarchy of streets and roads with varying functions. Arterial roads range from two-lane arterials to six-lane arterials that link residential and commercial districts with the freeway network and provide intercity connections. Arterial roads near the project site include Cedar Boulevard, Central Avenue, Cherry Street, Mowry Avenue, and Stevenson Boulevard.

Cedar Boulevard is a four-lane arterial located north of the project site. The roadway extends in the northsouth direction between Haley Street in the north and Stevenson Boulevard in the south. Near Mowry Avenue, Cedar Boulevard has an intermittent center median, Class II bike lanes, and continuous sidewalks on both sides of the street. The speed limit along Cedar Boulevard is 35-40 miles per hour (mph).

Cherry Street is a four-lane arterial located north of the project site. The roadway extends in the northsouth direction with a landscaped median or center two-way left turn lane and turn pockets. Class II bike lanes are provided for most of the street south of Central Avenue, although they are missing at several constraint points including the Mowry Avenue/Cherry Street intersection. Cherry Street provides connections to Fremont, and it becomes Boyce Road south of the Newark City limit. There are continuous



Draft Environmental Impact Report Environmental Impact Analysis

sidewalks on both sides of the street. The posted speed limit is 45 mph south of Central Avenue, and 35 mph north of Central Avenue.

Mowry Avenue, is a six-lane arterial between Cedar Boulevard and I-880, providing the main point of access to NewPark Mall and the project site. Between Cedar Boulevard and Cherry Street, Mowry Avenue narrows to four lanes and is designated a Class III bike route. Between Cherry Street and the UPRR tracks, Mowry Avenue continues as a four-lane road with Class II bike lanes and a Class I shared-use pathway. South of the railroad tracks, Mowry Avenue is one lane in each direction with no designated bicycle facilities. There are continuous sidewalks on both sides of the street north of the railroad tracks. The posted speed limit on Mowry Avenue is 35 mph.

Central Avenue is a four-lane arterial between I-880 and Newark Boulevard with Class II bike lanes. This section has a posted speed limit of 35 mph. West of Newark Boulevard, Central Avenue is designated a Class III bike route with posted speed limits between 35 and 40 mph. There are continuous sidewalks on both sides of the street.

Stevenson Boulevard is the southernmost east-west arterial in Newark. Stevenson Boulevard is a fourlane road with landscaped median with a speed limit of 40 mph, 35 mph west of Cherry Street. Class II bicycle lanes and continuous sidewalks are provided along the entire length of Stevenson Boulevard.

Bicycle Facilities

Existing bicycle facilities in the area consist of Class I paths, Class II lanes, and Class III routes. In the City's General Plan and the City's Pedestrian and Bicycle Master Plan (PBMP), Class I paths are defined as separate, bike paths or multi-use paths, Class II lanes are defined as striped lanes on a street or highway, and Class III routes are defined as signs or pavement markings for shared use with pedestrians or motor vehicles (City of Newark 2013a, 2017).

Most arterials in the project vicinity have an existing bicycle facility. Between I-880 and Cherry Street, Mowry Avenue is a Class III bike facility and Stevenson Boulevard is designated as a Class II. Both Mowry Avenue and Stevenson Boulevard provide both Class II bike lanes and a Class I bike path between Cherry Street and the UPRR tracks. Central Avenue provides a mix of Class II lanes and Class III routes south of I-880. East of Central Avenue, Cherry Street and Cedar Avenue also provide a mix of Class II lanes and Class III routes.

Currently there are no bicycle facilities along the project frontage on Mowry Avenue. There are no sidewalks facilities along the project frontage on Mowry Avenue; however, sidewalks exist on both sides of existing Mowry Avenue north of the UPRR tracks and extending to Cherry Street. The 2017 PBMP proposes several bicycle facility upgrades near the project site. The Class III segments of Mowry Avenue, Central Avenue, and Cedar Boulevard south of I-880 (including the segment of Mowry Avenue adjacent to the project site) are proposed as Class II lanes, while the Class II segment of Stevenson Boulevard is proposed as a Class IV separated bikeway. On Cherry Street, a Class II bike lane is proposed between Thornton and Central Avenue, and Class IV separated bikeways west of Central Avenue.

Transit System

Existing transit service near the project site include local bus service and regional rail services.



Draft Environmental Impact Report Environmental Impact Analysis

AC Transit operates local bus transit services in the East Bay and Transbay bus service to the Transbay Terminal in San Francisco. AC Transit is the primary bus service provider in 13 cities and adjacent unincorporated areas in Alameda and Contra Costa Counties, with Transbay service to destinations in San Francisco, San Mateo, and Santa Clara Counties. Three AC Transit bus routes operate near the project site – Lines 200, 216, and 251. Line 200 travels between the Union City and Fremont BART stations, with a stop at the NewPark Mall every day and at the Silliman Recreation Center on weekends. Lines 216 and 251 both have a weekday western terminus at Ohlone College on Cherry Street and a weekend terminus at the Silliman Recreation Center on Mowry Avenue. Both lines connect to the Fremont BART station, while Line 216 also stops at the NewPark Mall and continues north to the Fremont and Union City BART Stations.

The closest stop to the project site for Lines 200 and 216 is at the Mowry Avenue/Cherry Street intersection, and the closest stop for Line 251 is at the Cherry Street/Jasmine Avenue intersection. On weekends, the closest stop for Lines 216 and 251 is at the Silliman Recreation Center.

For regional rail service, BART provides regional rail service connecting San Francisco, northern San Mateo county, and the East Bay. ACE and Amtrak also provide regional rail service within the San Francisco Bay Area and beyond. Based on BART Monthly Ridership Reports, the average weekday ridership in 2018 was about 412,000 systemwide. The closest BART Station to the project site is the Fremont Station which is located about three miles north of the project site. The station is served by the Richmond-Warm Springs/South Fremont and Daly City-Warm Springs/South Fremont lines and is the penultimate BART station to the south followed by Warm Springs/South Fremont Station. Fremont Station is served by about eight trains per hour, per direction, during the peak periods. Based on BART Monthly Ridership Reports, in the first half of 2019, about 11,800 weekday daily passengers (entries plus exits) use the Fremont BART Station.

The San Joaquin Regional Rail Commission operates ACE commuter rail service of over 85 miles between Stockton and San Jose. It operates a limited number of trains per day with four westbound trains in the morning from Stockton and four eastbound trains in the afternoon from San Jose. The nearest ACE station is in Fremont and is located on Fremont Boulevard near Peralta Boulevard.

Amtrak provides intercity rail service on the Capitol Corridor, connecting Auburn, Sacramento, Emeryville, Oakland, and San Jose. The service provides a limited number of daily round trips. The nearest Amtrak station in Fremont is shared with the ACE station on Fremont Boulevard near Peralta Boulevard. Average ridership at Fremont Station was about 122 passengers per day in 2018, based on Amtrak State Fact Sheets.

3.17.2 Regulatory Setting

State

California Department of Transportation

Caltrans is responsible for planning, designing, constructing, operating, and maintaining all state-owned roadways in Alameda County. The state facilities providing regional access to and from the project site is I-880 and SR 84.



Draft Environmental Impact Report Environmental Impact Analysis

Senate Bill 743

On September 27, 2013, SB 743 was signed into law. The legislature found that with the adoption of the Sustainable Communities and Climate Protection Act of 2008 (SB 375), the state had signaled its commitment to encourage land use and transportation planning decisions and investments that reduce vehicle miles traveled and thereby contribute to the reduction of GHG emissions, as required by the California Global Warming Solutions Act of 2006 (AB 32). SB 743 started a process that will likely change transportation impact analysis as part of CEQA compliance. Changes include the elimination of auto delay, LOS, and similar measures of vehicular capacity or traffic congestion as the basis for determining significant impacts in many parts of California (if not statewide). The new criteria, "shall promote the reduction of GHG emissions, the development of multimodal transportation networks, and a diversity of land uses" (PRC Section 21099(b)(1)). On January 20, 2016, the OPR released revisions to its proposed Draft CEQA guidelines for the implementation of SB 743. In December 2018, the California Natural Resources Agency certified and adopted the CEQA Guidelines update package, including the Guidelines section implementing SB 743 (Section 15064.3). OPR developed a Technical Advisory on Evaluating Transportation Impacts in CEQA (December 2018), which contains OPR's technical recommendations regarding assessment of VMT, thresholds of significance, and mitigation measures. The provisions of CEQA Guidelines Section 15064.3 shall apply prospectively as described in Section 15007. A lead agency may elect to be governed by the provisions of this section immediately. Beginning on July 1, 2020, the provisions of this section shall apply statewide.

Regional

Regional Regulations Metropolitan Transportation Commission

MTC is the transportation planning, coordinating, and financing agency for the nine-county Bay Area, including Alameda County. It also functions as the federally mandated MPO for the region. Plan Bay Area is the Bay Area's Regional Transportation Plan and Sustainable Communities Strategy. Plan Bay Area, adopted jointly by ABAG and MTC July 18, 2013, lays out a development scenario for the region, which, when integrated with the transportation network and other transportation measures and policies, would reduce GHG emissions from transportation (excluding goods movement) beyond the per capita reduction targets identified by CARB. The regional housing and transportation plan adopted by MTC and ABAG on October 21, 2021, is the Plan Bay Area 2050. Plan Bay Area 2050 is a long-range blueprint to guide transportation investments and land-use decisions through 2050 while meeting the requirements of California's landmark 2008 SB 375, which calls on each of the state's 18 metropolitan areas to develop a Sustainable Communities Strategy to accommodate future population growth and reduce GHG emissions from cars and light trucks. The project's relationship to GHG emissions reductions is discussed in detail in Section 3.8, Greenhouse Gas Emissions, of the Mowry Village Project Draft EIR.

Plan Bay Area 2050

The MTC and ABAG's Plan Bay Area 2050 is the Bay Area's RTP/SCS. Plan Bay Area 2050 was adopted jointly by the MTC and ABAG on October 21, 2021 and is a regional long-range plan for housing, economic development, transportation and environmental resilience and charts the course for the future of the San Francisco Bay Area. Plan Bay Area 2050 sets a development pattern for the region that, when integrated with the transportation network and other transportation measures and policies, would reduce



Draft Environmental Impact Report Environmental Impact Analysis

GHG emissions from transportation (excluding goods movement) beyond the per capita reduction targets identified by CARB. An overarching goal of Plan Bay Area is to concentrate development in Priority Development Areas where there are existing services and infrastructure rather than allocate new growth to outlying areas where substantial transportation investments would be necessary to achieve the per capita passenger vehicle, VMT, and associated GHG emissions reductions.

Local

City of Newark General Plan

The following lists goals and policies from the City of Newark General Plan pertaining to transportation that are applicable to the proposed project.

Goal T-2: Pedestrian and Bicycle Circulation. Create a citywide pedestrian and bicycle network that provides safe access to destinations within the city, connects to an integrated regional network, and is accessible to users of all ages, abilities, and means.

- **Policy T-2.1: Promoting Bicycling and Walking.** Promote bicycling and walking as viable modes of transportation for everyday trips as well as for recreation to increase the number of people of all ages, abilities, and means who bicycle and walk.
- Policy T-2.2: Pedestrian Facilities. Work to close gaps in the pedestrian network and improve sidewalk connectivity between residential and commercial areas. Develop curbs, gutters, sidewalks on all remaining Newark streets not yet fully improved to encourage safe, convenient pedestrian travel. Where appropriate, include marked crosswalks at intersections and install pedestrian countdowns at traffic signals to facilitate safe pedestrian movement across City streets.
- Policy T-2.3: Bicycle Network. Maintain and expand an interconnected network of bicycle routes, paths and trails, serving the City's neighborhoods, shopping districts, workplaces, and park and open space areas. The existing bicycle network should be expanded to provide connections to developing areas, including the Dumbarton TOD, the Southwest Residential and Recreational Project, Old Town Newark, and the NewPark Mall vicinity.
- **Policy T-2.5: Connecting to the Region.** Develop bicycle and pedestrian facilities that connect across City boundaries, integrate with larger regional systems, and improve intermodal connections to local and regional public transportation systems.
- Policy T-2.6: Pedestrian and Bicycle Provisions within New Development. Ensure safe and convenient pedestrian and bicycle access to and through new public and private developments. The City will use the development review process to ensure and where appropriate to require provisions for pedestrians and bicycles in new development areas.
- **Policy T-2.9: Recreational Trails.** Develop and maintain trails in parks and open space areas, and between Newark neighborhoods and the City's open spaces.
- **Policy T-2.10: Railroad Crossings.** Ensure that any future grade separated railroad crossings include sidewalks and designated lanes for bicycles.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy T-2.11: Bicycle Parking.** Provide secure, adequate, and easily accessible bicycle parking at key destinations throughout the city, including municipal facilities, schools, and new development. The style and design of bike racks should contribute to overall neighborhood and architectural aesthetics.
- Policy T-2.12: Trails Along Railroads and Utilities. Consider the use of railroad, flood control, and utility rights of way for jogging, biking, and walking trails, provided that safety and operational issues can be fully addressed. Such trails may be considered where the right-of-way is sufficiently wide to address safety considerations, and where a trail project would not interfere with railroad, flood control, or utility operations.

Goal T-5: Vehicle Circulation. A safe, efficient, and well-maintained network of roadways that facilitates vehicle travel in and around the City.

- **Policy T-5.4: Level of Service Standards.** Strive for Level of Service (LOS) "D" or better at all major intersections in Newark. It is recognized that lower levels are projected at some intersections due to future increases in local and regional traffic. Decreased in the desired LOS may be acceptable in certain intersections due to conditions beyond the City's control, or to achieve other mobility and economic development objectives.
- **Policy T-5.9: Emergency Access.** Improve the street system as necessary to facilitate emergency vehicle response and to provide multiple route options in the event a road is blocked by a emergency or is otherwise made impassable.

City of Newark Pedestrian and Bicycle Master Plan

The PBMP was approved by Newark City Council on February 23, 2017 and is a comprehensive planning document that provides a vision for Newark's future biking and walking environment. The PBMP classifies the following five types of bicycle facilities (City of Newark 2017):

- **Class I Bicycle Paths or Multi-Use Paths** provide a completely separate right-of-way and is designated for the exclusive use of bicyclists and pedestrians with minimal vehicle and pedestrian cross-flow. Bike paths are for non-motorized use only.
- **Class II Bicycle Lanes** provide a restricted right-of-way and is designated for the use of bicycles with a striped lane on a street or highway. Bicycle lanes are generally at least five feet wide. Vehicle parking and vehicle/pedestrian cross-flow are permitted. Class II bicycle lanes are generally indicated on streets with speeds higher than 30 miles per hour.
- Class III Bicycle Routes provide a right-of-way designated for shared use with pedestrians or motor vehicles by signs or pavement markings. A Shared-Use Arrow (or "Sharrow") can be marked in the outside lane on a Class III route to show the suggested path of travel for bicyclists. A sign stating "Bicycles Allowed Full Use of Lane" citing the California Vehicle Code is often included.
- **Class III Bicycle Boulevards** are designed for shared bicycle use with motor vehicles, similar to bicycle routes. The key differentiator is that they are lower volume and lower speed roadways and typically include traffic calming.



Draft Environmental Impact Report Environmental Impact Analysis

> Class IV Separated Bikeways maximize protection for bicyclists in providing a physical separation between the bikeway and vehicular traffic. The separation may include, but is not limited to grade separation, inflexible physical barriers, or on-street parking. Separated bikeways, or cycle tracks, typically operate as one-way bikeway facilities in the same direction as vehicular traffic on the same side of the roadway.

Existing bicycle facilities in the area consist of Class I paths, Class II lanes, and Class III routes. The PBMP proposes several bicycle facility upgrades near the project site. The Class III segments of Mowry Avenue, Central Avenue, and Cedar Boulevards south of I-880 (including the segment of Mowry Avenue adjacent to the project site) are proposed as Class II lanes, while the Class II segment of Stevenson Boulevard is proposed as a Class IV separated bikeway. On Cherry Street, a Class II bike lane is proposed between Thornton and Central Avenues, and Class IV separated bikeways west of Central Avenue.

3.17.3 Environmental Impacts

This section analyzes the proposed project's potential to result in significant transportation impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for VMT Analysis

On September 27, 2013, California Governor Jerry Brown signed SB 743 into law and started a process that changed the way transportation impact analysis is conducted as part of CEQA compliance. These changes include elimination of automobile delay, LOS, and other similar measures of vehicular capacity or traffic congestion as a basis for determining significant impacts under CEQA. According to SB 743, these changes are intended to "more appropriately balance the needs of congestion management with statewide goals related to infill development, promotion of public health through active transportation, and reduction of greenhouse gas emissions."

In December 2018, the OPR completed an update to the CEQA Guidelines to implement the requirements of SB 743. The Guidelines state that generally, VMT is the most appropriate measure to determine significant transportation impacts. The Guidelines require all lead agencies in California to use VMT-based thresholds of significance in CEQA documents published after July 2020. In OPR's Technical Advisory on Evaluating Transportation Impacts in CEQA (December 2018), recommendations are provided on how local agencies can identify and address VMT impacts. The OPR Guidelines state that local agencies have the discretion to develop and adopt their own or rely on thresholds recommended by other agencies. The City of Newark has not adopted their own specific VMT guidelines.

The OPR Guidelines recommend developing screening criteria for development projects that meet certain criteria that can readily lead to the conclusion that they would not cause a significant impact on VMT. The OPR Guidelines also recommend evaluating VMT impacts using an efficiency-based version of the metric, such as VMT per resident for residential developments and/or VMT per worker for office or other employment-based developments. Since City of Newark has not developed their own screening criteria or thresholds of significance, this analysis uses the screening criteria and thresholds of significance recommended by the OPR Guidelines.



Draft Environmental Impact Report Environmental Impact Analysis

VMT refers to the amount and distance of automobile travel attributable to a project. More specifically, VMT measures the per capita number of car trips generated by a project and distances cars will travel to and from a project, rather than congestion levels at intersections (e.g., LOS, graded on a scale of A - F). VMT is typically an output from travel demand models and is calculated based on the estimated number of vehicles multiplied by the distance traveled by each vehicle. This analysis is based on total VMT per population, where VMT includes all automobile trips with an origin and/or destination within the analyzed geographic area generated on a typical weekday. Population is defined as the total number of residents in the analyzed geographic area.

Prior to a full VMT analysis, screening criteria is used to readily determine if the project would not cause a significant impact on VMT. The screening criteria considers the project size and estimated trip generation, location in a low-VMT generating area, and location near high-quality transit. A project must meet at least one of the screening criteria to be presumed to result in a less than significant impact. If the project does not meet any screening criteria, a detailed VMT study is carried out.

The VMT analysis uses the Alameda County Transportation Commission (CTC) Countywide Travel Demand Model to estimate VMT. The Model includes a year 2020 scenario, which approximates existing conditions. The Bay Area regional average daily VMT per capita is 19.8 and the City of Newark citywide average daily VMT per capita is 22.8 under 2020 conditions.

Consistent with OPR's Guidelines, the following thresholds are used to determine if the proposed project would have a significant impact on VMT:

• For residential uses, the project would cause substantial additional VMT if project generated VMT exceeds existing citywide average household VMT per capita minus 15 percent.

VMT Screening Analysis

According to the OPR Guidelines, screening thresholds can be used to quickly identify projects that can be expected to cause a less than significant impact without conducting a detailed study. OPR's recommended screening thresholds and their applicability to the proposed project are described below.

Small Projects – Projects that generate fewer than 110 trips per day generally may be assumed to cause a less than significant VMT impact. As shown in Table 3.17-1, the proposed project would generate more than 110 trips per day and would not meet this screening threshold.

Low-VMT Area – Residential projects located in areas with low-VMT (i.e., 15 percent below the citywide average), that incorporate similar features (i.e., density, mix of uses, transit accessibility), are expected to exhibit similarly low VMT and cause a less than significant VMT impact. Based on the results of the Alameda CTC Model, the proposed project is not located in an area with VMT per capita below the threshold. Thus, the proposed project is not located in a low-VMT area and does not meet this screening threshold.



Table 3.17-1: Project Trip Generation Summary

Land Use	Units ¹ ITE Code	Deilte	AM Peak Hour			PM Peak Hour			
		Code	Daily	In	Out	Total	In	Out	Total
New Uses									
Single-Family Detached Housing	2044	210 ²	2,000	37	113	150	127	74	201
Adjustments									
Existing Uses (Pick-n-Pull) ³			-920	-11	-3	-14	-16	-38	-54
Net New Vehicle Trips			1,080	26	110	136	111	36	147

Notes:

DU = Dwelling units. 1.

ITE Trip Generation (10th Edition) land use category 210 (Single-Family Detached Housing): 2

Daily: Ln(T) = 0.92 * Ln(X) + 2.71

AM Peak Hour: T = 0.71 * (X) + 4.80 (25% in, 75% out) PM Peak Hour: Ln(T) = 0.96 * Ln(X) + 0.20 (63% in, 37% out)

Existing use trip generation based on counts collected in March 2019. 3

The 2021 Fehr & Peers Draft Transportation Impact Analysis Report prepared for the proposed project references 204 DUs. 4. As 204 DUs is more conservative than the proposed 203 DUs, this table reflects project trip generation based on 204 DUs. Sources: ITE Trip Generation Manual, 10th Edition; Fehr & Peers, 2021.

Near Transit Stations – Projects located within 0.5-mile of an existing major transit stop² are expected to generate low VMT and cause a less than significant VMT impact. The Fremont BART Station is the nearest major transit top to the project site, and the proposed project is about three miles from the BART station. Since the project site is more than 0.5 miles walking distance from the BART station, the proposed project is not located near transit stations and does not meet this screening threshold.

The proposed project would not meet any of the OPR's applicable screening thresholds. Therefore, a detailed VMT analysis was carried out and is presented below.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's transportation impacts are significant. Would the proposed project:

- Conflict with a program, plan, ordinance or policy addressing the circulation system, including • transit, roadway, bicycle and pedestrian facilities?
- Conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b)?
- Substantially increase hazards due to a geometric design feature (e.g., sharp curves or • dangerous intersections) or incompatible uses (e.g., farm equipment)?

² According to the California Public Resources Code, § 21064.3, 'Major transit stop' is defined as a site containing an existing rail transit station, a ferry terminal served by either a bus or rail transit service, or the intersection of two or more major bus routes with a frequency of service interval of 15 minutes or less during the morning and afternoon peak commute periods.



Draft Environmental Impact Report Environmental Impact Analysis

• Result in inadequate emergency access?

Project Impact Analysis and Mitigation Measures

Conflict with Program, Plan, Ordinance, or Policy

Impact TRANS-1 The proposed project would not conflict with program plan, ordinance or policy addressing the circulation system, including transit roadway, bicycle and pedestrian facilities.

Impact Analysis

The proposed project does not conflict with the General Plan Circulation Element, or any other program, plan, ordinance or policy addressing the circulation system. The proposed project does not propose to amend or adjust roadway classifications, the roadway network, transit routes, or bicycle network as identified in the City's General Plan or PBMP.

Summary of Mowry Avenue Roadway Improvements

Mowry Avenue along the project frontage to the UPRR rail crossing is proposed to be widened to include a single travel lane provided in both directions, a center median with left turn pocket, bike lanes with buffers, landscaped parkways, stormwater treatment, and sidewalk along the project frontage. The proposed project would widen the ROW of Mowry Avenue, south of the UPRR tracks, from 49.5 feet to 96 to 104 feet to accommodate one 12 foot vehicle lane in the southbound direction, one 12 foot vehicle lane in the northbound direction, and a 12 foot wide median with left turn pocket to access the project site. A 6 foot bicycle lane with 3 foot buffer would also be provided in each direction of travel. A 5 foot parkway strip, 5 foot sidewalk, and 3 foot landscape strip on the northbound side would be provided with a 4 foot landscape strip and a minimum 10 foot setback from face of curb to the top of bank of the ACFC&WCD's Line B channel on the southbound side.

The proposed sidewalk along the eastern frontage of Mowry Avenue would conform to the existing UPRR crossing to the north. The proposed project would provide pedestrian crossing improvements at the UPRR crossing, which would be equipped with a crossing arm, upgraded roadway panels, signage, striping, and pedestrian path improvements to encourage safer pedestrian access to the proposed project, surrounding development, and recreation facilities. The UPRR crossing would also include any required gate signals, visual, and/or audio equipment, as required by UPRR or the Newark Municipal Code.

Additionally, existing Mowry Avenue north of the UPRR tracks and extending to Cherry Street will be restriped and a mid-block crossing to the Silliman Center will be constructed. Re-striping the road will eliminate one travel lane in the southbound direction to accommodate a single 14 foot vehicle travel lane, a 3 foot bike buffer, a 6 foot bike lane and a 10 foot parking lane matching the northbound side of Mowry Avenue. These striping improvements will accommodate the proposed mid-block crossing proposed at the Silliman Center.

The mid-block crossing will be located approximately mid-point between the UPRR tracks and Cherry Street, along the Silliman Center frontage. Improvements will include construction of high visibility crosswalk markings, RRFBs, advance pedestrian crossing yield markings, advance pedestrian crossing signage, median refuge, curb extensions, and ADA compliant curb ramps.


Draft Environmental Impact Report Environmental Impact Analysis

Sidewalks would be provided along both sides of each private street and would connect to sidewalks along Mowry Avenue to be developed as part of the proposed project. Construction of the proposed project would include a sidewalk along the project frontage on Mowry Avenue. The sidewalk would connect to the proposed crossing provided at the UPRR tracks.

The residential development would be accessible directly from Mowry Avenue and would be oriented along several internal streets serving the neighborhood. The proposed private streets include three east/west oriented roadways referred as, "A" Street, "B" Street, and "C" Street, with "A" Street and "B" Street functioning as the main arterials through the neighborhood. The proposed private streets also include four north/south oriented roadways referred as, "D" Avenue, "E" Avenue, "F" Avenue, and "G" Drive, all of which intersect with the east/west oriented roadways within the residential development. The proposed private streets would total approximately 7.3 acres of the project site. Each street would include two travel lanes, parking and sidewalks on both sides, and trees along the frontages. Two courts are proposed within the project site, branching off the private streets. The streets are proposed to be privately owned and maintained by a HOA. Bulb-outs are proposed at street intersections to promote traffic calming and provide shorter street crossings for pedestrians.

Parking supply requirements are based on the Newark Municipal Code, Section 17.23.040 – Required Number of On-Site Parking Spaces. Based on the City's requirements of two spaces per unit for single-family homes (detached), a total of 508 parking spaces would be required. The proposed project would provide 962 parking spaces, including 406 off-street covered spaces, 406 off-street driveway spaces, and 150 on-street guest spaces. With a planned supply of 962 spaces, the proposed parking supply would exceed the City's requirements with a surplus of 454 spaces.

Each of the components described above are further discussed in detail below.

Automobile Access and On-Site Circulation

Motorists would access the project site via two access points located on Mowry Avenue, the only current public access to the project site. These access points connect to the internal street network, which would be private. The internal private streets would have an overall width of 46 feet, with a 36-foot curb-to-curb width, accommodating two-way automobile travel and parallel on-street parking on both sides of the street.

Adjacent to the project site, Mowry Avenue does not have sidewalks or bicycle facilities and the pavement is in generally poor condition. The proposed project's site plan includes improvements to Mowry Avenue, along the project's north frontage. These improvements include widening the current roadway to 48 feet to accommodate two 12-foot vehicle lanes and two 6-foot bike lanes with 3-foot buffers in each direction of travel and a 12-foot wide left-turn lane to access the project site. Mowry Avenue would have a ROW width of 96 to 104 feet. Section 16.12.010 of the City of Newark Municipal code establishes a minimum of 56 feet ROW and a minimum of 36 feet curb-face to curb-face for minor streets. The proposed design for Mowry Avenue would meet the City's standards for both minimum ROW and curb-face to curb-face and the proposed project's private streets would meet the City's standard for minimum curb-face to curb-face.

The proposed project also has a 125-foot-long cul-de-sac with a circular end. Section 16.12.020 of the Municipal Code establishes a maximum cul-de-sac length of 600 feet, with a circular end, a minimum property line radius of 50 feet and a minimum curb radius of 45 feet. The proposed project's site plan



Draft Environmental Impact Report Environmental Impact Analysis

shows a 45-foot curb radius and a 50-foot overall width for the cul-de-sac. Therefore, the cul-de-sac dimensions are consistent with the Municipal Code.

The site plan also includes two courts at the northwest and southeast corners of the site, which are 134foot and 116-foot long respectively. These courts provide a 10.5-foot vehicle lane in each direction, a fivefoot sidewalk on one side only and provide access to the adjacent parcels only.

The proposed project would include several three-way and four-way intersections on-site. The proposed project's traffic study includes recommendations regarding specific traffic control measures to apply to the on-site intersections. Recommendations included in the traffic study include the development of internal intersection control guidelines and installation of stop signs at access points on Mowry Avenue. The internal intersections within the project site would include corner curb extensions, which would reduce the effective width of the street at the intersection approach to 20 feet for approaches with curb extension on both sides of the street and to 28 feet for approaches with curb extensions on only one side of the street. The site plan also shows mid-block curb extensions, which would reduce the street width to 30 feet.

Assuming a prevailing automobile speed of 25 mph or less, all internal project streets would provide adequate sight distance between vehicles traveling in conflicting directions and between vehicles and pedestrians. In addition, the bulb-outs at mid-block and intersection locations proposed through the site would reduce the effective width of the streets and result in lower travel speeds for automobiles.

A non-CEQA traffic operations analysis was also conducted that included seven study area intersections evaluated under existing and future conditions without and with the proposed project. The traffic operations analysis was prepared in compliance with the City's General Plan Policy T-5.4 which outlines the desired operational conditions for the City's roadways.

Based on the discussion above, the proposed project would not conflict with a program, plan, ordinance or policy addressing the circulation system, automobile access, and on-site circulation.

Bicycle Parking, Access, and On-Site Circulation

Bicycle users would access the project site via two access points located on Mowry Avenue, using the Class II bike lanes on Mowry Avenue proposed by the project. No short- or long-term bicycle parking is shown on the site plan and the Municipal Code does not require any bicycle parking for single-family housing units. Bicyclists could use their own garage space to park their bikes. Bicyclists would share the streets with vehicles within the project site, as no dedicated bicycle facilities are included in the project site plan. Though no designated bicycle facilities are provided on internal private streets within the project site as the streets are not wide enough to accommodate the inclusion of a designed bike lane, Class II bike lanes would be provided on Mowry Avenue.

The City of Newark PBMP includes a policy to ensure safe and convenient pedestrian and bicycle access to and through new public and private developments. It requires that new developments to provide secure, adequate and easily accessible bicycle parking. However, single-family dwelling units are exempted from this requirement. The PBMP also shows Mowry Avenue as an existing Class II bike lane between the project site and the railroad crossing, although the bike lanes do not appear striped on the road currently (City of Newark 2017).



Draft Environmental Impact Report Environmental Impact Analysis

The General Plan identifies Mowry Avenue as an arterial road and it emphasizes that design of arterial road should adopt the Complete Streets concept, where local thoroughfares are transformed by incorporating sidewalks, crosswalks, space for bicycles and other amenities that consider the needs of all road users (City of Newark 2013a).

Currently, no sidewalks or bike lanes are provided on either side of Mowry Avenue between the project site and the railroad crossing. The project proposes six-foot bike lanes in each direction of Mowry Avenue along the project frontage, through the railroad crossing to connect to the existing bike lanes just north of the railroad crossing. The proposed project would complete the installation of any bicycle crossing improvements at the railroad crossing to connect the new proposed bike lane to the existing bike lanes just north of the railroad crossing. Thus, the proposed project would be consistent with the PBMP and the General Plan.

Based on the discussion above, the proposed project would not conflict with a program, plan, ordinance or policy addressing bicycle parking, access, and on-site circulation.

Pedestrian Access and On-Site Circulation

Pedestrians would access the project site via the two access point on Mowry Avenue, along a 10-foot wide sidewalk/path along the east side of Mowry Avenue. All internal streets within the proposed project would provide five-foot sidewalks on both sides of the streets, except for the courts that would provide a sidewalk only on one side. The site plan shows directional curb ramps at all internal project intersections. Directional curb ramps would provide a more direct path for pedestrians to cross the street. Currently, no sidewalks are provided on either side of Mowry Avenue between the project site and the railroad tracks. The site plan proposes a 10-foot wide sidewalk/path on the east side of Mowry Avenue, along the project's frontage. The sidewalk constructed along the project frontage would connect to the proposed crossing at the UPRR tracks and the existing sidewalk/path just north of the railroad tracks.

The project site plan does not identify any marked crosswalks, either internally or at the two access driveways on Mowry Avenue. The crosswalk guidelines in the PBMP specify that new controlled intersections, which includes intersections with stop signs, should include marked crosswalks on all legs of the intersection that serve a key desire line, and advanced stop bars in advance of each crosswalk. The crosswalk guidelines provide treatment options for uncontrolled crossing locations with 20 or more pedestrians per hour. However, the internal project intersections are not expected to have any locations with 20 or more pedestrians per hour. The proposed project's traffic study includes recommendations regarding specific traffic control measures to apply to the on-site intersections. A midblock pedestrian crosswalk on Mowry Avenue adjacent to the Silliman Activity and Family Aquatic Center was evaluated to determine if a marked pedestrian crosswalk at this location would be warranted based on the City's Crosswalk Policy. In addition, a crosswalk safety enhancement analysis using the Treatment Toolbox provided in the City PBMP and the FHWA Guide for Improving Pedestrian Safety at Uncontrolled Crossing Locations (2018) was conducted to determine if additional crossing enhancements would be needed were the crosswalk to be marked (Fehr & Peers 2021b). A marked crosswalk on Mowry Avenue just south of Station Road, would be warranted per the City's Crosswalk Policy. In addition, the crosswalk concept should include high visibility crosswalk markings, flashing beacons, such as RRFBs, to enhance pedestrian safety, advance pedestrian crossing yield markings, advance pedestrian crossing signage, median refuge, curb extensions and ADA compliant ramps. These features would be consistent with



Draft Environmental Impact Report Environmental Impact Analysis

FHWA Guidelines and the City's Crosswalk Policy. Based on the discussion above, the proposed project would not conflict with a program, plan, ordinance or policy addressing pedestrian access, and on-site circulation.

Transit Access

AC Transit is the bus service provider for Alameda-Contra Costa counties, including the City of Newark. The nearest bus stops to the project site, as of 2019, are:

- About 0.5 miles east of the project site, on the east side of Mowry Avenue, in front of the Silliman Activity and Family Aquatic Center, that serves both directions of travel. This stop serves AC Transit lines 200, 216 and 251 during the weekends only and do not provide any amenities, except for a sign.
- About 0.8 miles east of the project site, on both sides of Cherry Street, just west of Mowry Avenue. These stops serve AC Transit lines 200, 216 and 629 and do not provide any amenities, except for a sign.
- About 0.8 miles east of the project site on both sides of Mowry Avenue. These stops serve AC Transit lines 251 and 269. The southbound stop provides a trash can and a sign, while the northbound stop provides only a sign.

Lines 200 and 216 provide service to both the Fremont and the Union City BART stations, while Line 251 provides service to the Fremont BART station. Line 629 is a school line that operates twice a day to and from Newark Memorial High School.

Pedestrians would travel between the project site and the bus stops by using the sidewalk on the east side of Mowry Avenue, including the segment of new sidewalk to be constructed by the proposed project between the project site to join the existing sidewalk just east of the railroad crossing. The proposed project would not conflict with a program, plan, ordinance or policy addressing transit access.

At-Grade Railroad Crossing

UPRR owns and operates the railroad crossing on Mowry Avenue to transport freight, while Amtrak uses the crossing for passenger transport. The crossing is a public, at-grade crossing with three tracks. Based on FRA data, about 24 trains use the tracks on a typical day, with a maximum speed of 60 mph. The railroad crossing is identified as US DOT crossing inventory number 749946C and has gate controls for vehicular approaches in both directions. The crossing only has sidewalks on the west side of the tracks but provides an even surface for crossing. However, there are no truncated domes or other detectable warning surfaces for pedestrians. Based on the FRA accident/incident reports, no collisions have been reported at the at-grade railroad crossing in the past ten years.

The General Plan includes a policy of replacing some of the at-grade railroad crossings with gradeseparated rail overpasses, to enhance safety, reduce travel delays and improve emergency access. According to the General Plan, grade separations are planned either at Mowry Avenue or Stevenson Avenue as part of the Southwest Newark Recreation and Residential Project, which includes the proposed project site (City of Newark 2013a). Although the General Plan states that the at-grade crossings would be replaced with grade-separated rail overpasses, the Newark Areas 3 and 4 Specific



Draft Environmental Impact Report Environmental Impact Analysis

Plan proposes an overpass at the Stevenson Boulevard railroad crossing and no improvements at the Mowry Avenue railroad crossing. According to the Specific Plan, advanced preliminary designs have been completed for the Stevenson Boulevard overpass.

The proposed project would provide crossing improvements at the UPRR crossing, which would be equipped with crossing arms, upgraded roadway panels, signage, striping, and pedestrian path and bicycle crossing improvements to encourage safer access to the proposed project, surrounding development, and recreation facilities. The UPRR crossing would also include any required gate signals, visual, and/or audio equipment, as required by UPRR or the Newark Municipal Code. The proposed project's traffic study includes recommendations regarding specific safety enhancements to the Mowry Avenue at-grade railroad crossing. Any proposed improvements must be coordinated with CPUC and affected railroads and all necessary permits/approvals obtained, including a GO 88-B Request (Authorization to Alter Highway Rail Crossings).

The proposed project would not conflict with a program, plan, ordinance or policy addressing at-grade railroad crossing.

Based on the discussion above, the proposed project would not conflict with an applicable plan, ordinance, or policy establishing measures of effectiveness for the performance of the circulation system and impact would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Vehicle Miles Travelled

Impact TRANS-2 The proposed project would conflict or be inconsistent with CEQA Guidelines Section 15064.3, Subdivision(b).

Impact Analysis

According to CEQA Guidelines §15064.3 Subdivision (b)(1), VMT exceeding an applicable threshold of significance may indicate a significant impact. OPR Guidelines recommend evaluating VMT impacts using an efficiency-based metric such as VMT per person. The OPR Guidelines also recommend setting significance thresholds as 15 percent below the citywide or regional average VMT per person. This analysis estimates the VMT per resident for the proposed project and compares it to 15 percent below the citywide average VMT per resident, consistent with the OPR guidelines.

VMT is typically an output from travel demand models and is calculated based on the number of vehicles multiplied by the distance traveled by each vehicle. This analysis uses VMT per resident, as estimated by the Alameda CTC Model. VMT per resident is defined as the total VMT generated by residents with an origin within a geographic area and tracked throughout the regional network on a typical weekday divided by the number of residents in that geographic area.



Draft Environmental Impact Report Environmental Impact Analysis

The Alameda CTC Model, which covers the entire nine county Bay Area, is a regional travel demand model that uses socio-economic data and roadway and transit network assumptions to forecast traffic volumes, transit ridership, and VMT using a four-step modeling process that includes trip generation, trip distribution, mode split, and trip assignment. This process accounts for changes in travel patterns due to future growth and expected changes in the transportation network. This analysis uses the latest version of the Alameda CTC Model, which was released in May 2019. The Alameda CTC Model is based on the prior MTC Plan Bay Area 2040 (i.e., Sustainable Communities Strategy) transportation network and land uses for 2020 and 2040.

As a regional planning tool, the Alameda CTC Model was developed through an extensive model validation process and is intended to replicate existing vehicular travel behavior. Therefore, it can provide a reasonable estimate of the VMT generated in various geographic areas on a typical weekday, as well as estimate future VMT that reflects planned local and regional land use and transportation system changes. The Alameda CTC Model was used to estimate VMT per resident generated by the proposed project, as well as average VMT per resident for the City of Newark under 2020 and 2040 conditions.

Table 3.17-2 summarizes the VMT estimates under 2020 and 2040 conditions. 2020 conditions are used to approximate existing conditions. It is estimated that the project residents would have an average VMT of 27.9 miles per resident per day in 2020 and 25.6 miles per resident per day in 2040.

	-	
	2020 ¹	2040 ¹
Proposed Project (TAZ 940)	27.9	25.6
Average, City of Newark	22.8	20.5
Average, City of Newark minus 15% (i.e., threshold of significance)	19.4	17.4
Percent Difference	44%	47%

Table 3.17-2: VMT Per Resident Summary

Notes:

Based on the outputs of the Alameda CTC Countywide Travel Demand Model

Traffic Analysis Zone (TAZ): A traffic analysis zone is a special area delineated by state and/or local transportation officials for tabulating traffic-related data.

Source: Fehr & Peers, 2021a.

Under 2020 and 2040 conditions, the average VMT per resident for the proposed project would be 44 and 47 percent higher, respectively, than the citywide average minus 15 percent, which is the threshold of significance. Therefore, the proposed project would cause a significant impact on VMT because it would exceed existing citywide household VMT per capita minus 15 percent. Mitigation Measure TRANS-1 would be implemented to reduce project-generated VMT. Mitigation Measure TRANS-1 requires the implementation of a Transportation Demand Management (TDM) Plan that would include measures that discourage the use of single-occupant automobiles and encourage the use of other travel modes. TDM Plan describe implementation of strategies to provide incentives that encourage walking, biking, and transit and reduce private automobile trips and parking demand. TDM strategies include, but are not limited to, offering to provide free parking spaces for car share vehicles and offering carpool matching to project residents. However, it is estimated that a TDM Plan would reduce the project-generated VMT by less than one percent which would not be adequate to reduce the project-generated VMT to below the



Draft Environmental Impact Report Environmental Impact Analysis

threshold of significance. Therefore, the proposed project would have a significant and unavoidable impact related to VMT and a statement of overriding considerations would be required.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

- **MM TRANS-1:** Implementation of a Transportation Demand Management (TDM) Plan. The Applicant shall prepare and implement a TDM Plan prior to the start of construction activities. Impacts on VMT can be reduced through implementing a robust TDM program to reduce VMT through measures that discourage the use of single-occupant automobiles and encourage the use of other travel modes. The TDM Plan would reduce VMT, as well as automobile trip generation and parking demand. Due to the project location, type of development, availability of transit service, and other area characteristics, limited TDM measures would be effective for the proposed project. The TDM Plan could include the following strategies:
 - Explore the feasibility and, if feasible, coordinate with other nearby developments and/or AC Transit to provide shuttle or bus service between the project site and a BART station and/or other major destinations.
 - Offer to provide free parking spaces for at least two car share vehicles (Zipcar, etc.) for residents to utilize to reduce the need for personal vehicle ownership.
 - Offer to provide carpool matching to project residents.

Level of Significance After Mitigation

Significant and Unavoidable Impact.

Hazardous Design Feature

Impact TRANS-3 The proposed project would not substantially increase hazards due to a geometric design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment).

Impact Analysis

The proposed project does not increase hazards due to a geometric design feature or incompatible uses. Development of the project site and site access improvements requires compliance with City development guidelines and code, which follow the General Plan policies and actions that encourage the safe design of arterial roads and streets. Vehicles would enter and exit the project site from the two access points on Mowry Avenue. The project driveways will provide access from Mowry Avenue to the 46-foot private internal street, with a 36-foot curb-to-curb width servicing the residential units. Assuming a prevailing automobile speed of 25 mph or less, all internal project streets would provide adequate sight distance between vehicles traveling in conflicting directions and between vehicles and pedestrians. In addition, the bulb-outs at mid-block and intersection locations proposed through the site would reduce the effective width of the streets and result in lower travel speeds for automobiles.



Draft Environmental Impact Report Environmental Impact Analysis

During construction, traffic management plans will be implemented to ensure the safety of roadway users accessing Mowry Avenue. During construction, the proposed project would generate vehicle traffic through the transport of workers, equipment, and materials to and from the project site. The use of roadways by heavy construction equipment can increase the risk to drivers and cyclists in the vicinity of the project site; however, construction equipment and materials would be stored on-site. The proposed project includes modifications and improvements to existing roadways and may result in partial or complete road closure during construction activities. If road closures or detours are required during construction, the proposed project would be required to prepare and implement a TCP to ensure that it would not result in increased roadway hazards during construction. The proposed project would comply with the City of Newark's TCP Requirements for work area traffic control for work performed in the City's ROWs. Also, there would be no incompatible uses introduced to the project area which could cause vehicle conflicts (e.g., farm equipment). Therefore, impact would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Emergency Access

Impact TRANS-4 The proposed project would not result in inadequate emergency access.

Impact Analysis

The proposed project will not result in inadequate emergency access. Development of the project site will not alter or impede emergency response routes or plans set in place by the City.

Emergency vehicles would access the project site through the same two vehicular access points and use the internal street network. According to the CFC (2016), fire apparatus access roads need to be no less than 20 feet wide and shall always be unobstructed, which the internal project streets meet. The minimum width available for driving or turning movements throughout the project site would be 20 feet. The neighborhood streets would be at least 36 feet wide. The project roadway and neighborhood design would provide adequate turning radii and drive areas for fire trucks and other emergency vehicles. Based on the project site plan, the internal streets and intersection, including the project cul-de-sacs, would accommodate a fire truck. The project site would not be gated and the fire truck would be able to access the project site freely.

The project driveways are designed to comply with turning radius requirements for emergency vehicles and will not cause hazardous driving conditions. The proposed project's detailed design will be completed in compliance with CFC requirements and not impair emergency vehicle access in the vicinity of the project site during construction and in ongoing operation. Compliance with the CFC and CBC will be mandated through the plan check and approval process. This process will also ensure that adequate access for emergency services is provided and the City's emergency response plan will be upheld during construction. Therefore, the proposed project would not result in inadequate emergency access and impacts would be less than significant.



Draft Environmental Impact Report Environmental Impact Analysis

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.18 TRIBAL CULTURAL RESOURCES

This section discusses impacts to cultural resources directly related to Native American tribal cultures that populated the area where the proposed project is located. The distinction for tribal cultural resources is that they are described as a site, feature, place, or cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe. PRC Section 21074(a)(1) and (2) defines tribal cultural resources as "sites, features, places, cultural landscapes, sacred places, and objected with cultural value to a California Native American Tribe" that are either included or determined to be eligible for inclusion in the California Register or included in a local register of historical resources, or a resource that is determined to be a tribal cultural resource by a lead agency, in its discretion and supported by substantial evidence. Cultural resources are generally considered as archaeological or paleontological resources that are typically beneath the surface of the ground and are discovered or uncovered through disturbance of the site. The potential tribal cultural resources impacts associated with the proposed project are identified and discussed herein.

Information in this section is based on the Cultural Resources Assessment prepared by Helix Environmental Planning in December 2021, and included as Appendix D. Where general information is applicable to both Section 3.5, Cultural Resources, and this section, the reader will be referred to Section 3.5 for additional detail.

3.18.1 Environmental Setting

Project Setting

The project site is located within an agricultural and industrial area in the southwestern portion of the City. A portion of the project site is developed with an auto parts and scrap metal salvage lot with three structures and a large asphalt parking area and the northern parcel of the project site is undeveloped, open land.

Ethnographic Setting

Prior to Euro-American occupation, the Newark area was occupied by various tribelets who spoke Ohlone (previously called Costanoan). The Ohlone group designates a language family consisting of eight branches of the Ohlone language that are considered too distinct to be dialects, with each being related to its geographically adjacent neighbors. These groups lived in approximately 50 separate and politically autonomous tribelet areas, each with one or more permanent villages, between the North San Francisco Bay and the lower Salinas River. The Ohlone were politically organized into autonomous tribelets that had distinct cultural territories. Individual tribelets contained one or more villages with several camps for resource procurement within the tribelet territory. The tribelet chief could be either male or female, and the position was inherited patrilineally, but approval of the community was required. The tribelet chief and council were essentially advisors to the community and were responsible for feeding visitors, directing hunting and fishing expeditions, ceremonial activities, and warfare on neighboring tribelets.

The various Ohlone tribes subsisted as hunter-gatherers and relied on local terrestrial and marine flora and fauna for subsistence. The predominant plant food source was acorn, but they also exploited a wide



Draft Environmental Impact Report Environmental Impact Analysis

range of other plants, including various seeds, buckeye, berries, and roots. Protein sources included grizzly bear, elk, sea lions, antelope, and black-tailed deer as well as smaller mammals such as raccoon, brush rabbit, ground squirrels, and wood rats. Waterfowl, including Canadian geese, mallards, green-winged teal, and American widgeon, were captured in nets using decoys to attract them. Fish also played an important role in the Ohlone diet and included steelhead, salmon, and sturgeon.

The Ohlone constructed watercraft from tule reeds and possessed bow and arrow technology, They fashioned blankets from sea otter pelts, fabricated basketry from twined reeds of various types, and assembled a variety of stone and bone tools in their assemblages. Ohlone villages typically consisted of domed dwelling structures, communal sweathouses, dance enclosures, and assembly houses constructed from thatched tule reeds and a combination of wild grasses, wild alfalfa, and ferns.

The Gold Rush brought disease to the native inhabitants and by the 1850s, nearly all of the Ohlone had adapted in some way or another to economies based on cash income. Hunting and gathering activities continued to decline and were rapidly replaced with economies based on ranching and farming (Helix 2021).

Native American Consultation

Helix Environmental Planning contacted NAHC on April 16, 2018, to request a search of their Sacred Lands File (SLF) for the presence of Native American sacred sites or human remains in the vicinity of the study area. A written response was received from the NAHC on May 23, 2018, which stated that the SLF failed to indicate the presence of Native American cultural resources in the vicinity of the project area. On May 29, 2018, Helix Environmental Consulting sent letters to six Native American tribal representatives named by NAHC as potential sources of information related to cultural resources in the area. The letters advised the tribal representatives of the proposed project and requested information regarding Native American resources in the immediate area, as well as feedback or concerns related to the proposed project. The letters noted that the requested information was not for AB 52 or SB 18 consultation, but merely for informational purposes.

Additionally, the City sent letters on February 8, 2022, to 12 Native American tribal representatives included on the NAHC's tribal consultation list for Alameda County for consultation under AB 52 and SB 18 for the proposed project. The AB 52 and SB 18 notification letters included a description of the proposed project and invited the tribe to consult with the City regarding the proposed project under AB 52 and SB 18. Under PRC Section 21080.3.1, AB 52 consultation process is not required to be initiated unless a tribe that is traditionally and culturally affiliated to the geographic area where a project is located requests, in writing, for consultation. The tribe must respond, in writing, within 30 days of receipt of the formal notification and request consultation. The notifications letters were sent on February 8, 2022 and to date, no responses have been received. Therefore, the timing to request AB 52 consultation has lapsed and AB 52 consultation was not conducted.

3.18.2 Regulatory Setting

Refer to Section 3.5, Cultural Resources, for additional federal and state regulations and local policies applicable to tribal cultural resources.



Draft Environmental Impact Report Environmental Impact Analysis

State

Assembly Bill 52 (PRC Section 21084.2)

AB 52 establishes a formal consultation process for California tribes as part of CEQA and equates significant impacts on "tribal cultural resources" with significant environmental impacts (PRC Section 21084.2). AB 52 defines a "California Native American tribe" as a Native American tribe located in California that is on the contact list maintained by NAHC. AB 52 requires formal consultation with California Native American tribes prior to determining the level of environmental documentation if a tribe has requested to be informed of proposed projects by the lead agency. AB 52 also requires that consultation address project alternatives and mitigation measures for significant effects, if requested by the California Native American tribe, and that consultation be considered concluded when either of the parties agrees to measures to mitigate or avoid a significant effect, or the agency concludes that mutual agreement cannot be reached. Under AB 52, such mitigation or avoidance measures must be recommended for inclusion in the environmental document and adopted mitigation monitoring program if determined to avoid or lessen a significant impact on a tribal cultural resource.

California Health and Safety Code and Public Resources Code

Broad provisions for the protection of Native American cultural resources are contained in the HSC, Division 7, Part 2, Chapter 5 (Sections 8010 through 8030). Several provisions of the PRC also govern archaeological finds of human remains and associated objects. Procedures are detailed under PRC Section 5097.98 through 5097.996 for actions to be taken whenever Native American remains are discovered.

Section 7050.5 of the HSC states that any person who knowingly mutilates or disinters, wantonly disturbs, or willfully removes human remains in or from any location other than a dedicated cemetery without authority of law is guilty of a misdemeanor, except as provided in Section 5097.99 of the PRC. Any person removing human remains without authority of law or written permission of the person or persons having the right to control the remains under PRC Section 7100 has committed a public offense that is punishable by imprisonment. PRC Chapter 1.7, Section 5097.5/5097.9 (Stats. 1965, c. 1136, p. 2792), entitled Archaeological and Historical Sites, defines any unauthorized disturbance or removal of remains on public land as a misdemeanor.

Senate Bill 18

SB 18 requires cities and counties to consult with California Native American tribes during the local planning process for the purpose of protecting Traditional Tribal Cultural Places. This allows Native American tribes the opportunity to provide input with respect to the possible preservation of, or the mitigation of impacts on, specified Native American places, features, and objects located within that jurisdiction. This consultation is required prior to amending or adopting any general plan or specific plan or designating land as open space. As noted above, the City contacted NAHC and local tribes in accordance with SB 18 requirements.



Local

City of Newark General Plan

The following lists goals and policies from the City of Newark General Plan pertaining to tribal cultural resources that are applicable to the proposed project.

Goal LU-5: Historic Preservation. Identify, preserve, and maintain historic structures and sites to enhance Newark's sense of place and create living reminders of the city's heritage.

• **Policy LU-5.5: Native American Resources.** Coordinate with local tribal representatives and the Native American Heritage Commission to ensure the protection of Newark's Native American resources ad to follow appropriate mitigation, preservation, and recovery procedures in the event that important resources are identified during development.

3.18.3 Environmental Impacts

This section analyzes the project's potential to result in significant tribal cultural resources impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid that impact.

Methodology for Analysis

The following impact analysis is based on the Cultural Resources Assessment prepared for the proposed project by Helix Environmental Planning in December 2021, which is included as Appendix D. The Cultural Resources Assessment included a records search at the NWIC, literature review, field survey, and search of the SLF from NAHC.

Threshold of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's tribal cultural resources impacts are significant. Would the proposed project:

- Cause a substantial adverse change in the significance of a tribal cultural resource, defined by PRC Section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe, and that is:
- Listed or eligible for listing in the CRHR, or in a local register of historical resources as defined in PRC Section 5020.1(k), or
- A resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of PRC Section 5024.1. In applying the criteria set forth in subdivision (c) of PRC Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe.



Draft Environmental Impact Report Environmental Impact Analysis

Project Impact Analysis and Mitigation Measures

Tribal Cultural Resources

Impact TRIB-1 The proposed project could cause a substantial adverse change in the significance of a tribal cultural resource, defined in Public Resources Code Section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to California Native American tribe, and that is:

- Listed or eligible for listing in the California Register of Historical Resources, or in a local register of historical resources as defined in Public Resources Code section 5020.1(k), or
- ii) A resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of Public Resources Code Section 5024.1. In applying the criteria set forth in subdivision (c) of Public Resource Code Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe.

Impact Analysis

A search of the NAHC's SLF was conducted as part of the Cultural Resources Assessment prepared for the proposed project for the presence of Native American sacred sites or human remains in the vicinity of the project site. A response received from the NAHC stated that the SLF failed to indicate the presence of Native American cultural resources in the vicinity of the project site and a records search of the NWIC did not identify any resources that have been previously recorded within the project site. Additionally, there is no resource within the project site that is listed or eligible for listing in the CRHR or in a local register of historical resources as defined in PRC Section 5020.1(k) or a resource determined by the City to be significant pursuant to criteria set forth in subdivision (c) of PRC Section 5024.1. Therefore, the proposed project is not anticipated to impact any known tribal cultural resources.

However, a records search of the NWIC identified 18 previous cultural resources studies that have been conducted in the area. Results of the NWIC records search identified seven resources that have been recorded within the vicinity of the project site, six of which are identified as prehistoric resources. The prehistoric resources include mounds, habitation debris, and human remains and therefore, the Cultural Resources Assessment identified that the project site has a high potential to contain unknown prehistoric resources which may include tribal cultural resources.

The City initiated AB 52 and SB 18 consultation requirements by sending letters on February 8, 2022, to 12 Native American tribal representatives included on the NAHC's tribal consultation list for Alameda County. Under PRC Section 21080.3.1, AB 52 consultation process is not required to be initiated unless a tribe that is traditionally and culturally affiliated to the geographic area where a project is located requests, in writing, for consultation. The tribe must respond, in writing, within 30 days of receipt of the formal notification and request consultation. The notifications letters were sent on February 8, 2022 and to date, no responses have been received. Therefore, the timing to request AB 52 consultation has lapsed and AB 52 consultation was not initiated.



Draft Environmental Impact Report Environmental Impact Analysis

As the project site is identified as having high potential for unknown prehistoric resources and construction activities required for development of the proposed project would require subsurface work including excavation and earthmoving activities which could potentially damage or destroy previously undiscovered tribal cultural resources, the proposed project would be required to implement mitigation measures to reduce potential impacts. Under Section 21084.3 of the PRC, public agencies shall, when feasible, avoid damaging effects on any tribal cultural resources. To ensure compliance with PRC Section 21084.3, the proposed project would be required to implement Mitigation Measures CUL-1 and CUL-2. identified in Section 3.5, Cultural Resources. Implementation of Mitigation Measure CUL-1 would require the preparation and implementation of a monitoring program which would include, but not limited to, identification of gualified archaeologists, preconstruction briefing for construction personnel on procedures to be followed in the event that unknown resources are encountered during construction, and construction monitoring. Additionally, Mitigation Measure CUL-1 requires consultation with a Native American tribal representative from a tribe that is traditionally and culturally affiliated to the geographic area where the project is located to determine the significance of the discovered resource and ensure proper handling of the resource. In the event of an accidental discovery or recognition of any human remains, the proposed project would be required to implement Mitigation Measure CUL-2 which requires the stoppage of all ground disturbing work so that the County Coroner may determine in the remains are Native American. With the implementation of Mitigation Measures CUL-1 and CUL-2, the proposed project would not cause a substantial adverse change in the significance of tribal cultural resources and impacts would be less than significant.

Level of Significance Before Mitigation

Potentially Significant Impact.

Mitigation Measures

Mitigation Measures CUL-1 and CUL-2 are required.

Level of Significance After Mitigation

Less Than Significant Impact with Mitigation.

3.19 UTILITIES AND SERVICE SYSTEMS

This section describes the environmental and regulatory setting for utilities and service systems. It also describes existing conditions and potential impacts related to utilities and service systems that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.19.1 Environmental Setting

Wastewater Collection/Treatment

The USD provides wastewater collection, treatment, and disposal services to the Cities of Newark, Fremont and Union City. The USD owns and maintains over 783 miles of sanitary sewer pipelines, four lift stations that serve low lying developments, three major pump stations, and 13 miles of dual force mains that transport wastewater to the Alvarado Wastewater Treatment Plan (AWWTP) in Union City (City of Newark 2013b). Wastewater from the City is conveyed to the Newark Pump Station and then to the AWWTP. The USD completed a \$11 million expansion project at the Newark Pump Station in 2010 which was upgraded to help accommodate any increases in flow rates for the foreseeable future. The AWWTP has the capacity to treat an average dry weather flow (ADWF) of up to 33 million gallons per day (mgd). Average wastewater treated in 2020 was 23.16 mgd.

Area 3 of the Specific Plan is within the existing USD service area boundaries, but USD has indicated Area 4 would need to be annexed into their jurisdictional boundaries through Alameda County Local Agency Formation Commission.

Stormwater Management

The ACFC&WCD provides stormwater collection system for the City of Newark. The stormwater system in Newark is managed through a joint partnership between the ACFC&WCD and the City of Newark. Stormwater in the City is managed through a system of gutters, storm drains, channels, and culverts that direct runoff to local creeks and the San Francisco Bay without treatment. All of the storm drainage catch basins and storm drain systems in Newark eventually connect to five different flood control channels.

Water Supply

The City is supplied water by the ACWD which supplies water to customers covering approximately 105 square miles, encompassing the Cities of Fremont, Newark and Union City. The ACWD's primary sources of water supply come from the Bay-Delta provided by the State Water Project (SWP), San Francisco Regional Water System (RWS), and local sources including groundwater from the Nile Cone Groundwater Basin and surface water from the Lake Del Valle (ACWD 2021a).

Before being delivered to ACWD customers, the source water supplies are treated to meet all state and federal drinking water standards. The ACWD operates two surface water treatment plant – Mission San Jose Water Treatment Plant (WTP) and WTP No. 2 - that treats SWP and local surface water from Del Valle Reservoir. According to the ACWD's latest UWMP, the Mission San Jose Treatment Plant is currently decommissioned as a cost saving measure due to low demands (ACWD 2021a). The Newark Desalination Facility (NDF) treats brackish groundwater to remove salts and other impurities, and the

Draft Environmental Impact Report Environmental Impact Analysis

Blending Facility blends San Francisco water with local fresh groundwater (with higher hardness) to provide a blended supply with lower overall hardness. Water resources provided by the ACWD are treated at the WTP No. 2, Newark Desalination Facility, and the Blending Facility before it is supplied to the City. In the 2020 to 2021 fiscal year, the ACWD had an average daily production of 40 mgd and maximum day production of 59.7 mgd. The WTP No. 2 has a 28 mgd capacity, NDF has a 12.5 mgd capacity, and the Blending Facility has a 48 mgd capacity (ACWD 2021b).

The 2020 – 2025 UWMP identified that under normal year water supply conditions, ACWD would have sufficient supplies to meet projected water demands and that during these conditions, ACWD would have sufficient supplied available in excess of the projected demands for placing into groundwater storage for later use in the area in dry years (ACWD 2021a). During critically dry and multiple dry years, the ACWD service area may be facing water supply shortages. Because the Areas 3 and 4 Specific Plan's demands are already factored into the UWMP, the development of these 203 homes would not result in increased shortages, during normal and dry years, beyond those which are already factored into ACWD's planning under current and foreseeable conditions (City of Newark 2014).

Solid Waste

Republic Services provides solid waste collection, disposal, recycling, and yard waste services in in the City. Materials collected by Republic Services of Alameda County are transferred to the Fremont Recycling and Transfer Station where materials are sorted for proper disposal. Non-recyclable materials are then taken to the Altamont Landfill for disposal and green waste/plant debris are transported to several composting facilities in the area. The Altamont Landfill site is approximately 2,063 acres and has a maximum permitted throughout of 11,150 tons per day. The landfill has a maximum permitted capacity of 124,400,000 CY and a remaining available capacity of 65,400,000 CY (CalRecycle 2021b).

Electric Power and Telecommunications

The proposed project would be 100 percent electric and electric services at the site would be provided by PG&E. Telecommunication services would be provided by AT&T and Comcast.

Proposed Project Utility Improvements

The proposed project would install new 8 inch sanitary sewer lines throughout the development, totaling approximately 5,950 linear feet, to connect to a proposed new 8 inch sanitary sewer line within Mowry Avenue. The wastewater flow would be directed north towards Mowry Avenue and routed to a proposed sewer pump station on Mowry Avenue, between the UPRR tracks and the project frontage. The flows would then be conveyed to the existing 8 inch sewer main on the north side of the UPRR tracks. The pump station system would provide redundant dual pump facilities, including backup generators, as required by USD for public use installations and would be designed to function independently in case of overload or mechanical failure. The sewer main extension would be constructed through a jack-and-bore operation under the UPRR tracks. The proposed project's wastewater would then be conveyed south to the Boyce Avenue pump station and ultimately to the AWWTP.

A new 8 inch water main, totaling approximately 6,300 linear feet, is proposed to serve the residents onsite and connect to a proposed new 12 inch water main within Mowry Avenue. The new 12 inch water main is proposed to be extended from the terminus of the existing 16 inch water main on the north side of



Draft Environmental Impact Report Environmental Impact Analysis

the UPRR tracks within Mowry Avenue toward the project site. The water main extension would be constructed through a jack-and-bore operation under the UPRR tracks, totaling approximately 1,850 linear feet. Fire hydrants are proposed throughout the development and along Mowry Avenue consistent with ACWD, ACFD and CFC standards.

In addition, potable and non-potable water mains would be extended approximately 900 linear feet from the southwest corner of the Sanctuary Development, within the old Addition Road alignment adjacent and parallel to ACFC&WCD Line D channel, to the UPRR ROW. The potable and non-potable water mains would be jack-and-bored, approximately 250 linear feet, under the ACFC&WCD Line D channel and the UPRR ROW. The non-potable water main would be stubbed on the western edge of the UPRR ROW for future connection. The potable water main would extend 2,500 linear feet northwest along the UPRR ROW, within an existing utility easement, to Mowry Avenue and then an additional 500 linear feet down Mowry Avenue to the project entrance at future 'A' Street.

The proposed project would utilize LID techniques which may include directing roof runoff to vegetated areas, storm drain stenciling, and site design that promotes infiltration. The storm drain system for the proposed project would consist of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes. New storm drain pipes installed throughout the project site would range from 15 to 24 inches in size and would convey stormwater to the two new on-site bioretention areas. The two bioretention would total approximately 24,665 square feet in size with 6 inches of ponding over the treatment areas. The bioretention treatment areas would be planted with water conserving grass species, shrubs, and trees that are adapted to bio-swale conditions. The bioretention treatment areas would discharge flow through the on-site storm drain system, into the adjacent City-owned open space parcel, consistent with the historic drainage path.

The proposed project would also involve off-site stormwater improvements associated with the widening of Mowry Avenue. These improvements would consist of new storm drains ranging from 15 to 24 inches in size within Mowry Avenue, which would collect and convey flow towards the ACFC&WCD Channel Line B at the terminus of Mowry Avenue. Additionally, the proposed roadway improvements would include four new bioretention treatment areas along Mowry Avenue, totaling approximately 3,212 square feet.

3.19.2 Regulatory Setting

Federal

Clean Water Act

Section 304 of the Clean Water Act establishes primary drinking water standards and requires states to ensure that potable water retailed to the public meets these standards. State primary and secondary drinking water standards are promulgated in California Code of Regulations Title 22, Sections 64431–64501. Secondary drinking water standards incorporate non health risk factors including taste, odor, and appearance. The National Pollutant Discharge Elimination System regulates the discharge of drainage to surface waters. Federal NPDES regulations are administered by the State Water Resources Control Board and through the Regional Water Resources Control Boards. Because the proposed project area drains to the San Francisco Bay, it is under the jurisdiction of the San Francisco Bay RWQCB.



Draft Environmental Impact Report Environmental Impact Analysis

State

Porter Cologne Water Quality Control Act

The State of California established the SWRCB, which oversees the nine RWQCBs, through Porter-Cologne. Through the enforcement of Porter Cologne, the SWRCB determines the beneficial uses of the waters (surface and groundwater) of the state, establishes narrative and/or numerical water quality standards, and initiates policies relating to water quality. The SWRCB and, more specifically, the RWQCB, is authorized to prescribe Waste Discharge Requirements for the discharge of waste, which may impact the waters of the State. Furthermore, the development of water quality control plans, or Basin Plans, are required by Porter-Cologne to protect water quality. The SWRCB issues both general construction permits and individual permits under the auspices of the federal NPDES program.

Urban Water Management Planning Act

In 1983, the California Legislature enacted the Urban Water Management Planning Act (Water Code Sections 10610–10656). The Urban Water Management Planning Act requires that every urban water supplier that provides water to 3,000 or more customers, or that provides over 3,000 acre-feet per year (AFY) shall prepare and adopt an UWMP. Water suppliers are required to prepare a UWMP within a year of becoming an urban water supplier and update the plan at least once every five years. The Urban Water Management Planning Act also specifies the content that is to be included in an UWMP. It is the intention of the legislature to permit levels of water management planning commensurate with the number of customers served and the volume of water supplied. The Urban Water Management Planning Act states that urban water suppliers should make every effort to ensure the appropriate level of reliability in its water service sufficient to meet the needs of its various categories of customers during normal, dry, and multiple-dry years. The Urban Water Management Planning Act also states that the management of urban water demands, and the efficient use of water shall be actively pursued to protect both the people of the state and their water resources. ACWD's 2020 - 2025 UWMP was adopted on May 13, 2021.

California Integrated Waste Management Act (AB 939 and AB 341)

To minimize the amount of solid waste that must be disposed of by transformation (i.e., recycling) and land disposal, the Legislature passed the California Integrated Waste Management Act of 1989 (AB 939), effective January 1990. According to AB 939, all cities and counties are required to divert 25 percent of all solid waste from landfill facilities by January 1, 1995, and 50 percent by January 1, 2000. Solid waste plans are required to explain how each city's AB 939 plan will be integrated within its respective county plan. They must promote (in order of priority) source reduction, recycling and composting, and environmentally safe transformation and land disposal. In 2010, the state legislature passed AB 341 (Chesbro) which set a statewide recycling goal of 75 percent by 2020, which is anticipated to be achieved through source reduction, recycling, and continued diversion of materials such as organic wastes.

Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:



Draft Environmental Impact Report Environmental Impact Analysis

Goal CS-3: Water Resources. Conserve and enhance Newark's water resources.

- **Policy CS-3.1: Protection of Water Resources.** Ensure that land use decisions consider the availability of water for domestic and non-domestic uses, potential impacts on groundwater quality and groundwater recharge capacity, and potential off-site impacts on water quality.
- **Policy CS-3.2: Water Conservation Standards.** Promote water conservation through development standards, building requirements, irrigation requirements, landscape design guidelines, and other applicable City policies and programs.
- **Policy CS-3.3: ACWD Conservation Incentives.** Support Alameda County Water District (ACWD) incentives, which encourage Newark residents and businesses to conserve water.
- **Policy CS-3.7: Wastewater Treatment.** Work with the Union Sanitary District to ensure that sanitary sewer collection and treatment systems are maintained and upgraded to reduce water pollution in San Francisco Bay.

Goal CS-8: Solid Waste Management. Reduce landfill waste through recycling, composting, and source reduction.

• **Policy CS-8.1: Recycling Program.** Actively promote recycling, composting, and waste reduction in order to minimize the amount of waste requiring disposal in landfills. Provide for residential recycling and green waste containers and weekly curbside recycling pickup, to make it as easy and convenient as possible for residents to reduce the volume of trash requiring landfill disposal.

Goal EH-3: Flooding Hazards. Reduce risks to life and property associated with flooding.

- **Policy EH-3.2: Maintaining Drainage Patterns.** Prohibit development, grading, and land modification activities that would adversely affect Newark's drainage system or create unacceptable erosion impacts.
- **Policy EH-3.3: Residential Development in the Flood Plain.** Require that new residential development, including streets and other surface improvements, be constructed above the 100-year flood elevation.
- **Policy EH-3.5: Storm Drain Maintenance.** Manage and maintain the storm drainage system to avoid flooding and reduce the negative effects of stormwater runoff.
- **Policy EH-3.8: Flood Control Improvements.** Work with Alameda County Flood Control and Water Conservation District (ACFC&WCD) on improvements to the storm drain, flood control channel, and levee system which ensure that these systems continue to protect Newark neighborhoods and business districts from flooding.

Goal CSF-5: Infrastructure. Provide safe, reliable, and efficiently operated infrastructure which meets Newark's long-term water, sewer, and stormwater management needs.



Draft Environmental Impact Report Environmental Impact Analysis

- **Policy CSF-5.1: Water Supply.** Work with the Alameda County Water District to ensure a stable supply of clean, safe drinking water for existing and future development in Newark.
- **Policy CSF-5.2: Sanitary Sewer.** Work with the Union Sanitary District to ensure that the sewer system is expanded to serve Newark's new development areas, existing facilities are regularly maintained, sufficient wastewater capacity is provided to meet projected growth, and wastewater effluent is treated to meet all state and federal standards.
- Policy CSF-5.5: Drainage with New Development. Ensure that new development provides drainage and flood protection improvements which reduce on-site and downstream hazards such as ponding, flooding, and erosion. New development areas should be designed to minimize impervious surfaces in order to reduce associated site runoff and maximize groundwater recharge.
- Policy CSF-5.6: Involving Utility Agencies in Development Review. Engage local water, sewer, and stormwater service providers in the review of new development projects to ensure that infrastructure, including water supply and wastewater treatment capacity, is available or will be made available to meet development-related needs.
- **Policy CSF-5.7: Infrastructure Cost.** Ensure that the cost of infrastructure improvements required for new development is the financial responsibility of that development and is allocated based on each project's expected impacts.
- **Policy CSF-5.8: Visual Impact of Utilities.** Minimize the visual impact of public utilities such as transmission lines and wireless communication facilities. Utility lines along new and redeveloped rights-of-way should be placed underground wherever feasible.

3.19.3 Environmental Impacts

This section analyzes the project's potential to result in significant utilities and service systems impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of documents pertaining to the project site, including the General Plan, the General Plan EIR, UWMP, and Section 2.0, Project Description, of this EIR. The following impact discussions consider the impacts of the proposed project related to utilities and service systems in the City.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's utilities and service system impacts are significant. Would the proposed project:



Draft Environmental Impact Report Environmental Impact Analysis

- Require or result in the relocation or construction of new or expanded water, wastewater treatment or storm water drainage, electric power, natural gas, or telecommunication facilities, the construction or relocation of which could cause significant environmental effects?
- Have sufficient water supplies available to serve the project and reasonably foreseeable future development during normal, dry and multiple dry years?
- Result in a determination by the wastewater treatment provider which serves or may serve the project that it has adequate capacity to serve the project's projected demand in addition to the provider's existing commitments?
- Generate solid waste in excess of state or local standards, or in excess of the capacity of local infrastructure, or other impair the attainment of solid waste reduction goals?
- Comply with federal, state and local management and reduction statues and regulations related to solid waste?

Project Impact Analysis and Mitigation Measures

Relocation or Construction of Utility Facilities

Impact UTIL-1 The proposed project would not require or result in the relocation or construction of new or expanded water, wastewater treatment or storm water drainage, electric power, natural gas, or telecommunications facilities, the construction or relocation of which could cause significant environmental effects.

Impact Analysis

Wastewater Treatment

The General Plan EIR determined that buildout of the General Plan would result in a projected wastewater generation rate at buildout of approximately 3.78 mgd. Even with the increase at buildout, the AWWTP would be operating at only 74 percent of its capacity and therefore, would not require the expansion of AWWTP (City of Newark 2013b). The General Plan EIR takes into account anticipated demand from existing Specific Plans and development projects within the City when developing or updating a General Plan and its EIR. Therefore, the Areas 3 and 4 Specific Plan's demands were factored into the General Plan EIR and the proposed project's demands would be adequately handled by the AWWTP's existing capacity.

The proposed project would be required to comply with the USD's Capacity Charge Ordinance and Newark Municipal Code Chapter 3.24, Development Impact Fees. The Capacity Charge Ordinance requires all development applicants to pay capacity fees and be subject to development review process. The Newark Municipal Code Chapter 3.24 requires new developments to pay a fee towards construction of facilities that are required by new development. The funds collected through the fee would ensure funding is provided and set aside to allow for future construction or expansion of facilities as needed to accommodate future growth A new sanitary sewer pump station is proposed on Mowry Avenue, between the UPRR tracks and the project frontage by the Areas 3 and 4 Specific Plan. The construction of the new



Draft Environmental Impact Report Environmental Impact Analysis

pump station is not proposed as part of the proposed project and the construction of the new pump station was planned as part of the development of the Areas 3 and 4 Specific Plan area and analyzed in the Specific Plan EIR. The pump station system would provide redundant dual pump facilities, including backup generators, as required by USD for public use installations and would be designed to function independently in case of overload or mechanical failure. The proposed pump station would pump water to a new 8 inch sewer main within Mowry Avenue, south of the UPRR tracks, prior to connecting to the existing 8 inch sewer main within Mowry Avenue, north of the UPRR tracks. The sewer main extension would be constructed through a jack-and-bore operation under the UPRR tracks. The proposed project's wastewater would be conveyed south to the Boyce Avenue Pump Station and ultimately to the AWWTP. Construction of the utility infrastructure would include excavation, jack-and-bore under the UPRR tracks, and grading. The construction of the new sewer main and extension of the existing sewer main would be completed in accordance with City engineering requirements for the installation of utilities and construction would comply with applicable air quality and noise reduction regulations to ensure that construction of utility infrastructure would not cause significant impacts to the environment. The proposed project would not require or result in the construction of additional wastewater treatment facilities and impacts would be less than significant.

Water Treatment

The proposed project would connect the new 8 inch main proposed to serve the residents on-site to a proposed 12 inch main within Mowry Avenue. The proposed 12 inch main would be extended from the terminus of the existing 16 inch water main on the north side of the UPRR tracks within Mowry Avenue toward the project site. Additionally, existing potable and non-potable water main would be extended from the southwest corner of the Sanctuary West Development, within the old Addition Road alignment adjacent and parallel to ACFC&WCD Line D channel, to the UPRR ROW. The potable and non-potable water mains would be jack-and-bored under the ACFC&WCD Line D channel and the UPRR ROW. The non-potable water main would be stubbed on the western edge of the UPRR ROW for future connection and the potable water main would extend northwest along the UPRR ROW, within an existing utility easement, to Mowry Avenue and then down Mowry Avenue to the project entrance at future 'A' Street. The construction and extension of water mains would be completed in accordance with City engineering requirements for the installation of utilities and construction would comply with applicable air quality and noise reduction regulations to ensure that construction of utility infrastructure would not cause significant impacts to the environment. As discussed under Impact UTIL-2, the proposed project would not require water supplies in excess of the project demand and the proposed project would not require the construction of or expansion of existing water facilities. Additionally, the proposed project would be required to pay its fair share of cost of infrastructure improvements required for new developments. Therefore, impacts would be less than significant.

Stormwater Drainage

The proposed project would include installation of an storm drain systems consisting of bioretention areas, curbs and gutters along the roadways, and underground storm drain pipes and would utilize LID techniques for the system. Storm drains installed on-site would convey stormwater from the site to the two on-site bioretention treatments areas which would then discharge flows through the on-site storm drain system into the adjacent City-owned open space parcel, consistent with the historic drainage path. Additionally, the proposed project would construct improvements to the off-site stormwater system and



Draft Environmental Impact Report Environmental Impact Analysis

storm drains located within Mowry Avenue would collect and convey flow towards the ACFC&WCD Line B channel at the terminus of Mowry Avenue. The construction of improvements would be completed in accordance with the City's engineering standards and construction activities would comply with existing regulations to reduce potential environmental impacts resulting from construction activities. The constructed improvements would not result in changes to the stormwater system that would result in environmental effects. The proposed project developed a Post Construction Stormwater Control Plan which outlines the proposed storm water infrastructure for the proposed project and includes design strategies, measures, and requirements to ensure that the proposed storm water infrastructure is constructed in accordance with City guidelines. As the City's storm water is directed towards local creeks and the bay without off-site treatment, the proposed project would not require the construction of new or expanded stormwater infrastructure. Stormwater infrastructure improvements included in the proposed project would adequately convey stormwater flows on- and off-site and impacts would be less than significant.

Other Infrastructure

The proposed project would be designed and constructed to be 100 percent electric and would not utilize natural gas during operation of the proposed project. PG&E is the electric provider to the City. Although the proposed project would demand additional electricity, electrical connections would be made with existing facilities located on-site. The proposed project would be subject to energy efficiency standards through the California Green Building Code and Title 24. No new expanded facilities would be required for electric facilities that could potentially cause a significant environmental impact.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Water Supply

Impact UTIL-2 The proposed project would have sufficient water supplies available to serve the project and reasonably foreseeable future development during normal, dry and multiple dry years.

Impact Analysis

The Areas 3 and 4 Specific Plan area which includes the project site is served by the ACWD. During the preparation of the Areas 3 and 4 Specific Plan RDEIR, a WSA was prepared by ACWD in November 2008 which indicated that sufficient water supplies exist to meet the ACWD's projected demands as well as the Specific Plan's demands, including the proposed project, under normal year conditions (City of Newark 2015b). Because the demands of the Specific Plan are already factored into the UWMP's demand analysis, the development of the proposed project would not result in increased shortages beyond those already factored into ACWD's planning under current and future conditions. The UWMP forecasts that the ACWD could withstand a repeat of past dry years conditions without any additional shortages.



Draft Environmental Impact Report Environmental Impact Analysis

The Areas 3 and 4 Specific Plan incorporates water efficiency measures recommended by the ACWD and the WSA concluded that there are sufficient supplies to meet the long term demands of the Specific Plan. All residential development located within the Areas 3 and 4 Specific Plan would be developed with water efficient plumbing fixtures and irrigation systems. If future updated ACWD demand forecast determines that there would be future water supply shortages, the ACWD would require the developers of the Specific Plan to provide funding for off-site conservation measures to offset the demand.

The proposed project would develop low density residential units in Sub Area D of Area 4 of the Newark Areas 3 and 4 Specific Plan which was originally proposed to be developed as a golf course. Although the proposed project would not develop the land as it was proposed in the Areas 3 and 4 Specific Plan, the proposed residential units would not result in a substantial increase in water demand for the project site compared to if the project site was developed as a golf course. The WSA prepared for the Areas 3 and 4 Specific Plan identified that with compliance of the requirements provided in the WSA, ACWD would have sufficient water supplies to support the development as part of the Areas 3 and 4 Specific Plan. Additionally, in a letter provided by the ACWD dated January 3, 2022, the ACWD determined that the total water demand for the Specific Plan area has been reduced since the preparation of the WSA in 2008 and therefore, the findings and determination of the WSA that there would be sufficient water supplies available to serve buildout of the Specific Plan area still apply. The WSA prepared for the Specific Plan EIR identified that single-family residential development would have a water demand of 250 gallons per day per unit (City of Newark 2008). Using the water demand factor identified in the WSA, the proposed project's 203 residential units would be anticipated to result in a water demand of 30,450 gallons per day or 34.13 acre feet per year. The golf course development proposed by the Specific Plan for the project site was estimated to result in water demand of 491 acre feet per year. Therefore, the proposed project would demand substantially less water at the site than what was identified in the Specific Plan and there would be sufficient water supplies available to serve the proposed project. Additionally, the WSA identified that total water demand for the buildout of the Specific Plan would be approximately 1,100 acre feet per year (City of Newark 2008). The proposed project's demand would represent approximately three percent of the total demand for buildout of the Specific Plan area. Therefore, the proposed project would not require water in excess of available existing entitlements and resources, and compliance with the requirements provided in the WSA would ensure that the proposed project would not result in insufficient water supplies to serve the proposed project. Therefore, impacts would be less than significant.

Level of Significance Before Mitigation Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Wastewater Treatment

Impact UTIL-3	The proposed project would not result in a determination by the wastewater
	treatment provider, which serves or may serve the project that it does not have
	adequate capacity to serve the project's projected demand in addition to the
	provider's existing commitments.

Impact Analysis

The AWWTP has the capacity to treat an ADWF of up to 33 mgd and average wastewater treated in 2020 was 23.16 mod (USD 2021). As discussed under Impact UTIL-1, the General Plan EIR determined that the increase in the generation of wastewater at buildout of the General Plan, which includes development under the Areas 3 and 4 Specific Plan, would be adequately handled by the AWWTP's existing capacity. As the proposed project's 203 single-family homes would be within the residential allocation identified for the Areas 3 and 4 Specific Plan, the proposed project's demands would be adequately handled by the AWWTP's existing capacity. The proposed project would be reviewed by USD and the City as required by the USD's Capacity Charge Ordinance. The Capacity Charge Ordinance requires all new development projects to be subject to the development review process to ensure that there is adequate capacity by the wastewater treatment provider to serve the proposed project, during which the Newark Public Works department and USD staff would require the applicant to upgrade or expand the USD's collection system if the USD determines the demand from the proposed project would exceed USD's system capacity. USD's Capacity Charge Ordinance also requires all development applicants to pay capacity fees, similar to the Newark Municipal Code Chapter 3.24 which requires new developments to pay a fee towards the construction of facilities that are required by new development. The proposed project would pay all required fees and would be required to complete the City and USD's development review process prior to the issuance of building permits. Therefore, the proposed project would not result in a determination by the wastewater treatment provider that it does not have adequate capacity to serve the proposed project and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Solid Waste

Impact UTIL-4 The proposed project would not generate solid waste in excess of state or local standards, or in excess of the capacity of local infrastructure, or otherwise impair the attainment of solid waste reduction goals.

Impact Analysis

The proposed project would be expected to generate waste during the construction and operation of the proposed project. Construction of the proposed project would involve the demolition of existing structures on-site. The proposed project is estimated to remove approximately 39,000 CY of vegetation, contaminated soil, demolition debris, and other cleared material prior to construction. Construction waste



Draft Environmental Impact Report Environmental Impact Analysis

generated would be disposed of by the project contractor in accordance with the City's established programs that facilitate the diversion and disposal of construction waste. The proposed project would be required to comply City's Green Ordinance under Newark Municipal Code Chapter 15.44, Green Building and Construction and Demolition Debris Recycling. All construction projects whose total costs are greater than \$100,000, or structure demolition projects whose total costs are greater than \$20,000, or pavement demolition projects involving 1,000 square feet of removed pavement are required to divert 100 percent of all Portland cement concrete and asphalt concrete and an average of no less than 50 percent of all remaining construction and/or demolition debris. Additionally, the project Applicant would be required to complete a waste management plan, which would describe the volume of construction/demolition debris that will be recycled, landfilled, or reused, and identification of the receiving facility, and submit the plan to the City prior to the issuance of any building permits.

According to California Department of Resources Recycling and Recovery's (CalRecycle) Disposal Rate Calculator, the City of Newark had a population disposal rate of 4.7 pounds per person per day in 2020 (CalRecycle 2021a). The proposed project is anticipated to generate up to 623 residents and therefore, using the 2020 disposal rate of the City, operation of the proposed project is anticipated to result in approximately 2,928 pounds of waste per day, or 485 tons per year. Solid waste from the proposed project would be disposed of at the Altamont Landfill which has a permitted maximum throughput of 11,150 tons per day and has a remaining capacity of 65,400,000 CY (CalRecycle 2021b). Due to the substantial amount of available capacity remaining at Altamont Landfill, sufficient capacity would be available to accommodate the proposed project's solid waste disposal needs. The proposed project would implement the City's goals, policies, and standards for reducing waste generation such as those under General Plan Goal CS-8. The proposed project would not be expected to generate solid waste in excess of state or local standards and would not impair attainment of solid waste reduction goals. Therefore, a less than significant impact related to solid waste would occur.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Solid Waste Statutes and Regulations

Impact UTIL-5 The proposed project would comply with federal, state, and local management and reduction statutes and regulations related to solid waste.

Impact Analysis

The proposed project would implement all applicable statutes and regulations related to solid waste that has been adopted or implemented by the City. The proposed project would be served by curbside solid waste and recycling services, which are standard services for residential uses in the City. Solid waste disposal must follow the requirements of the contracted waste hauler and disposal facility, which follows local, state, and federal statutes and regulations related to the collection and disposal of solid waste. There are numerous goals, policies, and actions in the City's General Plan related to reducing waste

Draft Environmental Impact Report Environmental Impact Analysis

generation in the City. The proposed project would implement all goals, policies and actions identified in the General Plan such as policies and actions under General Plan Goal CS-8 which requires compliance with the City's waste reduction program and standards set for construction and demolition debris. The proposed project would comply with all federal, state, and local statues and regulations related to solid waste and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

This page intentionally left blank.

3.20 WILDFIRE

This section describes the environmental and regulatory setting for wildfire. It also describes existing conditions and potential impacts related to wildfire that would result from implementation of the proposed project, and mitigation for potentially significant impacts, where feasible.

3.20.1 Environmental Setting

According to the City's General Plan, Newark is considered to be at low risk for wildfire. CAL FIRE has mapped areas at risks of fires around the state. Based on a review of the Fire Hazard Severity Zone maps developed by CAL FIRE, the project site is not within a state responsibility area and does not contain lands classified as very high fire hazard severity zone (CAL FIRE 2007, 2022). The project site is designated as being within a local responsibility area (LRA) and is not classified as being in a very high fire hazard severity zone (CAL FIRE 2008). The U.S. Forest Service (USFS) also has a Wildfire Hazard Potential map that is available online. The USFS Wildfire Hazard Potential map designates a majority of the project site as water/non-burnable and a small section along the east and northeast side of the project site is designated as having a low wildfire hazard potential (USFS 2020). However, the City's General Plan EIR designates the project site as having a high risk from wildfires (City of Newark 2013b). The General Plan EIR's designation is based on the 2007 CAL FIRE "Draft Fire Hazard Severity Zone in the LRA" map which designates the project site as a locally responsible high fire hazard severity zone. Though the General Plan EIR designates the site as having a high risk for wildfires, the updated 2008 CAL FIRE "Very High Fire Hazard Severity Zones in LRA" map does not identify the project site as being within a very high fire hazard severity zone and therefore, there is no significant risk of wildland fires at the project site (CAL FIRE 2008).

3.20.2 Regulatory Setting

State

California Office of Emergency Services

The California Emergency Management Agency was incorporated into the Governor's Office on January 1, 2009 by AB 38, and merged the duties, powers, purposes, and responsibilities of the Governor's Office of Emergency Services (Cal OES) with those of the Governor's Office of Homeland Security. Cal OES is responsible for the coordination of overall state agency response to major disasters in support of local government. The agency is responsible for ensuring the state's readiness to respond to and recover all hazards – natural, man-made, emergencies, and disaster – and for assisting local governments in their emergency preparedness, response, recovery, and hazard mitigation efforts. The Cal OES Fire and Rescue Division coordinates statewide response of fire and rescue mutual aid resources to all types of emergencies, including hazardous materials. The Operations Section under the Fire and Rescue Division coordinates the California Fire and Rescue Mutual Aid System, and coordinated response through the Mutual Aid System, includes responses to major fires, earthquakes, tsunamis, hazardous materials and other disasters.



Draft Environmental Impact Report Environmental Impact Analysis

Uniform Building Code

The Uniform Fire Code contains regulations relating to construction, maintenance, and use of buildings. Topics addressed in the code include fire department access, fire hydrants, automatic storage and use, provisions intended to protect and assist fire responders, industrial processes, and many other general and specialized fire-safety requirements for new and existing buildings and the surrounding premises. The code contains specialized technical regulations related to fire and life safety.

California Health and Safety Code

State fire regulations are set forth in Sections 13000 et seq. of the California HSC and include regulations for building standards (as also set forth in the California Building Standards Code), fire protection and notification systems, fire protection devices such as extinguishers and smoke alarms, high-rise building and childcare facility standards, and fire suppression training.

California Building Code

The state of California provided a minimum standard for building design through the CBC, which is located in Part 2 of Title 24 of the CCR. The 2010 CBC is based on the 1997 Uniform Building Code but has been modified for California conditions. It is generally adopted on a jurisdiction-by-jurisdiction basis, subject to further modification based on local conditions. Commercial and residential buildings are planchecked by local city and county building officials for compliance with the CBC. Typical fire safety requirements of the CBC included; the installation of sprinklers in all high-rise buildings; the establishment of fire resistance standards for fire doors, building materials, and particular types of construction; and the clearance of debris and vegetation within a prescribed distance from occupied structures in wildlife hazard areas.

California Fire Code

CCR, Title 24, also known as the California Building Standards Code, contains the CFC, included as Part 9 of that title. Updated every three years, the CFC includes provisions and standards for emergency planning and preparedness, fire service features, fire protection systems, hazardous materials, fire flow requirements, and fire hydrant locations and distribution. The ACFD provides fire protection services for the City of Newark and as such, implements and enforces the CFC in the Plan Area.

California Department of Forestry and Fire Protection

CAL FIRE protects the people of California from fires, responds to emergencies, and protects and enhances forest, range, and watershed values providing social, economic, and environmental benefits to rural and urban citizens. CAL FIRE's firefighters, fire engines, and aircraft respond to an average of more than 5,600 wildland fires each year.

The Office of the State Fire Marshal supports CAL FIRE's mission by focusing on fire prevention and provides support through a wide variety of fire safety responsibilities: regulating buildings in which people live, congregate, or are confined; controlling substances and products which may, in and of themselves, or by their misuse, cause injuries, death, and destruction by fire; providing statewide direction for fire prevention in wildland areas; regulating hazardous liquid pipelines; reviewing regulations and building standards; and providing training and education in fire protection methods and responsibilities.



Local

City of Newark General Plan

The City of Newark General Plan contains the following goals and policies applicable to the proposed project:

Goal EH-1: Reducing Hazard Exposure. Reduce the potential for injury, harm, property damage, and loss of life resulting from environmental hazards.

- **Policy EH-1.1: Development Regulations and Code Requirements.** Establish and enforce development regulations and building code requirements to protect residents and workers from flooding, liquefaction, earthquakes, fires, and other hazards.
- **Policy EH-1.5: Adequacy of Access.** Require adequate access and clearance for fire equipment, fire suppression personnel, and evacuation of new development.

3.20.3 Environmental Impacts

This section analyzes the project's potential to result in significant wildfire impacts. When an impact is determined to be significant, mitigation measures are identified that would reduce or avoid impacts.

Methodology for Analysis

The following analysis is based on a review of documents pertaining to the project site, including the General Plan, General Plan EIR, and review of CAL FIRE's Fire Hazard Severity Zone Maps and the USFS Wildfire Hazard Potential Map.

Thresholds of Significance

In accordance with the CEQA Guidelines' Appendix G Environmental Checklist, the following questions were analyzed and evaluated to determine whether the proposed project's wildfire impacts are significant. If located in or near state responsibility areas or lands classified as very high fire hazard severity zones, would the proposed project:

- Substantially impair an adopted emergency response plan or emergency evacuation plan?
- Due to slope, prevailing winds, and other factors, exacerbate wildfire risks, and thereby expose project occupants to, pollutant concentrations from a wildfire or the uncontrolled spread of a wildfire?
- Require the installation or maintenance of associated infrastructure (such as roads, fuel breaks, emergency water sources, power lines or other utilities) that may exacerbate fire risk or that may result in temporary or ongoing impacts to the environment?
- Expose people or structures to significant risks, including downslope or downstream flooding or landslides, as a result of runoff, post-fire slope instability, or drainage changes?



Environmental Impact Analysis

Project Impact Analysis and Mitigation Measures

Impact WF-1	The proposed project would not substantially impair an adopted emergency
	response plan or emergency evacuation plan.

Impact Analysis

The project site is not located within a state responsibility area and does not contain lands classified as being within a very high fire hazard severity zone (CAL FIRE 2007, 2022). The proposed project involves permanent modifications to Mowry Avenue and may include partial or complete road closures during the construction phase. There are no identified evacuation routes that would be potentially impacted by the construction of the proposed project. A TCP would be prepared which would identify all detours, appropriate traffic controls, and ensure adequate circulation and emergency access are provided during the construction phase. Additionally, all roadway improvements and construction would be designed to provide adequate emergency vehicle access to the project site during construction and operation of the proposed project. Therefore, project construction and operation activities would not interfere with an emergency evacuation or response plan, and impacts would be less than significant.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.

Exacerbate Wildfire Risk

Impact WF-2 The proposed project would not, due to slope, prevailing winds, and other factors, exacerbate wildfire risks, and thereby expose project occupants to pollutant concentrations from a wildfire or the uncontrolled spread of a wildfire.

Impact Analysis

The topography of the project site is flat and is located in an urban area with existing developments and roadways. The area surrounding the project site is similarly flat. The project site is not in a state responsibility area and does not contain lands classified as being within a very high fire hazard severity zone (CAL FIRE 2007, 2022). Furthermore, the risk of wildfire in this portion of the City is classified as low (USFS 2020). Given the characteristics of the project site, the proposed project would not exacerbate fire risk beyond what currently exists in the vicinity of the project site. Additionally, the proposed project would include fire safety measures and would be designed and constructed in accordance with the City's fire protection requirements and the CFC which would minimize the severity of wildfire impacts on structures and residents if a wildfire was to occur nearby. Development of the proposed project would not expose project occupants to pollutant concentrations from a wildfire or the uncontrolled spread of wildfire, and there would be no impact.

Level of Significance Before Mitigation

No Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Mitigation Measures

No mitigation is necessary.

Level of Significance After Mitigation

No Impact.

Associated Infrastructure

Impact WF-3 The proposed project would not require the installation or maintenance of associated infrastructure (such as roads, fuel breaks, emergency water sources, power lines or other utilities) that may exacerbate fire risk or that may result in temporary or ongoing impacts to the environment.

Impact Analysis

The project site is not located within a state responsibility area and does not contain lands classified as being within a very high fire hazard severity zone (CAL FIRE 2007, 2022). The project site is currently developed as an auto part and scrap metal salvage lot, and the construction of the proposed project would require the installation of associated infrastructure to serve the proposed project. Primary access to the project site would be via Mowry Avenue and on to private streets within the project site. The minimum width available for driving and turning movements through the project site would be 20 feet and the neighborhood streets would be at least 36 feet wide. The project roadway and neighborhood design would provide adequate turning radii and drive areas for fire trucks and other emergency vehicles. All utilities required for the new development would be located underground and the proposed project would include installation of fire hydrants at the project site to mitigate fire hazards. The City's fire flow requirements are based on the standards for fire flow requirements of the CFC and the proposed project's fire hydrants would be constructed to meet these requirements. The NFD would review the project plans to ensure that the proposed project meets the City's fire flow and fire protection requirements. The proposed project would be required to implement General Plan policies along with the implementation of the Uniform Fire Code, CBC and CFC which would reduce effects of development on wildland fire hazard impacts. The proposed project would require the installation of associated infrastructure to support the new development but would not exacerbate fire risk beyond what currently exists in the vicinity of the project site. Compliance with City's policies and applicable building and fire codes would minimize fire risk and there would be a less than significant impact.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation Less Than Significant Impact.



Draft Environmental Impact Report Environmental Impact Analysis

Expose People or Structures

Impact WF-4	The proposed project would not expose people or structures to significant risks,
	including downslope or downstream flooding or landslides, as a result of runoff,
	post-fire slope instability, or drainage changes.

Impact Analysis

The project site is not in a state responsibility area and does not contain lands classified as being within a very high fire hazard severity zone (CAL FIRE 2007, 2022). The project site and surrounding area is relatively flat and not in an area subject to landslides. Additionally, the project site is located in an area identified as having minimal flood hazards. The proposed project would use fill to elevate the proposed pad grades for the homes above the FEMA 100-year flood plain elevation which would offer flood damage protection, including potential flooding from sea level rise. The lowest proposed pad elevation for the proposed project is 13.0 feet NGVD with the average pad elevation for the proposed project at 14.2 feet NGVD. The proposed pad grades would exceed the City's standards of requiring a minimum elevation of 11.25 feet and would meet BCDC's currently adopted sea level rise guidance which recommends a minimum building pad elevation of 12.2 feet. Additionally, the proposed project's pad elevation would meet the minimum pad elevation requirements of the RPC-SAT projected likely range for sea level rise by the year 2100 of 3.4 feet, As such, the proposed project would not expose people or structures to significant risks, including downslope or downstream flooding or landslides, as a result of runoff, post-fire slope instability, or drainage changes and there would be a less than significant impact.

Level of Significance Before Mitigation

Less Than Significant Impact.

Mitigation Measures No mitigation is necessary.

Level of Significance After Mitigation

Less Than Significant Impact.
4.0 CUMULATIVE EFFECTS

4.1 INTRODUCTION

Section 15130(a) of the State CEQA Guidelines requires a discussion of the cumulative impacts of a project when the project's incremental effect is cumulatively considerable. Cumulatively considerable, as defined in CEQA Guidelines Section 15065(a)(3), means that the "incremental effects of an individual project are significant when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects." The State CEQA Guidelines Section 15355 defines a cumulative impact as two or more individual effects that, when considered together, are considerable or that compound or increase other environmental impacts. Cumulative impacts can result from individually minor but collectively significant projects taking place over time.

According to the CEQA Guidelines:

Cumulative impacts refer to two or more individual effects that, when considered together, are considerable and that compound or increase other environmental impacts.

- a) The individual effects may be changes resulting from a single project or multiple separate projects.
- b) "The cumulative impact from several projects is the change in the environment, which results from the incremental impact of the project when added to other closely related past, present, and reasonably foreseeable probably future projects. Cumulative impacts can result from individually minor but collectively significant projects taking place over a period of time." (CCR, Title 14, Division 6, Chapter 3, Section 15355)

In addition, as stated in CEQA Guidelines:

The mere existence of significant cumulative impacts caused by other projects alone shall not constitute substantial evidence that the proposed project's incremental effects are cumulatively considerable (CCR, Title 14, Division 6, Chapter 3, Section 15064[T][5]).

4.2 CUMULATIVE IMPACT SETTING

Cumulative impact discussions for each environmental issue area are provided within each individual impact section. As established in the CEQA Guidelines, related projects consist of "closely related past, present, and reasonably foreseeable probable future projects that would likely result in similar impacts and are located in the same geographic area" (CCR, Title 14, Division 6, Chapter 3, Section 15355).

The State CEQA Guidelines define a cumulative impact as two or more individual impacts that, when considered together, are significant or that compound or increase other significant environmental impacts. Cumulative impacts can result from individually minor but collectively significant projects taking place over time (State CEQA Guidelines Section 15355). The incremental impact of a project, although less than significant on its own, may be considerable when viewed in the cumulative context of other closely related



Draft Environmental Impact Report Cumulative Effects

past, present, and reasonably foreseeable projects. A considerable contribution is considered significant from the point of view of cumulative impact analysis.

CEQA Guidelines Section 15130 identifies two basic methods for establishing the cumulative environment in which a project is considered: the use of a list of past, present, and probable future projects or the use of adopted projections from a general plan, other regional planning document, or a certified EIR for such a planning document. This cumulative analysis uses a combination of the "list" approach and the "projections" approach to identify the cumulative setting. The plan and projections approach relies on an adopted plan or reliable projection that describes the significant cumulative impact. This Draft EIR combines both the project list and projection approaches to generate the most reliable future projections possible.

4.3 GEOGRAPHIC SCOPE

The geographic area analyzed for cumulative impacts is dependent on the resource being analyzed. The geographic area associated with the proposed project's environmental impacts defines the boundaries of the area used for compiling the list of past, present, and reasonably foreseeable projects considered in the cumulative impact analysis.

Each section of this Draft EIR considers the specific geographic area that is directly related to the individual topic addressed within that section. For example, the analysis of air quality is based on a regional level because air quality impacts are regional in nature, whereas analysis of aesthetic impacts only considers related projects in the vicinity of the project site because of the localized nature of the impact.

The geographic area that could be affected by implementation of the proposed project in combination with other projects varies depending on the type of environmental resource being considered. Table 4.3-1 provides the geographic area and the method of evaluation utilized in the cumulative analysis for each resource areas.

Resource Topic	Geographic Area	Method of Evaluation
Aesthetics	Immediate project vicinity	Projects
Agriculture and Forestry Resources	Region	Projects
Air Quality	Local (toxic air contaminants) air basin (construction-related and mobile sources)	Projects and Projections
Biological Resources	Immediate project vicinity and region	Projects
Cultural and Historical Resources	Project site only (does not contribute to cumulative impacts)	Projects
Energy	Immediate project vicinity and region	Projects and Projections
Geology and Soils	Immediate project vicinity	Projects
Greenhouse Gas Emissions and Climate Change	State	Projections

Table 4.3-1: Geographic Scope of Cumulative Impact and Method of Evaluation



Draft Environmental Impact Report Cumulative Effects

Resource Topic	Geographic Area	Method of Evaluation
Hazards and Hazardous Materials	Project site only (does not contribute to cumulative impacts)	Projects
Hydrology and Water Quality	Immediate project vicinity and region	Projects
Land Use and Planning	Immediate project vicinity	Projects
Mineral Resources	Immediate project vicinity and region	Projects
Noise and Vibration	Immediate project vicinity (effects are highly localized)	Projects
Population and Housing	Region	Projects and Projections
Public Services	Immediate project vicinity	Projects and Projections
Recreation	Immediate project vicinity	Projects
Transportation	Immediate project vicinity	Projects and Projections
Tribal Cultural Resources	Project site only (does not contribute to cumulative impacts)	Projects
Utilities and Service Systems	Local	Projects
Wildfire	Immediate project vicinity	Projects

Notes:

Projections = the use of projections contained in relevant planning documents

For those environmental resources that were evaluated based on the projections approach, the projections take into consideration future projects that are not included in the below list of related plans and projects.

4.4 LIST OF RELATED PLANS AND PROJECTS

The list of past, present, and probable future projects used for this cumulative analysis is restricted to those projects that have occurred or are planned to occur (i.e., pending applications at the time of the NOP release) within the vicinity of the project site. For the purposes of this discussion, these projects that may have a cumulative effect on the resources of the project area will be referred to as the "related projects." These related projects are described in Table 4.4-1 and shown in Figure 4-1.

CEQA defines "probable future projects" as those with an active application at the time the NOP was released for a project (in this case, November 19, 2021). The list of projects in Table 4.4-1 was used in the development and analysis of the cumulative settings and impacts for each resource topic. Past and current projects in the project vicinity were also considered as part of the cumulative setting as they contribute to the existing conditions upon which the project and each probable future project's environmental effects are compared.



Projects = the use of a list of past, present, and reasonably foreseeable projects

Draft Environmental Impact Report Cumulative Effects

This page intentionally left blank.



Disclaimer: This document has been prepared based on information provided by others as cited in the Notes section. Stantec has not verified the accuracy and/or completeness of this information and shall not be responsible for any errors or omissions which may be incorporated herein as a result. Stantec assumes no responsibility for data supplied in electronic format, and the recipient accepts full responsibility for verifying the accuracy and completeness of the data.

Draft Environmental Impact Report Cumulative Effects

This page intentionally left blank.

MOWRY VILLAGE PROJECT Draft Environmental Impact Report

Cumulative Effects

Unless otherwise specified, significance criteria are the same for cumulative impacts as they are for project impacts for each environmental topic area. When considered in relation to other reasonably foreseeable projects, cumulative impacts to some resources would be significant and more severe than those caused by the project alone.

No.	Lead Agency	Project Name	Project Address	Distance from Project Site	Project Description	Status
1	City of Newark	Bridgeway/Gateway	GW9W+9 G Newark, CA 37682 Bay Breeze St	2.25 miles	580 single- family units	Under construction
2	City of Newark	Filbert Townhomes	37243 Filbert Street	1.75 miles	Project involves development of 16 new 3-story townhomes and includes group open space, landscaping and street improvements.	Approved
3	City of Newark	Harbor Point	GW9V+G 7 Newark, CA	2.45 miles	192 residential units	Under construction
4	City of Newark	NewPark Phase A	Newpark Mall	1.15 miles	Residential/reta il mixed-use building with a total of 319 dwelling units, approximately 3,700 sf of ground floor retail, approximately 12,900 sf of amenities, a pool courtyard, and enclosed parking including 506 parking stalls on a 3.99 acre site.	Approved

Table 4.4-1: List of Related Projects



Draft Environmental Impact Report Cumulative Effects

No.	Lead Agency	Project Name	Project Address	Distance from Project Site	Project Description	Status
5	City of Newark	Timber Seniors	37660 Timber Street	1.80 miles	Senior affordable housing development with 79 rental units with resident services and amenities on a 0.99 acre site	Approved
6	City of Newark	FMC Willow and Grand Park Project	GWFX+5 9 Newark, CA; GWCX+P 6J Newark, CA	2.25 miles	Project involves redevelopment of a 22.1 acre site with 279- unit multifamily community along with a mixed-use area of 4,000 sf of retail and 90 multifamily affordable units, and a 1,485 square foot community building, along with approximately 1.8 acres set aside for a future transit station	Approved
7	City of Newark	Cedar Homes (Robson Homes)	38370 Cedar Boulevard	1.50 miles	130 residential units consisting of single-family homes and duplexes	Approved
8	City of Newark	Lepakshi at Thornton "Site A" Prelim Review	6781 Thornton Avenue	1.80 miles	59 residential units	Proposed
9	City of Newark	Lepakshi at Thornton "Site B" Prelim Review	6825 Thornton Avenue	1.80 miles	22 residential units	Proposed
10	City of Newark	Cedar Townhomes (Waymark)	38600 Cedar Boulevard	1.40 miles	76 three-story attached townhomes	Proposed



Draft Environmental Impact Report Cumulative Effects

No.	Lead Agency	Project Name	Project Address	Distance from Project Site	Project Description	Status
11	City of Newark	Area 4 – Sanctuary West	Area 4, Stevenso n Boulevard	0.40 mile	469 residential units	Approved
12	City of Newark	NewPark Mall – Costco	NewPark Mall	1.20 miles	161,000 square foot Costco	Under Construction
13	City of Newark	NewPark Plaza	NewPark Mall	1.25 miles	811 residential units, 72,000 square feet commercial	Proposed
14	City of Newark	AC Marriot Hotel	39888 Balentine Drive	1.30 miles	132 hotel rooms	Approved
15	City of Newark	Hyatt Place	5600 John Muir Drive	1.22 miles	113 hotel rooms	Constructed

Source: City of Newark

4.5 CUMULATIVE IMPACT ANALYSIS

For purposes of this EIR, the proposed project would result in a significant cumulative effect if:

- the cumulative effects of related projects (past, current, and probable future projects) are not significant, and the incremental impact of implementing the proposed project is substantial enough when added to the cumulative effects of related projects to result in a new cumulatively significant impact; or
- the cumulative effects of related projects (past, current, and probable future projects) are already significant, and implementation of the proposed project makes a considerable contribution to the effect. The standards used herein to determine a considerable contribution are that either the impact must be substantial or must exceed an established threshold of significance.

This cumulative analysis assumes that all mitigation measures identified in Sections 3.1 through 3.20 to mitigate project impacts are adopted. The analysis herein analyzes whether, after adoption of project-specific mitigation, the residual impacts of the project would cause a cumulatively significant impact or would contribute considerably to existing and anticipated (without the project) cumulatively significant effects. Where the project would so contribute, additional mitigation is recommended where feasible.

4.5.1 Aesthetics

The geographic scope of the cumulative aesthetics analysis is the area surrounding the project site. This is the area within the view of the project; therefore, the area most likely to experience changes in visual



Draft Environmental Impact Report Cumulative Effects

character or experience light and glare impacts. The project site is located in an urbanized area that is developed with agricultural and industrial uses and the majority of the project site is developed with existing uses; however, there are no substantial light and glare sources that exist on the site or along adjacent lands.

The project site would be developed with 203 two-story single-family detached homes. Although the proposed buildings would be taller than the existing structures on-site, they would be within the maximum building heights allowed by zoning regulations for the proposed RS-6000 zoning designation. While the proposed project would change the built environment on-site, the overall view of the project site in the context of views available would not be substantially different due to the amount of separation between the project site and the available views. Additionally, there are no officially designated scenic vistas or view corridors near the project site or within the City.

The proposed project would increase the number and intensity of lighting at the project site with the development of the single-family homes as it would provide exterior lighting throughout the site to illuminate the main entrances of the homes, private streets, sidewalks, common space areas, and driveways for security and safety purposes. Additional lighting would be installed along the project frontage on Mowry Avenue. The proposed project would design and construct lighting in accordance with the City's lighting standards and requirements and would include shielding to ensure light spillage does not occur.

All related projects listed in Table 4.4-1 are located more than one mile away from the project site, except for the Area 4 - Sanctuary West Project which is located approximately 0.4 mile east of the project site. Based on the distance of these related projects (except for the Area 4 - Sanctuary West Project) from the project site, they would not be associated with the visual character of the project area. However, the Area 4 - Sanctuary West Project identified a significant and unavoidable impact related to changes in visual character associated with the development of the project. The significant and unavoidable impact to visual resources for the Area 4 - Sanctuary West Project is due to the project seeking coverage under the existing Areas 3 and 4 Specific Plan which identified a significant and unavoidable aesthetic impact as it would result in development of approximately 850 acres of undeveloped area with urbanized uses and would change the existing character of the area. Most of the project site is developed as an auto part and scrap metal salvage lot, known as "Pick-n-Pull," that includes a 13,000 square foot warehouse, 1,500 square foot sales office, 3,000 square foot workshop, and a large parking area for storing vehicles.. Further, the Area 4 - Sanctuary West Project is already approved and would build 469 two-story singlefamily homes before the proposed project and therefore, the lands surrounding and in the vicinity of the project site would already be developed with urbanized uses and the proposed project would not result in changes to the visual character that would result in a significant cumulatively considerable aesthetics impact.

Similar to the proposed project, the related projects would be required to comply with the development standards of the City and would be subject to review by the City to ensure proposed buildings are compatible with surrounding development and to mitigate potential aesthetic impacts. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on aesthetics.**



Draft Environmental Impact Report Cumulative Effects

4.5.2 Agricultural and Forestry Resources

The proposed project would result in no impacts to agricultural and forestry resources. The project site is not designated for agricultural and forestry uses and no agricultural or forestry uses currently exist onsite. Other developments in the project vicinity would be required to demonstrate that development of the related project would not result in significant impacts to agricultural and forestry resources. Related projects identified in Table 4.4-1 are located in urbanized areas and would not result in the conversion of agricultural or forestry land to urban uses. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on agricultural and forestry resources.

4.5.3 Air Quality

The cumulative setting for air quality is the SFBAAB, which is under the jurisdiction of the BAAQMD. Air pollution is largely a cumulative impact by its very nature. No single project is sufficient in its overall emissions, in isolation, to result in nonattainment of ambient air quality standards. A project's individual emissions contribute to existing cumulatively significant adverse air quality impacts. Related projects in the vicinity of the proposed project and throughout the SFBAAB would also generate emissions that could contribute to air quality impacts. Generally, if a project is proposed in a city or county with a General Plan that is consistent with the Clean Air Plan, and the project is consistent with that General Plan (i.e., does not require a general plan amendment), then the project would not have a significant cumulative impact (provided, of course, that the project does not individually have any significant impacts).

The BAAQMD significance thresholds are intended to analyze whether a project's contribution to the cumulative impact is considerable. If a project exceeds the identified significance thresholds, its emissions would also be considered cumulatively considerable, resulting in a significant adverse air auality impact to the region's existing air guality conditions. As discussed in Section 3.3. Air Quality, the proposed project would be consistent with the Clean Air Plan with the implementation of Mitigation Measure AIR-1. Mitigation Measure AIR-1 requires the implementation of basic constriction mitigation measures recommended by the BAAQMD to reduce fugitive dust emissions. Additionally, as the proposed project's unmitigated remediation and construction emissions would exceed BAAQMD thresholds, the proposed project is required to implement Mitigation Measure AIR-2 which requires project construction to utilize Tier 4 certified construction equipment. Implementation of Mitigation Measure AIR-1 and AIR-2 would reduce project construction emissions and ensure emissions would not exceed BAAQMD thresholds. In addition to Mitigation Measures AIR-1 and AIR-2, the proposed project would be required to implement Mitigation Measure HAZ-1 to ensure that asbestos surveys are conducted for the existing structures on-site prior to demolition to ensure that demolition does not result in NOA pollutant concentrations to sensitive receptors nearby. Project operational emissions would be below the BAAQMD daily and annual significance thresholds.

The closest related project, the Area 4 – Sanctuary West Project, is located approximately 0.4 mile east of the project site and the analysis prepared for the Area 4 – Sanctuary West Project did not identify significant air quality impacts. All other related projects are located more than one mile away from the project site located at the existing Pick-n-Pull site. There are no related projects located near the project site that would result in construction activities during the same construction period as the proposed project. Related projects would be required to implement similar mitigation measures as the proposed



Draft Environmental Impact Report Cumulative Effects

project to reduce construction related emissions and ensure that construction related air quality impacts are less than significant. Additionally, related projects would require project specific air quality analysis be conducted to analyze impacts and determine project specific mitigation measures to reduce potential impacts. Therefore, the proposed project's impacts would not result in a cumulatively considerable impact. Operation of the proposed project would not expose future residents of the project site to substantial pollutant concentration that may cause harmful effects as the risk would be below the threshold of significance. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on air quality.

4.5.4 Biological Resources

The project site, which consists of three parcels totaling 29 acres is located in an area with agricultural and industrial uses in the southwestern portion of the City. Of the 29 acres, 19 acres are currently disturbed and developed with the existing auto parts and scrap metal salvage lot and is primarily covered by existing structure and pavement. The approximately 10 acre northern parcel is undeveloped ruderal/disturbed habitat. Additionally, the proposed project's construction would extend outside of the 29 acre project site to include off-site utility and circulation improvements along Mowry Avenue. Vegetation communities/land covers for the project area consist of developed land, ruderal/disturbed habitat, stormwater detention basins. Project site portions identified as developed land generally lacks significant habitat value for plants and wildlife and wildlife within developed areas is comprised of species that can tolerate regular human disturbance. The ruderal/disturbed area is comprised of vegetation dominated by naturalized or invasive non-native species and ruderal native annuals and provide marginal nesting and foraging habitat for bird species in the region as well as habitat for disturbance-tolerant wildlife. The Stormwater detention basins support patches of rapidly colonizing wetland plants.

The proposed project would implement Mitigation Measures BIO-1 through BIO-9 to reduce potential impacts during construction and operation of the proposed project to wildlife species and City protected trees to a less than significant level and would be consistent with the City's biological policies and tree ordinance. Other related projects in the project vicinity and region would be required to implement similar mitigation to reduce potential impacts to special status species and comply with local biological policies and ordinances. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on biological resources.

4.5.5 Cultural Resources

The geographic scope of the cumulative cultural resources analysis is the project site. The project site does not contain any historical resources and is not located within an identified historic district and therefore, would not result in the destruction of historical resources and there would be no impact. The project site does not contain any recorded archaeological or paleontological resources or burial sites. However, as discussed in Section 3.5, Cultural Resources, there is the possibility that previously undiscovered resources could be encountered by construction activities. Therefore, mitigation measure CUL-1 and CUL-2 were identified to reduce potential impacts to a less than significant level. Implementation of identified mitigation measures would ensure that undiscovered cultural resources are not adversely affected by project-related construction activities, which would prevent the destruction or degradation of potentially significant undiscovered cultural resources in the area and would reduce potential impacts to a less than significant significant level.



MOWRY VILLAGE PROJECT Draft Environmental Impact Report Cumulative Effects

implement similar mitigation measures to protect undiscovered cultural resources during construction. The Area 4 – Sanctuary West Project identified significant and unavoidable impacts to cultural resources; however, the geographic scope of impacts for cultural resources is the project site and does not contribute to cumulative impacts. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on cultural resources.**

4.5.6 Energy

The proposed project would be 100 percent electric and the proposed project's structures would be designed and constructed in accordance with CALGreen and Title 24 standards to increase energy efficiency and reduce energy demand. These standards include minimum energy efficiency requirements related to building envelope, mechanical systems (heating, ventilation, air conditioning, and water heating systems), indoor and outdoor lighting, and illuminated signs. Other planned projects in the vicinity and region would similarly be designed to meet existing CALGreen and Title 24 standards and would require compliance with applicable building code standards related to energy efficiency. This would ensure that the project would not result in the inefficient, unnecessary, or wasteful consumption of energy. **Thus, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on energy.**

4.5.7 Geology and Soils

The proposed project would result in significant impacts related to geology and soils if it would result in substantial adverse effects involving rupture of earthquake faults and seismic ground shaking, seismic related ground failure, expansive or unstable soils, erosion impacts, or destruction of a unique paleontological resource. The proposed project would not result in substantial geology and soils impacts; however, the proposed project would implement mitigation measures to reduce potential impacts from seismic related ground failure, expansive or unstable soils, erosion impacts and impacts to paleontological resources to a less than significant level. The proposed project would not result in any impacts from rupture of an earthquake fault as there are no identified fault lines passing through or near the project site. Mitigation Measures GEO-1 through GEO-4 requiring implementation of design recommendations identified in the preliminary and design level geotechnical reports prepared for the proposed project, as well as requiring preparation of dewatering and shoring plans, and outlining procedures for instances where paleontological resources are discovered during construction, would be implemented to reduce all potential geology and soils impacts to a less than significant level. Additionally, Mitigation Measure HYD-1 would be implemented to ensure the proposed project's construction would not result in substantial soil erosion impacts. Other related projects would be required to implement similar mitigation to reduce impacts related to geology and soils if investigations determine that they were necessary for the related project. Other related projects would be reviewed for impacts related to geology and soils and would be required to address any potential impacts with mitigation. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact related to geology and soils.

4.5.8 Greenhouse Gas Emissions and Climate Change

GHG impacts are a cumulative impact. On their own, GHG emissions from one project cannot result in changes in climate conditions; therefore, the emissions from one project must be considered in the



Draft Environmental Impact Report Cumulative Effects

context of their contribution to cumulative global emissions. As discussed in Section 3.8, Greenhouse Gases, the proposed project would not have a significant impact with regard to GHG emissions and would be consistent with the City's CAP and State plans for achieving GHG reductions and meeting established targets. Additionally, GHG emissions resulting from operation of the proposed project would not exceed the BAAQMD adjusted significance thresholds. The proposed project would be consistent with best practices for reducing GHGs through the incorporation of greater energy efficiency by developing the proposed project's structures to be 100 percent electric. Other projects in the region and the State would also have to show consistency with local and State GHG reduction plans and comply with the Title 24 and CALGreen requirements. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact related to GHG emissions.

4.5.9 Hazards and Hazardous Materials

The proposed project would not result in the use of substantial quantities of hazardous materials or the creation of new hazards. However, there is the potential for hazardous materials, including contaminated soil and water at the site, to be released into the environment resulting from construction activities. Mitigation Measures HAZ-1 and HYD-1 requiring preconstruction surveys to identify existing hazardous conditions on-site, remediation of on-site soils and water identified as being contaminated from previous site uses, and requiring the preparation and implementation of a SWPPP, would be incorporated into the proposed project to reduce potential impacts to a less than significant level. It is reasonable to assume that other related projects would be required to implement similar mitigation to reduce impacts from hazardous materials if investigations determine that they were necessary for the related project. Other related projects would be reviewed for impacts related to hazards and hazardous materials and would be required to address any potential impacts with mitigation. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact related to hazards and hazardous materials.**

4.5.10 Hydrology and Water Quality

The proposed project would not result in substantial degradation to water quality or result in substantial hydrology impacts. However, there is potential for construction related impacts to water quality and increased polluted runoff from the project site. Mitigation Measures HYD-1 and GEO-3 requiring preparation and implementation of a SWPPP and requiring implementation of design recommendations identified in the preliminary and design level geotechnical reports prepared for the proposed project, would be incorporated into the proposed project to reduce potential impacts to a less than significant level. It is reasonable to assume that related projects would be required to implement similar mitigation to reduce impacts related to hydrology and water quality if it were determined that they were necessary for the related project. Related projects would be reviewed for impacts related to hydrology and water quality and would be required to address any potential impacts with mitigation. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact related to hydrology and water quality.

4.5.11 Land Use and Planning

The land use analysis in Section 3.11, Land Use and Planning, found the proposed project to be consistent with applicable goals and policies of the City's General Plan and the General Plan land use



Draft Environmental Impact Report Cumulative Effects

designation of Low Density Residential. Additionally, the proposed project is located within the Areas 3 and 4 Specific Plan area which has been identified by the City as an area for future growth, including residential developments. Though the proposed project is not consistent with the existing zoning designation of the project site and designated use identified in the Areas 3 and 4 Specific Plan, the proposed project is requesting a rezoning and Specific Plan Amendment to better align the project site with the proposed use and existing General Plan land use designation and allow for the development of additional residential units above the maximum allowed number of residential units identified in the Areas 3 and 4 Specific Plan for the area. The proposed project would be consistent with the requirements of the Specific Plan residential single-family use and PD-RS-6000 zoning district with the approval of a Specific Plan Amendment and rezoning. Other development in the project vicinity would be required to demonstrate consistency with the General Plan and development standards through project design or the implementation of mitigation measures. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on land use.**

4.5.12 Mineral Resources

The proposed project would have no impact on mineral resources as the project site and its vicinity does not contain any known mineral resources of value and no mineral extraction activities exist within City limits. Other developments in the City would be required to demonstrate that development of the related project would not result in significant impacts to mineral resources. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on mineral resources**.

4.5.13 Noise and Vibration

The geographic scope of the cumulative noise analysis is the immediate project vicinity. Construction activities associated with the proposed project would be temporary. As discussed under Impact NOI-1 in Section 3.13, Noise, construction noise and vibration from the proposed project would increase noise levels. However, construction noise and vibration would be temporary and limited to the restrictions set by the Areas 3 and 4 Specific Plan RDEIR, as required by Mitigation Measure NOI-3. Operation of the proposed project would also be required to implement Mitigation Measure NOI-1, which includes compliance with City requirements to reduce interior noise levels within the buildings. Additionally, the proposed project would implement Mitigation Measure NOI-2 which requires compliance with requirements in Section 17.24.100, Paragraph A.2.a in the City of Newark Municipal Code and preparation of a noise analysis and incorporation of design measures as needed, such as shielding, barriers, and/or attenuators to reduce noise levels that may affect nearby properties.

Cumulative impacts from construction generated noise could result if other future planned construction activities take place near the project site and/or along the off-site improvements and cumulatively combine with construction noise from the project. The nearest related project to the project site is the Area 4 – Sanctuary West Project which is located approximately 0.4 mile east of the project site located at the existing Pick-n-Pull site. Similar to the proposed project, the Area 4 – Sanctuary West Project would be required to implement mitigation measure and comply with the City's construction noise requirements to limit construction-related noise impacts such as restricting construction hours and implementing measures such as shielding to reduce noise from construction equipment and activities. Additionally, the Area 4 – Sanctuary West Project would conduct a project-specific acoustical analysis and incorporate

Draft Environmental Impact Report Cumulative Effects

operational noise reduction measures to ensure that development does not result in operational noise impacts or contribute to a cumulative noise impact in the vicinity of the project, similar to the proposed project. The Area 4 – Sanctuary West Project's environmental analysis identified that there would be no significant environmental effects related to noise from development of the Area 4 – Sanctuary West Project and that impacts would be less than significant with incorporation of mitigation measures. All other related projects listed in Table 4.4-1 are located more than one mile away from the project site and given the distance between the other related projects and the project site, short-term construction noise and vibration generated by the proposed project would not combine with any other related construction project. Furthermore, the construction and operation of the related projects would be required to comply with the same City noise requirements and implement mitigation to reduce noise level and vibration. **Therefore, the proposed project, in conjunction with related project, would not have a cumulatively considerable impact on noise.**

4.5.14 Population and Housing

The geographic area for cumulative population and housing impacts aligns with the City boundaries and can be extended further to Alameda County and the San Francisco Bay region. The proposed project, in conjunction with the future development in the City, is within the planned growth projections provided by the City's General Plan. Additionally, the number of residential units proposed under the proposed project would be within the City's projected growth under the General Plan and would contribute to providing a housing quantity closer to the intended housing unit allocation envisioned within the General Plan. The proposed project would not have a significant impact on housing and job balance and would help the City meet its RHNA. Other related projects would be required to demonstrate consistency with the growth projections identified in the City's General Plan. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on population and housing.

4.5.15 Public Services

Police

The proposed project would increase service demands for police protection. Similar to the proposed project, related projects in the City would be required to pay a Development Impact Fee as required by Chapter 3.24 of the Newark Municipal Code to offset the cost of police protection demands associated with development of a new project. According to the NPD website, the NPD serves approximately 49,149 residents and the department includes a staff of 59 sworn officers which results in an approximate ratio of 1.2 officers per thousand residents (City of Newark 2021). The NPD does not identify a specific officer per resident ratio that is used as a standard to determine appropriate staffing levels. Though implementation of the proposed project would result in increased demand on police services, it is not anticipated to exacerbate any staffing issues by increasing the service population. Additionally, the General Plan EIR identified that continued implementation of required fees and General Plan policies related to the police department, including requiring police department review of major new development plans, would ensure that the NPD is actively involved in planning to accommodate future growth in the City and that funds are set aside to allow for construction of new facilities as needed to accommodate future growth (City of Newark 2013b). The proposed project is not anticipated to have a cumulative impact on police protection such that it would necessitate the construction of new or expanded police facilities the construction of which would have adverse environmental impacts. The General Plan EIR identified that though growth



Draft Environmental Impact Report Cumulative Effects

anticipated in the City would require a new building or further expansion of existing facilities in the future, associated impacts would be reduced to a less than significant level through the implementation of General Plan policies and collection of required fees from development (City of Newark 2013b). Other related projects would be reviewed for impacts on police protection and would be required to address any potential impacts with mitigation. Similar to the proposed project, related projects would be required to comply with General Plan policies and would pay required fees to offset the cost of increased demand from development of the related project. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on law enforcement.

Fire and Emergency

The proposed project would increase service demands for fire and emergency services. Similar to the proposed project, related projects in the City would be required to pay a Development Impact Fee as required by Chapter 3.24 of the Newark Municipal Code to offset the cost of fire protection demands associated with development of a new project. Additionally, the proposed project and related projects would be required to be designed and constructed in accordance with the City's fire protection requirements and the CBC. The proposed project is not anticipated to have a cumulative impact on fire and emergency services such that it would necessitate the construction of new or expanded fire and/or emergency facilities, the construction of which could have adverse environmental impacts. Other related projects would be required to address any potential impacts with mitigation. Therefore, the proposed project, in conjunction with other related projects, would not have a cumulatively considerable impact on fire and emergency services.

Schools

The proposed project would result in an increase in demand for school facilities and services. The proposed project would be required to pay school impact fees collected by NUSD which would offset school facility costs associated with serving new students that result from new residential developments. Under SB 50, payment of school impact fees is considered to fully mitigate impacts to schools from new development for the purposes of CEQA. Other related projects that would result in the development of new residential units would also be required to pay fees to offset their impacts. The proposed project is not anticipated to have a cumulative impact on schools such that it would necessitate the construction of new or expanded school facilities, the construction of which could have adverse environmental impacts. Other related projects would be required to pay schools and would be required to address any potential impacts with mitigation. Similar to the proposed project, related projects would be required to pay school impact fees is considered to mitigate impacts to schools from new development for the purposes of CEQA. As such, related projects are not anticipated to result in significant impacts to schools. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on schools.**

Parks

The proposed project would result in an increase in the demand for nearby parks; however, the proposed project would provide 4.89 acres of on-site open space, with the majority of on-site open space being



Draft Environmental Impact Report Cumulative Effects

provided through private open space, provided as a rear yard for each home. The proposed project would provide 0.94 acres of common open space through the construction of an on-site recreational area, landscaping, and bioretention areas. In addition, the proposed project would pay a parks impact fee. The proposed project is not anticipated to have a cumulative impact on parks such that it would necessitate the construction of new or expanded park facilities the construction of which would have adverse environmental impacts. Other related projects would be reviewed for impacts on parks and would be required to address any potential impacts with mitigation. Other related projects would be required to either dedicate land, or pay a fee in-lieu thereof, or both, at the option of the City as required for all new development within the City. The dedication of land and payment of fees would offset potential impacts from increased demand for park facilities. **Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on parks.**

4.5.16 Recreation

As discussed above under Section 4.5.15, the proposed project would result in an increase in demand for parks and recreational facilities but would have a less than significant impact on parks as the proposed project would develop on-site open space, including a new recreational area, and would pay the required parks impact fees. Related projects in the area would have the potential to result in a cumulative impact associated with increased demand for parks, resulting in deterioration of existing parks or requiring construction or expansion of recreational facilities. Related projects would be required to either dedicate land, or pay a fee in-lieu thereof, or both, at the option of the City as required for all new development within the City. The Area 4 – Sanctuary West Project would provide three parks and four boardwalk overlooks that would result in a combined park area of approximately 4.70 acres that would be available for public use, including for use by the proposed project's residents. Related projects would be reviewed for impacts on parks and would be required to address any potential impacts with mitigation. **As such, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on recreation.**

4.5.17 Transportation

According to OPR's Technical Advisory for *Evaluating Transportation Impacts in CEQA*, a project's cumulative impacts are based on an assessment of whether the "incremental effects of an individual project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects. The related projects listed in Table 4.4-1 would not have a bearing on the VMT analysis since the cumulative condition VMT data would come from the regional traffic model's 2040 horizon. A project that falls below an efficiency-based threshold that is aligned with long-term environmental goals and relevant plants would have no cumulative impact distinct from the project impact.

As discussed in Section 3.17, Transportation, the proposed project would not conflict with the City's General Plan Circulation Element, any program, plan, ordinance or policy addressing the circulation system. However, the proposed project would exceed the Countywide VMT threshold and result in a significant and unavoidable impact. Mitigation Measure TRANS-1, which requires the preparation and implementation of a TDM Plan, was identified to reduce VMT impacts. However, even with the implementation of Mitigation Measure TRANS-1, implementation of the TDM Plan would reduce project-generated VMT by less than one percent which would not be adequate to reduce the project-generated



VMT to below the threshold of significance. Therefore, the proposed project would have a cumulatively considerable impact with respect to VMT and result in a significant and unavoidable cumulative impact.

4.5.18 Tribal Cultural Resources

The geographic scope of the cumulative tribal cultural resources analysis is the project site. According to CEQA, the importance of tribal cultural resources is the value of the resource to California Native American tribes culturally affiliated with the project area. Therefore, the issue that must be explored in a cumulative analysis is the loss of tribal cultural resources. For tribal cultural resources that are avoided or preserved through dedication within open space, no impacts would occur. However, if avoidance or dedication of open space to preserve tribal cultural resources is infeasible, those impacts must be considered in combination with tribal cultural resources that would be impacted for other projects included in the cumulative project list.

Related projects located in the region would have the potential to result in a cumulative impact associated with the loss of tribal resources through development activities that could cause a substantial adverse change in the significance of a tribal cultural resource. Any related projects that involve ground disturbing activities would have the potential to result in significant impacts to tribal cultural resources. All projects would be regulated by applicable federal, state, and local regulations to avoid the destruction of tribal cultural resources. As discussed in Section 3.18, Tribal Cultural Resources, no tribal cultural resources have been identified within the project site during the cultural evaluation. In the event that tribal cultural resources are inadvertently discovered during construction, mitigation measures identified in Section 3.5, Cultural Resources, would be implemented to reduce potentially significant impacts to a less than significant level. As such, impacts to tribal cultural resources would be unlikely to occur with implementation of the project. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on tribal cultural resources.

4.5.19 Utilities and Service Systems

The proposed project would generate an increased demand for water and wastewater treatment, stormwater drainage, and solid waste services, and would increase demand for water supplies. However, the proposed project's increased demand would be adequately handled by existing systems and would not result in significant impacts to utilities and service systems. Additionally, a WSA prepared for the Areas 3 and 4 Specific Plan found that there were sufficient water supplies available to serve the developments proposed under the Specific Plan. Similar to the proposed project, other related projects would be required to demonstrate that existing systems have the capacity to handle the increase in water, wastewater, stormwater and solid waste generated from the project and that there would be sufficient water supplies to serve the increased demand during normal, dry, and multiple dry year scenarios. Other related projects would be reviewed for impacts on utilities and service systems and would be required to address any potential impacts with mitigation. Therefore, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on utilities and service systems.



Draft Environmental Impact Report Cumulative Effects

4.5.20 Wildfire

The proposed project would have no significant impacts related to wildfire as the project site is not located in an area identified as a very high fire hazard severity zone. Similar to the proposed project, all other related projects would be required to implement General Plan policies identified to reduce wildfire risk along with the Uniform Building Code, CBC, and the CFC which would reduce effects of development on wildfire hazard impacts. Other related projects would be reviewed for impacts on wildfire risk and would be required to address any potential impacts with mitigation. **As such, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on wildfires.**

5.0 ALTERNATIVES TO THE PROPOSED PROJECT

5.1 INTRODUCTION

The purpose of an alternatives analysis pursuant to CEQA is to identify feasible options that would attain most of the basic objectives of a proposed project while reducing its significant effects. Provisions of CEQA Guidelines (Section 15126.6) that address the number of project alternatives required in an EIR state the following:

The range of alternatives required in an EIR is governed by a "rule of reason;" the EIR must evaluate only those alternatives necessary to permit a reasonable choice. The alternatives shall be limited to those that would avoid or substantially lessen any of the significant effects of a proposed project while meeting most of the underlying project objectives.

5.2 **REQUIREMENTS FOR THE CONSIDERATION OF ALTERNATIVES**

An important aspect of EIR preparation is the identification and assessment of alternatives to the proposed project that have the potential to avoid or substantially lessen potentially significant impacts. In addition to mandating consideration of the no project alternative, CEQA Guidelines (Section 15126.6(e)) emphasize the selection of a reasonable range of feasible alternatives and adequate assessment, which allows decision-makers to use a comparative analysis. CEQA Guidelines (Section 15126.6(a)) states:

An EIR shall describe a reasonable range of alternatives to the project, or to the location of the project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project and evaluate the comparative merits of the alternatives. An EIR need not consider every conceivable alternative to a project. Rather it must consider a reasonable range of potentially feasible alternatives that will foster informed decision making and public participation.

In accordance with CEQA Guidelines 15126.6, this EIR contains a comparative impact assessment of alternatives to the proposed project. The primary purpose of this assessment is to provide decision-makers and the public with a reasonable number of feasible project alternatives that could attain most of the basic project objectives while avoiding or reducing any of the project's significant adverse environmental effects. Important considerations for these alternatives' analyses are provided below:

- An EIR need not consider every conceivable alternative to a project;
- An EIR should identify alternatives that were considered by the lead agency, but rejected as infeasible during the scoping process;
- Reasons for rejecting an alternative include:
 - Failure to meet most of the basic project objectives
 - o Infeasibility



Draft Environmental Impact Report Alternatives to the Proposed Project

o Inability to avoid significant environmental effects

5.2.1 No Project Alternative

CEQA Guidelines require that the alternatives be compared to the project's environmental impacts and that the "no project" alternative be considered (CEQA Guidelines Section 15126.6(d)(e)). Section 15126.6(d)(e)(1) states:

The specific alternative of "no project" shall also be evaluated along with its impact. The purpose of describing and analyzing a no project alternative is to allow decision makers to compare the impacts of approving the proposed project with the impacts of not approving the proposed project. The no project alternative analysis is not the baseline for determining whether the proposed project's environmental impacts may be significant, unless it is identical to the existing environmental setting analysis which does establish that baseline.

The purpose of describing and analyzing a no project alternative is to allow decision-makers to compare the impacts of approving the proposed project with the impacts of not approving the proposed project.

5.2.2 Consistency with Project Objectives

A project's statement of objectives describes the purpose of the project and the reasons for undertaking the project. To be considered for detailed analysis in the EIR, an alternative must meet most of the project objectives. Among the suite of project objectives identified by the applicants, the City as lead agency has identified the basic objectives for purposes of screening potential alternatives to the proposed project. The primary objective of the proposed project is to provide low density residential housing that incorporates multi-modal transportation for the future residents of Newark. Specific project objectives include the following:

- Implement the City's General Plan by developing the site with low-density residential;
- Support the City in meeting its RHNA target assigned by the ABAG;
- Provide high quality residential development including a mix of lot sizes;
- Minimize environmental impacts associated with residential development by siting the project on developed and disturbed lands;
- · Remediate contaminated soil on-site to levels suitable for residential development;
- Create a residential development that integrates multi-modal transportation (pedestrian, bicycle, automobile) and connects the development to existing, nearby bus transit stops and active centers by improving Mowry Avenue and upgrading the at-grade vehicular and pedestrian crossing along Mowry Avenue at the UPRR railroad tracks to increase safety.



5.2.3 Feasibility

According to CEQA Guidelines (Section 15126.6(f)(1):

Among the factors that may be taken into account when addressing the feasibility of alternatives are site suitability, economic viability, availability of infrastructure, general plan consistency, other plans or regulatory limitations, jurisdictional boundaries (projects with a regionally significant impact should consider the regional context), and whether the proponent can reasonably acquire, control, or otherwise have access to the alternative site (or the site is already owned by the proponent). No one of these factors establishes a fixed limit on the scope of reasonable alternatives.

Based on CEQA Guidelines, "feasible" is defined as, "capable of being accomplished in a successful manner within a reasonable period of time, taking into account economic, environmental, legal, social, and technological factors" (CEQA Guidelines Section 15364). CEQA does not require that an EIR determine the ultimate feasibility of a selected alternative, but rather that an alternative be potentially feasible.

For the screening analysis, the potential feasibility of potential alternatives was assessed using the following considerations:

Technological Feasibility: Is the alternative feasible from a technical perspective, considering available technology? Are there any construction, operation, or maintenance constraints that cannot be overcome?

Legal Feasibility: For example, do legal protections on lands or financing strategies preclude or substantially limit the feasibility of constructing the alternative?

Economic Feasibility: Is the alternative so costly that its costs would prohibit its implementation?

In determining what alternatives should be considered in the EIR, it is important to acknowledge the objectives of the project, the project's significant effects, and unique project considerations. These factors are crucial to the development of alternatives that meet the criteria specified in Section 15126.6(a). Although, as noted above, an EIR must contain a discussion of "potentially feasible" alternatives, the ultimate determination whether an alternative is feasible or infeasible is made by the lead agency's decision-making body (See PRC Section 21081[a][3]).

5.2.4 Potential to Avoid or Lessen Significant Environmental Effects

CEQA requires that alternatives to a proposed project have the potential to avoid or substantially lessen one or more significant effects of the project (CEQA Guidelines Section 15126.6). At the project and/or cumulative level, the Draft EIR has identified the following environmental issues that may result in significant impacts. This list only includes those impacts that were determined to be significant and unavoidable.

Transportation

• The proposed project would conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b).



Draft Environmental Impact Report Alternatives to the Proposed Project

5.3 METHODOLOGY AND SCREENING CRITERIA

A range of potential alternatives was developed and subjected to the screening criteria. Several representative alternatives were considered. There was no attempt to include every conceivable alternative. The following criteria were used to screen potential alternatives:

- Does the alternative meet most of the project objectives?
- Is the alternative potentially feasible?
- Would the alternative substantially reduce one or more of the significant impacts associated with the project?

5.4 ALTERNATIVES CONSIDERED AND REJECTED FROM FURTHER CONSIDERATION

As described above, State CEQA Guidelines Section 15126.6(c) provides that the range of potential alternatives for the project shall include those that could feasibly accomplish most of the basic objectives of the project and could avoid or substantially lessen one or more of the significant effects. Alternatives that fail to meet the fundamental project purpose need not be addressed in detail in an EIR. (In re Bay-Delta Programmatic Environmental Impact Report Coordinated Proceedings (2008) 43 Cal.4th 1143, 1165-1167.)

In determining what alternatives should be considered in the EIR, it is important to acknowledge the objectives of the project, the project's significant effects, and unique project considerations. These factors are crucial to the development of alternatives that meet the criteria specified in Section 15126.6(a). Although, as noted above, EIRs must contain a discussion of "potentially feasible" alternatives, the ultimate determination as to whether an alternative is feasible or infeasible is made by lead agency decision-makers. (See PRC, § 21081(a)(3).) At the time of action on the project, the decision-makers may consider evidence beyond that found in this EIR in addressing such determinations. The decision-makers, for example, may conclude that a particular alternative is infeasible (i.e., undesirable) from a policy standpoint, and may reject an alternative on that basis provided that the decision-makers adopt a finding, supported by substantial evidence, to that effect, and provided that such a finding reflects a reasonable balancing of the relevant economic, environmental, social, and other considerations supported by substantial evidence. (City of Del Mar v. City of San Diego [1982] 133 Cal.App.3d 401, 417; California Native Plant Society v. City of Santa Cruz [2009] 177 Cal.App.4th 957, 998.)

The EIR should also identify any alternatives that were considered by the lead agency but were rejected during the planning or scoping process and briefly explain the reasons underlying the lead agency's determination. The following alternatives were considered by the City but are not evaluated further in this Draft EIR for the reasons discussed below.

5.4.1 Alternative Location Alternative

In accordance with CEQA Guidelines Section 15126.6(f)(2), an alternative project site location should be considered if development of another site is feasible, and if development of another site would avoid or



Draft Environmental Impact Report Alternatives to the Proposed Project

substantially lessen significant impacts of the proposed project. Such alternatives are especially appropriate where a project would put a site to uses different than those contemplated in the governing general plan or zoning district, which presumably reflect land use policies reached after much deliberation and public involvement, and also in instances where there is an ample supply of similarly situated land that could be developed for a project. Factors that may be considered when identifying an alternative site location include the size of the site, its location, the General Plan or Community Plan land use designations, and availability of infrastructure.

The Applicant and the City considered the residential development of alternative site locations within the City limits that are located closer to transit hubs to reduce project generated VMT. Undeveloped lots within City limits were assessed by size and proximity to a transit hub. There are no undeveloped sites within City limits that are large enough to accommodate a project of similar size and scale to the proposed project and in proximity to a transit hub. Therefore, the Alternative Location Alternative was determined to be infeasible and was rejected from further consideration.

5.4.2 Restoration Alternative

A joint letter was received from Citizens Committee to Complete the Refuge, Greenbelt Alliance, Sierra Club, Center for Biological Diversity, San Francisco Baykeeper, Alameda Creek Alliance, Santa Clara Valley Audubon Society, Ohlone Audubon Society, and Mission Peak Conservancy on December 13, 2021, during the NOP comment period. The letter proposed a Restoration Alternative to remediate and reposition the project site for development as an alternate use, such as a park, open space, or recreational use.

Under this alternative, the project site would be remediated and repositioned for development of an alternative use, such as a park, open space, or recreational use. Under the Restoration Alternative, the project site would not be developed with 203 single-family homes and would instead be developed as a site providing a park, open space, or recreational uses. While this alternative would reduce potentially significant VMT impacts, it would not meet any of the proposed project's objectives. Additionally, currently, no known funding sources are available to provide financial assistance to acquire the property, demolish the existing uses, remediate groundwater and soils on-site, and restore the site for park, open space, or recreational uses. Therefore, the Restoration Alternative was determined to be infeasible and was rejected from further consideration.

5.4.3 Small Project Alternative

Under this alternative, the proposed project would be reduced to the maximum number of residential units that would result in a less than significant impact on transportation, specifically on VMT. OPR's screening threshold for Small Projects states that projects generating less than 110 trips per day may be assumed to have a less than significant impact on transportation. For the proposed project to generate less than 110 trips per day, the proposed project would be constructed with a maximum of seven detached single-family units or 16 low-rise multi-family residential units. The development of the project site with seven detached single-family units or 16 low-rise multi-family units would result in a generation of less than 110 trips per day and would meet the screening threshold for small projects. While this alternative would reduce potentially significant VMT impacts, it would not meet the proposed project's objectives to provide a mix of lot sizes, support the City in meeting its RHNA target, remediating contaminated soils on-site, or



creating a residential development that integrates multi-modal transportation. Additionally, the development of seven single-family units or 16 multi-family units on an approximately 29 acre project site would result in a density of less than one unit per acre and would underutilize the land. Therefore, the Small Project Alternative was determined to be infeasible and was rejected from further consideration.

5.5 ALTERNATIVES CONSIDERED

Section 15126 of CEQA Guidelines requires an EIR to identify and discuss a no project alternative, as well as a reasonable range of alternatives to the proposed project that would feasibly attain most of the basic objectives of the proposed project and would avoid or substantially lessen any of the significant environmental impacts.

Alternatives to the proposed project considered for analysis in this EIR are:

- No Project Alternative
- Multi-family Residential Alternative
- Reduced Density Alternative
- 100 Percent Affordable Housing Alternative

5.5.1 No Project Alternative

CEQA Guidelines Section 15126.6(e)(1) requires that the No Project Alternative be described and analyzed, "to allow decision-makers to compare the impacts of approving the project with the impacts of not approving the project." The no project analysis is required to discuss, "the existing conditions at the time the Notice of Preparation is published . . . as well as what would be reasonably expected to occur in the foreseeable future if the project were not approved, based on current plans and consistent with available infrastructure and community services" (Section 15126.6(e)(2)).

The No Project Alternative assumes that no additional development would occur on the project site. Under this alternative, the project site would remain developed with the existing structures for commercial uses. The existing structures and surface lot would continue to be used as an auto part and scrap metal salvage lot. The northern parcel of the project site would continue to be undeveloped land.

While the no project alternative would avoid the significant and unavoidable impact to VMT, it would not meet any of the proposed project's objectives.

Impact Analysis

Aesthetics

Under the No Project Alternative, the existing structures would remain and there would be no change to the project site's visual character. Therefore, the No Project Alternative would have no impact related to aesthetics.



Draft Environmental Impact Report Alternatives to the Proposed Project

Agricultural and Forestry Resources

Under the No Project Alternative, the project site would continue to be used for existing purposes and would not result in changes to the land use designations or zoning. Therefore, the No Project Alternative would have no impact related to agricultural and forestry resources.

Air Quality

Under the No Project Alternative, the existing uses would continue to operate on the project site, and there would be no change in air emissions. As discussed in Section 3.3, Air Quality, remediation and construction of project site required for the development of the proposed project would result in construction generated pollutant emissions that would be potentially significant. However, with the implementation of mitigation measures, no significant air quality impacts would occur during construction of the proposed project. The No Project Alternative would not involve demolition, remediation, or construction activities. Therefore, no impacts related to air quality due to construction or increased operational emissions would occur. This would be similar to the proposed project, which would not result in any significant air quality impacts with mitigation incorporated; however, air quality impacts would be lessened due the elimination of construction emissions and fewer operational emissions when compared to the proposed project.

Biological Resources

Under the No Project Alternative, the proposed project would not be constructed and operated, and the existing uses would continue to operate on the project site. Existing trees on the project site would not be removed and there would be no impacts to biological resources. Therefore, the No Project Alternative would lessen impacts to biological resources compared to the proposed project.

Cultural Resources

Under the No Project Alternative, the project would continue to have no impact to historical resources as there are no identified historical resources on-site. However, under the No Project Alternative, the project would not result in construction activities and there would be no subsurface ground disturbance activities that could impact undiscovered cultural resources. Therefore, the No Project Alternative would have no impact on cultural resources.

Energy

Under the No Project Alternative, the existing uses would continue to operate on the project site, and there would be no demolition and construction of more energy efficient buildings. As discussed in Section 3.6, Energy, though the proposed project would result in increased electricity demand from existing conditions, the electricity would be consumed more efficiently and compliance with future building code standards would result in increased energy efficiency. The No Project Alternative would not help build energy efficient buildings to serve growth and development and therefore, energy impacts may be greater than the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Geology and Soils

Under the No Project Alternative, the existing commercial uses would continue to operate on the project site and new structures would not be constructed. The proposed project includes the implementation of geotechnical design recommendations and includes mitigation for unstable soil and liquefaction induced impacts. These recommendations and mitigations may not occur under the No Project Alternative. Therefore, impacts related to geology and soils under the No Project Alternative may be greater compared to the proposed project.

Greenhouse Gas Emissions

Under the No Project Alternative, the existing uses would continue to operate on the project site, and there would be no change in GHG emissions. However, the No Project Alternative would not develop new housing near transit, would not construct more energy efficient structures, or construct pedestrian and bicycle improvements and therefore, would not help to reduce future GHG emissions. Accordingly, long-term impacts to GHG emissions may be greater than the proposed project.

Hazards and Hazardous Materials

If the No Project Alternative is implemented, the proposed project would not be constructed or operated, and the existing uses would continue on-site. The proposed project includes the implementation of mitigation and remediation for existing on-site contamination and other hazardous materials at the site, which may not occur under the No Project Alternative. Therefore, impacts related to hazards or hazardous materials under the No Project Alternative may be greater compared to the proposed project.

Hydrology and Water Quality

Under the No Project Alternative, the existing commercial uses would continue to occur on-site and would not result in changes to the existing drainage pattern, groundwater, or water quality at and near the project site. Under the No Project Alternative, the proposed improvements to off-site and on-site stormwater systems and the construction of bioretention treatment areas would not occur. Additionally, the existing structures would continue to be located below the FEMA 100 year flood plain elevation and would not construct flood damage protection. Therefore, hydrology and water quality impacts may be greater under the No Project Alternative.

Land Use and Planning

Under the No Project Alternative, the existing uses would continue to occur on-site; however, increased density and housing as envisioned by the City's General Plan and Specific Plan would not occur. The No Project Alternative would not meet the City and the proposed project's objective and goal to provide high quality housing within the City and to support the City meet its RHNA target. Land use and planning impacts may be greater under the No Project Alternative.

Mineral Resources

Under the No Project Alternative, the existing commercial uses would continue to operate on the project site and would not result in any ground disturbance activities. Section 3.12 of this EIR identified that the



Draft Environmental Impact Report Alternatives to the Proposed Project

project site is not a delineated mineral resource recovery site or is in area containing mineral resources. Therefore, there would be no impacts to mineral resources under the No Project Alternative.

Noise

Under the No Project Alternative, the existing uses would continue to operate on the project site, and there would be no change to the existing noise or vibration levels from construction or operation. While the proposed project would have some increase in noise and vibration during construction, the increases would be temporary and addressed through implementation of mitigation measures. Operationally, the proposed project would shield noise through building construction and mitigation by incorporating recommendations included in the required noise study. The inclusion of recommendations and design features included in the noise study which may include shielding, barriers, and attenuators to reduce noise levels would not occur under the No Project Alternative and noise impacts may be greater compared to the proposed project.

Population and Housing

Under the No Project Alternative, the project site would continue to be used for commercial purposes and would not construct residential developments that would result in an increase in the population. The No Project Alternative would not construct new infrastructure at or near the project site that could lead to indirect population growth. Therefore, there would be no impacts to population and housing under the No Project Alternative.

Public Services

The No Project Alternative would continue existing operations on the site. As discussed in Section 3.15, Public Service, demands for schools, parks, and service of police, fire and emergency services would increase with the development of the proposed project but would not increase demand to a substantial level. The No Project Alternative would not construct the proposed 203 single-family homes and would not result in increased calls for service. Therefore, there would be no impacts to public services under the No Project Alternative.

Recreation

The No Project Alternative would continue existing operation at the site and would not construct the 203 single-family homes and associated infrastructure and improvements. The No Project Alternative would not construct any new open space or parks. The proposed project included the development of new open spaces and park on-site which would have contributed to the City's parks and recreational opportunities; however, the No Project Alternative would not result in an increased demand for parks and recreational opportunities and would not result in the need for development of additional parks and recreational opportunities. Therefore, there would be no impacts to recreation under the No Project Alternative.

Transportation

Under the No Project Alternative, the existing uses would continue to operate on the project site. Under the No Project Alternative, traffic conditions at the site and in the vicinity of the project site would not change and would not generate additional traffic. Therefore, impacts to transportation would be less under the No Project Alternative.



Draft Environmental Impact Report Alternatives to the Proposed Project

Tribal Cultural Resources

Under the No Project Alternative, there would be no subsurface ground disturbance activities that could impact undiscovered tribal cultural resources. The No Project Alternative would not have an impact on tribal cultural resources.

Utilities and Service Systems

Under the No Project Alternative, the existing uses would continue to operate on the project site and would not construct the proposed residential development and associated infrastructures. Under the No Project Alternative, the demand for water, wastewater, stormwater, solid waste, and other utility services would remain the same as existing conditions and would not result in increased demand. Therefore, the No Project Alternative would not have any impacts on utilities and service systems.

Wildfire

If the No Project Alternative is adopted, the existing commercial uses at the site would continue to operate and the project site would remain the same as existing conditions. The adoption of the No Project Alternative would not exacerbate the risk of wildfires beyond what currently existing in the vicinity of the project site. Therefore, there would be no impacts to wildfire under the No Project Alternative.

Conclusion and Relationship to Project Objectives

The No Project Alternative would reduce the significant and unavoidable impact to VMT, but it would have greater impacts to energy, geology and soils, GHGs, hazards and hazardous materials, hydrology and water quality, land use and planning, and noise. The No Project Alternative would not achieve any of the project objectives shown below:

- Implement the City's General Plan by developing the site with low-density residential;
- Support the City in meeting its Regional Housing Needs Allocation (RHNA) target assigned by the Association of Bay Area Governments (ABAG);
- Provide high quality residential development including a mix of lot sizes;
- Minimize environmental impacts associated with residential development by siting the project on developed and disturbed lands;
- Remediate contaminated soil on-site to levels suitable for residential development;
- Create a residential development that integrates multi-modal transportation (pedestrian, bicycle, automobile) and connects the development to existing, nearby bus transit stops and active centers by improving Mowry Avenue and upgrading the at-grade vehicular and pedestrian crossing along Mowry Avenue at the UPRR railroad tracks to increase safety.

5.5.2 Alternative 2 – Multi-family Residential Alternative

Under the Multi-family Residential Alternative, the project would include the demolition of existing uses and remediation of existing contaminants on-site, same as the proposed project. However, under this



Draft Environmental Impact Report Alternatives to the Proposed Project

alternative, the project would develop 405 multi-family residential units. Under the Multi-family Residential Alternative, the project would construct 405 multi-family residential units on 29 acres, resulting in a density of approximately 14 units per acre and therefore, would require a General Plan Amendment from Low Density Residential to Medium Density Residential to allow for the increased density. This alternative would provide multi-family residential units that range from garden apartments and condominiums to townhomes and row houses that would have a maximum building height of 60 feet, in accordance with the development standards of the Areas 3 and 4 Specific Plan.

Impact Analysis

Aesthetics

Under the Multi-family Residential Alternative, the heights of the proposed buildings would be increased and the site would be developed with a higher density. The analysis in Section 3.1, Aesthetics, identified that the proposed project would not result in significant impacts to scenic resources given the amount of separation between the project site and scenic views. Though this alternative would increase building heights and change the character of the site, the distance between the project and scenic views would remain the same and impacts to aesthetics would be equivalent to the proposed project.

Agricultural and Forestry Resources

Under the Multi-family Residential Alternative, the project site would be developed with residential uses, similar to the proposed project. The alternative would be located within the same project site footprint as the proposed project and would not develop any agricultural and forestry uses. Therefore, the Multi-family Residential Alternative would have no impact related to agricultural and forestry resources and impacts would be equivalent to the proposed project.

Air Quality

Under the Multi-family Residential Alternative, construction of the project may require a longer construction period due to the increased number of residential units and may result in greater construction related air quality impacts. However, construction related emissions would be mitigated with the implementation of the same mitigation measures required for construction of the proposed project. Additionally, operational emissions under the Multi-family Residential Alternative would be greater when compared to the proposed project resulting from the increased number of residential units and the generation of more residents at the site. The net operational emissions under the proposed project were determined to be well below the BAAQMD project level threshold and it is not anticipated that implementation of the Multi-family Residential Alternative would not result in operational emissions that exceed the identified threshold. Though the Multi-family Residential Alternative would be increased from the proposed project and impact to air quality would be greater when compared to the proposed project and impact to air quality would be greater when compared to the proposed project.

Biological Resources

Under the Multi-family Residential Alternative, the entirety of the project site would continue to be developed with residential uses and the project would continue to require removal of existing trees on-



Draft Environmental Impact Report Alternatives to the Proposed Project

site. Additionally, development of the Multi-family Residential Alternative would result in the same biological resources impacts as the proposed project as it would be located within the same project footprint. Therefore, this alternative would require implementation of the same mitigation measures identified for the proposed project and impacts resulting from the Multi-family Residential Alternative would be equivalent to the proposed project.

Cultural Resources

The Multi-family Residential Alternative would continue involve subsurface ground disturbance activities which could result in the inadvertent discovery of unknown archeological resources at the project site. The amount of ground disturbance required would be similar to the proposed project as this alternative would include development of the same project site footprint as the proposed project. Potential impacts would be reduced to a less than significant level with the incorporation of the same mitigation measures identified for the proposed project to address the unanticipated discovery of cultural resources. Like the proposed project, the Multi-family Residential Alternative would continue to have no impact on historical resources. Therefore, impacts to cultural resources would be equivalent to the proposed project.

Energy

Under the Multi-family Residential Alternative, the project site would be developed with a higher density residential use than compared to the proposed project which could increase energy use at the site. However, the buildings constructed under this alternative would implement the same energy efficiency measures as the proposed project. Additionally, this alternative would be designed to comply with all energy efficiency requirements and would comply with future building code standards to increase energy efficiency, similar to the proposed project. Though this alternative would include energy efficiency measures and would increase energy efficiency at the site, the alternative would result in an increase in energy use from the proposed project and therefore, impacts to energy would be greater when compared to the proposed project.

Geology and Soils

Under this alternative, the project site would continue to be developed with residential uses and would result in the same level of impact as the proposed project. The Multi-family Residential Alternative would require implementation of the same mitigation measures identified for the proposed project in Section 3.7, Geology and Soils of this EIR and would continue to be required to comply with CBC and City guidelines related to seismic and geologic safety. This alternative would be required to implement geotechnical design recommendations and would include mitigation for unstable soil and liquefaction induced impacts, similar to the proposed project. Therefore, the impacts to geology and soils under this alternative would be equivalent to the proposed project.

Greenhouse Gas Emissions

Under the Multi-family Residential Alternative, the project site would be developed with a higher density residential development which could result in increased GHG emissions. However, the Multi-family Residential Alternative would develop new housing near transit, would construct more energy efficient structures, and construct pedestrian and bicycle improvements and therefore, would help to reduce future GHG emissions. Though this alternative would help to reduce future GHG emissions similar to the

Draft Environmental Impact Report Alternatives to the Proposed Project

proposed project, this alternative would generate more vehicle trips than the proposed project and therefore, would increase impacts to GHG from the proposed project. As such, long-term impacts to GHG emissions under this alternative would be greater when compared to the proposed project.

Hazards and Hazardous Materials

If the Multi-family Residential Alternative is implemented, the project would still require remediation of the project site to ensure that existing soil, soil gas, groundwater, and surface water contamination is mitigated. The increase in the number of units and density at the project site would not result in increased impacts to hazards and hazardous materials. The Multi-family Residential Alternative would require the same mitigation measures identified in Section 3.9, Hazards and Hazardous Materials for the proposed project to ensure that impacts are less than significant. Therefore, under this alternative, impacts to hazards and hazardous materials would be equivalent to the proposed project.

Hydrology and Water Quality

The Multi-family Residential Alternative is not anticipated to increase impacts to hydrology and water quality. New stormwater infrastructure constructed for the project would be required to be designed and constructed in accordance with City and C.3 requirements, similar to the proposed project. Additionally, this alternative would require the same mitigation measures identified in Section 3.10, Hydrology and Water Quality of this EIR for the proposed project to ensure that hydrologic and water quality impacts remain less than significant. Therefore, impacts to hydrology and water quality would be equivalent to the proposed project.

Land Use and Planning

Similar to the proposed project, the Multi-family Residential Alternative would provide residential uses at the project site. However, under this alternative, the residential units proposed would be multi-family and would increase the amount of units and density at the site. Similar to the proposed project, the Multi-family Residential Alternative would require rezoning of the site and a Specific Plan Amendment to allow the construction of 405 residential units that would be above the maximum allowable residential units for the Specific Plan area. In addition to the proposed rezoning and Specific Plan Amendment that is part of the proposed project, the Multi-family Residential Alternative would require a General Plan Amendment to increase the allowable density at the site. With the approval of the General Plan Amendment, this alternative would be consistent with the General Plan and land use and planning impacts under this alternative would be similar to the proposed project. Therefore, this alternative result in land use and planning impacts that are equivalent to the proposed project.

Mineral Resources

As discussed in Section 3.12, Mineral Resources of this EIR, the project site is not located in an area containing important mineral resources and the site is not a delineated mineral resource recovery site. Therefore, there would be no impacts to mineral resources under the Multi-family Residential Alternative and impacts would be equivalent compared to the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Noise

Under the Multi-family Residential Alternative, the project site would be developed with more residential units and the proposed density would be increased. Therefore, the construction period required for the alternative may be longer than the construction of the proposed project. However, construction noise and vibration levels would be similar to the proposed project and would be temporary. As with the proposed project, this alternative would be required to implement mitigation measures to reduce construction related noise. As discussed in Section 3.13, Noise of this EIR, the proposed project was determined to result in essentially no change in traffic noise along streets in the vicinity of the project site. This alternative would increase traffic along adjacent streets due to the increase in residents and would increase noise levels. As with the proposed project, this alternative would be required to project, this alternative would be required to project, this alternative would be required to project and would increase in residents and would increase noise levels. As with the proposed project, this alternative would be required to prepare an noise analysis and incorporate the recommendations and design features included in the noise analysis to shield noise. As the increase in residential units and anticipated residents would increase noise levels at the site, it would result in more impacts than what was identified for the proposed project. Therefore, impacts to noise under the Multi-family Residential Alternative would be greater compared to the proposed project.

Population and Housing

Implementation of the Multi-family Residential Alternative would result in an increase in the number of residents generated by the project. This alternative proposes the construction of 405 multi-family units and would generate approximately 1,345 residents, which is an increase from the anticipated 674 residents that would be generated by the proposed project. The Areas 3 and 4 Specific Plan planned for a maximum residential unit allocation of 1,260 units within the Specific Plan area. Currently, all 1,260 units are constructed and/or approved to be constructed within the Specific Plan area and therefore, implementation of this alternative would result in development of residential units above the maximum residential unit allocation of 1,260 units. Though the increase in the number of residential units and anticipated population generated by this alternative would be greater than what was planned in the Areas 3 and 4 Specific Plan, it would not result in unplanned population growth as the proposed number of units and anticipated population would be within the planned number of units within the General Plan and would be within the population and residential unit growth anticipated under the General Plan. Therefore, impacts to population and housing under this alternative would be equivalent to the proposed project.

Public Services

The Multi-family Residential Alternative would result in increased demand for public services above what is anticipated for the proposed project as it would increase the number of residential units and residents at the site. This alternative would require the payment of development impact fees and parks impact fees similar to the proposed project to reduce the increased demand on public services. However, since the alternative would result in more demand for public services compared to the proposed project, impacts to public services would be greater than the proposed project.

Recreation

The Multi-family Residential Alternative would result in greater impacts to parks and recreational facilities when compared to the proposed project. The alternative would result in the generation of more residents than the proposed project and therefore, would increase the demand on existing parks and recreational



Draft Environmental Impact Report Alternatives to the Proposed Project

facilities. The project under this alternative would be required to provide more park space or funding to reduce impacts, similar to the proposed project. Impacts to recreation would be greater under this alternative compared to the proposed project.

Transportation

Under the Multi-family Residential Alternative, the project would result in an increased number of residential units and residents generated compared to the proposed project. The increase in the number of residents anticipated would result in increased vehicle trips above what was identified for the proposed project.

The VMT for this alternative is calculated using California Air Pollution Control Officers Association's (CAPCOA) Strategy T-1, which correlates VMT with residential density. Since the alternative would have a density of 14 units per acre, it would generate 12 percent less VMT than the proposed project according to CAPCOA. Under this alternative, the project would have an average VMT per resident of 24.6, which is above the threshold of significance of 19.4 (i.e., 15 percent below citywide average). Though this alternative would reduce VMT by 12 percent from the proposed project's identified VMT of 27.9 to this alternative's identified VMT of 24.6, VMT under this alternative would continue to be above the threshold and would continue to have a significant impact. Therefore, this alternative would continue to have a significant and unavoidable impact on VMT and impacts to transportation would be equivalent compared to the proposed project.

Tribal Cultural Resources

The Multi-family Residential Alternative would continue to require subsurface ground disturbance activities which could result in the inadvertent discovery of unknown tribal cultural resources at the project site. The amount of ground disturbance required would be similar to the proposed project as this alternative would include development of the same project site footprint as the proposed project. Potential impacts would be reduced to a less than significant level with the incorporation of the same mitigation measures identified for the proposed project to address the unanticipated discovery of tribal cultural resources. Therefore, impacts to tribal cultural resources would be equivalent to the proposed project.

Utilities and Service Systems

The Multi-family Residential Alternative would increase demand for water, wastewater, and solid waste services and would require greater supplies to serve the proposed 405 unit multi-family residential development. However, the proposed number of units would be within the residential allocation identified in the Areas 3 and 4 Specific Plan and the WSA prepared for the Specific Plan identified that with compliance with requirements provided in the WSA, there would be sufficient water supplies to support the residential allocation identified in the Specific Plan. Though there would be sufficient supplies to serve this alternative, the increase in number of residential units under this alternative would place greater demand on water, wastewater, and solid waste service providers and therefore, impacts to utilities and service systems would be greater compared to the proposed project.



Wildfire

The Multi-family Residential Alternative would continue to construct residential uses at the site and would not result in construction in areas outside of the project site footprint that may be identified as having a higher potential fire hazard than the project site. The alternative would not introduce other uses that could increase wildfire hazard and therefore, impacts to wildfire would be equivalent to the proposed project.

Conclusion and Relationship to Project Objectives

The Multi-family Residential Alternative would not eliminate or reduce the significant and unavoidable impact to VMT. This alternative would have equivalent impacts compared to the proposed project on aesthetics, biological resources, cultural resources, geology and soils, hazards and hazardous materials, land use and planning, mineral resources, population and housing, tribal cultural resources, and wildfire. The Multi-family Residential Alternative would result in greater impacts to air quality, energy, GHG, noise public services, recreation, and utilities and service systems. The Multi-family Residential Alternative would not achieve the project objective shown below:

• Implement the City's General Plan by developing the site with low-density residential.

All other project objectives would be achieved under this alternative. This alternative would require modification of the site layout and conversion from single-family residential units to multi-family residential units. This alternative would result in a density of 14 units per acre and would require a General Plan Amendment to designate the site as medium density residential. As such, this alternative would not meet the project objective to provide low density residential uses at the site.

5.5.3 Alternative 3 – Reduced Density Alternative

Under the Reduced Density Alternative, the project site would be developed with 64 single-family detached residential homes on the 29 acre project site. The Reduced Density Alternative would have a resulting density of approximately 2.2 dwelling units per acre which is within the allowable density of 8.7 dwelling units per acre for the Low Density Residential General Plan land use designation. The Reduced Density Alternative would continue to involve the demolition of existing uses and structures at the project site, remediation of the project site, and would involve additional on- and off-site improvements, similar to the proposed project. Though the Reduced Density Alternative's density of 2.2 dwelling units per acre would be within the allowable density for Low Density Residential General Plan land use designation, it would underutilize the allowable density of the designation of the project site.

Impact Analysis

Aesthetics

Under the Reduced Density Alternative, the heights of the proposed buildings would be similar to the proposed project as the site would be developed with single-family homes. The analysis in Section 3.1, Aesthetics, identified that the proposed project would not result in significant impacts to scenic resources given the amount of separation between the project site and scenic views. Though this alternative would increase building heights from current conditions and change the character of the site, the distance


Draft Environmental Impact Report Alternatives to the Proposed Project

between the project and scenic views would remain the same and impacts to aesthetics would be equivalent to the proposed project.

Agricultural and Forestry Resources

Under the Reduced Density Alternative, the project site would be developed with low density residential uses, similar to the proposed project. The alternative would be located within the same project site footprint as the proposed project and would not develop any agricultural and forestry uses. Therefore, the Reduced Density Alternative would have no impact related to agricultural and forestry resources and impacts would be equivalent to the proposed project.

Air Quality

Under the Reduced Density Alternative, the amount and duration of construction required for development would be reduced from the proposed project as this alternative would only construct 64 single-family homes, compared to the 203 single-family homes proposed for the proposed project. Therefore, construction related air quality impacts would be reduced compared to the proposed project though the project would still be required to continue to implement construction mitigation measures identified in Section 3.3, Air Quality of this EIR. Additionally, operational emissions under the Reduced Density Alternative would be reduced when compared to the proposed project resulting from the decreased number of residential units and the generation of less residents at the site. The net operational emissions under the proposed project were determined to be well below the BAAQMD project level threshold and therefore, operational emissions would be further reduced below the BAAQMD threshold under this alternative. Therefore, air quality impacts under the Reduced Density Alternative would be less compared to the proposed project.

Biological Resources

Under the Reduced Density Alternative, the entirety of the project site would continue to be developed with residential uses and the project would continue to require removal of existing trees on-site. Additionally, development of the Reduced Density Alternative would result in the same biological resources impacts as the proposed project as it would be located within the same project footprint. Therefore, this alternative would require implementation of the same mitigation measures identified for the proposed project and impacts resulting from the Reduced Density Alternative would be equivalent to the proposed project.

Cultural Resources

The Reduced Density Alternative would continue to require subsurface ground disturbance activities which could result in the inadvertent discovery of unknown archeological resources at the project site. The amount of ground disturbance required would be similar to the proposed project as this alternative would include development of the same project site footprint as the proposed project. Potential impacts would be reduced to a less than significant level with the incorporation of the same mitigation measures identified for the proposed project to address the unanticipated discovery of cultural resources. Like the proposed project, the Reduced Density Alternative would continue to have no impact on historical resources. Therefore, impacts to cultural resources would be equivalent to the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Energy

Under the Reduced Density Alternative, the project site would be developed with a lower density residential use than the proposed project which would reduce energy use at the site. The buildings constructed under this alternative would implement the same energy efficiency measures as the proposed project. Additionally, this alternative would be designed to comply with all energy efficiency requirements and would comply with future building code standards to increase energy efficiency, similar to the proposed project. As this alternative would result in reduced energy use at the site from the proposed project, impacts to energy would be less when compared to the proposed project.

Geology and Soils

Under the Reduced Density Alternative, the project site would continue to be developed with residential uses and would result in the same level of impact as the proposed project. The Reduced Density Alternative would require implementation of the same mitigation measures identified for the proposed project in Section 3.7, Geology and Soils of this EIR and would continue to be required to comply with CBC and City guidelines related to seismic and geologic safety. Accordingly, the impacts to geology and soils under this alternative would be equivalent to the proposed project.

Greenhouse Gas Emissions

Under the Reduced Density Alternative, the project site would be developed with a lower density residential development which would result in less GHG emissions than the proposed project. The Reduced Density Alternative would develop new housing near transit, would construct more energy efficient structures, and construct pedestrian and bicycle improvements and therefore, would help to reduce future GHG emissions. This alternative would generate less vehicle trips than the proposed project as the alternative project would only develop 64 residential units compared to the proposed project's 203 residential units and therefore, would decrease impacts to GHG from the proposed project. As such, long-term impacts to GHG emissions under this alternative would be less when compared to the proposed project.

Hazards and Hazardous Materials

Under the Reduced Density Alternative, the project would still require remediation of the project site to ensure that existing soil, soil gas, groundwater, and surface water contamination is mitigated. The decrease in the number of units and density at the project site would not result in a change in the level of impacts to hazards and hazardous materials. The Reduced Density Alternative would require the same mitigation measures identified in Section 3.9, Hazards and Hazardous Materials for the proposed project to ensure that impacts are less than significant. Therefore, under this alternative, impacts to hazards and hazardous materials would be equivalent to the proposed project.

Hydrology and Water Quality

The Reduced Density Alternative is not anticipated to increase impacts to hydrology and water quality. This alternative would require construction of new stormwater infrastructure to serve the project and new stormwater infrastructure constructed for the project would be required to be designed and constructed in accordance with City and C.3 requirements, similar to the proposed project. Additionally, this alternative



Draft Environmental Impact Report Alternatives to the Proposed Project

would require the same mitigation measures identified in Section 3.10, Hydrology and Water Quality of this EIR for the proposed project to ensure that hydrologic and water quality impacts remain less than significant. Therefore, impacts to hydrology and water quality would be equivalent to the proposed project.

Land Use and Planning

Under the Reduced Density Alternative, the project site would continue to be developed with low density uses and this alternative would be consistent with the site's Low Density Residential General Plan land use designation. Though this alternative would contribute to increasing housing quantity within the City to support it in meeting its RHNA target and the Reduced Density Alternative's density of two units per acre would be within the allowable density for Low Density Residential General Plan land use designation, it would underutilize the allowable density of the designation of the project site. This alternative would comply with the General Plan land use designation of Low Density Residential as the designation required less than 8.7 units per acre. However, the Areas 3 and 4 Specific Plan did not envision lot sizes for residential developments with a two unit per acre density. This alternative would not require any additional approvals. Therefore, this alternative would be equally consistent with the City's General Plan and Zoning Code compared to the proposed project and would not result in increased land use and planning impacts. Impacts to land use and planning would be equivalent to the proposed project.

Mineral Resources

As discussed in Section 3.12, Mineral Resources of this EIR, the project site is not located in an area containing important mineral resources and the site is not a delineated mineral resource recovery site. Therefore, there would be no impacts to mineral resources under the Reduced Density Alternative and impacts would be equivalent to the proposed project.

Noise

Under the Reduced Density Alternative, the project site would be developed with less residential units and the proposed density would be reduced. Therefore, the construction period required for the alternative may be shorter than the construction period required for the proposed project. Though the construction period may be shorter, construction noise and vibrations levels would be similar to the proposed project and would be temporary. As with the proposed project, this alternative would be required to implement mitigation measures to reduce construction related noise level. Operation of this alternative would result in less vehicle trips on surrounding roadways as it would generate less residents and traffic noise would be reduced. As with the proposed project, this alternative would be required to prepare a noise analysis and incorporate the recommendations and design features included in the noise analysis to shield noise. As the decrease in residential units and anticipated residents would decrease noise levels at the site, it would result in less impacts than what was identified for the proposed project. Therefore, impacts to noise under the Reduced Density Alternative would be less than the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Population and Housing

Implementation of the Reduced Density Alternative would result in a decrease in the number of residents generated by the project. This alternative proposes the construction of 64 single-family units and would generate approximately 212 residents, which is a decrease from the anticipated 674 residents that would be generated by the proposed project. Similar to the proposed project, this alternative would require rezoning of the site and a Specific Plan Amendment to allow for the development of 64 residential units above the maximum allowed residential unit allocation of 1,260 units within the Specific Plan. Though the Reduced Density Alternative would not be within the number of residential units and anticipated population that was planned in the Areas 3 and 4 Specific Plan, the number of residential units proposed under this alternative and its anticipated population would not result in unplanned population growth as the proposed number of units would be within the planned number of units within the General Plan and would be within the population and residential unit growth anticipated under the General Plan. Therefore, impacts to population and housing under this alternative would be equivalent to the proposed project.

Public Services

The Reduced Density Alternative would result in decreased demand for public services from what is anticipated for the proposed project as it would decrease the number of residential units and residents at the site. This alternative would require the payment of development impact fees and parks impact fees similar to the proposed project to reduce the increased demand on public services from current conditions. However, since the alternative would result in less demand for public services compared to the proposed project, impacts to public services would be less than the proposed project.

Recreation

The Reduced Density Alternative would result in less impacts to parks and recreational facilities when compared to the proposed project. The alternative would result in the generation of less residents than the proposed project and therefore, would decrease the demand on existing parks and recreational facilities. The project under this alternative would be required to provide park space or funding to reduce impacts similar to the proposed project. Impacts to recreation would be less under this alternative compared to the proposed project.

Transportation

Under the Reduced Density Alternative, there would be less vehicle trips generated compared to the proposed project as this alternative would reduce the number of residential units and residents from what is proposed under the proposed project. VMT generated by the development of this alternative would not result in a net new increase in VMT in comparison to the existing VMT generated by the existing uses.

Existing daily VMT generated by the existing Pick-n-Pull uses on-site is identified to be 6,023. With a project daily VMT per person of 27.9 and household size of 3.36 used by Fehr & Peers for the preparation of the Transportation Impact Analysis for the proposed project, this alternative would result in a project daily VMT of 6,023. Therefore, this alternative would not result in net new VMT from current conditions. However, the daily VMT per person of 27.9 anticipated for the project would be above the threshold of significant of 19.4 (i.e., 15 percent below citywide average). Therefore, this alternative would continue to



Draft Environmental Impact Report Alternatives to the Proposed Project

have a significant and unavoidable impact to VMT and impacts to transportation would be equivalent to the proposed project.

Tribal Cultural Resources

The Reduced Density Alternative would continue to require subsurface ground disturbance activities which could result in the inadvertent discovery of unknown tribal cultural resources at the project site. The amount of ground disturbance required fort his alternative would be similar to the proposed project as this alternative proposed development of the entire 29 acre project site would include development of the same project site footprint as the proposed project. Potential impacts would be reduced to a less than significant level with the implementation of the same mitigation measures identified for the proposed project to address inadvertent discovery of tribal cultural resources. Therefore, impacts to tribal cultural resources under this alternative would be equivalent to the proposed project.

Utilities and Service Systems

The Reduced Density Alternative would decrease demand for water, wastewater, and solid waste services and would require less supplies to serve the proposed 64 single-family residential development. The proposed number of units would be within the residential allocation identified in the Areas 3 and 4 Specific Plan and the WSA prepared for the Specific Plan identified that with compliance with requirements provided in the WSA, there would be sufficient water supplies to serve this alternative and the decrease in number of residential units under this alternative would place less demand on water, wastewater, and solid waste service providers and therefore, impacts to utilities and service systems would be less compared to the proposed project.

Wildfire

The Reduced Density Alternative would continue to construct residential uses at the site and would not result in construction in areas outside of the project site footprint that may be identified as having a higher potential fire hazard than the project site. The alternative would not introduce other uses that could increase wildfire hazard and therefore, impacts to wildfire would be equivalent to the proposed project.

Conclusion and Relationship to Project Objectives

Though the Reduced Density Alternative would not result in a net increase in VMT above existing conditions, VMT generated by this alternative would be above the threshold of significance and therefore, this alternative would not eliminate the significant and unavoidable impact to VMT. This alternative would have equivalent impacts compared to the proposed project on aesthetics, agriculture and forestry resources, biological resources, cultural resources, geology and soils, hazards and hazardous materials, hydrology and water quality, land use and planning, mineral resources, population and housing, transportation, tribal cultural resources, and wildfire. The Reduced Density Alternative would have less impacts compared to the proposed project on air quality, energy, GHG, noise, public services, recreation, and utilities and service systems. The Reduced Density Alternative would achieve all of the project objectives.



Draft Environmental Impact Report Alternatives to the Proposed Project

Though the Reduced Density Alternative's density of two units per acre would be within the allowable density for Low Density Residential General Plan land use designation, it would underutilize the allowable density of the designation of the project site. Additionally, the Areas 3 and 4 Specific Plan did not envision lot sizes for residential developments with a two unit per acre density. The demolition of the existing uses, remediation of the project site, and construction of single-family homes within the project site would be extensive and costly and the high costs the Applicant would incur from development of this alternative would be greater than the revenue generated by development of this alternative.

5.5.4 Alternative 4 – 100 Percent Affordable Housing Alternative

Under the 100 Percent Affordable Housing Alternative, the project would include the demolition of existing uses and remediation of existing contaminants on-site, same as the proposed project. This alternative is similar to the Multi-family Residential Alternative; however, under this alternative, the project would develop 405 multi-family residential units that are 100 percent affordable. Under the 100 Percent Affordable Housing Alternative, the project would construct 405 affordable multi-family residential units on 29 acres, resulting in a density of approximately 14 units per acre and therefore, would require a General Plan Amendment from Low Density Residential to Medium Density Residential to allow for the increased density. This alternative assumed that no density bonus law would be applied. This alternative would provide multi-family residential units that range from garden apartments and condominiums to townhomes and row houses that would have a maximum building height of 60 feet, in accordance with the development standards of the Areas 3 and 4 Specific Plan.

Impact Analysis

Aesthetics

Under the 100 Percent Affordable Housing Alternative, the heights of the proposed buildings would be increased and the site would be developed with a higher density. The analysis in Section 3.1, Aesthetics, identified that the proposed project would not result in significant impacts to scenic resources given the amount of separation between the project site and scenic views. Though this alternative would increase building heights and change the character of the site, the distance between the project and scenic views would remain the same and impacts to aesthetics would be equivalent to the proposed project.

Agricultural and Forestry Resources

Under the 100 Percent Affordable Housing Alternative, the project site would be developed with residential uses, similar to the proposed project. The alternative would be located within the same project site footprint as the proposed project and would not develop any agricultural and forestry uses. Therefore, the Multi-family Residential Alternative would have no impact related to agricultural and forestry resources and impacts would be equivalent to the proposed project.

Air Quality

Under the 100 Percent Affordable Housing Alternative, construction of the project may require a longer construction period due to the increased number of residential units and may result in greater construction related air quality impacts. However, construction related emissions would be mitigated with the implementation of the same mitigation measures required for construction of the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Additionally, operational emissions under the 100 Percent Affordable Housing Alternative would be greater when compared to the proposed project resulting from the increased number of residential units and the generation of more residents at the site. The net operational emissions under the proposed project were determined to be well below the BAAQMD project level threshold and it is not anticipated that implementation of the 100 Percent Affordable Housing Alternative would not result in operational emissions that exceed the identified threshold. Though the 100 Percent Affordable Housing Alternative would not substantially increase short-term construction or longer-term operational emissions, emissions would be increased from what was identified for the proposed project and impact to air quality would be greater when compared to the proposed project.

Biological Resources

Under the 100 Percent Affordable Housing Alternative, the entirety of the project site would continue to be developed with residential uses and the project would continue to require removal of existing trees onsite. Additionally, development of the 100 Percent Affordable Housing Alternative would result in the same biological resources impacts as the proposed project as it would be located within the same project footprint. Therefore, this alternative would require implementation of the same mitigation measures identified for the proposed project and impacts resulting from the 100 Percent Affordable Housing Alternative would be equivalent to the proposed project.

Cultural Resources

The 100 Percent Affordable Housing Alternative would continue to require and involve subsurface ground disturbance activities which could result in the inadvertent discovery of unknown archeological resources at the project site. The amount of ground disturbance required would be similar to the proposed project as this alternative would include development of the same 29 acre project site as the proposed project. Potential impacts would be reduced to a less than significant level with the incorporation of the same mitigation measures identified for the proposed project to address the unanticipated discovery of cultural resources. Like the proposed project, the 100 Percent Affordable Housing Alternative would continue to have no impact on historical resources. Therefore, impacts to cultural resources would be equivalent to the proposed project.

Energy

Under the 100 Percent Affordable Housing Alternative, the project site would be developed with a higher density residential use than compared to the proposed project which could increase energy use at the site. However, the buildings constructed under this alternative would implement the same energy efficiency measures as the proposed project. Additionally, this alternative would be designed to comply with all energy efficiency requirements and would comply with future building code standards to increase energy efficiency, similar to the proposed project. Though the 100 Percent Affordable Housing Alternative would include energy efficiency measures and would increase energy efficiency at the site, the alternative would result in an increase in energy use from the proposed project and therefore, impacts to energy would be greater when compared to the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Geology and Soils

Under the 100 Percent Affordable Housing Alternative, the project site would be developed with 405 multi-family affordable housing units but would result in the same level of impact as the proposed project. The 100 Percent Affordable Housing Alternative would require implementation of the same mitigation measures identified for the proposed project in Section 3.7, Geology and Soils of this EIR and would continue to be required to comply with CBC and City guidelines related to seismic and geologic safety. The proposed project includes the implementation of geotechnical design recommendations and includes mitigation for unstable soil and liquefaction induced impacts and this alternative would be required to implement the same measures to ensure a less than significant level of impact. Therefore, the impacts to geology and soils under the 100 Percent Affordable Housing Alternative would be equivalent to the proposed project.

Greenhouse Gas Emissions

Under the 100 Percent Affordable Housing Alternative, the project site would be developed with a higher density residential development with 405 multi-family units which could result in increased GHG emissions. The 100 Percent Affordable Housing Alternative would develop new housing near transit, would construct more energy efficient structures, and construct pedestrian and bicycle improvements and therefore, would help to reduce future GHG emissions. Though this alternative would help to reduce future GHG emissions similar to the proposed project the 100 Percent Affordable Housing Alternative would generate more vehicle trips than the proposed project and therefore, would increase impacts to GHG from what was identified for the proposed project. As such, long-term impacts to GHG emissions under the 100 Percent Affordable Housing Alternative would be greater when compared to the proposed project.

Hazards and Hazardous Materials

Under the 100 Percent Affordable Housing Alternative, the project would continue to require remediation of the project site to ensure that existing soil, soil gas. groundwater, and surface water contamination is mitigated. The increase in the number of units and density at the project site would not result in increased impacts to hazards and hazardous materials. The 100 Percent Affordable Housing Alternative would require the same mitigation measures identified in Section 3.9, Hazards and Hazardous Materials for the proposed project to ensure that impacts are less than significant. Therefore, under the 100 Percent Affordable Housing Alternative, impacts to hazards and hazardous materials would be equivalent to the proposed project.

Hydrology and Water Quality

The 100 Percent Affordable Housing Alternative is not anticipated to increase impacts to hydrology and water quality. Under the 100 Percent Affordable Housing Alternative, the project would construct new stormwater infrastructure that would be required to be designed and constructed in accordance with City and C.3 requirements, similar to the proposed project. Additionally, this alternative would require the same mitigation measures identified in Section 3.10, Hydrology and Water Quality of this EIR for the proposed project to ensure that hydrologic and water quality impacts remain less than significant. Therefore, impacts to hydrology and water quality would be equivalent to the proposed project.



Draft Environmental Impact Report Alternatives to the Proposed Project

Land Use and Planning

The 100 Percent Affordable Housing Alternative would provide residential uses at the project site, similar to the proposed project. However, under the 100 Percent Affordable Housing Alternative, the residential units proposed would be multi-family and would increase the amount of units and density at the site. Additionally, the multi-family units would be 100 percent affordable. Similar to the proposed project, the 100 Percent Affordable Housing Alternative would require rezoning of the site and A Specific Plan Amendment to allow the construction of 405 residential units that would be above the maximum allowable residential units for the Areas 3 and 4 Specific Plan Area. In addition to the proposed rezoning and Specific Plan Amendment that is part of the proposed project, the 100 Percent Affordable Housing Alternative would require a General Plan Amendment to increase the allowable density at the site. With the approval of the General Plan Amendment, this alternative would be consistent with the General Plan and land use and planning impacts under this alternative would result in land use and planning impacts the Housing Alternative would result in land use and planning impacts that are equivalent to the proposed project.

Mineral Resources

As discussed in Section 3.12, Mineral Resources of this EIR, the project site is not located in an area containing important mineral resources and the site is not a delineated mineral resource recovery site. Therefore, there would be no impacts to mineral resources under the 100 Percent Affordable Housing Alternative and impacts would be equivalent compared to the proposed project.

Noise

Under the 100 Percent Affordable Housing Alternative, the project site would be developed with 405 multi-family residential units and the proposed density would be increased. Due to the increase in number of units, the construction period required for the alternative may be longer than the construction of the proposed project. However, construction noise and vibration levels would be similar to the proposed project and would be temporary. Under the 100 Percent Affordable Housing Alternative, the project would be required to implement mitigation measures to reduce construction related noise, similar to the proposed project. As discussed in Section 3.13, Noise of this EIR, the proposed project was determined to result in essentially no change in traffic noise along streets in the vicinity of the project site. This alternative would increase traffic along adjacent streets due to the increase in residents and would increase noise levels. As with the proposed project, this alternative would be required to prepare a noise analysis and incorporate the recommendations and design features included in the noise analysis to shield noise. The increase in residential units and anticipated residents would increase noise levels at the site which would result in more impacts than what was identified for the proposed project. Therefore, impacts to noise under the 100 Percent Affordable Housing Alternative would be greater compared to the proposed project.

Population and Housing

Implementation of the 100 Percent Affordable Housing Alternative would result in an increase in the number of residents generated by the project as the project would develop 405 residential units. This alternative proposes the construction of 405 multi-family units and would generate approximately 1,345 residents, which is an increase from the anticipated 674 residents that would be generated by the



Draft Environmental Impact Report Alternatives to the Proposed Project

proposed project. The Areas 3 and 4 Specific Plan planned for a maximum residential unit allocation of 1,260 units within the Specific Plan area. Currently, all 1,260 units are constructed and/or approved to be constructed within the Specific Plan area and therefore, implementation of this alternative would result in development of residential units above the maximum residential unit allocation of 1,260 units. Though the increase in the number of residential units and anticipated population generated by this alternative would be greater than what was planned in the Areas 3 and 4 Specific Plan, it would not result in unplanned population growth as the proposed number of units and anticipated population would be within the planned number of units within the General Plan and would be within the population and residential unit growth anticipated under the General Plan. Therefore, impacts to population and housing under the 100 Percent Affordable Housing Alternative would be equivalent to the proposed project.

Public Services

The 100 Percent Affordable Housing Alternative would result in increased demand for public services above what is anticipated for the proposed project as it would increase the number of residential units and residents at the site. This alternative would require the payment of development impact fees and parks impact fees to reduce the increased demand on public services, similar to the proposed project. As the alternative would result in more demand for public services compared to the proposed project, impacts to public services under the 100 Percent Affordable Housing Alternative would be greater than the proposed project.

Recreation

The 100 Percent Affordable Housing Alternative would result in a larger resident population than the proposed project and therefore, would result in increased demand and greater impacts to parks and recreational facilities when compared to the proposed project. The project under the 100 Percent Affordable Housing Alternative would be required to provide more park space or funding to reduce impacts, similar to the proposed project. Accordingly, impacts to recreation would be greater under the 100 Percent Affordable Housing Alternative when compared to the proposed project.

Transportation

Under the 100 Percent Affordable Housing Alternative, the project would result in increased number of residential units and residents generated compared to the proposed project. The increase in the number of residents anticipated would result in increased vehicle trips above what was identified for the proposed project.

Though the 100 Percent Affordable Housing Alternative is similar to the Multi-family Residential Alternative as it would provide the same number of multi-family residential units, VMT is calculated differently as this alternative include 100 percent affordable housing. The VMT for this alternative is calculated using CAPCOA's Strategy T-2c, which correlates VMT with affordability. Assuming the 100 percent of units are below market rate, CAPCOA estimates a maximum 1.2 percent reduction in VMT compared to market rate units. Therefore, this alternative would result in a 1.2 percent reduction in VMT from the Multi-family Residential Alternative. Under this alternative, the project would have an average VMT per resident of 24.3, which is above the threshold of significance of 19.4 (i.e., 15 percent below citywide average). Though this alternative would reduce VMT by 13 percent from the proposed project's identified VMT of 27.9 to this alternative's identified VMT of 24.3, VMT under this alternative would



Draft Environmental Impact Report Alternatives to the Proposed Project

continue to be above the threshold and would continue to have a significant impact. Therefore, this alternative would continue to have a significant and unavoidable impact on VMT and impacts to transportation would be equivalent compared to the proposed project.

Tribal Cultural Resources

Under the 100 Percent Affordable Housing Alternative, the project would continue to require subsurface ground disturbance activities which could result in the inadvertent discovery of unknown tribal cultural resources at the project site. The amount of ground disturbance required would be similar to the proposed project as this alternative would include development of the same project site footprint as the proposed project. Potential impacts to tribal cultural resources under this alternative would be reduced to a less than significant level with the incorporation of the same mitigation measures identified for the proposed project to address the unanticipated discovery of tribal cultural resources. Therefore, impacts to tribal cultural resources would be equivalent to the proposed project.

Utilities and Service Systems

The 100 Percent Affordable Housing Alternative would increase demand for water, wastewater, and solid waste services and would require greater supplies to serve the proposed 405 multi-family residential development. However, the proposed number of units would be within the residential allocation identified in the Areas 3 and 4 Specific Plan and the WSA prepared for the Specific Plan identified that with compliance with requirements provided in the WSA, there would be sufficient water supplies to support the residential allocation identified in the Specific Plan. Though there would be sufficient supplies to serve the 100 Percent Affordable Housing Alternative, the increase in number of residential units under this alternative would place greater demand on water, wastewater, and solid waste service providers and therefore, impacts to utilities and service systems would be greater compared to the proposed project.

Wildfire

The 100 Percent Affordable Housing Alternative would continue to construct residential uses at the site and would not result in construction in areas outside of the project site footprint that may be identified as having a higher potential fire hazard than the project site. The 100 Percent Affordable Housing Alternative would not introduce other uses that could increase wildfire hazard and therefore, impacts to wildfire would be equivalent to the proposed project.

Conclusion and Relationship to Project Objectives

Under this alternative, the project site would be constructed with 405 multi-family residential units with 100 percent of the units being allocated for affordable housing. The 100 Percent Affordable Housing Alternative would not eliminate or significantly reduce the significant and unavoidable impact to VMT. Under this alternative, the project would have equivalent impacts compared to the proposed project on aesthetics, biological resources, cultural resources, geology and soils, hazards and hazardous materials, land use and planning, mineral resources, population and housing, tribal cultural resources, and wildfire. The 100 Percent Affordable Housing Alternative would result in greater impacts to air quality, energy, GHG, noise, public services, recreation, and utilities and service systems. The 100 Percent Affordable Housing Alternative the project objective shown below:



Draft Environmental Impact Report Alternatives to the Proposed Project

• Implement the City's General Plan by developing the site with low-density residential.

All other project objectives would be met under this alternative. This alternative would require modifications to the site layout and conversion of units from single-family residential units to multi-family residential units. This alternative would result in a density of 14 units per acre and would require a General Plan Amendment to designate the site as medium density residential. As such, this alternative would not meet the project objective to provide low density residential uses at the site.

5.6 ENVIRONMENTALLY SUPERIOR ALTERNATIVE

CEQA Guidelines Section 15126.6(e)(2) requires an EIR to identify an "environmentally superior alternative." The qualitative environmental effects of each alternative in relation to the proposed project are summarized in Table 5.6-1. To quantitatively identify an environmentally superior alternative a value has been applied to each environmental effect. Additionally, Table 5.6-2 provides a comparison of the alternatives with the project objectives. Accordingly, the alternative with the fewest amounts of impacts and the ability to achieve the most project objectives is the environmentally superior alternative.

The Reduced Density Alternative is the environmentally superior alternative. Though the Reduced Density Alternative would not eliminate or reduce the significant and unavoidable impact to VMT, all other resource areas would be less than significant or less than significant with mitigation. Additionally, the Reduced Density Alternative would not result in greater impacts than the proposed project to any resource areas and would meet all project objectives.

Draft Environmental Impact Report Alternatives to the Proposed Project

Table 5.6-1: Project Alternative Impacts Comparison

Environmental Resource Area	Proposed Project	No Project (Alternative 1)	Multi-family Residential (Alternative 2)	Reduced Density (Alternative 3)	100 Percent Affordable (Alternative 4)
Aesthetics	LTS	<	=	=	=
Agriculture and Forestry Resources	NI	=	=	=	=
Air Quality	LTS/M	<	>	<	>
Biological Resources	LTS/M	<	=	=	=
Cultural Resources	LTS/M	<	=	=	=
Energy	LTS	>	>	<	>
Geology and Soils	LTS/M	>	=	=	=
Greenhouse Gases	LTS	>	>	<	>
Hazards and Hazardous Materials	LTS/M	>	=	=	=
Hydrology and Water Quality	LTS/M	>	=	=	=
Land Use and Planning	LTS	>	=	=	=
Mineral Resources	NI	=	=	=	=
Noise	LTS/M	>	>	<	>



Draft Environmental Impact Report Alternatives to the Proposed Project

Population and Housing	LTS	<	=	=	=
Public Services	LTS	<	>	<	~
Recreation	LTS	<	>	<	×
Transportation and Traffic	SU	<	=	=	=
Tribal Cultural Resources	LTS/M	<	=	=	=
Utilities and Service Systems	LTS	<	>	<	>
Wildfire	LTS	<	=	=	=

Notes:

Notes. NI = No Impact LTS = Less than Significant Impact LTS/M = Less than Significant Impact with Mitigation SU = Significant and Unavoidable < = Less impact than the proposed project = = Equivalent impact to the proposed project > = Constant and the proposed project

> = Greater impact than the proposed project



Draft Environmental Impact Report Alternatives to the Proposed Project

Table 5.6-2: Project Alternatives Comparison to Project Objectives

Project Objectives	Proposed Project	Multi-family Residential (Alternative 2)	Reduced Density (Alternative 3)	100 Percent Affordable (Alternative 4)
Implement the City's General Plan by developing the site with low-density residential.	Х		Х	
Support the City in meeting its RHNA target assigned by the ABAG.	Х	Х	Х	Х
Provide high quality residential uses including a mix of lot sizes.	Х	Х	Х	Х
Minimize environmental impacts associated with residential development by siting the project on developed and disturbed lands	X	Х	Х	Х
Remediate contaminated soil on-site to levels suitable for residential development;	Х	Х	Х	Х
Create a residential development that integrates multi-modal transportation (pedestrian, bicycle, automobile) and connects the development to existing, nearby bus transit stops and active centers by improving Mowry Avenue and upgrading the at-grade vehicular and pedestrian crossing along Mowry Avenue at the UPRR railroad tracks to increase safety.	X	Х	Х	X

Draft Environmental Impact Report Alternatives to the Proposed Project

This page intentionally left blank.

6.0 OTHER CEQA CONSIDERATIONS

This section describes the other statutorily required topics including growth inducing impacts, significant and unavoidable impacts, significant irreversible environmental changes, and mandatory findings of significance. It also provides a discussion of energy conservation as required by Section 15126.4 of the CEQA Guidelines.

6.1 GROWTH-INDUCING IMPACTS

Section 15126.2(d) of the CEQA Guidelines requires that an EIR evaluate the growth-inducing impacts of a proposed action:

Discuss the way in which a proposed project could foster economic or population growth, or the construction of additional housing, either directly or indirectly, in the surrounding environment. Included in this are projects that would remove obstacles to population growth (a major expansion of a wastewater treatment plant might, for example, allow for more construction in service areas). Increases in the population may tax existing community service facilities, requiring construction of new facilities that could cause significant environmental effects. Also discuss the characteristic of some projects which may encourage and facilitate other activities that could significantly affect the environment, either individually or cumulatively. It must not be assumed that growth in any area is necessarily beneficial, detrimental, or of little significance to the environment.

The State CEQA Guidelines do not distinguish between planned and unplanned growth for purposes of considering whether a project would foster additional growth. Therefore, for purposes of this EIR, to reach the conclusion that a project is growth-inducing as defined by CEQA, the EIR must find that it would foster (i.e., promote, encourage, or allow) additional growth in economic activity, population, or housing, regardless of whether the growth is already approved by and consistent with local plans. The conclusion does not determine that induced growth is beneficial or detrimental, consistent with Section 15126.2(d) of the State CEQA Guidelines. If the analysis conducted for the EIR results in a determination that a project is growth-inducing, the next question is whether that growth may cause adverse effects on the environment. Environmental effects resulting from induced growth (i.e., growth-induced effects) fit the CEQA definition of "indirect" effects in Section 15358(a)(2) of the State CEQA Guidelines. These indirect or secondary effects of growth may result in significant environmental impacts. CEQA does not require that the EIR speculate unduly about the precise location and site-specific characteristics of significant, indirect effects caused by induced growth, but a good-faith effort is required to disclose what is feasible to assess. Growth-inducing impacts can occur when development of a project imposes new burdens on a community by directly inducing population growth, or by leading to the construction of additional development in the project area. Also included in this category are projects that would remove physical obstacles to population growth, such as the construction of a new roadway into an undeveloped area or a wastewater treatment plant with excess capacity to serve additional new development. Construction of these types of infrastructure projects cannot be considered isolated from the immediate development that they facilitate and serve. Projects that physically remove obstacles to growth or projects that indirectly induce growth are those that may provide a catalyst for future unrelated development in the area (such as a new residential community that requires additional commercial uses to support residents). The growth-



inducing potential of a project could also be considered significant if it fosters growth in excess of what is assumed in the local master plans and land use plans, or in projections made by regional planning agencies.

6.1.1 Population Growth

The decision to allow/approve projects that result from induced growth (e.g., new commercial areas, new housing) is the subject of separate discretionary processes by individual lead agency (or agencies) responsible for considering such projects, in this case, the City Planning Commission or, on appeal, the City Council. Projects resulting from induced growth would themselves be discretionary and subject to CEQA. Therefore, the following discussion is intended to disclose the potential for environmental effects that could occur more generally because of the project rather than the site-specific impacts of induced growth. Its purpose is to inform the City decision-making body that additional environmental effects may be a possibility if growth-inducing projects are approved. However, the decision of whether projects are approved and the impacts associated with them still rests with the City decision-making body at such times as complete applications for development are submitted.

The proposed project would cause direct population growth by constructing 203 single-family residential units as part of Mowry Village. These dwelling units would directly generate population growth through an estimated 674 new residents to the City's population. The project is an in-fill development on existing developed but underutilized land and would not induce development in the area beyond that which has already been planned for as part of the General Plan and the Areas 3 and 4 Specific Plan. The proposed project does not include any employment components and would not result in an increase of jobs in the area. Therefore, the proposed project would not substantially induce population growth through the provision of new housing units or employment.

6.1.2 Removal of Barrier to Growth

The proposed project would construct improvements and install new water and wastewater lines at the project site which would connect to existing utilities in the project area. The proposed project would result in the extension of water and sanitary sewer infrastructure down Mowry Avenue from the terminus of the existing water and sanitary sewer main located on the north side of the UPRR tracks. Additionally, the proposed project would extend the existing potable and non-potable water mains from the southwest corner of the Sanctuary West Development. The proposed project would result in the extension of urban infrastructure: however, the extension of the water and sanitary sewer mains would not result in a removal of barrier of growth as the project site is located near the terminus of Mowry Avenue and vacant developable lands near the project site are limited. Due to the project site's close proximity to Mowry Slough and the use of nearby lands as salt production ponds, available developable lands near the project site are extremely limited and the developable land located to the east of the project site is already planned for development through the Area 4 - Sanctuary West Project which would construct its own utility infrastructure expansion and would not rely on mains located along Mowry Avenue. Therefore, the extension of utility infrastructure for the proposed project would not cause other property owners to develop adjacent or nearby properties. Additionally, the off-site roadway improvements to Mowry Avenue would not result in a removal of barrier of growth as the proposed project would not extension of Mowry Avenue from its current terminus.. The additional demand for utilities and public services generated by



operation of the proposed project would be met with existing facilities. The proposed project would be constructed within the City's Urban Growth Boundary. Therefore, the proposed project would not result in significant growth-inducing impacts.

6.2 SIGNIFICANT AND UNAVOIDABLE IMPACTS

CEQA Guidelines Section 15126(b) requires an EIR to "describe any significant impacts, including those which can be mitigated but not reduced to a level of insignificance. Where there are impacts that cannot be alleviated without imposing an alternative design, their implications and the reasons why the project is being proposed, notwithstanding their effect, should be described."

Section 3.0, Environmental Impact Analysis, provides a description of the potential environmental impacts of the proposed project and recommends mitigation measures to reduce impacts to a less than significant level, where possible. Section 4.0, Cumulative Impacts, determines whether the incremental effects of this project are significant when viewed in connection with the effects of past projects, other current projects, and probable future projects. After implementation of the recommended mitigation measures, the following resource area would have significant unavoidable impacts:

6.2.1 Transportation

VMT

- The proposed project would conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b).
- The proposed project would result in a cumulatively considerable transportation impact related to VMT.

6.3 SIGNIFICANT IRREVERSIBLE ENVIRONMENTAL CHANGES

As mandated by the CEQA Guidelines, the EIR must address any significant irreversible environmental change that would result from implementation of the proposed project. Specifically, pursuant to the CEQA Guidelines (Section 15126.2(c)), such an impact would occur if:

- The project would involve a large commitment of nonrenewable resources;
- Land area committed to new project facilities;
- Irreversible damage can result from environmental accidents associated with the project; and
- The proposed consumption of resources is not justified (e.g., the project results in the wasteful use of energy).

The proposed project involves the construction of 203 single-family detached homes on an approximately 29 acre project site. The proposed project would also include the demolition of existing structures on-site and remediation of the site. Additional improvements would include on-street parking, drive aisles, underground utilities, LID drainage and water quality treatment areas, lighting, sidewalks, and



Draft Environmental Impact Report Other CEQA Considerations

landscaping. The proposed project would be consistent with the Low Density Residential General Plan land use designation by providing a density of 7 units per acre. The proposed project site is zoned Park and would require rezoning to the RS-6000 zoning designation.

As discussed in Section 3.6, Energy, project construction would require electricity, gasoline, and diesel fuels primarily for on-road and off-road equipment. However, equipment operation would comply with BAAQMD basic construction measures recommended for all projects that are aimed at reducing air pollution, such as minimizing idling of construction off-road equipment and maintaining all equipment in accordance with manufacturer standards. Such measures would also minimize the wasteful consumption of energy resources during construction.

The proposed project would be designed and constructed to be 100 percent electric. Operation of the proposed project would be required to comply with energy efficiency standards set forth by Title 24 of the California Administrative Code and the Applicable Efficiency Regulations. Title 24 requires that the proposed project meet a number of conservation standards, including installation of water-efficient fixtures and energy-efficient appliances. Title 24 also regulated energy consumption for the heating, cooling, ventilation, and lighting of residential and nonresidential buildings, and in enforced by the City. Compliance with Title 24 would ensure reduction in the use of fuel, water, and energy by the proposed project. Additionally, the proposed project would comply with CALGreen and the Newark Municipal Code requirements related to energy and water conservation. Therefore, the proposed project would not result in wasteful or unnecessary consumption of energy.

The proposed project would not result in irreversible damage from environmental accidents, such as an accidental spill of a hazardous material. As further discussed in Section 3.9, Hazards and Hazardous Materials, the proposed project would involve demolition and construction activities that could potentially release hazardous materials into the environment. As part of the City's NPDES Construction General Permit, the proposed project would be required to prepare and implement a SWPPP that would include BMPs to prevent accidental spills of hazardous materials during construction, as required by Mitigation Measure HYD-1 identified under Section 3.10, Hydrology and Water Quality, of this EIR. The proposed project would also implement Mitigation Measure HAZ-1, which would require remediation of existing hazardous materials contamination such as contaminated soil, soil gas, groundwater, and surface water found at the project site. Additionally, Mitigation Measure HAZ-1 also includes the requirement for an asbestos survey to be completed prior to the demolition of existing buildings on-site to determine if asbestos remediation is required. If ACBMs are determined to be present by the survey, the ACBMs will be removed prior to the demolition of buildings in accordance with NESHAP guidelines. The proposed project would comply with all applicable federal, state, and local laws related to the transport, use, or disposal of hazardous materials, as overseen by the Cal EPA and DTSC.

During operation of the proposed project, the use of hazardous materials would be limited to those commonly found at residential facilities such as solvents, cleaners, paints, and pesticides for landscape maintenance activities. These common household hazardous materials would be used in limited quantities and would not create a substantial hazard to the public or the environment. As such, the proposed project would not have the potential to cause serious environmental accidents.



6.4 MANDATORY FINDINGS OF SIGNIFICANCE

PRC Section 21083 requires lead agencies to make a finding of a "significant effect on the environment" if one or more of the following conditions exist:

- A proposed project has the potential to degrade the quality of environment, substantially reduce the habitat of a fish or wildlife species, cause a fish or wildlife species to drop below selfsustaining levels, threaten to eliminate a plant or animal community, reduce the number or restrict the range of a rare, or endangered plant or animal or eliminate important examples of the major periods of California history or prehistory.
- 2. The possible effects of a project are individually limited but cumulatively considerable.
- 3. The environmental effects of a project will cause substantial adverse effects on human beings, either directly or indirectly.

Finding No. 1: The proposed project would not have the potential to significantly affect biological, cultural, or tribal cultural resources.

As discussed in Section 3.4, Biological Resources, all project-related impacts on biological resources can be mitigated to a level of less than significant with the implementation of Mitigation Measures BIO-1 through BIO-9. This pertains to potential impacts on sensitive bird species including nesting native residents and migratory birds, special status bat species, invasive plant species, and potential conflicts with the City's tree ordinance. With the implementation of mitigation measures identified in the EIR, the proposed project would not have the potential to significantly affect biological resources. Additionally, as discussed in Section 3.5, Cultural Resource and Section 3.18, Tribal Cultural Resources, though the presence of cultural or tribal cultural resources on-site are low, ground disturbing activities may result in the excavation of previously undiscovered resources. With implementation of Mitigation Measures CUL-1 and CUL-2 which outlines requirements and procedures in the event that resources are discovered during construction, all project-related impacts to cultural and tribal cultural resources were mitigated to a level of less than significant. Therefore, with the implementation of mitigation measures identified in the EIR, the proposed project would not have the potential to significantly affect cultural or tribal cultural resources.

Finding No. 2: The proposed project would have cumulatively considerable impacts.

Projects considered in the cumulative analysis are located within the City and are described in Section 4.0, Cumulative Effects and are listed in Table 4.4-1. As discussed in Section 4.0, the proposed project, in conjunction with related projects, would not have a cumulatively considerable impact on any resource topics analyzed in this EIR with the exception of transportation. The proposed project would have cumulatively considerable impacts to VMT. Since the proposed project would exceed the Countywide VMT threshold and result in a significant and unavoidable impact, the proposed project would have a cumulatively considerable impact with respect to VMT and result in a significant and unavoidable impact. All other cumulative impacts resulting from resource topics analyzed in this EIR would be less than significant or the proposed project would result in a les than cumulatively considerable contribution to cumulative impacts.



Draft Environmental Impact Report Other CEQA Considerations

Finding No. 3: The proposed project would not cause substantial adverse effects on human beings.

The proposed project would not directly or indirectly cause substantial adverse effects in human beings. As identified in Table ES-1, impacts related to air quality, geology and soils, GHG, hazards and hazardous materials, hydrology and water quality, and noise would be reduced to a less than significant level with the implementation of applicable mitigation measures.

7.0 LIST OF PREPARERS

Report Preparers

Stantec Consulting Services, Inc.

Principal-in-Charge	Trevor Macenski
Project Manager/ Principal Planner	Anna Radonich
Deputy Project Manager/Environmental Planner	Kaela Johnson
Air Quality Specialist	Kaitlyn Heck
Environmental Planner	Jennifer Webster
Senior Associate, Acoustics	Tracie Ferguson
Principal, Transportation Planning & Traffic Engineering	Daryl Zerfass
Senior Transportation Planner	Maria Morris
Archaeologist	Jenna Santy
Senior Biologist	Melissa Tu

MOWRY VILLAGE PROJECT Draft Environmental Impact Report List of Preparers

This page intentionally left blank.



MOWRY VILLAGE PROJECT Draft Environmental Impact Report References

8.0 **REFERENCES**

Section 1.0: Introduction

None.

Section 2.0: Project Description

- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2014. Newark Areas 3 and 4 Specific Plan Project Recirculated Draft Environmental Impact Report. <u>https://www.newark.org/home/showpublisheddocument/4773/636863371395100000</u>. Accessed November 2021.
- Haley and Aldrich. 2019. Report on Phase I Environmental Site Assessment and Phase II Assessment Pick and Pull Parcel Newark, California. PDF. Accessed November 2021.
- United States Census Bureau (U.S. Census). 2022. QuickFacts, Newark city, California. <u>https://www.census.gov/quickfacts/fact/table/newarkcitycalifornia/HSD310221#HSD310221</u>. Accessed December 2022.

Section 3.1: Aesthetics

- California Department of Transportation (Caltrans). 2021. California State Scenic Highway System Map. <u>https://www.arcgis.com/apps/webappviewer/index.html?id=465dfd3d807c46cc8e8057116f1aacaa</u> . Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.

Section 3.2: Agricultural and Forestry Resources

- California Department of Conservation (DOC). 2021. California Important Farmland Finder. <u>https://maps.conservation.ca.gov/dlrp/ciff/</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.



Section 3.3: Air Quality

- Bay Area Air Quality Management District (BAAQMD). 2017a. California Environmental Quality Act Air Quality Guidelines. Available online at: <u>https://www.baaqmd.gov/~/media/files/planning-and-</u><u>research/ceqa/ceqa_guidelines_may2017-</u> pdf.pdf?la=en&rev=0d2d971e661d41f28a56953f1776bdde. Accessed December 9, 2022.
- Bay Area Air Quality Management District (BAAQMD). 2017b. Final 2017 Clean Air Plan. Available online at: <u>https://www.baaqmd.gov/~/media/files/planning-and-research/plans/2017-clean-air-plan/attachment-a_proposed-final-cap-vol-1-pdf.pdf?la=en</u>. Accessed December 9, 2022.
- Bay Area Air Quality Management District (BAAQMD). 2022a. Health Risk Screening and Modeling. Available online at: <u>https://www.baaqmd.gov/plans-and-climate/california-environmental-quality-act-ceqa/ceqa-tools/health-risk-screening-and-modeling</u>. Accessed December 8, 2022.
- Bay Area Air Quality Management District (BAAQMD). 2022b. Stationary Source Screening Map. Available online at: <u>https://baaqmd.maps.arcgis.com/apps/webappviewer/index.html?id=845658c19eae4594b9f4b80</u> 5fb9d89a3. Accessed December 8, 2022.
- California Air Resources Board (CARB). 2016. Ambient Air Quality Standard. Available online at: <u>https://ww2.arb.ca.gov/sites/default/files/2020-03/aaqs2_0.pdf</u>. Accessed December 9, 2022.
- California Air Resources Board (CARB). 2022. ADAM: Air Quality Data Statistics Hayward La Mesa Station. Available online at: <u>https://www.arb.ca.gov/adam/topfour/topfourdisplay.php</u>. Accessed December 9, 2022.
- City of Newark. 2013a. General Plan. Available online at: <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed December 9, 2022.
- Fehr & Peers. 2021a. Mowry Avenue Residential Development Draft Transportation Impact Analysis Report. September.
- USEPA. 2022. Criteria Air Pollutants. Available online at: <u>https://www.epa.gov/criteria-air-pollutants</u>. Accessed December 9, 2022.

Section 3.4: Biological Resources

Bolster, B.C., editor (Bolster ed). 1998. Terrestrial Mammal Species of Special Concern in California.
Draft Final Report prepared by P.V. Brylski, P.W. Collins, E.D. Pierson, W.E. Rainey and T.E.
Kucera. Report submitted to California Department of Fish and Game Wildlife Management
Division, Nongame Bird and Mammal Conservation Program for Contract No. FG3146WM.

California Invasive Plant Council (CAL-IPC). 2023. <u>https://www.cal-ipc.org/</u>. Accessed January 2022.



Draft Environmental Impact Report References

- California Department of Fish and Game (CDFG). 2012. Staff Report on Burrowing Owl Mitigation. March 2012.
- California Department of Fish and Wildlife (CDFW). 2021. California Natural Diversity Database for San Leandro, Hayward, Dublin, Redwood Point, Newark, Niles, Palo Alto, Mountain View, and Milpitas USGS 7.5-minute series quadrangles, Sacramento, CA. <u>https://wildlife.ca.gov/Data/CNDDB</u>. Accessed November 2021.
- California Department of Fish and Wildlife (CDFW). 2018. Protocols for Surveying and Evaluating Impacts to Special-Status Native Plant Populations and Natural Communities. Prepared for: State of California Natural Resources Agency. March 2018.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- California Native Plant Society (CNPS). 2021. Inventory of Rare Plants (online edition, v8-03 0.39). For: San Leandro, Hayward, Dublin, Redwood Point, Newark, Niles, Palo Alto, Mountain View, and Milpitas USGS 7.5-minute series quadrangles, Sacramento, CA. <u>https://www.cnps.org/rareplants/cnps-inventory-of-rare-plants</u>. Accessed November 2021.
- Chipman, E.D., N.E. McIntyre, R.E. Strauss, M.C. Wallace, J.D. Ray, and C.W. Boal (Chipman et al.). 2008. Effects of Human Land Use on Western Burrowing Owl Foraging and Activity Budgets. Journal of Raptor Research 42(2):87-98.
- Gervais, J. A., and Anthony, R. G (Gervais et al.). 2003. Chronic organochlorine contaminants, environmental variability, and demographics of a Burrowing Owl population. Ecological Applications 13:1250-1262.
- Haug, E. A., and Oliphant, L. W (Haug and Oliphant). 1990. Movements, activity patterns, and habitat use of Burrowing Owls in Saskatchewan. Journal of Wildlife Management 54:27-35.
- Helix Environmental Planning, Inc. (Helix). 2022. Mowry Village Project Biological Resources Technical Report. Prepared by Helix Environmental Planning, Inc. June 2022.
- ICF International (ICF). 2012. Biological effectiveness monitoring for the Natomas Basin Habitat Conservation Plan Area 2011 annual survey results (agency version).
- San Francisco Bay Area Wetlands Ecosystem Goals Project. 2015. The Baylands and Climate Change: What We Can Do. Baylands Ecosystem Habitat Goals Science Update 2015 prepared by the San Francisco Bay Area Wetlands Ecosystem Goals Project. California State Coastal Conservancy, Oakland, CA.
- Spencer, W.D., P. Beier, K. Penrod, K. Winters, C. Paulman, H. Rustigian-Romsos, J. Strittholt, M. Parisi, and A. Pettler (Spencer et al.). 2010. California Essential Habitat Connectivity Project: A Strategy



for Conserving a Connected California. Prepared for California Department of Transportation, California Department of Fish and Game, and Federal Highways Administration.

- U.S. Army Corps of Engineers (USACE). 1987. Corps of Engineers Wetlands Delineation Manual, Technical Report Y-87-1, U.S. Army Engineer Waterways Experiment Station, Vicksburg, MS.
- U.S. Fish and Wildlife Service (USFWS). 2010. Draft Recovery Plan for Tidal Marsh Ecosystems of Northern and Central California. January 2010. U.S. Fish and Wildlife Service, Sacramento, California.
- U.S. Fish and Wildlife Service (USFWS). 2013. Recovery Plan for Tidal Marsh Ecosystems of Northern and Central California. Sacramento, California.
- U.S. Fish and Wildlife Service (USFWS). 2021a. Information for Planning and Consultation (IPaC) for the Proposed Project. List of threatened and endangered species that may occur in and/or may be affected by [the] proposed project. U.S. Fish and Wildlife Service, Sacramento Fish and Wildlife Office, Sacramento, California. Consultation Code 08ESMF00-2019-SLI-2623. Dated November 8, 2021.
- U.S. Fish and Wildlife Service (USFWS).2021b. National Wetlands Inventory. http://www.fws.gov/wetlands/Data/mapper.html. Accessed January 2019.
- Zeiner, D.C., W.F. Laudenslayer, Jr., K.E. Mayer, and M. White, eds.1988-1990. California's Wildlife. Vol. I-III. California Department of Fish and Game, Sacramento, California.

Section 3.5: Cultural Resources

- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- Helix Environmental Planning (Helix). 2021. Mowry Village Cultural Resources Assessment, December 2021. PDF.

Netronline. 2023. Historical Aerials. https://www.historicaerials.com/viewer. Accessed April 2023.

Section 3.6: Energy

- City of Newark. 2013a. General Plan. Available online at: <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed December 9, 2022.
- Pacific Gas and Electric (PG&E). 2014. Electric Service Area Maps. Available online at: <u>https://www.pge.com/tariffs/assets/pdf/tariffbook/ELEC_MAPS_Service%20Area%20Map.pdf</u>. Accessed December 9, 2022.



- References
- Pacific Gas and Electric (PG&E). 2022. Delivering Low-Emission Energy. Available online at: <u>https://www.pge.com/en_US/about-pge/environment/what-we-are-doing/clean-energy-solutions.page</u>. Accessed December 9, 2022.
- Solar Industry. 2018. PG&E Meets California's Renewable Energy Goal Three Years Early. Available online at: <u>https://solarindustrymag.com/pge-meets-californias-renewable-energy-goal-three-years-early</u>. Accessed December 9, 2022.

Section 3.7: Geology and Soils

- Berlogar Stevens and Associates. 2021. Draft Design Level Geotechnical Investigation Proposed Residential Development Pick-n-Pull – 7400 Mowry Avenue Newark, California. PDF. Accessed November 2021.
- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- Yuri Jewett. 2023. Personal communication with San Francisco Bay Conservation and Development Commission Principal Waterfront Planner Yuri Jewett, July 17, 2023.

Section 3.8: Greenhouse Gas Emissions

- Association of Environmental Professionals (AEP). 2016. Final White Paper Beyond 2020 and Newhall. Available online at: <u>https://califaep.org/docs/AEP-2016_Final_White_Paper.pdf</u>. Accessed December 2022.
- Bay Area Air Quality Management District (BAAQMD). 2017. California Environmental Quality Act Air Quality Guidelines. Available online at: <u>https://www.baaqmd.gov/~/media/files/planning-and-</u><u>research/ceqa/ceqa_guidelines_may2017-</u> pdf.pdf?la=en&rev=0d2d971e661d41f28a56953f1776bdde. Accessed December 2022.
- Bay Area Air Quality Management District (BAAQMD). 2019. Regulation 6 Particulate Matter and Visible Emissions Rule 3 Wood-Burning Devices. November 20, 2019. <u>https://www.baaqmd.gov/~/media/dotgov/files/rules/regulation-6-rule-</u> <u>3/documents/20191120_r0603_final-pdf.pdf?la=en</u>. Accessed December 2022.
- California Air Resources Board (CARB). 2017. Final Scoping Plan. Available online at: <u>https://ww2.arb.ca.gov/sites/default/files/classic/cc/scopingplan/scoping_plan_2017.pdf</u>. Accessed December 2022.



- California Air Resources Board (CARB). 2022a. 2022 Scoping Plan for Achieving Carbon Neutrality. Available online at: <u>https://ww2.arb.ca.gov/sites/default/files/2022-11/2022-sp.pdf</u>. Accessed December 2022.
- California Air Resources Board (CARB). 2022b. Current California GHG Emission Inventory Data. Available online at: <u>https://ww2.arb.ca.gov/ghg-inventory-data</u>. Accessed December 2022.
- California Air Resources Board (CARB). 2022c. SB 375 Regional Plan Climate Targets. Available online at: <u>https://ww2.arb.ca.gov/our-work/programs/sustainable-communities-program/regional-plantargets#:~:text=SB%20375%20Regional%20Plan%20Climate%20Targets.%20Senate%20Bill.of %20statewide%20public%20health%20and%20air%20quality%20objectives. Accessed December 2022.</u>
- City of Newark. 2010. Climate Action Plan. Available online at: <u>https://www.newark.org/home/showpublisheddocument/328/636502463456000000</u>. Accessed December 2022.

Section 3.9: Hazards and Hazardous Materials

- California Department of Forestry and Fire Protection (CAL FIRE). 2007. Fire Hazard Severity Zones in SRA Alameda County. <u>https://osfm.fire.ca.gov/media/7271/fhszs_map1.pdf</u>. Accessed November 2021.
- California Department of Forestry and Fire Protection (CAL FIRE). 2008. Very High Fire Hazard Severity Zones in LRA Alameda County. <u>https://osfm.fire.ca.gov/media/6638/fhszl_map1.pdf. Accessed November 2021</u>.
- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- Department of Toxic Substances Control (DTSC). 2021. EnviroStor Database. https://www.envirostor.dtsc.ca.gov/public/. Accessed November 2021.
- State Water Resources Control Board (SWRCB). 2021. GeoTracker Database. <u>https://geotracker.waterboards.ca.gov/</u>. Accessed November 2021.
- Haley and Aldrich. 2019. Report on Phase I Environmental Site Assessment and Phase II Assessment Pick and Pull Parcel Newark, California. PDF. Accessed November 2021.



Section 3.10: Hydrology and Water Quality

- Adapting to Rising Tides (ART). 2023. <u>https://explorer.adaptingtorisingtides.org/explorer</u>. Accessed July 2023.
- Berlogar Stevens, and Associates. 2019. Due Diligence Geotechnical Assessment Proposed Residential Development Pick-n-Pull – 7400 Mowry Avenue Newark, California. PDF. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- City of Newark. 2015b. Newark Areas 3 and 4 Specific Plan Project Recirculated Final Environmental Impact Report. <u>https://www.newark.org/home/showpublisheddocument/182/636502333562270000</u>. Accessed November 2021.
- Federal Emergency Management Agency (FEMA). 2021. National Flood Hazard Layer FIRMette Map 06001C0444G. <u>https://msc.fema.gov/portal/firmette?latitude=37.511065&longitude=-122.011988</u>. Accessed November 2021.
- Yuri Jewett. 2023. Personal communication with San Francisco Bay Conservation and Development Commission Principal Waterfront Planner Yuri Jewett, July 17, 2023.

Section 3.11: Land Use and Planning

- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.

Section 3.12: Mineral Resources

California Department of Conservation (DOC). 1996. California Geologic Survey Mineral Land Classification. Revised Mineral Land Classification Map – Newark Quadrangle. <u>https://maps.conservation.ca.gov/cgs/informationwarehouse/index.html?map=mlc</u>. Accessed November 2021.



City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.

City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.

Section 3.13: Noise

- California Department of Transportation (Caltrans). 2013. Technical Noise Supplement Traffic Noise Analysis Protocol, September 2013 (<u>https://dot.ca.gov/-/media/dot-</u> <u>media/programs/environmental-analysis/documents/env/tens-sep2013-a11y.pdf</u>. Accessed December 2022.
- California Department of Transportation (Caltrans). 2020. Transportation and Construction Vibration Guidance Manual, April 2020, <u>https://dot.ca.gov/-/media/dot-media/programs/environmental-analysis/documents/env/tcvgm-apr2020-a11y.pdf.</u> Accessed December 2022.
- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2014. Newark Areas 3 & 4 Specific Plan Draft Recirculated EIR. <u>https://www.newark.org/home/showpublisheddocument/4773/636863371395100000</u>. Accessed December 2022.

City of Newark. 2022. City of Newark Municipal Code. <u>https://library.municode.com/ca/newark/codes/code_of_ordinances?nodeld=CD_ORD_TIT17ZO_</u> <u>DIVIIICIRE_CH17.24PEST_17.24.100NO</u>. Accessed December 2022.

Environmental Protection Agency (EPA). 1978. Region 10 Environmental Impact Statement Guidelines, April 1973.

https://nepis.epa.gov/Exe/ZyNET.exe/2000RZBX.TXT?ZyActionD=ZyDocument&Client=EPA&Ind ex=Prior+to+1976&Docs=&Query=&Time=&EndTime=&SearchMethod=1&TocRestrict=n&Toc=& TocEntry=&QField=&QFieldYear=&QFieldMonth=&QFieldDay=&IntQFieldOp=0&ExtQFieldOp=0 &XmlQuery=&File=D%3A%5Czyfiles%5CIndex%20Data%5C70thru75%5CTxt%5C0000002%5 C2000RZBX.txt&User=ANONYMOUS&Password=anonymous&SortMethod=h%7C-&MaximumDocuments=1&FuzzyDegree=0&ImageQuality=r75g8/r75g8/x150y150g16/i425&Displ ay=hpfr&DefSeekPage=x&SearchBack=ZyActionL&Back=ZyActionS&BackDesc=Results%20pag e&MaximumPages=1&ZyEntry=1&SeekPage=x&ZyPURL. Accessed December 2022.

- Federal Highway Administration. 2006. Construction Noise Handbook. <u>https://www.fhwa.dot.gov/environment/noise/construction_noise/handbook/handbook02.cfm</u>. Accessed December 2022.
- Federal Highway Administration. Construction Noise Handbook, 2017 <u>https://www.fhwa.dot.gov/environment/noise/construction_noise/handbook/handbook02.cfm</u>. Accessed December 2022.



Draft Environmental Impact Report References

- Federal Transit Administration (FTA). 2018. Transit Noise and Vibration Impact Assessment Manual, September 2018, <u>https://www.transit.dot.gov/sites/fta.dot.gov/files/docs/research-innovation/118131/transit-noise-and-vibration-impact-assessment-manual-fta-report-no-0123_0.pdf</u>. Accessed December 2022.
- Santa Clara County. 2018. Comprehensive Land Use Plan Santa Clara County Moffett Federal Airfield. <u>https://stgenpln.blob.core.windows.net/document/ALUC_NUQ_CLUP.pdf</u>. Accessed December 2022.
- San Francisco International Airport. 2015. 2019 Noise Exposure Map, August 13, 2015. <u>https://www.flysfo.com/sites/default/files/media/sfo/noise-abatement/sfo_p150_2019-nem-36x24-plot-signed_ada.pdf</u>. Accessed December 2022.
- Port of Oakland. 2006. Oakland International Airport Master Plan, March 2006. Community Noise Equivalent Level (CNEL) Contours 2004 and 2010. <u>https://www.oaklandairport.com/wpcontent/uploads/2016/05/2010_CNEL_v_2004.pdf</u>. Accessed December 2022.
- San Jose International Airport. 2020. 2037 CNEL Contours Airport Master Plan (amended 4/28/20). Date: 7/28/2020. <u>https://www.flysanjose.com/sites/default/files/noise/2037_CNEL.pdf</u>. Accessed December 2022.

Section 3.14: Population and Housing

- Association of Bay Area Governments (ABAG). 2013. Regional Housing Needs Plan San Francisco Bay Area 2015-2023, adopted July 18, 2023. <u>https://abag.ca.gov/sites/default/files/2015-</u> <u>23 rhna plan.pdf</u>. Accessed November 2021.
- Association of Bay Area Governments (ABAG). 2018. Plan Bay Area Projections 2040, November 2018. <u>https://mtc.ca.gov/sites/default/files/Projections_2040-ABAG-MTC-web.pdf</u>. Accessed November 2021.
- City of Newark. 2015a. City of Newark Housing Element Update 2015. <u>https://www.newark.org/home/showpublisheddocument/74/636502245497230000</u>. Accessed November 2021.
- Employment Development Department (EDD). 2022. Monthly Labor Force Data for Cities and Census Designated Places December 2021 – Preliminary. <u>https://www.labormarketinfo.edd.ca.gov/data/unemployment-and-labor-force.html</u>. Accessed February 2022.
- Department of Finance (DOF). 2007. E-8 Historical Population and Housing Estimates for Cities, Counties and the State, 1990 2000. <u>https://dof.ca.gov/Forecasting/Demographics/Estimates/E-8/</u>. Accessed November 2021.



Draft Environmental Impact Report References

- Department of Finance (DOF). 2012. E-8 Historical Population and Housing Estimates for Cities, Counties and the State, 2000 2010. <u>https://dof.ca.gov/Forecasting/Demographics/Estimates/E-8/2000-10/</u>. Accessed November 2021.
- Department of Finance (DOF). 2022a. E-5 Population and Housing Estimates for Cities, Counties, and the State, 2010 2020 with 2010 Census Benchmark. <u>https://dof.ca.gov/forecasting/demographics/estimates/</u>. Accessed May 2022.
- Department of Finance (DOF). 2022b. E-5 Population and Housing Estimates for Cities, Counties, and the State, January 2021-2022, with 2020 Benchmark. <u>https://dof.ca.gov/Forecasting/Demographics/Estimates/estimates-e5-2010-2021/</u>. Accessed May 2022.
- United States Census Bureau (US Census). 2022. QuickFacts, Newark city, California. <u>https://www.census.gov/quickfacts/fact/table/newarkcitycalifornia/HSD310221#HSD310221</u>. Accessed December 2022.

Section 3.15: Public Services

- Alameda County Fire Department (ACFD). 2021. Response and Activity Statistics Fiscal Year 2019-20. <u>https://fire.acgov.org/AboutUs/stats.page</u>?. Accessed November 2021.
- California Department of Education. 2022. Data Quest. <u>https://dq.cde.ca.gov/dataquest/page2.asp?level=School&subject=Enrollment&submit1=Submit</u>. Accessed May 2022.
- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- City of Newark. 2021. Newark California Police Department About Newark Police. <u>https://www.newark.org/departments/police/office-of-the-chief-of-police/about-newark-police</u>. Accessed November 2021.
- Newark Unified School District (NUSD). 2020. District-Wide Facilities Master Plan Fall 2019. <u>https://www.newarkunified.org/uploaded/Business_Dept/Financial_Reports/Facilities_Master_Plan_n/District-Wide_Facilities_Master_Plan.pdf</u>. Accessed November 2021.
- Newark Unified School District (NUSD). 2021. Newark USD SchoolSite Locator. http://apps.schoolsitelocator.com/?districtCode=43094#. Accessed November 2021.



Section 3.16: Recreation

References

- City of Newark. 2013a. Newark California General Plan. <u>https://www.newark.org/home/showpublisheddocument/76/636502245500200000</u>. Accessed November 2021.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.

Section 3.17: Transportation

- City of Newark. 2013a. Newark California General Plan. Adopted December 12, 2013. https://www.newark.org/home/showpublisheddocument?id=70. Accessed November 2021.
- City of Newark. 2017. Pedestrian & Bicycle Master Plan. February. <u>https://www.newark.org/home/showpublisheddocument?id=1985</u>. Accessed November 2021.
- Fehr & Peers. 2021a. Mowry Avenue Residential Development Draft Transportation Impact Analysis Report. September. PDF.
- Fehr & Peers. 2021b. Mowry Village Residential Development Pedestrian Crossing Improvements Midblock on Mowry Avenue in Newark, California. October 15. PDF.
- State of California Governor's Office of Planning and Research. 2018. Technical Advisory on Evaluating Transportation Impacts in CEQA.

Section 3.18: Tribal Cultural Resources

Helix Environmental Planning (Helix). 2021. Mowry Village Cultural Resources Assessment, December 2021. PDF.

Section 3.19: Utilities and Service Systems

- Alameda County Water District (ACWD). 2021a. 2020 2025 Urban Water Management Plan. <u>https://www.acwd.org/DocumentCenter/View/3816/Final-2020-2025-UWMP</u>. Accessed November 2021.
- Alameda County Water District (ACWD). 2021b. ACWD Fact Sheet. <u>https://www.acwd.org/93/Fact-Sheet. Accessed November 2021</u>.
- California Department of Resources Recycling and Recovery (CalRecycle). 2021a. Disposal Rate Calculator Newark 2020.

https://www2.calrecycle.ca.gov/LGCentral/AnnualReporting/DisposalRateCalculator. Accessed November 2021.



- California Department of Resources Recycling and Recovery (CalRecycle). 2021b. SWIS Facility/Site Activity Details – Altamont Landfill and Resource Recovery (01-AA-0009). <u>https://www2.calrecycle.ca.gov/SolidWaste/SiteActivity/Details/7?siteID=7</u>. Accessed November 2021.
- City of Newark. 2008. Water Supply Assessment for Newark Areas 3 & 4 Specific Plan EIR Project November 2008.

https://www.newark.org/home/showpublisheddocument/196/636502338427870000. Accessed April 2023.

- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- City of Newark. 2014. Newark Areas 3 and 4 Specific Plan Project Recirculated Draft Environmental Impact Report. <u>https://www.newark.org/home/showpublisheddocument/4773/636863371395100000</u>. Accessed November 2021.
- City of Newark. 2015b. Newark Areas 3 and 4 Specific Plan Project Recirculated Final Environmental Impact Report. <u>https://www.newark.org/home/showpublisheddocument/182/636502333562270000</u>. Accessed November 2021.
- Union Sanitary District (USD). 2021. Mission, Organization, Facts, and History. <u>https://www.unionsanitary.com/about-us/about-us/mission-facts-history</u>. Accessed November 2021.

Section 3.20: Wildfire

- California Department of Forestry and Fire Protection (CAL FIRE). 2007. Fire Hazard Severity Zones in SRA Alameda County. <u>https://osfm.fire.ca.gov/media/7271/fhszs_map1.pdf</u>. Accessed November 2021.
- California Department of Forestry and Fire Protection (CAL FIRE). 2008. Very High Fire Hazard Severity Zones in LRA Alameda County. <u>https://osfm.fire.ca.gov/media/6638/fhszl_map1.pdf</u>. Accessed November 2021.
- California Department of Forestry and Fire Protection (CAL FIRE). 2022. State Responsibility Area Fire Hazard Severity Zones – Alameda County. <u>https://osfm.fire.ca.gov/media/1yelle2d/fhsz_county_sra_11x17_2022_alameda_ada.pdf</u>. Accessed January 2023.
- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.


U.S. Forest Service (USFS). 2020. Wildfire Hazard Potential.

https://usfs.maps.arcgis.com/home/webmap/viewer.html?useExisting=1&layers=55226e8547f84a ae8965210a9801c357. Accessed November 2021.

Section 4.0: Cumulative Effects

- City of Newark. 2013b. General Plan Tune Up Draft Program EIR for the City of Newark. <u>https://www.newark.org/home/showpublisheddocument/78/636502245503300000</u>. Accessed November 2021.
- City of Newark. 2021. Newark California Police Department About Newark Police. <u>https://www.newark.org/departments/police/office-of-the-chief-of-police/about-newark-police</u>. Accessed November 2021.

Section 5.0: Alternatives to the Proposed Project

None.

Section 6.0: Other CEQA Considerations

None.